Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273 Pg. 1 of Docket #1273 Date Filed: 01/25/2024 CSD 1001A [07/01/18] Name, Address, Telephone No. & I.D. No. Samuel R. Maizel (SBN 189301) Tania M. Moyron (SBN 235736) Order Entered on Rebecca M. Wicks (SBN 313608) January 25, 2024 **DENTONS US LLP** by Clerk U.S. Bankruptcy Court Southern District of California 601 South Figueroa Street, Suite 2500 Los Angeles, California 90017-5704 Telephone: 213 623 9300

UNITED STATES BANKRUPTCY COURT

SOUTHERN DISTRICT OF CALIFORNIA 325 West F Street, San Diego, California 92101-6991

In Re

BORREGO COMMUNITY HEALTH FOUNDATION

BANKRUPTCY NO. 22-02384-LT11

Date of Hearing: January 17, 2024 Time of Hearing: 10:00 a.m. Name of Judge: Laura S. Taylor

Debtor.

ORDER ON

First Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation [Docket No. 1168]

The court orders as set forth on the continuation pages attached and numbered 2 through 14 with exhibits, if any, for a total of 160 pages. Motion/Application Docket Entry No. 1168.

// // // // // // //

DATED:

January 25, 2024

Judge, United States Bankruptcy Court

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273 Pg. 2 of

CSD 1001A [07/01/18] (Page 2)

160

ORDER ON First Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation [Docket No. 1168]

DEBTOR: BORREGO COMMUNITY HEALTH FOUNDATION CASE NO: 22-02384-LT11

Order Confirming Borrego Plan

Upon consideration of the (I) final approval of the disclosures ("Disclosures") contained in the First Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation [Docket No. 1168], attached hereto as **Exhibit A** (together with all exhibits thereto, and as may be amended, modified, or supplemented,1 the "Plan"),2 jointly proposed by the above-captioned debtor and debtor in possession (the "Debtor") and the Official Committee of Unsecured Creditors (the "Committee" and together with the Debtor, the "Plan Proponents") and (II) confirmation of the Plan; the Disclosures having been approved on an interim bases pursuant to that certain Order on Joint Motion of the Debtor and the Official Committee of Unsecured Creditors for an Entry of an Order (I) Granting Interim Approval of the Adequacy of Disclosures in the Combined Joint Disclosure Statement and Plan: (II) Approving Solicitation Packages and Procedures; (III) Approving the Forms of Ballots; (IV) Setting Related Deadlines and (V) Granting Related Relief [Docket No. 1179] (the "Solicitation Order") granting the Joint Motion of the Debtor and the Official Committee of Unsecured Creditors for an Entry of an Order (I) Granting Interim Approval of the Adequacy of Disclosures in the Combined Joint Disclosure Statement and Plan; (II) Approving Solicitation Packages and Procedures; (III) Approving the Forms of Ballots; (IV) Setting Related Deadlines and (V) Granting Related Relief [Docket No. 1093] (the "Solicitation" Motion"); upon the declarations of service filed reflecting compliance with the notice and solicitation requirements of the Solicitation Order [Docket No. 1230] (the "Notice Declaration"); and as evidenced by the Notice Declaration, the applicable notices having been sent to the applicable creditors and interest holders entitled to notice of the Confirmation Hearing and/or non-voting status (the "Confirmation Notices"); upon the Certification of Sydney Reitzel Regarding the Solicitation and Tabulation of Votes on the First Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation [Docket No. 1243] (the "Voting Declaration"); and upon the Joint Memorandum of Law in Support of Confirmation of the First Amended Joint Combined Disclosure Statement and Plan of Liquidation of Borrego Community Health Foundation and Omnibus Reply to the Objections to Confirmation [Docket No. 1242] (the "Confirmation Memorandum"); and any and all objections to the Plan³ or related relief having been resolved and/or overruled by this Court pursuant to this Confirmation Order; and a hearing to consider Confirmation having been held on January 17, 2024 (the "Confirmation Hearing"); and upon the evidence adduced and proffered and the arguments of counsel made at the Confirmation Hearing; and this Court having reviewed all documents in connection with Confirmation and having heard all parties desiring to be heard; and upon the record of the Chapter 11 Case; and after due deliberation and consideration of all of the foregoing; and sufficient cause appearing therefore; this Court hereby makes the following:

Findings of Fact and Conclusions of Law

A. Findings of Fact and Conclusions of Law. The findings and conclusions set forth herein and in the Solicitation Order, together with the findings of fact and conclusions of law set forth in the record of the Confirmation Hearing, constitute this Court's findings of fact and conclusions of law pursuant to Rule 52 of the Federal Rules of Civil Procedure, made applicable to these proceedings pursuant to Bankruptcy Rules 7052 and 9014. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent that any of the following conclusions of law constitute findings of fact, they are adopted as such.

B. Capitalized Terms. Capitalized terms used herein, but not defined herein, shall have the respective meanings attributed to such terms in the Plan.

¹ On December 11, 2023, the Debtor filed the Notice of Plan Supplement to the First Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation [Docket No. 1182], which was amended on January 15, 2024 [Docket No. 1247] (as amended, the "Plan Supplement").

² Capitalized terms not defined herein shall have the same meaning ascribed in the Plan.

³ The following objections to confirmation of the Plan were filed: (i) Acting United States Trustee's Objection to Confirmation of the First Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation [Docket No. 1219] (the "UST Objection"); and (ii) Oracle America, Inc.'s Cure Objection and Reservation of Rights Regarding Debtor's Amended Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation [Docket No. 1232] (the "Oracle Objection").

- C. **Jurisdiction and Venue.** This Court has jurisdiction over the Chapter 11 Case pursuant to 28 U.S.C. §§ 157 and 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2), and this Court may enter a final order consistent with Article III of the United States Constitution, and the Plan Proponents consent to entry of this Confirmation Order under the Local Rules and Article III of the United States Constitution. Venue of this proceeding and the Chapter 11 Case is proper in this district and in this Court pursuant to 28 U.S.C. §§ 1408 and 1409.
- D. Chapter 11 Petition. On September 12, 2022 (the "Petition Date"), the Debtor filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code. Since the commencement of the Chapter 11 Case, the Debtor has been operating its business as a debtor in possession pursuant to §§ 1107 and 1108. No party has requested the appointment of a trustee or examiner in the Chapter 11 Case. The Committee has been appointed. [Docket No. 49].
- E. **Judicial Notice.** This Court takes judicial notice of the docket in the Chapter 11 Case, including the dockets of any adversary proceedings initiated in the Chapter 11 Case, maintained by the Clerk of this Court and/or its duly appointed agent, including, without limitation, all pleadings, notices, and other documents filed, all orders entered, and all evidence and arguments made, proffered or adduced at the hearings held before this Court during the Chapter 11 Case, including, without limitation, the Confirmation Hearing.
- F. **Disclosures.** The Disclosures within the Plan provides Holders of Claims entitled to vote on the Plan with adequate information to make an informed decision as to whether to vote to accept or reject the Plan in accordance with § 1125.
- G. **Plan Supplement.** Prior to the Confirmation Hearing, the Debtor filed the Plan Supplement. The Plan Supplement complies with the terms of the Plan, and the filing and notice of the Plan Supplement was appropriate and complied with the requirements of the Bankruptcy Code and the Bankruptcy Rules, and no other or further notice is or shall be required. The Plan Proponents are authorized to modify the Plan Supplement documents following entry of this Confirmation Order in a manner consistent with this Confirmation Order and/or the Plan.
- H. **Mailing of Solicitation and Confirmation Materials.** As is evidenced by the Voting Declaration and the Notice Declaration, the transmittal and service of the Plan, the Ballots and the Confirmation Hearing Notices were adequate and sufficient under the circumstances, and all parties required to be given notice of the Plan and the Confirmation Hearing (including the deadline for filing and serving objections to Confirmation of the Plan and final approval of the Disclosures) have been given due, proper, timely, and adequate notice thereof in accordance with the Solicitation Order and in compliance with the Bankruptcy Code, the Bankruptcy Rules, the Local Rules, and applicable non-bankruptcy law, and such parties have had an opportunity to appear and be heard with respect thereto. No other or further notice of the Plan and the Confirmation Hearing is required.
- I. **Voting.** The procedures by which the Ballots for acceptance or rejection of the Plan and Release Opt-Out Election Forms were distributed and tabulated under the circumstances of the Chapter 11 Case were fair, properly conducted, and complied with the Bankruptcy Code, the Bankruptcy Rules, the Local Rules, applicable non-bankruptcy law and the Solicitation Order.
- J. Bankruptcy Rule 3016(a). In accordance with Bankruptcy Rule 3016(a), the Plan is dated and identifies the Debtor and the Committee as the Plan Proponents.
- K. Plan Compliance with Bankruptcy Code (11 U.S.C.§ 1129(a)(1)). As set forth below, the Plan complies with all the applicable provisions of the Bankruptcy Code, thereby satisfying § 1129(a)(1).
- L. Proper Classification (11 U.S.C. §§ 1122, 1123(a)(1)). The classification of Claims under the Plan is proper under the Bankruptcy Code. In addition to Administrative Claims, Priority Tax Claims, Statutory Fees, and Professional Fee Claims, which need not be classified, the Plan designates four Classes of Claims. The Claims placed in each Class are substantially similar to other Claims, as the case may be, in each such Class. Valid business, factual, and legal reasons exist for separately classifying the various Classes of Claims created under the Plan, and such Classes do not unfairly discriminate between Holders of Claims. Thus, the Plan satisfies §§ 1122 and 1123(a)(1).
- M. Specification of Unimpaired Classes (11 U.S.C. §1123(a)(2)). Sections 9-10 of the Plan specifies that Class 1 (Priority Non-Tax Claims) and Class 2 (Other Secured Claims) are Unimpaired under the Plan. Thus, § 1123(a)(2) is satisfied.
- N. Specification of Treatment of Impaired Classes (11 U.S.C. § 1123(a)(3)). Sections 9-10 of the Plan designates Class 3 (General Unsecured Claims) and Class 4 (Allowed DHCS Claim) as Impaired and specifies the treatment of Claims in such Classes. Thus § 1123(a)(3) is satisfied.
- O. **No Discrimination (11 U.S.C. § 1123(a)(4)).** The Plan provides for the same treatment for each Claim in each respective Class unless the Holder of a particular Claim has agreed to a less favorable treatment of such Claim. Thus, CSD 1001A

§ 1123(a)(4) is satisfied.

- P. Implementation of the Plan (11 U.S.C. § 1123(a)(5)). The Plan and the Liquidating Trust Agreement provide adequate and proper means for the Plan's implementation. Thus, § 1123(a)(5) is satisfied.
- Q. Non-Voting Equity Securities (11 U.S.C. § 1123(a)(6)). The Plan does not provide for the issuance of any securities, including non-voting securities, and the Debtor is being dissolved as soon as practicable after the CHOW is approved for DAP Health. Therefore, § 1123(a)(6) is satisfied.
- R. Selection of Officers and Directors (11 U.S.C. § 1123(a)(7)). Section 15.5 of the Plan provides for the appointment of a three-member Post-Effective Date Board of Directors. The initial members of the Post-Effective Date Board of Directors are designated in the Plan Supplement. [Docket No. 1182]. Section 15.7 of the Plan provides that the Liquidating Trust shall be governed by the Liquidating Trustee and the Co-Liquidating Trustee. On the Effective Date, the Liquidating Trustee shall be Ankura, and Isaac Lee shall be acting in the capacity as the Liquidating Trustee. The Liquidating Trustee shall be the President of the Post-Effective Date Debtor. On the Effective Date, Scott Rinaldi of Ankura will be designated as Secretary and Treasurer of the Post-Effective Date Debtor. On the Effective Date, the Co-Liquidating Trustee shall be FTI, and Narendra Ganti shall be acting in the capacity as Co-Liquidating Trustee. In the event the Liquidating Trustee or Co-Liquidating Trustee is removed by this Court, resigns, or is replaced pursuant to the Liquidating Trust Agreement, or otherwise vacates its position, a successor Liquidating Trustee or Co-Liquidating Trustee, as applicable, shall be appointed as set forth in the Liquidating Trust Agreement. Therefore, § 1123(a)(7) is satisfied.
- S. Additional Plan Provisions (11 U.S.C. § 1123(b)). The Plan's provisions are appropriate, in the best interests of the Debtor and their Estate, and consistent with the applicable provisions of the Bankruptcy Code, Bankruptcy Rules, and Local Rules.
- T. Executory Contracts and Unexpired Leases (11 U.S.C. § 1123(b)(2)). The Debtor has exercised reasonable business judgment in determining to either assume, assume and assign, or reject each of the Debtor's remaining Executory Contracts and Unexpired Leases as provided for in the Plan, the Plan Supplement, and this Confirmation Order, and any such assumptions and rejections are justified and appropriate in the Chapter 11 Case. The Debtor's exercise of reasonable business judgment in determining to either assume, assume and assign, or reject each of the Debtor's remaining Executory Contracts and Unexpired Leases is justified and appropriate in the Chapter 11 Case for the reasons set forth in the Plan, Plan Supplement, and the Confirmation Memorandum.
- U. Releases, Injunctions, and Exculpations Under and in Connection with the Plan (11 U.S.C. § 1123(b)(3)). The Plan provides for, among other things, the Debtor's Release, the Third Party Release, Injunctions, and Exculpations. Accordingly, the Plan is consistent with § 1123(b).
- a. **Debtor Release.** The Debtor Release set forth in Section 17.2(a) of the Plan is in accordance with § 1123(b) and represents a valid exercise of the Debtor's business judgment under Bankruptcy Rule 9019. The Debtor Release is fair and equitable, is a key component of the Plan, and otherwise constitutes a settlement of Claims and Causes of Action under § 1123(b) and Bankruptcy Rule 9019. Releases of the Released Parties by the Debtor and the estate are critically important to the success of the Plan. The Plan reflects the settlement and resolution of several complex issues, and the Debtor Release is an integral part of the consideration to be provided in exchange for the compromises and resolutions embodied in the Plan. Entry of the Confirmation Order shall constitute this Court's approval, pursuant to § 1123(b) and Bankruptcy Rule 9019, of the Debtor Release, which includes by reference each of the related provisions and definitions contained in the Plan, and further, shall constitute this Court's finding that the Debtor Release is: (a) in exchange for the good and valuable consideration provided by the Released Parties, including, without limitation, the Released Parties' contributions to facilitating the restructuring and implementing the Plan; (b) a good faith settlement and compromise of the Claims and Causes of Action released by the Debtor Release; (c) in the best interests of the Debtor, the Estate and all Holders of Claims; (d) fair, equitable, and reasonable; (e) given and made after due notice and opportunity for hearing; and (f) a bar to any of the Debtor, the Liquidating Trust, or the Debtor's Estate asserting any Claim or Cause of Action released pursuant to the Debtor Release.
- b. **Third Party Release.** The Third Party Release set forth in Section 17.2(b) of the Plan is an essential provision of the Plan. The Confirmation Hearing Notices, Ballots, and Voting Instructions explicitly and conspicuously stated that potential Releasing Parties could opt out of the Third Party Release and included the language of the Third Party Release. Thus, each Releasing Party was given due and adequate notice that they would be granting the Third Party Release by failing to opt out of the Third Party Release prior to the deadline to do so. Further, as set forth in the Confirmation Memorandum, this Confirmation Order shall confirm that section 17.2(b) of the Plan is amended to provide that the Third Party Release is granted only by Creditors who (i) voted to accept the Plan (or were deemed to accept the Plan); and (ii) did not return a Release Opt-Out Election Form. Accordingly, the Third Party Release is consensual. The

Confirmation Order shall constitute this Court's approval, pursuant to § 1123 and Bankruptcy Rule 9019, of the Third Party Release, which includes by reference each of the related provisions and definitions contained in the Plan, and, further, shall constitute this Court's finding that the Third Party Release is: (a) consensual; (b) essential to the Confirmation of the Plan, important to the overall objectives of the Plan and an essential means of implementing the Plan pursuant to § 1123(a)(5); (c) given in exchange for the good and valuable consideration provided by the Released Parties, including, without limitation, the Released Parties' contributions to facilitating the restructuring and implementing the Plan; (d) a good faith settlement and compromise of the Claims released by the Third Party Release; (e) materially beneficial to, and in the best interests of the Debtor and the Estate and stakeholders; (f) fair, equitable, and reasonable; (g) given and made after due notice and opportunity for hearing; (h) a bar to any of the Releasing Parties asserting any Claim or Cause of Action released pursuant to the Third Party Release; (i) within the jurisdiction of this Court under 28 U.S.C. §§ 1334(a), 1334(b), and 1334(d); and (j) consistent with §§ 105, 1123, 1129, and 1141 and other applicable provisions of the Bankruptcy Code.

- c. **Injunction.** The injunction provisions set forth in Section 17.3 of the Plan are essential to the Plan and are necessary to implement the Plan and to preserve and enforce the Debtor Release, the Third Party Release, and the exculpation provisions in the Plan. As set forth in the Confirmation Memorandum, this Confirmation Order shall provide that, under section 17.3(a) of the Plan, the injunction is in place for as long as the Plan is effective.
- d. **Exculpation.** The exculpation provisions set forth in Section 17.4 of the Plan are essential to the Plan. The record in the Chapter 11 Case fully supports the exculpation provisions set forth in Section 17.4 of the Plan.
- e. **No Recourse.** To resolve the UST Objection, as set forth in the Confirmation Memorandum, section 17.5 is amended to confirm that notwithstanding any language in the Plan, the Liquidating Trustee and the Co-Liquidating Trustee are fiduciaries to the Liquidating Trust.
- V. **Debtor's Compliance with Bankruptcy Code (11 U.S.C. § 1129(a)(2)).** Pursuant to § 1129(a)(2), the Plan Proponents have complied with the applicable provisions of the Bankruptcy Code, including, without limitation, §§ 1122, 1123, 1124, 1125, and 1126, the Bankruptcy Rules, the Local Rules, and the Solicitation Order governing notice, disclosure, and solicitation in connection with the Plan, the Plan Supplement, and all other matters considered by this Court in connection with the Chapter 11 Case.
- W. Plan Proposed in Good Faith and Not by Means Forbidden by Law (11 U.S.C. § 1129(a)(3)). The Plan Proponents have proposed the Plan in good faith and not by any means forbidden by law, thereby satisfying § 1129(a)(3). In determining that the Plan has been proposed in good faith, this Court has examined the totality of the circumstances surrounding the filing of the Chapter 11 Case, the Plan itself, and the process leading to its formulation. The Plan is the product of months of extensive arm's-length independent and interrelated negotiations among the Debtor, the Committee, and DHCS with respect to the DHCS Settlement, which terms are incorporated into the Plan. The Plan promotes the objectives and purposes of the Bankruptcy Code.
- X. Payments for Services or Costs and Expenses (11 U.S.C. § 1129(a)(4)). The procedures set forth in the Plan for this Court's approval of the fees, costs, and expenses to be paid in connection with the Chapter 11 Case, or in connection with the Plan and incident to the Chapter 11 Case, satisfy the objectives of, and are in compliance with, § 1129(a)(4).
- Y. **Directors, Officers, and Insiders (11 U.S.C. § 1129(a)(5)).** On or before the Effective Date, the Post-Effective Date Board of Directors shall be established. The initial Liquidating Trustee and Co-Liquidating Trustee are identified in Section 15.7 of the Plan. The initial members of the Post-Effective Date Board of Directors shall be: (i) Jenna LeComte-Hinley, PhD; (ii) Frank Figueroa; and (iii) Martha Deichler. The Liquidating Trustee, Isaac Lee, shall be the President of the Post-Effective Date Debtor, and Scott Rinaldi shall be the Secretary and Treasurer of the Post-Effective Date Debtor. Thus, the Plan satisfies § 1129(a)(5).
- Z. No Rate Changes (11 U.S.C. § 1129(a)(6)). After the Confirmation Date, the Debtor will not have any businesses involving the establishment of rates over which any regulatory commission has or will have jurisdiction. Therefore, the provisions of § 1128(a)(6) do not apply to the Plan.
- AA. **Best Interests of Creditors (11 U.S.C. § 1129(a)(7)).** The "best interests" test is satisfied as to all Impaired Classes under the Plan, as each Holder of a Claim in such Impaired Classes will receive or retain property of a value, as of the Effective Date, that is not less than the amount that such Holder would so receive or retain if the Debtor was liquidated under chapter 7 of the Bankruptcy Code.
- BB. Acceptance by Certain Classes (11 U.S.C. § 1129(a)(8)). Class 1 (Priority Non-Tax Claims) and Class 2 (Secured Claims) are left unimpaired under the Plan. Class 3 (General Unsecured Claims) and Class 4 (Allowed DHCS

Claim) have voted to accept the Plan in accordance with the Bankruptcy Code. Therefore, § 1129(a)(8) is satisfied.

- CC. Treatment of Administrative Claims, Professional Fee Claims, Priority Tax Claims, and Statutory Fees (11 U.S.C. § 1129(a)(9)). The treatment of Administrative Claims, Priority Tax Claims, Statutory Fees, and Professional Fee Claims, and other Priority Claims pursuant to Sections 8 and 10 of the Plan satisfies § 1129(a)(9).
- DD. Acceptance by Impaired Classes (11 U.S.C. § 1129(a)(10)). Class 3 (General Unsecured Claims) and Class 4 (Allowed DHCS Claims) are Impaired Classes of Claims that voted to accept the Plan, determined without including any acceptance of the Plan by any insider. Therefore, § 1129(a)(10) is satisfied.
- EE. **Feasibility (11 U.S.C. § 1129(a)(11)).** The Plan provides for the dissolution of the Debtor as soon as practicable after the liquidation of the Debtor's property and the approval of the CHOW. Thus, § 1129(a)(11) is satisfied.
- FF. Payment of Fees (11 U.S.C. 1129(a)(12)). All fees payable under 28 U.S.C. § 1930 have been paid or shall be paid no later than thirty (30) days after the Effective Date pursuant to the Plan, thus satisfying § 1129(a)(12).
- GG. **Miscellaneous Provisions (11 U.S.C. §§ 1129(a)(13)-(15)).** Sections 1129(a)(13)-(15) are inapplicable to Confirmation, as the Debtor (i) is not obligated to pay retiree benefits (as defined in § 1114(a)) (§ 1129(a)(13)), (ii) has no domestic support obligations (§ 1129(a)(14)), and (iii) is not an individual (§1129(a)(15)).
- HH. **Transfers of Non-Profit Property (11 U.S.C. § 1129(a)(16)).** The Plan provides for this Court's approval of, or otherwise authorizes, any property transfers. Therefore, the Plan satisfies the requirements of § 1129(a)(16).
- II. **Principal Purpose (11 U.S.C. § 1129(d)).** The principal purpose of the Plan is neither the avoidance of taxes, nor the avoidance of the application of section 5 of the Securities Act of 1933, and no governmental unit has objected to Confirmation on any such grounds. Accordingly, § 1129(d) is inapplicable.
- JJ. **Satisfaction of Confirmation Requirements.** Based upon the foregoing, the Plan satisfies the requirements for Confirmation set forth in § 1129 and should be confirmed.
- KK. **Good Faith Solicitation (11 U.S.C. § 1125(e)).** The Plan Proponents and their officers, directors, employees, advisors, Professionals, and agents have acted in good faith within the meaning of § 1125(e), and in compliance with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the Local Rules, and the Solicitation Order in connection with all of their respective activities relating to the solicitation of acceptances of the Plan and their participation in the activities described in § 1125, and they are entitled to the protections afforded by § 1125(e) and the injunction and exculpation provisions set forth in Section 17 of the Plan and in this Confirmation Order.
- LL. **Implementation.** All documents and agreements necessary to implement transactions contemplated by the Plan, including those contained or summarized in the Plan Supplement (including, among others, the Liquidating Trust Agreement), have been negotiated in good faith and at arms'-length, are in the best interests of the Debtor and its Estate, and shall, upon completion of documentation and execution, be valid, binding, and enforceable documents and agreements not in conflict with any federal, state, or local law. The Debtor, the Liquidating Trustee, and the Co-Liquidating Trustee, as applicable, are authorized to take any action reasonably necessary or appropriate to consummate such agreements and the transactions contemplated thereby.
- MM. **Retention of Jurisdiction.** This Court may properly retain jurisdiction over the matters set forth in Section 19 of the Plan and/or § 1142.

Based on the foregoing findings, and upon the record made before this Court at the Confirmation hearing, and good and sufficient cause appearing therefore, it is hereby ORDERED ADJUDGED AND DECREED THAT:

Confirmation of the Plan

- 1. The Plan, as and to the extent modified by this Confirmation Order, is approved and confirmed pursuant to § 1129. Any objections to the Plan not otherwise withdrawn, resolved, or otherwise disposed of are overruled and denied.
- 2. The terms of the Plan are incorporated by reference into (except to the extent modified by this Confirmation Order), and are an integral part of, this Confirmation Order.

Final Approval of the Disclosures

3. The Disclosures are approved as adequate on a final basis pursuant to § 1125.

Compromises and Settlements Under the Plan

4. Pursuant to § 1123 and Bankruptcy Rule 9019, upon the Effective Date, all settlements and compromises set forth in the Plan, or announced on the record at the Confirmation Hearing are approved in all respects, and constitute good faith compromises and settlements.

Classification and Treatment

5. The Plan's classification scheme is approved. The classifications set forth on the Ballots (a) were set forth on the Ballots solely for purposes of voting to accept or reject the Plan, (b) do not necessarily represent, and in no event shall be deemed to modify or otherwise affect, the actual classification of such Claims under the Plan for distribution purposes, (c) may not be relied upon by any Holder as representing the actual classification of such Claim under the Plan for distribution purposes, and (d) shall not be binding on the Debtor, the Liquidating Trustee, and the Co-Liquidating Trustee except for Plan voting purposes.

Authorization to Implement the Plan

- 6. The Debtor, the Liquidating Trustee, and the Co-Liquidating Trustee, as applicable, are authorized to take or cause to be taken all corporate actions necessary or appropriate to implement all provisions of, and to consummate, the Plan, and to execute, enter into, or otherwise make effective all documents arising in connection therewith, including, without limitation, all Plan Documents, prior to, on, and after the Effective Date.
- 7. On the Effective Date, the officers of the Debtor, the Liquidating Trustee, and the Co-Liquidating Trustee are authorized to do all things and to execute and deliver all agreements, documents, instruments, notices, and certificates as are contemplated by the Plan and to take all necessary actions required in connection therewith, in the name of and on behalf of the Debtor. In accordance with § 1142(b), upon the entry of this Confirmation Order, the Debtor, the Liquidating Trustee, and the Co-Liquidating trustee, each acting by and through its respective officers and agents, are authorized to take any and all actions necessary or appropriate to implement the Plan, including, without limitation, (i) forming the Liquidating Trust, entering into the Liquidating Trust Agreement (substantially in the form attached hereto as **Exhibit B**). and complying with, and satisfying the obligations set forth under, the Liquidating Trust Agreement; and (ii) complying with, and satisfying the obligations set forth under, the Plan, in each case, without any further order of this Court. The Liquidating Trust shall be deemed for all purposes to have been created in connection with the Plan and this Confirmation Order. On the Effective Date, each of the Debtor's then-current directors and officers shall be deemed to have resigned and the Liquidating Trustee, Isaac Lee, shall automatically be the President of the Post-Effective Date Debtor, Scott Rinaldi, shall automatically be the Secretary and Treasurer of the Post-Effective Date Debtor, and the Post-Effective Date Board of Directors shall be formed without the need for any meeting of or resolution by the Post-Effective Date Board of Directors. Following the Effective Date, the Liquidating Trustee is given full power of attorney and has authority to execute and/or endorse any documentation on behalf of the Debtor in furtherance of the Plan and the Debtor's liquidation.
- 8. The approvals and authorizations specifically set forth in this Confirmation Order are not intended to limit the authority of the Liquidating Trustee, the Co-Liquidating Trustee, the Debtor, or any officer thereof to take any and all actions necessary or appropriate to implement, effectuate, and consummate any and all documents or transactions contemplated by the Plan or this Confirmation Order.
- 9. For the avoidance of doubt, the Plan and this Confirmation Order in no way authorize or approve any incentive payments under §§ 105 or 503.

Enforceability of the Plan

- 10. Pursuant to §§ 1123(a), 1141(a) and 1142, the Plan and all Plan Documents (including, but not limited to, the Liquidating Trust Agreement) shall be, and hereby are, valid, binding and enforceable.
- 11. Except to the extent otherwise provided in this Confirmation Order and the Plan and pursuant to Section 15 of the Plan, on the Effective Date, the Debtor shall irrevocably transfer and shall be deemed to have irrevocably transferred to the Liquidating Trust all of its rights, title, and interest in and to all of the Liquidating Trust Assets, then held by the Debtor free and clear of all liens, claims, and encumbrances.

Preservation of Causes of Action and Litigation

12. Except as otherwise provided in the Plan or this Confirmation Order (a) in accordance with § 1123(b)(3) and Section 17.7 of the Plan, any retained Causes of Action or Litigation that the Debtor may hold against any Entity shall vest upon the Effective Date in the Liquidating Trust and (b) after the Effective Date, the Liquidating Trustee shall have the exclusive right, authority, and discretion to determine and to initiate, file, prosecute, enforce, abandon, settle,

compromise, release, withdraw, or litigate to judgment any such retained Causes of Action and Litigation, in accordance with the terms of the Plan and the Liquidating Trust Agreement and without further order of this Court, in any court or other tribunal, including, without limitation, in an adversary proceeding filed in the Chapter 11 Case.

Reservation of Causes of Action

13. Unless a Cause of Action or Litigation against a Holder or other Entity is expressly waived, relinquished, released, compromised, or settled in the Plan or any Final Order (including this Confirmation Order), including pursuant to the Debtor Release or the Third Party Release, the Debtor and the Liquidating Trustee expressly reserve such Cause of Action or Litigation for later adjudication by the Liquidating Trustee, including, without limitation, Causes of Action or Litigation not specifically identified or described in the Plan Supplement or elsewhere or of which the Debtor may presently be unaware or which may arise or exist by reason of additional facts or circumstances unknown to the Debtor at this time or facts or circumstances which may change or be different from those the Debtor now believes to exist. No preclusion doctrine, including, without limitation, the doctrines of res judicata, collateral estoppel, issue preclusion, claim preclusion, waiver, estoppel (judicial, equitable or otherwise) or laches shall apply to such Causes of Action or Litigation upon, after, or as a consequence of the Confirmation or consummation or after the entry of this Confirmation Order or Effective Date based on the Plan, or this Confirmation Order, except where such Causes of Action or Litigation have been released in the Plan or any other Final Order. The Debtor and Liquidating Trustee's rights to pursue or adopt any claims alleged in any lawsuit in which the Debtor is a defendant or an interested party, against any Person or Entity, including, without limitation, the plaintiffs or co-defendants in such lawsuits, is expressly reserved.

Wind-Down and Dissolution of the Debtor

14. The Debtor shall be dissolved as provided in the Plan and this Confirmation Order. On the Effective Date, or as soon as practicable thereafter, the Liquidating Trustee and Co-Liquidating Trustee, as applicable, will administer the Class A Liquidating Trust Assets and Class B Liquidating Trust Assets and continue the wind-down and liquidation of the Debtor after the Effective Date. As set forth in the Plan, the Liquidating Trustee will cause the Post-Effective Date Debtor to be dissolved for all purposes after (i) the CHOW is approved, and (ii) the receipt of all payments related to Medi-Cal and Medicare. The Liquidating Trustee may dissolve the Post-Effective Date Debtor, earlier than as set forth herein, if it determines that the continued existence of the Post-Effective Date Debtor is not necessary to satisfy the foregoing conditions. The Liquidating Trustee shall wind-down the affairs of the Debtor, if any, subject to the Plan, the Confirmation Order and the Liquidating Trust Agreement, and the Liquidating Trustee shall prepare and file (or cause to be prepared and filed) on behalf of the Debtor, all tax returns, reports, certificates, forms, or similar statements or documents (collectively, "Tax Returns") required to be filed or that the Liquidating Trustee otherwise deems appropriate, including the filing of amended Tax Returns or requests for refunds, for all taxable periods ending on, prior to, or after the Effective Date. In connection therewith, subsequent to the approval of the CHOW, the Debtor and the Liquidating Trustee are authorized to surrender to the applicable governmental authorities all licenses that the Debtor used to operate their business.

Cancellation of Interests, Existing Securities, and Agreements

15. On the Effective Date, except to the extent otherwise provided in the Plan or in this Confirmation Order, all notes, instruments, certificates, and other documents evidencing Claims, shall be deemed cancelled and surrendered without any need for a Holder of a Claim to take further action with respect thereto and the obligations of the Debtor thereunder or in any way related thereto shall be deemed satisfied in full and discharged; *provided*, *however*, that notwithstanding Confirmation or consummation, any such agreement that governs the rights of the Holder of a Claim shall continue in effect solely for purposes of allowing Holders to receive distributions under the Plan.

Method of Distribution Under the Plan

- 16. The Liquidating Trustee and Co-Liquidating Trustee, as applicable, shall make all distributions required to be made to Holders of Allowed Claims pursuant to the Plan and the Liquidating Trust Agreement. The Liquidating Trust shall hold and distribute the Liquidating Trust Assets, the Administrative Claims Reserve, the Effective Date Professional Claim Reserves, and the Disputed Claim Reserve in accordance with the provisions of the Plan and the Liquidating Trust Agreement.
- 17. The Liquidating Trustee and Co-Liquidating Trustee, as applicable, shall make all distributions required to be made to such Holders of Allowed Claims pursuant to the Plan and the Liquidating Trust Agreement. The Liquidating Trustee and Co-Liquidating Trustee shall not be required to give any bond or surety or other security for the performance of the Liquidating Trustee's and Co-Liquidating Trustee's duties as disbursing agent unless otherwise ordered by this Court. At the option of the Liquidating Trustee or Co-Liquidating Trustee, as applicable, any Cash payment may be made from the Liquidating Trust by any means deemed appropriate by the Liquidating Trustee or Co-Liquidating Trustee, as

CSD 1001A

applicable, including by check or wire transfer, or as otherwise required or provided in applicable agreements.

The Liquidating Trust

- 18. **Administration of the Liquidating Trust.** The Liquidating Trust Agreement, substantially in the form attached hereto as **Exhibit B**, is hereby approved. The appointment of Ankura as the Liquidating Trustee with Isaac Lee acting in the capacity of the initial Liquidating Trustee, the appointment of FTI as the Co-Liquidating Trustee with Narendra Ganti acting in the capacity of the initial Co-Liquidating Trustee, and the retention of the professionals by the Liquidating Trust, on the terms set forth in the Plan and Liquidating Trust Agreement, is hereby approved. The Liquidating Trustee and Co-Liquidating Trustee shall be compensated in the manner set forth in and consistent with the Liquidating Trust Agreement. The Liquidating Trustee and the Co-Liquidating Trustee shall both have all powers, rights, duties, and protections afforded the Liquidating Trustee and the Co-Liquidating Trustee, respectively, under the Plan, this Confirmation Order, and the Liquidating Trust Agreement.
- 19. Interests in the Liquidating Trust. There shall be two classes of interests in the Liquidating Trust. The Liquidating Trust shall issue the Class A Liquidating Trust Interests to the Holders of Class 3 Claims and shall issue the Class B Liquidating Trust Interests to the Holder of Class 4 Claims in accordance with Sections 10.3 and 10.4 of the Plan. Trust Beneficial Interests shall be uncertificated. Holders of Trust Beneficial Interests shall be bound by the Liquidating Trust Agreement.
- 20. **Liquidating Trust Assets.** The Liquidating Trust shall consist of the Liquidating Trust Assets. On the Effective Date, the Liquidating Trust Assets shall vest in the Liquidating Trust free and clear of all liens, claims, and encumbrances, except to the extent otherwise provided in the Plan, including, without limitation, pursuant to Section 15.6 of the Plan or in this Confirmation Order. The transfer of the Liquidating Trust Assets to the Liquidating Trust shall not affect any attorney-client privilege, the work-product privilege, and any other applicable evidentiary privileges of the Debtor, which such privileges shall be expressly transferred and assumed by the Liquidating Trust.
- 21. **Distribution of Liquidating Trust Assets.** The Liquidating Trustee and Co-Liquidating Trustee, as applicable, shall distribute the appropriate proceeds of the Liquidating Trust Assets to the Holders of the Trust Beneficial Interests pursuant to the terms of the Plan and the Liquidating Trust Agreement.
- 22. On the Effective Date, or as soon thereafter as is practical and subject to the prior payment of the amounts required to be paid or reserved by the Post-Effective Date Debtor in cash on the Effective Date on account of Claims pursuant to this Plan, the Post-Effective Date Debtor shall establish and fund deposit accounts to serve as Operating Accounts for the Post-Effective Date Debtor for use in accordance with the Wind-down Budget, provided, however, the Post-Effective Date Debtor may, with the written consent of the Liquidating Trustee, utilize previously established deposit accounts for such purpose. The Liquidating Trustee shall be authorized to use the funds in the Operating Accounts as needed to preserve, administer, and continue the Operations for the Post-Effective Date Debtor, including paying all related costs and expenses associated, and collection of any amounts due to the Post-Effective Date Debtor under the Transition Services Agreement, each in accordance with the Wind-down Budget. After the Effective Date, all Cash or other proceeds generated by the Purchased Assets solely to the extent required to fund the Operating Accounts in accordance with the Wind-down Budget during the Transition Period shall be excluded from the definition of the Remaining Cash. Subject to payment in full of all Allowed Administrative Claims, and except as otherwise ordered by this Court, expenses incurred by the Liquidating Trust on or after the Effective Date shall be paid in accordance with the Liquidating Trust Agreement without further order of this Court.

Executory Contracts and Unexpired Leases

- 23. The assumption of the executory contracts or unexpired leases identified on the schedule filed with the Plan Supplement (the "Schedule of Assumed Contracts"), including without limitation, any Insurance Policies being assumed, is approved.
- 24. Except any executory contract or unexpired lease (a) identified on the Schedule of Assumed Contracts including without limitation, any assumed Insurance Policies, (b) which is the subject of a separate motion or notice to assume or reject filed by the Debtor and pending as of the Confirmation Hearing, (c) that previously expired or terminated pursuant to its own terms, or (d) that was previously assumed or rejected by any of the Debtor by Final Order, each of the Debtor's other executory contracts and unexpired leases shall be deemed rejected as of the Effective Date in accordance with the provisions and requirements of §§ 365 and 1123 and such rejection is approved.
- 25. Each executory contract and unexpired lease assumed pursuant to the Plan, this Confirmation Order, or other Order of this Court, and not assigned to a third party on or prior to the Effective Date, shall be transferred to the Liquidating Trust and be deemed a Liquidating Trust Asset.

CSD 1001A

26. Unless otherwise provided by a subsequent order of this Court, any Proofs of Claim arising from the rejection of executory contracts or unexpired leases ("Rejection Claims") pursuant to Confirmation of the Plan or otherwise must be filed with the Claims and Balloting Agent no later than the later of 30 days after the Effective Date or 30 days after the effective date of rejection. Rejection Claims shall be classified as Class 3 General Unsecured Claims and shall be treated in accordance with Section 10.3 of the Plan, as applicable. Any Rejection Claims that are not timely filed pursuant to Section 14.2 of the Plan or this Confirmation Order shall be forever disallowed and barred.

Disputed Claims

- 27. Except as otherwise provided in the Plan, the Liquidating Trustee and Co-Liquidating Trustee, as applicable, shall have the right to object to all Disputed Claims on any basis. Subject to further extension by this Court with or without notice, the Liquidating Trustee and Co-Liquidating Trustee, as applicable, may object to the allowance of all Disputed Claims on or before the Claim Objection Deadline. From and after the Effective Date, the Liquidating Trust shall succeed to all the rights, defenses, offsets, and counterclaims of the Debtor and the Estate in respect of all Claims, and in that capacity shall have the exclusive power to prosecute, defend, compromise, settle, and otherwise deal with all such objections.
- 28. On or as soon as practicable after the Effective Date, the Liquidating Trustee shall establish the Disputed Claim Reserves an amount equal to the asserted Disputed Claims. Such amounts shall be deposited in interest bearing accounts, which maximize value and maintain safety. The interest bearing accounts shall be invested in US 1 Month Treasury Bills or other US backed instruments. To the extent a Disputed Claim becomes an Allowed Claim, Holders of such Claims shall be entitled to the interest that accrues on the pro rata amount of their Claim.
- 29. All objections to Claims shall be Filed by the Claims Objection Deadline or such later date as this Court may establish upon a motion by the Liquidating Trustee in accordance with the Plan; <u>provided</u>, that no such objection may be filed with respect to any Claim or Interest after a Final Order has been entered Allowing such Claim or Interest.

Administrative Claims

- 30. On or as soon as practicable after the Effective Date, the Liquidating Trustee shall establish the Administrative Claims Reserve in the amount of approximately \$2 million in accordance with Section 20.2 of the Plan.
- 31. All requests for payment of an Administrative Claim (other than a Professional Fee Claim) arising or deemed to have arisen after September 12, 2022, must be filed with this Court, and served to the Debtor, the Liquidating Trustee, and the Co-Liquidating Trustee on or before the date that is 90 days after the Effective Date. In the event of an objection to allowance of an Administrative Claim, this Court shall determine the Allowed amount of such Administrative Claim. Nothing herein shall be deemed to extend existing deadlines established for filing certain Administrative Claims, including pursuant to the Bar Date Order.

Effective Date Professional Claim Reserves and Professional Fee Claims

- 32. On or as soon as practicable after the Effective Date, Liquidating Trustee shall establish the Effective Date Professional Claim Reserves and fund such reserve in an amount equal to the estimated Allowed Professional Fee Claims. The Effective Date Professional Claim Reserves shall be used solely for the payment of Allowed Professional Fee Claims in accordance with Section 8.2 of the Plan.
- 33. All final applications for payment of Professional Fee Claims must be Filed with this Court and served on the Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, and the U.S. Trustee on or before 45 days after the Effective Date or such later date as may be agreed to by the Liquidating Trustee. Upon approval of the fee applications by this Court, the Liquidating Trustee shall pay Professionals from the Effective Date Professional Claim Reserves all their respective Allowed Professional Fee Claims.

Release, Injunction, Exculpation and Related Provisions

- 34. The release, injunction, exculpation, and related provisions set forth in Section 17 of the Plan and as modified in this Confirmation Order are hereby approved and authorized in their entirety, and such provisions are effective and binding on all Persons and Entities as and to the extent provided for therein.
- 35. **Debtor Release.** Notwithstanding anything in the Plan to the contrary, pursuant to § 1123(b), and except as otherwise specifically provided in the Plan, for good and valuable consideration, on and after and subject to the occurrence of the Effective Date, the Debtor and its estate shall release each Released Party, and each Released Party CSD 1001A

is deemed released by the Debtor and the estate from any and all claims, obligations, rights, suits, damages, Causes of Action, remedies, and liabilities whatsoever, including any derivative claims, asserted or assertable on behalf of the Debtor or its estate, as applicable, whether known or unknown, foreseen or unforeseen, asserted or unasserted, accrued or unaccrued, matured or unmatured, determined or indeterminable, disputed or undisputed, liquidated or unliquidated, or due or to become due, existing or hereinafter arising, in law, equity, or otherwise, that the Debtor or the estate would have been legally entitled to assert in its own right, or on behalf of the Holder of any Claim or other entity, based on or relating to, or in any manner arising from, in whole or in part, the Debtor, the Debtor's liquidation, the Chapter 11 Case, the purchase, sale, transfer of any security, asset, right, or interest of the Debtor, the DAP Sale, the subject matter of, or the transactions or events giving rise to, any Claim that is treated in the Plan, the business or contractual arrangements between the Debtor and any Released Party, the treatment of Claims prior to or in the Chapter 11 Case, the negotiation, formulation, or preparation of the Plan or related agreements, instruments, or other documents, any other act or omission. transaction, agreement, event, or other occurrence taking place on and before the Petition Date, other than claims or liabilities arising out of or relating to any act or omission of a Released Party that constitutes fraud, willful misconduct, or gross negligence; provided, that, the foregoing Debtor Release shall not operate to waive or release any obligations of any party under the Plan or any other document, instrument, or agreement executed to implement the Plan. For avoidance of doubt, the foregoing Debtor Release does not release any of the Debtor's claims, obligations, rights, suits, damages, Causes of Action, remedies, and liabilities with respect thereto.

Entry of this Confirmation Order shall constitute this Court's approval, of the Debtor Release, which includes by reference each of the related provisions and definitions contained in the Plan, and further, shall constitute this Court's finding that the Debtor Release is: (a) in exchange for the good and valuable consideration provided by the Released Parties; (b) a good faith settlement and compromise of the Claims released by the Debtor Release; (c) in the best interests of the Debtor and all Holders of Claims; (d) fair, equitable and reasonable; (e) given and made after due notice and opportunity for hearing; and (f) a bar to the Debtor or its estate asserting any Claim or Cause of Action released pursuant to the Debtor Release.

36. Third Party Release. Notwithstanding anything in the Plan to the contrary, on, and as of, the Effective Date and for good and valuable consideration, the receipt and sufficiency of which are acknowledged, the Released Parties shall be forever released (the "Third Party Release") from any and all claims, obligations, actions, suits, rights, debts, accounts, causes of action, remedies, avoidance actions, agreements, promises, damages, judgments, demands, defenses, and liabilities throughout the world under any law or court ruling through the Effective Date (including all claims based on or arising out of factors or circumstances that existed as of or prior to the Effective Date, including claims based on negligence or strict liability, and further including any derivative claims asserted on behalf of the Debtor, whether known or unknown, foreseen or unforeseen, existing or hereinafter arising, in law, equity, or otherwise) which the Debtor, its estate, Creditors, or other persons receiving or who are entitled to receive distributions under the Plan may have against any of them in any way related to this Chapter 11 Case, the negotiation, formulation, or preparation of the Plan or related agreements, instruments, or other documents, any other act or omission, transaction, agreement, event, or other occurrence taking place on and before the Petition Date, and related to the Debtor (or its predecessors), its business and/or its assets; provided, however, that the foregoing releases are granted only by Creditors who (i) voted to accept the Plan (or were deemed to accept the Plan); and (ii) did not return a Release Opt-Out Election Form; provided, however, that the release provided in this section shall not apply to (A) any Creditor whose Claim is not Allowed either in whole or in part; or (B) any Creditor in category (b) above if the Solicitation Package or Release Opt-Out Election Form was returned to the Debtor as undelivered and that such Creditor did not otherwise submit a Ballot; and provided further, however, that the release provided in this Section shall not extend to any claims by any Governmental Unit with respect to criminal liability under applicable law, willful misconduct or bad faith under applicable law, or ultra vires acts under applicable law.

Entry of this Confirmation Order shall constitute this Court's approval of the Third Party Release, which includes by reference each of the related provisions and definitions contained in the Plan, and further, shall constitute this Court's finding that the Third Party Release is: (a) in exchange for the good and valuable consideration provided by the Released Parties; (b) in the best interests of the Debtor and all Holders of Claims; (c) fair, equitable, and reasonable; (d) given and made after due notice and opportunity for hearing; and (e) a bar to any of the Releasing parties asserting any Claim or Cause of Action released pursuant to the Third Party Release.

37. **General Injunction**. In implementation of the Plan, except as otherwise expressly provided in the Confirmation Order or the Plan, all Persons who have held, currently hold or may hold a Claim against the Debtor are permanently enjoined on and after the Effective Date from taking any action in furtherance of such Claim or any other Cause of Action released and discharged under the Plan, including, without limitation, the following actions against any Released Party: (a) commencing, conducting or continuing in any manner, directly or indirectly, any action or other proceeding with respect to a Claim; (b) enforcing, levying, attaching, collecting or otherwise recovering in any manner or by any means, whether directly or indirectly, any judgment, award, decree or order with respect to a Claim; (c) creating, perfecting or enforcing in any manner, directly or indirectly, any lien or encumbrance of any kind with respect to a Claim; (d) asserting CSD 1001A

any setoff, right of subrogation or recoupment of any kind, directly or indirectly, against any debt, liability or obligation due to the Debtor, the Post-Effective Date Debtor or the Liquidating Trust with respect to a Claim; or (e) commencing, conducting or continuing any proceeding that does not conform to or comply with or is contradictory to the provisions of this Plan; provided, however, that nothing in this injunction shall preclude the Holders of Claims against the Debtor from enforcing any obligations of the Debtor, the Post-Effective Date Debtor, the Liquidating Trust, the Liquidating Trustee, or Co-Liquidating Trustee under the Plan and the contracts, instruments, releases and other agreements delivered in connection herewith, including, without limitation, this Confirmation Order, or any other order of this Court in the Chapter 11 Case. By accepting a Distribution made pursuant to this Plan, each Holder of an Allowed Claim shall be deemed to have specifically consented to the injunctions set forth in this Paragraph. For the avoidance of doubt, the injunction set forth in this paragraph is in place for as long as the Plan is effective.

- 38. Other Injunctions. In implementation of the Plan, except as otherwise expressly provided in the Confirmation Order or the Plan, the Post-Effective Date Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, the Post-Effective Date Board of Directors, or the Liquidating Trust and their respective members, directors, officers, agents, attorneys, advisors or employees shall not be liable for actions taken or omitted in its or their capacity as, or on behalf of, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, or the Liquidating Trust (as applicable), except those acts found by Final Order to arise out of its or their willful misconduct, gross negligence, fraud, and/or criminal conduct, and each shall be entitled to indemnification and reimbursement for fees and expenses in defending any and all of its or their actions or inactions in its or their capacity as, or on behalf of the Post-Effective Date Board of Directors, the Post-Effective Date Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, or the Liquidating Trust (as applicable), except for any actions or inactions found by Final Order to involve willful misconduct, gross negligence, fraud, and/or criminal conduct. Any indemnification claim of the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, and the other parties entitled to indemnification under this subsection shall be satisfied from the Liquidating Trust Assets. The parties subject to this paragraph shall be entitled to rely, in good faith, on the advice of retained professionals, if any.
- 39. Exculpation. To the maximum extent permitted by applicable law, each Exculpated Party shall not have or incur any liability for any act or omission in connection with, related to, or arising out of the Chapter 11 Case (including, without limitation, the filing of the Chapter 11 Case), the marketing of the Debtor's assets and the DAP Sale, the Plan and any related documents (including, without limitation, the negotiation and consummation of the Plan, the pursuit of the Effective Date, the administration of the Plan, or the property to be distributed under the Plan), or each Exculpated Party's exercise or discharge of any powers and duties set forth in the Plan, except with respect to the actions found by Final Order to constitute willful misconduct, gross negligence, fraud, or criminal conduct, and, in all respects, each Exculpated Party shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan. Without limitation of the foregoing, each such Exculpated Party shall be released and exculpated from any and all Causes of Action that any Person is entitled to assert in his/her/their own right or on behalf of any other Person, based in whole or in part upon any act or omission, transaction, agreement, event, or other occurrence in any way relating to the subject matter of this paragraph.

Entry of this Confirmation Order shall constitute this Court's approval of the Exculpation, which includes by reference each of the related provisions and definitions contained in the Plan, and further, shall constitute this Court's finding that the Exculpation is: (a) in exchange for the good and valuable consideration provided by the Exculpated Parties; (b) in the best interests of the Debtor and all Holders of Claims; (c) fair, equitable, and reasonable; (d) given and made after due notice and opportunity for hearing; and (e) a bar to any party from asserting any Claim or Cause of Action released pursuant to the Exculpation.

40. No Recourse. If a Claim is Allowed in an amount for which after application of the payment priorities established by this Plan (including, without limitation, in Sections 8 and 10 of the Plan) there is insufficient value to provide a recovery equal to that received by other Holders of Allowed Claims in the respective Class, no Claim Holder shall have recourse for any such deficiency against any of the Released Parties, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, or the Liquidating Trust. However, except as specifically stated otherwise in this Plan, nothing in this Plan shall modify any right of a Holder of a Claim under § 502(j). The obligations under the Plan of the Debtor's Estate shall (i) be contractual only and shall not create any fiduciary relationship and (ii) be obligations of the Debtor's Estate only and no individual acting on behalf of the Debtor, the Committee, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, or otherwise, shall have any personal or direct liability for these obligations. For the avoidance of doubt, notwithstanding anything to the contrary in the Plan or this Confirmation Order, any obligations under the Plan of the Debtor's Estate are contractual only. Further, for the avoidance of further doubt, the Ligidating Trustee and the Co-Liquidating Trustee are fiduciaries.

Resolution of Formal Objections to the Plan

- 41. United States Trustee. To resolve the UST Objection, the Plan Proponents have, among other things, provided language in paragraphs 36, 37, and 40 above to address the US Trustee's concerns. Such language amends sections 17.2(b), 17.3(a), and 17.5 of the Plan. Further, Section 3.80 of the Plan is amended to provide that "Initial Distribution Date means the Effective Date, or as soon as practicable thereafter, but no later than six months after the Effective Date, when the initial Distribution of Cash shall be made to the Holders of Allowed Claims, as determined by the Debtor, the Liquidating Trustee, or the Co-Liquidating Trustee, as applicable.
- 42. Oracle. The Oracle Objection is resolved pursuant to the Stipulation Regarding Oracle Americ Inc.'s Cure Objection and Reservation of Rights Regarding Debtor's Amended Joint Disclosure Statement and Champter 11 Plan of Liquidation [Docket No. 1250], which was approved by this Court [Docket No. 1256].
- 43. Premier Creditors. In resolution to an informal objection by DRP Holdings, LLC ("DRP"), Inland Valley Investments, LLC ("IVI"), Premier Healthcare Management, Inc. ("Premier"), and Promenade Square, LLC ("Promenade," collectively with DRP, IVI, and Premier, the "Premier Creditors"), the Plan Proponents filed the Stipulation By and Among the Debtor, the Official Committee of Unsecured Creditors and Creditors DRP Holdings, LLC, Inland Valley Investments, LLC, Premier Healthcare Management, Inc., and Promenade Square, LLC [Docket No. 1238] (the "Premier Stipulation"), which was approved by this Court [Docket No. 1239]. Pursuant to the Premier Stipulation, the following language is hereby adopted and approved:
- a. The Plan will reserve for the full amount of the Premier Creditor's claims and such amounts shall be deposited in interest bearing accounts, which maximize value and maintain safety. The interest-bearing accounts shall be invested in US 1 Month Treasury Bills or other US backed instruments.
- b. To the extent a Disputed Claim becomes an Allowed Claim, Holders of such Claims shall be entitled to the interest that accrues on the pro rata amount of their Claim.

Payment of Statutory Fees

44. Statutory Fees shall be paid by the Liquidating Trustee in the ordinary course of business until the closing, dismissal, or conversion of the Chapter 11 Case to another chapter of the Bankruptcy Code. Any unpaid Statutory Fees that accrued before the Effective Date shall be paid no later than thirty (30) days after the Effective Date. For the avoidance of doubt, if the Chapter 11 Case is reopened, the Liquidating Trustee shall pay any Statutory Fees in the ordinary course of business until the closing, dismissal, or conversion of the Chapter 11 Case to another chapter of the Bankruptcy Code.

Liquidating Trust Professionals

45. Dentons US LLP, Pachulski, Stang, Ziehl & Jones LLP and such other professionals as the Liquidating Trustee and Co-Liquidating Trustee may retain hereinafter (the "Retained Liquidating Trust Professionals") may be retained as counsel to the Liquidating Trust, pursuant to the Liquidating Trust Agreement. Any Retained Liquidating Trust Professional may be compensated pursuant to the terms of the Liquidating Trust Agreement and, for the avoidance of doubt, shall not be required to comply with §§ 330 and 331 solely with respect to such Retained Liquidating Trust Professional's retention and compensation or payment for work performed for the Liquidating Trust.

Notice of Entry of Confirmation Order and Effective Date

46. Pursuant to Bankruptcy Rules 2002 and 3020(c), the Plan Proponents are hereby authorized to serve a notice of entry of this Confirmation Order and the occurrence of the Effective Date, substantially in the form attached hereto as Exhibit C (the "Notice of Confirmation and Effective Date") on the Effective Date, on all Holders of Claims against the Debtor and all other Persons on whom the Confirmation Hearing Notice was served. The form of the Notice of Confirmation and Effective Date is hereby approved in all respects. The Notice of Confirmation and Effective Date shall constitute good and sufficient notice of the entry of this Confirmation Order and of the relief granted herein, including, without limitation, any bar dates and deadlines established under the Plan and this Confirmation Order, and no other or further notice of the entry of this Confirmation Order, the occurrence of the Effective Date, and any such bar dates and deadlines need be given.

Retention of Jurisdiction

47. Under §§ 105(a) and 1142, and notwithstanding entry of this Confirmation Order and the occurrence of the Effective Date, on and after the Effective Date, this Court shall retain exclusive jurisdiction over all matters arising out of, or related to, the Chapter 11 Case, the Plan, and each of the Plan Documents, to the extent provided under applicable CSD 1001A

law, including, among other things, to take the actions specified in Section 19 of the Plan.

References to Plan Provisions

48. The failure to specifically include or to refer to any particular section, or provision of the Plan or any related document in this Confirmation Order shall not diminish or impair the effectiveness of such section, or provision, and such section, or provision shall have the same validity, binding effect, and enforceability as every other section, or provision of the Plan, it being the intent of this Court that the Plan (as and to the extent modified by this Confirmation Order) be confirmed in its entirety.

Rules Governing Conflicts Between Documents

49. In the event that any provision of the Plan is inconsistent with the provisions of the Plan Supplement, and any other Order in the Chapter 11 Case, or any other agreement to be executed by any Person pursuant to the Plan, the provisions of the Plan shall control and take precedence; provided, however, that this Confirmation Order shall control and take precedence in the event of any inconsistency between this Confirmation Order, any provision of the Plan, and any of the foregoing documents.

Extension of Injunctions and Stays

50. Unless otherwise provided in the Plan or in this Confirmation Order, all injunctions or stays provided for in the Chapter 11 Case under §§ 105 or 362 or otherwise, and extant on the Confirmation Date (including any injunctions or stays contained in or arising from the Plan or this Confirmation Order), shall remain in full force and effect.

Section 1146 Exemption

51. Pursuant to § 1146(a), the issuance, transfer or exchange of any security under the Plan or the making or delivery of any instrument of transfer pursuant to, in implementation of, or as contemplated by the Plan, or the re-vesting, transfer or sale of any real or personal property of the Debtor pursuant to, in implementation of, or as contemplated by the Plan, shall not be taxed under any state or local law imposing a stamp tax, transfer tax or any similar tax or fee.

Headings

52. Headings utilized herein are for convenience and reference only, and do not constitute a part of the Plan or this Confirmation Order for any other purpose.

No Stay of Confirmation Order

53. Notwithstanding Bankruptcy Rules 3020(e) and 6004(h) and any other Bankruptcy Rule to the contrary, to the extent applicable, there is no reason for delay in the implementation of this Confirmation Order and, thus, this Confirmation Order shall be effective and enforceable immediately upon entry.

Approved as to form:

Haeji Hong

Office of the United States Trustee

EXHIBIT A

FIRST AMENDED JOINT COMBINED DISCLOSURE STATEMENT AND CHAPTER 11 PLAN OF LIQUIDATION OF BORREGO COMMUNITY HEALTH FOUNDATION

Docket #1168 Date Filed: 12/4/2023 of 160 1 SAMUEL R. MAIZEL (SBN 189301) JEFFREY N. POMERANTZ (SBN samuel.maizel@dentons.com 143717) TANIA M. MOYRON (SBN 235736) jpomerantz@pszjlaw.com STEVEN W. GOLDEN (Admitted *Pro* tania.moyron@dentons.com 3 REBECCA M. WICKS (SBN 313608) Hac Vice) sgolden@pszjlaw.com PACHULSKI STANG ZIEHL & JONES rebecca.wicks@dentons.com Dentons US LLP 4 601 South Figueroa Street, Suite 2500 Los Angeles, California 90017-5704 Telephone: 213 623 9300 Facsimile: 213 623 9924 5 10100 Santa Monica Blvd., 13th Floor Los Angeles, California 90067 Telephone: 310 227 6910 Facsimile: 310 201 0760 6 Attorneys for Chapter 11 Debtor and 7 Attorneys to the Official Committee of Debtor In Possession 8 **Unsecured Creditors** 9 UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF CALIFORNIA 10 11 Case No. 22-02384 In re: Chapter 11 Case 12 **BORREGO COMMUNITY** HEALTH FOUNDATION, FIRST AMENDED JOINT COMBINED 13 DISCLOSURE STATEMENT AND **CHAPTER 11 PLAN OF LIQUIDATION** Debtor and Debtor In Possession. 14 OF BORREGO COMMUNITY HEALTH FOUNDATION 15 Judge: Hon. Laura S. Taylor 16 Hearing: 17 Date: January 17, 2024 Time: 10:00 a.m. 18 Place: Department 3 19 20 21 22 23 24 25 26

1875 CENTURY PARK EAST, SUITE 1600 LOS ANGELES, CALIFORNIA 90067 TEL (310) 551-8111 • FAX (310) 551-8181

27



2202384231204000000000006

 $1 \parallel$

TABLE OF CONTENTS

2	SECTION	1. INT	RODUCTION	4
3	SECTION		IMARY OF CLASSIFICATION OF CLAIMS UNDER	
4			AND IMPORTANT SOLICITATION AND IRMATION DATES AND DEADLINES	6
5		2.1	Summary of Classification of Claims.	6
6		2.2	Important Dates and Deadlines.	6
7	SECTION	3. DEF	INITIONS AND INTERPRETATION	7
8	A.	Definit	ions	7
9	В.	Interpre	etation and Rules of Construction.	25
10	C.	Contro	lling Document	25
11	SECTION	4. BAC	KGROUND	26
12		4.1	Overview of the Debtor.	26
13		4.2	Events Leading to Chapter 11 Filing	27
14	SECTION	5. THE	CHAPTER 11 CASE	28
		5.1	Material First-Day Motions Filed on the Petition Date	28
15 16		5.2	Motion for Entry of Order: Authorizing (I) Key Employee Retention Program and (II) Key Employee Incentive Program.	20
17 18		5.3	Estate Professionals, the Committee, and the Patient Care Ombudsman.	
19		5.4	Administrative Matters, Reporting and Disclosures	30
20		5.5	Adversary Proceeding against DHCS.	31
21		5.6	The Sale of Substantially All Assets	33
22		5.7	Claims Bar Date and Reconciliation.	35
23	SECTION	6. CON	FIRMATION AND VOTING PROCEDURES	36
24		6.1	Confirmation Hearing	36
25		6.2	Procedures for Objections.	36
26		6.3	Requirements for Confirmation.	36
27		6.4	Classification of Claims.	37
28		6.5	Impaired Claims.	39
40				

	6.6	Confirmation Without Necessary Acceptances; Cramdown.	40
	6.7	Feasibility.	41
	6.8	Best Interests Test and Liquidation Analysis	41
	6.9	Eligibility to Vote on the Combined Plan and Disclosure Statement.	42
	6.10	Solicitation Package / Release Opt-Out Election Form	42
	6.11	Voting Procedures, Voting Deadline, and Deadline to Submit the Release Opt-Out Election.	43
	6.12	Acceptance of the Combined Plan and Disclosure Statement.	44
SECTION		RTAIN RISK FACTORS, TAX CONSEQUENCES,	
	AND (OTHER DISCLOSURES	45
	7.1	Certain Risk Factors to be Considered	45
	7.2	Certain U.S. Federal Income Tax Consequences	48
	7.3	Releases, Exculpations, and Injunctions.	55
	7.4	Alternatives to the Combined Plan and Disclosure Statement.	55
SECTION	8. UNC	CLASSIFIED CLAIMS	56
	8.1	Administrative Claims	56
	8.2	Professional Claims.	57
	8.3	Statutory Fees.	57
	8.4	Priority Tax Claims.	58
SECTION	9. CLA	ASSIFICATION OF CLAIMS AND VOTING	58
	9.1	Classification in General.	58
	9.2	Summary of Classification.	58
	9.3	Special Provision Governing Unimpaired Claims	59
	9.4	Elimination of Vacant Classes.	59
	9.5	Voting; Presumptions; Solicitation in Good Faith	59
	9.6	Cramdown.	59
SECTION	10. TR	EATMENT OF CLAIMS	60
	10.1	Class 1: Priority Non-Tax Claims	

- 1				
1		10.2	Class 2: Secured Claims.	60
2		10.3	Class 3: General Unsecured Claims.	61
3		10.4	Class 4: Allowed DHCS Claim.	61
4	SECTION	11. DIS	STRIBUTIONS	62
5		11.1	Party Responsible for Making Distributions	62
6		11.2	Appointment of Disbursing Agent	62
		11.3	Timing of Distributions	62
7 8		11.4	Manner of Cash Payments Under the Plan and Liquidating Trust Agreement.	63
9		11.5	Distribution Procedures.	63
10		11.6	Withholding of Distributions.	64
11		11.7	Delivery of Distributions and Undeliverable Distributions.	65
12		11.8	Setoffs and Recoupments.	
13		11.9	De Minimis Distributions.	
14		11.10	Allocation of Plan Distribution Between Principal and	
15		11110	Interest.	66
16		11.11	Entry of Final Decree in Chapter 11 Case	66
17		11.12	Distribution Cap.	67
18		11.13	Distributions Free and Clear	67
19	SECTION		OCEDURES FOR RESOLVING CONTINGENT, QUIDATED, AND DISPUTED CLAIMS	67
20		12.1	No Distributions Pending Allowance	67
21		12.2	Resolution of Disputed Claims.	67
22		12.3	Objection Deadline.	68
23		12.4	Allowance and Estimation of Claims.	68
24		12.5	Disallowance of Claims.	69
25		12.6	Adjustment Without Objection.	69
26		12.7	Reserve Provisions for Disputed Claims	70
27		12.8	Rounding.	70
28		12.9	Cumulative Effect	70

1	SECTION	13. LIQ	QUIDATING TRUST BENEFICIARIES	71
2		13.1	Identification of Liquidating Trust Beneficiaries	71
3		13.2	Beneficial Interests Only	71
4		13.3	Evidence of Beneficial Interests	71
5		13.4	Conflicting Claims.	71
6		13.5	Limitation on Transferability.	72
7	SECTION	14. EX	ECUTORY AGREEMENTS	72
		14.1	General Treatment.	72
8		14.2	Rejection Bar Date.	73
9		14.3	Insurance Policies.	73
10	SECTION	15. ME	ANS FOR IMPLEMENTATION OF THE PLAN	74
11		15.1	General Settlement of Claims.	74
12		15.2	Sale Transaction.	74
13		15.3	DHCS Settlement Agreement.	75
14		15.4	Plan Funding	76
15		15.5	Post-Effective Date Governance	76
16		15.6	Liquidating Trust.	79
17		15.7	Appointment of the Liquidating Trustee and Co- Liquidating Trustee.	80
18 19		15.8	Rights and Powers of Liquidating Trustee and Co- Liquidating Trustee.	80
20		15.9	Fees and Expenses.	81
21		15.10	Transfer of Beneficial Interests in the Liquidating Trust	81
		15.11	Litigation of Debtor's Causes of Action.	81
22		15.12	Full and Final Satisfaction.	82
23		15.13	Employment and Compensation of Professionals	82
24		15.14	No Further Court Authorization.	82
25		15.15	Dissolution of the Committee	82
26		15.16	Coordination Between Post-Effective Date Debtor and the	
27			Liquidating Trust.	83
28		15.17	Destruction and Abandonment of Books and Records	83

1		15.18	Mutuality Preserved.	.84
2	SECTION 1	16. EFI	FECT OF CONFIRMATION	.84
3		16.1	Binding Effect of the Plan.	.84
4		16.2	Vesting of Assets.	.84
5		16.3	No Discharge.	. 84
6		16.4	Pursuant to § 1141(d), the Debtor will not receive a discharge under this Plan.	. 84
7		16.5	Property Free and Clear	. 84
8	SECTION 1	17. EX	CULPATIONS, INJUNCTIONS, AND RELEASES	.85
9		17.1	Extension of Existing Injunctions and Stays	. 85
10		17.2	Releases.	.85
11		17.3	Injunctions	. 87
12		17.4	Exculpation	.88
13		17.5	No Recourse.	.89
14	:	17.6	Post-Confirmation Liability of Liquidating Trustee and the Co-Liquidating Trustee.	.89
15		17.7	Preservation of Rights of Action.	.90
16 17		17.8	Termination of Responsibilities of the Patient Care Ombudsman	.93
18	SECTION 1	18. CO	NDITIONS PRECEDENT TO EFFECTIVE DATE	.93
19		18.1	Conditions Precedent to Confirmation of Plan	.93
20		18.2	Conditions to Effective Date.	.93
		18.3	Waiver of Condition	.94
21	SECTION 1	19. RE	ΓENTION OF JURISDICTION	.95
22		19.1	Bankruptcy Court Jurisdiction.	.95
23	SECTION 2	20. MIS	SCELLANEOUS PROVISIONS	.97
2425	2	20.1	Termination of All Employee, Retiree and Workers' Compensation Benefits	.97
26		20.2	Administrative Claims Bar Date.	.98
27		20.3	Exemption from Transfer Taxes.	.98
		20.4	Amendments	
28				

20.5	Revocation or Withdrawal of Plan.	.99
20.6	Severability	.99
20.7	Request for Expedited Determination of Taxes	.99
20.8	Securities Exemption.	100
20.9	U.S. Trustee Quarterly Fees and Post-Confirmation Status Reports.	100
20.10	Courts of Competent Jurisdiction.	100
20.11	Governing Law.	100
20.12	Continuing Effect of the Bankruptcy Court Orders and Settlement Stipulations.	100
20.13	Substantial Consummation.	101
20.14	Waiver of Fourteen-Day Stay	101
20.15	Reservation of Rights.	101
20.16	Successors and Assigns.	101
20.17	Time	101
20.18	Business Day Transactions	101
20.19	Headings.	102
20.20	Exhibits	102
20.21	Notices.	102
20.22	Post-Effective Date Notices.	103
20.23	Conflict of Terms.	103
20.24	Entire Agreement.	103

2

3

4

5

6

7

8

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

DISCLAIMERS

EACH HOLDER OF A CLAIM AGAINST THE DEBTOR ENTITLED TO VOTE TO ACCEPT OR REJECT THE COMBINED PLAN AND DISCLOSURE STATEMENT SHOULD READ THE COMBINED PLAN AND DISCLOSURE STATEMENT IN ITS ENTIRETY BEFORE VOTING. NO SOLICITATION OF VOTES TO ACCEPT OR REJECT THE COMBINED PLAN AND DISCLOSURE STATEMENT MAY BE MADE EXCEPT PURSUANT TO THE TERMS HEREOF AND SECTIONS 1121 AND 1125 OF THE BANKRUPTCY CODE. 1 IF YOU ARE ENTITLED TO VOTE TO ACCEPT THE COMBINED PLAN AND DISCLOSURE STATEMENT, YOU ARE RECEIVING A BALLOT WITH YOUR NOTICE OF THE COMBINED PLAN AND DISCLOSURE STATEMENT. THE DEBTOR URGES YOU TO VOTE TO ACCEPT THE COMBINED PLAN AND DISCLOSURE STATEMENT. THIS COMBINED PLAN AND DISCLOSURE STATEMENT PROPOSED WITH THE OFFICIAL COMMITTEE UNSECURED CREDITORS (THE "COMMITTEE") WHICH SUPPORTS THE COMBINED PLAN AND DISCLOSURE STATEMENT AND URGES CREDITORS IN CLASS 3 TO VOTE TO ACCEPT THE COMBINED PLAN AND DISCLOSURE STATEMENT.

THE COMBINED PLAN AND DISCLOSURE STATEMENT HAS BEEN PREPARED IN ACCORDANCE WITH §§ 1121 AND 1125 AND BANKRUPTCY RULES 3016 AND 3017, AND NOT IN ACCORDANCE **FEDERAL** OR **STATE SECURITIES** LAWS OR APPLICABLE **NON-BANKRUPTCY** LAW. PERSONS OR ENTITIES PURCHASING, **TRADING OR OTHERWISE** SELLING. TRANSFERRING CLAIMS AGAINST THE DEBTOR SHOULD EVALUATE THE COMBINED PLAN AND DISCLOSURE STATEMENT IN LIGHT OF THE PURPOSE FOR WHICH IT WAS PREPARED. THE COMBINED PLAN AND DISCLOSURE STATEMENT SHALL NOT BE CONSTRUED TO BE ADVICE ON THE TAX, SECURITIES, OR OTHER LEGAL EFFECTS OF THE COMBINED PLAN AND DISCLOSURE STATEMENT AS TO HOLDERS OF CLAIMS AGAINST THE DEBTOR. YOU SHOULD CONSULT YOUR PERSONAL COUNSEL OR TAX ADVISOR ON ANY QUESTIONS OR CONCERNS RESPECTING TAX, SECURITIES, OR OTHER LEGAL CONSEQUENCES OF THE COMBINED PLAN AND

27

²⁶

¹ All references to section or chapter herein are to the Bankruptcy Code, 11 U.S.C. §§ 101, *et seq.*, as amended. All references to "Bankruptcy Rules" are to the Federal Rules of Bankruptcy Procedure.

DISCLOSURE STATEMENT.

1

2

3

4

5

6

7

9

10

11

14

15

16

18

19

20

21

22

23

24

25

26

27

28

THE COMBINED **PLAN AND DISCLOSURE STATEMENT** CONTAINS SUMMARIES OF CERTAIN STATUTORY PROVISIONS, DOCUMENTS RELATED TO THE COMBINED PLAN AND DISCLOSURE STATEMENT, EVENTS IN THE CHAPTER 11 CASE, AND FINANCIAL INFORMATION. ALTHOUGH THE DEBTOR AND THE COMMITTEE BELIEVE THAT THE STATEMENTS AND DESCRIPTIONS CONTAINED IN THE COMBINED PLAN AND DISCLOSURE STATEMENT ARE TRUE AND ACCURATE, THEY ARE QUALIFIED TO THE EXTENT THAT THEY DO NOT SET FORTH THE ENTIRE TEXT OF THE DOCUMENTS RELATED TO THE COMBINED PLAN AND DISCLOSURE STATEMENT AND APPLICABLE STATUTORY PROVISIONS. THE TERMS OF THE DOCUMENTS RELATED TO THE COMBINED PLAN AND DISCLOSURE STATEMENT AND APPLICABLE STATUTES GOVERN IN THE EVENT **DISCREPANCY WITH** THE **COMBINED PLAN OF** ANY DISCLOSURE STATEMENT. CREDITORS AND OTHER INTERESTED PARTIES SHOULD READ THE COMBINED PLAN AND DISCLOSURE STATEMENT, THE DOCUMENTS RELATED TO THE COMBINED PLAN AND DISCLOSURE STATEMENT, AND THE APPLICABLE STATUTES THEMSELVES FOR THE FULL AND COMPLETE STATEMENTS OF SUCH TERMS AND PROVISIONS.

THE **FACTUAL STATEMENTS** AND REPRESENTATIONS CONTAINED IN THE COMBINED PLAN AND DISCLOSURE STATEMENT ARE MADE BY THE DEBTOR AND THE COMMITTEE AS OF THE DATE HEREOF, UNLESS OTHERWISE SPECIFIED, AND THE DEBTOR AND THE COMMITTEE DISCLAIM ANY OBLIGATION TO UPDATE ANY SUCH STATEMENTS AFTER THE SOLICITATION OF VOTES TO ACCEPT OR REJECT THE COMBINED PLAN AND DISCLOSURE STATEMENT. THE DELIVERY OF THE COMBINED PLAN AND DISCLOSURE STATEMENT SHALL NOT BE DEEMED OR CONSTRUED **CREATE** ANY **IMPLICATION THAT** THE **INFORMATION** CONTAINED HEREIN IS CORRECT AT ANY TIME AFTER THE DATE HEREOF.

THE FINANCIAL INFORMATION CONTAINED HEREIN HAS NOT BEEN AUDITED BY A CERTIFIED PUBLIC ACCOUNTANT AND HAS NOT NECESSARILY BEEN PREPARED IN ACCORDANCE WITH GENERALLY ACCEPTED ACCOUNTING PRINCIPLES.

2

3

5

6

7

9

10

11

14

15

16

17

18

19

20

21

22

23

24

26

27

28

ANY PROJECTED RECOVERIES TO CREDITORS SET FORTH IN THIS DISCLOSURE STATEMENT ARE BASED UPON THE ANALYSES PERFORMED BY THE DEBTOR AND ITS ADVISORS. ALTHOUGH THE DEBTOR AND ITS ADVISORS HAVE MADE EVERY EFFORT TO VERIFY THE ACCURACY OF THE INFORMATION PRESENTED HEREIN, THE DEBTOR AND ITS ADVISORS CANNOT MAKE ANY REPRESENTATIONS OR WARRANTIES REGARDING THE ACCURACY OF THIS INFORMATION.

IN CONNECTION WITH THE DEBTOR'S AND THE COMMITTEE'S SOLICITATION OF ACCEPTANCES OF THE COMBINED PLAN AND DISCLOSURE STATEMENT PURSUANT TO § 1126(b), THE DEBTOR AND THE COMMITTEE ARE FURNISHING A SOLICITATION PACKAGE, **OF** THE **COMBINED PLAN** CONSISTING **AND DISCLOSURE** STATEMENT, THE EXHIBIT A HERETO, CONFIRMATION NOTICE, AND A BALLOT AND/OR A RELEASE OPT-OUT ELECTION FORM, AS APPLICABLE, TO EACH RECORD HOLDER OF CLAIMS ELIGIBLE TO VOTE OR ITS COUNSEL. THE COMBINED PLAN AND DISCLOSURE STATEMENT IS TO BE USED BY EACH SUCH ELIGIBLE HOLDER SOLELY IN CONNECTION WITH ITS EVALUATION OF THE COMBINED PLAN AND DISCLOSURE STATEMENT; USE OF THE COMBINED PLAN AND DISCLOSURE STATEMENT FOR ANY OTHER PURPOSE IS NOT AUTHORIZED. NOTHING STATED IN THE COMBINED PLAN AND DISCLOSURE STATEMENT SHALL BE DEEMED OR CONSTRUED AS AN ADMISSION OF ANY FACT OR LIABILITY BY ANY PARTY, OR BE ADMISSIBLE IN ANY PROCEEDING INVOLVING THE DEBTOR, THE

COMMITTEE OR ANY OTHER PARTY. THE COMBINED PLAN AND DISCLOSURE STATEMENT MAY NOT BE REPRODUCED OR PROVIDED TO ANYONE OTHER THAN ADVISORS TO THE RECIPIENT WITHOUT THE PRIOR WRITTEN CONSENT OF THE DEBTOR.

SECTION 1. INTRODUCTION

Borrego Community Health Foundation, (the "<u>Debtor</u>"), the debtor and debtor-in-possession in the above-captioned chapter 11 case, and the Official Committee of Unsecured Creditors (the "<u>Committee</u>") hereby jointly propose the following combined disclosure statement and plan pursuant to §§ 1121(a) and 1125(b) (the disclosure statement portion hereof, the "<u>Disclosure Statement</u>" and the chapter 11 plan portion hereof, the "<u>Plan</u>," as may be modified and/or amended from time to time, and collectively, the "<u>Combined Plan and Disclosure Statement</u>"). Capitalized terms used in the Combined Plan and Disclosure Statement and not otherwise defined have the meanings ascribed to such terms in Section 3.

The Plan proposes to pay or otherwise satisfy Allowed Administrative Claims, Allowed Secured Claims, Allowed General Unsecured Claims, and a portion of the Allowed DHCS Claim, in full on the Effective Date or as soon as practicably thereafter. The Plan creates Class A and Class B Liquidating Trust Interests and proposes to pay Allowed General Unsecured Claims and the Allowed DHCS Claim in accordance with the DHCS Settlement. The Plan also proposes the resolution of certain other Claims and the distribution of proceeds to Holders of Allowed Claims. Claims against the Debtor—other than Unclassified Claims—are classified in Section 9 and treated in accordance with Section 10 hereof. The Plan provides that (i) the Liquidating Trustee will administer the Class B Liquidating Trust Assets and continue the wind-down and liquidation of the Debtor after the Effective Date, and (ii) the Co-Liquidating Trustee will administer the Class A Liquidating Trust Assets to pay Holders of Allowed General Unsecured Claims.

The Debtor and the Committee will distribute the Combined Plan and Disclosure Statement to all holders of Claims in accordance with § 1125(b); Bankruptcy Rules 2002, 3016, and 3017; and the Bankruptcy Court's order conditionally approving the Combined Plan and Disclosure Statement [Docket No. _] (the "Conditional Approval and Procedures Order").

The Combined Plan and Disclosure Statement and the exhibit hereto include a discussion of: (i) the nature and history of the Debtor's business and liabilities; (ii) events during the Chapter 11 Case; (iii) the requirements for confirmation of the Plan and procedures for voting to accept or reject the Plan; (iv) additional factors and

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

disclosures to be considered, including risk factors and certain U.S. federal income tax consequences of the Plan; and (v) the terms of the Plan, including the treatment of holders of Claims under the Plan. The Disclosure Statement was prepared with the intent to provide "adequate information" (as defined in the Bankruptcy Code) to enable holders of Claims against the Debtor to make informed judgments about the Plan.

Subject to the restrictions on modifications set forth in § 1127 and Bankruptcy Rule 3019 and those restrictions on modifications set forth in Section 20.4 of the Combined Plan and Disclosure Statement, the Debtor and the Committee expressly reserve the right to alter, amend, or modify the Combined Plan and Disclosure Statement, including the Plan Supplement, one or more times, before substantial consummation thereof.

Please read the Combined Plan and Disclosure Statement, the exhibit, other supporting materials, and any appropriate ballot carefully and follow the instructions set forth below and on the appropriate ballot to vote on the Combined Plan and Disclosure Statement. The Debtor and the Committee believe that the Combined Plan and Disclosure Statement provides the best method of maximizing the recoveries for the holders of Claims against the Debtor. **Therefore, the Debtor and the Committee recommend that all creditors who are entitled to vote should vote in favor of the Combined Plan and Disclosure Statement.**

Unless otherwise specified, all section or exhibit references in the Combined Plan and Disclosure Statement are to the respective section in, or exhibit to, the Combined Plan and Disclosure Statement, as the same may be amended, waived, or modified from time to time. The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Combined Plan and Disclosure Statement as a whole and not to any particular section, subsection, or clause contained herein. The headings in the Combined Plan and Disclosure Statement are for convenience of reference only and shall not limit or otherwise affect the provisions hereof. For purposes herein: (a) in the appropriate context, each term, whether stated in the singular or the plural, shall include both the singular and the plural, and pronouns stated in the masculine, feminine, or neuter gender shall include the masculine, feminine, and the neuter gender; (b) any reference herein to a contract, lease, instrument, release, or other agreement or document being in a particular form or on particular terms and conditions means that the referenced document shall be substantially in that form or substantially on those terms and conditions; and (c) unless otherwise noted above, the rules of construction set forth in § 102 shall apply.

SECTION 2.

SUMMARY OF CLASSIFICATION OF CLAIMS UNDER PLAN AND IMPORTANT SOLICITATION AND CONFIRMATION DATES AND DEADLINES

2.1 Summary of Classification of Claims.

The following table designates the Classes of Claims against the Debtor and specifies which of those Classes are (a) not Impaired by the Plan, (b) Impaired by the Plan, and (c) entitled to vote to accept or reject the Plan in accordance with § 1126. In accordance with § 1123(a)(1), Administrative Claims, Professional Claims, Statutory Fees, and Priority Tax Claims, have not been classified. All of the potential Classes for the Debtor are set forth herein. If, ultimately, the Debtor does not have Holders of Claims in a particular Class or Classes, such Classes shall be treated as set forth in Section 10.

Class	Designation	Impairment	Entitled to Vote	
1	Priority Non-Tax Claims	Not Impaired	No (deemed to	
1	Friority Non-Tax Claims	Not impaired	accept)	
$\parallel _{2}$	Secured Claims	Not Impaired	No (deemed to	
	Secured Claims	Not impaired	accept)	
3	General Unsecured Claims	Impaired	Yes	
4	Allowed DHCS Claim	Impaired	Yes	

2.2 Important Dates and Deadlines.

Event	Proposed Date
Voting Record Date	November 28, 2023
Solicitation commences	December 11, 2023
Deadline to file Plan Supplement	December 11, 2023
Voting Objection Deadline	December 22, 2023
Deadline for Creditors to file Rule 3018 Motions	December 29, 2023
Deadline to respond to Voting Objection	December 29, 2023
Deadline for Debtor to respond to Rule 3018 Motions	January 5, 2024
Voting Deadline and deadline to submit the Release Opt- Out Election Form	January 8, 2024, at 4:00 p.m., Pacific Time

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

Event	Proposed Date
Combined Plan and Disclosure Statement Objection Deadline	January 8, 2024, at 4:00 p.m., Pacific Time
Deadline to file Confirmation Brief and other evidence supporting the Combined Plan and Disclosure Statement	January 11, 2024
Deadline to file Voting Tabulation Affidavit	January 11, 2024
Confirmation Hearing	January 17, 2024, at 10:00 a.m.

SECTION 3. DEFINITIONS AND INTERPRETATION

A. Definitions.

As used in the Combined Plan and Disclosure Statement, capitalized terms not otherwise defined have the meanings set forth below. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, shall have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.

- 3.1 Administrative Claim means a Request for Payment of an administrative expense of a kind specified in § 503(b) and entitled to priority pursuant to § 507(a)(2) or § 507(b), including: (i) the actual, necessary costs and expenses, incurred on or after the Petition Date, of preserving the Estate and operating the business of the Debtor through the Effective Date; (ii) the value of goods received by the Debtor within 20 days before the Petition Date in which the goods have been sold to the Debtor in the ordinary course of the Debtor's business pursuant to § 503(b)(9); (iii) Allowed Claims that are entitled to be treated as Administrative Claims pursuant to a Final Order of the Bankruptcy Court; (iv) Professional Claims; (v) and Statutory Fees.
- 3.2 Administrative Claims Bar Date means the deadline set by an order of the Bankruptcy Court by which Holders of Administrative Claims, other than Administrative Claims arising in the ordinary course of business for the Debtor or Professional Claims, must assert Administrative Claims or be forever barred, which shall be (a) thirty days after the Effective Date for Administrative Claims other than Professional Claims, and (b) 45 days after the Effective Date for Professional Claims.
- 3.3 Administrative Claims Objection Bar Date means the deadline for filing objections to requests for Administrative Claims required to be filed, which shall be

345

6 7

8 9 10

11 12

131415

1617

18 19

20

212223

2425

27

28

26

90 days following the Effective Date, *provided, however,* that the Administrative Claims Objection Bar Date may be extended by the Bankruptcy Court.

- 3.4 *Administrative Claims Reserve* means Cash to be set aside by the Debtor on the Effective Date in an aggregate amount estimated to fund the required amounts for payment of all unpaid Allowed Administrative Claims that will be paid after the Effective Date and all Administrative Claims that are not yet Allowed as of the Effective Date.
- 3.5 *Adversary Proceeding* shall have the meaning provided below in Section 5.5.
- 3.6 Allowed means for Distribution purposes, a Claim, or any portion thereof, or a particular Class of Claims: (a) that is Allowed by a Final Order of the Bankruptcy Court (or such other court as provided by the Plan or as the Liquidating Trustee or Co-Liquidating Trustee, as applicable, and the Holder of such Claim agree may adjudicate such Claim and objections thereto); (b) that is Allowed by this Plan and/or Confirmation Order; (c) is scheduled as not contingent, not unliquidated, and not disputed, and for which no superseding Proof of Claim has been timely filed; (d) for which a Proof of Claim in a liquidated amount has been timely filed with the Bankruptcy Court pursuant to the Bankruptcy Code or deemed timely filed by any Final Order of the Bankruptcy Court or other applicable bankruptcy law, and as to which (i) no objection to its allowance has been filed prior to the date of entry of the Confirmation Order or is not listed on the Disputed Claims Schedule; or (ii) any filed objection to its allowance has been settled or withdrawn, or has been denied by a Final Order of the Bankruptcy Court; (e) following the Effective Date, with respect to General Unsecured Claims, as otherwise may be determined as Allowed by the Liquidating Trust in accordance with the Plan and the Liquidating Trust Agreement; or (f) that is expressly allowed in a liquidated amount pursuant to this Plan.
- 3.7 *Allowed DHCS Balance Claim* means the Allowed DHCS Claim after: (i) the application of the DHCS Allowed Offset Amount, which amounts have been withheld by DHCS; and (ii) the payment of the DHCS Sales Proceeds Recovery. As set forth herein, the Allowed DHCS Balance Claim shall be subordinated to the prior payment in full of: (a) Allowed Administrative Claims; (b) Allowed Priority Claims; (c) Allowed General Unsecured Claims; and (d) the Effective Date Professional Claim Reserves.
- 3.8 *Allowed DHCS Claim* means the Allowed General Unsecured Claim of DHCS in the total amount of \$112,000,000, as accounted for and adjusted as set forth in the DHCS Settlement Agreement.

2

9

10

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

DENTONS US LLP

16 17

18 19

20 21

22 23

24 25

26 27

- *Appeal* shall have the meaning provided below in Section 5.5. 3.9
- 3.10 Assets means all legal or equitable interests of the Estate in any and all (a) property of every kind, nature, character and description, whether real, personal, or mixed, whether tangible or intangible (including contract rights), wherever situated and by whomever possessed, and any goodwill related thereto, including any real estate, buildings, structures, improvements, privileges, rights, easements, leases, subleases, goods, materials, supplies, furniture, fixtures, equipment, work in process, accounts, chattel paper, cash, deposit accounts, reserves, deposits, contractual rights, intellectual property rights, claims, Causes of Action, securities, investments and any other general intangibles, and (b) the proceeds, products, offspring, rents or profits thereof, including all assets of the Debtor constituting "property of the estate" as described in § 541.
- 3.11 Asset Purchase Agreement means the agreement between the Debtor and DAP Health setting forth the terms of the DAP Sale, pursuant to § 363, and approved by the Sale Order.
- 3.12 Avoidance Actions means any Causes of Action arising under any section of chapter 5 of the Bankruptcy Code, including, without limitation, §§ 502, 510, 541, 542, 543, 544, 545, 547, 548, 549, 550, 551, and 553 or under similar or related state or federal statutes and common law, including state fraudulent transfer laws.
- 3.13 **Ballots** means the ballots upon which the Holders of Impaired Claims shall indicate their acceptance or rejection of the Plan, in accordance with the Plan and the Voting Instructions.
- 3.14 **Ballot Deadline** means the date by which all Ballots must be properly executed, completed and delivered by First Class Mail, overnight courier, or hand delivery, to Kurtzman Carson Consultants LLC, at 222 N. Pacific Coast Highway, 3rd Floor, El Segundo, CA 90245, so as to be actually received by Kurtzman Carson Consultants LLC no later than 4:00 p.m. (Pacific Time), on the date set by the Bankruptcy Court in the Conditional Approval and Procedures Order.
- 3.15 Bankruptcy Code means title 11 of the United States Code, 11 U.S.C. §§ 101, et seq., as amended.
- 3.16 Bankruptcy Court means the United States Bankruptcy Court for the Southern District of California, or any other court having jurisdiction over the Chapter 11 Case, including, to the extent the jurisdictional reference of the Bankruptcy Court has been withdrawn to the United States District Court for the Southern District of California, pursuant to section 157(d) of title 28 of the United States Code.

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

DENTONS US LLP

16

21

24

26

27 28

3.17 *Bankruptcy Rules* means the Federal Rules of Bankruptcy Procedure as promulgated by the United States Supreme Court under section 2075 of title 28 of the United States Code, as may be amended from time to time.

- 3.18 *Bar Date* means the applicable deadlines by which a Proof of Claim or Request for Payment must be, or must have been, filed in in this Chapter 11 Case, as established by either the Bar Date Order, an order of the Bankruptcy Court or this Plan, including without limitation: (a) November 21, 2022 deadline to file Proofs of Claim relating to prepetition Claims; (b) the March 13, 2023 deadline to file Proofs of Claim for Governmental Units; (c) the Administrative Claims Bar Date; (d) the Extended DHCS Bar Date; and (e) the Rejection Bar Date.
- 3.19 Bar Date Order means any order of the Bankruptcy Court establishing Bar Dates for filing Proofs of Claim or Requests for Payment in this Chapter 11 Case, as the same may be amended, modified or supplemented including, but not limited to, the order at Docket No. 16.
- 3.20 Business Day means any day other than a Saturday, a Sunday, or any other day on which banking institutions in the State of California are required or authorized to close by law or executive order.
- Cash means the legal tender of the United States of America and its 3.21 equivalent.
- 3.22 Causes of Action means any and all present or future claims, rights, interests, legal and equitable defenses, offsets, recoupments, actions in law or equity or otherwise, choses in action, obligation, guaranty, controversy, demand, action suits, damages, judgments, third-party claims, counter-claims, cross-claims against any Person, whether known or unknown, liquidated or unliquidated, foreseen or unforeseen, existing or hereafter arising, whether based on legal or equitable relief, whether arising under the Bankruptcy Code or federal, state, common, or other law or equity, whether or not the subject of a pending litigation or proceedings on the Effective Date or thereafter of the Estate, the Debtor, or the Liquidating Trust, as applicable, including without limitation: (a) all Avoidance Actions; (b) all other claims in avoidance, recovery, and/or subordination; (c) all claims for the turnover of property to the Debtor or the Liquidating Trust; (d) all claims for compensation for damages incurred by the Debtor; (e) all claims arising in connection with *Husam E*. Aldairi, et al. v. Borrego Community Health Foundation, Case No. 37-2021-00046200-CU-BC-CTL (Cal. Sup. Ct. San Diego); (f) all claims in Borrego Community Health Foundation v. Inland Valley, LLC, et al., Case No. 3:21-cv-01417-AJB-AGS (S.D. Cal.); (g) all claims in Borrego Community Health Foundation v. Karen Hebets, et al., Case No. 3:22-cv-01056-AJB-AGS (S.D. Cal.); (h) all claims in

1

4 5

6

7 8 9

11

10

12 13

14

1516

18 19

17

20

2223

21

2425

27

28

26

Borrego Community Health Foundation v. Travelers Casualty and Surety Company of America, Case No. 3:22-CV-161-L-MDD (S.D. Cal.); and (i) all other actions described in this Combined Plan and Disclosure Statement, the Confirmation Order, the Schedules, or the Plan.

- 3.23 *Chapter 11 Case* means the voluntary chapter 11 case filed by the Debtor on the Petition Date under the caption, *In re Borrego Community Health Foundation*, Case No. 22-02384, currently pending before the Bankruptcy Court.
- 3.24 *CHOW* means the change of ownership application submitted pursuant to 42 C.F.R § 489.18 by the Debtor and DAP Health for approval by CMS that will result in the transfer of the Debtor's Medicare Identification Number and Medicare Provider Agreement to DAP Health.
- 3.25 *CHOW Effective Date* means date of the approval of the CHOW by CMS.
 - 3.26 *Claim* has the meaning set forth in § 101(5).
- 3.27 *Claims and Balloting Agent* means Kurtzman Carson Consultants LLC, which was appointed as the Debtor's claims, noticing, and balloting agent.
- 3.28 *Claims Objection Deadline* means the first Business Day that is the later of (a) two hundred ten (210) days after the Effective Date, or (b) such other later date as the Bankruptcy Court may establish upon a motion by the Liquidating Trustee in accordance with the Plan.
 - 3.29 *Claimant* means the Holder of a Claim.
 - 3.30 *Class* means a class of Claims established pursuant to Section 9 herein.
- 3.31 *Class A Liquidating Trust Assets* means, collectively: (i) the Remaining Cash; and (ii) (a) 67% of the first \$1 million of Net Recovery, (b) 33% of the second \$1 million of Net Recovery, and (c) for any Net Recovery thereafter, the Pro Rata share of such Net Recovery among the Holders of Class A Trust Beneficial Interests and Class B Trust Beneficial Interests.
- 3.32 *Class A Trust Beneficial Interests* means the interests in the Liquidating Trust of the Holders of Allowed Claims in Class 3 and their concomitant entitlement to Distributions to be made by the Liquidating Trust on account of Allowed General Unsecured Claims from the Class A Liquidating Trust Assets.

67

8

9 10

11

- 12 13
- 1415
- 16
- 18

17

- 19 20
- 2122
- 24

23

26

25

- 3.33 Class B Liquidating Trust Assets means collectively: (i) the DHCS Sales Proceeds Recovery; (ii) (a) 33% of the first \$1 million of Specified Litigation Recoveries, (b) 67% of the second \$1 million of Specified Litigation Recoveries, and (c) for any Specified Litigation Recoveries thereafter, the Pro Rata share of such Specified Litigation Recoveries among the Holders of Class A Trust Beneficial Interests and Class B Trust Beneficial Interests; and (iii) solely to the extent any Class A Liquidating Trust Assets remain after Holders of Class A Trust Beneficial Interests are paid in full, the Class A Liquidating Trust Assets.
- 3.34 *Class B Trust Beneficial Interests* means the interest in the Liquidating Trust of the Holders of Allowed Claims in Class 4 and their concomitant entitlement to Distributions to be made by the Liquidating Trust on account of the Allowed DHCS Balance Claim from the Class B Liquidating Trust Assets.
- 3.35 *Closing* means the consummation of the transactions contemplated by the DAP Sale pursuant to the terms of the Asset Purchase Agreement.
- 3.36 *Closing Date* means July 31, 2023, pursuant to the Asset Purchase Agreement, as set forth in the *Notice of Occurrence of Closing of Sale to DAP Health, Inc.* [Docket No. 823].
 - 3.37 *CMS* means the Centers for Medicare and Medicaid Services.
- 3.38 *Committee* means the Official Committee of Unsecured Creditors appointed on September 26, 2022, by the U.S. Trustee in this Chapter 11 Case pursuant to § 1102 [Docket No. 49].
- 3.39 *Conditional Approval and Procedures Order* means the order entered by this Court (i) conditionally approving the disclosures set forth in the Combined Plan and Disclosure Statement, and (ii) approving the solicitation procedures set forth in the Combined Plan and Disclosure Statement on December [_], 2023 [Docket No. _].
- 3.40 *Confirmation* means the entry of the Confirmation Order, subject to all conditions specified in Section 18.1 hereof having been satisfied.
- 3.41 *Confirmation Date* means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the docket.
- 3.42 *Confirmation Hearing* means the hearing to be held by the Bankruptcy Court to consider Confirmation of the Plan, as such hearing may be adjourned or continued from time to time.

- 1 2
- 3 4
- 56
- 7 8
- 10

- 11 12
- 13 14
- 151617
- 18 19
- 2021
- 23

24

22

- 25
- 2627

- 3.43 *Confirmation Order* means the order of the Bankruptcy Court confirming this Plan pursuant to § 1129.
- 3.44 *Co-Liquidating Trustee* means such person selected pursuant to Section 15 of the Plan or any successor or replacement officer appointed under the terms of the Plan and Liquidating Trust Agreement.
 - 3.45 *DAP Health* means Desert Aids Project d/b/a DAP Health.
- 3.46 *DAP Sale* means the sale of substantially all of the Debtor's assets to DAP Health pursuant to § 363.
- 3.47 *Debtor* means Borrego Community Health Foundation, in its capacity as debtor and debtor in possession in this Chapter 11 Case.
 - 3.48 **DHCS** means the California Department of Health Care Service.
- 3.49 **DHCS 9019 Order** means the Order granting the Debtor's Motion to Approve Compromise Among Debtor, Official Committee of Unsecured Creditors, and California Department of Health Care Services [Docket No. 544] and approving the DHCS Settlement Agreement.
- 3.50 **DHCS** Allowed Offset Amount means the approximately \$20,600,000 that DHCS is withholding from the Debtor, which amount is inclusive of approximately \$6,200,000 in monies otherwise payable to the Debtor for the provision of in-house dental services to Medi-Cal beneficiaries, consistent with the definition in the DHCS Settlement Agreement.
- 3.51 **DHCS Findings of Fact** shall have the meaning provided below in Section 5.5.
 - 3.52 *DHCS Order* shall have the meaning provided below in Section 5.5.
- 3.53 **DHCS Sales Proceeds Recovery** means the 40% of the Net Cash Proceeds of the DAP Sale received by DHCS on or around the Effective Date of this Plan, consistent with the definition in the DHCS Settlement Agreement.
- 3.54 **DHCS Settlement Agreement** means the settlement agreement by and between the Debtor, the Committee, and DHCS that resolved all disputes between the Debtor, the Committee, and DHCS, approved by the DHCS 9019 Order.
- 3.55 **DHHS** means the United States Department of Health and Human Services.

2

3

4

5

6

7

8

9

10

11

12

13

- 9300
- 1415
- 1617
- 1819

20

- 2223
- 2425
- 2627
- 28

- 3.56 *Disbursing Agent* means the individual or individuals as may be retained by the Liquidating Trustee or Co-Liquidating Trustee, as applicable, to assist him or her distribute the Liquidating Trust Assets in accordance with this Plan and the Liquidating Trust Agreement.
- 3.57 *Dispute Resolution* shall have the meaning set forth below in Section 13.4.
- 3.58 *Disputed Claim* means a Claim that is either: (i) as of the date of entry of the Confirmation Order, the subject of a pending objection; or (ii) listed on a schedule included with the Plan Supplement (the "<u>Disputed Claims Schedule</u>").
- 3.59 *Disputed Claim Reserve* shall have the meaning set forth below in Section 12.7.
- 3.60 *Distribution Date* means the Initial Distribution Date or any Subsequent Distribution Date, as applicable.
- 3.61 *Distributions* means the distributions of Cash to be made in accordance with the Plan and the Liquidating Trust Agreement.
- 3.62 *Distribution Record Date* means the close of business on the Business Day immediately preceding the Effective Date.
- 3.63 *District Court* means the United States District Court for the Southern District of California.
- 3.64 *Effective Date* means a day, as determined by the Plan Proponents, that is a Business Day as soon as reasonably practicable after all conditions to the Effective Date specified in Section 18.2 hereof have been satisfied or waived.
- 3.65 *Effective Date Professional Claim Reserves* means cash to be set aside by the Liquidating Trustee on the Effective Date sufficient in the aggregate to fund a reserve on account of accrued and unpaid Professional Claims not yet fixed and allowed by the Bankruptcy Court prior to or on the Effective Date.
 - 3.66 *Entity* means an entity as defined in § 101(15).
 - 3.67 *Estate* means the estate created upon the Petition Date pursuant to § 541.
- 3.68 *Exculpated Party* means, individually and collectively: (a) the Debtor; (b) the Debtor's trustees, officers, and managers serving in such capacity on and after the Petition Date; (c) members of the Committee (solely in their capacities as Committee members); and (d) Bankruptcy Court-approved Estate and Committee

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

professionals. Notwithstanding the foregoing, none of the Prepetition Fraud Parties are an Exculpated Party.

- 3.69 Excluded Party means all enumerated defendants in the Causes of Action and the Prepetition Fraud Parties, as set forth in the Plan Supplement.
- 3.70 *Executory Agreement* means any executory contract or unexpired lease subject to § 365, excluding any executory contract or unexpired lease entered into after the Petition Date and approved by an order of the Bankruptcy Court.
- 3.71 Extended DHCS Bar Date means December 29, 2023, which is the deadline by which DHCS must file a Proof of Claim for any further General Unsecured Claims against the Debtor for Medi-Cal overpayments, consistent with the definition in the DHCS Settlement Agreement.
- 3.72 *Final Decree* means the decree contemplated under Bankruptcy Rule 3022.
- 3.73 Final Distribution means the last payment to Holders of Allowed Claims in accordance with the provisions of the Plan.
- 3.74 *Final Order* means an order or judgment, the operation or effect of which has not been reversed, stayed, modified, or amended, is in full force and effect, and as to which order or judgment (or any reversal, stay, modification, or amendment thereof) (a) the time to appeal, seek *certiorari*, or request reargument, further review, or rehearing has expired and no appeal, petition for *certiorari*, request for reargument or further review, or rehearing has been timely filed, or (b) any appeal that has been or may be taken, or any petition for certiorari or request for reargument or further review or rehearing that has been or may be filed, has been resolved by the highest court to which the order or judgment was appealed, from which certiorari was sought, or to which the request was made, and no further appeal, petition for *certiorari*, request for reargument, or further review or rehearing has been or can be taken or granted; provided, however, that the possibility that a motion under Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed relating to such order shall not prevent such order from being a Final Order; provided, further, that the Debtor or Liquidating Trustee, as applicable, reserve the right to waive any appeal period for an order or judgment to become a Final Order.
- 3.75 General Unsecured Claim means any Claim against the Debtor that is not a/an: (i) Administrative Claim; (ii) Professional Claim; (iii) Secured Claim; (iv) Priority Claim; (v) Allowed DHCS Claim; (vi) Allowed DHCS Balance Claim; or (vii) Statutory Fee.

- 3.76 *Governmental Unit* has the definition set forth in § 101(27).
- 3.77 *Holder* means a holder of a Claim against the Debtor.
- 3.78 *Impaired* means, with respect to a Class of Claims, that such Class is "impaired" within the meaning of § 1124.
- 3.79 *Indemnified Parties* shall have the meaning set forth below in Section 17.6.
- 3.80 *Initial Distribution Date* means the Effective Date, or as soon as practicable thereafter when the initial Distribution of Cash shall be made to the Holders of Allowed Claims, as determined by the Debtor, the Liquidating Trustee, or the Co-Liquidating Trustee, as applicable.
- 3.81 *Insurance Policy* means any insurance policy maintained by or for the benefit of the Debtor, regardless of whether such Insurance Policy is set forth in a schedule to the Plan Supplement.
- 3.82 *Liquidating Trust* means the liquidating trust created pursuant to the Liquidating Trust Agreement.
- 3.83 *Liquidating Trust Agreement* means the *Liquidating Trust Agreement*, to be dated on or prior to the Effective Date, as may be modified from time to time, between the Debtor and the Liquidating Trustee.
- 3.84 *Liquidating Trust Assets* means all assets of the Debtor that (i) exist immediately prior to the Effective Date and are not otherwise used by the Debtor to make Distributions or create Reserves on the Effective Date to or for the benefit of Holders of Allowed Administrative Claims, Priority Non-Tax Claims, Professional Claims and Secured Claims, and (ii) are not Purchased Assets. The Liquidating Trust Assets also include, without limitation, to the extent not otherwise expressly excluded by this definition: (i) the Remaining Cash; (ii) all Causes of Action the Debtor holds or may hold against any Person or Entity as of the Effective Date (except to the extent they are the subject of any of the Releases set forth in the Plan); (iii) all Claims and rights of the Debtor under any Insurance Policies; (iv) any and all other non-Cash assets, interests, rights, claims and defenses of the Debtor or the Estate, including, without limitation, all rights under any order of the Bankruptcy Court; (v) any and all tax refunds to which the Debtor may be entitled; and (vi) any and all proceeds of any of the foregoing.
- 3.85 *Liquidating Trust Beneficiaries* means the Holders of Trust Beneficial Interests, as of any point in time.

1

456

7

8 9 10

121314

11

15

16

17

18

19 20

2122

2324

25

2627

- 3.86 *Liquidating Trustee* means such person selected pursuant to Section 15 of the Plan or any successor or replacement officer appointed under the terms of the Plan and Liquidating Trust Agreement.
- 3.87 *Litigation* means the interest of the Estate, the Debtor, or the Liquidating Trust, as applicable, in any and all claims, rights, and Causes of Action that have been or may be commenced by the Debtor or the Liquidating Trust, as applicable, except to the extent concerning any Released Parties. Litigation includes, without limitation not otherwise stated herein, any action: (i) to avoid and recover any transfers of property determined to be preferential, fraudulent, or avoidable pursuant to §§ 544, 545, 547, 548, 549(a), and 550; (ii) for the turnover of property to the Debtor or Liquidating Trust, as applicable; (iii) for the recovery of property or payment of money that belongs to or can be asserted by the Debtor or the Liquidating Trust, as applicable; (iv) for compensation for damages incurred by the Debtor; and (v) equitable subordination actions against Creditors.
- 3.88 *Litigation Recoveries* means any Cash or other property received by the Debtor or the Liquidating Trust, as applicable, from all or any portion of the Litigation of any of the Causes of Action, including, but not limited to, awards of damages, attorneys' fees and expenses, interest, and punitive damages, whether recovered by way of settlement, execution on judgment, or otherwise. If any litigation of any of the Causes of Action is pursued on a contingent-fee basis, the Litigation Recovery will be net of any expenses of such litigation and any contingent fee paid to legal counsel.
- 3.89 *Local Bankruptcy Rules* means the Local Rules of the United States Bankruptcy Court of the Southern District of California, as amended from time to time.
- 3.90 *Management Services Support Agreement* means the agreement between the Debtor and DAP Health for DAP Health to provide a broad range of management support services prior to the Closing Date as set forth in the Asset Purchase Agreement.
- 3.91 *Medi-Cal* means the program administered by the State of California for medical assistance under title XIX of the Social Security Act.
- 3.92 *Medicare* means the federal health insurance program administered by CMS under title XVIII of the Social Security Act.
- 3.93 *Net Cash Proceeds* means the aggregate cash consideration paid to or retained by the Debtor at Closing of the DAP Sale net of the Sale Proceeds Holdback.

1

- 4 5
- 7

8

6

- 9 10
- 11 12
- 13 14
- 15 16
- 17
- 181920
- 2122
- 2324
- 26

25

- 3.94 *Net Recovery* means the aggregate amount of any cash recovery from litigation of any of the Causes of Action realized by the Debtor or its successor-in-interest net of the fees (including attorney's fees) and expenses of such litigation.
- 3.95 *Nonprofit Laws* means any and all federal, state, local and other laws and governmental regulations applicable to nonprofit corporations, including without limitation, any administrative and judicial interpretations thereof (as applicable).
- 3.96 *Nonprofit Status* means status as a nonprofit corporation under applicable Nonprofit Laws.
 - 3.97 *Ombudsman Parties* has the meaning set forth in Section 17.8 herein.
- 3.98 *Operate* (and any such variations, such as "Operation") means to operate, oversee, manage, administer, coordinate, control, supervise and/or direct the business and operations of any and/or all of the Purchased Assets, whether in the ordinary course of business or in accordance with the Asset Purchase Agreement, Transition Services Agreement, or otherwise, and including undertaking or pursuing strategies, activities, or actions with the intent of furthering the objectives of, and otherwise to effectuate the Plan as contemplated by the provisions hereof, including any strategies, activities or actions aimed at retaining, renewing, amending, extending or Transferring any of the Purchased Assets.
- 3.99 *Operating Account* means one or more deposit accounts of Cash established and/or maintained by the Liquidating Trustee as set forth in Section 15.5(c).
- 3.100 *Ordinary Course Professionals Order* means the order [Docket No. 400] entered by the Bankruptcy Court granting the Debtor's motion to retain and compensate professionals utilized by the Debtor in the ordinary course of business [Docket No. 271].
- 3.101 *Ordinary Course Professionals* means the professionals retained by the Debtor in the ordinary course of their business operations, pursuant to the Ordinary Course Professionals Order.
- 3.102 *Patient Care Ombudsman* means Dr. Jacob Nathan Rubin, MD, FACC, appointed by the U.S. Trustee to serve as the patient care ombudsman in the Chapter 11 Case, pursuant to § 333(a), in accordance with the order entered by the Bankruptcy Court on September 16, 2022 [Docket No. 25].
- 3.103 *Person* means an individual, partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint

3

4

5

6

7

8

9

11

12

13

14

15

17

26

27

venture,	governmental	authority,	Governmental	Unit	or	other	entity	of	whatever
nature.									

- 3.104 **Petition Date** means September 12, 2022, which is the date that the Debtor filed a voluntary chapter 11 petition.
- 3.105 *Plan* means this plan of liquidation proposed by the Plan Proponents, including the Plan Supplement and the exhibit hereto and thereto, as the same may be amended, modified or supplemented from time to time in accordance with the provisions of the Bankruptcy Code and its terms.
 - 3.106 *Plan Proponent* means each of the Debtor and the Committee.
- 3.107 *Plan Supplement* means a supplemental appendix to this Plan, as may be amended from time to time on or prior to the Effective Date, which will contain the following items:
 - a) the Schedule of Assumed Contracts;
 - b) the list of Retained Causes of Action;
 - c) the list of Excluded Parties;
 - d) the Disputed Claims Schedule;
 - e) the schedule of Insurance Policies;
 - f) the initial Wind-down Budget;
 - g) the identity of the directors serving on the Post-Effective Date Board of Directors;
 - h) the identity of the initial Liquidating Trustee;
 - i) the identity of the initial Co-Liquidating Trustee;
 - j) the form of Liquidating Trust Agreement; and
 - k) the schedule of estimated costs of administration and any other funds required to be distributed upon the Effective Date as required by Local Rule 3020-1.

456

9 10

7

8

111213

14 15

16 17

> 18 19

202122

2425

23

27

28

26

provided that, the Debtor shall file items (a) through (k) by December 11, 2023. The Debtor may file separately each of the foregoing documents. The Plan Supplement shall be in substance and form acceptable to each of the Plan Proponents.

- 3.108 *Post-Effective Date Board of Directors* means the three (3) member board of directors for the Debtor that will be formed prior to or on the Effective Date in accordance with Section 15.5(b) hereof. The initial members of the Post-Effective Date Board of Directors shall be: (i) Jenna LeComte-Hinley, PhD; (ii) Frank Figueroa; and (iii) Martha Deichler. All proposed members of the Post-Effective Date Board of Directors are members of the Debtor's existing Board of Directors.
- 3.109 *Post-Effective Date Debtor* means the Debtor, in existence as of the Effective Date, which shall exist solely for the limited duration and purposes set forth in the Plan.
- 3.110 *Prepetition Fraud* means the fraud discovered by the Debtor orchestrated by certain of the Debtor's prior management, landlords and contractors and community contract dentists that involved filing false claims for dental services provided by the contract dentists.
- 3.111 *Prepetition Fraud Parties* means the parties who are alleged to have participated in the Prepetition Fraud as set forth in the Plan Supplement.
 - 3.112 *Priority Claim* means a Priority Non-Tax Claim or a Priority Tax Claim.
- 3.113 *Priority Non-Tax Claim* means any Claim entitled to priority in payment as specified in § 507(a)(4), (5), (6), (7) or (9) other than Administrative Claims and Priority Tax Claims.
- 3.114 *Priority Tax Claims* means Claims of any Governmental Unit entitled to priority under § 507(a)(8) and 507(c).
- 3.115 *Professional* means any Person (a) retained in the Chapter 11 Case by Final Order, pursuant to §§ 327, 363, and 1103 or otherwise; or (b) awarded compensation and reimbursement by the Bankruptcy Court, pursuant to § 503(b)(4).
- 3.116 *Professional Claim* means an Administrative Claim of a Professional for compensation for services rendered or reimbursement of costs, expenses, or other charges and disbursements incurred relating to services rendered or expenses incurred after the Petition Date and prior to and including the Effective Date.
 - 3.117 *Proof of Claim* means a proof of claim filed in this Chapter 11 Case.

2

3

4

5

6

7

8

10

11

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

- 3.118 *Provider Agreements* means (i) the Medicare Health Insurance Benefits Agreements between the Debtor and DHHS, and (ii) the Medi-Cal Provider Agreements between the Debtor and DHCS.
- 3.119 *Purchased Assets* means collectively all of the Assets listed in Section 1.7 of the Asset Purchase Agreement.
- 3.120 Rejection Bar Date means the last date for any Entity whose claims arise out of the Bankruptcy Court approved rejection of an executory contract or unexpired lease to file a Proof of Claim for damages related to such rejection. The Rejection Bar Date for such Claims will be, (i) with respect to executory contracts and unexpired leases rejected pursuant to a Bankruptcy Court order other than the Confirmation Order, the date provided by an order approving the rejection, and, (ii) with respect to executory contracts and unexpired leases rejected pursuant to the Confirmation Order, the date that is thirty (30) days after the Effective Date.
- 3.121 *Related Persons* means, subject to any exclusions expressly set forth in the Plan, with respect to a specific Person, said Person's successors and assigns, as applicable, its current and former shareholders, affiliates, subsidiaries, employees, agents, investment managers, subagents, officers, directors, managers, trustees, partners, members, professionals, representatives, advisors, attorneys, financial advisors, accountants, and consultants.
 - 3.122 *Releases* shall have the meaning provided below in Section 17.2.
- 3.123 *Release Opt-Out Election Form* means a form for Holders of Claims to opt out of being a Releasing Party in connection with the Third Party Release set forth in Section 17 of the Plan.
- 3.124 *Released Party* means, individually and collectively: (a) the Debtor, (b) the Committee, (c) the following members of the Committee: McKesson Corporation; Greenway Health, LLC; We Klean Inc.; Mustafa Bilal, DDS, Inc.; Vista Village Family Dentistry; Vitamin D Public Relations, LLC; and Pourshirazi & Youssefi Dental Corporation; and (d) each of the Related Persons of each of the Entities in the foregoing clauses (a)-(c); provided, however, that notwithstanding anything to the contrary herein, including the definition of "Related Persons," none of the Prepetition Fraud Parties are a Released Party.
- 3.125 *Releasing Party* means (a) the Released Parties; and (b) all Claimants that (i) vote to accept the Plan (or are deemed to accept the Plan), and (ii) do not affirmatively opt out of Third Party Releases pursuant to a duly executed Release Opt-Out Election Form; provided, that, notwithstanding anything contained herein to the

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

contrary, in no event shall an Entity be a Releasing Party that (x) does not vote to accept or reject the Plan, (y) votes to reject the Plan, or (z) appropriately marks the Release Opt-Out Election Form to opt out of the Third Party Releases and returns such Release Opt-Out Election Form in accordance with the Plan and the Voting Instructions.

- 3.126 *Remaining Cash* means the actual sum of Cash that constitutes Liquidating Trust Assets after: (i) the payment of Cash necessary to satisfy all Allowed Unclassified Claims that are Allowed on or prior to the Effective Date; (ii) the funding of reserves for disputed Unclassified Claims to the extent required by the Bankruptcy Court on or prior to the Effective Date; (iii) the payment of all Allowed Claims payable on the Effective Date as set forth in Classes 1-2; and (iv) the Transfer into or maintenance of funds in the Operating Accounts for the Post-Effective Date Debtor on the Effective Date in accordance with Section 15.5(c).
- 3.127 *Remaining Estate Funds* means the actual sum of Cash held by the Debtor on the Effective Date.
- 3.128 *Request for Payment* means a request for payment of an Administrative Claim filed in the Chapter 11 Case.
- 3.129 *Retained Causes of Action* shall have the meaning set forth below in Section 17.7(b).
- 3.130 *Sale Effective Time* means midnight (Pacific Coast Time) following the Closing Date of the DAP Sale.
- 3.131 *Sale Order* means the Final Order approving the DAP Sale, pursuant to § 363, titled Order (A) Authorizing the Sale of Property to Desert Aids Project d/b/a DAP Health Free and Clear of Liens, Claims, Encumbrances, and Other Interests; (B) Approving the Assumption and Assignment of an Unexpired Lease Related Thereto; and (C) Granting Related Relief [Docket No. 559].
- 3.132 *Sale Proceeds Holdback* means \$16,000,000, which amount is the Debtor's good faith estimate of the aggregate amount of Secured Claims, Priority Claims, and Administrative Claims, which amount may be adjusted only in a manner consistent with the DHCS Settlement Agreement.
- 3.133 *Schedule of Assumed Contracts* means the schedule listing the Executory Agreements to be assumed pursuant to the Plan.
- 3.134 *Scheduled* means, with respect to any Claim, the status, priority, and amount, if any, of such Claim as set forth in the Schedules.

1

67

8

9

1011

12

13 14

> 16 17

15

19

20

18

21

23

24

25

22

2627

- 3.135 *Schedules* means the schedules of assets and liabilities and the statements of financial affairs filed by the Debtor in the Chapter 11 Case pursuant to § 521 and Bankruptcy Rule 1007 [Docket Nos. 97, 98], which incorporate by reference the global notes and statement of limitations, methodology, and disclaimer regarding the Debtor's schedules and statements, as such schedules or statements have been or may be further modified, amended, or supplemented from time to time in accordance with Bankruptcy Rule 1009 or Final Orders of the Bankruptcy Court.
- 3.136 **Secured Claim** means a Claim that is (a) secured by a lien on any of the Assets, which lien is valid, perfected, and enforceable under applicable law or by reason of a Final Order, to the extent of the value of the claimant's interest in such Asset, or (b) entitled to setoff under § 553, to the extent of the amount subject to such setoff, as determined pursuant to § 506(a) or § 1129(b).
- 3.137 *Solicitation Package* shall have the meaning set forth below in Section 6.10.
- 3.138 *Specified Litigation* means the following lawsuits: (i) Husam E. Aldairi, et al. v. Borrego Community Health Foundation, Case No. 37-2021-00046200-CU-BC-CTL (Cal. Sup. Ct. San Diego); (ii) Borrego Community Health Foundation v. Inland Valley, LLC, et al., Case No. 3:21-cv-01417-AJB-AGS (S.D. Cal.); (iii) Borrego Community Health Foundation v. Karen Hebets, et al., Case No. 3:22-cv-01056-AJB-AGS (S.D. Cal.); and (iv) Borrego Community Health Foundation v. Travelers Casualty and Surety Company of America, Case No. 3:22-CV-161-L-MDD (S.D. Cal.).
- 3.139 *Specified Litigation Defendant* means any counterparty to the Debtor in the Specified Litigation.
- 3.140 *Specified Litigation Recoveries* means the Net Recovery, if any, realized by the Debtor or the Liquidating Trust from any or all of the Specified Litigation.
- 3.141 *Statutory Fees* means the fees payable pursuant to section 1930 of title 28 of the United States Code that were incurred in connection with the Chapter 11 Case.
- 3.142 **Subsequent Distribution Date** means any date after the Initial Distribution Date upon which the Liquidating Trust makes a Distribution to any Holders of Allowed Claims, as determined by the Liquidating Trustee or Co-Liquidating Trustee, as applicable.
- 3.143 *Tax* means any tax, charge, fee, levy, impost, or other assessment by any federal, state, local, or foreign taxing authority, including, without limitation, income,

4

5

6

7

10

9

111213

14 15

> 16 17

19

20

18

2122

2324

2526

2728

excise, property, sales, transfer, employment, payroll, franchise, profits, license, use, ad valorem, estimated, severance, stamp, occupation, and withholding tax.

- 3.144 *Third Party Releases* shall have the meaning provided below in Section 17.2(b).
- 3.145 *Transition Period* means the time between the Sale Effective Time and the occurrence of both (i) the CHOW Effective Date, and (ii) the dissolution of the Post-Effective Date Debtor.
- 3.146 *Transition Services Agreement* means the agreement between the Debtor and DAP Health in which the parties (or their successors-in interest, including the Post-Effective Date Debtor) agree to provide certain services and support after the Closing of the Sale pending the approval of the CHOW by CMS.
- 3.147 *Transfer (and any variations such as "Transferring")* means to, directly or indirectly, sell, convey, assign, pledge, encumber, hypothecate, gift, contribute, subject to a joint venture, partnership, or similar arrangement, abandon, convey, or transfer or otherwise dispose of, either voluntarily or involuntarily, any Asset or enter into any contract for any Asset that will effectuate the foregoing whether or not the foregoing is subject to approvals or conditions.
- 3.148 *Trust Assets Accounts* means interest-bearing bank account(s) or money-market account(s) to be established and held in trust by the Liquidating Trustee or Co-Liquidating Trustee, as applicable, on or after the Effective Date, for the purpose of holding the Liquidating Trust Assets to be distributed pursuant to the Plan and any interest, dividends, or other income earned upon the investment of the Liquidating Trust Assets.
- 3.149 *Trust Beneficial Interests* mean, collectively, (i) Class A Trust Beneficial Interests, and (ii) the Class B Trust Beneficial Interests. The Trust Beneficial Interests shall be evidenced as set forth in Section 13.3 and shall not be transferable, except to the limited extent provided in the Liquidating Trust Agreement.
- 3.150 *Unclassified Claims* means, collectively, Administrative Claims, Professional Claims, Statutory Fees, and Priority Tax Claims.
- 3.151 *Unimpaired Claim* means a Claim that is not impaired because the Plan leaves unaltered the legal, equitable, and contractual rights to which such Claim entitles the Holder of such Claim, as set forth in § 1124(1).
- 3.152 *U.S. Trustee* means the Office of the United States Trustee for the Southern District of California.

5

8

9

12 13

> 14 15 16

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

17 18

19 20

21 22

24 2.5

23

26 27

28

3.153 Vacatur Stipulation shall have the meaning provided below in Section 5.5.

- 3.154 Voting Instructions means the instructions for voting on the Plan contained on the Ballots.
- 3.155 *Wind-down Budget* means the budget (as the same may be amended or modified from time to time as provided in the Liquidating Trust Agreement) setting forth the projected costs and expenses associated with winding down the Debtor and the Estate and for the Liquidating Trustee and Co-Liquidating Trustee to discharge their duties under the Plan and the Liquidating Trust Agreement.

В. **Interpretation and Rules of Construction.**

Unless otherwise specified, all Section or exhibit references in the Combined Plan and Disclosure Statement are to the respective Section in, or exhibit to, the Combined Plan and Disclosure Statement, as the same may be amended, waived, or modified from time to time. The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Combined Plan and Disclosure Statement as a whole and not to any particular Section, subsection, or clause contained therein. The headings in the Combined Plan and Disclosure Statement are for convenience of reference only and shall not limit or otherwise affect the provisions hereof. For purposes herein: (1) in the appropriate context, each term, whether stated in the singular or the plural, shall include both the singular and the plural, and pronouns stated in the masculine, feminine, or neuter gender shall include the masculine, feminine, and the neuter gender; (2) any reference herein to a contract, lease, instrument, release, indenture, or other agreement or document being in a particular form or on particular terms and conditions means that the referenced document shall be substantially in that form or substantially on those terms and conditions; (3) the rules of construction set forth in § 102 shall apply; and (4) any term used in capitalized form herein that is not otherwise defined, but that is used in the Bankruptcy Code or the Bankruptcy Rules, shall have the meaning assigned to that term in the Bankruptcy Code or the Bankruptcy Rules, as the case may be.

C. **Controlling Document.**

The Combined Plan and Disclosure Statement (without reference to the Plan Supplement) shall govern and control in the event of an inconsistency between the terms and provisions in the Combined Plan and Disclosure Statement (without reference to the Plan Supplement), the Plan Supplement, any other instrument or document created or executed pursuant to the Combined Plan and Disclosure Statement, or any order (other than the Confirmation Order) referenced in the

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Combined Plan and Disclosure Statement (or any exhibit, schedules, appendices, supplements or amendments to any of the foregoing); <u>provided that</u>, notwithstanding anything herein to the contrary, the Confirmation Order shall govern and control in all respects in the event of a conflict between the Confirmation Order and any provision of the Combined Plan and Disclosure Statement or the Plan Supplement.

SECTION 4. BACKGROUND

4.1 Overview of the Debtor.

The Debtor is a non-profit public charity, tax-exempt under section 501(c)(3) of the Internal Revenue Code. The Debtor strives to be the community leader in improving the health of the populations in its service area, many of whom struggle with job and housing insecurity or disabilities. Its primary focus is the underserved, with an empowered workforce providing measurable quality and compassionate care to its patients. On the Petition Date, the Debtor had 24 brick and mortar sites including administrative sites, 2 pharmacies, and 6 mobile units covering a service area consisting of a 250-mile corridor on the eastern side of San Diego and Riverside Counties, CA. During 2021, the Debtor provided approximately 386,000 patient care visits. As of 2021, of Debtor's patients: (a) 94% had incomes below 200% of the FPL; (b) 71% lived in poverty; (c) approximately 75% were Medi-Cal (Medicaid) recipients or participate in other public health programs; (d) 36% were under the age 18; and (e) 93% were under the age of 65. The Debtor's services included comprehensive primary care, pediatric care, urgent care, behavioral health, dental services, specialty care, transgender health, women's health, prenatal care, veteran's health, chiropractic services, telehealth, and pharmacy. The Debtor was also an active partner in the training of medical residents, medical students, nurse practitioner students, physician assistant students, nursing students and other healthcare professionals.

As of the Petition Date, the Debtor operated as a federally qualified health center ("FQHC"). FQHCs are federally designated entities that receive federal grants and enhanced state payments to provide health care services to low-income and rural patients. The Debtor's health services were targeted to families with incomes below 200% of the Federal Poverty Level. As an FQHC, the Debtor strived to deliver high quality, comprehensive, compassionate primary health care to people in the surrounding area, regardless of ability to pay.

The Debtor also operated as a Federal Tort Claims Act ("FTCA") "deemed facility." Under section 224 of the Public Health Service Act, as amended by the

7

1

10

11

12

13

DENTONS US LLP

17

21

Federally Supported Health Centers Assistance Acts of 1992 and 1995, employees of eligible health centers like the Debtor may be deemed as federal employees for the purposes of liability protections under the FTCA for acts or omissions in the performance of medical, surgical, dental, or related functions resulting in personal injury, including death, and occurring within the scope of employment. Congress extended eligibility for FTCA protections to health centers like the Debtor in order to increase the availability of funds for health centers to provide primary health care services by reducing or eliminating health centers' malpractice insurance premiums.

Further information concerning the Debtor's operations is available in the Declaration of Isaac Lee, Chief Restructuring Officer, in Support of Debtor's Emergency First Day Motions [Docket No. 7] (the "First-Day Declaration").

4.2 Events Leading to Chapter 11 Filing.

In 2020, the Debtor's Board of Trustees (the "Board") became aware that members of the Debtor's leadership, certain landlords/contractors, and community dentists orchestrated what appears to be significant fraud for their own personal enrichment which involved filing false claims for dental services provided by contract dentists (i.e., the Prepetition Fraud). In November 2020, DHCS issued a temporary suspension of payments for Medi-Cal services because of an ongoing fraud investigation into the outside, contract dental program. There were no accusations of any fraudulent acts associated with the in-house dental program or medical services. During the pendency of the suspension by DHCS, the Debtor provided thousands of services without compensation.

Following the participation in a formal administrative meet and confer process, on February 26, 2021, the Debtor entered into a settlement agreement with DHCS, pursuant to which DHCS agreed to lift the temporary suspension as it pertained to the reimbursement of Medi-Cal medical services. The temporary payment suspension remained in place through the Petition Date with respect to dental services.

The Debtor took strong corrective actions and fully cooperated with state and federal investigators. Among other steps taken by the Debtor in the fall of 2020, the Debtor (a) terminated individual executives and others suspected of involvement in the apparent fraud; (b) removed any tainted members from the Board; (c) hired new leadership and added new board members with a high level of integrity and experience; (d) agreed to the appointment of a monitor, and fully cooperated with that monitor; (e) diligently completed corrective action plans; (f) initiated lawsuits against the bad actors; (g) created a corporate compliance department and instituted a robust compliance program; and (h) engaged reputable legal, IT, accounting, and financial

2

3

4

5

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

consultants to guide it on its path to full responsibility and integrity.

The Debtor initiated an internal investigation to identify wrongdoing by former associates, which led to the filing of pending litigation against the former trustees, officers, and contract dentists. The schemes included selling useless assets to the Debtor at inflated prices, entering into one-sided agreements with the Debtor to its detriment, committing and/or covering up healthcare fraud though improper billing of dental services, entering into leases with the Debtor that were many times fair market rates and terms, paying themselves above-market salaries and benefits, and hiring friends and family members to work for the Debtor and paying them abovemarket salaries.

While there were no accusations of any fraudulent acts associated with the inhouse dental program, as set forth above, the Medi-Cal suspension on the Debtor's in-house dental payments resulted in DHCS withholding more than \$7 million otherwise payable to the Debtor.

The Debtor commenced this Chapter 11 Case as a result of the issues discussed in this Section IV with the objective to maintain the Debtor's business operations; to preserve value for the Debtor, its stakeholders, and parties in interest; and, most importantly, to protect the health and wellbeing of the patients who are being treated at the facilities operated by the Debtor and the employees of the Debtor.

SECTION 5. THE CHAPTER 11 CASE

The following is a brief description of certain material events that have occurred during the Chapter 11 Case.

5.1 Material First-Day Motions Filed on the Petition Date.

a) Emergency Motion to Pay the Debtor's Prepetition Priority Wages

The Debtor filed an emergency motion [Docket No. 3] (the "Wage Motion") for authority to pay the Debtor's prepetition priority wages and related benefits in the ordinary course of business to avoid the disruption to the Debtor's business from failing to do so. The Bankruptcy Court granted the Wage Motion. [See Docket No. 20].

b) Emergency Motion to Maintain Cash Management Systems The Debtor filed an emergency motion [Docket No. 4] (the "Cash Management

Motion") for authority to maintain their cash management systems, which was imperative to avoid significant disruption to the Debtor's business operations. The Bankruptcy Court granted the Cash Management Motion. [See Docket. No. 28].

c) Emergency Motion to Provide Adequate Assurance of Payment to the Debtor's Utilities

The Debtor filed an emergency motion [Docket No. 5] (the "<u>Utilities Motion</u>") for an order authorizing the Debtor to provide adequate assurance of future payment to certain utility companies pursuant to § 366(c). The Bankruptcy Court granted the Utilities Motion. [See Docket No. 22].

d) Emergency Motion to Maintain Insurance Programs

The Debtor filed an emergency motion [Docket No. 6] (the "<u>Insurance Motion</u>") for authority to maintain insurance programs, pay premiums and other obligations in the ordinary course, and prevent insurance companies from enforcing ipso facto provisions or otherwise terminating insurance policies without first seeking relief from the automatic stay. The Bankruptcy Court granted the Insurance Motion. [See Docket No. 21].

e) <u>Emergency Motion to Set Insider Compensation</u>

The Debtor filed an emergency motion [Docket No. 11] (the "Motion to Set Insider Compensation") for authority to pay insiders their pre-petition salaries as they come due during the Chapter 11 Case. The Bankruptcy Court granted the Motion to Set Insider Compensation on an interim basis [see Docket No. 24] and later on a final basis [see Docket No. 90].

5.2 Motion for Entry of Order: Authorizing (I) Key Employee Retention Program and (II) Key Employee Incentive Program.

On December 7, 2022, the Debtor filed a motion [Docket No. 279] (the "KERP/KEIP Motion") seeking authorization of (a) a key employee retention plan and (b) a key employee incentive plan. The United States Trustee filed an objection to the KERP/KEIP Motion [Docket No. 324] and the Bankruptcy Court later approved the KERP/KEIP Motion on February 9, 2023 [Docket No. 436].

5.3 Estate Professionals, the Committee, and the Patient Care Ombudsman.

On September 27, 2022, the Bankruptcy Court entered an order approving the

6

7

4

10

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

DENTONS US LLP

23 24

25 26

27 28 employment of Kurtzman Carson Consultants LLC, as claims and noticing agent to the Debtor. [Docket No. 54]. On November 16, 2022, the Bankruptcy Court entered an order approving the employment of Ankura Consulting Group, LLC, as financial advisor to the Debtor. [Docket No. 176]. On November 18, 2022, the Bankruptcy Court entered an order [Docket No. 198] approving the employment of Hooper, Lundy & Bookman P.C., as special healthcare regulatory counsel. On December 13, 2022, the Bankruptcy Court entered an order [Docket No. 292] approving the employment of Dentons US LLP, as counsel to the Debtor.

Additionally, on December 6, 2022, the Debtor filed a motion [Docket No. 271] to employ various ordinary course professionals. On January 24, 2023, the Bankruptcy Court entered an order [Docket No. 400] granting the motion. Since the Petition Date, the Debtor has employed, pursuant to various filings, ordinary course professionals that provide an array of important services to the Debtor in the ordinary course of business, including legal, accounting, and consulting services.

On September 26, 2022, the U.S. Trustee appointed [Docket No. 49] an Official Committee of Unsecured Creditors (the "Committee") to represent the interests of general unsecured creditors. The Committee comprises the following seven members: (i) McKesson Corporation, (ii) Greenway Health, LLC, (iii) We Klean Inc., (iv) Mustafa Bilal, DDS, Inc., (v) Vista Village Family Dentistry, (vi) Vitamin D Public Relations, LLC, and (vii) Pourshirazi & Youssefi Dental Corporation. On December 9, 2022, the Bankruptcy Court entered an order [Docket No. 287] approving the employment of Pachulski Stang Ziehl & Jones LLP, as counsel to the Committee. On November 29, 2022, the Bankruptcy Court entered an order [Docket No. 242] approving the employment of FTI Consulting, Inc., as financial advisor to the Committee.

The U.S. Trustee appointed Dr. Jacob Nathan Rubin, MD, (the "Patient Care Ombudsman") to serve as the patient care ombudsman in this Chapter 11 Case, pursuant to § 333(a). [Docket No. 25]. On October 18, 2022, the Bankruptcy Court entered orders approving the employment of the following professionals to the Patient Care Ombudsman: Levene, Neale, Bender, Yoo & Brill LLP, as bankruptcy counsel [Docket No. 100]; and Dr. Tim Stacy DNP, ACNP-BC, as consultant [Docket No. 101]. The Patient Care Ombudsman has filed three reports in the Chapter 11 Case [Docket Nos. 169, 348, 560].

5.4 Administrative Matters, Reporting and Disclosures.

The Debtor was required to address the various administrative matters attendant to the commencement of this Chapter 11 Case, which required an extensive amount

5

10

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

DENTONS US LLP

16 17

18 19

20

21 22

23 24

25 26

27 28 of work by the Debtor's employees and its professionals. These matters included the preparation of the Schedules of Assets and Liabilities and Statements of Financial Affairs for the Debtor's Chapter 11 Case [see Docket Nos. 97, 98], and preparation of the materials required by the U.S. Trustee, including, without limitation, the 7-Day Package.

The Debtor has made every effort to comply with its duties under §§ 521, 1106 and 1107 and all applicable U.S. Trustee guidelines, including the filing of the Debtor's monthly operating reports with the U.S. Trustee. [Docket Nos. 172, 312, 432, 517.] The Debtor also attended its initial interview with the U.S. Trustee and the meeting of creditors required under § 341(a).

5.5 Adversary Proceeding against DHCS.

On September 26, 2022, the Debtor filed the Debtor's Complaint for Declaratory Judgment and Preliminary and Permanent Injunctive Relief, or in the Alternative, for Writ of Mandate Under Code of Civil Procedure 1085 [Docket No. 1] (the "Complaint"), commencing an adversary proceeding (Adv. Pro. No. 22-90056) against DHCS (the "Adversary Proceeding"). The Complaint sought, among other things, (i) an order temporarily and permanently enjoining DHCS from suspending the Debtor from the Medi-Cal program until and unless DHCS affords the Debtor the rights to which it is entitled under federal law and under the Constitution and (ii) a Writ of Mandate under Code of Civil Procedure 1085: (1) setting aside DHCS' suspension of the Debtor's Medi-Cal payments; (2) ordering DHCS to rescind any notices issued to third-parties, including but not limited to Medi-Cal health plans, directing or otherwise compelling them to (x) block transfer patients already assigned to the Debtor, and (y) assign patients that would otherwise be assigned to the Debtor to other providers; and (3) compel the payment of approximately \$6.7 million that was being withheld related to prior provision of in-house dental services to Medi-Cal beneficiaries.

On September 27, 2022, the Debtor filed its *Emergency Motion: (I) to Enforce* the Automatic Stay Pursuant to 11 U.S.C. § 362; or, Alternatively, (II) for Temporary Restraining Order; Memorandum of Points and Authorities in Support Thereof; and Declarations in Support Thereof [Adv. Dkt. No. 3] (the "Motion to Enforce") and certain other supporting declarations. The Motion to Enforce sought (A) the entry of an order enforcing the automatic stay to prevent DHCS from suspending all Medi-Cal payments and taking other related acts; or, alternatively, (2) the entry of order restraining and enjoining DHCS from causing immediate and irreparable harm to the Debtor, its estate, and thousands of patients by suspending all Medi-Cal payments and taking other related acts which would, inevitably, have caused the Debtor to close its

On October 26, 2022, the Bankruptcy Court issued its *Findings of Fact and Conclusions of Law re: Emergency Motion to (I) Enforce the Automatic Stay or (II) Alternatively for Temporary Restraining Order* [Adv. Dkt. No. 65] (the "DHCS Findings of Fact") and *Order on Emergency Motion to (I) Enforce the Automatic Stay or (II) Alternatively for Temporary Restraining Order* [Adv. Dkt. No. 66] (the "DHCS Order") granting, in part, the Motion to Enforce on the terms and conditions set forth in the DHCS Order. DHCS filed a *Notice of Appeal and Statement of Election*. [Adv. Dkt. No. 90] with respect to the DHCS Findings of Fact and DHCS Order, thereby commencing Case No. 22-CV-01751-GPC-MSB (S.D. Cal.) (the "Appeal").

On October 26, 2022, DHCS filed its *Answer to Complaint for Declaratory Judgment and Preliminary and Permanent Injunctive Relief, or in the Alternative, for Writ of Mandate Under Code of Civil Procedure 1085* [Adv. Dkt. No. 67].

On November 4, 2022, the Debtor, the Committee and DHCS entered into a stipulation (the "<u>Stipulation</u>") [Adv. Dkt. No. 73], to participate in a non-binding mediation to resolve all active disputes between the parties. On November 7, 2022, the Bankruptcy Court approved the Stipulation. [Adv. Dkt. No. 74].

On February 27, 2023, the Debtor filed a *Motion to Approve Compromise Among Debtor, Official Committee of Unsecured Creditors and California Department of Health Care Services Pursuant to Federal Rule of Bankruptcy Procedure 9019* [Docket No. 510] (the "Settlement Motion"). The Bankruptcy Court approved the Settlement Motion on March 7, 2023 [Docket No. 544] (the "DHCS 9019 Order"). The DHCS 9019 Order sets forth the treatment of the DHCS Claim and provides for the allocation of the DAP Sale proceeds among DHCS and Holders of other Allowed Claims.

On September 26, 2023, the Debtor filed the *Notice of Filing of Executed Settlement Agreement Among the Debtor, the Official Committee of Unsecured Creditors, and the California Department of Health Care Services [Docket Nos. 510, 544]* [Docket No. 923]. Pursuant to the DHCS 9019 Order, the Debtor, the Committee, and DHCS prepared and executed a finalized settlement agreement.

8

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

DENTONS US LLP

15

20

21

25

24

26 27 28

On October 17, 2023, the Debtor, DHCS, and the Committee jointly filed the Stipulation Among the Debtor, the California Department of Health Care Services and the Official Committee of Unsecured Creditors to (I) Vacate the (A) Findings of Fact and Conclusions of Law re: Emergency Motion to Enforce the Automatic Stay or Alternatively for Temporary Restraining Order [Docket No. 65] and (B) Order on Emergency Motion to Enforce the Automatic Stay or Alternatively for Temporary Restraining Order [Docket No. 66]; and (II) Dismiss the Adversary Proceeding [Adv. Docket No. 133] (the "Vacatur Stipulation") pursuant to the DHCS Settlement Agreement. On October 18, 2023, the Bankruptcy Court entered an order approving the Vacatur Stipulation [Adv. Docket No. 134] and (a) dismissing the Adversary Proceeding with prejudice and (b) vacating the DHCS Findings of Fact and DHCS Order.

On November 6, 2023, the Debtor and DHCS filed the Joint Stipulation to Dismiss the Appeal [App. Docket No. 12] (the "Appeal Stipulation") pursuant to the DHCS Settlement Agreement. On November 7, 2023, the District Court entered an order approving the Appeal Stipulation [App. Docket No. 13] and dismissing the Appeal.

5.6 The Sale of Substantially All Assets.

On November 10, 2022, the Debtor filed a motion [Docket No. 161] (the "Sale Motion") requesting entry of an order (i) authorizing the proposed sale of substantially all of the Debtor's assets, (ii) approving the form of the Asset Purchase Agreement (the "APA"), (iii) approving certain procedures governing the sale process (the "Bid Procedures"), and (iv) approving certain procedures governing assumption and rejection of Executory Agreements in connection with the sale. On December 5, 2022, the Debtor filed a supplement to the Sale Motion with revised Bid Procedures [Docket No. 276]. On December 8, 2022, the Debtor filed a second supplement to the Sale Motion, which attached a form APA [Docket No. 285].

On December 19, 2022, the Bankruptcy Court entered an order [Docket No. 321] approving the Bid Procedures. The order provided that all objections to the proposed Bid Procedures were overruled and that remaining objections concerning the proposed sale were premature.

On January 16, 2023, the Debtor filed a notice [Docket No. 389] to counterparties of Executory Agreements that may be assumed and assigned in connection with the sale. The Debtor filed a supplemental notice [Docket No. 409] on January 27, 2023. Certain counterparties to executory agreements filed objections (collectively, the "Cure Objections") to the notices concerning assumption and

assignment. [See Docket Nos. 426, 431, 440, 441, 445, 447, 455, 458, 487].

On February 2, 2023, the Debtor filed a notice [Docket No. 418] that the Debtor received Qualified Bids pursuant to the Bid Procedures and selected designated DAP Health as the stalking horse bidder. The notice further provided that the Debtor would conduct an auction on February 6, 2023.

On February 9, 2023, the Debtor filed a notice [Docket No. 438] that the Debtor had adjourned the auction and requested final highest and best bids by February 10, 2023. The next day, on February 10, 2023, the Bankruptcy Court entered an order vacating deadlines for objections to the sale [Docket No. 443].

On February 15, 2023, the Debtor filed a notice [Docket No. 465] that the Debtor had selected (i) DAP Health as the winning bidder, and (ii) Altamed Health Services Corporation as the back-up bidder.

On February 16, 2023, the Bankruptcy Court entered an order approving modified deadlines and scheduling a hearing to approve the sale. [Docket No. 476].

On February 16, 2023, the Debtor filed a notice [Docket No. 478] to counterparties to Executory Agreements designated by DAP Health for assumption and assignment in connection with the sale.

On March 1, 2023, the Bankruptcy Court held a hearing to approve the DAP Sale pursuant to the Sale Motion. At the hearing, the Bankruptcy Court considered the Cure Objections as well as certain objections (collectively, the "Sale Objections") to the DAP Sale as well as any withdrawals thereof. [See Docket Nos. 270, 356, 489, 491]. As set forth in further detail below, the Bankruptcy Court overruled the Sale Objections.

On March 13, 2023, the Bankruptcy Court entered an order [Docket No. 559] granting the Sale Motion and approving the DAP Sale (the "Sale Order"). In connection with the DAP Sale, the Debtor and DAP Health entered into that certain Transition Services Agreement, wherein the parties agree to provide certain services and support after the Closing of the Sale pending the approval of the CHOW by CMS. Subsequent to approval of the CHOW, the Post-Effective Date Debtor will be dissolved wherein the Debtor provides certain services to DAP Health.

On July 31, 2023, the Debtor filed the *Notice of Occurrence of Closing of Sale to DAP Health, Inc.* [Docket No. 823], which informed the Bankruptcy Court and all parties in interest of the occurrence of the Closing Date.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

After the Effective Date and Closing of the Sale, the Post-Effective Date Debtor, as described herein, will need to continue to operate until the CHOW submitted, pursuant to 42 C.F.R § 489.18, by the Debtor and DAP Health is approved by CMS, which will result in the transfer of the Debtor's Medicare Identification Number and Medicare Provider Agreement to DAP Health. After the Closing and Effective Date, DAP Health and the Post-Effective Date Debtor will operate pursuant to the Transition Services Agreement.

5.7 Claims Bar Date and Reconciliation.

a) General Bar Date

On September 13, 2022, the Bankruptcy Court set a general claims bar date of November 21, 2022, and March 13, 2023, as the deadline for governmental entities to file claims [Docket No. 16]. A Notice of Bar Date was published on October 27, 2022, and October 28, 2022, in the Los Angeles Times, the San Diego Union-Tribune, The Desert Sun, The Press-Enterprise, the El Latino, the La Prensa Hispana Bilingual News Paper, the San Bernadino County Sun, and the San Diego Union-Tribune Español. [Docket No. 157].

b) Extended DHCS Bar Date

Pursuant to the DHCS Settlement Agreement, the Extended DHCS Bar Date is set for December 29, 2023, which is the deadline by which DHCS must file a Proof of Claim for any further General Unsecured Claims against the Debtor for Medi-Cal overpayments, consistent with the definition in the DHCS Settlement Agreement.

c) Administrative Claims Bar Date

The Plan contemplates that a deadline will be set by the Bankruptcy Court by which Holders of Administrative Claims, other than Administrative Claims arising in the ordinary course of business for the Debtor or Professional Claims, must assert Administrative Claims or be forever barred, which shall be (a) thirty (30) days after the Effective Date for Administrative Claims other than Professional Claims, and (b) sixty (60) days after the Effective Date for Professional Claims. Such requests for payment may include estimates of amounts through the Effective Date of the Plan.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

19

20

21

22

23

24

25

26

27

28

SECTION 6. CONFIRMATION AND VOTING PROCEDURES

6.1 Confirmation Hearing.

On [___], the Bankruptcy Court entered the Conditional Approval and Procedures Order. The Confirmation Hearing has been scheduled for January 17, 2024, at 10:00 a.m. (prevailing Pacific Time) to consider (a) final approval of the Combined Plan and Disclosure Statement as providing adequate information pursuant to § 1125 and (b) confirmation of the Combined Plan and Disclosure Statement pursuant to § 1129. The Confirmation Hearing may be adjourned from time to time by the Debtor without further notice, except for an announcement of the adjourned date made at the Confirmation Hearing or by filing a notice with the Bankruptcy Court.

6.2 Procedures for Objections.

Any objection to final approval of the Combined Plan and Disclosure Statement as providing adequate information pursuant to § 1125 and/or confirmation of the Combined Plan and Disclosure Statement must be made in writing and filed with the Bankruptcy Court by no later than January 8, 2024, at 4:00 p.m. (prevailing Pacific Time) and be served in accordance with the Local Bankruptcy Rules of the Bankruptcy Court on the following parties: (i) counsel to the Debtor: Dentons US LLP, 601 South Figueroa Street, Suite 2500, Los Angeles, California 90017 (Attn: Samuel R. Maizel (samuel.maizel@dentons.com) and Tania M. Moyron (tania.moyron@dentons.com)); (ii) financial advisor to the Debtor: Ankura, 2021 McKinney Avenue, Suite 340, Dallas, Texas 75201 (Attn: Charles Pease (charles.pease@ankura.com)); (iii) the Office of the United States Trustee: 880 Front San Diego, California Street, Room 3230, 92101 (Attn: Haeji Hong (haeji.hong@usdoj.gov)); (iv) counsel to the Committee: Pachulski Stang Ziehl & Jones LLP, 10100 Santa Monica Blvd., 13th Floor, Los Angeles, California 90067 (Attn: Jeffrey N. Pomerantz (jpomerantz@pszjlaw.com) and Steven W. Golden (sgolden@pszjlaw.com)); and (v) financial advisor to the Committee: FTI, 350 South Grand Avenue, Suite 3000, Los Angeles, California 90071 (Attn: Cliff Zucker (cliff.zucker@fticonsulting.com) and Narendra Ganti (narendra.ganti@fticonsulting.com)). Unless an objection is timely filed and served, it may not be considered by the Bankruptcy Court at the Confirmation Hearing.

6.3 Requirements for Confirmation.

The Bankruptcy Court will confirm the Combined Plan and Disclosure Statement only if it meets all the applicable requirements of § 1129. Among the

7

5

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

15

21

24

requirements for confirmation in the Chapter 11 Case is that the Combined Plan and Disclosure Statement be: (i) accepted by all Impaired Classes of Claims or, if rejected by an Impaired Class, that the Combined Plan and Disclosure Statement "does not discriminate unfairly" against, and is "fair and equitable" with respect to, such Class; and (ii) feasible. The Bankruptcy Court must also find that:

- a. the Combined Plan and Disclosure Statement has classified Claims in a permissible manner;
- b. the Combined Plan and Disclosure Statement complies with the requirements of Chapter 11 of the Bankruptcy Code; and
- c. the Combined Plan and Disclosure Statement has been proposed in good faith.

The Plan Proponents believe that the Combined Plan and Disclosure Statement complies, or will comply, with all such requirements.

6.4 Classification of Claims.

Section 1123 provides that a plan must classify the claims of a debtor's creditors. In accordance with § 1123, the Combined Plan and Disclosure Statement divides Claims into Classes and sets forth the treatment for each Class (other than those claims which pursuant to § 1123(a)(1) need not be and have not been classified).

Section 1122 requires the Combined Plan and Disclosure Statement to place a Claim in a particular Class only if such Claim is substantially similar to the other Claims in such class. The Combined Plan and Disclosure Statement creates separate Classes to deal respectively with Priority Non-Tax Claims, various Secured Claims, General Unsecured Claims, and an Allowed DHCS Claim. The Plan Proponents believe that the Combined Plan and Disclosure Statement's classifications place substantially similar Claims in the same Class and, thus, meet the requirements of § 1122.

The Bankruptcy Code also requires that a plan provide the same treatment for each claim of a particular class unless the claim holder agrees to a less favorable treatment of its claim. The Plan Proponents believe that the Combined Plan and Disclosure Statement complies with such standard. If the Bankruptcy Court finds otherwise, however, it could deny confirmation of the Combined Plan and Disclosure Statement if the holders of Claims affected do not consent to the treatment afforded them under the Combined Plan and Disclosure Statement.

A Claim is placed in a particular Class only to the extent that the Claim falls within the description of that Class and is classified in other Classes to the extent that any portion of the Claim falls within the description of such other Classes. A Claim also is placed in a particular Class for the purpose of receiving distributions pursuant to the Combined Plan and Disclosure Statement only to the extent that such Claim is an Allowed Claim in that Class, and such Claim has not been paid, released, or otherwise settled prior to the Effective Date.

The Plan Proponents believe that the Combined Plan and Disclosure Statement has classified all Claims in compliance with the provisions of § 1122 and applicable case law. It is possible that a holder of a Claim may challenge the Plan Proponents' classification of Claims and that the Bankruptcy Court may find that a different classification is required for the Combined Plan and Disclosure Statement to be confirmed. If such a situation develops, the Plan Proponents intend, in accordance with the terms of the Combined Plan and Disclosure Statement, to make such permissible modifications to the Combined Plan and Disclosure Statement as may be necessary to permit its confirmation. Any such reclassification could adversely affect holders of Claims by changing the composition of one or more Classes and the vote required of such Class or Classes for approval of the Combined Plan and Disclosure Statement.

EXCEPT AS SET FORTH IN THE COMBINED PLAN AND DISCLOSURE STATEMENT, UNLESS SUCH MODIFICATION OF CLASSIFICATION MATERIALLY ADVERSELY AFFECTS THE TREATMENT OF A HOLDER OF A CLAIM AND REQUIRES RE-SOLICITATION, ACCEPTANCE OF THE COMBINED PLAN AND DISCLOSURE STATEMENT BY ANY HOLDER OF A CLAIM PURSUANT TO THIS SOLICITATION WILL BE DEEMED TO BE A CONSENT TO THE COMBINED PLAN AND DISCLOSURE STATEMENT'S TREATMENT OF SUCH HOLDER OF A CLAIM REGARDLESS OF THE CLASS AS TO WHICH SUCH HOLDER ULTIMATELY IS DEEMED TO BE A MEMBER.

The amount of any Impaired Claim that ultimately is Allowed by the Bankruptcy Court may vary from any estimated Allowed amount of such Claim and, accordingly, the total Claims that are ultimately Allowed by the Bankruptcy Court with respect to each Impaired Class of Claims may also vary from any estimates contained herein with respect to the aggregate Claims in any Impaired Class. Thus, the actual recovery ultimately received by a particular holder of an Allowed Claim may be adversely or favorably affected by the aggregate amount of Claims Allowed in the applicable Class. Additionally, any changes to any of the assumptions

4 | 5 | 6 |

8 9

7

1112

10

1314

15

16 17

19 20

18

2122

23

242526

27

28

underlying the estimated Allowed amounts could result in material adjustments to recovery estimates provided herein or the actual Distribution received by creditors. The projected recoveries are based on information available to the Debtor as of the date hereof and reflect the Debtor's view as of the date hereof only.

The classification of Claims and the nature of distributions to members of each Class are summarized herein. The Plan Proponents believe that the consideration, if any, provided under the Combined Plan and Disclosure Statement to holders of Allowed Claims reflects an appropriate resolution of their Allowed Claims taking into account the differing nature and priority of such Claims. The Bankruptcy Court must find, however, that a number of statutory tests are met before it may confirm the Combined Plan and Disclosure Statement. Many of these tests are designed to protect the interests of holders of Claims who are not entitled to vote on the Combined Plan and Disclosure Statement, or do not vote to accept the Combined Plan and Disclosure Statement, but who will be bound by the provisions of the Combined Plan and Disclosure Statement if it is confirmed by the Bankruptcy Court.

6.5 Impaired Claims.

Pursuant to § 1126, only the Holders of Claims in Classes Impaired by the Combined Plan and Disclosure Statement and receiving a payment or Distribution under the Combined Plan and Disclosure Statement may vote to accept or reject the Combined Plan and Disclosure Statement. Pursuant to § 1124, a Class of Claims may be Impaired if the Combined Plan and Disclosure Statement alters the legal, equitable, or contractual rights of the holders of such Claims treated in such Class. The Holders of Claims not Impaired by the Combined Plan and Disclosure Statement are deemed to accept the Combined Plan and Disclosure Statement and do not have the right to vote on the Combined Plan and Disclosure Statement. The Holders of Claims in any Class which will not receive any payment or Distribution or retain any property pursuant to the Combined Plan and Disclosure Statement are deemed to reject the Combined Plan and Disclosure Statement and do not have the right to vote. Finally, the Holders of Claims whose Claims are not classified under the Combined Plan and Disclosure Statement are not entitled to vote on the Combined Plan and Disclosure Statement.

Under the Combined Plan and Disclosure Statement, Holders of Claims in the Voting Classes – Class 3 (General Unsecured Claims) and Class 4 (Allowed DHCS Claim) – are Impaired and are entitled to vote to accept or reject the Combined Plan and Disclosure Statement. Holders of Claims in Class 1 (Priority Non-Tax Claims) and Class 2 (Secured Claims) are Unimpaired and, therefore, not entitled to vote on

$\begin{bmatrix} 3 & \mathbf{O} \\ 4 & \mathbf{P} \end{bmatrix}$

6

7

5

9 10

111213

15 16 17

14

19 20

18

2223

21

24

2526

27

28

the Combined Plan and Disclosure Statement and are deemed to accept the Combined Plan and Disclosure Statement.

ACCORDINGLY, A BALLOT FOR ACCEPTANCE OR REJECTION OF THE COMBINED PLAN AND DISCLOSURE STATEMENT IS BEING PROVIDED ONLY TO HOLDERS OF CLAIMS IN THE VOTING CLASSES.

6.6 Confirmation Without Necessary Acceptances; Cramdown.

In the event that any impaired class of claims does not accept a plan, a debtor nevertheless may move for confirmation of the plan. A plan may be confirmed, even if it is not accepted by all impaired classes, if the plan has been accepted by at least one impaired class of claims, determined without including any acceptance of the plan by any insider holding a claim in that class, and the plan meets the "cramdown" requirements set forth in § 1129(b). Section 1129(b) requires that a court find that a plan (a) "does not discriminate unfairly" and (b) is "fair and equitable," with respect to each non-accepting impaired class of claims. Here, the Plan Proponents believe that such requirements are satisfied, as no holder of a Claim or Interest junior to those in the Impaired Classes is entitled to receive any property under the Combined Plan and Disclosure Statement.

The concept of "unfair discrimination" is not defined in the Bankruptcy Code, but case law suggests it exists when a difference in a plan's treatment of two classes of equal priority results in a materially lower percentage recovery for the non-accepting class. The Plan Proponents do not believe that the Plan unfairly discriminates against any Class of Claims.

The Bankruptcy Code provides a nonexclusive definition of the phrase "fair and equitable." To determine whether a plan is "fair and equitable," the Bankruptcy Code establishes "cramdown" tests for secured creditors, unsecured creditors, and equity holders, as follows:

a. <u>Secured Creditors</u>. Either (i) each impaired secured creditor retains its liens securing its secured claim and receives on account of its secured claim deferred Cash payments having a present value equal to the amount of its allowed secured claim, (ii) each impaired secured creditor realizes the "indubitable equivalent" of its allowed secured claim or (iii) the property securing the claim is sold free and clear of liens with such liens to attach to the proceeds of the sale and the treatment of such liens on proceeds to be as provided in clause (i) or (ii) above.

67

8

5

13 14 15

16171819

2223

24

2.5

20

21

2627

28

b. <u>Unsecured Creditors</u>. Either (i) each impaired unsecured creditor receives or retains under the plan property of a value equal to the amount of its allowed claim or (ii) the holders of claims that are junior to the claims of the dissenting class will not receive any property under the plan.

As discussed above, the Plan Proponents believe that the distributions provided under the Combined Plan and Disclosure Statement satisfy the absolute priority rule, where required.

6.7 Feasibility.

Section 1129(a)(11) requires that confirmation of a plan not be likely to be followed by the liquidation, or the need for further financial reorganization, of the Debtor or any successor to the Debtor (unless such liquidation or reorganization is proposed in the Combined Plan and Disclosure Statement). Based on the Debtor's analysis and subject to the financing contingency described below, the Liquidating Trustee and Co-Liquidating Trustee will have sufficient assets to accomplish their tasks under the Combined Plan and Disclosure Statement. Therefore, the Plan Proponents believe that the liquidation pursuant to the Combined Plan and Disclosure Statement will meet the feasibility requirements of the Bankruptcy Code.

6.8 Best Interests Test and Liquidation Analysis.

Even if a plan is accepted by the holders of each class of claims, the Bankruptcy Code requires the Bankruptcy Court to determine that such plan is in the best interests of all holders of claims that are impaired by that plan and that have not accepted the plan. The "best interests" test, as set forth in § 1129(a)(7), requires a court to find either that all members of an impaired class of claims have accepted the plan or that the plan will provide a member who has not accepted the plan with a recovery of property of a value, as of the effective date of the plan, that is not less than the amount that such holder would recover if the debtor were liquidated under chapter 7 of the Bankruptcy Code.

To calculate the probable distribution to holders of each impaired class of claims if the debtor was liquidated under chapter 7, a court must first determine the aggregate dollar amount that would be generated from a debtor's assets if its chapter 11 case was converted to a case under chapter 7 of the Bankruptcy Code. To determine if a plan is in the best interests of each impaired class, the present value of the distributions from the proceeds of a liquidation of the debtor's unencumbered assets and properties, after subtracting the amounts attributable to the costs, expenses, and administrative claims associated with a chapter 7 liquidation, must be compared with

1

5

4

8 9 10

7

12 13

11

14 15

161718

1920

21

222324

242526

27

28

the value offered to such impaired classes under the plan. If the hypothetical liquidation distribution to holders of claims in any impaired class is greater than the distributions to be received by such parties under the plan, then such plan is not in the best interests of the holders of claims in such impaired class.

The Debtor, with the assistance of its advisors, has prepared a liquidation analysis that summarizes the Debtor's best estimate of recoveries by holders of Claims if the Chapter 11 Case was converted to a case under chapter 7 (the "Liquidation Analysis"), which is attached hereto as **Exhibit A**.

Based upon the Debtor's current projections, Holders of Allowed Administrative Claims, Allowed Secured Claims, Allowed General Unsecured Claims, and a portion of the Allowed DHCS Claim, will be paid in full on the Effective Date or as soon as practicably thereafter. *See Exhibit A* (Liquidation Analysis Supplement) attached hereto.

Based upon <u>Exhibit A</u> (Liquidation Analysis Supplement) attached hereto, the Plan Proponents believe that creditors will receive at least as much or more under the Plan than they would receive if the Chapter 11 Case was converted to a chapter 7 case.

6.9 Eligibility to Vote on the Combined Plan and Disclosure Statement.

Unless otherwise ordered by the Bankruptcy Court, only Holders of Allowed Claims in the Voting Classes may vote on the Combined Plan and Disclosure Statement. Further, subject to the tabulation procedures that were approved by the Conditional Approval and Procedures Order, in order to vote on the Combined Plan and Disclosure Statement, Creditors must hold an Allowed Claim in the Voting Classes, or be the Holder of a Claim that has been temporarily Allowed for voting purposes only pursuant to the approved tabulation procedures or under Bankruptcy Rule 3018(a).

6.10 Solicitation Package / Release Opt-Out Election Form.

All Holders of Allowed Claims in the Voting Classes will receive a solicitation package (the "Solicitation Package"). The Solicitation Packages will contain: (i) the Combined Plan and Disclosure Statement; (ii) the Conditional Approval and Procedures Order; (iii) notice of the Confirmation Hearing; (iv) a form of Ballot, including Voting Instructions and a pre-addressed return envelope; (v) a Release Opt-Out Election Form; and (vi) such other materials as the Bankruptcy Court may direct or approve or that the Debtor deems appropriate.

All other Creditors and parties in interest not entitled to vote on the Combined Plan and Disclosure Statement will receive only a copy of the notice of Confirmation Hearing and a Release Opt-Out Election Form.

Copies of the Combined Plan and Disclosure Statement shall be available on the Claims and Balloting Agent's website at https://www.kccllc.net/BorregoHealth. Any creditor or party-in-interest can request a hard copy of the Combined Plan and Disclosure Statement be sent to them by regular mail by calling the Claims and Balloting Agent at (866) 967-0670 (U.S. & Canada) during regular business hours.

IN ADDITION TO OTHER PARTIES WHO WILL BE CONSIDERED RELEASING PARTIES, ANY HOLDER OF A CLAIM THAT DOES NOT AFFIRMATIVELY OPT-OUT OF THE THIRD PARTY RELEASE CONTAINED IN SECTION 17 HEREOF BY TIMELY AND PROPERLY COMPLETING AND RETURNING A RELEASE OPT-OUT ELECTION FORM WILL BE CONSIDERED A RELEASING PARTY IN RELATION TO THE THIRD PARTY RELEASE UNDER THE COMBINED PLAN AND DISCLOSURE STATEMENT.

6.11 Voting Procedures, Voting Deadline, and Deadline to Submit the Release Opt-Out Election.

The Voting Record Date for determining which Holders of Claims in the Voting Classes may vote on the Combined Plan and Disclosure Statement is November 28, 2023.

In order for a Creditor's Ballot to count, the Creditor must (1) complete, date, and properly execute the Ballot and (2) properly deliver the Ballot to the Claims and Balloting Agent by either (a) mailing the Ballot via First Class Mail to the Claims and Balloting Agent at the following address: Borrego Health Ballot Processing c/o Kurtzman Carson Consultants LLC, 222 N. Pacific Coast Highway, Suite 300, El Segundo, CA 90245; (b) overnight delivery; (c) hand delivery; or (d) submitting the Ballot by electronically uploading the Ballot on the Claims and Balloting Agent's online balloting platform at https://www.kccllc.net/BorregoHealth. Instructions for casting a Ballot will be available on the Claims and Balloting Agent's website.

Ballots must be submitted electronically, or the Claims and Balloting Agent must actually receive physical, original Ballots by mail, overnight, or hand delivery, on or before the Voting Deadline, which is **January 8, 2024, at 4:00 p.m.** (prevailing **Pacific Time**). Subject to the tabulation procedures approved by the Conditional Approval and Procedures Order, Creditors may not change their vote once a Ballot is submitted electronically or the Claims and Balloting Agent receives their original

2

3

4

5

6

7

8

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

paper Ballot. Subject to the tabulation procedures approved by the Conditional Approval and Procedures Order, any Ballot that is timely and properly submitted electronically or received physically will be counted and will be deemed to be cast as an acceptance, rejection, or abstention, as the case may be, of the Combined Plan and Disclosure Statement.

In order to be effective, Release Opt-Out Election Form for Holders of Claims entitled to opt out of being a Releasing Party in connection with the Third Party Release contained in Section 17.2(b) must be received by the Claims and Balloting Agent by the Voting Deadline, which is **January 8, 2024, at 4:00 p.m.** (prevailing Pacific Time). Each Release Opt-Out Election Form must be properly delivered to the Claims and Balloting Agent by either (a) mailing via First Class mail the Release Opt-Out Election Form to the Claims and Balloting Agent at the following address: Borrego Health Ballot Processing c/o Kurtzman Carson Consultants LLC, 222 N. Pacific Coast Highway, Suite 300, El Segundo, CA 90245; (b) overnight courier; (c) hand delivery; or (d) uploading the Release Opt-Out Election Form on the Claims and Balloting Agent's online opt-out portal at https://www.kccllc.net/BorregoHealth.

IF YOU ARE ENTITLED TO VOTE ON THE COMBINED PLAN AND DISCLOSURE STATEMENT, YOU ARE URGED TO COMPLETE, DATE, SIGN, AND PROMPTLY MAIL OR UPLOAD THE BALLOT YOU RECEIVE. PLEASE BE SURE TO COMPLETE ALL BALLOT ITEMS PROPERLY AND LEGIBLY. IF YOU ARE A HOLDER OF A CLAIM ENTITLED TO VOTE ON THE COMBINED PLAN AND DISCLOSURE STATEMENT AND YOU DID NOT RECEIVE A BALLOT, YOU RECEIVED A DAMAGED BALLOT, OR YOU LOST YOUR BALLOT, OR IF YOU HAVE **OUESTIONS CONCERNING** ANY THE **COMBINED PLAN** DISCLOSURE STATEMENT OR PROCEDURES FOR VOTING ON THE COMBINED PLAN AND DISCLOSURE STATEMENT, PLEASE CONTACT THE CLAIMS AND BALLOTING AGENT BY (I) TELEPHONE AT (888) 647-174 (U.S./CANADA) OR (II) EMAIL AT BorregoHealthinfo@kccllc.com. THE CLAIMS AND BALLOTING AGENT IS NOT AUTHORIZED TO, AND WILL NOT, PROVIDE LEGAL ADVICE.

6.12 Acceptance of the Combined Plan and Disclosure Statement.

If you are a Holder of a Claim in one of the Voting Classes, your acceptance of the Combined Plan and Disclosure Statement is important. In order for the Combined Plan and Disclosure Statement to be accepted by an Impaired Class of Claims, a majority in number (*i.e.*, more than half) and two-thirds in dollar amount of the Claims voting (of each Impaired Class of Claims) must vote to accept the Combined Plan and Disclosure Statement. At least one Impaired Class of Creditors, excluding the votes

1

6

5

4

67

8 9

101112

13

1415

16

17 18

19

202122

23

2425

26

27

28

of insiders, must actually vote to accept the Combined Plan and Disclosure Statement. The Debtor urges that you vote to accept the Combined Plan and Disclosure Statement.

SECTION 7. CERTAIN RISK FACTORS, TAX CONSEQUENCES, AND OTHER DISCLOSURES

7.1 Certain Risk Factors to be Considered.

THE COMBINED PLAN AND DISCLOSURE STATEMENT AND ITS IMPLEMENTATION ARE SUBJECT TO CERTAIN RISKS, INCLUDING, BUT NOT LIMITED TO, THE RISK FACTORS SET FORTH BELOW. HOLDERS OF CLAIMS WHO ARE ENTITLED TO VOTE ON THE COMBINED PLAN AND DISCLOSURE STATEMENT SHOULD READ AND CAREFULLY CONSIDER THE RISK FACTORS, AS WELL AS THE OTHER INFORMATION SET FORTH IN THE COMBINED PLAN AND DISCLOSURE STATEMENT AND THE DOCUMENTS DELIVERED TOGETHER HEREWITH OR REFERRED TO OR INCORPORATED BY REFERENCE HEREIN, BEFORE DECIDING WHETHER TO VOTE TO ACCEPT OR REJECT THE COMBINED PLAN AND DISCLOSURE STATEMENT. THESE FACTORS SHOULD NOT, HOWEVER, BE REGARDED AS CONSTITUTING THE ONLY RISKS INVOLVED IN **PLAN** CONNECTION WITH THE **COMBINED** AND **DISCLOSURE** STATEMENT AND ITS IMPLEMENTATION.

(a) <u>The Combined Plan and Disclosure Statement May Not Be Accepted.</u>

The Plan Proponents can make no assurances that the requisite acceptances of the Combined Plan and Disclosure Statement will be received, and the Debtor may need to obtain acceptances of an alternative plan for the Debtor, or otherwise, that may not have the support of the creditors and/or may be required to liquidate the Estate under chapter 7. There can be no assurance that the terms of any alternative arrangement or plan would be similar to or as favorable to creditors as those proposed in the Combined Plan and Disclosure Statement.

(b) <u>The Combined Plan and Disclosure Statement May Not Be</u> <u>Confirmed.</u>

Even if the Plan Proponents receive the requisite acceptances, there is no assurance that the Bankruptcy Court, which may exercise substantial discretion as a court of equity, will confirm the Combined Plan and Disclosure Statement. Even if

1

11 12 13

141516

18 19 20

17

21 22

23

24

252627

28

the Bankruptcy Court determined that the Combined Plan and Disclosure Statement and the balloting procedures and results were appropriate, the Bankruptcy Court could still decline to confirm the Combined Plan and Disclosure Statement if it finds that any of the statutory requirements for confirmation had not been met. As is described in greater detail in Section 6.3, § 1129 sets forth the requirements for confirmation of a chapter 11 plan. While, as more fully set forth Section 6, the Plan Proponents believe that the Combined Plan and Disclosure Statement complies with or will comply with all such requirements, there can be no guarantee that the Bankruptcy Court will agree.

Moreover, there can be no assurance that modifications to the Combined Plan and Disclosure Statement will not be required for Confirmation or that such modifications would not necessitate the re-solicitation of votes. If the Combined Plan and Disclosure Statement is not confirmed, it is unclear what distributions holders of Claims ultimately would receive with respect to their Claims in a subsequent plan of liquidation. If an alternative could not be agreed to, it is possible that the Debtor would have to liquidate its remaining assets in chapter 7, in which case it is likely that the Holders of Allowed Claims would receive substantially less favorable treatment than they would receive under the Combined Plan and Disclosure Statement.

(c) <u>Distributions to Holders of Allowed Claims Under the Combined</u> <u>Plan and Disclosure Statement May be Inconsistent with</u> <u>Projections.</u>

Projected Distributions are based upon good faith estimates of the total amount of Claims ultimately Allowed and the funds available for Distribution. There can be no assurance that the estimated Claim amounts set forth in the Combined Plan and Disclosure Statement are correct. These estimated amounts are based on certain assumptions with respect to a variety of factors. Both the actual amount of Allowed Claims in a particular Class and the funds available for distribution to such Class may differ from the Debtor's estimates. If the total amount of Allowed Claims in a Class is higher than the Debtor's estimates, or the funds available for distribution to such Class are lower than the Debtor's estimates, the percentage recovery to Holders of Allowed Claims in such Class will be less than projected.

(d) Objections to Classification of Claims.

Section 1122 provides that a plan may place a claim in a particular class only if such claim is substantially similar to the other claims in such class. As is described in greater detail in Section 6.4, the Plan Proponents believe that the classification of Claims under the Combined Plan and Disclosure Statement complies with the requirements set forth in the Bankruptcy Code. Nevertheless, there can be no assurance the Bankruptcy Court will reach the same conclusion.

2

3

4

5

7

9

10

18

19

20

21

22

23

24

25

26

27

28

To the extent that the Bankruptcy Court finds that a different classification is required for the Combined Plan and Disclosure Statement to be confirmed, the Plan Proponents would seek to (i) modify the Combined Plan and Disclosure Statement to provide for whatever classification might be required for Confirmation and (ii) use the acceptances received from any Holder of Claims pursuant to this solicitation for the purpose of obtaining the approval of the Class or Classes of which such holder ultimately is deemed to be a member. Any such reclassification of Claims, although subject to the notice and hearing requirements of the Bankruptcy Code, could adversely affect the Class in which such Holder was initially a member, or any other Class under the Combined Plan and Disclosure Statement, by changing the composition of such Class and the vote required for approval of the Combined Plan and Disclosure Statement. There can be no assurance that the Bankruptcy Court, after finding that a classification was inappropriate and requiring a reclassification, would approve the Combined Plan and Disclosure Statement based upon such reclassification. Except to the extent that modification of classification in the Combined Plan and Disclosure Statement requires re-solicitation, the Plan Proponents will, in accordance with the Bankruptcy Code and the Bankruptcy Rules, seek a determination by the Bankruptcy Court that acceptance of the Combined Plan and Disclosure Statement by any Holder of Claims pursuant to this solicitation will constitute a consent to the Combined Plan and Disclosure Statement's treatment of such Holder, regardless of the Class as to which such holder is ultimately deemed to be a member. The Plan Proponents believe that under the Bankruptcy Rules, they would be required to resolicit votes for or against the Combined Plan and Disclosure Statement only when a modification adversely affects the treatment of the Claim of any Holder.

The Bankruptcy Code also requires that the Combined Plan and Disclosure Statement provide the same treatment for each Claim of a particular Class unless the Holder of a particular Claim agrees to a less favorable treatment of its Claim. The Plan Proponents believe that the Combined Plan and Disclosure Statement complies with the requirement of equal treatment. To the extent that the Bankruptcy Court finds that the Combined Plan and Disclosure Statement does not satisfy such requirement, the Bankruptcy Court could deny confirmation of the Combined Plan and Disclosure Statement. Issues or disputes relating to classification and/or treatment could result in a delay in the confirmation and consummation of the Combined Plan and Disclosure Statement and could increase the risk that the Combined Plan and Disclosure Statement will not be consummated.

> Failure to Consummate the Combined Plan and Disclosure (e) Statement.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Although the Plan Proponents believe that the Effective Date will occur and may occur quickly after the Confirmation Date, there can be no assurance as to such timing, or as to whether the Effective Date will, in fact, occur.

(f) The Releases May Not Be Approved.

There can be no assurance that the releases, as provided in Section 17, will be granted. Failure of the Bankruptcy Court to grant such relief may result in a plan that differs from the Combined Plan and Disclosure Statement or the Plan not being confirmed.

(g) Reductions to Estimated Creditor Recoveries.

The Allowed amount of Claims in any Class could be greater than projected, which, in turn, could cause the amount of Distributions to creditors in such Class to be reduced substantially. The amount of cash realized from the monetization of the Debtor's remaining assets could be less than anticipated, which could cause the amount of Distributions to creditors to be reduced substantially.

7.2 Certain U.S. Federal Income Tax Consequences.

The following discussion is a summary of certain material U.S. federal income tax consequences of the Combined Plan and Disclosure Statement to the Debtor and to certain Holders (which solely for purposes of this discussion means the beneficial owner for U.S. federal income tax purposes) of Claims. The following summary does not address the U.S. federal income tax consequences to holders of Claims not entitled to vote on the Combined Plan and Disclosure Statement. This summary is based on the Internal Revenue Code, Treasury Regulations promulgated and proposed decisions, published administrative thereunder, iudicial and pronouncements of the IRS, all as in effect on the date hereof and all of which are subject to change or differing interpretations, possibly with retroactive effect. No legal opinions have been requested or obtained from counsel with respect to any of the tax aspects of the Combined Plan and Disclosure Statement and no rulings have been or will be requested from the IRS with respect to the any of the issues discussed below. The discussion below is not binding upon the IRS or the courts. No assurance can be given that the IRS would not assert, or that a court would not sustain, a different position than any position discussed herein.

This discussion does not purport to address all aspects of U.S. federal income taxation that may be relevant to the Debtor or to certain holders of Claims in light of their individual circumstances, nor does the discussion deal with tax issues with respect to holders of Claims subject to special treatment under the U.S. federal income

2

3

4

6

7

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

If a partnership (or other entity or arrangement classified as a partnership for U.S. federal income tax purposes) is a holder of Claims, the U.S. federal income tax treatment of a partner in the partnership will generally depend on the status of the partner and the activities of the partnership. A holder of a Claim that is a partnership and the partners in such partnership should consult their tax advisors with regard to the U.S. federal income tax consequences of the Combined Plan and Disclosure Statement.

FOLLOWING SUMMARY IS **FOR INFORMATIONAL** PURPOSES ONLY AND IS NOT A SUBSTITUTE FOR CAREFUL TAX **PLANNING AND ADVICE BASED UPON** THE **INDIVIDUAL** CIRCUMSTANCES PERTAINING TO A HOLDER OF A CLAIM. EACH HOLDER OF A CLAIM IS URGED TO CONSULT WITH SUCH HOLDER'S TAX ADVISORS CONCERNING THE U.S. FEDERAL, STATE, LOCAL, FOREIGN, AND OTHER TAX CONSEQUENCES OF THE COMBINED PLAN AND DISCLOSURE STATEMENT.

(a) <u>Tax Consequences for U.S. Holders of Certain Claims.</u>

Generally, a Holder of a Claim should in most, but not all, circumstances recognize gain or loss equal to the difference between the "amount realized" by such Holder in exchange for its Claim and such Holder's adjusted tax basis in the Claim. The "amount realized" is equal to the sum of the cash and the fair market value of any other consideration received under a plan of reorganization in respect of a Holder's

Claim. The tax basis of a Holder in a Claim will generally be equal to the Holder's cost therefor. To the extent applicable, the character of any recognized gain or loss (e.g., ordinary income, or short-term or long-term capital gain or loss) will depend upon the status of the Holder, the nature of the Claim in the Holder's hands, the purpose and circumstances of its acquisition, the Holder's holding period of the Claim, and the extent to which the Holder previously claimed a deduction for the worthlessness of all or a portion of the Claim. Generally, if the Claim is a capital asset in the Holder's hands, any gain or loss realized will generally be characterized as capital gain or loss, and will constitute long-term capital gain or loss if the Holder has held such Claim for more than one year.

A creditor who receives Cash in satisfaction of its Claims may recognize ordinary income or loss to the extent that any portion of such consideration is characterized as accrued interest. A creditor who did not previously include in income accrued but unpaid interest attributable to its Claim, and who receives a distribution on account of its Claim pursuant to the Plan, will be treated as having received interest income to the extent that any consideration received is characterized for U.S. federal income tax purposes as interest, regardless of whether such creditor realizes an overall gain or loss as a result of surrendering its Claim. A creditor who previously included in its income accrued but unpaid interest attributable to its Claim should recognize an ordinary loss to the extent that such accrued but unpaid interest is not satisfied, regardless of whether such creditor realizes an overall gain or loss as a result of the distribution it may receive under the Plan on account of its Claim.

Under the Plan, the Holders of certain Claims, including General Unsecured Claims in Class 3, will likely receive only a partial distribution of their Allowed Claims. Whether the applicable Holder of such Claims will recognize a loss or any other tax treatment will depend upon facts and circumstances that are specific to the nature of the Holder and its Claims. Creditors should consult their own tax advisors.

(b) <u>Tax Consequences in Relation to Liquidating Trust.</u>

As of the Effective Date, the Liquidating Trust will be established for the benefit of the holders of certain Allowed Claims. The tax consequences of the Plan in relation to the Liquidating Trust and the Beneficiaries thereof are subject to uncertainties due to the complexity of the Plan and the lack of interpretative authority regarding certain changes in the tax law.

Allocations of taxable income of the Liquidating Trust (other than taxable income allocable to the Liquidating Trust's claims reserves) among Holders of Claims will be determined by reference to the manner in which an amount of cash equal to such taxable income would be distributed (were such cash permitted to be distributed

at such time) if, immediately prior to such deemed distribution, the Liquidating Trust had distributed all of its assets (valued at their tax book value) to the holders of the beneficial interests in the Liquidating Trust, adjusted for prior taxable income and loss and taking into account all prior and concurrent distributions from the Liquidating Trust. Similarly, taxable loss of the Liquidating Trust will be allocated by reference to the manner in which an economic loss would be borne immediately after a liquidating distribution of the remaining trust assets.

The tax book value of the trust assets for this purpose will equal their fair market value on the Effective Date, adjusted in accordance with tax accounting principles prescribed by the Tax Code, applicable Treasury Regulations, and other applicable administrative and judicial authorities and pronouncements. Uncertainties with regard to federal income tax consequences of the Plan may arise due to the inherent nature of estimates of value that will impact tax liability determinations.

Subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary (including the receipt of an IRS private letter ruling if the Liquidating Trustee so requests one, or the receipt of an adverse determination by the IRS upon audit if not contested by the Liquidating Trustee), the Liquidating Trustee may (a) elect to treat any trust assets allocable to, or retained on account of, Disputed Claims (the "Trust Claims Reserve") as a "disputed ownership fund" governed by Treasury Regulation section 1.468B-9, and (b) to the extent permitted by applicable law, report consistently with the foregoing for state and local income tax purposes. Accordingly, any Trust Claims Reserve will be subject to tax annually on a separate entity basis on any net income earned with respect to the trust assets in such reserves, and all distributions from such reserves will be treated as received by holders in respect of their Claims as if distributed by the Debtor. All parties (including, without limitation, the Liquidating Trustee and the holders of beneficial interests in the Liquidating Trust) will be required to report for tax purposes consistently with the foregoing.

The Liquidating Trust is intended to qualify as a liquidating trust for federal income tax purposes. In general, a liquidating trust is not a separate taxable entity but rather is treated for federal income tax purposes as a "grantor" trust (i.e., a pass-through entity). The IRS, in Revenue Procedure 94-45, 1994.28 I.R.B. 124, set forth the general criteria for obtaining an IRS ruling as to the grantor trust status of a liquidating trust under a chapter 11 plan. The Liquidating Trust has been structured with the intention of complying with such general criteria. Pursuant to the Plan and Liquidating Trust Agreement, and in conformity with Revenue Procedure 94-45, supra, all parties (including the Liquidating Trustee and the holders of beneficial interests in the Liquidating Trust) are required to treat for federal income tax purposes, the Liquidating Trust as a grantor trust of which the holders of the applicable Allowed

Claims are the owners and grantors. While the following discussion assumes that the Liquidating Trust would be so treated for federal income tax purposes, no ruling has been requested from the IRS concerning the tax status of the Liquidating Trust as a grantor trust. Accordingly, there can be no assurance that the IRS would not take a contrary position to the classification of the Liquidating Trust as a grantor trust. If the IRS were to challenge successfully such classification, the federal income tax consequences to the Liquidating Trust and the Beneficiaries thereof could materially vary from those discussed herein.

In general, each creditor who is a Liquidating Trust Beneficiary will recognize gain or loss in an amount equal to the difference between (i) the "amount realized" by such Liquidating Trust Beneficiary in satisfaction of its applicable Allowed Claim, and (ii) such Liquidating Trust Beneficiary's adjusted tax basis in such Claim. The "amount realized" by a Liquidating Trust Beneficiary will equal the sum of cash and the aggregate fair market value of the property received by such party pursuant to the Plan (such as a Liquidating Trust Beneficiary's undivided beneficial interest in the assets transferred to the Liquidating Trust). Where gain or loss is recognized by a Liquidating Trust Beneficiary in respect of its Allowed Claim, the character of such gain or loss (*i.e.*, long-term or short-term capital, or ordinary income) will be determined by a number of factors including the tax status of the party, whether the Claim constituted a capital asset in the hands of the party and how long it had been held, whether the Claim was originally issued at a discount or acquired at a market discount and whether and to what extent the party had previously claimed a bad debt deduction in respect of the Claim.

After the Effective Date, any amount that a creditor receives as a Distribution from the Liquidating Trust in respect of its beneficial interest in the Liquidating Trust should not be included, for federal income tax purposes, in the party's amount realized in respect of its Allowed Claim, but should be separately treated as a distribution received in respect of such party's beneficial interest in the Liquidating Trust.

In general, a Liquidating Trust Beneficiary's aggregate tax basis in its undivided beneficial interest in the assets transferred to the Liquidating Trust will equal the fair market value of such undivided beneficial interest as of the Effective Date and the Liquidating Trust Beneficiary's holding period in such assets will begin the day following the Effective Date. Distributions to any Liquidating Trust Beneficiary will be allocated first to the original principal portion of the Liquidating Trust Beneficiary's Allowed Claim as determined for federal tax purposes, and then, to the extent the consideration exceeds such amount, to the remainder of such Claim. However, there is no assurance that the IRS will respect such allocation for federal income tax purposes.

2

3

4

5

7

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

The Liquidating Trust's taxable income will be allocated to the holders of beneficial interests in the Liquidating Trust in accordance with each such holder's pro rata share. The character of items of income, deduction and credit to any holder and the ability of such holder to benefit from any deductions or losses may depend on the particular situation of such holder.

The federal income tax reporting obligation of a holder of a beneficial interest in the Liquidating Trust is not dependent upon the Liquidating Trust distributing any cash or other proceeds. Therefore, a holder of a beneficial interest in the Liquidating Trust may incur a federal income tax liability regardless of the fact that the Liquidating Trust has not made, or will not make, any concurrent or subsequent distributions to the holder. If a holder incurs a federal tax liability but does not receive distributions commensurate with the taxable income allocated to it in respect of its beneficial interests in the Liquidating Trust it holds, the holder may be allowed a subsequent or offsetting loss.

The Liquidating Trustee will file with the IRS returns for the Liquidating Trust as a grantor trust pursuant to Treasury Regulations section 1.671-4(a). The Liquidating Trust will also send to each holder of a beneficial interest in the Liquidating Trust a separate statement setting forth the holder's share of items of income, gain, loss, deduction, or credit and will instruct the holder to report such items on its federal income tax return.

Events subsequent to the date of this Disclosure Statement, such as the enactment of additional tax legislation, could also change the federal income tax consequences of the Plan and the transactions contemplated thereunder.

(c) <u>Information Reporting and Withholding.</u>

In connection with the Combined Plan and Disclosure Statement, the Debtor will comply with all applicable withholding and information reporting requirements imposed by U.S. federal, state, local, and foreign taxing authorities, and all Distributions under the Combined Plan and Disclosure Statement will be subject to those withholding and information reporting requirements. Holders of Claims may be required to provide certain tax information as a condition to receiving Distributions pursuant to the Combined Plan and Disclosure Statement.

In general, information reporting requirements may apply to Distributions pursuant to the Combined Plan and Disclosure Statement. Additionally, under the backup withholding rules, a holder may be subject to backup withholding with respect to Distributions made pursuant to the Combined Plan and Disclosure Statement, unless a U.S. holder provides the applicable withholding agent with a taxpayer identification number, certified under penalties of perjury, as well as certain other information, or otherwise establish an exemption from backup withholding. Backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules will be allowed as a credit against a U.S. holder's U.S. federal income tax liability, if any, hand may entitle a U.S. holder to a refund, provided the required information is timely furnished to the IRS.

In addition, from an information reporting perspective, Treasury Regulations generally require disclosure by a taxpayer on its U.S. federal income tax return of certain types of transactions in which the taxpayer participated, including, among other types of transactions, certain transactions that result in the taxpayer's claiming a loss in excess of specified thresholds. Holders of Claims are urged to consult their tax advisors regarding these regulations and whether the transactions contemplated by the Combined Plan and Disclosure Statement would be subject to these regulations and require disclosure on the holder's tax returns.

THE U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE COMBINED PLAN AND DISCLOSURE STATEMENT ARE COMPLEX. THE FOREGOING SUMMARY DOES NOT DISCUSS ALL ASPECTS OF U.S. FEDERAL INCOME TAXATION THAT MAY BE RELEVANT TO A PARTICULAR HOLDER OF A CLAIM IN LIGHT OF SUCH HOLDER'S CIRCUMSTANCES. EACH HOLDER OF A CLAIM IS URGED TO CONSULT WITH SUCH HOLDER'S TAX ADVISORS CONCERNING THE U.S. FEDERAL, STATE, LOCAL, FOREIGN, AND OTHER TAX CONSEQUENCES OF THE COMBINED PLAN AND DISCLOSURE STATEMENT.

7.3 Releases, Exculpations, and Injunctions.

This Combined Plan and Disclosure Statement contains certain releases, exculpations, and injunction language. Parties are urged to read these provisions carefully to understand how Confirmation and consummation of the Plan will affect any Claim, interest, right, or action with regard to the Debtor and certain third parties.

THE COMBINED PLAN AND DISCLOSURE STATEMENT SHALL BIND ALL HOLDERS OF CLAIMS AGAINST THE DEBTOR TO THE FULLEST EXTENT AUTHORIZED OR PROVIDED UNDER THE APPLICABLE PROVISIONS OF THE BANKRUPTCY CODE AND ALL OTHER APPLICABLE LAW.

7.4 Alternatives to the Combined Plan and Disclosure Statement.

If the requisite acceptances are not received or the Combined Plan and Disclosure Statement is not confirmed and consummated, the theoretical alternatives to the Combined Plan and Disclosure Statement would be (a) formulation of an alternative chapter 11 plan, (b) conversion of the Chapter 11 Case to a case under chapter 7 of the Bankruptcy Code, or (c) dismissal of the Chapter 11 Case. As discussed below, the Plan Proponents do not believe that any of these alternatives, even if viable, would afford holders of Claims a greater recovery than what is provided by the Combined Plan and Disclosure Statement.

If the Combined Plan and Disclosure Statement is not confirmed, then the Debtor or any other party in interest could attempt to formulate a different plan. The additional costs, including, among other amounts, additional professional fees, all of which would constitute Administrative Claims (subject to allowance thereof), however, may be so significant that one or more parties in interest could request that the Chapter 11 Case be converted to chapter 7. At this time, the Plan Proponents do not believe that there are viable alternative plans available to the Debtor.

If the Combined Plan and Disclosure Statement is not confirmed, the Chapter 11 Case may be converted to a case under chapter 7 of the Bankruptcy Code, pursuant to which a trustee would be elected or appointed to liquidate and distribute the Debtor's remaining assets in accordance with the priorities established by the Bankruptcy Code. As discussed above and indicated in the Liquidation Analysis, the Plan Proponents believe that the Combined Plan and Disclosure Statement provides a better outcome for holders of Claims than a chapter 7 liquidation would provide.

If the Combined Plan and Disclosure Statement is not confirmed, the Chapter 11 Case also could be dismissed. Among other effects, dismissal would result in the

8

5

15

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

16 17

18 19

20 21

22 23

24 26

27 28 termination of the automatic stay, thus permitting creditors to assert state-law rights and remedies against the Debtor and its assets, likely to the detriment of other creditors. While it is impossible to predict precisely what would happen in the event the Chapter 11 Case is dismissed, it is unlikely that dismissal would result in a ratable distribution of the Debtor's assets among creditors as provided in the Combined Plan and Disclosure Statement. Thus, the vast majority of creditors could expect to receive less in the dismissal scenario than they would receive under the Combined Plan and Disclosure Statement.

SECTION 8. UNCLASSIFIED CLAIMS

In accordance with § 1123(a)(1), the following Claims are not classified and are excluded from the Classes set forth in Section 9 hereof and shall receive the treatment discussed below:

8.1 Administrative Claims.

Except to the extent that the Debtor (or the Liquidating Trust) and the Holder of an Allowed Administrative Claim agree to less favorable treatment, a Holder of an Allowed Administrative Claim (other than a Professional Claim, which shall be subject to Section 8.2 or Statutory Fees, which shall be subject to Section 8.3) shall receive, in full satisfaction, settlement, release, and discharge of, and in exchange for, such Administrative Claim, Cash equal to the unpaid portion of such Allowed Administrative Claim either (a) on the Effective Date, (b) if the Allowed Administrative Claim is based on liabilities incurred by the Debtor in the ordinary course of their businesses after the Petition Date, in the ordinary course of business in accordance with the terms and conditions of the particular transaction giving rise to such Allowed Administrative Claim, without any further action by the Holder of such Allowed Administrative Claim, (c) on such other date as agreed between the Debtor (or the Post-Effective Date Debtor) and such Holder of an Allowed Administrative Claim, or (d) to the extent the Allowed Administrative Claim had not yet been Allowed on the Effective Date, from the Administrative Claims Reserve pursuant to Section 20.2 hereof.

Holders of Administrative Claims (including, without limitation, Professionals requesting compensation or reimbursement of such expenses pursuant to §§ 327, 328, 330, 331, 503(b), or 1103) that do not file such requests by the applicable deadline provided for herein may be subject to objection for untimeliness and may be prohibited by order of the Bankruptcy Court from asserting such claims against the Debtor, the Post-Effective Date Debtor, the Estate, the Liquidating Trust, or their

successors or assigns, or their property. Any objection to Professional Fee Claims shall be filed on or before the objection deadline specified in the application for final compensation or order of the Bankruptcy Court.

8.2 Professional Claims.

All Professionals seeking an award by the Bankruptcy Court of a Professional Claim (other than the Ordinary Course Professionals) shall file their respective final applications for allowance of compensation for services rendered and reimbursement of expenses incurred by the date that is forty-five (45) days after the Effective Date, and shall receive, in full satisfaction of such Claim, Cash in an amount equal to 100% of such Allowed Professional Claim promptly after entry of an order of the Bankruptcy Court allowing such Claim or upon such other terms as may be mutually agreed-upon between the Holder of such Professional Claim and the Debtor, which Cash shall be paid out of the Effective Date Professional Claim Reserve. Objections to any final applications covering Professional Claims must be filed and served on the Post-Effective Date Debtor, the Liquidating Trustee, and the requesting Professional no later than ninety (90) days after the Effective Date (unless otherwise agreed by the requesting Professional).

On the Effective Date, or as soon as practicable thereafter, the Liquidating Trustee shall establish the Effective Date Professional Claim Reserves based upon estimates of anticipated fees provided by Professionals for services rendered and expenses incurred prior to the Effective Date, including estimated fees for services rendered, and actual and necessary costs incurred, in connection with the filing, service and prosecution of any applications for allowance of Professional Fees pending on the Effective Date or filed and/or served after the Effective Date. The Liquidating Trustee shall supplement the Effective Date Professional Claim Reserves if the amount originally established is insufficient to pay Allowed Professional Fee Claims.

Upon approval of the fee applications by the Bankruptcy Court, the Liquidating Trustee shall pay Professionals from the Effective Date Professional Claim Reserves all of their respective Allowed Professional Fee Claims.

8.3 Statutory Fees.

Statutory Fees shall be paid by the Liquidating Trustee in the ordinary course of business until the closing, dismissal or conversion of the Chapter 11 Case to another chapter of the Bankruptcy Code. Any unpaid Statutory Fees that accrued before the Effective Date shall be paid no later than thirty (30) days after the Effective Date. For the avoidance of doubt, if the Chapter 11 Case is reopened, the Liquidating Trustee

4

5

6

7

8

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Code.

8.4 Priority Tax Claims.

Except to the extent that a Holder of an Allowed Priority Tax Claim agrees to less favorable treatment, each Holder of an Allowed Priority Tax Claim shall receive, in full and final satisfaction of such Allowed Priority Tax Claim, at the option of the Liquidating Trustee: (a) Cash in an amount equal to such Allowed Priority Tax Claim on, or as soon thereafter as is reasonably practicable, the later of (i) the Effective Date, to the extent such Claim is an Allowed Priority Tax Claim on the Effective Date, and (ii) the first Business Day after the date that is thirty (30) days after the date such Priority Tax Claim becomes an Allowed Priority Tax Claim; or (b) equal annual Cash payments in an aggregate amount equal to the amount of such Allowed Priority Tax Claim, together with interest at the applicable rate pursuant to § 511, over a period not exceeding five (5) years from and after the Petition Date; *provided, however*, the Post-Effective Date Debtor and Liquidating Trustee, as applicable, reserve the right to prepay all or a portion of any such amounts at any time under this option at the discretion of the Post-Effective Date Debtor and the Liquidating Trustee.

shall pay any Statutory Fees in the ordinary course of business until the closing,

dismissal, or conversion of the Chapter 11 Case to another chapter of the Bankruptcy

SECTION 9. CLASSIFICATION OF CLAIMS AND VOTING

9.1 Classification in General.

A Claim is placed in a particular Class for all purposes, including voting, confirmation, and distribution under the Plan and under §§ 1122 and 1123(a)(1); provided, that a Claim is placed in a particular Class for the purpose of receiving Distributions pursuant to the Plan only to the extent that such Claim is an Allowed Claim in that Class and such Allowed Claim has not been satisfied, released, or otherwise settled prior to the Effective Date.

9.2 Summary of Classification.

The following table designates the Classes of Claims against the Debtor and specifies which of those Classes are (a) not Impaired by the Plan, (b) Impaired by the Plan, and (c) entitled to vote to accept or reject the Plan in accordance with § 1126. In accordance with § 1123(a)(1), Administrative Claims, Professional Claims, Statutory Fees, and Priority Tax Claims, have not been classified. All of the potential Classes for the Debtor are set forth herein. If, ultimately, the Debtor does not have

1

5 6 7

9 10

8

12 13

11

14 15

17 18

16

20 21

19

2324

22

25

26

2728

Holders of Claims in a particular Class or Classes, such Classes shall be treated as set forth in Section 9.4.

Class	Designation	Impairment	Entitled to Vote
1	Priority Non-Tax Claims	Not Impaired	No (deemed to
		Not impaned	accept)
2	Secured Claims	Not Impaired	No (deemed to
			accept)
3	General Unsecured Claims	Impaired	Yes
4	Allowed DHCS Claim	Impaired	Yes

9.3 Special Provision Governing Unimpaired Claims.

Except as otherwise provided in the Plan, nothing under the Plan shall affect the rights of the Debtor or the Liquidating Trust with respect to Unimpaired Claims, including all legal and equitable defenses to, or setoffs or recoupments against, any such Unimpaired Claims.

9.4 Elimination of Vacant Classes.

Any Class of Claims, as of the commencement of the Confirmation Hearing, that does not have at least one (1) Holder of a Claim in an amount greater than zero for voting purposes shall be considered vacant, deemed eliminated from the Plan for purposes of voting to accept or reject the Plan, and disregarded for purposes of determining whether the Plan satisfies § 1129(a)(8) with respect to that Class.

9.5 Voting; Presumptions; Solicitation in Good Faith.

Only Holders of Allowed Claims in Class 3 and Class 4 are entitled to vote to accept or reject this Plan. Holders of Claims in a Voting Class will receive Ballots containing detailed voting instructions.

The Plan Proponents will solicit votes on the Plan from the Voting Classes in good faith and in compliance with the applicable provisions of the Bankruptcy Code. Accordingly, the Plan Proponents and each of their Related Persons shall be entitled to, and upon the Confirmation Date, will be granted the protections of § 1125(e).

9.6 Cramdown.

If any Class of Claims is deemed and/or presumed to reject this Plan or is entitled to vote on this Plan and does not vote to accept this Plan, the Plan Proponents intend to (i) seek confirmation of this Plan under § 1129(b), or (ii) amend or modify this Plan in accordance with the terms hereof and the Bankruptcy Code. If a

4 5

6

7 8 9

10 11

12 13

> 14 15

16

18

17

1920

2122

23

2425

2627

28

controversy arises as to whether any Claims, or any Class of Claims, are Impaired, the Bankruptcy Court shall, after notice and a hearing, determine such controversy on or before the Confirmation Date.

SECTION 10. TREATMENT OF CLAIMS

In full and final satisfaction of all of the Claims against the Debtor (except with respect to Unclassified Claims that are satisfied in accordance with Section 8 above), the Claims shall receive the treatment described below. Except to the extent expressly provided in this Section 10, the timing of Distributions is addressed in Section 11 hereof.

10.1 Class 1: Priority Non-Tax Claims.

- a) *Classification*. Class 1 consists of Priority Non-Tax Claims.
- b) <u>Treatment.</u> Except to the extent that a Holder of an Priority Non-Tax Claim agrees to a less favorable treatment of such Claim, each such Holder shall receive payment in Cash in an amount equal to the amount of such Allowed Claim, payable on the later of (i) the Effective Date; and (ii) after the date on which such Priority Non-Tax Claim becomes an Allowed Priority Non-Tax Claim, in each case, or as soon as reasonably practicable thereafter in accordance with the priority scheme set forth in the Bankruptcy Code.
- c) <u>Voting.</u> Class 1 is Unimpaired. Holders of Priority Non-Tax Claims are deemed to have accepted the Plan, pursuant to § 1126(f), and are not entitled to vote to accept or reject the Plan.

10.2 Class 2: Secured Claims.

- a) Classification. Class 2 consists of Secured Claims.
- b) **Treatment.** The legal, equitable, and contractual rights of Holders of Allowed Secured Claims are unaltered by the Plan, except as altered pursuant to the Sale Order, and the Liens of the Holders of the Secured Claims will continue to attach to their respective collateral, provided, that, all such Claims shall remain subject to any and all defenses, challenges, counterclaims, and setoff or recoupment rights with respect thereto. Except to the extent that a Holder of an Allowed Secured Claim agrees to a less favorable treatment of such Claim, each such Holder shall, at the option of the Post-Effective Date Debtor and Liquidating Trustee, as applicable, receive one of

2

3

4

5

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

the following treatments: (i) the Holder of such Secured Claim shall retain its Lien on its collateral until such collateral is sold, and the proceeds of such sale, less costs and expenses of disposing of such collateral, shall be paid to such Holder in full satisfaction of such Secured Claim; (ii) on or as soon as practicable after the later of (A) the Effective Date; or (B) the date upon which the Bankruptcy Court enters a Final Order determining or allowing such Secured Claim, the Holder of such Secured Claim will receive a Cash payment equal to the amount of its Secured Claim in full and final satisfaction of such Secured Claim; or (iii) the collateral securing the Secured Claim shall be abandoned to such Holder, in full and final satisfaction of such Secured Claim. receive payment in Cash in an amount equal to the amount of such Allowed Claim.

The Bankruptcy Court shall retain jurisdiction and power to determine the amount necessary to satisfy any Allowed Secured Claim. Upon receipt of Cash in an amount equal to the amount of such Allowed Secured Claim, the Holder of such Allowed Secured Claim shall release (by the Confirmation Order shall be deemed to release) all Liens against any Liquidating Trust Assets.

c) Voting. Class 2 is Unimpaired. Holders of Secured Claims are deemed to have accepted the Plan, pursuant to § 1126(f), and are not entitled to vote to accept or reject the Plan.

10.3 Class 3: General Unsecured Claims.

- a) Classification. Class 3 consists of the General Unsecured Claims against the Debtor.
- b) **Treatment.** As soon as practicable after the Effective Date, each Holder of an Allowed General Unsecured Claim shall receive in full satisfaction, settlement, discharge, and release of, and in exchange for such Allowed General Unsecured Claim, its Pro Rata share of the Class A Trust Beneficial Interests.
- c) Voting. Class 3 is Impaired. Holders of General Unsecured Claims are entitled to vote to accept or reject the Plan.

10.4 Class 4: Allowed DHCS Claim.

- a) **Classification.** Class 4 consists of the Allowed DHCS Claim.
- b) **Treatment.** As soon as practicable after the Effective Date, DHCS as holder of the Allowed DHCS Claim, shall receive in full satisfaction, settlement, discharge, and release of, and in exchange for, such Allowed DHCS Claim: (i) the application of the DHCS Allowed Offset Amount to the Allowed DHCS Claim by

4 5

7

8

9

6

10 11

12

1314

15

16 17

18 19

20

2122

2324

252627

28

DHCS; (ii) distribution of the DHCS Sales Proceeds Recovery by the Liquidating Trustee; and (iii) distribution of its Pro Rata share of the Class B Trust Beneficial Interests.

c) **Voting. Class 4 is Impaired.** The Holder of the DHCS Claim is entitled to vote to accept or reject the Plan.

SECTION 11. DISTRIBUTIONS

11.1 *Party Responsible for Making Distributions*. Subject to the prior payment of the amounts required to be paid by the Debtor in Cash on the Effective Date pursuant to this Plan, all Plan Distributions shall be made by the Liquidating Trustee (or by the Disbursing Agent, to the extent directed by the Liquidating Trustee), or, with respect to Distributions made to Holders of Class A Trust Beneficial Interests, by the Co-Liquidating Trustee (or by the Disbursing Agent, to the extent directed by the Co-Liquidating Trustee).

Neither the Liquidating Trustee nor the Co-Liquidating Trustee shall be required to give any bond or surety or other security for the performance of their duties unless otherwise ordered by the Bankruptcy Court. The Liquidating Trustee, Co-Liquidating Trustee, or Disbursing Agent, as applicable, shall be empowered to (a) effect all actions and execute all agreements, instruments, and other documents necessary to perform their duties under the Plan; (b) make all Distributions contemplated hereby; (c) employ transfer agents and registrars to represent them with respect to their responsibilities; and (d) exercise such other powers as may be vested in the Liquidating Trustee or Co-Liquidating Trustee by order of the Bankruptcy Court, pursuant to the Plan, the Liquidating Trustee, applicable, to be necessary and proper to implement the provisions hereof.

11.2 *Appointment of Disbursing Agent.* A Disbursing Agent may be appointed either pursuant to the Confirmation Order or by the Liquidating Trustee.

11.3 Timing of Distributions.

a) Distributions on Account of All Claims Other Than the DHCS Claim. Subject to Section 11.1 of this Plan, on the Effective Date (or if a Claim is not an Allowed Claim on the Effective Date, on the date that such Claim becomes an Allowed Claim), or as soon as reasonably practicable thereafter, each Holder of an Allowed Claim (except for the Allowed DHCS Claim) against the Debtor shall receive full amount of the Distributions that the Plan provides for Allowed Claims in

1

5

6

7 8

9

1011

12 13

1415

1617

18 19

2021

22

23

2425

2627

28

the applicable Class and in the manner provided herein. If and to the extent there are Disputed Claims, Distributions on account of any such Disputed Claims shall be made pursuant to the provisions provided in the Plan. Except as otherwise provided in the Plan, Holders of Claims shall not be entitled to interest, dividends, or accruals on the Distributions provided for therein, regardless of whether Distributions are delivered on or at any time after the Effective Date.

Upon the Effective Date, all debts of the Debtor shall be deemed fixed and adjusted pursuant to the Plan and the Debtor and the Liquidating Trust shall have no liability on account of any Claims except as set forth in the Plan and in the Confirmation Order. All payments and all Distributions made by the Liquidating Trustee, the Co-Liquidating Trustee, or the Disbursing Agent under the Plan shall be in full and final satisfaction, settlement, and release of all Claims against the Debtor.

b) Distributions on Account of the Allowed DHCS Claim. Distributions on account of the Allowed DHCS Claim shall be made in accordance with Section 10.4 hereof, provided, however, that Distributions on account of the Allowed DHCS Balance Claim (i.e., to the Holder of Class B Trust Beneficial Interests) need not be made to the extent there is no Cash in the Trust Assets Accounts to distribute.

11.4 Manner of Cash Payments Under the Plan and Liquidating Trust Agreement.

Cash payments made pursuant to the Plan and Liquidating Trust Agreement, shall be in United States dollars by checks drawn on a domestic bank selected by the Liquidating Trustee, Co-Liquidating Trustee, or Disbursing Agent, as applicable, or by wire or ACH transfer from a domestic bank, at the option of the Liquidating Trustee, Co-Liquidating Trustee, or Disbursing Agent, as applicable.

11.5 Distribution Procedures.

- a) **Distribution Dates.** The Liquidating Trustee and Co-Liquidating Trustee, as applicable, shall make Distributions to Holders of Claims as provided in Section 11 of the Plan.
- b) **Subsequent Distributions.** Any Distribution not made on the Initial Distribution Date or a Subsequent Distribution Date because the Claim relating to such Distribution had not been Allowed on that Distribution Date shall be held by the Liquidating Trust for Distribution on any Subsequent Distribution Date after such Claim is Allowed. No interest shall accrue or be paid on the unpaid amount of any Distribution.

2

3

4

5

6

7

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

2.5

26

27

28

- Distribution Record Date. Except as otherwise provided in a Final Order of the Bankruptcy Court, the transferees of Claims that are transferred pursuant to Bankruptcy Rule 3001 on or prior to the Distribution Record Date will be treated as the Holders of those Claims for all purposes, notwithstanding that any period provided by Bankruptcy Rule 3001 for objecting to the transfer may not have expired by the Distribution Record Date. Neither the Liquidating Trustee nor the Co-Liquidating Trustee shall have any obligation to recognize any transfer of any Claim occurring after the Distribution Record Date. In making any Distribution with respect to any Claim, the Liquidating Trustee and Co-Liquidating Trustee shall be entitled instead to recognize and deal with, for all purposes hereunder, only the Entity that is listed on the proof of claim filed with respect thereto or on the Schedules as the Holder thereof as of the close of business on the Distribution Record Date and upon such other evidence or record of transfer or assignment that is known to the Liquidating Trustee or Co-Liquidating Trustee as of the Distribution Record Date.
- d) Time Bar to Cash Payments by Check. Checks issued by the Liquidating Trustee or Co-Liquidating Trustee, as applicable, on account of Allowed Claims shall be null and void if not negotiated within ninety (90) days after the date of issuance thereof. Requests for the reissuance of any check that becomes null and void pursuant to this Article shall be made directly to the Liquidating Trustee or Co-Liquidating Trustee, as applicable, by the Holder of the Allowed Claim to which the check was originally issued. Any Claim in respect of such voided check shall be made in writing on or before the later of six months from the Effective Date or 30 days after the date on which such check is voided. After that date, all Claims in respect of voided checks shall be disallowed and forever barred and the proceeds of those checks shall revest in and become the property of the Liquidating Trust as unclaimed property in accordance with § 347(b).
- 11.6 Withholding of Distributions. In connection with making Distributions under this Plan, to the extent applicable, the Liquidating Trustee and Co-Liquidating Trustee shall comply with all tax withholding and reporting requirements imposed on them by any Governmental Unit, and all Distributions pursuant to this Plan shall be subject to such withholding and all related agreements shall be subject to any applicable withholding and reporting requirements. The Liquidating Trustee or Co-Liquidating Trustee, as applicable, may withhold the entire Distribution to any Holder of an Allowed Claim until such time as such Holder provides the necessary information to comply with any withholding requirements of any Governmental Unit. Any property so withheld will then be paid by the Liquidating Trustee or the Co-Liquidating Trustee, as applicable, to the appropriate authority. If the Holder of an Allowed Claim fails to provide the information to comply with any withholding requirements of any Governmental Unit within three months after the date of first

4

1

111213

10

14 15

161718

1920

21 22

232425

26

27

28

notification to the Holder of the need for such information for the Cash necessary to comply with any applicable withholding requirements, then such Holders' Distribution shall be treated in accordance with Section 11.6 of the Plan.

11.7 Delivery of Distributions and Undeliverable Distributions. Subject to Bankruptcy Rule 9010 and except as otherwise provided herein, Distributions to the Holders of Allowed Claims shall be made by the Liquidating Trustee or Co-Liquidating Trustee, as applicable, at (a) the address of each Claimant as set forth in the Schedules, unless superseded by the address set forth on proof(s) of claim filed by such Claimant, or (b) the last known address of such Claimant if no proof of claim is filed or the Debtor, Post-Effective Date Debtor, Liquidating Trustee, or Co-Liquidating Trustee has been notified in writing of a change of address. If any Distribution is returned as undeliverable, the Liquidating Trustee or Co-Liquidating Trustee may, in its discretion, make reasonable efforts to determine the current address of the Holder of the Claim with respect to which the Distribution was made as the Liquidating Trustee or Co-Liquidating Trustee deems appropriate, but no Distribution to any such Holder shall be made unless and until the Liquidating Trustee or Co-Liquidating Trustee has determined the then-current address of such Holder, at which time the Distribution to such Holder shall be made without interest. Amounts in respect of any undeliverable Distributions shall be returned to, and held in trust by, the Liquidating Trustee or Co-Liquidating Trustee until the Distributions are claimed or are deemed to be unclaimed property under § 347(b), as set forth in Section 11.7 of the Plan. The Liquidating Trustee or Co-Liquidating Trustee shall have the discretion to determine how to make Distributions in the most efficient and costeffective manner possible; provided, however, that its discretion may not be exercised in a manner inconsistent with any express requirements of the Plan or Liquidating Trust Agreement.

Except with respect to property not distributed because it is being held in the Disputed Claim Reserve, Distributions that are not claimed by the later of ninety (90) days from the Effective Date or ninety (90) days after the date of a Distribution shall be deemed to be unclaimed property under § 347(b) and shall vest or revest in the Liquidating Trust, and the Claims with respect to which those Distributions are made shall be automatically cancelled. Nothing contained in the Plan shall require the Debtor or the Liquidating Trust to attempt to locate any Holder of an Allowed Claim. All funds or other property that vest or revest in the Liquidating Trust pursuant to this Section shall be distributed to the other Holders of Allowed Claims in accordance with the provisions of the Plan and the Liquidating Trust Agreement.

11.8 **Setoffs and Recoupments.** The Liquidating Trustee or Co-Liquidating Trustee may, to the extent permitted under applicable law, setoff or recoup against

any Allowed Claim and any distributions to be made pursuant to the Plan on account of such Allowed Claim, the claims, rights, and Causes of Action of any nature that the Debtor or the Liquidating Trust may hold against the Holder of such Allowed Claim that are not otherwise waived, released, or compromised in accordance with the Plan; provided, however, that neither such a setoff, recoupment, nor the allowance of any Claim hereunder shall constitute a waiver or release by the Liquidating Trustee or Co-Liquidating Trustee of any such claims, rights, and Causes of Action that the Debtor or the Liquidating Trust possesses against such Holder; provided, further, that neither the failure to effect such a setoff or recoupment nor the allowance of any Claim hereunder shall constitute a waiver or release by the Liquidating Trustee or the Co-Liquidating Trustee of any such Claims, rights, or Causes of Action that the Debtor, the Estate, or the Liquidating Trust possess against such Holder. Any Holder of an Allowed Claim subject to such setoff or recoupment reserves the right to challenge any such setoff or recoupment in the Bankruptcy Court or any other court with jurisdiction with respect to such challenge.

- Holder of an Allowed Claim if the amount of Cash to be distributed on any Distribution Date under the Plan on account of such Claim is \$50 or less. Any Holder of an Allowed Claim on account of which the amount of Cash to be distributed is \$50 or less will have its Claim for such Distribution discharged and will be forever barred from asserting any such Claim against each Released Party, the Post-Effective Date Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, and the Liquidating Trust. Any Cash not distributed pursuant to this Section will, in the Liquidating Trustee's discretion, be included in the Trust Assets Accounts, free of any restrictions thereon, and will be distributed in accordance with the Plan.
- 11.10 *Allocation of Plan Distribution Between Principal and Interest.* All Distributions by the Liquidating Trustee with respect to any Allowed Claim shall be allocated first to the principal amount of such Allowed Claim, as determined for federal income tax purposes, and thereafter, to the remaining portion of such Allowed Claim (including the interest portion of the Allowed Claim), if any.
- 11.11 *Entry of Final Decree in Chapter 11 Case.* After all the Disputed Claims have become Allowed Claims or have been disallowed by Final Order, and all Distributions in respect of Allowed Claims have been made in accordance with this Plan, or at such earlier time as the Liquidating Trustee deems appropriate, the Liquidating Trustee (i) shall seek authority from the Bankruptcy Court for entry of the Final Decree closing the Chapter 11 Case in accordance with the Bankruptcy Code and the Bankruptcy Rules and (ii) shall be authorized under the Plan to take any necessary corporate action with respect to the Debtor's continued existence without

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

the necessity for approvals or notices under any applicable state or other law, including under the Nonprofit Laws. Notwithstanding the foregoing, actions with respect to the Post-Effective Date Debtor shall be taken by the Liquidating Trustee. The entry of Final Decree closing this Chapter 11 Case shall not affect the Nonprofit Status of the Post-Effective Date Debtor to the extent it has not dissolved in accordance with the Plan.

11.12 Distribution Cap. In no event shall any Holder of an Allowed Claim receive Distribution(s) of an aggregate value exceeding one hundred percent (100%) of the amount of such Holder's Allowed Claim.

11.13 *Distributions Free and Clear*. Except as otherwise provided herein, any Distributions under the Plan shall be free and clear of any Liens, Claims, and encumbrances, and no Entity, including the Debtor, shall have any interest (legal, beneficial, or otherwise) in any property of the Estate distributed pursuant to the Plan.

SECTION 12. PROCEDURES FOR RESOLVING CONTINGENT, UNLIQUIDATED, AND DISPUTED CLAIMS

No Distributions Pending Allowance.

Notwithstanding any other provision of the Plan, the Debtor, the Liquidating Trustee or the Co-Liquidating Trustee, as applicable, shall not distribute any Cash or other property on account of any Disputed Claim unless and until such Claim becomes Allowed. Nothing contained herein, however, shall be construed to prohibit or require payment or Distribution on account of any undisputed portion of a Claim.

12.2 Resolution of Disputed Claims.

- a) General Unsecured Claims. From and after the Confirmation Date, all objections with respect to Disputed General Unsecured Claims shall be litigated to a Final Order by the Co-Liquidating Trustee, (i) except to the extent that the Co-Liquidating Trustee elects to withdraw any such objection or the Co-Liquidating Trustee and the Claimant elects to compromise, settle, or otherwise resolve any such objection, in which event they may settle, compromise or otherwise resolve any Disputed General Unsecured Claim without approval of the Bankruptcy Court; or (ii) as otherwise provided in the Liquidating Trust Agreement. The costs of pursuing objections to General Unsecured Claims shall be borne by the Liquidating Trust.
- b) Other Claims. From and after the Confirmation Date, all objections with respect to Disputed Claims (other than Disputed General Unsecured Claims) shall be litigated to a Final Order by the Liquidating Trustee, except to the extent the

5

6

7 8

9

1011

12

13 14

> 16 17

15

18 19

20

21222324

2526

27

28

Claims) shall be borne by the Liquidating Trust.

12.3 *Objection Deadline*. All objections to Claims shall be filed and served upon the Claimant not later than the Claims Objection Deadline, as such may be extended by order of the Bankruptcy Court.

Liquidating Trustee elects to withdraw any such objection or the Liquidating Trustee

and the Claimant elect to compromise, settle, or otherwise resolve any such objection,

in which event they may settle, compromise, or otherwise resolve any Disputed Claim (other than Disputed General Unsecured Claims) without approval of the Bankruptcy

Court. The costs of pursuing the objections to Claims (other than General Unsecured

- 12.4 *Allowance and Estimation of Claims*. Following the date on which a Disputed Claim becomes an Allowed Claim after the Distribution Date, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, shall pay directly to the Holder of such Allowed Claim, as soon as reasonably practicable, the amount provided for under the Plan, as applicable, and in accordance therewith.
- a) Allowance of Claims. Notwithstanding anything to the contrary herein, after the Effective Date and subject to the other provisions of the Plan, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, will have and will retain any and all rights and defenses under bankruptcy or nonbankruptcy law that the Debtor or its Estate had with respect to any Claim immediately before the Effective Date, except with respect to any Claim deemed Allowed under the Plan or by orders of the Bankruptcy Court. Except as expressly provided in the Plan or in any order entered in the Chapter 11 Case prior to the Effective Date (including the Confirmation Order), no Claim will become an Allowed Claim unless and until such Claim is deemed Allowed under the Plan or the Bankruptcy Code, or the Bankruptcy Court has entered a Final Order, including the Confirmation Order, in the Chapter 11 Case allowing such Claim.
- b) **Prosecution of Objections to Claims.** From and after the Effective Date, unless otherwise provided in the Plan, the Confirmation Order, or the Liquidating Trust Agreement, (i) the Liquidating Trustee will have the sole authority to file objections to Claims (other than General Unsecured Claims) and settle, compromise, withdraw, or litigate to judgment objections to any and all Claims (other than General Unsecured Claims), regardless of whether such Claims are in an Unimpaired Class or otherwise; and (ii) the Co-Liquidating Trustee will have the sole authority to file objections to general Unsecured Claims and settle, compromise, withdraw, or litigate to judgment objections to any and all General Unsecured Claims. From and after the Effective Date, the Liquidating Trustee may settle or compromise any Disputed Claim without any further notice to or action, order, or approval of the Bankruptcy Court. The Liquidating Trustee will have the sole authority to administer

1

5

6

7 8 9

10

11 12 13

14

15 16

17 18

192021

222324

2526

2728

and adjust the Claims Register to reflect any such settlements or compromises without any further notice to or action, order, or approval of the Bankruptcy Court; provided, however, that with respect to General Unsecured Claims, the Co-Liquidating Trustee shall have the sole authority to administer and adjust the Claims Register to reflect any compromises or settlements of General Unsecured Claims without any further notice to or action, order, or approval of the Bankruptcy Court.

c) Estimation of Claims. At any time, (a) prior to the Effective Date, the Debtor, and (b) after the Effective Date, the Liquidating Trustee or, solely with respect to General Unsecured Claims, the Co-Liquidating Trustee may request that the Bankruptcy Court estimate any contingent or unliquidated Claim to the extent permitted by § 502(c) regardless of whether the Debtor or the Liquidating Trust has previously objected to such Claim or whether the Bankruptcy Court has ruled on any such objection, and the Bankruptcy Court shall have jurisdiction to estimate any Claim at any time to any such objection. If the Bankruptcy Court estimates any contingent or unliquidated Claim, that estimated amount shall constitute either the Allowed amount of such Claim or a maximum limitation on the Claim, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on the Claim, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, may elect to pursue supplemental proceedings to object to the ultimate allowance of the Claim.

12.5 Disallowance of Claims.

- a) Except as otherwise agreed, any and all proofs of claim filed after the Bar Date are Disputed, and Holders of such Claims will not receive any Distributions thereon unless and until such Claim is Allowed.
- b) Subject to and in accordance with this Plan, any Claims held by Entities from which property is recoverable under §§ 542, 543, 550, or 553 or Entities that are transferees of transfers avoidable under §§ 522(f), 522(h), 544, 545, 547, 548, 549, or 724(a), provided, that, such Cause of Action is retained by the Liquidating Trust, shall be deemed disallowed pursuant to § 502(d), and Holders of such Claims may not receive any Distributions on account of such Claims until such time as such Causes of Action the Debtor holds or may hold against any Entity have been resolved or a Bankruptcy Court order with respect thereto has been entered and all sums due, if any, to the Estate by that Entity have been turned over or paid to the Debtor or Liquidating Trust.
- 12.6 Adjustment Without Objection. Any Claim that has been paid or satisfied, or any Claim that has been amended or superseded, may be adjusted or expunged from the claims register at the direction of the Debtor, the Liquidating

without further notice to or action, order, or approval of the Bankruptcy Court.

12.7 Reserve Provisions for Disputed Claims.

a) **Establishment of Disputed Reserves.** On or prior to each Distribution Date, the Liquidating Trustee shall each reserve Cash required for Distribution on Disputed Claims as if such Claims were Allowed as filed with any Disputed Claims that are unliquidated or contingent being reserved in an amount reasonably determined by the Liquidating Trustee, as applicable (the "<u>Disputed Claim Reserve</u>"). On each Distribution Date after the Effective Date in which the Liquidating Trustee or Co-Liquidating Trustee makes Distributions to Holders of Allowed Claims, the Liquidating Trustee shall retain on account of Disputed Claims an amount the Liquidating Trustee estimates would be necessary to fund the Pro Rata Share of such Distributions to Holders of Disputed Claims is such Claims were Allowed, with any Disputed Claims that are unliquidated or contingent being reserved in an amount reasonably determined by the Liquidating Trustee or Co-Liquidating Trustee, as applicable.

Trustee, or the Co-Liquidating Trustee, as applicable, without an objection filed and

- b) Maintenance of Disputed Reserves. The Liquidating Trustee shall hold property in the Disputed Claim Reserve in trust for the benefit of the Holders of the applicable Disputed Claims that are ultimately determined to be Allowed. The Disputed Claim Reserve shall be closed (or deemed closed) by the Liquidating Trust when all Distributions and other dispositions of Cash of other property required to be made hereunder will have been made in accordance with the terms of the Plan. Upon closure of the Disputed Claim Reserve, all Cash or other property held in that Disputed Claim Reserve shall revest in and become unrestricted property of the Liquidating Trust to be distributed in accordance with the Plan and the Liquidating Trust Agreement.
- c) Limitation on Funding Disputed Claim Reserves. Except as expressly set forth in the Plan, neither the Debtor nor the Liquidating Trustee shall have any duty to fund the Disputed Claim Reserve except from the Liquidating Trust Assets.
- 12.8 *Rounding*. Whenever any payment of a fraction of a cent would otherwise be called for, the actual Distribution shall reflect a rounding of such fraction down to the nearest cent.
- 12.9 *Cumulative Effect.* All the objection, estimation, and resolution procedures set forth in this Section are intended to be cumulative (where possible) and not exclusive of one another.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

SECTION 13. LIQUIDATING TRUST BENEFICIARIES

- 13.1 *Identification of Liquidating Trust Beneficiaries*. Each of the Liquidating Trust Beneficiaries shall be recorded and set forth in a schedule maintained by the Liquidating Trustee expressly for such purpose based upon its Allowed Claim in Class 3 or Class 4.
- 13.2 *Beneficial Interests Only.* The ownership of Trust Beneficial Interests shall not entitle any Trust Beneficiary to any title in or to the Liquidating Trust Assets or to any right to call for a partition or division of such Liquidating Trust Assets or to require an accounting, except as may be specifically provided herein.
- 13.3 *Evidence of Beneficial Interests.* Ownership of a Trust Beneficial Interest (a) shall be noted in the books and records of the Liquidating Trust and (b) shall not be evidenced by any certificate, note, or receipt or in any other form or manner whatsoever, except as maintained on the books and records of the Liquidating Trust by the Liquidating Trustee, including the Schedules.
- 13.4 *Conflicting Claims*. Except as otherwise provided in the Liquidating Trust Agreement, if any conflicting claims or demands are made or asserted with respect to a Trust Beneficial Interest, the Liquidating Trustee shall be entitled, at its sole election, to refuse to comply with any such conflicting claims or demands. In so refusing, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, may elect to make no payment or Distribution with respect to the beneficial interest represented by the claims or demands involved, or any part thereof, and the Liquidating Trustee or Co-Liquidating Trustee, as applicable, shall refer such conflicting claims or demands to the Bankruptcy Court, which shall have exclusive jurisdiction over resolution of such conflicting claims or demands and file a motion with the Bankruptcy Court to adjudicate any such conflicting claims or demands. In so doing, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, shall not be or become liable to any party for his/her refusal to comply with any of such conflicting claims or demands. The Liquidating Trustee or Co-Liquidating Trustee, as applicable, shall be entitled to refuse to act until either (a) the rights of the adverse claimants have been adjudicated by a Final Order or (b) all differences have been resolved by a written agreement among all of such parties and the Liquidating Trustee or Co-Liquidating Trustee, as applicable, which agreement shall include a complete release of the Liquidating Trust and the Liquidating Trustee (the occurrence of either (a) or (b) being referred to as a "Dispute Resolution" in this Section). Until a Dispute Resolution is reached with respect to such conflicting claims or demands, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, shall hold in a segregated interest-bearing account with a United States financial institution any

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

26

27

28

payments or Distributions from the Liquidating Trust as applicable, to be made with respect to the Beneficial Interest at issue. Promptly after a Dispute Resolution is reached, the Liquidating Trustee shall Transfer the payments and Distributions, if any, held in the segregated account, together with any interest and income generated thereon, in accordance with the terms of such Dispute Resolution.

13.5 *Limitation on Transferability*. As set forth in more detail in the Liquidating Trust Agreement, the Trust Beneficial Interests may not be transferred, sold, assigned, hypothecated, or pledged, except as they may be assigned or transferred by will, intestate succession, or operation of law.

SECTION 14. EXECUTORY AGREEMENTS

14.1 *General Treatment.* On the Effective Date, all Executory Agreements to which the Debtor is a party shall be deemed rejected as of the Effective Date, except for those Executory Agreements that (a) have been assumed or rejected pursuant to a Final Order of the Bankruptcy Court (including pursuant to any Sale Order), (b) are the subject of a separate motion to assume, assume and assign, or reject filed under § 365 on or before the Effective Date, (c) are specifically designated as a contract or lease to be assumed pursuant to the DAP Sale and no timely objection to the proposed assumption was filed; or (d) are specifically designated as a contract or lease to be assumed on the Schedule of Assumed Contracts and no timely objection to the proposed assumption has been filed. If the party to an Executory Agreement listed to be assumed in the Schedule of Assumed Contracts wishes to object to the proposed assumption (including with respect to the cure amounts), it shall do so within thirty (30) days from the service of the Schedule of Assumed Contracts. However, nothing in this Section shall cause the rejection, breach, or termination of any contract of insurance benefiting the Debtor and the Estate, the Debtor's officers, managers and directors and/or the Liquidating Trust. The Confirmation Order shall constitute an order of the Bankruptcy Court approving such assumptions and rejections, as applicable, pursuant to § 365, as of the Effective Date. Nothing in this Section shall be construed as an acknowledgement that a particular contract or agreement is executory or is properly characterized as a lease.

The non-Debtor parties to any rejected personal property leases shall be responsible for taking all steps necessary to retrieve the personal property that is the subject of such executory contracts and leases, and neither the Debtor nor the Liquidating Trust shall bear any liability for costs associated with such matters.

1

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

DENTONS US LLP

16 17

19

18

20 21

22 23

24

14.2 *Rejection Bar Date.* Claims arising out of the rejection of an Executory Agreement pursuant to the Plan must be filed with the Bankruptcy Court (or as otherwise provided for in the Debtor's notice of rejection) no later than thirty (30) days after the Effective Date. Any Claims not filed within such time period will be forever barred from assertion against the Debtor, the Liquidating Trustee, and the Liquidating Trust Assets. All such Claims for which Proofs of Claim are timely and properly filed and ultimately Allowed will be treated as General Unsecured Claims.

14.3 *Insurance Policies.* For the avoidance of doubt, the Debtor's rights with respect to all Insurance Policies under which the Debtor may be an insured beneficiary or assignee (including all Insurance Policies that may have expired prior to the Petition Date, all Insurance Policies in existence on the Petition Date, all Insurance Policies entered into by the Debtor after the Petition Date, and all Insurance Policies under which the Debtor holds rights to make, amend, prosecute, and benefit from claims) shall be Transferred to the Liquidating Trust (including, without limitation, for the Liquidating Trustee to pursue and prosecute any Causes of Action) on the Effective Date, unless any such Insurance Policy is otherwise cancelled by the Liquidating Trustee in its discretion. Notwithstanding any provision providing for the rejection of Executory Agreements, any Insurance Policy that is deemed to be an Executory Agreement shall neither be rejected nor assumed by operation of this Plan and shall be the subject of a specific motion by the Liquidating Trust, which shall retain the right to assume or reject any such Executory Agreements pursuant to and subject to the provisions of § 365 following the Effective Date, with all rights of the Insurers to object or otherwise contest such assumption or rejection being expressly reserved provided, that, the Liquidating Trustee may not reject (a) any extended reporting period (tail) coverage purchased by the Debtor and (b) any Insurance Policies assumed by the Debtor pursuant to an order of the Bankruptcy Court.

The Confirmation Order shall constitute a determination that no default by the Debtor exists with respect to any of the Insurance Policies requiring a cure payment and that nothing in the Sale Order, any underlying agreements or this Plan shall be construed or applied to modify, impair, or otherwise affect the enforceability of the Insurance Policies or any coverage thereunder with regard to any Claims or Causes of Action.

Notwithstanding anything to the contrary in the Confirmation Order or the Plan (including any other provision that purports to be preemptory or supervening), nothing shall in any way operate to impair, or have the effect of impairing the Insurers' legal, equitable or contractual rights, if any, in respect of any Claims (as defined by § 101(5)), and the rights of Insurers shall be determined under the Insurance Policies and under applicable nonbankruptcy law; provided, that, any Claim by an Insurer

2.5

against the Debtor or the Liquidating Trust shall also be determined under applicable bankruptcy law, and Plan and Confirmation Order provisions.

Nothing in the Plan or in the Confirmation Order shall preclude any Person from asserting in any proceeding any and all Claims, defenses, rights or causes of action that it has or may have under or in connection with any Insurance Policy, and nothing in the Plan or the Confirmation Order shall be deemed to waive any claims, defenses, rights or causes of action that any Person (including any Insurer) has or may have under the provisions, terms, conditions, defenses and/or exclusions contained in the subject Insurance Policies; *provided, that*, any Claims by an Insurer against the Debtor or the Liquidating Trust shall also be determined under applicable bankruptcy law, and Plan and Confirmation Order provisions. Nothing in this Plan shall diminish, impair or otherwise affect payments from the proceeds or the enforceability of any Insurance Policies that may cover (a) Claims by the Debtor, or (b) Claims against the Debtor or covered Persons.

SECTION 15. MEANS FOR IMPLEMENTATION OF THE PLAN

- 15.1 *General Settlement of Claims.* The Liquidating Trustee and Co-Liquidating Trustee are authorized and directed to make Distributions of the distributable Assets pursuant to and in accordance with the Plan. All Plan Distributions made to Holders of Allowed Claims in any Class are intended to be and shall be final. For the avoidance of doubt, the Plan itself shall not be deemed to be a settlement.
- 15.2 *Sale Transaction.* On March 13, 2023, the Bankruptcy Court entered the Sale Order authorizing and approving the DAP Sale, pursuant to § 363. Pursuant to the DAP Sale, the Debtor sold the Purchased Assets to DAP Health pursuant to the terms in the Asset Purchase Agreement.
- a) Cooperation During Transition Period. Between the entry of the Sale Order and the Closing of the DAP Sale, the Debtor and DAP Health cooperated pursuant to the Management Services Support Agreement. On July 31, 2023, the DAP Sale Closed and the Management Services Support Agreement terminated on its terms and that to the extent the Debtor or DAP Health may have claims, rights, or ongoing obligations to each other under the Management Services Support Agreement that survive the termination, such claims, rights, or ongoing obligations shall cease to exist on the Effective Date, and that the Post-Effective Date Debtor shall not inherit any claims, rights, benefits, or Causes of Action arising under,

345

67

8

9

10 11

12

13 14

15 16

17

18 19

202122

2324

252627

28

relating to, or in connection with the Asset Purchase Agreement, the Management Services Support Agreement, or any other agreement relating to the foregoing.

- b) **No Effect on Sale Order.** Nothing in the Plan is intended to, nor shall be construed to, alter any of the terms and conditions upon which the DAP Sale was approved as set forth in the Asset Purchase Agreement. Neither the Plan nor the Confirmation Order shall limit or otherwise affect any of the Bankruptcy Court's findings, conclusions, orders, and judgments as set forth in the Sale Order, and insofar as any of the protections afforded DAP Health by the Sale Order conflict with or contradict certain terms and conditions in the Plan or any findings, conclusions, orders, or judgments in the Confirmation Order, the Sale Order shall govern and control with respect to DAP Health.
- c) No Claims against DAP Health. Notwithstanding anything that may suggest otherwise in the Plan, the Plan Supplement, or any Schedule or Exhibit to either of the foregoing, or any other document executed in connection with Confirmation or this Chapter 11 Case, neither the Post-Effective Date Debtor nor any Holder of any Claim shall have any claim, cause of action, right, or recourse against DAP Health or the Purchased Assets on account of or in connection with the DAP Sale, the Asset Purchase Agreement, the Master Services Support Agreement, or any other agreement or document executed in connection with the consummation of the DAP Sale.
- 15.3 **DHCS Settlement Agreement.** On March 7, 2023, the Bankruptcy Court entered the DHCS 9019 Order authorizing and approving the DHCS Settlement among the Debtor, the Committee, and DHCS, which resolved the various disputes between the Debtor, the Committee, and DHCS. The DHCS Settlement Agreement sets forth the treatment of the DHCS Claim and provides for the allocation of the DAP Sale proceeds among DHCS and Holders of other Allowed Claims. DHCS will apply the DHCS Offset Amount to the DHCS Claim as set forth in the DHCS Settlement Agreement. Pursuant to the DHCS Settlement Agreement, the DHCS Claim will be reduced dollar for dollar by the DHCS' receipt of Specified Litigation Recoveries, which will be allocated as set forth therein. Finally, the DHCS Settlement Agreement provides for the Extended DHCS Bar Date in order for DHCS to file an additional Proof of Claim for Medi-Cal overpayments not included within the DHCS Claim as a General Unsecured Claim.

Nothing in the Plan is intended to, nor shall be construed to, alter any of the terms and conditions upon which the DHCS Settlement Agreement was approved as set forth therein. Neither the Plan nor the Confirmation Order shall limit or otherwise affect any of the Bankruptcy Court's findings, conclusions, orders, and judgments as set forth in the DHCS 9019 Order, and insofar as any of the protections afforded

1

5678

10 11

9

1213

14

1516

1718

19

2021

2223

24

2526

27

28

DHCS by the DHCS 9019 Order conflict with or contradict certain terms and conditions in the Plan or any findings, conclusions, orders, or judgments in the Confirmation Order, the DHCS 9019 Order shall govern and control with respect to DHCS

15.4 *Plan Funding*. This Plan will be funded from the following sources: (i) the Remaining Estate Funds; (ii) the Remaining Cash; (iii) Net Cash Proceeds; (iv) any refunds, deposits, or other monies owing to the Debtor which were not sold to DAP Health; (v) the Litigation Recoveries; (vi) any other monetary recoveries obtained by the Debtor prior to the Effective Date; and (vii) any other monetary recoveries obtained by the Liquidating Trustee after the Effective Date that do not constitute Purchased Assets.

15.5 Post-Effective Date Governance¶.

a) Post-Effective Date Debtor.

- Continued Limited Existence. On and after the Effective Date, i. the Post-Effective Date Debtor shall continue in existence for the purposes set forth herein, and retain its Nonprofit Status to the same extent as such status existed immediately prior to the Petition Date. No party shall take any action to interfere with, alter, terminate, or otherwise adversely affect the Nonprofit Status of the Post-Effective Date Debtor. Specifically, the Post-Effective Date Debtor shall continue in existence (i) to maintain the Provider Agreements for Medi-Cal and Medicare, and participate in the Medi-Cal and Medicare programs, until the CHOW is approved, and (ii) to collect or otherwise liquidate all amounts owing under the Provider Agreements until all payments due under such agreements have been received by the Post-Effective Date Debtor and, if appropriate, Transferred to the Liquidating Trust.
- ii. *No Further Approvals Required*. In performance of its duties hereunder, the Post-Effective Date Debtor shall have the rights and powers of a debtor in possession under § 1107, and such other rights, powers, and duties necessary, appropriate, advisable or convenient to effectuate the provisions of the Plan. Except to the extent provided in this Plan or the Confirmation Order, on and after the Effective Date, the Post-Effective Date Debtor shall not be required to obtain any approvals from the Bankruptcy Court, any court or Governmental Unit and/or provide any

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

notices under the Nonprofit Laws to implement the terms of the Plan.

- iii. <u>The Post-Effective Date Debtor's Books and Records.</u> The Debtor shall Transfer dominion and control over all of its books and records, in whatever form, manner or media, to the Liquidating Trustee on or as soon as reasonably practicable after the Effective Date.
- **Dissolution.** The Liquidating Trustee will cause the Postiv. Effective Date Debtor to be dissolved for all purposes under applicable non-bankruptcy law after (i) the CHOW is approved and (ii) the receipt of all payments related to Medi-Cal and Medicare. The Liquidating Trustee may dissolve the Post-Effective Date Debtor, earlier than as set forth herein, if it determines that the continued existence of the Post-Effective Date Debtor is not necessary to satisfy the foregoing conditions. Such dissolution shall occur without the necessity for any other or further actions to be taken by or on behalf of the Post-Effective Debtor, or payment of any fees, charges, penalties or other amounts required by applicable non-bankruptcy law; provided, however, that the Liquidating Trustee may in its discretion file any certificates of cancellation as may be appropriate in connection with dissolution of the Post-Effective Date Debtor.

b) Post-Effective Date Board of Directors.

- i. <u>Duties and Obligations</u>. The Post-Effective Date Board of Directors shall (i) fulfill its duties and obligations under the Post-Effective Date Debtor's bylaws and state and federal law, and (ii) oversee the Liquidating Trustee solely in his/her capacity as president of the Post-Effective Date Debtor consistent with the terms of this Plan.
- ii. <u>Resignation</u>. Any member of the Post-Effective Date Board of Directors may resign at any time upon not less than thirty (30) days' written notice to the Liquidating Trustee; provided, that, the Liquidating Trustee may waive such notice period.
- iii. <u>Replacement</u>. Notwithstanding anything in the bylaws to the contrary, in the event that a director serving on the Post-Effective

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Date Board of Directors resigns or is duly removed for cause, or in the event of the death of any such director or other occurrence rendering such director incapacitated or unavailable for a period of thirty (30) consecutive days, a replacement director shall be designated by the remaining members of the Post-Effective Date Board of Directors in consultation with the Liquidating Trustee. If a replacement director cannot be designated pursuant to this Plan, the Liquidating Trustee may file a motion with the Bankruptcy Court seeking appointment of a replacement director.

- iv. <u>Termination</u>. The terms of the Post-Effective Date Board of Directors shall expire upon the date they are no longer required under state law as to the Debtor, as applicable.
- v. <u>Limitation of Liability of the Post-Effective Date Board of Directors</u>. The liability of the Post-Effective Date Board of Directors shall be limited to the maximum extent permitted by law, including any exculpations under the articles of incorporation or bylaws of the Post-Effective Date Debtor.
- Operating Accounts for the Post-Effective Date Debtor. On the vi. Effective Date, or as soon thereafter as is practical and subject to the prior payment of the amounts required to be paid or reserved by the Post-Effective Date Debtor in cash on the Effective Date on account of Claims pursuant to this Plan, the Post-Effective Date Debtor shall establish and fund deposit accounts to serve as Operating Accounts for the Post-Effective Date Debtor for use in accordance with the Wind-down Budget, provided, however, the Post-Effective Date Debtor may, with the written consent of the Liquidating Trustee, utilize previously established deposit accounts for such purpose. The Liquidating Trustee shall be authorized to use the funds in the Operating Accounts as needed to preserve, administer, and continue the Operations for the Post-Effective Date Debtor, including paying all related costs and expenses associated, and collection of any amounts due to the Post-Effective Date Debtor under the Transition Services Agreement, each in accordance with the Wind-down Budget. After the Effective Date, all Cash or other proceeds generated by the Purchased Assets solely to the extent required to fund the Operating Accounts in accordance with the Wind-down Budget

3

4

5

15.6 Liquidating Trust.

6789

101112

13 14

1516

17

23

24

25

2627

28

during the Transition Period shall be excluded from the definition of the Remaining Cash.

Formation of the Liquidating Trust. On the Effective Date, the Liquidating Trust shall be established pursuant to the Liquidating Trust Agreement for the purpose of, inter alia, (a) administering the Liquidating Trust Assets including the Distributions and payments contemplated under the Plan, (b) prosecuting and/or resolving all Disputed Claims, (c) investigating and pursuing any Causes of Action the Debtor holds or may hold against any Entity, and (d) making all Distributions to the Beneficiaries provided for under the Plan. The Liquidating Trust is intended to qualify as a liquidating trust pursuant to Treas. Reg. § 301.7701-4(d), with no objective to continue or engage in the conduct of the trade or business, except to the extent reasonably necessary to, and consistent with, the liquidating purpose of the Liquidating Trust. Accordingly, the Liquidating Trustee shall, in an expeditious but orderly manner, make distribution to Holders of Allowed Claims subject to the terms of this Plan, liquidate and convert to Cash the remaining Liquidating Trust Assets, and make timely Distributions to the Beneficiaries of the proceeds thereof, and not unduly prolong the duration of the Liquidating Trust. Neither the Liquidating Trust nor the Liquidating Trustee or Co-Liquidating Trustee shall be or shall be deemed a successor-in-interest of the Debtor for any purpose other than as specifically set forth herein or in the Liquidating Trust Agreement.

b) Funding of the Liquidating Trust. On the Effective Date, the Liquidating Trust Assets shall vest automatically in the Liquidating Trust. The Plan shall be considered a motion pursuant to §§ 105, 363, and 365 for such relief. The transfer of the Liquidating Trust Assets to the Liquidating Trust shall be made for the benefit and on behalf of the Liquidating Trust Beneficiaries. The assets comprising the Liquidating Trust Assets will be treated for tax purposes as being transferred by the Debtor to the Liquidating Trust Beneficiaries pursuant to the Plan in exchange for their Allowed Claims and then by the Liquidating Trust Beneficiaries to the Liquidating Trust in exchange for the beneficial interests in the Liquidating Trust. The Liquidating Trust Beneficiaries shall be treated as the grantors and owners of the Liquidating Trust. Upon the transfer of the Liquidating Trust Assets, the Liquidating Trust Assets, and the Debtor will have no further interest in or with respect to the Liquidating Trust Assets.

Except to the extent definitive guidance from the IRS or a court of competent jurisdiction (including the issuance of applicable Treasury Regulations or the receipt by the Liquidating Trustee of a private letter ruling if the Liquidating Trustee so

1

6

7

5

8 9 10

111213

141516

17

18 19

2122

23

20

2425

2627

28

requests one) indicates that such valuation is not necessary to maintain the treatment of the Liquidating Trust as a liquidating trust for purposes of the Internal Revenue Code and applicable Treasury Regulations, as soon as possible after the Effective Date, the Liquidating Trustee shall make a good-faith valuation of the Liquidation Trust Assets. The valuation shall be used consistently by all parties (including, without limitation, the Debtor, the Liquidating Trust, and the Liquidating Trust Beneficiaries) for all federal income tax purposes.

The Liquidating Trustee shall be selected by the Debtor with the consent of the Committee, such consent not to be unreasonably withheld. The initial Liquidating Trustee shall be a representative from Ankura Consulting Group, LLC. The Co-Liquidating Trustee shall be selected by the Committee with the consent of the Debtor, such consent not to be unreasonably withheld. The initial Co-Liquidating Trustee shall be a representative from FTI Consulting, Inc. The Liquidating Trustee and Co-Liquidating Trustee shall be deemed appointed on the Effective Date, without application, notice, hearing or other order of the Bankruptcy Court. The appointment, duties, rights, and powers of the Liquidating Trustee and the Co-Liquidating Trustee are as set forth in the Liquidating Trust Agreement.

15.8 Rights and Powers of Liquidating Trustee and Co-Liquidating Trustee.

Liquidating Trustee. The Liquidating Trustee shall be deemed a) the Estate's representative in accordance with § 1123 and shall have all the rights and powers set forth in the Liquidating Trust Agreement, including, without limitation, the powers of a trustee under §§ 704 and 1106 and Bankruptcy Rule 2004 to act on behalf of the Liquidating Trust. Without limiting the foregoing, and except as provided for below with respect to the Co-Liquidating Trustee, the Liquidating Trustee will have the right to, among other things, (1) effect all actions and execute all agreements, instruments and other documents necessary to implement the provisions of the Plan and the Liquidating Trust Agreement; (2) liquidate the Liquidating Trust Assets; (3) investigate, prosecute, settle, abandon or compromise any Causes of Action the Debtor holds or may hold against any Entity; (4) make Distributions as contemplated hereby, (5) establish and administer any necessary reserves for Disputed Claims that may be required; (6) object to the Disputed Claims and prosecute, settle, compromise, withdraw or resolve in any manner approved by the Bankruptcy Court such objections; (7) assert or waive any attorney-client privilege on behalf of the Debtor and Estate with regard to acts or events during time periods prior to the Petition Date; and (8) employ and compensate professionals and other agents, including, without limitation, existing Professionals employed by the Debtor in accordance with the Liquidating Trust Agreement or the Plan, provided, however,

4

5

678

10 11

9

12 13

14

1516

17 18

19 20

22

23

21

242526

2728

that any such compensation shall be made only out of the Liquidating Trust Assets, to the extent not inconsistent with the status of the Liquidating Trust as a liquidating trust within the meaning of Treas. Reg. § 301.7701-4(d) for federal income tax purposes.

- b) **Co-Liquidating Trustee.** The Co-Liquidating Trustee shall have all the rights and powers set forth in the Liquidating Trust Agreement. To ensure expeditious resolution of any Disputed General Unsecured Claims and distributions to Holders of Allowed General Unsecured Claims, the Co-Liquidating Trustee will have the right to, in consultation with the Liquidating Trustee, to: (1) make Distributions to Holders of Allowed General Unsecured Claims; (2) object to the Disputed General Unsecured Claims and prosecute (or continue the prosecution of), settle, compromise, withdraw or resolve in any manner approved by the Bankruptcy Court such objections; and (3) employ and compensate professionals and other agents, including, without limitation, existing Professionals employed by the Debtor or Committee in accordance with the Liquidating Trust Agreement or the Plan, provided, however, that any such compensation shall be made only out of the Liquidating Trust Assets. The Co-Liquidating Trustee will resign (by written submission, email being sufficient to the Liquidating Trustee) upon (i) fully adjudicating all Disputed General Unsecured Claims except any Disputed General Unsecured Claims against Specified Litigation Defendants, and (ii) making the Final Distribution to the Holders of General Unsecured Claims except Specified Litigation Defendants.
- 15.9 *Fees and Expenses.* Subject to payment in full of all Allowed Administrative Claims, and except as otherwise ordered by the Bankruptcy Court, expenses incurred by the Liquidating Trust on or after the Effective Date shall be paid in accordance with the Liquidating Trust Agreement without further order of the Bankruptcy Court.
- 15.10 *Transfer of Beneficial Interests in the Liquidating Trust.* Liquidating Trust Interests shall not be transferable except upon death of the interest holder or by operation of law. The Liquidating Trust shall not have any obligation to recognize any transfer of Claims occurring after the Distribution Record Date.
- 15.11 *Litigation of Debtor's Causes of Action*. Except as otherwise provided in this Plan, all Causes of Action are retained, vested in the Liquidating Trust, and preserved pursuant to § 1123(b). From and after the Effective Date, all Causes of Action will be prosecuted or settled by the Liquidating Trustee. Except as otherwise provided in this Plan, to the extent any Causes of Action are already pending on the Effective Date, the Liquidating Trustee, as successor to the Debtor (in any derivative capacity or as an intervening party), will continue the prosecution of such Causes of Action and shall be substituted as plaintiff, defendant, or in any other capacity for the

Debtor pursuant to this Plan and the Confirmation Order on the Effective Date without need for any further motion practice or notice in any case, action, or matter.

- 15.12 *Full and Final Satisfaction*. Commencing upon the Effective Date, subject to the terms of this Plan and the Liquidating Trust Agreement, the Liquidating Trustee and Co-Liquidating Trustee, as applicable, shall be authorized and directed to distribute the amounts required under the Plan to the Holders of Allowed Claims according to the provisions of the Plan. Upon the Effective Date, all Debts of the Debtor shall be deemed fixed and adjusted pursuant to this Plan, and the Liquidating Trust shall have no liability on account of any Claims except as set forth in this Plan and in the Liquidating Trust Agreement. All payments and all distributions made by the Liquidating Trustee or Co-Liquidating Trustee under the Plan shall be in full and final satisfaction of all Claims against the Liquidating Trust; provided, however, that nothing contained in this Article 9 of the Plan, or in any other provision of this Plan, shall be deemed to constitute or result in a discharge of the Debtor under § 1141(d).
- 15.13 *Employment and Compensation of Professionals*. In accordance with the Liquidating Trust Agreement, the Liquidating Trust may employ such counsel, advisors, and other professionals selected by the Liquidating Trustee or Co-Liquidating Trustee (which may be the same professionals employed by the Post-Effective Date Debtor or the Committee) that the Liquidating Trustee and Co-Liquidating Trustee reasonably require to perform its responsibilities under the Plan without further order from the Bankruptcy Court.
- 15.14 *No Further Court Authorization*. Except as provided herein or the Confirmation Order, the Liquidating Trustee and Co-Liquidating Trustee will continue the orderly administration of the Liquidating Trust Assets and otherwise implement the provisions of this Plan without necessity of any further order of the Bankruptcy Court or approval or consent of any Governmental Unit, including under the Nonprofit Laws. Further, except as provided herein or the Confirmation Order, the Liquidating Trustee and Co-Liquidating Trustee will continue their oversight and related responsibilities pursuant to the Plan and Liquidating Trust Agreement without necessity of any further order of the Bankruptcy Court or other Governmental Unit, including under the Nonprofit Laws.
- 15.15 *Dissolution of the Committee*. On the Effective Date, the Committee will dissolve, and the members of the Committee and the Committee's Professionals will cease to have any role arising from or relating to the Chapter 11 Case, except in connection with final fee applications of Professionals for services rendered prior to the Effective Date (including the right to object thereto). The Professionals retained by the Committee and the members thereof will not be entitled to assert any fee claims for any services rendered to the Committee or expenses incurred in the service of the

5

6

7

8

1

9 10 11

12 13 14

> 16 17 18

15

19 20

21 22 23

25

24

26

27

28

Committee after the Effective Date, except for reasonable fees for services rendered, and actual and necessary costs incurred, in connection with any applications for allowance of Professional Fees pending on the Effective Date or filed and served after the Effective Date. Nothing in the Plan shall prohibit or limit the ability of the Debtor's or Committee's Professionals to represent the Liquidating Trustee or to be compensated or reimbursed per the Plan and the Liquidating Trust Agreement in connection with such representation.

15.16 Coordination Between Post-Effective Date Debtor and the Liquidating Trust. Notwithstanding anything herein to the contrary, in furtherance of the purposes of the Liquidating Trust, at the request of the Liquidating Trustee or Co-Liquidating Trustee, as applicable, the Post-Effective Date Debtor (including, without limitation, the Post-Effective Date Debtor's employees, agents and/or professionals) shall be authorized to provide assistance and services to, or otherwise act on behalf of, the Liquidating Trustee or Co-Liquidating Trustee, as applicable, in the performance of the Liquidating Trustee's or Co-Liquidating Trustee's duties, as applicable, under the Plan and the Liquidating Trust. Without limitation on the foregoing, the Post-Effective Date Debtor shall be authorized to assist in the reconciliation and administration of Claims, and assist in the liquidation and/or collection of Liquidating Trust Assets (including, without limitation, litigation claims). The Liquidating Trustee shall oversee all such services provided on behalf of the Liquidating Trust.

15.17 Destruction and Abandonment of Books and Records. Except as otherwise provided in this subsection or the Liquidating Trust Agreement, on or after the Effective Date, pursuant to § 554(a), the Liquidating Trustee is authorized, from time to time, without further application to the Bankruptcy Court or notice to any party, to abandon or otherwise destroy documents and records (whether in electronic or paper format) that he or she determines, in his/her reasonable business judgment, are no longer necessary to the administration of either the Chapter 11 Case or the Plan, notwithstanding any federal, state, or local law or requirement requiring the retention of the applicable documents or records; provided, that, sixty (60) days prior to any abandonment or destruction, the Liquidating Trustee will give notice to any Insurer requesting notice prior to the Confirmation Date and a general description of the documents to be abandoned or destroyed, and the Insurer shall have thirty (30) days thereafter to request, at its sole expense, copies of the documents relevant to the defense or indemnity claims covered by that Insurer. The Insurer and the Liquidating **DENTONS US LLP**

3

1

4 5

67

8

9

111213

141516

17 18 19

2122

23

20

242526

27

28

indemnity of claims covered by that Insurer.

15.18 *Mutuality Preserved.* Unless specifically agreed

15.18 *Mutuality Preserved.* Unless specifically agreed in writing by the Debtor, the Liquidating Trustee, or Co-Liquidating Trustee, as applicable, nothing in the Plan constitutes a waiver of the requirements for setoff under § 553.

Trustee shall cooperate in limiting the request to documents relevant to defense or

SECTION 16. EFFECT OF CONFIRMATION

- 16.1 *Binding Effect of the Plan.* The provisions of the confirmed Plan shall bind the Debtor, the Liquidating Trust, the Liquidating Trustee, the Co-Liquidating Trustee, any Entity acquiring property under the Plan, any Beneficiary, and any Creditor, whether or not such Creditor has filed a Proof of Claim in the Chapter 11 Case, whether or not the Claim of such Creditor is impaired under the Plan, and whether or not such Creditor has accepted or rejected the Plan. All Claims and Debts shall be fixed and adjusted pursuant to the Plan. The Plan shall also bind any taxing authority, recorder of deeds, or similar official for any county, state, or Governmental Unit or parish in which any instrument related to under the Plan or related to any transaction contemplated under the Plan is to be recorded with respect to any taxes of the kind specified in § 1146(a).
- 16.2 *Vesting of Assets.* Upon the Effective Date, title to all property of the Estate of the Debtor in the Chapter 11 Case shall vest in the Liquidating Trust and shall be retained by the Liquidating Trust for the purposes contemplated under this Plan pursuant to the Liquidating Trust Agreement. Without limiting the generality of the foregoing, all Causes of Action the Debtor holds or may hold against any Entity, recoveries from any Causes of Action, and all resulting Liquidating Trust Assets shall vest in the Liquidating Trust upon the Effective Date and shall no longer constitute property of the Estate.
- 16.3 No Discharge. Pursuant to § 1141(d), the Debtor will not receive a discharge under this Plan.
- 16.5 **Property Free and Clear.** Except as otherwise provided in the Plan or the Confirmation Order, all property that shall vest in the Liquidating Trust shall be free and clear of all Claims, Liens, charges, or other encumbrances of Creditors, and in relevant documents, agreements, and instruments contained in the Plan Supplement. Following the Effective Date, the Liquidating Trustee may Transfer and dispose of any such property free of any restrictions imposed by the Bankruptcy Code or the Bankruptcy Rules and without further approval of the Bankruptcy Court or

4

9

7

12 13

> 14 15

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

DENTONS US LLP

16 17

18 19

20 21

22

23 24

25

26

27

notice to Creditors, except as may otherwise be required under the Plan or the Confirmation Order.

SECTION 17. EXCULPATIONS, INJUNCTIONS, AND RELEASES

17.1 Extension of Existing Injunctions and Stays. Unless otherwise provided herein, all injunctions or stays arising under §§ 105 or 362, any order entered during the Chapter 11 Case under §§ 105 or 362 or otherwise, and in existence on the Effective Date, shall remain in full force and effect until the closing of the Chapter 11 Case.

17.2 Releases.

Debtor Release. Pursuant to § 1123(b), and except as otherwise specifically provided in the Plan, for good and valuable consideration, on and after and subject to the occurrence of the Effective Date, the Debtor and its estate shall release each Released Party, and each Released Party is deemed released by the Debtor and the estate from any and all claims, obligations, rights, suits, damages, Causes of Action, remedies, and liabilities whatsoever, including any derivative claims, asserted or assertable on behalf of any of the Debtor or its estate, as applicable, whether known or unknown, foreseen or unforeseen, asserted or unasserted, accrued or unaccrued, matured or unmatured, determined or indeterminable, disputed or undisputed, liquidated or unliquidated, or due or to become due, existing or hereinafter arising, in law, equity, or otherwise, that the Debtor or the estate would have been legally entitled to assert in its own right, or on behalf of the Holder of any Claim or other entity, based on or relating to, or in any manner arising from, in whole or in part, the Debtor, the Debtor's liquidation, the Chapter 11 Case, the purchase, sale, transfer of any security, asset, right, or interest of the Debtor, the DAP Sale, the subject matter of, or the transactions or events giving rise to, any Claim that is treated in the Plan, the business or contractual arrangements between any Debtor and any Released Party, the treatment of Claims prior to or in the Chapter 11 Case, the negotiation, formulation, or preparation of the Plan or related agreements, instruments, or other documents, any other act or omission, transaction, agreement, event, or other occurrence taking place on and before the Petition Date, other than claims or liabilities arising out of or relating to any act or omission of a Released Party that constitutes fraud, willful misconduct, or gross negligence; provided, that, the foregoing Debtor Release shall not operate to waive or release any obligations of any party under the Plan or any other document, instrument, or agreement executed to implement the Plan. For avoidance of doubt, the foregoing Debtor Release does not

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

release any of the Debtor's claims, obligations, rights, suits, damages, Causes of Action, remedies, and liabilities with respect thereto.

Entry of the Confirmation Order shall constitute the Bankruptcy Court's approval, pursuant to Bankruptcy Rule 9019, of the Debtor Release, which includes by reference each of the related provisions and definitions contained herein, and further, shall constitute the Bankruptcy Court's finding that the Debtor Release is: (a) in exchange for the good and valuable consideration provided by the Released Parties; (b) a good faith settlement and compromise of the Claims released by the Debtor Release; (c) in the best interests of the Debtor and all Holders of Claims; (d) fair, equitable and reasonable; (e) given and made after due notice and opportunity for hearing; and (f) a bar to the Debtor or its estate asserting any Claim or Cause of Action released pursuant to the Debtor Release.

Third Party Release. On, and as of, the Effective Date and for b) good and valuable consideration, the receipt and sufficiency of which are acknowledged, the Released Parties shall be forever released (the "Third Party Release") from any and all claims, obligations, actions, suits, rights, debts, accounts, causes of action, remedies, avoidance actions, agreements, promises, damages, judgments, demands, defenses, and liabilities throughout the world under any law or court ruling through the Effective Date (including all claims based on or arising out of factors or circumstances that existed as of or prior to the Effective Date, including claims based on negligence or strict liability, and further including any derivative claims asserted on behalf of the Debtor, whether known or unknown, foreseen or unforeseen, existing or hereinafter arising, in law, equity, or otherwise) which the Debtor, its estate, Creditors, or other persons receiving or who are entitled to receive distributions under the Plan may have against any of them in any way related to this Chapter 11 Case, the negotiation, formulation, or preparation of the Plan or related agreements, instruments, or other documents, any other act or omission, transaction, agreement, event, or other occurrence taking place on and before the Petition Date, and related to the Debtor (or its predecessors), its business and/or its assets; provided, however, that the foregoing releases are granted only by (a) Creditors who returned a Ballot; and (b) Creditors who were sent a Solicitation Package or a Release Opt-Out Election Form, but either (i) did not vote; or (ii) did not return a Release Opt-Out Election Form; provided, however, that the release provided in this section shall not apply to (A) any Creditor whose Claim is not Allowed either in whole or in part; or (B) any Creditor in category (b) above if the Solicitation Package or Release Opt-Out Election Form was returned to the Debtor as undelivered and that such Creditor did not otherwise submit a Ballot; and provided further, however, that the release provided in this Section shall not extend to any claims by any Governmental Unit

3

with respect to criminal liability under applicable law, willful misconduct or bad faith under applicable law, ultra vires acts under applicable law.

456

c) Limitation of Claims Against the Liquidating Trust. As of the Effective Date, except as provided in this Plan or the Confirmation Order, all Persons shall be precluded from asserting against the Liquidating Trust any other or further Claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action, and liabilities whatsoever, relating to the Debtor based upon any acts, omissions, liabilities, transactions, occurrences, or other activity of any nature that occurred prior to the Effective Date.

8 9

7

10 11

12

1314

15

1617

181920

2122

2324

25

26

2728

d) <u>WAIVER OF LIMITATIONS ON RELEASES</u>. THE LAWS OF SOME STATES (FOR EXAMPLE, CALIFORNIA CIVIL CODE § 1542) PROVIDE, IN WORDS OR SUBSTANCE, THAT A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS/HER/ITS FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY THE RELEASING PARTY MUST HAVE MATERIALLY AFFECTED THE RELEASING PARTY'S DECISION TO RELEASE. THE RELEASING PARTIES IN SECTIONS 17.2(a)-(c) OF THE PLAN ARE DEEMED TO HAVE WAIVED ANY RIGHTS THEY MAY HAVE UNDER SUCH STATE LAWS AS WELL AS UNDER ANY OTHER STATUTES OR COMMON LAW PRINCIPLES OF SIMILAR EFFECT.

17.3 Injunctions.

General Injunction. Except as otherwise expressly provided a) herein, all Persons who have held, currently hold or may hold a Claim against the Debtor are permanently enjoined on and after the Effective Date from taking any action in furtherance of such Claim or any other Cause of Action released and discharged under the Plan, including, without limitation, the following actions against any Released Party: (a) commencing, conducting or continuing in any manner, directly or indirectly, any action or other proceeding with respect to a Claim; (b) enforcing, levying, attaching, collecting or otherwise recovering in any manner or by any means, whether directly or indirectly, any judgment, award, decree or order with respect to a Claim; (c) creating, perfecting or enforcing in any manner, directly or indirectly, any lien or encumbrance of any kind with respect to a Claim; (d) asserting any setoff, right of subrogation or recoupment of any kind, directly or indirectly, against any debt, liability or obligation due to the Debtor, the Post-Effective Date Debtor or the Liquidating Trust with respect to a Claim; or (e) commencing, conducting or continuing any proceeding that does not conform to or comply with or is contradictory to the provisions of this Plan; provided, however, that nothing in this injunction shall preclude the Holders of Claims against the Debtor

1

6

7

5

8 9 10

11 12 13

> 14 15

16 17

18

19 20

21 22 23

24 2.5

27

28

26

from enforcing any obligations of the Debtor, the Post-Effective Date Debtor, the Liquidating Trust, the Liquidating Trustee, or Co-Liquidating Trustee under this Plan and the contracts, instruments, releases and other agreements delivered in connection herewith, including, without limitation, the Confirmation Order, or any other order of the Bankruptcy Court in the Chapter 11 Case. By accepting a Distribution made pursuant to this Plan, each Holder of an Allowed Claim shall be deemed to have specifically consented to the injunctions set forth in this Section.

- Other Injunctions. The Post-Effective Date Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, the Post-Effective Date Board of Directors, or the Liquidating Trust and their respective members, directors, officers, agents, attorneys, advisors or employees shall not be liable for actions taken or omitted in its or their capacity as, or on behalf of, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, or the Liquidating Trust (as applicable), except those acts found by Final Order to arise out of its or their willful misconduct, gross negligence, fraud, and/or criminal conduct, and each shall be entitled to indemnification and reimbursement for fees and expenses in defending any and all of its or their actions or inactions in its or their capacity as, or on behalf of the Post-Effective Date Board of Directors, the Post-Effective Date Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, or the Liquidating Trust (as applicable), except for any actions or inactions found by Final Order to involve willful misconduct, gross negligence, fraud, and/or criminal conduct. Any indemnification claim of the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, and the other parties entitled to indemnification under this subsection shall be satisfied from the Liquidating Trust Assets. The parties subject to this Section shall be entitled to rely. in good faith, on the advice of retained professionals, if any.
- 17.4 *Exculpation.* To the maximum extent permitted by applicable law, each Exculpated Party shall not have or incur any liability for any act or omission in connection with, related to, or arising out of the Chapter 11 Case (including, without limitation, the filing of the Chapter 11 Case), the marketing and the DAP Sale, the Plan and any related documents (including, without limitation, the negotiation and consummation of the Plan, the pursuit of the Effective Date, the administration of the Plan, or the property to be distributed under the Plan), or each Exculpated Party's exercise or discharge of any powers and duties set forth in the Plan, except with respect to the actions found by Final Order to constitute willful misconduct, gross negligence, fraud, or criminal conduct, and, in all respects, each Exculpated Party shall be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan. Without limitation of the foregoing, each such Exculpated Party shall be released and exculpated from any and all Causes of Action

4

1

567

8 9 10

11121314

16 17 18

15

19 20

212223

2425

27

28

26

that any Person is entitled to assert in his/her/their own right or on behalf of any other Person, based in whole or in part upon any act or omission, transaction, agreement, event or other occurrence in any way relating to the subject matter of this Section.

17.5 No Recourse. If a Claim is Allowed in an amount for which after application of the payment priorities established by this Plan (including, without limitation, in Sections 8 and 10 hereof) there is insufficient value to provide a recovery equal to that received by other Holders of Allowed Claims in the respective Class, no Claim Holder shall have recourse for any such deficiency against any of the Released Parties, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, or the Liquidating Trust. However, except as specifically stated otherwise in this Plan, nothing in this Plan shall modify any right of a Holder of a Claim under § 502(j). The obligations under this Plan of the Debtor's Estate shall (i) be contractual only and shall not create any fiduciary relationship and (ii) be obligations of the Debtor's Estate only and no individual acting on behalf of the Debtor, the Committee, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, the Co-Liquidating Trustee, the Oversight Committee, or otherwise, shall have any personal or direct liability for these obligations. Approval of the Plan by the Confirmation Order shall not in any way limit the foregoing.

17.6 Post-Confirmation Liability of Liquidating Trustee and the Co-Liquidating Trustee. The Liquidating Trustee and the Co-Liquidating Trustee, together with their respective consultants, agents, advisors, attorneys, accountants, financial advisors, other representatives and the professionals engaged by the foregoing (collectively, the "Indemnified Parties") shall not be liable for any and all liabilities, losses, damages, claims, causes of action, costs and expenses, including but not limited to attorneys' fees arising out of or due to their actions or omissions, or consequences of such actions or omissions, to the Holders of Claims for any action or inaction taken in good faith in connection with the performance or discharge of their duties under this Plan, except the Indemnified Parties will be liable for actions or inactions that are grossly negligent, fraudulent, or which constitute willful misconduct (in each case, liability shall be subject to determination by final order of a court of competent jurisdiction). However, any act or omission taken with the approval of the Bankruptcy Court, and not inconsistent therewith, will be conclusively deemed not to constitute gross negligence, fraud or willful misconduct. In addition, the Liquidating Trust and the Estate shall, to the fullest extent permitted by the laws of the State of California, indemnify and hold harmless the Indemnified Parties from and against and with respect to any and all liabilities, losses, damages, claims, costs and expenses, including but not limited to attorneys' fees arising out of or due to their actions or omissions, or consequences of such actions or omissions, with respect to the

1

5 6

7

8

9

10

11 12

13

601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704

213 623 9300

DENTONS US LLP

14 15

16 17

18

19

20 21 22

23 24

25 26

28

27

Liquidating Trust and the Estate or the implementation or administration of the Plan if the Indemnified Party acted in good faith and in a manner reasonably believed to be in or not opposed to the best interest of the Liquidating Trust and the Estate. To the extent the Liquidating Trust indemnifies and holds harmless the Indemnified Parties as provided above, the legal fees and related costs incurred by counsel to the Liquidating Trustee or the Co-Liquidating Trustee in monitoring and participating in the defense of such claims giving rise to the right of indemnification shall be paid as expenses of the Liquidating Trust. All rights of the Persons exculpated and indemnified pursuant hereto shall survive confirmation of the Plan.

17.7 Preservation of Rights of Action.

Vesting of Causes of Action. Except as otherwise provided in the Plan or Confirmation Order, including any Cause of Action that is expressly waived, relinquished, exculpated, released, settled, or compromised under the Plan or Confirmation Order (including, without limitation, pursuant to the Debtor/Estate Release), (i) in accordance with § 1123(b)(3), any Causes of Action that the Debtor holds or may hold against any Entity shall vest upon the Effective Date in the Liquidating Trust; (ii) after the Effective Date, the Liquidating Trustee shall have the exclusive right to institute, prosecute, abandon, settle, or compromise any Causes of Action the Estate holds or may hold against any Entity constituting Liquidating Trust Assets, in accordance with the terms of the Plan and the Liquidating Trust Agreement, as applicable, and without further order of the Bankruptcy Court, in any court or other tribunal, including, without limitation, in an adversary proceeding filed in the Chapter 11 Case; and (iii) Causes of Action and recoveries therefrom shall remain the sole property of the Liquidating Trust, and Holders of Claims shall have no direct right or interest in to any such Causes of Action or recoveries.

Preservation of All Causes of Action Not Expressly Settled or b) Released. Unless a Cause of Action against a Holder of a Claim or other Entity is expressly waived, relinquished, released, compromised, or settled in the Plan (including, without limitation, pursuant to the Debtor/Estate Release) and/or or any Final Order (including the Confirmation Order), the Debtor and the Liquidating Trustee expressly reserve such retained Cause of Action (collectively, the "Retained Causes of Action") for later adjudication by the Liquidating Trustee (including, without limitation, Causes of Action not specifically identified or described in the Plan Supplement or elsewhere, or of which the Debtor may be presently unaware, or which may arise or exist by reason of additional facts or circumstances unknown to the Debtor at this time, or facts or circumstances that may change or be different from those the Debtor now believe to exist) and, therefore, no preclusion doctrine, including, without limitation, the doctrines of res judicata, collateral estoppel, issue

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

preclusion, claim preclusion, waiver, estoppel (judicial, equitable, or otherwise) or laches shall apply to such Retained Causes of Action upon or after the entry of the Confirmation Order or Effective Date based on the Plan or Confirmation Order, except where such Causes of Action have been released or otherwise resolved by a Final Order (including the Confirmation Order). In addition, the Debtor and Liquidating Trustee expressly reserve the right to pursue or adopt claims alleged in any lawsuit in which a Debtor is a defendant or interested party against any Entity, including, without limitation, the plaintiffs or co-defendants in such lawsuits.

- i. The Retained Causes of Action preserved hereunder include, without limitation, the following claims, rights, or other causes of action:
- Against the Excluded Parties and/or any other party not ii. expressly released pursuant to this Plan;
- iii. that constitute Avoidance Actions:
- relating to pending litigation, including, without limitation, the iv. suits, administrative proceedings, executions, garnishments, and attachments listed in the Debtor's Schedules;
- against vendors, suppliers of goods or services (including v. attorneys, accountants, consultants, or other professional service providers), utilities, contract counterparties, and other parties for, including but not limited to: (A) services rendered; (B) over- and under-payments, back charges, duplicate improper holdbacks, payments, deposits, warranties, guarantees, indemnities, setoff, or recoupment; (C) failure to fully perform or to condition performance on additional requirements under contracts with any one or more of the Debtors; (D) wrongful or improper termination, suspension of services, or supply of goods, or failure to meet other contractual or regulatory obligations; (E) indemnification and/or warranty claims; or (F) turnover Causes of Action arising under §§ 542 or 543;
- vi. against health plans, payors, and other related providers;
- against landlords or lessors, including, without limitation, for vii. erroneous charges, overpayments, returns of security deposits, indemnification, or for environmental claims;

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

26

27

28

- viii. arising against current or former tenants or lessees, including, without limitation, for non-payment of rent, damages, and holdover proceedings;
 - ix. arising from damage to any of the Debtor's property;
 - x. relating to claims, rights, or other Causes of Action the Debtor may have to interplead third parties in actions commenced against the Debtor;
 - xi. for collection of a debt or other amount owed to the Debtor;
 - xii. against insurance carriers, reinsurance carriers, underwriters, surety bond issuers or other related or similar parties relating to coverage, indemnity, contribution, reimbursement, or other matters; and
- xiii. arising under or relating to the Asset Purchase Agreement and related documents including, but not limited to, enforcement of such agreements by the Debtor's Estate and/or breaches of any and/or all of such agreements by the applicable non-Debtor parties.
- Transactions Subject to Review by Liquidating Trustee or Co-Liquidating Trustee. Subject to the immediately preceding paragraph, any Entity to which the Debtor has incurred an obligation (whether on account of services, the purchase or sale of goods, or otherwise), or that has received services from the Debtor or a Transfer of money or property of the Debtor, or that has received services from the Debtor or a Transfer or money or property of the Debtor, or that has transacted business with the Debtor, or that has leased property from the Debtor, should assume and is hereby advised that any such obligation, Transfer, or transaction may be reviewed by the Liquidating Trustee or Co-Liquidating Trustee, as applicable, subsequent to the Effective Date and may be the subject of an action after the Effective Date, regardless of whether (i) such Entity has filed a Proof of Claim against the Debtor in the Chapter 11 Case; (ii) the Debtor, the Committee, the Liquidating Trustee, or Co-Liquidating Trustee have objected to any such Entity's Proof of Claim; (iii) any such Entity's Claim was included in the Schedules; (iv) the Debtor, the Committee, the Liquidating Trustee, or Co-Liquidating Trustee have objected to any such Entity's Scheduled Claim; (v) any such Entity's Scheduled Claim has been identified by the Debtor, the Committee, the Liquidating Trustee, or Co-Liquidating Trustee as disputed, contingent, or unliquidated; or (vi) the Debtor, the Liquidating

1

3 4

5

6 7

8

9 10

11 12 13

> 14 15

16 17

18

19 20

22

21

24

23

2526

27

28

Trustee, or Co-Liquidating Trustee have identified any potential claim or Cause of Action against such Entity herein.

17.8 Termination of Responsibilities of the Patient Care Ombudsman. On the Closing Date, the duties and responsibilities of the Patient Care Ombudsman were terminated and the Patient Care Ombudsman was discharged from his duties as Patient Care Ombudsman and is not required to file any further reports or perform any additional duties as Patient Care Ombudsman. No person or entity may seek discovery in any form, including but not limited to by motion, subpoena, notice of deposition or request or demand for production of documents, from the Patient Care Ombudsman or his agents, professionals, employees, other representatives, designees or assigns (collectively, with the Patient Care Ombudsman, the "Ombudsman Parties") with respect to any matters arising from or relating in any way to the performance of the duties of the Patient Care Ombudsman in this Chapter 11 Case, including, but not limited to, pleadings, reports or other writings filed by the Patient Care Ombudsman in connection with this Chapter 11 Case. Nothing herein shall in any way limit or otherwise affect the obligations of the Patient Care Ombudsman under confidentiality agreements, if any, between the Patient Care Ombudsman and any other person or entity or shall in any way limit or otherwise affect the Patient Care Ombudsman's obligation, under § 333(c)(1) or other applicable law or Bankruptcy Court Orders, to maintain patient information, including patient records, as confidential, and no such information shall be released by the Patient Care Ombudsman without further order of the Bankruptcy Court.

SECTION 18. CONDITIONS PRECEDENT TO EFFECTIVE DATE

- 18.1 *Conditions Precedent to Confirmation of Plan.* The confirmation of the Plan shall be conditioned upon the Bankruptcy Court entering the Confirmation Order in form and substance satisfactory to the Plan Proponents.
- 18.2 *Conditions to Effective Date.* The following are conditions precedent to the Effective Date:
 - a) The Confirmation Order shall have been entered by this Court in form and substance acceptable to the Plan Proponents, which Confirmation Order shall not have been terminated, suspended, vacated or stayed, and shall not have been amended or modified after entry without the consent of the Plan Proponents;

- b) The Liquidating Trustee shall have accepted the terms of the Liquidating Trustee's service and compensation, and such terms and compensation shall have been approved by the Court in the Confirmation Order;
- c) The Co-Liquidating Trustee shall have accepted the terms of the Co-Liquidating Trustee's service and compensation, and such terms and compensation shall have been approved by the Court in the Confirmation Order;
- d) With respect to all other documents and agreements necessary to implement the Plan: (1) all conditions precedent to such documents and agreements (other than any conditions precedent related to the occurrence of the Effective Date) shall have been satisfied or waived pursuant to the terms of such documents or agreements; (2) such documents and agreements shall have been tendered for delivery to the required parties and have been approved by any required parties and, to the extent required, filed with and approved by the applicable authorities in the relevant jurisdiction; and (3) such documents and agreements shall have been effected or executed; and
- e) All other actions, authorizations, consents, and regulatory approvals required (if any) and necessary to implement the provisions of the Plan shall have been obtained, effected, or executed in a manner acceptable to the Plan Proponents or, if waivable, waived by the Person or Persons (or Entity or Entities) entitled to the benefit thereof.
- 18.3 *Waiver of Condition.* The Plan Proponents may waive the conditions to effectiveness of this Plan, set forth in Section 18.2 hereof, without leave of the Bankruptcy Court and without any formal action, other than proceeding with confirmation of this Plan and filing a notice of confirmation with the Bankruptcy Court. To the extent that the Debtor believes that it is unable to comply with the conditions to the effectiveness of this Plan, set forth in Section 18.2 hereof, the Plan Proponents reserve the right to amend the Plan at such time (in accordance with the terms hereof) to address such inability.

SECTION 19. RETENTION OF JURISDICTION

- 19.1 *Bankruptcy Court Jurisdiction*. Unless otherwise provided herein or in the Confirmation Order, on and after the Effective Date, the Bankruptcy Court shall retain jurisdiction over all matters arising in, arising under, or related to the Chapter 11 Case. Without limiting the foregoing, the Bankruptcy Court shall retain jurisdiction to:
 - (a) allow, disallow determine, liquidate, classify, estimate, or establish the priority or secured or unsecured status of any Claim, including the resolution of any request for payment of any Administrative Claim or Professional Claim and the resolution of any objections to the allowance or priority of Claims, and the resolution of any claim objections brought by the Debtor and/or the Committee, by the Liquidating Trustee and Co-Liquidating Trustee on behalf of the Liquidating Trust;
 - (b) resolve any matters related to the assumption, assumption and assignment, or rejection of any Executory Agreement to which the Debtor is a party and to hear, determine and, if necessary, liquidate, any Claims arising from, or cure amounts related to, such assumption or rejection;
 - (c) determine any motion, adversary proceeding, application, contested matter, and other litigated matter pending on or commenced after the Effective Date, including, without limitation, any and all Causes of Action preserved under the Plan commenced prior to, on, or after the Effective Date;
 - (d) ensure that Distributions to Holders of Allowed Claims are accomplished in accordance with the Plan;
 - (e) hear and determine matters relating to claims with respect to the Debtor's director and officer insurance;
 - (f) enter, implement or enforce such orders as may be appropriate in the event that the Confirmation Order is for any reason stayed, reversed, revoked, modified, or vacated;
 - (g) issue injunctions, enter and implement other orders, and take such other actions as may be necessary or appropriate to restrain interference by any Person with the consummation,

implementation or enforcement of this Plan, the Confirmation Order or any other order of the Bankruptcy Court, including, without limitation, any actions relating to the Nonprofit Status of the Post-Effective Date Debtor;

- (h) resolve a dispute with respect to and/or otherwise appoint a replacement of the Liquidating Trustee or Co-Liquidating Trustee;
- (i) hear and determine any application to modify this Plan in accordance with § 1127, to remedy any defect or omission or reconcile any inconsistency in this Plan, the Disclosure Statement, any contract, instrument, release, or other agreement or document created in connection therewith, or any order of the Bankruptcy Court, including the Confirmation Order, in such a manner as may be necessary to carry out the purposes and effects thereof;
- (j) hear and determine all applications under §§ 330, 331, and 503(b) for awards of compensation for services rendered and reimbursement of expenses incurred prior to the Effective Date;
- (k) hear and determine disputes arising in connection with the interpretation, implementation, obligation or enforcement of this Plan, the Confirmation Order, any transactions or payments contemplated in the Plan, or any agreement, instrument, or other document governing or relating to any of the foregoing;
- (l) take any action and issue such orders as may be necessary to construe, enforce, implement, execute and consummate this Plan, including all contracts, instruments, releases, and other agreements or documents created in connection therewith, or to maintain the integrity of this Plan following consummation;
- (m) determine such other matters and for such other purposes as may be provided in the Plan and/or the Confirmation Order;
- (n) hear and determine matters concerning state, local, and federal taxes in accordance with §§ 346, 505, and 1146, including without limitation, (i) any requests for expedited determinations under § 505(b) filed, or to be filed, with respect to tax returns for any and all taxable periods ending after the Petition Date through, and including, the date of Final Distribution under the Plan, and (ii)

Case	22-02384-LT11	Filed 02/05/23 Entered 02/05/23 13:00:50 Doc 1208 Pg. 109 of 160									
1		any other matters relating to the Nonprofit Status of the Post-									
2		Effective Date Debtor;									
3	(o)	(o) hear and determine any other matters related hereto and not inconsistent with the Bankruptcy Code and Title 28 of the United									
4		States Code;									
5	(p)	(p) authorize recovery of all assets of any of the Debtor and propert									
6	G.	of the Debtor's Estate, wherever located;									
7	(q)	(q) consider any and all claims against each Released Party involving									
8		or relating to the administration of the Chapter 11 Case, any									
9		rulings, orders, or decisions in the Chapter 11 Case, or any aspects of the Debtor's Chapter 11 Case and the events leading up to the commencement of the Chapter 11 Case, including the decision to commence the Chapter 11 Case, the development and implementation of the Plan, the decisions and actions taken prior									
10											
11											
12		to or during the Chapter 11 Case and any asserted claims based									
13		upon or related to prepetition obligations of the Debtor for the purpose of determining whether such claims belong to the Estate or third parties. In the event it is determined that any such claims									
14											
15		belong to third parties, then, subject to any applicable subject matter jurisdiction limitations, the Bankruptcy Court shall have									
16		exclusive jurisdiction with respect to any such litigation, subject									
17		to any determination by the Bankruptcy Court to abstain and consider whether such litigation should more appropriately									
18		proceed in another forum;									
19	(r)										
20		hereunder, including without limitation, disputes regarding the amounts of such reserves or the amount, allocation and timing of									
21		any releases of such reserved funds; and									
22	(s)	enter the Final Decree closing the Chapter 11 Case.									
23											
24		SECTION 20.									
25		MISCELLANEOUS PROVISIONS									
26	20.1 Te	ermination of All Employee, Retiree and Workers' Compensation									

20.1 *Termination of All Employee, Retiree and Workers' Compensation Benefits.* All existing employee benefits (including, without limitation, workers' compensation benefits, health care plans, disability plans, severance benefit plans, incentive plans, and life insurance plans) and retiree benefits (as such term is defined

DENTONS US LLP 601 SOUTH FIGUEROA STREET, SUITE 2500 LOS ANGELES, CALIFORNIA 90017-5704 213 623 9300

1

4

5

6 7

8

9 10

12

11

13 14

16 17

15

18

19 20

21

2223

24

2526

27

28

under § 1114(a)) not previously terminated by the Debtor, or assumed by the Debtor in the Schedule of Assumed Contracts, shall be terminated on or before the Effective Date.

20.2 Administrative Claims Bar Date. All Requests for Payment of an Administrative Claim must be filed with the Bankruptcy Court and served on the Debtor no later than the Administrative Claims Bar Date. Such Requests for Payment may include estimates of amounts through the Effective Date. The Administrative Claims Reserve shall be established on the Effective Date in an amount determined by the Bankruptcy Court in order to satisfy all Administrative Claims that have not been Allowed as of the Effective Date and all Allowed Administrative Claims that will be paid after the Effective Date. In the event that the Debtor, the Liquidating Trustee objects to an Administrative Claim, the Bankruptcy Court shall determine the Allowed amount of such Administrative Claim. Notwithstanding the foregoing: (a) no Request for Payment need be filed with respect to an undisputed postpetition obligation which was paid or is payable by the Debtor in the ordinary course of business; provided, however, that in no event shall a postpetition obligation that is contingent or disputed and subject to liquidation through pending or prospective litigation, including, but not limited to, alleged obligations arising from personal injury, property damage, products liability, consumer complaints, employment law (excluding claims arising under workers' compensation law), secondary payor liability, or any other disputed legal or equitable claim based on tort, statute, contract, equity, or common law, be considered to be an obligation which is payable in the ordinary course of business; and (b) no Request for Payment need be filed with respect to fees payable pursuant to 28 U.S.C. § 1930. All Administrative Claims that become Allowed after the Effective Date shall be paid solely from the Administrative Claims Reserve, and shall not constitute a claim against the Liquidating Trust, the Liquidating Trustee, or any of the Liquidating Trust Assets. No Holder of an Administrative Claim shall have recourse for any deficiency in the payment of its Administrative Claim against any of the Released Parties, the Post-Effective Date Debtor, the Post-Effective Date Board of Directors, the Liquidating Trustee, or the Liquidating Trust.

20.3 *Exemption from Transfer Taxes.* Pursuant to § 1146(c), the assignment or surrender of any lease or sublease, or the delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, this Plan, including any deeds, bills of sale or assignments executed in connection with any disposition of assets contemplated by this Plan, whether real or personal property, shall not be

subject to any stamp, real estate transfer, mortgage recording, sales, use or other similar tax.

- 20.4 *Amendments.* The Plan Proponents reserve the right, in accordance with the Bankruptcy Code and the Bankruptcy Rules, to amend or modify this Plan at any time prior to the entry of the Confirmation Order. After the entry of the Confirmation Order, the Plan Proponents may, upon order of the Bankruptcy Court, amend or modify this Plan, in accordance with § 1127(b), or remedy any defect or omission or reconcile any inconsistency in this Plan in such manner as may be necessary to carry out the purpose and intent of this Plan. A Holder of an Allowed Claim that is deemed to have accepted this Plan shall be deemed to have accepted this Plan as modified if the proposed modification does not materially and adversely change the treatment of the Claim of such Holder.
- 20.5 *Revocation or Withdrawal of Plan*. The Plan Proponents may withdraw or revoke this Plan at any time prior to the Effective Date. If the Plan Proponents revoke or withdraw this Plan prior to the Effective Date, or if the Effective Date does not occur, then this Plan shall be deemed null and void. In such event, nothing contained herein shall be deemed to constitute a waiver or release of any Claim by or against the Debtor or any other Person or to prejudice in any manner the rights of the Debtor or any other Person in any further proceedings involving the Debtor.
- 20.6 Severability. In the event that the Bankruptcy Court determines, prior to the Effective Date, that any provision of this Plan is invalid, void or unenforceable, the Bankruptcy Court shall, with the consent of the Plan Proponents, have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistently with the original purpose of the term or provision held to be invalid, void or unenforceable, and such term or provision shall then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of this Plan shall remain in full force and effect and shall in no way be affected, impaired or invalidated by such holding, alteration or interpretation. The Confirmation Order shall constitute a judicial determination and shall provide that each term and provision of this Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 20.7 Request for Expedited Determination of Taxes. The Plan Proponents or the Liquidating Trustee, as applicable, shall have the right to request an expedited determination under § 505(b) with respect to tax returns filed, or to be filed, for any

4 5 6

8

7

10111213

1415

161718

192021

232425

22

252627

28

and all taxable periods ending after the Petition Date through and including the date of Final Distribution under the Plan.

- 20.8 **Securities Exemption.** To the extent the Trust Beneficial Interests are deemed or asserted to constitute securities, the Trust Beneficial Interests and the issuance and distribution thereof shall be exempt from Section 5 of the Securities Act, if applicable, and from any state or federal securities laws requiring registration for offer or sale of a security or registration or licensing of an issuer of, underwriter of, or broker or dealer in, a security, and shall otherwise enjoy all exemptions available for distributions of securities under a plan of reorganization in accordance with all applicable law, including without limitation, § 1145.
- 20.9 *U.S. Trustee Quarterly Fees and Post-Confirmation Status Reports.* All fees payable under 28 U.S.C. § 1930(a)(6) shall be paid by the Debtor in the amounts and at the times such fees may become due up to and including the Effective Date. The Liquidating Trust shall pay all fees payable by the Debtor under 28 U.S.C. § 1930(a)(6) until the Chapter 11 Case is closed, dismissed or converted. After the Effective Date, the Liquidating Trust and Post-Effective Date Debtor shall file and serve the quarterly status reports required by § 1106(a)(7), Bankruptcy Rule 2015(a)(5), and 28 C.F.R. § 58.8.
- 20.10 *Courts of Competent Jurisdiction*. If the Bankruptcy Court abstains from exercising, or declines to exercise, jurisdiction or is otherwise without jurisdiction over any matter arising out of this Plan, such abstention, refusal or failure of jurisdiction shall have no effect upon and shall not control, prohibit or limit the exercise of jurisdiction by any other court having competent jurisdiction with respect to such matter.
- 20.11 *Governing Law.* Except to the extent that the Bankruptcy Code or Bankruptcy Rules are applicable, the rights, duties and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with, the laws of the State of California, without giving effect to the principles of conflict of laws thereof.
- 20.12 Continuing Effect of the Bankruptcy Court Orders and Settlement Stipulations. Unless otherwise set forth in the Plan or the Confirmation Order or otherwise ordered by the Bankruptcy Court, the orders of the Bankruptcy Court and any other settlement stipulations entered into by the Debtor (including without limitation, the DHCS 9019 Order, the DHCS Settlement Agreement, agreements to lift the automatic stay, resolve litigation claims and limit recoveries to available insurance proceeds) shall not be modified, limited or amended by the Plan and shall remain in full force and effect. To the extent of any direct conflict between the terms

of this Plan and any settlement agreements, the conflicting provisions of such settlement agreements shall govern with respect to the treatment of Allowed Claims as provided for therein.

- 20.13 *Substantial Consummation*. On the Effective Date, the Plan shall be deemed substantially consummated under §§ 1101 and 1127(b).
- 20.14 *Waiver of Fourteen-Day Stay*. The Plan Proponents request as part of the Confirmation Order a waiver from the Bankruptcy Court of the 14-day stay of Bankruptcy Rule 3020(e) and, to the extent applicable, a waiver of the 14-day stay of Bankruptcy Rule 6004(g).
- 20.15 *Reservation of Rights.* Neither the filing of the Plan nor any statement or provision contained in the Plan, nor the taking by any party in interest of any action with respect to the Plan, shall: (a) be or be deemed to be an admission against interest and (b) until the Effective Date, be or be deemed to be a waiver of any rights any party in interest may have (i) against any other party in interest, or (ii) in or to any of the assets of any other party in interest, and, until the Effective Date, all such rights are specifically reserved. In the event that the Plan is not confirmed or fails to become effective, neither the Plan nor any statement contained in the Plan may be used or relied upon in any manner in any suit, action, proceeding, or controversy within or without the Chapter 11 Case involving the Debtor, except with respect to Confirmation of the Plan.
- 20.16 *Successors and Assigns*. The rights, benefits, and obligations of any Entity named or referred to in the Plan shall be binding on, and shall inure to the benefit of, the heirs, executors, administrators, successors, and/or assigns of such Entity.
- 20.17 *Time*. In computing any period of time prescribed or allowed by this Plan, unless otherwise set forth herein or determined by the Bankruptcy Court, the provisions of Bankruptcy Rule 9006 shall apply. Any reference to "day" or "days" shall mean calendar days, unless otherwise specified herein.
- 20.18 *Business Day Transactions*. In the event that any payment or act under this Plan is required to be made or performed on a date that is not a Business Day, then the making of such payment or the performance of such act may be completed

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

on or as soon as reasonably practicable on the next succeeding Business Day, but shall be deemed to have been completed as of the initial due date.

- 20.19 *Headings*. Headings are used in this Plan for convenience and reference only and shall not constitute a part of this Plan for any other purpose.
- 20.20 *Exhibits*. Exhibit A, any schedules to this Plan, and the Plan Supplement are incorporated into and are a part of this Plan as if set forth in full herein.

20.21 *Notices.* Any notices to or requests by parties in interest under or in connection with this Plan shall be in writing and served either by (i) certified mail, return receipt requested, postage prepaid, (ii) hand delivery or (iii) reputable overnight delivery service, all charges prepaid, and shall be deemed to have been given when received by the following parties:

If to the Debtor:

Borrego Community Health Foundation 587 Palm Canyon Dr. Suite 208 Borrego Springs, California 92004 Attn: Doug Habig

with copies to:

Dentons US LLP
Attorneys for the Debtor and Debtor-In-Possession
601 South Figueroa Street, Suite 2500
Los Angeles, California 90017
1 213 623 9300
Attn: Samuel R. Maizel
Tania M. Moyron

If to the Committee:

Rebecca M. Wicks

Pachulski Stang Ziehl & Jones LLP 10100 Santa Monica Blvd., 13th Floor Los Angeles, California 90067 1 310 227 6910 Attn: Jeffrey N. Pomerantz Steven W. Golden

Case	ase 22-02384-LT11 Filed 02/05/23 Entered 02/05 of 160	5/23 13:00:5 9	Doc 1268	Pg. 126								
1	1											
2	If to the Liquidating	If to the Liquidating Trustee:										
3	3 Ankura Consulting	Ankura Consulting Group, LLC										
4	2000 K Street NW, 12 th Floor											
5	Washington, DC 20006 1 202 507 5499											
6	Attn: Isaac Lee											
7	If to the Co-Liquidating Trustee:											
8	FTI Consulting, Inc.											
9	8251 Greensboro Drive, Suite 400											
10	McLean, VA 22101 1 202 262 4778	McLean, VA 22101										
11		1 202 262 47/8 Attn: Narendra Ganti										
12												
13	20.22 Post-Effective Date Notices. Fo	ollowing the F	Effective Da	ate, except as								
14	otherwise provided herein, notices shall only be served on the Post-Effective Date Debtor, the Liquidating Trustee, the Co-Liquidating Trustee, the U.S. Trustee, and											
15	- II	those Persons who file with the Court and serve upon the Liquidating Trust a request,										
16	 	which includes such Person's name, contact person, address, telephone number, facsimile number, and email, that such Person receive notice of post-Effective Date										
17	· • • • • • • • • • • • • • • • • • •		-	-								
18	of the proceedings and other filings in the Chapost-Effective Date matters unless such Person											
19	11		•									
20		20.23 <i>Conflict of Terms.</i> In the event of a conflict between the terms of this										
21	Plan and the Disclosure Statement, the terms	Plan and the Disclosure Statement, the terms of this Plan shall control.										
22												
23	<i>[]</i> []	Order shall supersede all previous and contemporaneous negotiations, promises, covenants, agreements, understandings, and representations on such subjects, all of										
24		-		.								
25	25											
26	Dated: San Diego, California											
27	$\Lambda_{\alpha} \circ f D_{\alpha} \circ mb \circ m A = 2022$	Λ_{α} of D_{α} ombow Λ_{α} 2022										
28	$_{28}\parallel$											

Exhibit A² Liquidation Analysis

Background

Section 1129(a)(7) of the Bankruptcy Code, which is often referred to as the "best interests of creditors" test, requires that as a condition to confirmation of the Plan, each holder of a Claim in each Impaired Class must either (i) accept the Plan, or (ii) receive or retain under the Plan property of value that is not less than the amount the Holder of an Impaired Class Claim would receive or retain in a hypothetical liquidation under chapter 7 of the Bankruptcy Code (the "*Liquidation Analysis*").

The Debtor prepared this Liquidation Analysis to include: (i) estimated cash proceeds that a chapter 7 trustee would generate if the Debtor's Chapter 11 Case were converted to a chapter 7 case and the assets of the Debtor's Estate were liquidated; (ii) estimated distribution that each Holder of a Claim would receive from the liquidation proceeds; and (iii) compared each Holder's estimated distribution and recovery under a chapter 7 liquidation to the projected distribution and recovery under the Plan.

The Liquidation Analysis is based upon certain assumptions, and, as such, certain aspects may be different from those represented in the Plan. The Liquidation Analysis should be read in conjunction with the *Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation* [Docket No. 1091].

THE PLAN PROPONENTS MAKE NO REPRESENTATIONS OR WARRANTIES REGARDING THE ACCURACY OF THESE ESTIMATES AND ASSUMPTIONS CONTAINED HEREIN, OR A CHAPTER 7 TRUSTEE'S ABILITY TO ACHIEVE THE PROJECTED RESULTS. IN THE EVENT THAT THE CHAPTER 11 CASE IS CONVERTED TO A CHAPTER 7 LIQUIDATION, ACTUAL RESULTS MAY VARY MATERIALLY FROM THE ESTIMATES AND PROJECTIONS SET FORTH IN THIS LIQUIDATION ANALYSIS.

Presentation

The Liquidation Analysis has been prepared assuming the Debtor converted its

² Capitalized terms not otherwise defined herein shall have the meanings described to them in the *Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Borrego Community Health Foundation* [Docket No. 1091].

Chapter 11 Case to chapter 7 of the Bankruptcy Code on or about January 19, 2024 (the "*Liquidation Date*").³

On the Liquidation Date, it is assumed the Bankruptcy Court would appoint a chapter 7 trustee to liquidate the Debtor's remaining assets, wind down the Debtor and the Estate, and provide distributions to creditors considering the DHCS Settlement Agreement. The distributable value is assumed to be applied in the following order: (i) payment of the Allowed Secured Claims and the DHCS Sales Proceeds Recovery; (ii) chapter 7 administrative costs, including trustee fees, professional fees, and wind-down expenses; (iii) Allowed Chapter 11 Administrative Claims and Priority Claims; (iv) Allowed General Unsecured Claims; and (v) the Allowed DHCS Claim. The treatment of creditors in the context of a chapter 7 liquidation would be the same as they are under the Plan, but the distribution to those creditors would be significantly less.

The Liquidation Analysis assumes the Liquidating Trust under the Plan would employ existing Debtor and/or Committee professionals. Under the chapter 7 scenario, a chapter 7 trustee would be appointed to administer the Debtor's assets. A chapter 7 trustee would be completely unfamiliar with the complexities of this Chapter 11 Case. Following the appointment of a chapter 7 trustee, the chapter 7 trustee would presumably hire new professionals who are equally unfamiliar with the complexities of this Chapter 11 Case. For example, there is significant litigation pending where the Debtor is a plaintiff, and those cases could eventually represent a meaningful source of recoveries for the Debtor's Estate. The Debtor's professionals are intimately familiar with that litigation. The result of a chapter 7 trustee's employment of a substantial number of professionals unfamiliar with this complex Chapter 11 Case would be the incurrence of an extraordinary amount of additional professional fees. By contrast, the Debtor's and the Committee's professionals are skilled and already intimately familiar with the Chapter 11 Case.

The "Best Interest Test" requires a liquidation analysis that demonstrates that, if a claimant or interest holder is in an impaired class and that claimant or interest holder does not vote to accept the Plan, than that claimant or interest holder must receive or retain under the plan property of a value not less than the amount that such holder would receive or retain if the Debtor were forced to liquidate under chapter 7 of the Bankruptcy Code. It is not at all clear that this test applies in the bankruptcy of a nonprofit company. Unlike in the bankruptcy of a for-profit entity, the Bankruptcy Code and state law may preclude or restrict the forced sale of a nonprofit's assets. 11 U.S.C. §§ 1112(c), 303. By way of example, under § 1112(c), a nonprofit's creditors cannot force a nonprofit to convert its chapter 11 case to a chapter 7, nor under § 303 can they file an involuntary petition against a nonprofit. Similarly, state statutes impose stringent requirements on the transfer or sale of a nonprofit debtor's assets, see, e.g., CAL. CORP. CODE §§ 5913, 7913, 9633.5, and the involuntary dissolution of a nonprofit, see, e.g., CAL. CORP. CODE §§ 6510-6519, 8510-8519, 9680.

2

3

4

5

6

7

9

10

11

12

13

14

15

16

17

18

19

20

21

2.2

23

24

25

26

27

28

Additionally, a chapter 7 trustee(s) would be under a statutory duty to liquidate the Debtor's assets as expeditiously as possible. See 11 U.S.C. § 704(a)(1). However, the Debtor must remain extant, with operating management and a board of directors until DAP Health obtains its own Medicare and Medi-Cal provider agreements, among other things. Since the Bankruptcy Code does not automatically authorize the chapter 7 trustee to operate the Debtor's businesses following a conversion to chapter 7, the chapter 7 trustee would be required to seek authority to continue operating the Debtor after obtaining approval from the U.S. Trustee to make such request. See, e.g., 11 U.S.C. § 721 ("The court may authorize the trustee to operate the business of the debtor for a limited period, if such operation is in the best interest of the estate and consistent with the orderly liquidation of the estate."); Executive Office for the United States Trustee, U.S. Dept. of Justice *Handbook for Chapter 7 Trustees* (Oct. 1, 2012), at 4-31 ("The trustee must consult with the United States Trustee prior to seeking authority to operate the business[.]"). The chapter 7 trustee's discretion to move for an operating order under § 721, and the willingness of the U.S. Trustee and Court to grant such request, presents significant potential risks to creditor recoveries in chapter 7 for several important reasons. The Debtor must be in a position to monitor DAP Health's operations for as long as DAP Health does not have its own provider agreements. Failure to allow DAP Health continued access to Medi-Cal receivables would imperil the continued viability of the Debtor's former clinics and breach the agreement between the Debtor and DAP Health. Thus, the risk that the Debtor would not continue to operate in a hypothetical chapter 7 case represents a substantial risk to creditor recoveries in comparison with the Plan. Additionally, the chapter 7 trustee and/or their professionals would have to have health care industry operational experience and experience collecting health care receivables, in general, and FQHC operational experience specifically, to provide the necessary oversight and ensure sufficient liquidation of Estate assets. Retaining this expertise will result in greatly increasing the cost to the chapter 7 trustee and the Estate.

The advantages of finishing a liquidation in chapter 11 are not just "common knowledge" among professionals. Experts have also concluded that conversion to chapter 7 offers few advantages over liquidation in chapter 11: cases converted from chapter 11 to chapter 7 take significantly longer to resolve than a "pure" chapter 11 liquidation, and such cases require similar, if not greater, fees, and in the end provide creditors with statistically lower recovery rates than a comparable chapter 11 procedure. See Arturo Bris, Ivo Welch and Ning Zhu, The Costs of Bankruptcy: Chapter 7 Liquidation versus Chapter 11 Reorganization, J. OF FINANCE, vol. 61(3), June 2006, at 1253.

Conclusion

The Plan demonstrates the Allowed DHCS Claim will receive a substantial recovery under the Plan. However, under a chapter 7 liquidation, the remaining value available for the Allowed DHCS Claim would be materially less compared to the remaining value available under the proposed Plan. Through the significant cost savings of the confirmed Plan as compared to conversion to chapter 7, holders of allowed claims will receive more under the Plan than they would receive in a converted chapter 7 bankruptcy (and certainly at least as much as under the Plan).

Borrego Community Health Foundation Liquidation Analysis

Proposed Chapter 11 Plan vs. Chapter 7 Plan of Liquidation

\$ in USD 000's					
	Prop	oosed Plan	С	hapter 7	FN
Current Cash on Hand	\$	69,332	\$	69,332	[A]
Cash Burn to Plan Effective Date (ED)		(1,628)		(1,628)	[B]
Distributable Value at ED	\$	67,705	\$	67,705	
Secured Claims		(3)		(3)	
DHCS Sales Proceeds Recovery		(13,600)		(13,600)	[C]
US Trustee Fees		(250)		(250)	[O]
Administrative Claims - Excluding Professional Fees		(1,939)		-	
Administrative Professional Fee Claims		(2,015)		_	
Priority Claims		(1,435)		-	
Claims to be Paid at ED	\$	(19,242)	\$	(13,853)	
Excess Distributable Value at ED	\$	48,463	\$	53,851	
Litigation Settlement Proceeds		1,500		1,500	
Jared Settlement Proceeds		105		105	[D]
Estimated Post-ED Net Payments Due to Buyer		(1,425)		(1,425)	[E]
Distributable Value to be Realized Post-ED	\$	48,643	\$	54,031	
Post-ED US Trustee Fees		(750)		(750)	
Wind-Down Expenses		(1,523)		(1,523)	[F]
Liquidating Trust Fees		(3,110)		-	
Chapter 7 Trustee Fee (3% of disbursements)		-		(2,372)	
Chapter 7 Trustee Legal Counsel		-		(3,000)	[G]
Chapter 7 Trustee Financial Advisor		-		(1,100)	[G]
Cost of Transition from Chapter 11 Debtor's Professionals		-		(1,000)	[H]
Administrative Claims - Excluding Professional Fees		-		(1,939)	
Administrative Professional Fee Claims		-		(2,015)	
Priority Claims		-		(1,435)	
Administrative & Priority Claims Paid Post-ED	\$	(5,382)	\$	(15,133)	
Remaining Value Available for Unsecured Creditors	\$	43,260	\$	38,898	
Estimated General Unsecured Claims		11,668		11,668	[1]
Remaining Value for DHCS Claim	\$	31,592	\$	27,230	[C]
Negative Impact of Chapter 7 Scenario				(4,362)	

2

3

4

5

6

7

9

10

11

12

13

14

15

16

17

18

19

20

21

2.2

23

24

25

26

27

28

Footnotes to the Liquidation Analysis

[A]: This reflects the Debtor's Cash on hand as of November 3, 2023.

[B]: The projected uses of Cash include estimated payment of Administrative Claims and Chapter 11 administrative costs incurred before the Effective Date, net of estimated accounts receivable collections.

[C]: Pursuant to the DHCS Settlement Agreement, DHCS will be paid the first 40% of Net Cash Proceeds of the DAP Sale. The remaining cash available after disbursements to Allowed General Unsecured Creditors will be distributed to DHCS in accordance with the Plan and DHCS Settlement Agreement.

[D]: This reflects payments pursuant to the compromise among Debtor, George Jared, D.D.S., and George Jared D.D.S., Inc. [Docket Nos. 761], as approved by this Court's order [Docket No. 824].

[E]: This represents accounts receivable collected on behalf of DAP Health pursuant to the Asset Purchase Agreement after Closing [Docket No. 465], net of any Debtor-related accounts receivable collections.

[F]: This represents the costs of maintaining the Debtor's Estate until the regulatory change of ownership applications are approved, and costs of the subsequent wind-down of the Estate. Expenses include, but are not limited to, regulatory reporting and audits, tax related filings, business software licenses, document storage, and bank fees.

[G]: Chapter 7 trustee advisor fees are estimated to be materially greater than the Liquidating Trust advisor fees given the need for the chapter 7 trustee's professionals to familiarize themselves with the Chapter 11 Case and its complexities.

[H]: This represents an estimate of the costs to transition from the Debtor's current professionals in the Chapter 11 Case to incoming chapter 7 professionals including the costs of information and knowledge transfer by the Debtor's professionals related to outstanding causes of action.

[I]: General Unsecured Claims were based on the mid-point of an estimated allowed claims range of \$6.2 million to \$17.1 million. As set forth herein and in the Plan, General Unsecured Claimants are anticipated to receive 100% recovery on their Allowed Claims.

EXHIBIT B LIQUIDATING TRUST AGREEMENT

LIQUIDATING TRUST AGREEMENT

This Liquidating Trust Agreement ("<u>Liquidating Trust Agreement</u>" or "<u>Agreement</u>"), effective as of [●], 2024, by and between (i) Borrego Community Health Foundation (the "<u>Debtor</u>")¹ and (ii) [●] (the "<u>Liquidating Trustee</u>") and [●] (the "<u>Co-Liquidating Trustee</u>"), as cotrustees (together with any successor trustee, collectively the "<u>Liquidating Trustees</u>" or "<u>Trustees</u>") for the benefit of the Liquidating Trust Beneficiaries.

RECITALS

WHEREAS, on September 12, 2022 (the "<u>Petition Date</u>"), Borrego Community Health Foundation (the "<u>Debtor</u>") filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code with the United States Bankruptcy Court for the Southern District of California (the "<u>Bankruptcy Court</u>" or "<u>Court</u>"), which commenced *In re Borrego Community Health Foundation*, Case No. 22-02384 (the "<u>Bankruptcy Case</u>");

WHEREAS, on $[\bullet]$, the Debtor and the Committee jointly filed the *Joint Combined Disclosure Statement and Chapter 11 Plan of Liquidation of Debtor Borrego Community Health Foundation* [Docket No. \bullet] (as may be amended, supplemented, or otherwise modified from time to time, the "<u>Plan</u>"), which was confirmed by the Bankruptcy Court on $[\bullet]$, 2024, pursuant to the $[\bullet]$ [Docket No. \bullet] (the "<u>Confirmation Order</u>");

WHEREAS, the Plan and the Confirmation Order provide for the establishment of a Liquidating Trust (as defined in the Plan, the "<u>Liquidating Trust</u>") pursuant to this Liquidating Trust Agreement and the appointment of the Liquidating Trustees;

WHEREAS, The Liquidating Trust is intended to qualify as a "grantor trust" for U.S. federal income tax purposes pursuant to sections 671-677 of the Internal Revenue Code of 1986, as amended, with the Beneficiaries treated as the grantors and owners of the Liquidating Trust.

WHEREAS, this Agreement, including all exhibits hereto, is the "Liquidating Trust Agreement" described in the Plan and shall be executed on or before the Effective Date in order to facilitate implementation of the Plan.

DECLARATION OF TRUST

NOW, THEREFORE, in order to declare the terms and conditions hereof, and in consideration of the premises and mutual agreements herein contained, the confirmation of the Plan and of other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Debtor and the Liquidating Trustees have executed this Agreement for the benefit of the Liquidating Trust Beneficiaries entitled to share in the Liquidating Trust Assets as provided for in the Plan.

¹ As used herein, "Debtor" shall include, as applicable, the Post-Effective Date Debtor, as such term is defined in the Plan.

TO HAVE AND TO HOLD unto the Liquidating Trustees and their successors or assigns in trust, under and subject to the terms and conditions set forth herein and for the benefit of the Liquidating Trust Beneficiaries, and for the performance of and compliance with the terms hereof and of the Plan; <u>provided</u>, <u>however</u>, that upon termination of the Liquidating Trust in accordance with Article VIII hereof, this Liquidating Trust Agreement shall cease, terminate, and be of no further force and effect, unless otherwise specifically provided for herein.

IT IS FURTHER COVENANTED AND DECLARED that the Liquidating Trust Assets are to be strictly held and applied by the Liquidating Trustees subject to the specific terms set forth below.

ARTICLE I. DEFINITION AND TERMS

- 1.1 <u>Certain Definitions</u>. Unless the context shall otherwise require and except as contained in this Section 1.1 or as otherwise defined herein, the capitalized terms used herein shall have the respective meanings assigned thereto in the Plan. For all purposes of this Agreement, the following terms shall have the following meanings:
- (a) "Cause" means (i) a Person's willful failure to perform his material duties hereunder, which is not remedied within 30 days of notice; (ii) a Person's commission of an act of fraud, theft, or embezzlement during the performance of his or her duties hereunder; (iii) a Person's conviction of a felony (other than a felony that does not involve fraud, theft, embezzlement, or jail time) with all appeals having been exhausted or appeal periods lapsed; or (iv) a Person's gross negligence, bad faith, willful misconduct, or knowing violation of law in the performance of his or her duties hereunder.
- (b) "<u>Co-Liquidating Trustee</u>" means [●], as the initial "Co-Liquidating Trustee" hereunder and as defined in the Plan, and any successor Co-Liquidating Trustee that may be appointed pursuant to the terms of this Agreement.
- (c) "<u>GUC Claim Objection</u>" means any objection filed by the Debtor and/or the Committee prior to the Effective Date or the Liquidating Trustee and/or Co-Liquidating Trustee after the Effective Date to the allowance of any General Unsecured Claim, whether filed before or after the Effective Date.
- (d) "<u>Liquidating Trust</u>" means the "Borrego Liquidating Trust" established in accordance with Treasury Regulation Section 301.7701-4(d) pursuant to this Agreement.
 - (e) "Liquidating Trust Agreement" means this Agreement.
- (f) "<u>Liquidating Trust Beneficiaries</u>" means the holders of Class A Trust Beneficial Interests and Class B Trust Beneficial Interests.
- (g) "<u>Liquidating Trust Expense Reserve</u>" means \$[●] in Cash to be funded by the Liquidating Trustees from the Liquidating Trust Assets into a bank account of the Liquidating Trust on or shortly after the Effective Date for the purpose of paying Liquidating Trust Expenses in accordance herewith.

- (h) "<u>Liquidating Trust Expenses</u>" means the costs, expenses, liabilities and obligations incurred by the Liquidating Trust and/or the Liquidating Trustees in administering and conducting the affairs of the Liquidating Trust, and otherwise carrying out the terms of the Liquidating Trust and the Plan on behalf of the Liquidating Trust, including without any limitation, indemnification obligations, any taxes owed by the Liquidating Trust, and the fees and expenses of the Liquidating Trustees and professional persons retained by the Liquidating Trust or Liquidating Trustees in accordance with this Agreement.
- (i) "<u>Liquidating Trustee</u>" means [●], as the initial "Liquidating Trustee" hereunder and as defined in the Plan, and any successor Liquidating Trustee that may be appointed pursuant to the terms of this Agreement.
- (j) "Material Litigation" means (i) Husam E. Aldairi, et al. v. Borrego Community Health Foundation, Case No. 37-2021-00046200-CU-BC-CTL (Cal. Sup. Ct. San Diego); (ii) Borrego Community Health Foundation v. Inland Valley, LLC, et al., Case No. 3:21-cv-01417-AJB-AGS (S.D. Cal.); (iii) Borrego Community Health Foundation v. Karen Hebets, et al., Case No. 3:22-cv-01056-AJB-AGS (S.D. Cal.); (iv) Borrego Community Health Foundation v. Travelers Casualty and Surety Company of America, Case No. 3:22-CV-161-L-MDD (S.D. Cal.); and (v) any other Litigation other than the foregoing that is determined by the Liquidating Trustees to be Material Litigation.
- (k) "<u>Tax Code</u>" means the Internal Revenue Code of 1986, 26 U.S.C. § 1 *et seq.*, as amended from time to time, and corresponding provisions of any subsequent federal revenue act. A reference to a section of the Tax Code shall include a reference to any and all Treasury Regulations interpreting, limiting or expanding such section of the Tax Code.
- (l) "<u>Treasury Regulations</u>" means regulations promulgated under the Tax Code, including, but not limited to the Procedure and Administration Regulations, as such regulations may be amended from time to time.
- (m) "<u>Trust Beneficial Interests</u>" means, collectively, the Class A Trust Beneficial Interests and Class B Trust Beneficial Interests.
- 1.2 General Construction. As used in this Agreement, the masculine, feminine and neuter genders, and the plural and singular numbers shall be deemed to include the others in all cases where they would apply. "Includes" and "including" are not limiting and "or" is not exclusive. References to "Articles," "Sections" and other subdivisions, unless referring specifically to the Plan or provisions of the Bankruptcy Code, the Bankruptcy Rules, or other law, statute or regulation, refer to the corresponding Articles, Sections and other subdivisions of this Agreement, and the words "herein," "hereafter" and words of similar import refer to this Agreement as a whole and not to any particular Article, Section, or subdivision of this Agreement. Amounts expressed in dollars or following the symbol "\$" shall be deemed to be in United States dollars. References to agreements or instruments shall be deemed to refer to such agreements or instruments as the same may be amended, supplemented, or otherwise modified in accordance with the terms thereof.

1.3 <u>Incorporation of the Plan</u>. The Plan is hereby incorporated into this Agreement and made a part hereof by this reference.

ARTICLE II. ESTABLISHMENT OF THE LIQUIDATING TRUST

2.1 <u>Creation and Purpose of Trust.</u> The Debtor and the Liquidating Trustees, pursuant to the Plan and the Confirmation Order, and in accordance with the Bankruptcy Code and applicable tax statutes, rules, and regulations, to the extent incorporated in this Agreement, hereby constitute and create a trust (*i.e.*, the Liquidating Trust) for the purpose of winding down certain affairs of the Debtor and liquidating the Liquidating Trust Assets for the benefit of the Liquidating Trust Beneficiaries, with no objective to continue or engage in the conduct of a trade or business (except to the extent reasonably necessary to carry out, and consistent with, the liquidation purpose of the Liquidating Trust, provided that any such conduct will not affect the Liquidating Trust's tax status as a liquidation trust). The Liquidating Trust shall not have authority to engage in a trade or business, and no portions of the Liquidating Trust Assets shall be used in the conduct of a trade or business, except as is reasonably necessary for the prompt and orderly collection and reduction to Cash of the Liquidating Trust Assets (and any non-Cash proceeds thereof) and an orderly wind down of the Debtor's affairs, with the goal of maximizing such assets for the benefit of the Liquidating Trust Beneficiaries.

2.2 Objectives.

- (a) The Liquidating Trust is established for the purpose of (a) administering the Liquidating Trust Assets including the Distributions and payments contemplated under the Plan, (b) prosecuting and/or resolving all Disputed Claims, (c) investigating and pursuing any Causes of Action the Debtor holds or may hold against any Entity, and (d) making all Distributions to the Liquidating Trust Beneficiaries provided for under the Plan. The Liquidating Trust shall not continue or engage in any trade or business except to the extent reasonably necessary to monetize and distribute the Liquidating Trust Assets consistent with this Agreement and the Plan and act as president of the Post-Effective Date Debtor.
- (b) It is intended that the Liquidating Trust be classified for federal income tax purposes as a "liquidating trust" within the meaning of section 301.7701-4(d) of the Treasury Regulations. In furtherance of this objective, the Liquidating Trustees shall, in their business judgment, make continuing best efforts to (i) dispose of or monetize the Liquidating Trust Assets and resolve Claims, (ii) make timely distributions, and (iii) not unduly prolong the duration of the Liquidating Trust, in each case in accordance with this Agreement.
- 2.3 Nature and Purposes of the Liquidating Trust. The Liquidating Trust is organized and established as a trust for the purpose of monetizing the Liquidating Trust Assets and making Distributions to Liquidating Trust Beneficiaries in a manner consistent with "liquidating trust" status under Treasury Regulation Section 301.7701-4(d). The Liquidating Trust shall retain all rights to commence and pursue all Causes of Action of the Debtor. For the avoidance of doubt, the Liquidating Trust, pursuant to section 1123(b)(3)(B) of the Bankruptcy Code and applicable

state trust law, is appointed as the successor-in-interest to, and representative of, the Debtor and its Estate for the retention, enforcement, settlement, and adjustment of all Claims.

2.4 <u>Transfer of Assets and Rights to the Liquidating Trust.</u>

- (a) On the Effective Date, pursuant to the Plan, the Debtor shall irrevocably transfer, assign, and deliver, and shall be deemed to have transferred, assigned, and delivered, all Liquidating Trust Assets and related Privileges held by the Debtor to the Liquidating Trust free and clear of all Claims, Interests, Liens, and other encumbrances, and liabilities, except as provided in the Plan and this Agreement. To the extent certain assets comprising the Liquidating Trust Assets, because of their nature or because such assets will accrue or become transferable subsequent to the Effective Date, and cannot be transferred to, vested in, and assumed by the Liquidating Trust on such date, such assets shall be considered Operating Assets.
- (b) On or before the Effective Date, and continuing thereafter, the Debtor or Post-Effective Date Debtor, as applicable, shall provide (i) for the Liquidating Trustees' reasonable access to all records and information in the Debtor's and Post-Effective Date Debtor's possession, custody or control, (ii) that all Privileges related to the Liquidating Trust Assets shall transfer to and vest exclusively in the Liquidating Trust, and (iii) subject to Section 3.12, the Debtor and Post-Effective Date Debtor shall preserve all records and documents (including all electronic records or documents) until such time as the Liquidating Trustee directs the Post-Effective Date Debtor that such records are no longer required to be preserved. For the purposes of transfer of documents, the Liquidating Trust is an assignee and successor to the Debtor in respect of the Liquidating Trust Assets and shall be treated as such in any review of confidentiality restrictions in requested documents.
- (c) Until the Liquidating Trust terminates pursuant to the terms hereof, legal title to the Liquidating Trust Assets and all property contained therein shall be vested at all times in the Liquidating Trust as a separate legal entity, except where applicable law in any jurisdiction requires title to any part of the Liquidating Trust Assets to be vested in the Liquidating Trustees, in which case title shall be deemed to be vested in the Liquidating Trustees, solely in their capacity as Liquidating Trustees. For purposes of such jurisdictions, the term Liquidating Trust, as used herein, shall be read to mean the Liquidating Trustees.
- 2.5 <u>Acceptance</u>. The Liquidating Trustees accept the Liquidating Trust imposed by this Agreement and agrees to observe and perform that Liquidating Trust, on and subject to the terms and conditions set forth herein and in the Plan.
- 2.6 <u>Further Assurances</u>. The Debtor, Post-Effective Date Debtor, and any successors thereof will, upon reasonable request of the Liquidating Trustees, execute, acknowledge and deliver such further instruments and do such further acts as may be necessary or proper to transfer to the Liquidating Trustees any portion of the Liquidating Trust Assets intended to be conveyed hereby and in the Plan in the form and manner provided for hereby and in the Plan and to vest in the Liquidating Trustees the powers, instruments or funds in trust hereunder.

2.7 <u>Incidents of Ownership</u>. The Liquidating Trust Beneficiaries shall be the sole beneficiaries of the Liquidating Trust and the Liquidating Trustees shall retain only such incidents of ownership as are necessary to undertake the actions and transactions authorized herein.

ARTICLE III. THE LIQUIDATING TRUSTEES

3.1 Role.

- (a) <u>Liquidating Trustee</u>. In furtherance of and consistent with the purpose of the Liquidating Trust, the Plan, and this Agreement, the Liquidating Trustee, subject to the terms and conditions contained herein, in the Plan, and in the Confirmation Order, shall serve as the Liquidating Trustee with respect to the Class B Liquidating Trust Assets for the benefit of the holders of Class B Trust Beneficial Interests and maintain, manage, and take action on behalf of the Liquidating Trust.
- (b) <u>Co-Liquidating Trustee</u>. In furtherance of and consistent with the purpose of the Liquidating Trust, the Plan, and this Agreement, the Co-Liquidating Trustee, subject to the terms and conditions contained herein, in the Plan, and in the Confirmation Order, shall serve as the Co-Liquidating Trustee with respect to the Class A Liquidating Trust Assets for the benefit of holders of Class A Trust Beneficial Interests and maintain, manage, and take action on behalf of the Liquidating Trust.

3.2 <u>Liquidating Trustee Authority</u>.

- (a) In connection with the administration of the Liquidating Trust, in addition to any and all of the powers enumerated elsewhere herein, the Liquidating Trustee shall, in an expeditious but orderly manner, monetize the Class B Liquidating Trust Assets, make timely Distributions with respect to the holders of Class B Trust Beneficial Interests, and not unduly prolong the duration of the Liquidating Trust. The Liquidating Trustee shall have the power and authority and is authorized to perform any and all acts necessary and desirable to accomplish the purposes of this Agreement and the provisions of the Plan and the Confirmation Order relating to the Liquidating Trust, within the bounds of this Agreement, the Plan, the Confirmation Order, and applicable law. The Liquidating Trustee will monetize the Class B Liquidating Trust Assets with a view toward maximizing value in a reasonable time.
- (b) The Liquidating Trustee, subject to the limitations set forth in Sections 3.4 of this Agreement shall have the right to prosecute, defend, compromise, adjust, arbitrate, abandon, estimate, or otherwise deal with and settle any and all Claims and Causes of Action that are part of the Class B Liquidating Trust Assets as the Liquidating Trustee determines is in the best interests of the Liquidating Trust. To the extent that any action has been taken to prosecute, defend, compromise, adjust, arbitrate, abandon, or otherwise deal with and settle any such Claims and Causes of Action prior to the Effective Date, on the Effective Date the Liquidating Trustee shall be substituted for the Debtor in connection therewith in accordance with Rule 25 of the Federal Rules of Civil Procedure, made applicable by Rule 7025 of the Federal Rules of Bankruptcy Procedure, and the caption with respect to such pending action shall be changed to the following

"[Liquidating Trustee], not individually but solely as Liquidating Trustee for the Liquidating Trust, et al. v. [Defendant]".

- (c) Subject in all cases to any limitations contained herein, in the Confirmation Order, or in the Plan, the Liquidating Trustee shall have the power and authority to:
- (i) solely as required by Section 2.4(d), hold any and all rights of the Liquidating Trust and Class B Trust Beneficial Interests in or arising from the Class B Liquidating Trust Assets, including collecting and receiving any and all money and other property belonging to the Liquidating Trust and the right to vote or exercise any other right with respect to any claim or interest relating to the Class B Liquidating Trust Assets in any case under the Bankruptcy Code and receive any distribution with respect thereto;
- (ii) open accounts for the Liquidating Trust and make Distributions of Class B Liquidating Trust Assets in accordance herewith;
- (iii) as set forth in Section 3.11, exercise and perform the rights, powers, and duties held by the Debtor with respect to the Class B Liquidating Trust Assets, including the authority under section 1123(b)(3) of the Bankruptcy Code, and shall be deemed to be acting as a representative of the Debtor's Estate with respect to the Class B Liquidating Trust Assets, including with respect to the sale, transfer, or other disposition of the Class B Liquidating Trust Assets;
- (iv) settle or resolve any Claims other than the General Unsecured Claims (but including, for the avoidance of doubt, any Litigation with respect to any Holders of Disputed General Unsecured Claims);
- (v) exercise and perform the rights, powers, and duties arising from the Liquidating Trustee's role as president of the Post-Effective Date Debtor;
- (vi) take all steps necessary to wind-down and dissolve the Post-Effective Date Debtor;
- (vii) protect and enforce the rights to the Class B Liquidating Trust Assets by any method deemed appropriate, including by judicial proceedings or pursuant to any applicable bankruptcy, insolvency, moratorium or similar law and general principles of equity;
- (viii) obtain reasonable insurance coverage with respect to any liabilities and obligations of the Liquidating Trustees in the form of fiduciary liability insurance, a directors and officers policy, an errors and omissions policy, or otherwise. The cost of any such insurance shall be a Liquidating Trust Expense and paid by the Liquidating Trustees from the Liquidating Trust Assets;
- (ix) without further order of the Bankruptcy Court, but subject to the terms of this Agreement, employ various consultants, third-party service providers, and other professionals, including counsel, tax advisors, consultants, brokers, investment bankers, valuation counselors, and financial advisors, as the Liquidating Trustee reasonably deems necessary to aid it in fulfilling its obligations under this Agreement; such consultants, third-party service providers,

and other professionals shall be retained pursuant to whatever fee arrangement the Liquidating Trustee deems appropriate as long as such fees do not exceed what is allocated in the Wind-Down Budget (which may be amended or modified from time to time by the Liquidating Trustee), including contingency fee arrangements and any fees and expenses incurred by such professionals engaged by the Liquidating Trustee shall be Liquidating Trust Expenses and paid by the Liquidating Trustees from the Liquidating Trust Assets;

- (x) prepare and file as necessary (A) tax returns for the Liquidating Trust treating the Liquidating Trust as a grantor trust pursuant to Treasury Regulation section 1.671-4(a) which includes a valuation of assets transferred to the Liquidating Trust based upon the Liquidating Trustee's good faith determination of their fair market value, (B) an election pursuant to Treasury Regulation 1.468B-9(c) to treat the Disputed Claim Reserve as a disputed ownership fund, in which case the Liquidating Trustees will file federal income tax returns and pay taxes for the Disputed Claim Reserve as a separate taxable entity, or (C) any periodic or current reports that may be required under applicable law;
- (xi) prepare and send annually to the Liquidating Trust Beneficiaries, as required by and in accordance with applicable tax law, a separate statement stating a Liquidating Trust Beneficiary's interest in the Liquidating Trust and its share of the Liquidating Trust's income, gain, loss, deduction or credit, and to instruct all such Liquidating Trust Beneficiaries to report such items on their federal tax returns;
- (xii) to the extent applicable, assert, enforce, release, or waive any attorney-client communication, attorney work product or other Privilege or defense on behalf of the Liquidating Trust (including as to any Privilege that the Debtor held prior to the Effective Date), including to provide any information to insurance carriers that the Liquidating Trustees deem necessary to utilize applicable insurance coverage for any Claim or Claims;
- (xiii) request any appropriate tax determination with respect to the Liquidating Trust, including a determination pursuant to section 505 of the Bankruptcy Code;
- (xiv) take or refrain from taking any and all actions the Liquidating Trustee reasonably deems necessary for the continuation, protection, and maximization of the value of the Class B Liquidating Trust Assets consistent with purposes hereof;
- (xv) take all steps and execute all instruments and documents necessary to effectuate the purpose of the Liquidating Trust and the activities contemplated herein and in the Confirmation Order and the Plan, and take all actions necessary to comply with the Confirmation Order, the Plan, and this Agreement and the obligations thereunder and hereunder;
- (xvi) exercise such other powers and authority as may be vested in or assumed by the Liquidating Trustees by any Final Order;
- (xvii) evaluate and determine strategy with respect to the Class B Liquidating Trust Assets, and hold, pursue, prosecute, adjust, arbitrate, compromise, release, settle or abandon the Class B Liquidating Trust Assets on behalf of the Liquidating Trust;

- (xviii) with respect to the holders of Class B Trust Beneficial Interests, perform all duties and functions of the Distribution Agent as set forth in the Plan, including distributing Cash from the Disputed Claim Reserve, solely on account of Disputed Claims that were Disputed as of the Effective Date, but become Allowed, (the foregoing subparagraphs (i)-(xvii) being collectively, the "Authorized Acts"); and
 - (xix) exercise such other powers as may be vested in or assumed by the Liquidation Trust or the Liquidating Trustee pursuant to the Plan, Bankruptcy Court order, or as may be necessary, proper and appropriate to carry out the provisions of the Plan.
- (d) The Liquidating Trustee has the power and authority to act as trustee of the Liquidating Trust and perform the Authorized Acts through the date such Liquidating Trustee resigns, is removed, or is otherwise unable to serve for any reason.

3.3 <u>Co-Liquidating Trustee Authority.</u>

- (a) In connection with the administration of the Class A Liquidating Trust Assets, in addition to any and all of the powers enumerated elsewhere herein, the Co-Liquidating Trustee shall, in an expeditious but orderly manner, in consultation with the Liquidating Trustee, prosecute all GUC Claim Objections and, on behalf of the Liquidating Trustees, make Distributions to holders of Class A Trust Beneficial Interests. The Co-Liquidating Trustee shall have the power and authority and is authorized to perform any and all acts, after consultation with the Liquidating Trustee, necessary and desirable to accomplish the purposes of this Agreement and the provisions of the Plan and the Confirmation Order relating to the Class A Liquidating Trust Assets, within the bounds of this Agreement, the Plan, the Confirmation Order, and applicable law.
- (b) The Co-Liquidating Trustee, subject to the limitations set forth in Section 3.4 of this Agreement and after consultation with the Liquidating Trustee shall have the right to prosecute, commence, compromise, adjust, estimate, or otherwise deal with and settle any and all GUC Claim Objections.
- (c) Subject in all cases to any limitations contained herein, in the Confirmation Order, or in the Plan, and after consultation with the Liquidating Trustee, the Co-Liquidating Trustee shall have the power and authority to:
- (i) hold any and all rights in or arising from the Class A Liquidating Trust Assets, including, but not limited to, the right to collect any and all money and other property belonging to the Class A Liquidating Trust Assets (including any proceeds of the Class A Liquidating Trust Assets);
- (ii) perform the duties, exercise the powers, and asserts the rights of a trustee under sections 1123(b)(3)(B) of the Bankruptcy Code with respect to the Class A Liquidating Trust Assets, including the right to assert claims, defenses, offsets, and privileges;
- (iii) protect and enforce the rights of the Liquidating Trust with respect to any Class A Liquidating Trust Assets by any method deemed appropriate, including, without

limitation, by judicial proceeds, or pursuant to any applicable bankruptcy, insolvency, moratorium, or similar law and general principles of equity;

- (iv) subject to applicable law, seek the examination of any Entity or Person with respect to GUC Claim Objections;
- (v) make all payments relating to the holders of Class A Trust Beneficial Interests;
- (vi) assess, enforce, release, or waive any privilege or defense on behalf of the Liquidating Trust, the Class A Liquidating Trust Assets, or the holders of Class A Trust Beneficial Interests, if applicable;
- (vii) without further order of the Bankruptcy Court, but subject to the terms of this Agreement, employ various consultants, third-party service providers, and other professionals, including counsel, tax advisors, consultants, brokers, investment bankers, valuation counselors, and financial advisors, as the Co-Liquidating Trustee deems reasonably necessary, after consultation with the Liquidating Trustee, to aid him in fulfilling his obligations under this Agreement; such consultants, third-party service providers, and other professionals shall be retained pursuant to whatever fee arrangement the Co-Liquidating Trustee deems appropriate as long as such fees do not exceed what is allocated in the Wind-Down Budget, and any fees and expenses incurred by such professionals engaged by the Co-Liquidating Trustee shall be a Liquidating Trust Expense and paid by the Liquidating Trustees from the Liquidating Trust Assets;
- (viii) take all steps and execute all instruments and documents necessary to effectuate the purpose of the Liquidating Trust and the activities contemplated herein and in the Confirmation Order and the Plan, and take all actions necessary to comply with the Confirmation Order, the Plan, and this Agreement and the obligations thereunder and hereunder;
- (ix) with respect to the holders of Class A Trust Beneficial Interests, perform all duties and functions of the Distribution Agent as set forth in the Plan, including distributing Cash from the Disputed Claim Reserve, solely on account of Disputed Claims that were Disputed as of the Effective Date, but become Allowed; and
- (x) exercise such other powers and authority as may be vested in or assumed by the Co-Liquidating Trustee by any Final Order (the foregoing subparagraphs (i)-(x) being collectively, the "Co-Liquidating Trustee Authorized Acts").
- (d) The Co-Liquidating Trustee has the power and authority to act as a Co-Liquidating Trustee and perform the Co-Liquidating Trustee Authorized Acts through the date such Co-Liquidating Trustee resigns, is removed, or is otherwise unable to serve for any reason.
- (e) Any determinations by the Co-Liquidating Trustee with respect to the amount or timing of settlement or other disposition of any GUC Claim Objection settled in accordance with the terms of this Agreement shall be conclusive and binding on the holders of Class A Trust Beneficial Interests and all other parties of interest following the entry of an order of a court of competent jurisdiction approving such settlement or other disposition to the extent

required or obtained; *provided, however*, that any such determination that affects any Material Litigation must be made with the express consent of the Liquidating Trustee.

- 3.4 <u>Limitation of Authority</u>. Notwithstanding anything herein to the contrary, the Liquidating Trust and the Liquidating Trustees shall not (i) be authorized to engage in any trade or business, (ii) take any actions inconsistent with the management of the Liquidating Trust Assets as are required or contemplated by applicable law, the Confirmation Order, the Plan, and this Agreement, or (iii) take any action in contravention of the Confirmation Order, the Plan, or this Agreement.
- 3.5 <u>Binding Nature of Actions</u>. All actions taken and determinations made by the Liquidating Trustees in accordance with the provisions of this Agreement shall be final and binding upon any and all Liquidating Trust Beneficiaries.
- 3.6 <u>Term of Service</u>. The Liquidating Trustee shall serve as the Liquidating Trustee for the duration of the Liquidating Trust, subject to death, resignation or removal. The Co-Liquidating Trustee shall serve as the Co-Liquidating Trustee until the Class A Liquidating Trust Assets are fully administered in accordance with the Plan, upon which time the Co-Liquidating Trustee shall resign in accordance with Section 3.7(b), subject to earlier death, resignation or removal.

3.7 Resignation.

- (a) The Liquidating Trustees may resign as Liquidating Trustees of the Liquidating Trust by an instrument in writing delivered to the Bankruptcy Court at least thirty (30) days before the proposed effective date of resignation. The Liquidating Trustees shall continue to serve as Liquidating Trustees after delivery of the Liquidating Trustees' resignation until the proposed effective date of such resignation, unless otherwise ordered by the Bankruptcy Court, which earlier effective date shall be no earlier than the date of appointment of a successor Liquidating Trustees in accordance with Section 3.9 hereof becomes effective, unless (i) the Liquidating Trustee or Co-Liquidating Trustee, as applicable, is no longer able to serve in such role, (ii) continuing in such role imposes a material economic burden, or (iii) continuing in such role violates law, regulation, or ethical responsibilities. In the case of (i), (ii), or (iii) in the immediately preceding sentence, the resignation effective date shall be the proposed effective date provided that at least thirty (30) days' prior written notice is given to the Bankruptcy Court by the resigning party.
- (b) Once the Liquidating Trustees determine that the Class A Liquidating Trust Assets have been fully administered in accordance with the Plan, the Liquidating Trustees shall deliver a notice in writing to the Bankruptcy Court and the Liquidating Trustee effectuating the Co-Liquidating Trustee's resignation as the Co-Liquidating Trustee and no successor Co-Liquidating Trustee shall be appointed.

3.8 Removal.

(a) A Liquidating Trustee or Co-Liquidating Trustee may be removed for Cause immediately upon an order of the Bankruptcy Court. Upon the removal of a Liquidating Trustee or Co-Liquidating Trustee pursuant hereto, such Liquidating Trustee or Co-Trustee will

resign, or be deemed to have resigned, from any role or position he or she may have at the Post-Effective Date Debtor unless such Liquidating Trustee or Co-Liquidating Trustee agree otherwise.

(b) To the extent there is any dispute regarding the removal of a Liquidating Trustee or Co-Liquidating Trustee (including any dispute relating to any compensation or expense reimbursement due under this Agreement) the Bankruptcy Court shall retain jurisdiction to consider and adjudicate such dispute. Notwithstanding the foregoing, unless the Co-Liquidating Trustee is deemed to resign under Section 3.7(b), the Liquidating Trustees will continue to serve as the Liquidating Trustees after their removal until the earlier of (i) the time when a successor Liquidating Trustee will become effective in accordance with Section 3.9 of this Agreement or (ii) such date as the Bankruptcy Court otherwise orders.

3.9 Appointment of Successor.

- (a) Appointment of Successor. In the event of a vacancy by reason of the death or disability (in the case of a Liquidating Trustee or Co-Liquidating Trustee that is a natural person), dissolution (in the case of a Liquidating Trustee or Co-Liquidating Trustee that is not a natural person), or removal of a Liquidating Trustee or Co-Liquidating Trustee, or prospective vacancy by reason of resignation (unless the Co-Liquidating Trustee is deemed to resign under Section 3.6(b)), a successor Liquidating Trustee shall be selected by the Bankruptcy Court. If a final decree has been entered closing the Chapter 11 Case, the Liquidating Trustees may seek to reopen the Chapter 11 Case for the limited purpose of determining the successor Liquidating Trustee, and the costs for such motion and costs related to re-opening the Chapter 11 Case shall be paid by the Liquidating Trust. The successor Liquidating Trustee shall be appointed as soon as practicable, but in any event no later than sixty (60) days after the occurrence of the vacancy or, in the case of resignation, on the effective date of the resignation of the then acting Liquidating Trustee.
- (b) Vesting or Rights in Successor Liquidating Trustee. Every successor Liquidating Trustee appointed hereunder shall execute, acknowledge, and deliver to the Liquidating Trust and the exiting Liquidating Trustee, and file with the Bankruptcy Court, an instrument accepting such appointment subject to the terms and provisions hereof. The successor Liquidating Trustee, without any further act, deed, or conveyance shall become vested with all the rights, powers, trusts and duties of the exiting Liquidating Trustee, except that the successor Liquidating Trustee shall not be liable for the acts or omissions of the retiring Liquidating Trustee. In no event shall the retiring Liquidating Trustee be liable for the acts or omissions of the successor Liquidating Trustee.
- 3.10 Continuance of Liquidating Trust. The death, resignation, or removal of the Liquidating Trustees shall not operate to terminate the Liquidating Trust created by this Agreement or to revoke any existing agency (other than any agency of the Liquidating Trustees as the Liquidating Trustees) created pursuant to the terms of this Agreement or invalidate any action taken by the Liquidating Trustees. In the event of the resignation or removal of the Liquidating Trustees, the Liquidating Trustees shall promptly (i) execute and deliver, by the effective date of resignation or removal, such documents, instruments, records, and other writings as may be reasonably requested by their successor to effect termination of the exiting Liquidating Trustees' capacity under this Agreement and the conveyance of the Liquidating Trust Assets then held by

the exiting Liquidating Trustees to the successor Liquidating Trustees; (ii) deliver to the successor Liquidating Trustees all non-privileged documents, instruments, records, and other writings relating to the Liquidating Trust as may be in the possession or under the control of the exiting Liquidating Trustees, provided, the exiting Liquidating Trustees shall have the right to make and retain copies of such documents, instruments, records and other writings delivered to the successor Liquidating Trustees and the cost of making such copies shall be a Liquidating Trust Expense to be paid by the Liquidating Trust; and (iii) otherwise assist and cooperate in effecting the assumption of the exiting Liquidating Trustees' obligations and functions by his successor, provided the fees and expenses of such assistance and cooperation shall be paid to the exiting Liquidating Trustees by the Liquidating Trust. The exiting Liquidating Trustees shall irrevocably appoint the successor Liquidating Trustees as their attorney-in-fact and agent with full power of substitution for it and its name, place and stead to do any and all acts that such exiting Liquidating Trustees are obligated to perform under this Section 3.10.

- 3.11 <u>Liquidating Trustee as an "Estate Representative"</u>. The Liquidating Trustee will be a trustee of the Liquidating Trust Assets for purposes of 31 U.S.C. § 3713(b) and 26 U.S.C. § 6012(b)(3). The Liquidating Trustee shall be a representative of the Estate appointed pursuant to section 1123(b)(3)(B) of the Bankruptcy Code, with all rights and powers attendant thereto, in addition to all rights and powers granted in the Plan and in this Agreement. The Liquidating Trustee will be the successor-in-interest to the Debtor with respect to any action pertaining to the Liquidating Trust Assets, which was or could have been commenced by the Debtor prior to the Effective Date, except as otherwise provided in the Plan or Confirmation Order. All actions, claims, rights, or interest constituting Liquidating Trust Assets are preserved and retained and may be enforced by the Liquidating Trustee.
- 3.12 <u>Liquidating Trust Expense Reserve</u>. Each of the Liquidating Trustee and the Co-Liquidating Trustee may maintain the Liquidating Trust Expense Reserve in an amount and for the duration as determined by each of the Liquidating Trustee and the Co-Liquidating Trustee as applicable for its respective Liquidating Trust Expense Reserve.

3.13 Books and Records.

- (a) The Liquidating Trustee shall maintain in respect of the Liquidating Trust and the Liquidating Trust Beneficiaries books and records reflecting Liquidating Trust Assets in its possession and the income of the Liquidating Trust and payment of expenses, liabilities, and claims against or assumed by the Liquidating Trust in such detail and for such period of time as may be necessary to enable it to make full and proper accounting in respect thereof. Such books and records shall be maintained as reasonably necessary to facilitate compliance with the tax reporting requirements of the Liquidating Trust and the requirements of Article VI herein. Except as otherwise provided herein, nothing in this Agreement requires the Liquidating Trustees to file any accounting or seek approval of any court with respect to the administration of the Liquidating Trust, or as a condition for managing any payment or distribution out of the Liquidating Trust Assets.
- (b) The Liquidating Trustee may dispose some or all of the books and records maintained by the Liquidating Trustee at the later of (i) such time as the Liquidating Trustee determines that the continued possession or maintenance of such books and records is no longer

necessary for the benefit of the Liquidating Trust, or (ii) upon the termination and winding up of the Liquidating Trust under Article VIII of this Agreement. Notwithstanding the foregoing, the Liquidating Trustee shall cause the Post-Effective Date Debtor to retain such books and records, and for such periods, as are required to be retained pursuant to any applicable laws, rules, or regulations.

3.14 <u>Compensation and Reimbursement; Engagement of Professionals.</u>

(a) <u>Compensation and Expenses.</u>

- (i) <u>Compensation</u>. As compensation for any services rendered by the Liquidating Trustees in connection with this Agreement, the Liquidating Trustees shall receive compensation as set forth in the Plan Supplement.
- (ii) <u>Expense Reimbursements</u>. All reasonable out-of-pocket expenses of the Liquidating Trustees in the performance of their duties hereunder shall be reimbursed as Liquidating Trust Expenses paid by the Liquidating Trust.

(b) Professionals.

- (i) <u>Engagement of Professionals</u>. The Liquidating Trustees shall engage professionals from time to time in conjunction with the services provided hereunder.
- (ii) <u>Fees and Expenses of Professionals</u>. The Liquidating Trustees shall pay the reasonable fees and expenses of any retained professionals as Liquidating Trust Expenses.
- Reliance by Liquidating Trustees. Except as otherwise provided herein, the 3.15 Liquidating Trustees may rely, and shall be fully protected in acting or refraining from acting, on any resolution, statement, certificate, instrument, opinion, report, notice, request, consent, order or other instrument or document that the Liquidating Trustees have no reason to believe to be other than genuine and to have been signed or presented by the proper party or parties or, in the case of facsimiles, to have been sent by the proper party or parties, and the Liquidating Trustees may conclusively rely as to the truth of the statements and correctness of the opinions or direction expressed therein. The Liquidating Trustees may consult with counsel and other professionals, and any advice of such counsel or other professionals shall constitute full and complete authorization and protection in respect of any action taken or not taken by the Liquidating Trustees in accordance therewith. The Liquidating Trustees shall have the right at any time to seek instructions from the Bankruptcy Court, or any other court of competent jurisdiction concerning the Liquidating Trust Assets, this Agreement, the Plan, or any other document executed in connection therewith, and any such instructions given shall be full and complete authorization in respect of any action taken or not taken by the Liquidating Trustees in accordance therewith. The Liquidating Trust shall have the right to seek Orders from the Bankruptcy Court as set forth in Article IX of the Plan.
- 3.16 <u>Commingling of Liquidating Trust Assets</u>. The Liquidating Trustees shall not commingle any of the Liquidating Trust Assets with their own property or the property of any other Person.

ARTICLE IV. TRUST INTERESTS

4.1 <u>Liquidating Trust Interests.</u>

- (a) <u>Class A Trust Beneficial Interests</u>. On the date hereof, or on the date such Claim becomes Allowed under the Plan, the Liquidating Trust shall issue Class A Trust Beneficial Interests to Holders of Allowed General Unsecured Claims. The Co-Liquidating Trustee shall allocate to each Holder of an Allowed General Unsecured Claim a Class A Trust Beneficial Interest equal to the ratio that the amount of each Holder's Allowed General Unsecured Claim bears to the total amount of the Allowed General Unsecured Claims. The Class A Trust Beneficial Interests shall be entitled to Distributions from the Class A Liquidating Trust Assets in accordance with the terms of the Plan and this Agreement.
- (b) <u>Class B Trust Beneficial Interests</u>. On the date hereof, the Liquidating Trust shall issue the Class B Trust Beneficial Interests to the Holder of the Allowed DHCS Balance Claim. The Class B Trust Beneficial Interests shall be entitled to Distributions from the Class B Liquidating Trust Assets in accordance with the terms of the Plan and this Agreement.
- 4.2 <u>Interests Beneficial Only.</u> The ownership of the beneficial interests in the Liquidating Trust shall not entitle the Liquidating Trust Beneficiaries to any title in or to the Liquidating Trust Assets (which title shall be vested in the Liquidating Trust) or to any right to call for a partition or division of the Liquidating Trust Assets or to require an accounting. No Liquidating Trust Beneficiary shall have any governance right or other right to direct Liquidating Trust activities.
- 4.3 <u>Transferability of Trust Beneficial Interests.</u> Trust Beneficial Interests are not transferable.
- 4.4 <u>Effect of Death, Incapacity, or Bankruptcy</u>. The death, incapacity, or bankruptcy of any Liquidating Trust Beneficiary during the term of the Liquidating Trust shall not (i) entitle the representatives or creditors of the deceased Liquidating Trust Beneficiary to any additional rights under this Agreement, or (ii) otherwise affect the rights and obligations of any of other Liquidating Trust Beneficiary under this Agreement.
- 4.5 <u>Change of Address</u>. Any Liquidating Trust Beneficiary may, after the Effective Date, select an alternative distribution address by providing notice to the Liquidating Trustees identifying such alternative distribution address. Such notification shall be effective only upon receipt by the Liquidating Trustees. Absent actual receipt of such notice by the Liquidating Trustees, the Liquidating Trustees shall not recognize any such change of distribution address.
- 4.6 <u>Standing</u>. No Liquidating Trust Beneficiary shall have standing to direct the Liquidating Trustees to do or not to do any act or to institute any action or proceeding at law or in

equity against any party upon or with respect to the Liquidating Trust Assets. No Liquidating Trust Beneficiary shall have any direct interest in or to any of the Liquidating Trust Assets.

4.7 <u>Limitations on Rights of Liquidating Trust Beneficiaries.</u>

- (a) The Liquidating Trust Beneficiaries shall have no rights other than those set forth in this Agreement, the Confirmation Order, or the Plan (including any Plan Supplement documents incorporated therein).
- (b) In any action taken by a Liquidating Trust Beneficiary against the Liquidating Trust or a current or former Liquidating Trustee or Co-Liquidating Trustee, the prevailing party will be entitled to reimbursement of attorneys' fees and other costs; <u>provided</u>, <u>however</u>, that any fees and costs shall be borne by the Liquidating Trust on behalf of any such Liquidating Trustee or Co-Liquidating Trustee, as set forth herein.
- (c) Any action brought by a Liquidating Trust Beneficiary must be brought in the United States Bankruptcy Court for the Southern District of California. Liquidating Trust Beneficiaries are deemed to have waived any right to a trial by jury.
- (d) The rights of Liquidating Trust Beneficiaries to bring any action against the Liquidating Trust, a current or former Liquidating Trustee or Co-Liquidating Trustee shall not survive the final Distribution by the Liquidating Trust.

ARTICLE V. DISTRIBUTIONS

5.1 Distributions.

All Distributions to be made by the Liquidating Trustees to any Person or Entity shall be made only in accordance with the Plan, the Confirmation Order, and this Liquidating Trust Agreement, and from the Cash or Cash proceeds of Liquidating Trust Assets, and only to the extent that the Liquidating Trust has sufficient Cash to make such payments in accordance with and to the extent provided for in the Plan, the Confirmation Order, and this Liquidating Trust Agreement. The Liquidating Trustees shall distribute to the Liquidating Trust Beneficiaries the net income of the Liquidating Trust plus all net proceeds from the monetization of Liquidating Trust Assets at least annually; *provided* that (i) the Liquidating Trust may retain an amount of net proceeds or net income reasonably necessary to maintain the value of its assets, pay Allowed Claims, and fund the Disputed Claim Reserve and such other reserves provided for in, or otherwise not inconsistent with, the terms of the Plan, the Confirmation Order, or this Liquidating Trust Agreement, including a reserve for the payment of Liquidating Trust Expenses, and (ii) such Distribution is not otherwise inconsistent with the terms of the Plan and Confirmation Order. Any distribution made by the Liquidating Trustees in good faith shall be binding and conclusive on all interested parties, absent manifest error.

5.2 <u>Delivery of Distributions</u>. All distributions under this Agreement to any Liquidating Trust Beneficiary shall be made, as applicable, at the address of such Liquidating Trust

Beneficiary (a) as set forth on the proof of claim (where applicable) or Schedules filed with the Bankruptcy Court or (b) on the books and records of the Debtor or their agents, as applicable, unless the Liquidating Trustees have been notified in writing of a change of address pursuant to Section 4.5 hereof.

- 5.3 <u>Disputed Claim Reserves</u>. There will be no Distributions under this Agreement or the Plan on account of Disputed Claims pending Allowance. The Liquidating Trustees will maintain a Disputed Claim Reserve, as set forth in the Plan and will make Distributions from the Disputed Claim Reserve as set forth in the Plan.
- 5.4 <u>Undeliverable Distributions and Unclaimed Property</u>. All undeliverable distributions and unclaimed property shall be treated in the manner set forth in the Plan.
- 5.5 <u>De Minimis Distributions</u>. Distributions with a value of less than \$100 will be treated in accordance with the Plan.
- Trust shall pay as a Liquidating Trust Expense, all fees incurred under 28 U.S.C. § 1930(a)(6) by reason of the Liquidating Trust's disbursements until the Chapter 11 Case is closed. After the Effective Date, the Liquidating Trust shall prepare and serve on the Office of the United States Trustee such quarterly disbursement reports for the Liquidating Trust as required by the Office of the United States Trustee Office for as long as the Chapter 11 Case remains open, with the assistance of the Co-Liquidating Trustee, who shall timely provide the Liquidating Trustee with all information concerning the Class A Liquidating Trust Assets and professionals retained by the Co-Liquidating Trustee necessary for the Liquidating Trustee to complete and timely file and serve such disbursement reports.

ARTICLE VI. TAX MATTERS

6.1 Income Tax Status.

- (a) It is intended that the Liquidating Trust be classified for federal income tax purposes as a liquidating trust pursuant to Treasury Regulations Section 301.7701-4(d) and IRS Revenue Procedure 94-45, 1994-2 C.B. 684 and as a "grantor trust" subject to the provisions of Subtitle A, Chapter 1, Subchapter J, Part I, Subpart E of the Tax Code that is owned by the Liquidating Trust Beneficiaries. Accordingly, the parties hereto intend that the Liquidating Trust Beneficiaries be treated as if they had received a distribution of the applicable assets transferred to the Liquidating Trust and then contributed such assets to the Liquidating Trust. As such, notwithstanding anything set forth herein, the transfer of assets to the Liquidating Trust shall be treated for all purposes of the Tax Code as a transfer from the Estate to creditors to the extent the creditors are Liquidating Trust Beneficiaries followed by a deemed transfer by the Liquidating Trust Beneficiaries will be treated as grantors and deemed owners of the Liquidating Trust.
- (b) All parties, including the Debtor, the Liquidating Trustees, and all Liquidating Trust Beneficiaries, must value all assets transferred to the Liquidating Trust consistently, and those valuations must be used for all federal, state, and local income tax purposes.

The Liquidating Trustee must file returns for the Liquidating Trust as a grantor trust pursuant to Treasury Regulation Section 1.671-4(a). The assets shall be valued based upon the Liquidating Trustee's good faith determination of their fair market value.

- (c) Anything set forth herein to the contrary notwithstanding, the Liquidating Trustee shall not receive or retain Cash or Cash equivalents in excess of a reasonable amount to meet claims and contingent liabilities or to maintain the value of Class B Liquidating Trust Assets during liquidation. Further, anything set forth herein to the contrary notwithstanding, the Co-Liquidating Trustee shall not receive or retain Cash or Cash equivalents in excess of a reasonable amount to meet claims and contingent liabilities or to maintain the value of Class A Liquidating Trust Assets during liquidation. All income of the Liquidating Trust must be subject to tax on a current basis, including income retained in a disputed claims reserve. The taxable income of the Liquidating Trust will be allocated to and among Liquidating Trust Beneficiaries who are grantors of the Liquidating Trust as required by virtue of their being grantors and deemed owners of the Liquidating Trust, and they shall each be responsible to report and pay taxes due on their appropriate share of Liquidating Trust income.
- Treasury Regulations Section 301.7701-4(d) and IRS Revenue Procedure 94-45, 1994-2 C.B. 684, and in the event of any inconsistency between any term or provision herein, in the Plan, or in the Confirmation Order necessary for the Liquidating Trust to be deemed at all times a liquidating trust pursuant to Treasury Regulations Section 301.7701-4(d) and IRS Revenue Procedure 94-45, 1994-2 C.B. 684 and any other term or provision herein, in the Plan, or in the Confirmation Order, the term(s) and provision(s) necessary for the Liquidating Trust to be deemed a liquidating trust pursuant to Treasury Regulations Section 301.7701-4(d) and IRS Revenue Procedure 94-45, 1994-2 C.B. 684 shall govern. Similarly, anything to the contrary set forth herein, in the Plan, or in the Confirmation Order notwithstanding, to the extent any term or provision herein, in the Plan, in the Confirmation Order would result in the Liquidating Trust not being classified as a liquidating trust at all times pursuant to Treasury Regulations Section 301.7701-4(d) and IRS Revenue Procedure 94-45, 1994-2 C.B. 684, such term or provision shall be ineffective and reformed to the extent necessary for the Liquidating Trust to be classified at all times as a liquidating trust pursuant to Treasury Regulations Section 301.7701-4(d) and IRS Revenue Procedure 94-45, 1994-2 C.B. 684.
- 6.2 <u>Tax Returns</u>. The Liquidating Trustee shall prepare and provide to, or file with, the appropriate parties such notices, tax returns, information returns, and other filings as may be required by the Tax Code and may be required by applicable law of other jurisdictions. The Liquidating Trustee shall be responsible for filing all required federal, state, and local tax returns and information returns of the Liquidating Trust and the Post-Effective Date Debtor, which shall be prepared with the assistance of the Co-Liquidating Trustee who shall provide the Liquidating Trustee will any information necessary to complete such returns. The Liquidating Trustees shall, when specifically requested by a Liquidating Trust Beneficiary in writing, make such tax information available to the Liquidating Trust Beneficiary for inspection and copying at the Liquidating Trust Beneficiary's expense, as is necessary for the preparation by such Liquidating Trust Beneficiary of its income tax return.

ARTICLE VII. STANDARD OF CARE AND INDEMNIFICATION

- Trustee Standard of Care; Exculpation; Limitation on Liability; Indemnification. The Liquidating Trustees shall not be liable for any act the Liquidating Trustees may do or omit to do as Liquidating Trustees under the Plan and this Liquidating Trust Agreement while acting in good faith and in the exercise of the Liquidating Trustees' business judgment; nor will the Liquidating Trustees be liable in any event except for an act or omission that is determined by Final Order of a court of competent jurisdiction to have constituted gross negligence, willful misconduct, or fraud. The foregoing limitation on liability also will apply to any Person or Entity employed by the Liquidating Trustees and acting on behalf of the Liquidating Trustees in the fulfillment of the Liquidating Trustees' duties hereunder or under the Plan. In addition, the Liquidating Trustees and any Person or Entity employed by the Liquidating Trustees and acting on behalf of the Liquidating Trustees shall be entitled to indemnification and defense out of the assets of the Liquidating Trust against any losses, liabilities, expenses (including attorneys' fees and disbursements), damages, taxes, suits, or claims that they may incur or sustain by reason of being or having been employed by the Liquidating Trust or the Liquidating Trustees or for performing any functions incidental to such service; provided, however, that the foregoing shall not relieve the Liquidating Trustees or any Person employed by the Liquidating Trustees from liability, nor entitle any of the foregoing entities to indemnification pursuant to the Plan or this Liquidating Trust Agreement, for an act or omission that is determined by Final Order to have constituted gross negligence, willful misconduct, or fraud. Persons or Entities dealing with the Liquidating Trustees or any of the foregoing parties in their respective capacities as such, or seeking to assert claims against the Liquidating Trustees or any of the foregoing parties in their capacities as such, shall have recourse only to the Trust Assets (excluding any fund or reserve to pay Liquidating Trust Expenses) to satisfy any liability incurred by the Liquidating Trustees or any of the foregoing parties in their respective capacities as such to such Persons or Entities in carrying out the terms of the Plan and this Liquidating Trust Agreement. Satisfaction of any obligation of the Liquidating Trust arising pursuant to the terms of this section shall be payable only from the Trust Assets, and such right to payment shall be prior and superior to any other rights to receive on behalf of any Liquidating Trust Beneficiary any distribution of Trust Assets or proceeds thereof. The Liquidating Trust shall have the right to control the defense and settlement of claims as to which it is obligated to indemnify.
- 7.2 <u>Bond</u>. The Liquidating Trustees shall not be obligated to give any bond or surety for the performance of any of the Liquidating Trustees' duties unless otherwise ordered by the Bankruptcy Court. If so ordered, all costs and expenses of procuring a bond or surety shall be deemed a Liquidating Trust Expense and payable from the Liquidating Trust Assets.
- 7.3 No Personal Liability. Except as otherwise provided herein, the Liquidating Trustees shall not be subject to any personal liability whatsoever, whether in tort, contract, or otherwise, to any Person in connection with the affairs of the Liquidating Trust to the fullest extent provided under applicable law, and all Persons asserting claims against the Liquidating Trustees or otherwise asserting claims of any nature in connection with the affairs of the Liquidating Trust, shall look solely to the Liquidating Trust Assets for satisfaction of any such claims.

7.4 <u>No Liability for Acts of Predecessor Liquidating Trustees.</u> No successor Liquidating Trustees shall be in any way liable for the acts or omissions of any predecessor Liquidating Trustee unless a successor Liquidating Trustee expressly in writing assumes such responsibility.

ARTICLE VIII. TERMINATION

- 8.1 Duration. The Liquidating Trustees and the Liquidating Trust shall be discharged or dissolved, as the case may be, at such time as: (a) the Liquidating Trustee determines that the pursuit of Causes of Action are not likely to yield sufficient additional proceeds to justify further pursuit of such Causes of Action; (b) the Liquidating Trustee determines that the pursuit of monetization of other Class B Liquidating Trust Assets is not likely to yield sufficient additional proceeds to justify further pursuit of such monetization of other Class B Liquidating Trust Assets; (c) the Co-Liquidating Trustee determines that the pursuit of monetization of other Class A Liquidating Trust Assets is not likely to yield sufficient additional proceeds to justify further pursuit of such monetization of other Class A Liquidating Trust Assets; (d) all objections to Disputed Claims are fully resolved; (e) the Post-Effective Date Debtor is dissolved; (f) all Liquidating Trust Expenses are paid and any claims for which the Liquidating Trustee or the Co-Liquidating Trustee would need to seek indemnification are resolved; and (g) all Distributions required to be made by the Liquidating Trustees to the Liquidating Trust Beneficiaries under the Plan have been made, but in no event shall the Liquidating Trust be dissolved later than three years from the Effective Date unless the Bankruptcy Court, upon motion made within the six-month period before such third anniversary (and, in the event of further extension, by order of the Bankruptcy Court, upon motion made at least six months before the end of the preceding extension), determines that a fixed period extension (not to exceed two years, together with any prior extensions) is necessary to facilitate or complete the recovery on, and liquidation of, the Liquidating Trust Assets.
- 8.2 <u>Termination of Duties</u>. Except as otherwise specifically provided herein, upon the termination of the Liquidating Trust, the Liquidating Trustees shall have no further duties or obligations hereunder.

ARTICLE IX. AMENDMENTS AND WAIVER

The Liquidating Trustees may amend this Agreement to correct or clarify any non-material provisions. This Agreement may not otherwise be amended, supplemented, otherwise modified, or waived in any respect except by an instrument in writing signed by the Liquidating Trustees and the approval of the Bankruptcy Court, after notice and a hearing.

ARTICLE X. MISCELLANEOUS

10.1 <u>Trust Irrevocable</u>. Except as set forth in this Agreement, establishment of the Liquidating Trust by this Agreement shall be irrevocable and shall not be subject to revocation, cancellation or rescission by the Liquidating Trust Beneficiaries.

- 10.2 <u>Bankruptcy of Liquidating Trust Beneficiaries</u>. The dissolution, termination, bankruptcy, insolvency or other similar incapacity of any Liquidating Trust Beneficiary shall not permit any creditor, trustee, or any other Liquidating Trust Beneficiary to obtain possession of, or exercise legal or equitable remedies with respect to, the Liquidating Trust Assets.
- 10.3 <u>Liquidating Trust Beneficiaries have No Legal Title to Liquidating Trust Assets.</u>
 No Liquidating Trust Beneficiary shall have legal title to any part of the Liquidating Trust Assets.
- 10.4 <u>Agreement for Benefit of Parties Only</u>. Nothing herein, whether expressed or implied, shall be construed to give any Person other than the Liquidating Trustees and the Liquidating Trust Beneficiaries any legal or equitable right, remedy or claim under or in respect of this Agreement. The Liquidating Trust Assets shall be held for the sole and exclusive benefit of the Liquidating Trust Beneficiaries.
- 10.5 <u>Notices</u>. All notices, directions, instructions, confirmations, consents and requests required or permitted by the terms hereof shall, unless otherwise specifically provided herein, be in writing and shall be sent by first class mail, facsimile, overnight mail, electronic mail or in the case of mailing to a non-United States address, air mail, postage prepaid, addressed to:
 - (a) If to the Liquidating Trustees:

 $[\bullet]$

With a copy to:

[•]

Notice mailed shall be effective on the date mailed or sent. Any Person may change the address at which it is to receive notices under this Agreement by furnishing written notice pursuant to the provisions of this Section 10.5 to the entity to be charged with knowledge of such change.

- 10.6 <u>Severability</u>. Any provision hereof which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provisions in another jurisdiction.
- 10.7 <u>Counterparts</u>. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute but one and the same instrument.
- 10.8 <u>Binding Effect, etc.</u> All covenants and agreements contained herein shall be binding upon, and inure to the benefit of, the Liquidating Trust, the Liquidating Trustees, and the Liquidating Trust Beneficiaries, and their respective successors and assigns. Any notice, direction, consent, waiver or other instrument or action by any Liquidating Trust Beneficiary shall bind its successors and assigns.

- 10.9 <u>Headings</u>; <u>References</u>. The headings of the various Sections herein are for convenience of reference only and shall not define or limit any of the terms or provisions hereof.
- 10.10 Governing Law. This Agreement shall in all respects be governed by, and construed in accordance with the laws of the State of California, including all matters of constructions, validity and performance.
- 10.11 <u>Consent to Jurisdiction</u>. Each of the parties hereto consents and submits to the exclusive jurisdiction of the Bankruptcy Court for any action or proceeding instituted for the enforcement and construction of any right, remedy, obligation, or liability arising under or by reason of this Agreement, the Plan or any act or omission of the Liquidating Trustees (acting in their capacity as the Liquidating Trustees or in any other capacity contemplated by this Agreement or the Plan); *provided, however*, that if the Bankruptcy Court either declines to exercise jurisdiction over such action or cannot exercise jurisdiction over such action, such action may be brought in the state or federal courts located in the Southern District of California.
- 10.12 <u>Transferee Liabilities</u>. The Liquidating Trust shall have no liability for, and the Liquidating Trust Assets shall not be subject to, any claim arising by, through or under the Debtor except as expressly set forth in the Plan or in this Agreement. In no event shall the Liquidating Trustees or the Liquidating Trust Beneficiaries have any personal liability for such claims. If any liability shall be asserted against the Liquidating Trust or the Liquidating Trustees as the transferee of the Liquidating Trust Assets on account of any claimed liability of, through or under the Debtor or Post-Effective Date Debtor, the Liquidating Trustees may use such part of the Liquidating Trust Assets as may be necessary to contest any such claimed liability and to pay, compromise, settle or discharge same on terms reasonably satisfactory to the Liquidating Trustees as a Liquidating Trust Expense.

[Remainder of Page Intentionally Blank]

IN WITNESS HEREOF, the parties hereto have caused this Liquidating Trust Agreement to be duly executed by their respective officers thereunto duly authorized on the day and year first written above.

Borrego Community Health Foundation
By:
[•] [•]
Liquidating Trustees
By:
[•], not individually but solely in his capacity as the Liquidating Trustee
By:
[•], not individually but solely in his
capacity as the Co-Liquidating Trustee

EXHIBIT C

NOTICE OF (I) CONFIRMATION AND EFFECTIVE DATE OF FIRST
AMENDED JOINT COMBINED DISCLOSURE STATEMENT AND CHAPTER
11 PLAN OF LIQUIDATION OF BORREGO COMMUNITY HEALTH
FOUNDATION AND (II) DEADLINE UNDER THE PLAN AND
CONFIRMATION ORDER TO FILE ADMINISTRATIVE CLAIMS,
PROFESSIONAL FEE CLAIMS AND REJECTION CLAIMS

DENTONS US LLP 601 South Figueroa Street, Suite 2500 Los Angeles, California 90017-5704 213 623 9300

be amended, modified, or supplemented, the "<u>Plan</u>")¹ in the chapter 11 case of the above-captioned debtor and debtor in possession (the "<u>Debtor</u>").

- 2. **Effective Date of the Plan.** The Effective Date of the Plan was _____, 2024.
- 3. Administrative Claims Bar Date. As provided for in Section 3 of the Plan (defined term Administrative Claims Bar Date) and in the Confirmation Order, all requests for payment of an Administrative Claim incurred after September 12, 2022, must be filed with the Bankruptcy Court and served on counsel to the Debtor, counsel to the Liquidating Trustee, counsel to the Co-Liquidating Trustee, and counsel to the U.S. Trustee no later than _______, 2024 (the date that is 45 days after the Effective Date). Nothing herein shall be deemed to extend existing deadlines established for filing certain Administrative Claims, including pursuant to the Bar Date Order.
- 4. Rejection Bar Date. As provided for in Section 3 of the Plan (defined term Rejection Bar Date) and in the Confirmation Order, any Proofs of Claim arising from the rejection of executory contracts or unexpired leases ("Rejection Claims") pursuant to Confirmation of the Plan or otherwise must be filed with the Claims and Balloting Agent no later than the later of ______, 2024 (the date that is 30 days after the Effective Date), or 30 days after the effective date of rejection. Rejection Claims shall be classified as Class 3 General Unsecured Claims and shall be treated in accordance with Section 10.3 of the Plan, as applicable. Any Rejection Claims that are not timely filed pursuant to Section 14.2 of the Plan or the Confirmation Order shall be forever disallowed and barred.
- 5. Deadline to File Professional Fee Claims. As provided for in Section 8.2 of the Plan and in the Confirmation Order, all final applications for payment of the Professional Fee Claims must be filed with this Court and served on counsel to

¹ Capitalized terms not defined herein shall have the meanings ascribed to them in the Plan.

1	the Debtors, counsel to the Liquidating Trustee, counsel to the Co-Liquidating					
2	Trustee, and the U.S. Trustee no later					
3	after the Effective Date), unless otherwise ordered by this Court or such later date is					
4	agreed to by the Liquidating Trustee.					
5	6. Inquiries by Interested Parties. Copies of the Confirmation Order (to					
6	which the Plan is attached as Exhibit A) may be examined free of charge at					
7	http://www.kccllc.net/Borregohealth. The Confirmation Order is also on file with the					
8	Bankruptcy Court and may be viewed by accessing the Bankruptcy Court's website					
9	at <u>www.cacb.uscourts.gov</u> . To access documents on the Bankruptcy Court's website,					
10	you will need a PACER password and login, which can be obtained at					
11	www.pacer.uscourts.gov.					
12						
13						
14	Dated: January 16, 2024 DENTONS US LLP SAMUEL R. MAIZEL					
15	TANIA M. MOYRON REBECCA M. WICKS					
16	By: /s/ Tania M. Moyron					
17	Tania M. Moyron					
18	Attorneys for the Chapter 11 Debtor and Debtor In Possession					
19						
20						
21						
22						
23						
24						
25						
26						
27						
28						

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-1 Pg. 1 of 1

NCP Rev. 12/03 United States Bankruptcy Court Southern District of California Jacob Weinberger U.S. Courthouse 325 West F Street San Diego, CA 92101–6991 Telephone: 619–557–5620 Website: www.casb.uscourts.gov Hours: 8:30am – 4:30pm Monday–Friday

BORREGO COMMUNITY HEALTH FOUNDATION,

587 Palm Canyon Dr. Suite 208

Borrego Springs, CA 92004

Debtor Aliases: Desert Home Care

Case number: 22-02384-LT11

Chapter: 11

Judge Laura S. Taylor

NOTICE OF ORDER CONFIRMING CHAPTER 11 PLAN

Notice is hereby given that an Order Confirming the Chapter 11 plan was entered on 1/25/24. The confirmed plan may include a discharge of debts.

Dated: 1/25/24 By order of the court:

Michael Williams

Clerk of the Bankruptcy Court

Notice Recipients

District/Off: 0974-3 User: Admin. Date Created: 1/25/2024

Case: 22-02384-LT11 Form ID: 149 Total: 5398

Recipients of Notice of Electronic Filing: ust United States Trustee u ustp.region15@usdoj.gov Haeji Hong Haeji.Hong@usdoj.gov Tania M. Moyron tania.moyron@dentons.com aty aty

TOTAL: 3

	TOTAL: 3
	submitted to the BNC (Bankruptcy Noticing Center):
db	BORREGO COMMUNITY HEALTH FOUNDATION, 587 Palm Canyon Dr. Suite 208 Borrego
	Springs, CA 92004 Devid Ordin DOL Het 980 Front Street Devid A Ordin Series 2220 See Diseas CA 92101
aty	David Ortiz DOJ-Ust 880 Front Street David A. Ortiz Suite 3230 San Diego, CA 92101 Samuel Ruven Maizel Dentons US LLP 601 South Figueroa Street Suite 2500 Los Angeles,
aty	Samuel Ruven Maizel Dentons US LLP 601 South Figueroa Street Suite 2500 Los Angeles, CA 90017
smg	
smg	Franchise Tax Board Attn: Bankruptcy P.O. Box 2952 Sacramento, CA 95812–2952 IRS–Insolvency Division P.O. Box 7346 Philadelphia, PA 19101–7346
14962282	1000Bulbs 2140 Merritt Dr Garland, TX 75041
14962283	1060 Graphics 41 N 2nd St Saint Clair, PA 17970–1027
14962284	211 SAN DIEGO 3860 Calle Fortunada Ste 101 San Diego, CA 92123
14962285	24/7 MOBILE LOCKSMITH SERVICE 717 N Union St Ste 42 Wilmington, DE 19805
14962286	340B HEALTH 1101 15th Street NW, Suite 910 340B Coalition Conference
	Registration Washington, DC 20005
14962287	3M HEALTH INFORMATION SYSTEMS, INC. P.O. Box 120881 DEPT 0881 Dallas, TX
	75312–0881
14962288	40/30 Dental 1166 East Main Street El Cajon, CA 92021
14962289	40/30 Dental Inc. 6175 El Cajon Blvd. San Diego, CA 92215
14962290	4IMPRINT 101 Commerce Street Oshkosh, WI 54901
14962291	501(c) AGENCIES TRUST 400 Race St., Suite 200 San Jose, CA 95126 A 2 Z TOWING 72310 Ouarry Trl, Suite B Thousand Palms, CA 92276
14962292 14962293	A 2 Z TOWING 72310 Quarry Trl, Suite B Thousand Palms, CA 92276 A AARDVARK SELF STORAGE 4490 Murphy Canyon Rd San Diego, CA 92123
14962294	A&E Factory Services, LLC 1560 Cable Ranch Rd San Antonio, TX 78245
14962295	A&H PARTY RENTALS INC P.O. Box 7457 10241 HOLE AVE Riverside, CA 92503
14962296	A+ SIGN SHOPS 9119 Baseline Rd. Rancho Cucamonga, CA 91730
14962297	A2Z LOCK & SECURITY 915 E Main St Barstow, CA 92311
14962298	AAMES LOCK & SAFE COMPANY 818 W Chapman Ave Orange, CA 92868–2823
14962299	AAPC P.O. Box 639237 Cincinnati, OH 45263–9237
14962304	AB STAFFING SOLUTIONS 3451 S Mercy Road, Ste 102 Gilbert, AZ 85297
14962303	AB Staffing 3451 Mercy Rd. #102 Gilbert, AZ 85297
14970331	AB Staffing Solutions, LLC c/o Bryan Cave Leighton Paisner, LLP Attention: Mark Mersel and Olivia
1.40.60205	Scott 1920 Main Street, Ste. 1000 Irvine, CA 92614 ABC Company 123 Anywhere St San Diego, CA 92101
14962305	ABC Company 123 Anywhere St San Diego, CA 92101 ABC Recovery Center 44374 Palm St Indio, CA 92201
14962306 14962307	ABC TRANSLATING & INTERPRETING SERVICES P.O. Box 6474 San Diego, CA 92166
14962308	ABEO (ASMG) 3750 Convoy St Suite 125 San Diego, CA 92111
14962314	ABILITY NETWORK, INC. P.O. Box 856015 Minneapolis, MN 55485–6015
14962316	ABP BACKFLOW 6069 Rancho Mission Road #103 San Diego, CA 92108
14962318	ABRAHAM GHORBANIAN MOBILE DENTAL CORP. 31 MUSICK Irvine, CA 92618
14962320	ARRAMSON I FVIN & GINDI I I P 11846 Ventura Blvd Suite 100 Studio City CA 91604
14962322	ABSOLUTE RV CARE 541 N. Palm Ave ACADEMY OF ONCOLOGY NURSE (AONN+) ACE COOLERS, INC 7852 Raytheon Road ACE HARDWARE 1785 E Palm Canyon Dr ACE MARDWARE 1785
14962323	ACADEMY OF ONCOLOGY NURSE (AONN+) 1249 S River Rd Ste 202A Cranbury, NJ 08512
14962325	ACE COOLERS, INC 7852 Raytheon Road San Diego, CA 92111
14962326	ACE HARDWARE 1/85 E Paim Canyon Dr Paim Springs, CA 92264
14962328	ACGME 29376 Network Place Chicago, IL 60673–1293
14962329 14962330	ACTIVE EVENTS 717 North Harwood Street, Suite 2500 Dallas, TX 75201 ACTSOFT 10006 N. Dale Mabry Hwy. Suite 100 Tampa, FL 33618
14962334	ADAP/PrEP–AP Enrollment Site Contract P.O. Box 997377 MS 0500 Sacramento, CA
14702334	95899–7377 Mis 0500 Sacramento, e.r.
14962335	ADDRESSERS 15700 Texaco Ave Paramount, CA 90723
14962336	ADECCO EMPLOYMENT SERVICES DEPT LA 21403 Pasadena, CA 91185–1403
14962338	ADESSA SOLUTIONS, LLC 2300 N. Creek Vista Dr. Tucson, AZ 85749
14962341	ADOBE CREATIVE CLOUD 345 Park Avenue San Jose, CA 95110–2704
14962342	ADP, Inc. P.O. Box 31001–1874 Pasadena, CA 91110–1874
14962346	ADRIANA CHAVEZ RAMIREZ Address Redacted
14962351	ADT 1501 Yamato Road Boca Raton, FL 33431
14962352	ADT COMMERCIAL P.O. Box 219044 PROTECTION ONE SECURITY-61633590 Kansas City,
14062252	MO 64121–9044 ADT COMMEDIAL DO Poy 210044 PROTECTION ONE SECURITY 61622070 Kansas City
14962353	ADT COMMERCIAL P.O. Box 219044 PROTECTION ONE SECURITY–61633970 Kansas City, MO 64121–9044
14962354	ADT SECURITY SERVICES, INC P.O. Box 371956 Pittsburgh, PA 15250–7956
14962355	ADT SECURITY SERVICES, INC 1.0. Box 371930 Tittsburgh, PA 15250–7878 ADT SECURITY SERVICES, INC STE A P.O. Box 371878 Pittsburgh, PA 15250–7878

```
ADT SECURITY SERVICES, INC. ADMIN B2 ADT SECURITY SERVICES, INC. ADMIN E
                                                        P.O. Box 371956
                                                                           Pittsburgh, PA 15250-7956
14962357
                                                       P.O. Box 371956
                                                                          Pittsburgh, PA 15250-7956
          ADT SECURITY SVCS, INC
                                        P.O. Box 371878
                                                            Pittsburhg, PA 15250-7878
14962358
                                      83509 HWY 111
14962359
                                                          INDIO, CA 92201
          ADVANCE AUTO & AIR
14962360
          ADVANCED DENTAL ARTS, INC.
                                               185 W. Madison Ave
                                                                      El Cajon, CA 92020
14962362
          ADVANCED EXERCISE EQUIPMENT INC.
                                                      861 SouthPark Drive, #100
                                                                                  Littleton, CO 80120
14962363
                           1800 International Park Drive, Suite 110
                                                                  Birmingham, AL 35243
                                         Lexington, KY 40512
14962364
                      P.O. Box 14079
          AETNA
14962368
          AFC URGENT CARE OF SANTEE
                                              750 Otay Lakes Road, #272
                                                                          Chula Vista, CA 91910
14962369
          AFFORDABLE RV HIGH DESERT
                                              11854 MARIPOSA ROAD
                                                                          HESPERIA, CA 92345
14962370
          AFFORDABLE TELEPHONE SYSTEMS
                                                   P.O. Box 300784
                                                                       Escondido, CA 92030
14962371
                       1932 Wynnton Rd
                                           Columbus, GA 31993-8601
          AFLAC
14962373
          AFRICAN AMERICAN HEALTH COALITION
                                                        1505 West Highland, Suite 9
                                                                                      San Bernardino, CA
14962375
          AG-PCPI SKY PARK OWNER, L.P.
                                               245 PARK AVENUE, FL 24
                                                                            NEW YORK, NY 10167
14962374
          AGENICS
                        862 Hwy 105
                                        Palmer Lake, CO 80133
14962376
                       233 N. Michigan Avenue 21st Floor
                                                          Chicago, IL 60601-5809
          AHIMA
14962377
          AHLEE BACKFLOW SERVICES, INC
                                                 9920 Prospect Ave, Ste. 104
                                                                              Santee, CA 92071
14962381
                                             Durham, NC 27707-8110
          AICPA
                      220 Leigh Farm Road
14962386
          AIRGAS USA - 1683345
                                     P.O. Box 102289
                                                         Pasadena, CA 91189-2289
14962388
          AIRMEDCARE NETWORK
                                        1800 Air Medical Drive Suite 1
                                                                       West Plains, MO 65775
                                   1965 Gillespie Way #105
          AKW MEDICAL, INC.
14962390
                                                              El Cajon, CA 92020
14962391
          AKW Medical-Equip Calib & Maintenance
                                                   1965 Gillespie Way #105
                                                                              El Cajon, CA 92020
14962490
          AL-TAR SERVIĈEŜ, INC
                                     P.O. Box 398138
                                                          San Francisco, CA 94139–8138
14962393
          ALAN M. FLAUM
                                Address Redacted
14962394
          ALARM MONITORING SERVICES, INC
                                                   P.O. Box 14323
                                                                      Palm Desert, CA 92255
14962395
          ALASKA AIRLINES
                                  P.O. Box 68900
                                                    Seattle, WA 98168
14962404
          ALCOHOL COUNTERMEASURE SYSTEM CORP
                                                            60 International Boulevard
                                                                                       Etobicoke, ON M9W
          6J2 Canada
14962423
          ALEPH GROUP, INC.
                                   1900 East Alessandro Blvd. #105
                                                                    Riverside, CA 92508
14962438
          ALFRED SANTOS &BARBARA SANTOS & TRUSTEES
                                                                 P.O. Box 1217
                                                                                  Pico Rivera, CA 90660
                                                             La Jolla, CA 92037
14962439
          ALFREDO RATNIEWSKI
                                      8695 Glenwick Lane
          ALFREDOS PIZZA & PASTA
                                         251 W Base Line St
                                                                San Bernardino, CA 92410
14962440
14962445
          ALICES VEHICLE REGISTRATIONS OF RAMONA
                                                             1140 Main Street, Ste 204
                                                                                        Ramona, CA
14962454
          ALIVE SCAN
                            2707 Garnet Ave, #3
                                                   San Diego, CA 92109
14962455
          ALL CITY PLUMBING
                                    9360 7TH ST UNIT B
                                                            Rancho Cucamonga, CA 91730
          ALL-AMERICAN PUBLISHING
14962456
                                            P.O. Box 100
                                                             Caldwell, ID 83606
14962459
          ALLIANCE DIESEL
                              1309 S. Juanita Ste. A
                                                         San Jacinto, CA 92583
                                                                       West Chester, PA 19382
14962460
          ALLIANCE PROTECTION
                                       1646 West Chester Pike Suite 31
          ALLIANCE PROTECTION – CC 2211460
                                                   45130 Golf Center Parkway Suite A
14962461
                                                                                      Indio, CA 92201
14962462
          ALLIANCE PROTECTION/ALLIANCE INTEGRATION
                                                                45130 Golf Center Parkway Suite A
                                                                                                   Indio, CA
                                     5701 Golden Hills Drive
14962463
          ALLIANZ INSURANCE
                                                              Minneapolis, MN 55416-1297
14962469
          ALLIED X-RAY, INC
                                   18307 Septo Street
                                                        Northridge, CA 91325
14962485
          ALPHA CARE MEDICAL GROUP
                                             P.O. Box 570215
                                                                 Tarzana, CA 91357
14962486
          ALPHA FUND BETA CORP
                                        1443 Danville Boulevard
                                                                  Alamo, CA 94507-1911
14962487
                             1321 N. Gene Autry Trail
                                                       Palm Springs, CA 92262
          ALPHA MEDIA
                           3231 S. Florence Ln.
14962489
          ALSCO, INC
                                                  Yuma, AZ 85365
14962491
          ALTIUM HEALTHCARE
                                      P.O. Box 505525
                                                          St. Louis, MO 63150-5525
14962493
                                   4475 Campus Ave #4
          ALVA G ALVARADO
                                                           San Diego, CA 92116
          AMAZON 410 Terry Ave. N
AMERICAN ACADEMY OF CME
14962502
                                             Seattle, WA 98109
                                              129 N County Line Rd
14962509
                                                                      Jackson, NJ 08577
14962512
          AMERICAN AIRLINES
                                    P.O. Box 619616
                                                        DFW Airport
                                                                     Fort Worth, TX 75261-9616
          AMERICAN BOARD OF FAMILY MEDICINE
                                                        1648 McGrathiana Parkway, Suite 550
14962513
                                                                                             Lexington, KY
          40511-1247
          AMERICAN BOARD OF PODIATRIC
14962514
                                                 1060 Aviation Blvd. #100
                                                                            Hermosa Beach, CA 90254
14962515
          AMERICAN CANCER SOCIETY
                                             6377 Riverside Ave. Suite 203
                                                                           Riverside, CA 92506
14962516
          AMERICAN COACH LIMOUSINE
                                              1100 Jorie Blvd Suite 314
                                                                         Oak Brook, IL 60523
14962517
          AMERICAN CONFERENCE
                                        545 East John Carpenter Freeway
                                                                          Third Floor
                                                                                        Irving, TX
          75062
14962520
          AMERICAN DIABETES ASSOCIATION
                                                   611 Wilshire Blvd Ste 900
                                                                              Los Angeles, CA 90017
14962521
          AMERICAN ENVIRONMENTAL GROUP
                                                                                      Westlake Village, CA
                                                    5665 Lindero Canyon Rd, Suite 701
                                    P.O. Box 53852
                                                       Phoenix, AZ 85072-3852
14962523
          AMERICAN EXPRESS
          AMERICAN FLOOR MATS
14962524
                                        152 Rollins Ave #102
                                                               Rockville, MD 20852
                                                   155 N. Wacker Dr.
14962525
          AMERICAN HOSPITAL ASSOCIATION
                                                                        Chicago, IL 60606
          AMERICAN INTERNET SERVICES / NFINIT
                                                       9305 Lightwave Ste 100
14962527
                                                                                 San Diego, CA 92123
                                                  330 North Wabash Ave Ste 39300
14962528
          AMERICAN MEDICAL ASSOCIATION
                                                                                    Chicago, IL
          60611-5885
          AMERICAN NURSE ASSOCIATION
14962530
                                                1107 9th St. #350
                                                                    Sacramento, CA 95864
14962532
          AMERICAN POWER SECURITY SERVICE, INC
                                                         1451 S Rimpau Ave., Ste. #207
                                                                                          Corona, CA
          AMERICAN SOCY-RADIOLOGIC TECHNOLOGISTS
14962533
                                                                15000 Central Ave. SE
                                                                                        Albuquerque, NM
           87123-3909
                                                                     Barstow, CA 92311
          AMERICAN TOWING & RECOVERY
14962534
                                                 2520 E MAIN ST
                                                               Cincinnati, OH 45263-3553
14962535
          AMERICORP FINANCIAL, LLC
                                            P.O. Box 633553
14962536
          AMERIGAS-201711540 CVHS
                                           45900 Commerce St. Suite 1
                                                                       Indio, CA 92201-3402
```

```
14962538
           AMERISOURCEBERGEN DRUG CORP
                                                     P.O. BOX 100565
                                                                           Pasadena, CA 91189-0565
14962539
                                        Newton, MA 02458
           AMION
                        248 Park St
                                74794 42ND AVE, SUITE 1
14962541
           AMS CONNECT
                                                               PALM DESERT,, CA 92260
                                74794 42nd Avenue, Suite 1
14962542
           AMS CONNECT
                                                              Palm Desert, CA 92260
14962543
           AMTRAK.COM
                                1 Massachusetts Ave. NW
                                                            Washington, DC 20001
14962544
           AMY COLLEEN FRENCH
                                         Address Redacted
14962577
           ANDREW J GELLENS
                                      Address Redacted
14962578
                                     49950 California-111
           ANDYS BURGERS #5
                                                              Coachella, CA 92236
14962587
           ANGELA T DEARINGER
                                        6120 Hiddenaway Ln.
                                                                 Versailles, KY 40383
           ANNAMARIA ORTIZ SOLIS
14962608
                                           Address Redacted
14962614
           ANSOM PRODUCTIONS, INC.
                                             3755 Mactibby St
                                                                  San Diego, CA 92117
                                       P.O. Box 933657
                                                            Atlanta, GA 31193-3657
14962616
           ANTHEM BLUE CROSS
14962618
           ANTHEM BLUE CROSS PPO - PRUDENT BUYER
                                                                220 Virginia Ave
                                                                                    Indianapolis, IN 46204
14962622
           ANTHROPICS
                                               London, W12 7SB United Kingdom
                              56 Wood Ln
                                         2345 4th Avenue
14962627
           ANYTHINGTRUCK.COM
                                                             Moline, IL 61265
14962630
           ANZA-BORREGO DESERT NATURAL HISTORY
                                                               P.O. Box 310
                                                                                 Borrego Springs, CA 92004
14962632
                      2530 Professional Road, Suite 200
                                                          Richmond, VA 23235
           APMS
14962633
           APPLE STORE
                              One Apple Park Way
                                                       Cupertino, CA 95014
                                        19410 Business Ctr Dr
                                                                  Northridge, CA 91324
14962634
           APPLIANCE PARTS PRO
14962635
           APPLIANCE PARTS PROS.COM
                                               19410 Business Ctr Dr.
                                                                         Northridge, CA 91324
14962642
           ARCONE, RAFAEL M.D
                                        Address Redacted
                                   P.O. Box 230515
14962658
           ARMYNAVY.COM
                                                       Brooklyn, NY 11223
14962659
           ARRABELLE GOMEZ
                                     Address Redacted
           ASAP Drain Guys & Plumbing ASD SPECIALTY HEALTHCARE
                                            999 Rancheros Dr. Suite B
14962670
                                                                        San Marcos, CA 92069
                                                3101 GAYLORD PARKWAY
14962671
                                                                                Frisco, TX 75034
14962682
           ASIS International
                                 1625 Prince Street
                                                       Alexandria, VA 22314-2882
                                                                                  CORONA, CA 92879-3332
14962683
           ASPEN MEDICAL GROUP, INC
                                              341 MAGNOLIA AVE, STE #201
14962684
           ASPEN MILLS BREAD COMPANY
                                                 555 S Sunrise Way #101
                                                                            Palm Springs, CA 92264
                                                         Indio, CA 92203
                                41700 Jackson Street
14962685
           ASPHALT M.D.S
14962688
           AT & T - LM - 288665022
                                         P.O. Box 5014
                                                            Carol Stream, IL 60197-5014
           AT &T-LM-133528765
60197-5014
                                                         Now under AT&T-LM 288665022
14962689
                                      P.O. Box 5014
                                                                                             Carol Stream, IL
           AT T -LM - 619-463-1047
                                                            Carol Stream, IL 60197-5014
14962690
                                          P.O. Box 5014
14962691
           AT&T
                       208 S. Akard St.
                                          AT&T Payment Center
                                                                     Dallas, TX 75202
                                         eet Dallas, TX 75202
Carol Stream, IL 60197–5025
14962692
           AT&T
                       208 South Akard Street
14962693
                       P.O. Box 5025
           AT&T
14962694
           AT&T-858-292-1474 289 2 Sky Park
                                                   208 S Akard St
                                                                      Dallas, TX 75202
                                                   New York, NY 10285-3106
14962695
           AT&T-AMEX
                              200 Vesey Street
14962696
                     901 N. Glebe Rd Ste 850
                                                 Arlington, VA 22203
           ATA
14962699
           ATHENS SERVICES-RV0002918-EHC BLDG B
                                                             P.O. Box 60009
                                                                                 City of Industry, CA
           91716-0009
                                                                         City of Industry, CA 91716-0009
14962700
           ATHENS SERVICES-RV0419784-EHC
                                                     P.O. Box 60009
14962701
           ATHENS SERVICES-RV0440324-ARLANZA
                                                           P.O. Box 60009
                                                                               City of Industry, CA
           91716-0009
14962705
           ATLAS ENVIRONMENTAL SERVICES, INC.
                                                           9032 Olive Dr.
                                                                             Spring Valley, CA 91977
14962706
           ATTEST HEALTH CARE ADVISORS, LLC
                                                        24042 VECCHIO
                                                                             SAN ANTONIO, TX 78260
14962708
           AURORA AUDIO LLC
                                      520 W Lake Mary Blvd, Ste 200
                                                                       Sanford, FL 32773
14962711
                              2575 E Camelback Rd Suite 700
           AXA Equitable
                                                                Phoenix, AZ 85016
           AXIUM SERVICES
14962712
                                   9187 Clairemont Mesa Blvd., Suite 6-741
                                                                             San Diego, CA 92123
14962714
           AYRES HOTEL
                               355 Bristol Street
                                                    Costa Mesa, CA 92626
14962300
           Aaron Caminos
                              Address Redacted
14962301
           Aaron Gonzalez
                               Address Redacted
14962302
           Aaron Peck
                           Address Redacted
14962309
           Abigail Lara
                            Address Redacted
14962310
           Abigail Medina
                              Address Redacted
14962311
           Abigail Salas
                            Address Redacted
14962312
           Abigail Serrano
                              Address Redacted
14962313
           Abijail Guillen Lopez
                                    Address Redacted
14962315
           Abner Vilches
                             Address Redacted
14962317
           Abraham Ezrilov
                                Address Redacted
14962319
           Abraham Vergara
                                Address Redacted
14962321
           Abril Soto
                         Address Redacted
14962324
           Access Clinic Services
                                    477 N. El Camino Real, Suite A100
                                                                         Encinitas, CA 92024
14962327
           Ace Pediatrics
                             621 North State Street #3
                                                         San Jacinto, CA 92583
                                                74839 42nd Ave
14962350
           AdRythm Advertising & Design, Inc
                                                                    Palm Desert, CA 92260
14962331
           Adam Delgadillo
                                Address Redacted
14962332
           Adam Newcomb
                               Address Redacted
14962333
           Adamari Hernandez Avendano
                                           Address Redacted
14962337
           Adelanto Elementary School District
                                                11824 Air Expressway
                                                                          Adelanto, CA 92301
14962339
           Adeyinka Shoroye
                                 Address Redacted
           Adilenee Torrez Lopez
14962340
                                    Address Redacted
14962343
           Adrian Hernandez
                                 Address Redacted
14962344
           Adrian Lanza
                             Address Redacted
           Adrian Valencia
14962345
                               Address Redacted
14962347
           Adriana Gonzales-Magallon
                                         Address Redacted
14962348
                                Address Redacted
           Adrianna Aceves
14962349
           Adrienne Brown
                               Address Redacted
14962361
           Advanced Dynamo Industries, Inc.
                                               19425 SW 89th Ave
                                                                      Tualatin, OR 97062
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 4 of 76

```
14962365
           Aetna
                       P.O. Box 639715
                                             Cincinnati, OH 45263
14962366
           Aetna Better Health of California
                                                10260 Meanley Drive
                                                                          San Diego, CA 92131
14962367
            Afaf Al Hayani
                                Address Redacted
14962372
           Afrah Halabia
                               Address Redacted
14962378
            Ahmad Zayed
                               Address Redacted
14962379
            Ahmed Al-Tameemi
                                     133 West Main Street, Suite 100
                                                                        El Cajon, CA 92020
14962380
            Ahmed Al-Tameemi
                                     Address Redacted
14962382
            Aide Anava
                             Address Redacted
14962383
            Aids Healthcare Foundation
                                           6255 Sunset Blvd., 21st Fl.
                                                                          FKA Hillcrest Pharmacy
                                                                                                      Los Angeles, CA
           90028
14962384
            Airgas
                        259 North Radnor-Chester Road, Suite 100
                                                                      Radnor, PA 19087
14962385
            Airgas Puritan Medical
                                       259 N Radnor Chester Rd #100
                                                                                      Radnor, PA 19087
                                                                          Airgas
14962387
            Airmed
                         1800 Air Medical Drive
                                                     West Plains, MO 65775
14962389
           Akanksha Kumar
                                  Address Redacted
14962392
            Alan Kuehn
                             Address Redacted
14962396
            Albert Co
                           Address Redacted
14962397
            Albert Guzman
                                Address Redacted
14962398
           Alberto Garcia
                               Address Redacted
14962399
           Albertsons LLC
                                250 East Parkcenter Blvd
                                                             Boise, ID 83706
14962400
            Alborz Mehdizadeh, DDS
                                         2801 West Empire Avenue
                                                                        Bleau Fox, a PLC - Megan Ann
                          Burbank, CA 91504
           Childress
            Alborz Mehdizadeh, Inc
14962401
                                        15080 7th Street, Suite 7
                                                                     Victorville, CA 92395
14962402
           Alborz Mehdizadeh, Inc
                                        286 N. San Jacinto Street
                                                                     Hemet, CA 92543
14962403
            Alborz Mehdizadeh, Inc.
                                        2800 Pacific Avenue, Suite B
                                                                         Long Beach, CA 90806
14962405
            Aldairi DDS, Inc.
                                  101 Broadway, Suite 3A
                                                              The Maul Firm PC-Anthony F. Maul
                                                                                                       Oakland, CA
            94607
14962406
            Aldairi DDS, Inc.
                                  6175 El Cajon Blvd.
                                                           El Cajon, CA 92115
14962407
            Aldo Villada
                             Address Redacted
14962408
            Alejandra Alvarez
                                  Address Redacted
14962409
            Alejandra Cortez Ramirez
                                         Address Redacted
14962410
            Alejandra Dealba
                                  Address Redacted
14962411
                                 Address Redacted
            Alejandra Garcia
14962412
            Alejandra Lopez
                                 Address Redacted
14962413
            Alejandra Magana
                                   Address Redacted
14962414
            Alejandra Mariscal
                                   Address Redacted
14962415
            Alejandra Martinez
                                    Address Redacted
14962416
            Alejandra Moreno Ibarra
                                        Address Redacted
14962417
            Alejandra Valenzuela Naranjo
                                             Address Redacted
14962418
            Alejandro Cervantes
                                    Åddress Redacted
14962419
           Alejandro Garcia Bravo
                                        Address Redacted
14962420
           Alejandro Martinez
                                    Address Redacted
14962421
            Alejandro Mondragon
                                      Address Redacted
14962422
            Alejandro Vargas
                                  Address Redacted
           Alepio Quillopo
14962424
                                 Address Redacted
14962425
            Alex Adamian
                               Address Redacted
14962426
           Alexa Palomino
                                Address Redacted
14962427
            Alexandra Lopez
                                 Address Redacted
14962428
           Alexandra Miller
                                  Address Redacted
                                   Address Redacted
14962429
            Alexandra Sanchez
14962430
           Alexandria Zavala
                                   Address Redacted
14962431
                             Address Redacted
            Alexia Diaz
14962432
            Alexis Ahumada
                                 Address Redacted
14962433
           Alexis Dinero
                               Address Redacted
14962434
           Alexis Espadas
                               Address Redacted
14962435
           Alexis Garcia
                              Address Redacted
14962436
           Alexis Leal
                            Address Redacted
                                Address Redacted
14962437
            Alexus Sustaita
14962441
            Alhan Halabia
                               Address Redacted
14962442
           Ali Al-Qaissi
                              Address Redacted
14962443
           Ali Moosa
                           Address Redacted
14962444
           Ali Nori
                         Address Redacted
14962446
           Alicia Arias Rojas
                                  Address Redacted
14962447
           Alicia Beltran-Beltran
                                      Address Redacted
14962448
           Alicia Canales
                               Address Redacted
14962449
            Alicia Elizalde
                               Address Redacted
14962450
                               Address Redacted
            Alicia Elizalde
14962451
            Alicia Herrera
                               Address Redacted
14962452
            Alicia Salazar
                               Address Redacted
14962453
            Alinea Medical Imaging
                                        1818 N Orange Grove Ave, Suite 101
                                                                                Pomona, CA 91767-3028
14962457
            Allan Arce
                            Address Redacted
14962458
           Allen Ford
                            Address Redacted
14962464
           Allied World National Assurance Company
                                                         1690 New Britain Ave., Suite 101
                                                                                              Farmington, CT
           06032
14962465
            Allied World National Assurance Company
                                                         27 Richmond Road
                                                                                 Pembroke, HM 08 Bermuda
14962466
            Allied World Specialty Insurance Company
                                                         199 WATER STREET
                                                                                    New York, NY 10038
                                                                                 Pembroke, HM 08 Bermuda
                                                         27 Richmond Road
14962467
            Allied World Specialty Insurance Company
14962468
            Allied World Surplus Lines Insurance Co
                                                        199 WATER STREET
                                                                                  New York, NY 10038
14962470
           Allison Medina
                               Address Redacted
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 5 of 76

		0170	
14962471 14968280	Ally Bank AIS Portfolio Services, LLC	ENIX, AZ 85062–8369 4515 N Santa Fe Ave. Dept. APS	Oklahoma City, OK
14968281	73118 Ally Bank AIS Portfolio Services, LLC	4515 N Santa Fe Ave. Dept. APS	Oklahoma City, OK
14968389	73118 Ally Bank AIS Portfolio Services, LLC 73118	4515 N Santa Fe Ave. Dept. APS	Oklahoma City, OK
14976660	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14976662	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14976676	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14976677	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14977284	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14977289	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14977290	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14978355	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14978427	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14981201	Ally Bank c/o AIS Portfolio Services, LLC 73118	4515 N. Santa Fe Ave. Dept. APS	Oklahoma City, OK
14962472 14962473	Ally Financial 500 Woodward Ave Alma Anguiano Address Redacted	Ally Detroit Center Detroit, MI 48	3226
14962474	Alma Beltran Address Redacted		
14962475	Alma Felix Address Redacted		
14962476	Alma Felix Address Redacted Alma Felix Silva Address Redacted		
14962477	Alma Hernandez Address Redacted		
14962477	Alondra Bernal Address Redacted Address Redacted		
14962478	Alondra Bernal Address Redacted Alondra Bernal Address Redacted		
14962480			
14962481	Alondra Felix Address Redacted		
14962482	Alondra Hurtado Address Redacted		
14962483	Alonso Araujo Vega Address Redacted	* 11 4 #200 411 1 G4 0100	.4
14962484		ield Ave, #200 Alhambra, CA 9180	
14962488		Sarfield Ave., 2nd Fl Alhambra, CA	91801
14962492	Alva Alvarado Address Redacted		
14962494	Alvaro Delgadillo Address Redacted		
14962495	Amanda Abarca Address Redacted		
14962496	Amanda Ardessi Address Redacted		
14962497	Amanda Dashiell Address Redacted		
14962498	Amanda Guina Address Redacted		
14962499	Amanda Troncoso Address Redacted		
14962500	Amanda Zora Address Redacted		
14962501	Amar Najjar Address Redacted		
14962503	Ambar Gonzalez Address Redacted		
14962504	Amber Browning Address Redacted		
14962505	Amber Perez Address Redacted		
14962506	Amber Roesch Address Redacted		
14962507	Ambrosia Healthcare 750–60 Gerald For Desert, CA 92211	rd Dr, Units 1–2 Ambrosia Compou	inding Pharmacy Palm
14962508			
14962510	AmeriBen P.O. Box 7186 Boise, I	D 83707-1186	
			eawood, KS 66211
14962511	American Academy of Family Physicians	11400 Tomahawk Creek Pkwy Le	
	American Academy of Family Physicians American Academy of Pediatrics P.O. B	11400 Tomahawk Creek Pkwy Le ox 776442 Chicago, IL 60677–6442	2
14962511 14962518	American Academy of Family Physicians American Academy of Pediatrics P.O. B American Dental Association 28094 NE	11400 Tomahawk Creek Pkwy Leox 776442 Chicago, IL 60677–6442 FWORK PLACE Chicago, IL 6067	2
14962511 14962518 14962519	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. P.O. B 28094 NE 19425 S	11400 Tomahawk Creek Pkwy Leox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706	2 3 52
14962511 14962518	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street	11400 Tomahawk Creek Pkwy Leox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706 Laureen E. Seeger New York, N	2 3 52 Y 10285
14962511 14962518 14962519 14962522 14962526	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services 9305 Lightw	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212	2 3 52 Y 10285
14962511 14962518 14962519 14962522 14962526 14962529	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express American Internet Services American Medical Response P.O. B 28094 NE 19425 S 28094 NE 280	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412	2 3 52 Y 10285
14962511 14962518 14962519 14962522 14962526	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association 142 F	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 O Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–25	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962537	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response File 554120 American Osteopathic Association Amerisource Bergen Drug Corporation	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–25	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962537	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Control Industries American Osteopathic Association American Magana Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 TWORK PLACE Chicago, IL 6067 SW 89th Avenue Tualatin, OR 9706 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 O Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–25	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962537 14962540 14962545	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association Amerisource Bergen Drug Corporation Amparo Magana Address Redacted Amy Elenbaum Arediatrics P.O. B 28094 NE 19425 S 28094 NE 19425 S	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962537 14962540 14962545 14962546	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association American Costeopathic Association American Magana Address Redacted Amy Elenbaum Address Redacted Ana Bonilla Rodriguez Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962537 14962540 14962545 14962546 14962547	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association American Magana Address Redacted Amy Elenbaum Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Carrillo Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962537 14962540 14962545 14962547 14962547	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association Amerisource Bergen Drug Corporation Amparo Magana Address Redacted Any Elenbaum Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Carrillo Address Redacted Ana Fernandez Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962540 14962545 14962546 14962547 14962548 14962549	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association American Osteopathic Association American Osteopathic Association Amparo Magana Address Redacted Any Elenbaum Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Carrillo Address Redacted Ana Fernandez Address Redacted Ana Garcia Londono Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962540 14962545 14962546 14962547 14962548 14962549 14962550	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association American Osteopathic Association Amparo Magana Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Fernandez Address Redacted Ana Fernandez Address Redacted Ana Garcia Londono Address Redacted Ana Garcia Londono Address Redacted Ana Gianni Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962537 14962540 14962545 14962547 14962548 14962549 14962549 14962550 14962551	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association American Osteopathic Association Amparo Magana Address Redacted Any Elenbaum Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Fernandez Address Redacted Ana Garcia Londono Address Redacted Ana Garcia Londono Address Redacted Ana Gianni Address Redacted Ana Hernandez Address Redacted Ana Hernandez Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962537 14962540 14962545 14962546 14962547 14962548 14962549 14962550 14962551 14962551	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association Amparo Magana Address Redacted Amy Elenbaum Address Redacted Ana Bonilla Rodriguez Andress Redacted Ana Fernandez Andress Redacted Ana Garcia Londono Address Redacted Ana Gianni Address Redacted Ana Hernandez Address Redacted Ana Hernandez Address Redacted Ana Hernandez Address Redacted Ana Jimenez Meraz Address Redacted Ana Jimenez Meraz Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962540 14962545 14962546 14962547 14962548 14962549 14962550 14962551 14962552 14962553	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association Amparo Magana Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Carrillo Address Redacted Ana Garcia Londono Address Redacted Ana Gianni Address Redacted Ana Hernandez Address Redacted Ana Jimenez Meraz Address Redacted Ana Jimenez Meraz Address Redacted Ana Lomeli Address Redacted Ana Lomeli Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962540 14962545 14962546 14962547 14962548 14962549 14962550 14962551 14962553 14962553	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association Amparo Magana Address Redacted Ana Bonilla Rodriguez Anderss Redacted Ana Garrillo Anderss Redacted Ana Garnil Address Redacted Ana Gianni Address Redacted Ana Gianni Address Redacted Ana Gianni Address Redacted Ana Gianni Address Redacted Ana Jimenez Meraz Address Redacted Ana Lomeli Address Redacted Ana Lomeli Address Redacted Ana Lomeli Address Redacted Ana Lomeli Address Redacted Ana Loza Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3
14962511 14962518 14962519 14962522 14962526 14962529 14962531 14962540 14962545 14962546 14962547 14962548 14962549 14962550 14962551 14962552 14962553	American Academy of Family Physicians American Academy of Pediatrics American Dental Association American Dental Industries, Inc. American Express 200 Vesey Street American Internet Services American Medical Response American Osteopathic Association American Osteopathic Association Amparo Magana Address Redacted Ana Bonilla Rodriguez Address Redacted Ana Carrillo Address Redacted Ana Garcia Londono Address Redacted Ana Gianni Address Redacted Ana Hernandez Address Redacted Ana Jimenez Meraz Address Redacted Ana Jimenez Meraz Address Redacted Ana Lomeli Address Redacted Ana Lomeli Address Redacted	11400 Tomahawk Creek Pkwy ox 776442 Chicago, IL 60677–6442 IWORK PLACE SW 89th Avenue Chicago, IL 6067 Laureen E. Seeger New York, N' ave, Suite 100 San Diego, CA 9212 D Los Angeles, CA 90074–5412 E. Ontario St. Chicago, IL 60611–23 West First Avenue Conshohocken	2 3 52 Y 10285 3

```
14962556
           Ana Mann-Cervera
                                   Address Redacted
                           Address Redacted
14962557
            Ana Mora
14962558
           Ana Oliveros
                              Address Redacted
            Ana Ortega Rojas
14962559
                                  Address Redacted
14962560
           Ana Robles
                            Address Redacted
14962561
            Ana Vazquez
                              Address Redacted
14962562
            Ana Villafana
                              Address Redacted
14962563
           Anabel Balderas
                                Address Redacted
14962564
            Anabel Fuentes
                                Address Redacted
14962565
           Anabel Ortiz
                             Address Redacted
14962566
            Anadaisy Guerrero
                                   Address Redacted
14962567
            Anahi Gonzalez Vergara
                                        Address Redacted
14962568
            Anay Marquez
                               Address Redacted
14962569
            Andre Harnanto
                                Address Redacted
           Andrea Fragoso
14962570
                                Address Redacted
14962571
            Andrea Garavito Olivares
                                        Address Redacted
14962572
           Andrea Taylor
                               Address Redacted
14962573
            Andrea Urquidez
                                 Address Redacted
14962574
           Andres De La Llana
                                    Address Redacted
14962575
            Andres Rodriguez
                                  Address Redacted
14962576
            Andres Smith
                              Address Redacted
14962579
            Angel Garcia
                              Address Redacted
14962580
           Angel Garcia Saavedra
                                      Address Redacted
14962581
            Angel Marin
                             Address Redacted
            Angel Sanchez
14962582
                               Address Redacted
14962583
            Angela Elias
                             Address Redacted
14962584
            Angela Martinez
                                 Address Redacted
14962585
            Angela Olivares
                                Address Redacted
14962586
            Angela Spieckerman
                                    Address Redacted
            Angelica Arciga
14962588
                                Address Redacted
14962589
            Angelica Ayala
                                Address Redacted
14962590
            Angelica Beltran
                                 Address Redacted
14962591
            Angelica Briseno
                                 Address Redacted
14962592
            Angelica Campos
                                  Address Redacted
14962593
            Angelica Khoder
                                 Address Redacted
14962594
            Angelica Lopez
                                Address Redacted
14962595
            Angelica Ortiz
                               Address Redacted
14962596
            Angham Joseph
                                Address Redacted
14962597
            Anissa Madrid
                               Address Redacted
14962598
                         Address Redacted
            Ann Ala
14962599
            Ann Cadoret
                             Address Redacted
14962600
           Ann Roberts
                             Address Redacted
14962601
            Ann Vita
                          Address Redacted
14962602
           Anna Cervantes
                                Address Redacted
14962603
            Anna Liza Floresca
                                   Address Redacted
14962604
           Anna Lopez
                            Address Redacted
14962605
            Anna Najera
                             Address Redacted
           Anna Navarro
14962606
                              Address Redacted
            Annalisa Rodriguez
14962607
                                    Address Redacted
14962609
            Annamaria Solis
                                Address Redacted
14962610
            Annette Elizondo Alvarezsoto
                                             Address Redacted
14962611
            Annette Perez
                              Address Redacted
14962612
            Annie Le
                          Address Redacted
14962613
            Annmarie Charles
                                  Address Redacted
14962615
            Anthem Blue Cross
                                   P.O. Box 73651
                                                        Cleveland, OH 44193
14962617
            Anthem Blue Cross (Blue View Vision)
                                                     3075 Vandercar Way
                                                                               Cincinnati, OH 45209
            Anthony A Foster
                                  Address Redacted
14962619
14962620
            Anthony Bullum
                                 Address Redacted
14990210
                                                                                          360 E. 2nd Street, Suite
           Anthony R. Bisconti
                                    BIENERT KATZMAN LITTRELL WILLIAMS LLP
                     Los Angeles, CA 90012
                                                 Telephone (213 528–3400
                                                                              Email: tbisconti@bklwlaw.com
14962621
            Anthony Torres
                                Address Redacted
14962623
            Antoinia Simpson
                                  Address Redacted
14962624
           Antonia Dorsay
                                Address Redacted
14962625
           Antonio Ramos
                                Address Redacted
14962626
            Antonio Rosselli
                                 Address Redacted
                          P.O. Box 391909
                                                58470 Highway 371
                                                                         Anza, CA 92539
14962628
            Anza ECI
14962629
            Anza Valley Pharmacy/Wellpartner
                                                  56555 Highway 371
                                                                          Anza, CA 92539
            Apex Medical Group
                                     311 West I Street
                                                          Los Banos, CA 93635
14962631
            April Reyes
14962636
                            Address Redacted
14962637
            Araceli Ğarcia-Perez
                                     Address Redacted
14962638
            Aracelly Palma
                               Address Redacted
14962639
            Aram Arakelyan, D.D.S
                                       Address Redacted
14962640
            Aram Arakelyan, DDS
                                      2801 West Empire Avenue
                                                                    Bleau Fox, a PLC - Megan Ann
                          Burbank, CA 91504
            Childress
14962641
            Arcelia Gonzalez
                                 Address Redacted
14962643
            Areej Gabola
                              Address Redacted
14962644
            Aresenia P. Jared
                                 Address Redacted
           Argelia Tello
14962645
                              Address Redacted
```

```
14962646
           Argo Capital Group
                                  1555 BOTELHO DR #433,
                                                                WALNUT CREEK, CA 94596
14962647
           Ariana Arciniega
                                Address Redacted
14962648
           Aries Amerson
                              Address Redacted
           Arisahi Lizarraga
                                Address Redacted
14962649
14962650
           Ark Homes Agency
                                  9645 Arrow Route, Bldg 5, Suite A
                                                                       Ark Homes Foster Family Agency
                                                                                                          Rancho
           Cucamonga, ČA 91730
14962651
           Arlanza Family Health Center
                                           8856 Arlington Avenue
                                                                     Riverside, CA 92503
14962652
                                    8856 Arlington Avenue
           Arlanza Health Center
                                                               Riverside, CA 92503
14962653
           Arlene Gonzalez
                               Address Redacted
           Arlene Martinez-Partida
14962654
                                      Address Redacted
14962655
           Arlene Plascencia-Frias
                                      Address Redacted
           Arlete Pichardo
14962656
                               Address Redacted
14962657
           Arline Ramirez
                              Address Redacted
14962660
                          Address Redacted
           Arsenio Sy
14962661
           Arthur J. Gallagher & Co
                                       505 N Brand Blvd, Suite 600
                                                                      Glendale, CA 91203
14962662
           Arthur J. Gallagher & Co.
                                       500 N. Brand Blvd. #100
                                                                   Michelle Gonzalez
                                                                                         Glendale, CA 91203
14962663
           Arthur Santos Smile Dental
                                         12110 Woodside Avenue
                                                                                      Lakeside, CA 92040
                                                                    Arthur Santos
14962664
           Arturo Amador Perez
                                   Address Redacted
14962665
           Arturo Castro Ramos
                                   Address Redacted
14962666
           Arturo Duenas-Rosales
                                      Address Redacted
14962667
           Arturo Reyes
                            Address Redacted
                            Address Redacted
14962668
           Arturo Rojas
14962669
           Aruba Airwave
                              6280 America Center Dr
                                                          San Jose, CA 95002
14962672
           Aseel Bunni
                            Address Redacted
14962673
           Asembia
                         200 Park Avenue Suite 300
                                                      Florham Park, NJ 07932
14962674
           Ashley Barron
                              Address Redacted
14962675
           Ashley Mercado
                               Address Redacted
14962676
           Ashley Miranda
                               Address Redacted
           Ashley Moreno
                              Address Redacted
14962677
           Ashley Orozco
14962678
                              Address Redacted
           Ashley Perea
                            Address Redacted
14962679
14962680
           Ashraf Zadshir
                              Address Redacted
           Ashwin Ravisankar
14962681
                                  Address Redacted
14962687
           Assoc OF FAMILY MEDICINE ADMINISTRATION
                                                                 11400 Tomahawk Creek Parkway
                                                                                                    Leawood, KS
           66211-2672
14962686
           Assoc for Hospital Medical Education
                                                  P.O. Box 725
                                                                    Indiana, PA 15701
14962697
           Atheer Butrus
                             Address Redacted
14962698
           Athens Services
                               14048 East Valley Blvd
                                                         City of Industry, CA 91746
14962702
           Athil Badeel
                            Address Redacted
14962703
           Athziri Morales
                              Address Redacted
14962704
           Atlantic Specialty Insurance Company
                                                  605 Highway 169 North, St 800
                                                                                    Plymouth, MN 55441
14962707
           Audrey Sanchez
                               Address Redacted
14962709
           Autumn Ashcraft
                                Address Redacted
14962710
           Aven Kelo
                          Address Redacted
14962713
           Ayed Hawatmeh, DDS
                                     101 Broadway, Suite 3A
                                                                The Maul Firm PC–Anthony F. Maul
                                                                                                      Oakland, CA
           94607
           B&B APPLIANCE
14962715
                                  1743 S. Escondido Boulevard
                                                                 Escondido, CA 92025
14962716
                             3691 W 34th St.
                                                New York, NY 10001
           B&H PHOTO
           B.S. YOUTH & SENIOR CENTER
                                                P.O. Box 1362
14962717
                                                                   Borrego Springs, CA 92004
           BABCOCK LABORATORIES, INC.
                                                                    Riverside, CA 92502
14962718
                                                 P.O. BOX 432
                               1201 Connecticut Avenue NW, Suite 200H
14962719
           BABYSCRIPTS
                                                                           Washington, DC 20036
14962721
           BACTES (Now SHARECARE)
                                             8344 Clairemont Mesa Blvd. Suite 201
                                                                                    San Diego, CA 92111
14962722
           BAD QUAIL
                            7217 Saddleback Rd
                                                    Joshua Tree, CA 92252
           BALLOONY TUNES 67–502 E Palm Cyn Drive
BARBARA JONES CLEANING SERVICE 6045
14962723
                                                                 Cathedral City, CA 92234
14962727
                                                                                  Anza, CA 92539
                                                       60450 Yucca Valley Rd
14962730
           BARRERA & ASSOCIATES
                                           2298 E. MAPLE AVENUE
                                                                         EL SEGUNDO, CA 90245
           BARSTOW AREA CHAMBER OF COMMERCE
                                                             P.O. BOX 698
14962735
                                                                                Barstow, CA 92312-0698
14962736
           BARSTOW COMMUNITY HOSPITAL
                                                    820 E. Mountain View Street
                                                                                    MEDICAL STAFF
           ASSOC
                        Barstow, CA 92311
           BARSTOW GLASS & MIRROR
14962737
                                              1110 W MAIN ST
                                                                    Barstow, CA 92311
14962738
           BARSTOW HEALTH PARTNERS
                                                820 E. Mountain View St.
                                                                            Barstow, CA 92311
14962739
           BARSTOW TIRE AND BRAKE INC.
                                                   100 W Main St
                                                                     Barstow, CA 92311-2221
           BARSTOW TOWING, INC.
                                                                 Barstow, CA 92311
14962740
                                          2241 West Main St.
14962744
           BASIC
                       P.O. Box 1914
                                          Borrego Springs, CA 92004
           BAY ALARM COMPANY
14962750
                                         P.O. Box 51041
                                                            Los Angeles, CA 90051-5337
14962751
           BAY ALARM COMPANY
                                         P.O. Box 7137
                                                            San Francisco, CA 94120
14962752
           BAY ALARM COMPANY -2941932
                                                                                  Ontario, CA 91761
                                                  740 S Rochester Ave, Suite D
           BAY CITY ELECTRIC WORKS,INC
14962753
                                                  P.O. Box 641054
                                                                       Dallas, TX 75264-1054
                                                                 Leverkusen, 51373 Germany
Borrego Springs, CA 92004
14962754
           BAYER
                       Kaiser-Wilhelm-Allee 1
                                                    Bayer Global
                                              P.O. Box 2369
14962755
           BCHF Supplemental Savings Plan
           BEACON HEALTH
14962757
                                   200 State Street
                                                      Boston, MA 02109
                                                5225 Sheridan Drive
14962763
           BEC INTEGRATED SOLUTIONS
                                                                       Georgetown Square
                                                                                               Williamsville, NY
           14221
           BECK OIL, INC - 601653 SF
14962764
                                           16640 D Street
                                                              Victorville, CA 92395
           BEN FRANKLIN FINANCE – DAVIDSON, JOHN
14962768
                                                              P.O. Box 771852
                                                                                   DETROIT, MI 48277-1852
14962769
           BENCO DENTAL
                                 P.O. Box 731372
                                                      Dallas, TX 75373-1372
14962772
           BENLO
                        15633 Arrow Hwy
                                              Irwindale, CA 91706
```

P.O. Box 743074

Los Angeles, CA 90074–3074

14962779

BEST BEST & KRIEGER, LLP

```
14962780
           BEST BUY
                          7601 Penn Ave. S.
                                               Richfield, MN 55423
           BEST WESTERN DATE TREE HOTEL
                                                  6201 N. 24th Parkway
14962781
                                                                            Phoenix, AZ 85016
                              15373 Innovation Dr # 120
                                                            San Diego, CA 92128
14962783
           BETA Healthcare
          BETA Healthcare Group C/o Geraiu IN. 51

Avenue. Suite 400 San Diego, CA 92101
                                     c/o Gerald N. Sims
14992238
                                                           Pyle Sims Duncan & Stevenson, APC
                                                                                                 1620 Fifth
          BETA Healthcare Group c/o Susan C. Stevenson
Avenue, Suite 400 San Diego, CA 92101
14992239
                                                               Pyle Sims Duncan & Stevenson, APC
                                                                                                     1620 Fifth
           BETA Risk Management Authority
                                              1443 Danville Boulevard
14962784
                                                                         BETA Healthcare Group
                                                                                                    Alamo, CA
14962789
           BHC PHARMACEUTICALS
                                          P.O. Box 10435
                                                             Palatine, IL 60055-0435
           BIG SKY COMMUNICATIONS INC
14962792
                                                 124 TAILGATE LANE
                                                                           HAMILTON, MT 59840
           BILLING PARADISE 24X7 INC
                                             53, Emerald Road
                                                                  Robbinsville, NJ 08691
14962793
           BIOLYTICAL LABORATORIES
14962794
                                             406 – 13251 Delf Place
                                                                       Richmond, BC V6V 2A2 Canada
14962795
           BIOMED
                       P.O. Box 463076
                                             Escondido, CA 92046-3076
           BIOPLUS SPECIALTY PHARMACY SERVICES, INC
14962796
                                                                376 Northlake Boulevard
                                                                                            Altamonte Springs, FL
           32701
          BIZ FURNITURE 4350 Ball Ground Highway Canton, GA 30114
BIZMANUALZ, INC. 7733 Forsyth Blvd. Suite 1100
14962797
14962798
14962799
                                                                   Clayton, MO 63105
14962800
           BKD, LLP (Now FORVIS, LLP)
                                            P.O. Box 1190
                                                              910 E. St. Louis Street, Suite 200
                                                                                                 Springfield, MO
           65801-1190
14962807
           BLESH & ASSOC. REAL ESTATE, INC
                                                    112 E. State St.
                                                                       Redlands, CA 92373
                                                             Mesa, AZ 85205-3306
14962808
           BLINDS CHALET
                                1946 N Higley Rd Ste 201
14962809
                    5511 Virginia Way Suite 150
                                               Brentwood, TN 37027
                                                777 N Ashley Drive
14962813
           BLUE MOUNTAIN CONSULTING
                                                                      Tampa, FL 33602
                             P.O. Box 272540
                                                 Chico, CA 95927–2540
14962814
           BLUE SHIELD
14962815
           BLUE SHIELD OF CA
                                    P.O. Box 241012
                                                         Lodi, CA 95241-9512
           BLUE SHIELD-PROMISE
14962821
                                        601 Potrero Grande Drive
                                                                    FORMALLY KNOWN AS CARE 1ST SAN
                         Monterey Park, CA 91755–7430
ARMACY 2720 Gateway Oaks Drive, Suite 100
           DIEGO CTY
           BOARD OF PHARMACY
14962823
                                                                             Sacramento, CA 95833
14962824
                           112 S Main St Ste 2F Saint Charles, MO 63301
           BOARDPAQ
           BOB STALL CHEVROLET
14962825
                                       P.O. Box 338
                                                          La Mesa, CA 91941
14962826
           BOB TURNERS CRANE SERVICE, INC. 12101 Highway 67
                                                                          Lakeside, CA 92040
                                                             Madison, WI 53713
14962827
           BOOKS4SCHOOLS.COM
                                       201 E. Badger Rd.
14962829
           BORREGO AUTO PARTS & SUPPLY CO.
                                                      P.O. Box 343
                                                                       2476 Stirrup Rd
                                                                                           Borrego Springs, CA
           92004
                                                                     Borrego Springs, CA 92004
14962831
           BORREGO FESTIVALS FOUNDATION
                                                    P.O. Box 420
           BORREGO PEST CONTROL P.O. Box 2428
                                                            Borrego Springs, CA 92004
14962833
14962836
           BORREGO SPRINGS CHAMBER OF COMMERCE
                                                              P.O. Box 420
                                                                               Borrego Springs, CA
           92004-0420
14962838
           BORREGO SPRINGS COMMUNITY CONCERT Assoc
                                                                 P.O. Box 2426
                                                                                    Borrego Springs, CA
           92004
14962839
           BORREGO SPRINGS LITTLE LEAGUE
                                                    P.O. Box 844
                                                                     Borrego Springs, CA 92004
14962840
                                                   1112 Tilting T Rd.
                                                                         Borrego Springs, CA 92004
           BORREGO SPRINGS RESORT HOTEL
                                                           642 Palm Canyon Drive
14962841
           BORREGO SPRINGS ROTARY FOUNDATION
                                                                                     Borrego Springs, CA
           92004
14962842
           BORREGO SPRINGS SENIOR CENTER
                                                    P.O. Box 1316
                                                                       Senior Center Benefit
                                                                                               Borrego Springs,
           CA 92004
14962843
           BORREGO SPRINGS UNIFIED SCHOOL DISTRICT
                                                               1315 Palm Canyon Drive
                                                                                           Borrego Springs, CA
           92004
           BORREGO SPRINGS YOUTH & SENIOR CENTER
14962844
                                                               P.O. Box 1362
                                                                                 Borrego Springs, CA
           92004-1362
           BORREGO SUN
14962845
                               P.O. Box 249
                                                Borrego Springs, CA 92004
14962847
           BORREGO VALLEY HANGARS
                                              615 Avenida Acapulco
                                                                      San Clemente, CA 92672
                                     405 PALM CANYON DR
14962848
           BORREGO VALLEY INN
                                                                    Borrego Springs, CA 92004
14962850
           BORREGO WATER DISTRICT 005002-000 W SIDE
                                                               P.O. Box 1870
                                                                                  Borrego Springs, CA
           92004-1870
14962851
           BORREGO WATER DISTRICT 005002-001 E SIDE
                                                               P.O. Box 1870
                                                                                 Borrego Springs, CA
           92004-1870
14962852
           BORREGO WATER DISTRICT 005002-002 -AL
                                                            P.O. Box 1870
                                                                               806 Palm Canyon
                     Borrego Springs, CA 92004–1870
14962853
           BORREGO WATER DISTRICT 005002-003 AL
                                                           P.O. Box 1870
                                                                             Borrego Springs, CA
           92004-1870
14962854
           BOSTON SCIENTIFIC CORPORATION
                                                    P.O. Box 512638
                                                                         Los Angeles, CA 90051-0638
           BOX APPLIANCE 13851 DANIELSON ST Poway, BOXWOOD TECHNOLOGY IC 1430 Spring Hill Rd # 6
                                                           Poway, CA 92064
14962855
                                                                       Mc Lean, VA 22102-3000
14962856
           BOYS & GIRLS CLUBS OF GREATER SAN DIEGO
14962857
                                                               P.O. Box 1051
                                                                                 630 Cahuilla Rd
                                                                                                     Borrego
           Springs, CA 92004
           BRADLEY BUILDERS, INC P.O. Box 891805 Temecula, CA 92589
BRAND NEW DAY – UNIVERSAL CARE 5455 Garden Grove Blvd, Suite 500
14962858
14962863
                                                                                           Westminster, CA
           92683
                                                                  Seattle, WA 98136-1833
14962864
           BRANDPRIDE LLC
                                  6523 California Ave SW, #329
           BRE CA Office Owner LLC
14962867
                                        P.O. Box 209259
                                                            BLDG ID 18592
                                                                                Austin, TX 78720-9259
                      P.O. BOX 849991
                                           DALLAS, TX 75284
14962874
           BREG
14962904
           BROADWELL BEHAVIORAL HEALTH SERVICES INC
                                                                   2048 Edinburg Ave
                                                                                          Cardiff, CA 92007
14962905
           BROCK & GONZALES, LLP
                                          6701 Center Drive West, Ste. 610
                                                                            Los Angeles, CA 90045
          BRUDVIK, INC
BSN MEDICAL INC.

600 Eugene Road
5825 Camegic
14962912
                                                Palm Springs, CA 92264
14962915
                                   5825 Camegie Blvd.
                                                          Charlotte, NC 28209
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 9 of 76

```
14962916
           BUDGET BLINDS
                                   3425 N Indian Canyon Drive, Suite 6
                                                                          Palm Springs, CA 92262
                                   501 Congress Ave. Ste 150 Austin, TX 78701
SUPPLY 2502 South 78h St. Tacoma, W
14962917
           BUILDERS AREA
14962919
           BURKHART DENTAL SUPPLY
                                                                      Tacoma, WA 98409-9053
                                                          P.O. Box 7187
           BURRTEC WASTE & RECYCLING SVCS.
14962923
                                                                             Payment Processing Center
                                                                                                            Buena
           Park, CA 90622-7187
           BURRTEC WASTE & RECYCLING SVCS.CVHS
14962924
                                                                P.O. Box 7187
                                                                                   Payment Processing
                       Buena Park, CA 90622-7187
14962925
           BURRTEC WASTE & RECYCLING-CV Calhoun
                                                                P.O. Box 7187
                                                                                   Payment Processing
           Center
                       Buena Park, CA 90622-7187
           BURRTEC WASTE INDUSTRIES, INC - BARSTOW
14962926
                                                                   P.O. Box 6240
                                                                                       Buena Park, CA 90622
14962928
           BUYDOORHARDWARE
                                         1009 NE Jib Ct, Ste C
                                                                   Lees Summit, MO 64064
14962720
                                                                       Franklin, TN 37067
           Bactes-Imaging-Sharecare\\
                                            701 Cool Springs Blvd
           Banning Unified School District
14962724
                                              161 West Williams St.
                                                                        Banning, CA 92220-4746
14962725
           Barbara Cretan-Cuciz
                                     Address Redacted
14962726
           Barbara Egerdahl
                                 Address Redacted
           Barbara Lytle
14962728
                              Address Redacted
14962729
           Barbara Scott Luca
                                  Address Redacted
14962731
           Barrie Abbott
                              Address Redacted
14962732
           Barry Lumkin
                              Address Redacted
           Barry Tischler
Barry Tischler, M.D.
14962733
                              Address Redacted
14962734
                                    Address Redacted
14962741
           Barstow Unified School District
                                              551 South Avenue H
                                                                       Barstow, CA 92311-2500
14962742
           Bart Danby
                            Address Redacted
14962743
           Barton Associates
                                 3700 S Maryland Pkwy #525d
                                                                   Las Vegas, NV 89119
14962745
           Basila Sabagh
                              Address Redacted
14962746
           Basilio Llora
                             Address Redacted
                                Address Redacted
14962747
           Basima Odeesho
14962748
           Basma Al Shaikhli
                                  Address Redacted
14962749
           Basma Saleem
                               Address Redacted
                        505 Huntmar Park Drive Suite #300
14962756
                                                              Herndon, VA 20170
           Beacon
14962758
                               Address Redacted
           Beatrice Brown
14962759
           Beatrice Garcia
                               Address Redacted
14962760
           Beatriz Garibay
                                Address Redacted
14962761
           Beatriz Gonzalez Medina
                                        Address Redacted
14962762
           Beatriz J Culbertson
                                    Address Redacted
14962765
           Belgica Esquivel
                                 Address Redacted
14962766
           Belgica Paulino
                                Address Redacted
14962767
           Belinda Anderson
                                 Address Redacted
14962770
           Benjamin Lam
                               Address Redacted
14962771
           Benjamin Yeung
                                Address Redacted
14962773
           Bennie Velasquez
                                 Address Redacted
14962774
           Berenice Garibaldi
                                  Address Redacted
14962775
           Berkeley Research Group
                                        550 S. Hope Street, Suite 2150
                                                                          Frank Stevens
                                                                                             Los Angeles, CA
           90071
14962776
           Berkeley Research Group
                                        P.O. Box 676158
                                                              Dallas, TX 75267-6158
           Berkeley Research Group, LLC
Bernard Wosk Address Re
14962777
                                             P.O. Box 676158
                                                                   Dallas, TX 75267-6158
14962778
                              Address Redacted
14962782
           Beta Health Group
                                 P.O. Box 619084
                                                        Roseville, CA 95661
           Betsy Sanchez Briones
                                      Address Redacted
14962785
14962786
           Bettina Collins
                              Address Redacted
14962787
           Beverly Nicole Callaway
                                        Address Redacted
           Beverly Radiology Medical Group
14962788
                                                8750 Wilshire Blvd
                                                                        Beverly Hills, CA 90211-2713
14962790
           Bianca Alomar
                               Address Redacted
14962791
           Bianca Cabrales
                                Address Redacted
14962801
                             Address Redacted
           Blair Honore
14962802
           Blaire Heath
                             Address Redacted
14962803
           Blake Wylie
                             Address Redacted
14962804
           Blanca Arredondo
                                  Address Redacted
14962805
                                 Address Redacted
           Blanca Rodriguez
14962806
           Blanca Ruiz
                            Address Redacted
14962810
           Blue Cross Blue Shield of Tennessee
                                                  1 Cameron Hill Circle
                                                                            Chattanooga, TN 37402
14962811
           Blue Cross of California
                                       P.O. Box 10557
                                                           Atlanta, GA 30348-5557
           Blue Cross of California.
                                                            Los Angeles, CA 90060
14962812
                                       P.O. Box 60007
                                   P.O. Box 272620
14962819
           Blue Shield Promise
                                                         Chino, CA 95927-2620
                                                                            Monterey Park, CA 91755
           Blue Shield Promise Health Plan
14962820
                                               601 Potreto Grande Drive
14962818
           Blue Shield of CALIFORNIA
                                            P.O. BOX 272560
                                                                   Chico, CA 95927–2560
                                                       Chico, CA 95927-2620
                                 P.O. BOx 272620
14962816
           Blue Shield of Ca
                                                             Lodi, CA 95241
14962817
           Blue Shield of California
                                       P.O. Box 241012
14962822
                            3130 West Maple Loop
           BlueEO LLC
                                                        Ste 103
                                                                    Lehi, UT 84043
14962828
                                   P.O. Box 2383
                                                       Borrego Springs, CA 92004
           Borrego Art institute
14962830
           Borrego Community Health Foundation
                                                     655 Palm Canvon Dr
                                                                              Borrego Springs, CA 92004
14962832
           Borrego Management Services Organization
                                                         955 HARBOR ISLAND DR STE 162
                                                                                                 SAN DIEGO, CA
           92101
14962834
           Borrego Pharmacy-Barstow
                                           Rx Options
                                                           EnvisionRx
                                                                            Twinsburg, OH 44087
14962835
           Borrego Pharmacy-Cathedral City
                                                Rx Options
                                                                 EnvisionRx
                                                                                 Twinsburg, OH 44087
           Borrego Springs Childrens Center
                                                580 CÎRCLE J DRIVE
14962837
                                                                          BORREGO SPRINGS, CA 92004
14962846
           Borrego Valley Endowment Fund
                                               P.O.BOX 2714
                                                                   BORREGO SPRINGS, CA 92004
           Borrego Water District
14962849
                                     806 Palm Canyon Drive
                                                                 Borrego Springs, CA 92004
```

```
14962859
           Bradley S Fischl
                               Address Redacted
           Bradley University
                                 1501 W Bradley Ave
                                                          Peoria, IL 61625
14962860
14962861
           Braille Institute
                               741 N. Vermont Avenue
                                                          Los Angeles, CA 90029
14962862
           Brand New Day
                               5455 Garden Grove Blvd, Suite 500
                                                                    Westminster, CA 92683
14962865
           Bravo Dental Group of Corona
                                            1185 Magnolia Avenue #K & #L
                                                                              Corona, CA 92879
14962866
                                       391 W Central Ave
           Bravo Dental of Banning
                                                              Brea, CA 92821-3025
14962868
           Breana Hicks
                             Address Redacted
14962869
           Breanda Troncoso
                                 Address Redacted
14962870
           Breanna Gutierrez
                                 Address Redacted
14962871
           Breanna Plata
                             Address Redacted
14962872
           Breanna Ramirez
                                Address Redacted
14962873
           Bree Francis
                            Address Redacted
14962875
           Brenda Cabrera Landeros
                                       Address Redacted
14962876
           Brenda Duenas
                              Address Redacted
14962877
           Brenda Figueroa
                                Address Redacted
14962878
           Brenda Macias Ortiz
                                   Address Redacted
14962879
           Brenda McDermott
                                  Address Redacted
14962880
           Brenda Palomares
                                 Address Redacted
                               Address Redacted
14962881
           Brenda Ramirez
14962882
           Brendha Flores
                              Address Redacted
14962883
           Brendi S Rodriguez
                                  Address Redacted
14962884
           Brian Beck
                           Address Redacted
14962885
           Brian Blacksher
                               Address Redacted
14962886
           Brian Jensen
                            Address Redacted
14962887
           Brian Reves
                            Address Redacted
14962888
           Brian Scannell
                              Address Redacted
14962889
           Briana Wischnack
                                 Address Redacted
14962890
           Brianda M Duran
                                Address Redacted
14962891
           Brianna Fields
                              Address Redacted
14962892
           Brianna Hennessy
                                 Address Redacted
14962893
           Brianna Widman
                                Address Redacted
14962894
           Bridget Moran
                              Address Redacted
14962895
           Bridgette Aviles
                               Address Redacted
14962896
           Brigette Briones
                               Address Redacted
14962897
           BrightEdge Technologies, Inc.
                                           989 E. Hillsdale Blvd., Suite 300
                                                                              Foster City, CA 94404
14962898
           Briseida Mora
                              Address Redacted
14962899
           Briselda Mendoza
                                 Address Redacted
           Brittney Harrell
                               Address Redacted
14962900
14962901
           Brittney Mothersill
                                  Address Redacted
           Brittney Young
14962902
                               Address Redacted
           Brittny Ramirez
14962903
                               Address Redacted
14962906
           Brodwell Behavioral Health Services, Inc
                                                                            EL CAJON, CA 92020
                                                     181 REA AVE, 201
14962907
           Brownstein Hyatt Farber Schreck
                                              410 Seventeenth Street Suite 2200
                                                                                  Denver, CO 80202-4432
           Bruce Bainbridge
14962908
                                Address Redacted
14962909
           Bruce Hinton
                             Address Redacted
14962910
           Bruce Janzen
                             Address Redacted
14962911
           Bruce Penny
                            Address Redacted
14962913
           Bryan Hendry
                             Address Redacted
14962914
           Bryan Young
                             Address Redacted
           Burke, Williams & Sorensen LLP 90071–2953
14962918
                                              444 South Flower Street Suite 2400
                                                                                   Los Angeles, CA
14962920
           Burrell College of Osteopathic Medicine
                                                    3501 Arrowhead Dr
                                                                            Las Cruces, NM 88001
                                       P.O. Box 5518
14962921
           Burrtec Environmental SF
                                                          Buena Park, CA 90622-6859
14962922
           Burrtec Waste & Recycling Services
                                                P.O. Box 7187
                                                                   Buena Park, CA 90622-7187
           Burrtec Waste Industries, Inc.
                                           9400 Cherry Ave
                                                                               Fontana, CA 92335
14962927
                                                                Building C
14962929
           Byron Jones
                            Address Redacted
14962930
           C&M FENCE CO
                                 2280 E. Main St.
                                                      Barstow, CA 92311
14962931
           CA DEPT OF PUBLIC HEALTH
                                               ATTN RADIOLOGIC HEALTH BRANCH
                                                                                           P.O. BOX
                       SACRAMENTO, CA 95899-7414
           997414
14962932
           CA DEPT OF PUBLIC HEALTH
                                               P.O. Box 997434
                                                                   Sacramento, CA 95899
14962933
           CA DEPT OF PUBLIC HEALTH -CDPH
                                                      P.O. Box 997376, Unit MS1601
                                                                                        Accounting
           Section/Cashiering Unit
                                      Sacramento, CA 95899-7376
           CA DEPT OF PUBLIC HEALTH RHB
14962934
                                                   P.O. Box 997414
                                                                         MS 7610
                                                                                       Sacramento, CA
           95899-7414
14962935
           CA DEPT OF SOCIAL SERVICES
                                                P.O. Box 944243
                                                                     MS 9-3-67
                                                                                      Sacramento, CA
           94244-2430
14962936
           CA Dept of Tax &Fee ADMIN (CDTFA)
                                                     35900 Bob Hope Dr, Ste 280
                                                                                    Rancho Mirage, CA 92270
                                                            San Marcos, CA 92078
14962937
           CA OFFICE DESIGN
                                    1797 Lemon Tree Ct
                                                   P.O. Box 942533
14962938
           CA STATE BOARD OF PHARMACY
                                                                        DEPARTMENT OF CONSUMER
           AFFAIRS
                          Sacramento, CA 94258-0533
           CA STATE DISBURSEMENT UNIT
14962939
                                                  P.O. Box 989067
                                                                       CARRILLO, FRANK #0535
                                                                                                      WEST
           SACRAMENTO, CA 95798-9067
14962940
           CA STATE DISBURSEMENT UNIT
                                                  P.O. Box 989067
                                                                       CARRILLO, FRANK #1187
                                                                                                      WEST
           SACRAMENTO, CA 95798–9067
14962941
           CA STATE DISBURSEMENT UNIT
                                                  P.O. Box 989067
                                                                       CARRILLO, FRANK #3453
                                                                                                      WEST
           SACRAMENTO, CA 95798-9067
14962942
           CA STATE DISBURSEMENT UNIT
                                                  P.O. Box 989067
                                                                       DAVIDSON, JOHNATHAN
                                                                                                       WEST
           SACRAMENTO, CA 95798–9067
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 11 of 76

		01 /6		
14962943	CA STATE DISBURSEMENT UNIT	P.O. Box 989067	DAVILA, CRISTIAN	WEST
14962944	SACRAMENTO, CA 95798–9067 CA STATE DISBURSEMENT UNIT	P.O. Box 989067	ESPINOZA, ELMER	W WEST
14962945	SACRAMENTO, CA 95798–9067 CA STATE DISBURSEMENT UNIT	P.O. Box 989067	ESPINOZA, GILBER	T WEST
14962946	SACRAMENTO, CA 95798–9067 CA STATE DISBURSEMENT UNIT	P.O. Box 989067	MARTINEZ, FABIA	N WEST
14962947	SACRAMENTO, CA 95798–9067 CA STATE DISBURSEMENT UNIT 95798–9067	P.O. Box 989067	REED, RICK W	VEST SACRAMENTO, CA
14962948	CA STATE DISBURSEMENT UNIT SACRAMENTO, CA 95798–9067	P.O. Box 989067	ROSALES, JESUS	WEST
14962949	CA STATE DISBURSEMENT UNIT SACRAMENTO, CA 95798–9067	P.O. Box 989067	SHANAE ROGERS	WEST
14962950	CA STATE DISBURSEMENT UNIT CA 95798–9067	P.O. Box 989067	SMITH, DANNY	WEST SACRAMENTO,
14962951 14962952	CA STATE DISBURSEMENT UNIT CADENCE COMMUNICATIONS	P.O. Box 989067 34–400 Date Palm Driv	WEST SACRAMENT e Suite Q Cathedra	ГО, CA 95798–9067 ll City, CA 92234
14962953 14962954		egas Blvd Las Veg Van Ness Ave Tor	gas, NV 89109 rrance, CA 90501	
14962955	CAL COAST OPHTHALMIC 327 CAL DREAMSCAPE LANDSCAPE COI TERRACE, CA 92313	MPANY, INC. 22	421 BARTON ROAD #	286 GRAND
14962956	CAL FIRE-RIVERSIDE CTY FIRE DEP 92211	T 77–933 Las Mo	ontanas Rd., Ste 201	Palm Desert, CA
14962958			Francisco, CA 94139–83	42
14962960 14962961	CALI AUTO GLASS 15370 Cholar CALIFORNIA ACADEMY OF FAMILY 94109		ctorville, CA 92392 520 PACIFIC AVENUE	San Francisco, CA
14962964	CALIFORNIA BREASTFEEDING COAL	ITION 3665 Kee	ırny Villa Rd San I	Diego, CA 92123
14962965		D. Box 1015 Palm	Springs, CA 92263–101	5
14962966	CALIFORNIA COMMERCIAL SECURI 92123–1668	TY 9560 Ridgeha	ven Court Suite C	San Diego, CA
14962967 14962970	CALIFORNIA DENTAL ASSOCIATION CALIFORNIA DEPT PUBLIC HEALTH Unit Sacramento, CA 95899–7376	1201 K Street, 14 P.O. Box 997376	4th Floor Sacramer Acct Section/ Casl	nto, CA 95814 hiering
14962971	CALIFORNIA DEPT TAX & FEE ADMI 94279	N P.O. Box 9428	79 450 N Street	Sacramento, CA
14962972	CALIFORNIA DEPT. OF PUBLIC HEAI 95899–7434			Sacramento, CA
14962975	CALIFORNIA ELECTRIC SUPPLY			0-1915
14962976 14962978	CALIFORNIA HEALTH & WELLNESS CALIFORNIA HEALTH AND WELLNE 95833		yth Boulevard St L le Oaks Drive, Suite 200	ouis, MO 63105 Sacramento, CA
14962979	CALIFORNIA HEALTH INFORM	5055 E MCKINLEY A	VE Fresno, CA 93°	727
14962980	CALIFORNIA MEDICAL ASSOCIATIO			
14962983	CALIFORNIA PRIMARY CARE ASSOC	CIATION 1231 I S		cramento, CA 95814
14962984 14962986	CALIFORNIA SCHOOL–BASED HEAL 302 Oakland, CA 94612 CALIFORNIA SECRETARY OF STATE			Vay, Suite ento, CA 95814
14962980	CALIFORNIA TELEHEALTH RESOUR			n, CA 95763
14962991	CALIFORNIA WATER OPERATORS, L 92399	LC 34428 YUCA		
14962996	CAM COMMERCE SOLUTIONS, LLC 92683		re Blvd., Suite 100	Westminster, CA
14962997 14962998	CAMBRELL WINDOW EU M		Huntington Booch CA	02647
14962998		Lynbrook, NY 11:	Huntington Beach, CA	1 92047
14963000	CAPITOL ADMINISTRATORS, INC.		Nashville, TN 37230	L-6312
14963002	CARDINAL CVS 1140 S. MAIN S	T SALINAS, CA		0312
14963004	CARDINAL EL CAJON EXPRESS-7628 92262	32 1717 E VISTA	CHINO STE B2 P	ALM SPRINGS, CA
14963006	CARDINAL MEDPILOT PHARM INCO DIEGO, CA 92123			STE C1 SAN
14963007	CARDINAL PHARM - 69019 2768			02211
14963008 14963009	CARDINAL PHARM-340B 77770 BARS CARDINAL PHARM-COMM. OTC 692	00 File #56/12	IN ST BARST, CA Los Angeles, CA 900	
14963009	CARDINAL PHARM-COMM. 01C 692 CARDINAL PHARM-RETAIL 70984 CA 92004	590 PALM CANYO	ON DR STE 203 &	
			1 TNT 46545 4420	
14963011	CARDINAL PROMOTIONS 3102		waka, IN 46545–4438	
14963012	CARDINAL PROMOTIONS 3102 CARDINAL WALGREENS – 340 BT – 7	76210 464 CARD	INAL LN Green B	ay, WI 54313
14963012 14963014	CARDINAL PROMOTIONS 3102 CARDINAL WALGREENS – 340 BT – 3 CARE 1ST HEALTH PLAN 601 PC	76210 464 CARD otrero Grande Dr	INAL LN Green B Monterey Park, CA 9175	ay, WI 54313 5
14963012 14963014 14963016	CARDINAL PROMOTIONS 3102 CARDINAL WALGREENS – 340 BT – 7 CARE 1ST HEALTH PLAN 601 PC CAREERBUILDER.COM 200 N. L	76210 464 CARD otrero Grande Dr I aSalle St. Suite 1100	INAL LN Green B Monterey Park, CA 9175 Chicago, IL 60601	ay, WI 54313 5
14963012 14963014 14963016 14963017	CARDINAL PROMOTIONS 3102 CARDINAL WALGREENS – 340 BT – 7 CARE 1ST HEALTH PLAN 601 PC CAREERBUILDER.COM 200 N. I	76210 464 CARD otrero Grande Dr 1 aSalle St. Suite 1100 Street New York, I	INAL LN Green B Monterey Park, CA 9175. Chicago, IL 60601 NY 10011	ay, WI 54313 5
14963012 14963014 14963016 14963017 14963018	CARDINAL PROMOTIONS 3102 CARDINAL WALGREENS – 340 BT – 3 CARE 1ST HEALTH PLAN 601 PC CAREERBUILDER.COM 200 N. I CAREERMD.COM 110 East 28th S CAREPOINT, INC 997 Morrison D	76210 464 CARD otrero Grande Dr 1 LaSalle St. Suite 1100 Street New York, I rive, Suite 300 Ch	INAL LN Green B Monterey Park, CA 9175. Chicago, IL 60601 NY 10011 narleston, SC 29403	ay, WI 54313 5
14963012 14963014 14963016 14963017 14963018 14963019	CARDINAL PROMOTIONS CARDINAL WALGREENS – 340 BT – 3 CARE IST HEALTH PLAN CAREERBUILDER.COM 200 N. L CAREERMD.COM 110 East 28th S CAREPOINT, INC 997 Morrison D CARIBE ROYALE 8101 World Ce	76210 464 CARD otrero Grande Dr 1 LaSalle St. Suite 1100 Street New York, I rive, Suite 300 Ch nter Dr Orlando, F	INAL LN Green B Monterey Park, CA 9175. Chicago, IL 60601 NY 10011 narleston, SC 29403 FL 32821	ay, WI 54313 5
14963012 14963014 14963016 14963017 14963018 14963019 14963025	CARDINAL PROMOTIONS CARDINAL WALGREENS – 340 BT – 3 CARE 1ST HEALTH PLAN CAREERBUILDER.COM CAREERMD.COM CAREPOINT, INC CARIBE ROYALE CARIUM INC. 201 1ST STREET #	76210 464 CARD otrero Grande Dr 1 LaSalle St. Suite 1100 Street New York, I rive, Suite 300 Ch nter Dr Orlando, F 2211 PETALUMA	INAL LN Green B Monterey Park, CA 9175. Chicago, IL 60601 NY 10011 narleston, SC 29403 FL 32821 L, CA 94952	ay, WI 54313 5
14963012 14963014 14963016 14963017 14963018 14963019	CARDINAL PROMOTIONS CARDINAL WALGREENS – 340 BT – 3 CARE 1ST HEALTH PLAN CAREERBUILDER.COM CAREERMD.COM CAREPOINT, INC CARIBE ROYALE CARIUM INC. 201 1ST STREET #	76210 464 CARD otrero Grande Dr LaSalle St. Suite 1100 Street New York, I rive, Suite 300 Ch nter Dr Orlando, F 211 PETALUMA NYON DR Borres	INAL LN Green B Monterey Park, CA 9175. Chicago, IL 60601 NY 10011 narleston, SC 29403 FL 32821 L, CA 94952	ay, WI 54313 5

```
14963056
           CASA DIAZ MEXICAN KITCHEN
                                               11090 Limonite Ave
                                                                      Jarupa Valley, CA 91752
           CASH. - CLAUDIA RODRIGUEZ
                                               9632 Peridot Ave
                                                                   Hesperia, CA 92344-8017
14963058
                                                              Van Nuys, CA 91408
14963062
           CASTLE PUBLICATIONS, LTD
                                             P.O. Box 580
           CATHEDRAL CITY DOWNTOWN FOUNDATION
                                                              68700 Avenida Lalo Guerrero
14963063
                                                                                            Cathedral City, CA
           CATHEDRAL CITY FIRE DEPARTMENT
14963064
                                                      32100 Desert Vista Rd.
                                                                               Cathedral City, CA 92234
14963065
           CATHEDRAL CITY POLICE DEPT-ALARM UNIT
                                                              P.O. Box 670
                                                                               Cathedral City, CA
           92235-0670
14963067
           CATHEDRAL VILLAGE SELF STORAGE
                                                      32-500 Date Palm Drive
                                                                                 Cathedral City, CA 92234
                                                               910 16th Street, Suite 426
          CATHEDRAL VILLAGE SHOPPING CENTER, LLLP
14963068
                                                                                           Denver, CO
           80202
           CBRE, Inc
14963076
                         P.O. Box 281620
                                             Atlanta, GA 30384-1620
          CCC VERIFY 03053
14963077
                             50 Nashua Road
                                                Verifications – Corporate Cost Control
                                                                                      Londonderry, NH
           CDPH
14963078
                      850 Marina Bay Parkway, Bld P1st floor
                                                              Richmond, CA 94804
14963079
           CDPH-RHB
                           P.O. Box 997414
                                               Sacramento, CA 95899-7414
14963080
          CDW-G
                        75 Remittance Dr., Suite 1515
                                                       Chicago, IL 60675-1515
14963085
           CELLULAR OUTLET
                                   5627 LANKERSHIM BLVD
                                                                  North Hollywood, CA 91601
                                                     3980 Sherman Street, Suite 170
           CENTER FOR SUSTAINABLE ENERGY
                                                                                    San Diego, CA 92110
14963088
                                                   Borrego Springs, CA 92004
Phoenix, AZ 85072–2187
14963089
           CENTER MARKET
                                 P.O. Box 935
14963092
          CENTURY LINK
                               P.O. Box 52187
                            2100 Powell St Suite 400
14963094
           CEP America
                                                       Emeryville, CA 94608
14963095
           CERIDIAN
                          P.O. Box 772830
                                              Chicago, IL 60677-2830
           CERTIFIED SERVICE CENTER
14963097
                                             151 N. Rose St., Suite F101
                                                                          Escondido, CA 92027
           CHA MEDIATIONS SERVICES
                                                                              Beverly Hills, CA 90210
14963101
                                             9440 Santa Monica Blvd., Ste 301
           CHAD FINCH (CMF SERVICES, INC)
                                                                   La Quinta, CA 92248
14963102
                                                  P.O. Box 5634
           CHANTERELLE CATERING COMPANY
14963103
                                                     7950 Silverton Ave
                                                                           San Diego, CA 92126-6342
                                                    3288 Morgan Dr. Suite 100
14963112
           CHARTCAPTURE - Under Aesto Health
                                                                                Birmingham, AL 35216
           CHEVROLET CADILLAC OF LA QUINTA
                                                       79–225 Hwy 111
14963115
                                                                           La Quinta, CA 92253
14963116
                     7259 S. Bingham, Junction Blvd.,
                                                       Midvale, UŤ 84047
          CHG
          CHIA 44 Montgomery Street 3rd Floor
CHICAGO FAUCET SHOPPE 3844 N
CHIEF 13 EAST 19TH STREET
                                                      San Francisco, CA 94104
14963117
                                          3844 N ELSTON AVE
14963118
                                                                    Chicago, IL 60618
                                                 NEW YORK, NY 10003
14963120
14963123
           CHIPOTLE
                          610 Newport Center Dr
                                                    Newport Beach, CA 92660
           CHIROPRACTIC INTEGRATIVE PARTNERS, INC
                                                              746 S. Main Avenue, Suite D
14963124
                                                                                            Fallbrook, CA
           92028
14963126
           CHOICE MEDICAL GROUP
                                          19111 Town Center Drive
                                                                     HORIZONS VALLEY MEDICAL
          GROUP Apple Valley, CA 92308
CHOW NOW 12181 Bluff Creek Drive
14963128
                                                      Los Angeles, CA 90094
                                    Address Redacted
           CHRISTINA REAVES
14963138
           CHRISTOPHER VELARDEZ
14963152
                                          Address Redacted
14963154
           CHSB FOUNDATION
                                   P.O. Box 2637
                                                      San Bernardino, CA 92406
14963158
           CHUYITAS TACOS
                                  11726 Hazeldell Dr.
                                                         Riverside, CA 92505
                                                Bloomfield, CT 06002
14963160
          CIGNA
                       900 Cottage Grove Road
           CIGNA BEHAVIORAL HEALTH
                                              900 Cottage Grove Road
14963162
                                                                        Bloomfield, CT 06002
14963167
           CINTAS FIRE 636525
                                   P.O. Box 636525
                                                       Cincinnati, OH 45263-6525
14963169
          CIP-Chiropractic Integrative Partner
                                               746 S Main Ave Ste D
                                                                        Fallbrook, CA 92028
14963171
           CIRCLE J STORAGE
                                  P.O. Box 1173 Borrego Springs, CA 92004
                                      P.O. Box 936497
          CITRIX SYSTEMS, INC.
                                                          Atlanta, GA 31193-6497
14963175
14963177
           CITY CLERK SAN BERNARDINO
                                               P.O. Box 1318
                                                                San Bernardino, CA 92402
           CITY OF CATHEDRAL CITY
                                           68–700 Ave. Lalo Guerrero
                                                                       Cathedral City, CA 92234-7031
14963182
14963185
           CITY OF EL CAJON
                                  P.O. Box 51159
                                                      Los Angeles, CA 90051-5459
          CITY OF EL CAJON- ALARM CLERK
                                                   200 Civic Center Way
                                                                           El Cajon, CA 92020-3916
14963186
14963187
           CITY OF EL CAJON-BUSINESS LICENSE
                                                      P.O. Box 511378
                                                                           Los Angeles, CA 90051-7933
14963188
          CITY OF ESCONDIDO
                                     201 North Broadway
                                                            Escondido, CA 92025
14963190
          CITY OF ESCONDIDO FIRE DEPARTMENT
                                                        1163 N. Centre City Parkway
                                                                                       Escondido, CA
           92026
                                                                       Irving, TX 75014-2708
14963192
          CITY OF INDIO – ALARM PROGRAM
                                                   P.O. Box 142708
14963193
           CITY OF JURUPA VALLEY
                                         8930 Limonite Ave
                                                               Jurupa Valley, CA 92509
14963194
           CITY OF LA MESA
                                 8130 Allison Ave
                                                      La Mesa, CA 91942
14963195
           CITY OF LAKE ELSINORE
                                         130 South Main Street
                                                                 Lake Elsinore, CA 92530
           CITY OF PALM DESERT
                                                                  Palm Desert, CA 92260-2578
14963196
                                       73510 Fred Waring Drive
           CITY OF PALM SPRINGS
14963197
                                       P.O. Box 2743
                                                         BUSINESS LICENSE
                                                                                  Palm Springs, CA 92263
14963199
           CITY OF RIVERSIDE
                                    8839 N. Cedar Ave #212
                                                              Fresno, CA 93720
           CITY OF RIVERSIDE - PRCSD
                                            6927 Magnolia Ave
14963200
                                                                   Second Floor
                                                                                    Riverside, CA 92506
           CITY OF RIVERSIDE FIRE PREVENTION
                                                      3900 Main Street
                                                                          City Hall – 3rd Floor
14963201
                                                                                                  Riverside, CA
           92522
14963202
          CITY OF SAN BERNARDINO
                                                             Business Registration
                                           P.O. Box 1318
                                                                                     San Bernardino, CA
           92402-1318
14963203
          CITY OF SAN JACINTO
                                      P.O. Box 488
                                                       595 S. San Jacinto Avenue
                                                                                   San Jacinto, CA 92581
                              200 Civic Center Dr
14963204
          CITY OF VISTA
                                                      Vista, CA 92084
14963205
           CITY TREASURER
                                 1200 Third Ave., Suite 100
                                                              San Diego, CA 92101
          CITY TREASURER-DELINQUENT ACCT PROGRAM 92112–9039
14963206
                                                                 P.O. Box 129037
                                                                                     San Diego, CA
           CITY TREASURER-DELINQUENT ACCT PROGRAM
14963207
                                                                 P.O. Box 129039
                                                                                     San Diego, CA
           92112-9039
14963208
          CLAIREMONT EQUIPMENT
                                          7651 Ronson Road
                                                                San Diego, CA 92111-1511
          CLERK OF THE COURT
                                                      Brawley, CA 92227
14963227
                                      220 Main St
```

```
14963228
          CLEVELAND UNIVERSITY
                                        2121 Euclid Avenue
                                                               Cleveland, OH 44115-2214
14963229
          CLIA LABORATORY PROGRAM
                                                               Portland, OR 97208-3056
                                             P.O. Box 3056
          CLINICAL LABORATORY OF SAN BERNARDINO
                                                             P.O. Box 329
14963231
                                                                              San Bernardino, CA 92402
14963232
          CLOUDMED
                           P.O. Box 20872
                                              Dallas, TX 75320-8272
                                        P.O. Box 25517
14963233
          CMK PALM CANYON, LLC
                                                           Tampa, FL 33622
                                          1151 E Washington Ave Ste C
14963234
          CML Del Valley Medical Group
                                                                         Escondido, CA 92025
          CMS MEDICARE 7500 Security Boulevard Baltin
COACH SUPPLY DIRECT 69039 ELKHART ROAD
14963235
                                                         Baltimore, MD 21244
                                                                Edwardsburg, MI 49112
14963236
14963237
          COACHELLA DESERT OASIS OB
                                              2262 North Carillo Rd
                                                                      GOMEZ LOAN PYMT
                                                                                               Palm Springs,
          CA 92262
14963239
          COACHELLA VALLEY UNIFIED SCHOOL DIST.
                                                           P.O. Box 847
                                                                            Thermal, CA 92274
          COLLINS FAMILY TRUST 3215 Indian Mills Lane
COLUMBIA BOOKS 1530 Wilson Boulevard, Suite 400
                                                                Jamul, CA 91935
14963243
                                            on Boulevard, Suite 400 Arlington, VA 22209
10368 ALDER CREEK LN Riverside CA 9
14963246
14963247
          COMMERCIAL DOOR MEDICS
                                                                        Riverside, CA 92505
          COMMERCIAL LIGHTING INDUSTRIES, INC.
                                                         81161 Indio Boulevard
14963248
                                                                                 Indio, CA 92201
14963249
          COMMERCIAL PROTECTIVE SERVICES, INC
                                                         3400 E. AIRPORT WAY
                                                                                   LONG BEACH, CA
          90806
14963252
          COMMUNITY CONTROL
                                       2101 N Tustin Ave
                                                            Santa Ana, CA 92705
          COMMUNITY HEALTH ACTION NETWORK (CHAN)
                                                                15000 7th Street, Ste 208-G
14963253
                                                                                             Victorville, CA
14963254
          COMMUNITY HEALTH ASSOC INLAND S REGION
                                                              621 E. Carnegie Drive, Suite 180
                                                                                               San
          Bernardino, CA 92408
14963261
          COMPENSATION RESOURCES, INC.
                                                 310 Route 17 North
                                                                       Upper Saddle River, NJ 07458
14963263
          COMPHEALTH
                             P.O. Box 972651
                                                 Dallas, TX 75397-2651
14963264
          COMPLIANCEIO.COM
                                  2345 Yale Street, 1st Floor
                                                                Palo Alto, CA 94306
          COMPLIANCESIGNS.COM
14963265
                                        16228 Flight Path Drive
                                                                 ComplianceSigns, LLC
          34604
          COMPUTACENTER UNITED STATES, INC.
14963266
                                                       P.O. Box 847403
                                                                          Los Angeles, CA 90084-7403
          CONNECT CONSULTING SERVICES
                                                 1510 J Street Suite 115
14963270
                                                                          Sacramento, CA 95814
          CONNECTANZA ANZA ELECTRIC COOPERATIVE
                                                              P.O. Box 391909
14963271
                                                                                  Anza, CA 92539-1909
                                 LP P.O. Box 12920
95 Corporate Drive
          COOK STREET OFFICE, LP
                                                           Palm Desert, CA 92255
14963280
          COOPER SURGICAL
14963281
                                                        Trumbull, CT 06611
          CORDOVA PHOTOGRAPHY 44489 Town Center Way, D456
14963282
                                                                         Palm Desert, CA 92260
14963285
          CORINTECH LIMITED
                                   118-122 Station Rd
                                                          Ashford Mill
                                                                           Fordingbridge, SP6 1DZ United
          Kingdom
          CORNERS PLAZA LLC
14963286
                                    124 West Main Street, Ste 240
                                                                   El Cajon, CA 92020
          CORODATA MEDIA STORAGE, INC
14963287
                                                P.O. BOX 846143
                                                                     LOS ANGELES, CA 90084-6143
                                                      P.O. Box 842638
                                                                            Los Angeles, CA 90084-2638
          CORODATA RECORDS MANAGEMENT, INC
14963288
14963289
          CORODATA RECORDS MANAGEMENTS-016589
                                                            P.O. Box 842638
                                                                                Los Angeles, CA
          90084-2638
14963290
          CORODATA RECORDS MANAGMENT, INC.-049204
                                                               P.O. Box 842638
                                                                                   Los Angeles, CA
          90084-2638
14963291
          CORONA TRANSGENDER PROJECT
                                                 P.O. Box 6103
                                                                   Corona, CA 92878
          CORONIS HEALTH LLC
                                     P.O. Box 790372
14963292
                                                         St. Louis, MO 63179-0372
                                             Chula Vista, CA 91910
                               825 E St
14963294
          COSTA VISTA RV
          COULTER COMMUNICATIONS
14963295
                                            4177 Manchester Ave.
                                                                     Encinitas, CA 92024
                                     10704 Prospect Ave., Suite D
14963297
          COUNTERTOPS 4 LESS
                                                                   Santee, CA 92071
14963298
          COUNTRY CLUB SELF STORAGE & RV
                                                  749 N. Main St. Riverside, CA 92501
                                     7888 Mission Grove Parkway, Suite 201
P.O. Box 751 Riverside, CA 92502
          COUNTY OF RIV. MISP
14963300
                                                                            Riverside, CA 92508
          COUNTY OF RIVERSIDE
14963301
          COUNTY OF RIVERSIDE ASSESSOR P.O. Box 751
                                                                  Riverside, CA 92502-0751
14963302
14963303
          COUNTY OF RIVERSIDE DEH
                                          P.O. Box 7909
                                                           Department of Environmental Health
                                                                                                 Riverside,
          CA 92513
14963305
          COUNTY OF SAN BERNARDINO
                                             150 S. Lena Road
                                                                  San Bernardino, CA 92415–0515
          COUNTY OF SAN BERNARDINO – PUBLIC HEALTH
                                                                385 N Arrowheald Ave, 2nd Floor
14963306
                                                                                                  San
          Bernardino, CA 92415
          COUNTY OF SAN DIEGO DEH
                                                               San Diego, CA 92112-9261
14963307
                                           P.O. Box 129261
14963308
          COUNTY OF SAN DIEGO RECORDER / COUN P.O. Box 121750
                                                                            1600 Pacific Highway, Suite
                   San Diego, CA 92112–1750
14963309
          COURT-ORDERED DEBT COLLECTIONS
                                                    P.O. BOX 1328, FRANCHISE TAX BOARD
          MUNOZ, GABRIELA
                                 RANCHO CORDOVA, CA 95741-1328
          COVE ELECTRIC, INC.
                                    77–971 Wilcat Drive, Suite F Palm Desert, CA 92211
14963311
                                                 5210 E. Williams Circle, Suite 220
          COVENTRY / FIRST HEALTH – CCN
14963312
                                                                                   Tucson, AZ 85711
                                                                    Phoenix, AZ 85072
14963313
          COX BUSINESS - 001 3110 128556001
                                                  P.O. Box 53214
          COX COMMUNICATIONS
                                                          Phoenix, AZ 85072-3214
14963316
                                       P.O. Box 53214
          COX COMMUNICATIONS – 133 EL CAJON
                                                      P.O. Box 53214
14963317
                                                                         Phoenix, AZ 85072-3214
                         4749 E. Wesley Drive
          CPAC INC
                                               Anaheim, CA 92807
14963318
          CPACINC.Com
                             4749 E. Wesley Drive
14963319
                                                     Anaheim, CA 92807
                                             Sacramento, CA 95814
14963320
                     1231 I Street, Suite 400
          CPCA
          CR&R INCORPORATED
14963322
                                     P.O. Box 7183
                                                       Pasadena, CA 91109-7183
                                                     P.O. Box 7183
                                                                     Pasadena, CA 91109-7183
14963323
          CR&R INCORPORATED – BH-0000158 2
          CR&R INCORPORATED - CHL CTR
                                                P.O. Box 7183
                                                                  Pasadena, CA 91109-7183
14963324
14963325
          CR&R INCORPORATED- BH-00006318
                                                  P.O. Box 7183
                                                                     Pasadena, CA 91109-7183
          CR&R Incorporated 11292 Western Avenue Stanton, CA 94108
CRAIGSLIST 222 Sutter St, 9th Floor San Francisco, CA 94108
14963321
14963328
                                             2201 Francisco Drive, Suite 140–183
          CREATIVE MEDIA GROUP, LLC
                                                                                 El Dorado Hills, CA
14963329
          95762
          CREATIVE SAFETY SUPPLY
                                        8030 SW NIMBUS AVE
14963330
                                                                    Beaverton, OR 97008
```

```
14963331
           CREST CHEVROLET
                                     909 W 21st St
                                                       San Bernardino, CA 92405
           CRG Financial LLC
                                   84 Herbert Ave, Building B – Suite 202
15002661
                                                                          Closter, NJ 07624
14963335
           CRUCIAL.COM
                                P.O. Box 6, 8000 South Federal Way
                                                                       Micron Technology Inc
                                                                                                 Boise, ID
           83716
14963344
           CSU San Bernardino
                                   5500 University Parkway
                                                                San Bernardino, CA 92407–2393
                                                                  Riverside, CA 92503
14963345
           CT COACHWORKS, LLC
                                         9700 INDIANA AVE,
14963346
           CTC CONSTANT
                                  11801 Mississippi Avenue, Suite 200
                                                                        Los Angeles, CA 90025
                                     3133 Mission Inn Avenue
                                                                 ECONOMIC DEVELOPMENT
14963347
           CTY OF RIVERSIDE
           AGENCY
                          Riverside, CA 92507
           CTY OF RIVERSIDE
14963348
                                     4080 Lemon Street, 8th Floor
                                                                     Business Stormwater Registration
                                                                                                        Riverside.
           CA 92501
14963349
           CTY OF RIVERSIDE BUILDING & SAFETY DEPT
                                                                 4080 Lemon St 9th Floor
                                                                                             Riverside, CA
           92502
14963350
           CUBICLE CURTAIN FACTORY
                                               7810 S DIXIE HWY
                                                                        WEST PALM BEACH, FL 33405-4820
14963352
                                             P.O. Box 2903
           CULLIGAN OF ESCONDIDO
                                                                Wichita, KS 67201-2903
14963353
           CULLIGAN OF ONTARIO
                                          P.O. Box 2903
                                                             Wichita, KS 67201-2903
14963354
           CULLIGAN OF SAN DIEGO
                                                               Wichita, KS 67201-2903
                                            P.O. Box 2903
14963355
           CUMMINS INC
                               P.O. BOX 848731
                                                     LOS ANGELES, CA 90084-8731
           CUNNINGHAM LAW, APC CLIENT TRUST
                                                                                          Temecula, CA 92590
14963356
                                                           1 Ridgegate Drive, Suite 205
14963357
           CURT DEGENFELDER CONSULTING, INC
                                                          2536 Granville Avenue
                                                                                     Los Angeles, CA 90064
14963358
                                575 Sir Francis Drake Boulevard, Suite 1
                                                                          Greenbrae, CA 94904
           CURTIS CRANE
           CURVATURE, INC.
14963359
                                    14416 Collections Ctr. Dr.
                                                                 Chicago, IL 60693
14963361
           CV COMMERCIAL TRUCK REPAIR INC.
                                                        P.O. Box 1504
                                                                           Indio, CA 92202
                                                       Woonsocket, RI 02895
14963362
           CVS Pharmacy, Inc.
                                   One CVS Drive
14963373
           CYNTHIA MARCHANT D.B.A. ANZA VALLEY AUTO
                                                                     Address Redacted
14963376
           CYNTHIA PRECIADO
                                      Address Redacted
                                              333 S. Twin Oaks Valley Rd.
                                                                             San Marcos, CA 92096-0001
14962957
           Cal State University San Marcos
           CalHIPSO Service Partner Program
14962959
                                                3053 Freeport Blvd 156
                                                                           Sacramento, CA 95818
                             1201 K Street, Suite 800
14962994
           CalMed Force
                                                         Sacramento, CA 95814
14962962
           California Attorney General
                                          1300 I Street
                                                           Sacramento, CA 95814
                                          8432 Magnolia Avenue
           California Baptist University
14962963
                                                                    Riverside, CA 92504
                                             1515 Člay Street, Suite 1302
14962968
           California Department of Labor
                                                                            Oakland, CA 94612
14962969
           California Dept Health Care SVCS (DHCS)
                                                       P.O. Box 997413, MS 0010
                                                                                     c/o Jon-Paul
                           Sacramento, CA 95899
           Valcarenghi
14962973
           California Deptartment of Public Health
                                                    P.O. Box 997377
                                                                         SACRAMENTO, CA 95899-7377
14962974
           California Deptof Health Care Services
                                                   P.O. Box 997413
                                                                                      Sacramento, CA 95899
                                                                        MS 0000
14962977
           California Health and Human Services
                                                   1215 O Street
                                                                     Sacramento, CA 95814
14962981
           California Medical Behavioral Health
                                                  1800 N Western Ave #305
                                                                               San Bernardino, CA 92411
14962982
           California Nurses Educational Insitute
                                                   5200 E. Ramon Road BLD STE 1
                                                                                      Palm Springs, CA 92264
           California Secretary of State
14962985
                                          1500 11th Street
                                                              Sacramento, CA 95814
14962987
           California Special Patrol
                                      8350 Archibald Ave Ste 200
                                                                     Rancho Cucamonga, CA 91730
14962988
           California State Controller
                                                              Sacramento, CA 94250-5873
                                        P.O. Box 94285B
14962989
           California State University Long Beach
                                                    1250 Bellflower Blvd
                                                                             Long Beach, CA 90840-5801
14962992
           Call 4 Health / Call 4 Nurse
                                         2855 S. Congress Ave, Suites A & B
                                                                               Delray Beach, FL 33445
14962993
           Call 4 Health-Over the Phone Triage Co.
                                                     2855 S. Congress Avenue Suite A & B
                                                                                             Delray Beach, FL
           33445
14962995
           Calpers
                        P.O. Box 70000
                                            Van Nuys, CA 91470
14963001
           Cardinal – 78074 – 78570 – 78075 – 78873
                                                       PARKVIEW MED PLAZA PHY 340B ST
                                                                                                   3975 JACKSON
                        Riverside, CA 92503
           ST #109
                                            1666 MEDICAL CENTER DR. STE. 1
14963003
           Cardinal Delta Drugs - 76359
                                                                                    SAN BERNARDINO, CA
           92411
14963005
           Cardinal Health
                               7000 Cardinal Place
                                                       Steve Falk
                                                                     Dublin, OH 43017
14963013
                               7000 Cardinal Place
           CardinalHealth
                                                      Dublin, OH 43017
                                                    Tullytown, PA 19007-6108
14963015
           Care Organix
                             100 Main St Ste 1G
14963020
                             Address Redacted
           Carina Anson
14963021
           Carina Cuevas
                              Address Redacted
                            Address Redacted
14963022
           Carina Elias
14963023
           Carina Flores
                             Address Redacted
14963024
           Carina Ordas
                             Address Redacted
14963026
           Carium, Inc.
                            2020 Carium 201
                                                 Petaluma, CA 94952
14963027
           Carla Chavez
                             Address Redacted
14963028
           Carla Chavez
                             Address Redacted
14963029
           Carla Tyndall
                             Address Redacted
14963031
           Carlina Torres
                              Address Redacted
14963032
           Carlo Garibaldi
                               Address Redacted
14963033
           Carlos Cruz
                            Address Redacted
14963034
           Carlos H Morales
                                 Address Redacted
14963035
                             Address Redacted
           Carlos Huizar
                             Address Redacted
14963036
           Carlos Lucas
14963037
                              Address Redacted
           Carlos Moreno
14963038
           Carlos Sv
                         Address Redacted
           Carmelo Pizano
14963039
                               Address Redacted
14963040
           Carmelo Ruiz
                             Address Redacted
14963041
           Carmen Diaz
                             Address Redacted
14963042
           Carmen Estrada
                               Address Redacted
14963043
                                 Address Redacted
           Carmen Garibaldi
14963044
           Carmen Navarro
                                Address Redacted
14963045
           Carmen Ochoa
                              Address Redacted
```

```
14963046
            Carmen Ruiz
                              Address Redacted
14963047
            Carmen Ruiz
                              Address Redacted
14963048
            Carmen Tamez
                                Address Redacted
14963050
            Carolina Cazeau
                                 Address Redacted
14963051
            Carolina Jimenez
                                  Address Redacted
14963052
            Carolina Ramirez
                                  Address Redacted
14963053
            Caroline Dale
                               Address Redacted
14963054
            Caroline Schurter
                                  Address Redacted
14963055
            Carolyn Jensen-Francis
                                        Address Redacted
                                    7 Ashbrook Lane
14963057
            CaseNetwork, LLC
                                                          Newtown Square, PA 19073
14963059
            Cass Information Systems
                                          P.O. Box 271172
                                                                c/o Change Healthcare
                                                                                           Salt Lake City, UT 84127
14963060
            Cassandra Boren
                                  Address Redacted
14963061
            Cassandra Faiello
                                  Address Redacted
                                                                Denver, CO 80204
14963066
            Cathedral City Shopping Center
                                               910 16th St
                                     Address Redacted
14963069
            Catherine Coucoules
14963070
            Catherine Koreski
                                   Address Redacted
14963071
                                  Address Redacted
            Catherine Roland
14963072
            Cathleen Shaffer
                                 Address Redacted
14963073
            Cathlene Thompson
                                    Address Redacted
14963074
            Cathy Aguilar
                               Address Redacted
14963075
            Caylin McKiel
                                Address Redacted
14963081
            Cecilia Iacoviello
                                  Address Redacted
            Cecilia Murillo
14963082
                                Address Redacted
14963083
            Celeste Miranda
                                 Address Redacted
14963084
            Celina Uribe
                              Address Redacted
14963086
            Center for Care Innovations
                                            1438 Webster Street, Suite 101
                                                                              Oakland, CA 94612
14963087
            Center for Community Solutions
                                                4508 Mission Bay Dr
                                                                          San Diego, CA 92109
14963090
            Centers for Medicare & Medicaid Services
                                                         7500 Security Boulevard
                                                                                      Baltimore, MD 21244
14963091
            Centro Medico
                                10368 Arlington Ave
                                                         Riverside, CA 92505
                                                        Denver, CO 80202
14963093
            CenturyLink
                              1801 California Street
14963096
            Ceridian HCM, INC.
                                     3311 East Old Shakopee Road
                                                                       Minneapolis, MN 55425-1640
14963098
            Cesar Acuna
                              Address Redacted
                              Address Redacted
14963099
            Cesar Garcia
14963100
            Cesar Vasquez
                                Address Redacted
14963104
            Chapcare
                          455 W Montana St
                                                  Pasadena, CA 91103
14963105
            Charita Robinson
                                  Address Redacted
14963106
            Charlene Avalos
                                 Address Redacted
14963107
            Charles Collum
                                 Address Redacted
14963108
            Charles Fraley
                               Address Redacted
14963109
            Charlesetta Smith
                                  Address Redacted
14963110
            Charlton Byun
                                Address Redacted
14963111
            Chartavia Francis
                                  Address Redacted
14963113
            Cherilyn Mones
                                 Address Redacted
14963114
            Chervl Welch
                               Address Redacted
14963119
            Chidi Duru
                             Address Redacted
14963121
            Childrens Physicians Medical Group
                                                    P.O. Box 23076
                                                                         San Diego, CA 92193
14963122
            Chinyere Opara
                                Address Redacted
            Chochezi Garcia-Griggs
                                         Address Redacted
14963125
14963127
            Choice Physicians Network
                                           19111 Town Center Drive
                                                                         Choice Medical Group
                                                                                                     Apple Valley, CA
            92308
14963129
            Chris Glenn
                             Address Redacted
14963130
            Christen Cantrell
                                  Address Redacted
14963131
                                  Address Redacted
            Christina Aguilar
14963132
            Christina Felipe
                                 Address Redacted
14963133
                                 Address Redacted
            Christina Fierros
14963134
            Christina Hedrick
                                  Address Redacted
                                     Address Redacted
14963135
            Christina Hernandez
14963136
            Christina Kordi
                                Address Redacted
14963137
            Christina OHare
                                 Address Redacted
14963139
            Christine Aguon Amos
                                       Address Redacted
14963140
            Christine Au
                              Address Redacted
14963141
            Christine Bielawa
                                  Address Redacted
14963142
            Christine Flores
                                 Address Redacted
14963143
            Christine Marquez
                                   Address Redacted
14963144
            Christine Shaker
                                 Address Redacted
14963145
            Christine Tellis
                                Address Redacted
14963146
            Christopher Cortez
                                    Address Redacted
            Christopher Day
14963147
                                 Address Redacted
14963148
            Christopher Fairfield
                                     Address Redacted
14963149
            Christopher Mariscal
                                     Address Redacted
            Christopher Ramirez
14963150
                                     Address Redacted
            Christopher Swanson
14963151
                                     Address Redacted
            Christy Kerr
Chuck Kimball
14963153
                              Address Redacted
14963155
                                Address Redacted
14963156
            Chuck Kimball
                                Address Redacted
14963157
            Chutney Washington
                                     Address Redacted
            Chw, LLP
14963159
                            7797 N. First Street #15
                                                        Fresno, CA 93720
                              P.O. Box 952366
                                                     St Louis, MO 63195
14963161
            Cigna Accent
```

```
14963163
           Cigna Health Care Proclaim
                                           P.O. Box 542007
                                                                 9140 West Dodge Road
                                                                                             Omaha, NE
            68154-8007
14963164
           Cindy Meza
                             Address Redacted
14963165
           Cindy Nakhla
                              Address Redacted
14963166
           Cindy Patereau
                                Address Redacted
14963168
           Cinthia Garnica
                                Address Redacted
14963170
           Cipriano Martinez
                                  Address Redacted
14963172
           Ciria Olivas
                             Address Redacted
14963173
           Cirilo Ramirez
                               Address Redacted
14963174
           Citrix Systems
                               851 Cypress Creek Road
                                                            Fort Lauderdale, FL 33309
14963176
           Citrix XenApp
                               851 Cypress Creek Road
                                                            Fort Lauderdale, FL 33309
                                    3484 Central Avenue
14963178
           City National Bank
                                                             Priscilla Lopez, VP, Account Officer
                                                                                                     Riverside, CA
            92506
14963179
           City of Adelanto
                                 11600 Air Expressway
                                                            Adelanto, CA 92301
14963180
           City of Banning
                                P.O. Box 998
                                                   99 E Ramsey St
                                                                       Banning, CA 92220
14963181
           City of Barstow
                                220 E Mountain View St.
                                                              Barstow, CA 92311
           City of Coachella
14963183
                                  1515 Sixth Street
                                                       Coachella, CA 92236
14963184
           City of Desert Hot Springs
                                          65950 Pierson Blvd.
                                                                  Desert Hot Springs, CA 92240
                                                 P.O. Box 460009
14963189
           City of Escondido – Utility Billing
                                                                       Escondido, CA 92046-0009
14963191
           City of Indio
                             100 Civic Center Mall
                                                        Indio, CA 92201
14963198
           City of Rialto
                              150 S. Palm Ave
                                                    Rialto, CA 92376
           Clara A Sanchez
14963209
                                 Address Redacted
14963210
           Claridilis Bencosme De Ruiz
                                            Address Redacted
14963211
           Clarisse Valencia
                                  Address Redacted
14963212
           Classic Auto Restyling
                                       67360 Sarah St
                                                           Cathedral City, CA 92234
14963213
           Claudia Aguayo
                                 Address Redacted
14963214
           Claudia Alvarez
                                Address Redacted
14963215
           Claudia Barron
                                Address Redacted
           Claudia Felipe-Villegas
14963216
                                        Address Redacted
14963217
           Claudia Magallanes
                                    Address Redacted
14963218
           Claudia Nava
                              Address Redacted
14963219
           Claudia Padilla
                                Address Redacted
14963220
           Claudia Rodriguez
                                   Address Redacted
14963221
           Claudia Rosas Garcia
                                     Address Redacted
14963222
           Claudia Salas
                              Address Redacted
14963223
           Claudian Barron
                                 P.O. Box 1517
                                                    Borrego Springs, CA 92004
14963224
           Claudine Castro
                                Address Redacted
14963225
           Clayton Barbour
                                 Address Redacted
14963226
           Cleaning Equipment Direct.
                                           4007 Richards Rd.
                                                                  North Little Rock, AR 72117
14963230
                             Address Redacted
           Clifford Mix
           Coachella Valley Housing Coalition
14963238
                                                   45701 Monroe Street, Suite G
                                                                                    Indio, CA 92201
14963240
           Coachella Valley Unified School District
                                                       87225 Church St
                                                                            Thermal, CA 92274
14963241
           Colleen Jones
                              Address Redacted
           College of the Desert
                                     43500 Monterey Ave
14963242
                                                              Palm Desert, CA 92260
14963244
           Colonial Life Insurance Company
                                                1200 Colonial Life Blvd.
                                                                             Colombia, SC 29210
14963245
           Colony Palms Hotel
                                    572 N Indian Canyon Dr
                                                                Palm Springs, CA 92262
14963260
           CommVault
                             1 Commvault Way
                                                    Tinton Falls, NJ 07724
14963250
           Community Care IPA
                                    5000 Airport Plaza Dr Suite 150
                                                                         Long Beach, CA 90815-1260
           Community Care IPA INC.
                                                                  Tarzana, CA 91357
14963251
                                          P.O. Box 7020-04
                                         2420 Fenton St Ste 100
                                                                     Chula Vista, CA 91921
14963255
           Community Health Group
                                                              Chula Vista, CA 91914
14963256
           Community Health Partner
                                          2420 Fenton St
14963257
            Community Hospital of San Bernardino 1805 Medical Center Dr
                                                                                  San Bernardino, CA 92411
           Community Med Ctr Phy – 340B ST – 78965
                                                           610 Gateway Center Way Ste A
14963258
                                                                                               San Diego, CA
            92102
14963259
           Community Valley Bank (CVB)
                                               571 A Palm Canyon Drive
                                                                             Corina Garcia, Operations
                         Borrego Springs, CA 92004
7259 S. Bingham Jct. Blvd
           Manager
                                                            Midvale, UT 84047
14963262
           CompHealth
14963267
           Concilio Child Development Centers
                                                   2010 Martin Luther King Luther Blvd
                                                                                            Riverside, CA 92507
                                         601 108th Ave NE, Suite 1000
14963268
           Concur Technologies Înc.
                                                                           Bellevue, WA 98004
14963269
           Conduent, INC
                                820 Stillwater Road
                                                        MS 06
                                                                    West Sacramento, CA 95605
14963272
           Conrado Martinez
                                  Address Redacted
14963273
           Consejosano Inc.
                                 5200 Lankershim Blvd, Ste 310
                                                                    SameSky Health
                                                                                         North Hollywood, CA
           91601
14963274
           Consuelo Arellano
                                   Address Redacted
14963275
           Consuelo Delgado
                                   Address Redacted
           Consuelo Jauregui
14963276
                                   Address Redacted
14963277
           Consuelo Jimenez
                                  Address Redacted
14963278
           ContextMedia Health
                                     330 North Wabash Avenue, Suite 2500
                                                                               Chicago, IL 60611
14963279
                                4880 Stevens Creek Blvd. Suite 204
           ControlUp, Inc.
                                                                       San Jose, CA 95129
15002662
           Cook Street Office, LP
                                       P.O. Box 12920
                                                           Palm Desert, CA 92255
           Corina Velasquez
                                  Address Redacted
14963283
14963284
           Corinne Blackwell
                                   Address Redacted
14963293
           Corwin Woodward
                                   Address Redacted
14963296
           Council of Community Clinics
                                             3710 Ruffin Rd
                                                                  Health Center Partners of San Diego
                                                                                                         San Diego, CA
            92123
14963299
           County Medical Services – San Diego
                                                    1600 Pacific Highway, Room 206
                                                                                         Health and Human Services
            Agency
                        San Diego, CA 92101
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 17 of 76

```
14963304
           County of Riverside Deptof Public Health
                                                     4065 County Circle Drive
                                                                                 Health Administration
                        Riverside, CA 92503
           Building
                                                                                 Brea, CA 92821-4370
14963310
           Coury & Buehler Physical Therapy
                                               3230 E. Imperial Hwy, Suite 100
                                                                    Phoenix, AZ 85072-3214
           Cox Business - 001 3110 130126501
14963314
                                                P.O. Box 53214
14963315
           Cox Communications
                                    6205-B Peachtree Dunwoody Road NE
                                                                            Atlanta, GA 30328
14963326
                                 1111 W Jefferson Street
           CradlePoint Cloud
                                                           Boise, ID 83702-5389
14963327
           Craig Steinberg
                              Address Redacted
14963332
           Cristal Gonzalez
                               Address Redacted
14963333
           Cristian Davila
                              Address Redacted
14963334
           Crown Dental Group
                                   2405 Transportation Avenue
                                                                 National City, CA 91950
14963336
           Crystal Ball Room
                                 414 N Magnolia Ave
                                                         El Cajon, CA 92020
14963337
           Crystal Cortez
                             Address Redacted
14963338
           Crystal Erickson
                               Address Redacted
14963339
           Crystal Garcia
                             Address Redacted
                                 Address Redacted
14963340
           Crystal Hernandez
14963341
           Crystal Knowlton
                                Address Redacted
14963342
           Crystal Valenzuela
                                 Address Redacted
14963343
           Crystal Westgate
                               Address Redacted
                                              7040 Avenida Encinas
           Cubicles Office Environments Inc.
14963351
                                                                        Carlsbad, CA 92011
14963360
           Customer Contact Services
                                        14525 Highway 7, Suite 315
                                                                       Minneapolis, MN 55345
14963363
                                     Address Redacted
           Cyndi Herrera Villalba
14963364
           Cynthia Amezcua
                                Address Redacted
14963365
           Cynthia Arzeta
                              Address Redacted
           Cynthia Bainbridge
14963366
                                  Address Redacted
14963367
           Cynthia Benitez
                               Address Redacted
14963368
           Cynthia Brigham
                                Address Redacted
           Cynthia Cervantes
14963369
                                 Address Redacted
14963370
           Cynthia Garcia
                              Address Redacted
14963371
           Cynthia Gonzalez
                                Address Redacted
14963372
           Cynthia Lopez
                              Address Redacted
           Cynthia Mauricio Solano
14963374
                                      Address Redacted
14963375
           Cynthia Morales
                               Address Redacted
           Cynthia Rivera
14963377
                              Address Redacted
14963378
           Cynthia Rocha
                              Address Redacted
           Cynthia Valencia
14963379
                                Address Redacted
           Cynthia Valle
14963380
                             Address Redacted
14963381
           Cynthia Vega
                             Address Redacted
14963382
           Cynthia Zavala
                              Address Redacted
14963383
           Czar Sacramento
                                Address Redacted
14963384
                                  915 E First Street
                                                      Los Angeles, CA 90012
           DAILY JOURNAL
14963385
           DAILY PRESS
                              13891 Park Avenue
                                                     Victorville, CA 92393
14963396
           DALE L HENDERSON
                                      Address Redacted
14963400
           DANA ERWIN
                              Address Redacted
14963405
           DANIEL CAMP
                               Address Redacted
14963408
           DANIEL E ZELAC
                                  Address Redacted
           DANIEL, MARTHA
DANIELS TIRE SERVICE
14963413
                                   P.O. Box 854
                                                     Borrego Springs, CA 92004
                                                             Indio, CA 92201-6655
14963419
                                         81943 HWY 111
           DANSEREAU DENTAL EQUIPMENT
14963420
                                                    1581 COMMERCE ST
                                                                             Corona, CA 92880
           DAVE BALDWIN
14963429
                                 Address Redacted
                                 OFFICE OF THE UNITED STATES TRUSTEE
14967694
           DAVID A. ORTIZ
                                                                                 880 FRONT STREET, SUITE
                     SAN DIEGO, CA 92101
           3230
14963440
           DAVIS WRIGHT TREMAIN, LLP
                                                1201 Third Avenue, Suite 2200
                                                                                Seattle, WA 98101-3045
14963442
           DAYS INN
                           1501 Carmelo Dr
                                                Oceanside, CA 92054
14963443
           DCS - DEAF COMMUNITY SERVICES
                                                     1545 Hotel Circle South, Suite 300
                                                                                         San Diego, CA
           92108
14963444
           DE ANZA READY MIX & MATERIALS, INC.
                                                           P.O. Box 758
                                                                            Borrego Springs, CA 92004
14963445
                                                                             Philadelphia, PA 19101-1602
           DE LAGE LANDEN 25010151-DESERT OA
                                                         P.O. Box 41602
           DE LAGE LANDEN 25247098- DESERT OASIS
14963446
                                                             P.O. Box 41602
                                                                                 Philadelphia, PA 19101–1602
14963447
           DEALS FOR PRIME, INC
                                                            Pembroke, MA 02359
                                        32 Riverside Dr
14963448
           DEAN MOFIDI INSURANCE SERVICES, LLC
                                                            4633 East Ramon Road
                                                                                     Palm Springs, CA 92264
14963459
           DECKED STORAGE
                                    P.O. Box 885
                                                     Ketchum, ID 83340
14963460
           DECOR SHOPPE
                                P.O. Box 506
                                                  587 Palm Canyon Drive
                                                                            Borrego Springs, CA 92004
14963461
           DEHARTE GROUP
                                  329 W. Mariscal Rd
                                                          Palm Springs, CA 92262
14963464
           DELL MARKETING LP
                                       P.O. Box 910916
                                                           Pasadena, CA 91110-0916
14963467
           DELTA AIR LINES
                                   30320, 1020 Cargo Service Rd
                                                                   Atlanta, GA 30337
14963470
                                              Cincinnati, OH 45274-2572
           DELUXE
                         P.O. Box 742572
14963487
           DENNIS VANBAALE
                                     Address Redacted
           DENOVO
14963490
                          6400 Lookout Road, Suite 101
                                                          Boulder, CO 80301
14963491
           DENTAL BOARD
                                 1625 N MARKET BLVD
                                                             Sacramento, CA 95834
14963492
           DENTSPLY SIRONA, INC
                                         P.O. Box 31001-1205
                                                                  DEPT. TUL
                                                                                  Pasadena, CA 91110-1205
14963495
           DEPARTMENT OF INDUSTRIAL RELATIONS
                                                            P.O. Box 511266
                                                                                 Los Angeles, CA 90051-7821
14963496
           DEPARTMENT OF MOTOR VEHICLES
                                                     P.O. Box 825339
                                                                          Sacramento, CA 94232-5339
14963497
           DEPARTMENT OF SOCIAL SERVICES
                                                     P.O. Box 944243
                                                                          MS 9-3-67
                                                                                          Sacramento, CA
           94244-2430
14963498
           DEPARTMENT OF THE TREASURY
                                                   Internal Revenue Service
                                                                              Ogden, UT 84201-0009
14963500
           DEPARTMENT OF THE TREASURY - SHERYL DREW
                                                                    P.O. BOX 24017
                                                                                         ACS
           SUPPORT
                          FRESNO, CA 93779-4017
```

```
14963501
           DEPARTMENT OF TOXIC SUBSTANCES CONTROL
                                                                  P.O. Box 1288
                                                                                     Accounting
                    Sacramento, CA 95812-1288
14963504
           DEPT OF HOUSING & COMMUNITY DEVELOPMENT
                                                                    P.O. Box 26060
                                                                                        Registration and Tilting
                        Sacramento, CA 95826-0060
           Program
14963505
           DEPT OF MOTOR VEHICLES
                                            P.O. Box 942897
                                                                 Sacramento, CA 94297-0899
14963510
                                                                        Cathedral City, CA 92234
           DESERT CITY GLASS, INC
                                          68743 Perez Rd., Ste. D-36
14963512
           DESERT DAYLIGHT LODGE 873
                                                83610 Leeds Ct
                                                                   Indio, CA 92203
14963513
           DESERT ELECTRIC SUPPLY
                                            4605 E. Sunny Dunes Road
                                                                          Palm Springs, CA 92264
14963514
           DESERT FIRE EXTINGUISHER CO., INC.
                                                       P.O. Box 1607
                                                                          Palm Springs, CA 92263–1607
                                                                  1140 N. Indian Canyon Drive
           DESERT HEALTHCARE DISTRICT & FOUNDATION
14963515
                                                                                                  Palm Springs,
           CA 92262
                                                    Borrego Springs, CA 92004–2532
68–915 Vista Chino C
14963522
           DESERT PANTRY
                                  P.O. Box 2128
           DESERT PROMOTIONAL & EMBROIDERY
14963523
                                                                                 Cathedral, CA 92234
14963524
           DESERT REGIONAL MEDICAL CENTER
                                                        1150 N. Indian Canyon Dr.
                                                                                     MEDICAL STAFF
           OFFICE
                        Palm Springs, CA 92262
           DESERT REGIONAL MEDICAL CENTER
DESERT SUNRISE LANDSCAPE AND DESIGN
14963525
                                                       1150 N. Indian Canyon Dr.
                                                                                     Palm Springs, CA 92262
                                                             P.O. Box 580495
14963527
                                                                                  North Palm Springs, CA
                                                  P.O. Box 2720
14963529
           DESERT VALLEY DISPOSAL, INC.
                                                                     Palm Springs, CA 92263-2720
           DESERT VIP URGENT CARE DESERT WIRELESS SYSTEMS
14963530
                                            72630 Fred Waring Dr. Ste. 101
                                                                              Palm Desert, CA 92260-5004
                                              41175 Stimson Court
                                                                      Indio, CA 92203
14963531
14963532
           DESERTWRAPS.COM
                                      74839 42ND AVE SUITE D
                                                                    Palm Desert, CA 92260
14963540
                      1501 Capitol Avenue
                                               MS 4100, Suite 71.3052
                                                                         Sacramento, CA 95814
           DHCS
                                                                                 Washington, DC 20201
14963541
           DHHS - 330 Grant
                                330 Independence Ave., S.W.
                                                                 Room 5309
14963542
           DHS Health & Wellness
                                      11750 Cholla Drive, Suite B
                                                                    Desert Hot Springs, CA 92240
14963544
           DIAMOND VALLEY RV
                                        21462 HARVILL AVE STE 2A
                                                                         Perris, CA 92570
14963545
           DIAMONDS LIMOUSINE
                                        8130 Old Leonardtown Road
                                                                       Hughesville, MD 20637
14963567
           DIBS SAFE AND LOCK SERVICE
                                                342 W. 6th Street
                                                                     San Bernardino, CA 92401
14963570
           DIGITAL RIVER
                                10380 Bren Road West
                                                          Minnetonka, MN 55343
                            18124 Wedge Parkway, #1073
                                                            Reno, NV 89511
14963572
           DIIGO, INC.
14963574
                            2230 E. Imperial Hwy
                                                    El Segundo, CA 90245
           DIRECT TV
14963575
           DIRECTV, LLC
                               2260 East Imperial Highway
                                                             El Segundo, CA 90245
                                    400 Hwy 169 S Ste 300 M
GE 1620 E Miraloma Ave
14963577
           DISC PROFILE.COM
                                                              Minneapolis, MN 55426
14963578
           DISCOUNT LOW VOLTAGE
                                                                     Placentia, CA 92870
           DISCOUNT MUGS 12610
DISCOUNT SCHOOL SUPPLY
14963579
                                  12610 NW 115TH AVE
                                                             Miami, FL 33178
14963580
                                             P.O. Box 6013
                                                                Carol Stream, IL 60197-6013
14963581
           DISCOUNT TIRE
                                 13801 Danielson St.
                                                        Poway, CA 92064
14963583
                                 P.O. Box 7203
           DISH NETWORK
                                                   Pasadena, CA 91109-7303
14963584
           DISPLAYS2GO
                               81 Commerce Drive
                                                      Fall River, MA 02720
                                             15500 W. SAND ST. SUITE #1
14963586
           DIXON PEST CONTROL, INC.
                                                                               VICTORVILLE, CA 92392
14963587
           DLA PIPER LLP
                                P.O. Box 75190
                                                   Baltimore, MD 21275
                                Address Redacted
P.O. Box 735445
14963588
           DLVEEN N ALRAIS
14963589
           DOCUSIGN INC
                                                     Dallas, TX 75373-5445
14963593
                               500 Volvo Pkwy
           DOLLAR TREE
                                                    Chesapeake, VA 23320
                                 30 Frank Lloyd Wright Drive
14963595
           DOMINOS PIZZA
                                                                 Ann Arbor, MI 48105
           DON KENT – RIVERSIDE COUNTY TREAS.
DOOR HARDWARE CENTER 150 Long
14963596
                                                          38-686 El Cerrito Rd
                                                                                   Palm Desert, CA 92211
                                             150 Longwater Dr Ste 102B
14963600
                                                                           Norwell, MA 02061-1647
14963602
           DOTSQUARE INC., DBA GEOCODIO
                                                    3628 21st Ave N
                                                                         Arlingtion, VA 22207
           DOUBLETREE by HILTON
                                                            Phoenix, AZ 85008
14963603
                                          320 N 44th St
           DOUGLAS S. CUMMING-ATTORNEY AT LAW
14963606
                                                              Address Redacted
           DOUGLAS YOUNGLOVE
14963609
                                         Address Redacted
14963610
           DR MANAGEMENT SERVICES LLC
                                                   11612 S Breeze Grass Way
                                                                                Parker, CO 80134
                                        P.O. Box 25169
14963612
           DRC CITRUS OFFICE LP
                                                            Santa Ana, CA 92799
14963614
           DRI FARONICS
                               609 Granville St, Suite 1400
                                                              Vancouver, BC V7Y 1G5 Canada
14963615
           DRIVE SAVERS - DATA RECOVERY
                                                    400 Bel Marin Keys Blvd.
                                                                                 Novato, CA 94949
                                                                  El Cajon, CA 92020
14963616
           DRP HOLDINGS LLC
                                     124 Main Street, Suite 240
                                   124 W Main Street
                                                         Ste 240
                                                                     El Cajon, CA 92020
14963617
           DRP Holdings, LLC
14963618
           DRUG ENFORCEMENT ADMINISTRATION
                                                          P.O. BOX 2639
                                                                              Springfield, VA 22152
14963619
           DRUG PACKAGE LLC
                                                                 OFallon, MO 63366
                                      901 Drug Package Lane
14963620
           DSK ULTRASOUND
                                    1471 Pomona Rd. Suite G
                                                                 Corona, CA 92882
14963621
           DST PHARMACY SOLUTIONS
                                              P.O. Box 419019
                                                                   Kansas City, MO 64141-6019
14963624
           DUNLAP, WILLIAM
                                    Address Redacted
14963625
           DWIGHT M. MONTGOMERY - ATTORNEY-AT-LAW
                                                                    700 East Redlands Boulevard, Suite
           U/350
                      Redlands, CA 92373
           DYNAFOLD USA INC
14963626
                                      7439 La Palma Ave
                                                             STE 200
                                                                          Buena Park, CA 90620
14963386
           Daisy Arredondo Acosta
                                      Address Redacted
           Daisy Cervantes
14963387
                               Address Redacted
14963388
           Daisy Jimenez
                              Address Redacted
           Daisy Meza
Daisy Oregon
14963389
                           Address Redacted
14963390
                             Address Redacted
14963391
           Daisy Perez
                           Address Redacted
14963392
           Daisy Sandoval
                               Address Redacted
           Daisy Terron
Daisy Torres
14963393
                            Address Redacted
14963394
                            Address Redacted
14963395
           Dalana Perry
                            Address Redacted
                                 Address Redacted
14963397
           Damaris Alderman
14963398
           Damoun Rezai
                              Address Redacted
14963399
                           Address Redacted
           Dana Erwin
```

```
14963401
            Danicee King
                              Address Redacted
14963402
            Daniel Aguilar
                               Address Redacted
14963403
            Daniel Arzate
                               Address Redacted
            Daniel Camp
14963404
                              Address Redacted
14963406
            Daniel Cotton
                               Address Redacted
14963407
            Daniel Devine
                               Address Redacted
14963409
            Daniel Fishel
                              Address Redacted
14963410
            Daniel P Anderson
                                   Address Redacted
14963411
            Daniel Pearce
                               Address Redacted
14963412
            Daniel Rivas
                              Address Redacted
14963414
            Daniela Cervantes-Elia
                                        Address Redacted
            Daniela Dominguez
14963415
                                    Address Redacted
14963416
            Daniela Salazar
                                Address Redacted
14963417
            Daniela Zurita
                               Address Redacted
14963418
            Daniella Rodriguez
                                    Address Redacted
14963421
            Darek Fonder
                              Address Redacted
                                       707 Third Street, Second Floor
                                                                         CA DGS - Procurement
14963422
            Darfur Contracting Act
            Division
                          Sacramento, CA 95605
14963423
            Dariana Aguilera Sanchez
                                          Address Redacted
14963424
            Darlene Taylor
                                Address Redacted
14963425
            Darrel Gibson
                               Address Redacted
14963426
                             Address Redacted
            Daryl Priest
14963427
            Darynel Sandoval
                                  Address Redacted
14963428
            Dauod Ghafari
                               Address Redacted
14963430
            David Aguilar
                               Address Redacted
14963431
            David Armando Ballardo Martinez
                                                  Address Redacted
14963432
            David Baldwin
                                Address Redacted
14963433
            David Dick
                            Address Redacted
                             Address Redacted
14963434
            David Flick
14963435
            David Hernandez
                                  Address Redacted
14963436
            David Irwin
                             Address Redacted
14963437
                              Address Redacted
            David Kaufer
14963438
            David Rivera
                              Address Redacted
            David Villalobos
                                 Address Redacted
14963439
14963441
            Davna Devereaux
                                  Address Redacted
14968311
            Dean T. Kirby, Jr.
                                  KIRBY & McGUINN, A P.C.
                                                                    707 Broadway, Suite 1750
                                                                                                  San Diego, CA
            92101
14963449
            Deanette Simental
                                   Address Redacted
14963450
            Deanna McLaughlin
                                     Address Redacted
14963451
            Deanna Noll
                             Address Redacted
14963452
            Deanna Tolentino
                                  Address Redacted
14963453
            Deanna Townsend
                                   Address Redacted
14963454
            Deborah Dollins
                                 Address Redacted
14963455
            Deborah Streletz
                                 Address Redacted
14963456
            Deborah Wondoloski
                                     Address Redacted
14963457
            Debra Casserly
                                Address Redacted
14963458
            Debra Doran
                              Address Redacted
14963462
            Delfina Garcia
                               Address Redacted
14963463
            Delia Ibarra
                             Address Redacted
14963465
            Dellisa Garcia
                               Address Redacted
14963466
            Delmy Alvarado
                                 Address Redacted
14963468
            Delta Dental
                             P.O. Box 1810
                                                 Alpharelta, GA 30023
            Delta Drugs/Wellpartner
                                        437 Fernando Court
14963469
                                                                 Glendale, CA 91204
14963471
            Demetria Robertson
                                    Address Redacted
14963472
            Demetria Robertson
                                    Address Redacted
14963473
            Denevrie Martinez
                                   Address Redacted
14963474
            Denise Chacon
                                Address Redacted
14963475
            Denise Chapin
                               Address Redacted
14963476
            Denise Delatorre
                                 Address Redacted
14963477
            Denise Figueroa
                                 Address Redacted
14963478
            Denise Gordon
                                Address Redacted
14963479
            Denise Razo
                             Address Redacted
14963480
            Denise Rodriguez
                                  Address Redacted
14963481
            Denise Trevino Perez
                                      Address Redacted
14963482
            Denise Velasquez
                                  Address Redacted
14963483
            Denisse Espinosa
                                  Address Redacted
14963484
            Dennis Hom
                             Address Redacted
14963485
            Dennis Nourse
                               Address Redacted
14963486
            Dennis Roberts
                                Address Redacted
14963488
                                Address Redacted
            Dennise Munoz
            Dennise Walker
14963489
                                Address Redacted
14963493
            Deo Campbell
                               Address Redacted
14963494
            Department of Education
                                         P.O. Box 790356
                                                               St. Louis, MO 63179-0356
14963499
            Department of The Treasury
                                            P.O. Box 51320
                                                                Philadelphia, PA 19115-6320
                                        1500 Pennsylvania Avenue, NW
14963502
            Department of Treasury
                                                                            Washington, DC 20220
            Dept of Health Care SVCS, State of CA
14963503
                                                      Address Redacted
14963506
            Derakhsh Fozouni
                                   Address Redacted
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 20 of 76

Oust	of 76
14963507	Derakhsh Fozouni OB Hospital Coverage 555 Tachevah Drive, Suite 2w–103 Palm Springs, CA 92262
14963508	Desert Aids Project–DAP–Ryan White 1695 N. Sunrise Way Palm Springs, CA 92262
14963509 14963511	Desert Bumper Pros 74–841 42nd Ave Palm Desert, ČA 92260 Desert Community College 4800 Magnolia Avenue District of Riverside CTY Riverside, CA 92506–1299
14963516	Desert Healthcare Foundation 1140 N. Indian Canyon Drive Palm Springs, CA 92262
14963517	Desert Hot Springs Wellness Center 11750 Cholla Drive, Suite B Desert Hot Springs, CA 92240
14963518	Desert Medical Group 275 N. El Cielo Rd Desert Oasis Healthcare Med Palm Springs, CA 92262
14963519	Desert Medical Imaging Desert Oasis Healthcare 74785 Hwy 111, Suite 101 Palm Springs, CA 92210 Palm Springs, CA 92262
14963520	Desert Oasis Healthcare 275 N El Cielo Road Palm Springs, CA 92262
14963521	Desert Pain – Tobias Moeller–Bertram 36101 Bob Hope Dr Ste B2 Rancho Mirage, CA 92270
14963526 14963528	Desert Sands Unified School District 47–950 Dune Palms Road La Quinta, CA 92253 Desert Valley Disposal, Inc. 4690 East Mesquite Avenue Palm Springs, CA 92264
14963533	Desert Valley Disposal, Inc. 4690 East Mesquite Avenue Palm Springs, CA 92264 Deshea Wilson Address Redacted
14963534	Design Benefits 3447 W Central Ave Wichita, KS 67203
14963535	Desiree Duran Address Redacted
14963536	Desktop Central 4141 Hacienda Drive Zoho Corporation c/o ManageEngine Pleasanton, CA
	94588
14963537	Destiny Lopez Address Redacted
14963538	Dhafir Petros 3530 Camino Del Rio N Suite 109 San Diego, CA 92108 Dhahir Abumakki Address Redacted
14963539 14963543	Diamond Fowler Address Redacted
14963546	Diana Arxer Address Redacted Diana Arxer Address Redacted
14963547	Diana Bodo Address Redacted
14963548	Diana Borunda Address Redacted
14963549	Diana Camacho Address Redacted
14963550	Diana Esho Address Redacted
14963551	Diana Esho Address Redacted Diana Gardner Address Redacted
14963552 14963553	Diana Gardner Address Redacted Diana Hernandez Address Redacted
14963554	Diana Jaramillo Address Redacted Diana Jaramillo Address Redacted
14963555	Diana Marrufo Michel Address Redacted
14963556	Diana Morales Address Redacted
14963557	Diana Owen Address Redacted
14963558	Diana Ramirez Address Redacted
14963559 14963560	Diana Ramos Address Redacted Diana Sanchez Address Redacted
14963561	Diana Thompson Address Redacted Diana Thompson Address Redacted
14963562	Diana Thompson (Troncoso) Address Redacted
14963563	Diana Uriostegui Address Redacted
14963564	Dianna Fragoso Address Redacted
14963565	Dianne Tungol Address Redacted
14963566 14963568	Dib Pharmaceutical Inc. 528 E Main St El Cajon, CA 92020–4008 Diego Barajas Address Redacted
14963569	Diego Rodriguez Maldonado Address Redacted
14963571	Dignity Health Foundation Inland Empire P.O. Box 2637 San Bernardino, CA 92406
14963573	Dina Basali Address Redacted
14963576	Diri Marroquin Address Redacted
14963582	Dish Network 9601 South Meridian Boulevard Englewood, CO 80112
14963585 14963590	DiversityDHS 14080 Palm Drive Ste D423 Desert Hot Springs, CA 92240 Doe Gasque Address Redacted
14963591	Doe Gasque Address Redacted Dogon Behavioral Medical Group P.O. Box 51030 Riverside, CA 92517
14963592	Dogon–The Neighborhood Med Clinic 4960 Arlington Ave., Suite B Riverside, CA 92504
14963594	Dolores Taylor Address Redacted
14963597	Donna Nunez Address Redacted
14963598	Donna Rochelle Address Redacted Dona Drugg Wollnesteer 1222 W 5th St. Son Remarking CA 02411
14963599 14963601	Dons Drugs/ Wellpartner 1222 W 5th St San Bernardino, CA 92411 Dora Murillo Address Redacted
14963604	Douglas Habig Address Redacted Address Redacted
14963605	Douglas Ness, DDS 600 B Street, Suite 2491 Sharif Faust Lawyers Ltd–Matthew J Faust San Diego,
	CA 92101
14963607	Douglas Smith Address Redacted
14963608	Douglas Younglove Address Redacted
14963611 14963613	Dr. Alfredo Ratniewski Address Redacted Drexel University 3141 Chestnut Street Philadelphia, PA 19104
14963622	Dulce Acosta Villarreal Address Redacted
14963623	Dulce Marquez Vega Address Redacted
14963627	E LANDSĆAPING AND POOLS 38085 Dorn RD Cathedral City, CA 92234
14963674	E-FILE MAGIC P.O. BOX 20576 Portland, OR 97294
14963629	EA COMPENSATION RESOURCES LLC 111 Wood Avenue South, 6th Floor Iselin, NJ
14963630	08830–2700 EAGLE GRAPHICS 600 City Pkwy W STE 600 Orange, CA 92868–2945
14963631	EAGLE GRAPHICS 600 City Pkwy w STE 600 Grange, CA 92808–2943 EAGLE PRINT DYNAMICS 1430 W Katella Aver Orange, CA 92867
14963633	EAST COUNTY CAR DETAIL 456 North Magnolia El Cajon, CA 92020
14963635	EAST COUNTY LOCK & SAFE 4270 North Cordoba Ave Spring Valley, CA 91977
14963638	EASY GLIDE 3738 WALNUT PARK WAY Hemet, CA 92544

```
14963639
           EASY PARTY RENTALS
                                       66184 Pierson Blvd
                                                             Desert Hot Springs, CA 92240
                                               San Jose, CA 95125
14963640
          EBAY
                    2025 Hamilton Avenue
14963641
           EBSCOHealth
                            10 Estes Street
                                               EBSCO Information Services
                                                                             Ipswich, MA 01938
14963642
           ECG MANAGEMENT CONSULTANTS
                                                   11512 El Camino Real, Suite 200
                                                                                     San Diego, CA 92130
14963643
                           P.O. Box 1171130
                                                Atlanta, GA 30368-7113
           ECMC – VELASQUEZ, ESPERANZA
14963644
                                                  Address Redacted
14963645
           ECOLA SERVICES-RIVERSIDE
                                              15314 Devonshire St.
                                                                     Mission Hills, CA 91345
14963646
           ECONO FIRESTONE TIRE
                                        7445 Arlington Ave
                                                              Riverside, CA 92503
14963647
           ECRI Institute
                           5200 Butler Pike
                                                Plymouth Meeting, PA 19462
14963656
          EDGE ENDO LLC
                                5600 WYOMING BĽVD
                                                           Albuquerque, NM 87109
          EDGE HOTEL 514 W 16TH ST New York, NY 10032
EDWARD STEVE RUDKIN BACKFLOW 7851 Bonnie Street
14963657
14963672
                                                                           San Bernardino, CA 92410
14963675
           EFILEFORBUSINESS.COM
                                         3300 Gateway Dr
                                                             eFile4Biz.com
                                                                               Pompano Beach, FL 33069
14963677
           EHS, INC / GREENWAY
                                      P.O. Box 203658
                                                          Dallas, TX 75320-3658
           EL POLLO LOCO
                               3535 Harbor Blvd #100
14963684
                                                          Costa Mesa, CA 92626
                                                             Apartment 3A
14963686
           ELEGANT THEMES, INC.
                                        1233 Howard Street
                                                                                San Francisco, CA 94103
14963688
          ELEPHANT FIRE EXTINGUISHER SERVICE
                                                        3277 Appalachian Drive
                                                                                   Perris, CA 92570
                                           Address Redacted
                                             P.O. Box 5299
14963691
           ELISA S. ROGERS, M.D. INC.
           ELISE BROWN, MD-Now UC San Diego Health
14963696
14963699
           ELITE FIRE PROTECTION, INC
                                                              Hemet, CA 92544-0299
14963700
           ELITE MOBILE RV SERVICES, INC.
                                                 16412 El Revino Dr.
                                                                        Fontana, CA 92336
                                      Address Redacted
14963703
           ELIZABETH CERRONE
14963727
           ELLIS FENCE COMPANY, INC.
                                             28633 US Hwy 58
                                                                  Barstow, CA 92311
                             9555/9595 Springboro Pike
14963734
           ELSEVIER INC.
                                                          Miamisburg, OH 45342
14963735
           ELSEVIER, CLINICAL SOLUTIONS NA
                                                    1600 John F. Kennedy Blvd, Ste 1800
                                                                                          Philadelphia, PA
           19103-2899
                                                                 Memphis, TN 38117
14963739
           EMBASSY SUITES HOTELS
                                          755 Crossover Lane
                                     One Technology Place
14963741
           EMD Serono – Serostim
                                                             Rockland, MA 02370
14963751
           EMPIRE PHYSICIANS MEDICAL GROUP
                                                     P.O. Box 19646
                                                                         Minneapolis, MN 55419
                                         200 Route 130 North
                                                                Cinnaminson, NJ 08077
14963752
           EMSL ANALYTICAL, INC.
           ENDEAVOR BUSINESS MEDIA
                                            30 Burton Hills Blvd, Suite 185
14963754
                                                                             Nashville, TN 37215
                                 11810 Wills Road, Suite 100
                                                              Alpharetta, GA 30009
14963755
           ENDOCHOICE INC
14963762
           ENVIRO MATRIX ANALYTICAL, INC.
                                                   9590 Chesapeake Dr., Ste. 5
                                                                                San Diego, CA 92123
14963764
           EPIC – Insurance Brokers
                                      1 California Street, Suite 400
                                                                    San Francisco, CA 94111
                      34-160 Gateway Dr., Suite 100
                                                      Empire Physicians Medical Group
14963766
           EPMG
                                                                                        Palm Desert, CA
           92211
                                                       720 North Broadway
14963796
           ESCONDIDO CHAMBER OF COMMERCE
                                                                              Escondido, CA 92025
                                                              Escondido, CA 92025
           ESCONDIDO DISPOSAL
14963797
                                     1021 W Mission Ave
                                                                                    Ste #701
14963798
           ESCONDIDO DOWNTOWN BUSINESS ASSOC.
                                                            243 S. Escondido Blvd
                                                                                                  Escondido,
           CA 92025
14963799
           ESCONDIDO GIRLS SOFTBALL LEAGUE
                                                       P.O. Box 1208
                                                                        Escondido, CA 92033
14963800
           ESCONDIDO POLICE DEPARTMENT - ALARM
                                                            P.O. Box 1783
                                                                              Escondido, CA 92033
14963811
           ESSENTIAL ACCESS
                                    3600 Wilshire Blvd
                                                          Ste 600
                                                                     Los Angeles, CA 90010–2610
14963817
           ESUPPLYBOX
                             105 Harrison Ave
                                                  Harrison, NJ 07029-1332
14963819
           ETR ASSOCIATES
                                 5619 Scotts Valley Drive, Suite 140
                                                                     Scotts Valley, CA 95066
14963820
                               1507 HIGHWAY A
                                                      WENTZVILLE, MO 63385
           ETRAILER.COM
14963836
           EVENT BRITE
                              209 10th Ave S #300
                                                     Cummins Station
                                                                         Nashville, TN 37203
14963840
           EXPERTS EXCHANGE
                                     7301 Morro Rd Ste 105A
                                                                Atascadero, CA 93422-4458
                                                    Saint Louis, MO 63121
14963842
           EXPRESS SCRIPTS
                                  1 Express Way
14963843
           EXTENDEDSTAY AMERICA
                                          11525 N. Community House Road
                                                                              Charlotte, NC 28277
14963632
           Eartha Reed
                          Address Redacted
14963634
           East County Family Dental
                                       13465 Camino Canada Road, Suite 110-A
                                                                                El Cajon, CA 92021
14963636
           Eastside Health Center
                                   1970 University Avenue
                                                             Riverside, CA 92507
14963637
           Eastside/Arlanza
                              8856 Arlington Avenue
                                                       Riverside, CA 92503
14963648
                            Address Redacted
           Eddie Lemon
14963649
           Edgar Bulloch
                            Address Redacted
14963650
           Edgar Ledezma
                             Address Redacted
14963651
           Edgar Montalva
                              Address Redacted
14963652
           Edgar Oropeza Ramirez
                                    Address Redacted
           Edgar Rojo Valenzuela
14963653
                                    Address Redacted
14963654
           Edgar Zendejas
                             Address Redacted
14963655
           Edgardo Felix
                            Address Redacted
           Edgenet Consulting, LLC
                                      5716 Wild Orchid Trail
14963658
                                                               Raleigh, NC 27613-8546
14963659
           Edilma Ramirez
                              Address Redacted
           Edit Cebreros
14963660
                            Address Redacted
14963661
                             Address Redacted
           Edith Gonzalez
14963662
           Edith Gonzalez Lozano
                                    Address Redacted
                          Address Redacted
14963663
           Edith Lopez
14963664
           Edith Mendez Flores
                                 Address Redacted
14963665
           Edlin Zavala
                          Address Redacted
                                  Address Redacted
14963666
           Edna Gomez Duenas
14963667
           Edna Ramirez-Ramirez
                                    Address Redacted
14963668
           Eduardo Gutierrez
                                Address Redacted
14963669
           Eduardo Macias
                              Address Redacted
14963670
           Eduardo Zavala
                              Address Redacted
           Edward Requeno
14963671
                               Address Redacted
14963673
           Effuah Harris
                            Address Redacted
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 22 of 76

```
14963676
           Ehab Samaan
                              1358 W. 6th Street Suite 101
                                                               Corona Family Dental Group-Ehab Samaan
                                                                                                             Corona,
            CA 92882
14963678
           Eileen Bahena Hernandez
                                         Address Redacted
                               Address Redacted
14963679
            Eileen Tavares
                               Address Redacted
14963680
            Eileen Weilein
14963681
           Eisenhower Medical Center
                                           39000 Bob Hope Drive
                                                                      Rancho Mirage, CA 92270-3221
14963682
            El Cajon Express Pharmacy/Wellpartner
                                                       330 S Magnolia Ave Ste 102
                                                                                       El Cajon, CA 92020
           El Mirador Medical Plaza Pharmacy
                                                   1180 N Indian Canyon Dr Ste E140
14963683
                                                                                         dba Desert Hosp Out Patient
            PHCY
                        Palm Springs, CA 92262-4883
                           Address Redacted
14963685
           Elba Sainz
14963687
            Elena Walter
                              Address Redacted
                              Address Redacted
14963689
            Elias Koutros
14963690
           Elisa Rogers
                             Address Redacted
14963692
                              Address Redacted
           Elisa Salazar
14963693
            Elisabeth Marquez
                                   Address Redacted
14963694
            Elisabeth Ramirez
                                   Address Redacted
                                  Address Redacted
14963695
           Elisandro Damian
14963697
           Elissa Garcia
                              Address Redacted
14963698
           Elissa Hayes
                              Address Redacted
14963701
            Elizabet Medina
                                 Address Redacted
14963702
           Elizabeth Alonzo
                                  Address Redacted
            Elizabeth Claire Tapang
14963704
                                        Address Redacted
14963705
                                    Address Redacted
           Elizabeth Contreras
14963706
           Elizabeth Dominguez
                                      Address Redacted
           Elizabeth Dominguez
14963707
                                     Address Redacted
14963708
           Elizabeth Hernandez
                                     Address Redacted
14963709
           Elizabeth Hernandez
                                     Address Redacted
14963710
           Elizabeth Jimenez
                                   Address Redacted
14963711
           Elizabeth Labha
                                 Address Redacted
14963712
           Elizabeth Martinez
                                   Address Redacted
14963713
            Elizabeth Melendrez
                                     Address Redacted
                                 Address Redacted
14963714
           Elizabeth Munoz
14963715
           Elizabeth Newell
                                 Address Redacted
14963716
           Elizabeth Ojeda
                                Address Redacted
            Elizabeth Pavne
                                 Address Redacted
14963717
14963718
           Elizabeth Pinedo
                                 Address Redacted
14963719
           Elizabeth Ramos
                                 Address Redacted
14963720
                                 Address Redacted
           Elizabeth Rochin
14963721
            Elizabeth Rodarte
                                  Address Redacted
14963722
                                Address Redacted
           Elizabeth Silva
14963723
           Elizabeth Tan
                              Address Redacted
14963724
           Elizabeth Vallejo
                                  Address Redacted
14963725
            Elizabeth Vallejo v. BCHF
                                          8889 W Olympic Blvd # 200
                                                                           Lavi & Ebrahimian
                                                                                                   Beverly Hills, CA
            90211
14963726
            Ellie Lopez
                            Address Redacted
14963728
            Ellisa Branch
                              Address Redacted
14963729
            Elmer Espinoza
                                Address Redacted
14963730
           Elmer Hilo
                            Address Redacted
14963731
           Eloisa Contreras Carmona
                                          Address Redacted
            Eloisa Madrigal-Ventura
14963732
                                         Address Redacted
14963733
                               Address Redacted
           Eloise Skelton
14963736
           Elvia Chambers
                                Address Redacted
                                   Address Redacted
14963737
            Emanuel Coronado
14963738
            Emanuel Morales Guzman
                                          Address Redacted
14963740
                                      P.O. Box 2833
                                                          New York, NY 10116-2833
           EmblemHealth - GHI
14963742
           Emelia Bernal
                               Address Redacted
           Emelissa Borro
                                Address Redacted
14963743
14963744
            Emily Gentry
                              Address Redacted
14963745
           Emily Rosas
                             Address Redacted
14963746
           Emily Zavala
                              Address Redacted
14963747
            Emma Bermudez
                                  Address Redacted
14963748
            Emmanuel Scaife
                                  Address Redacted
14963749
           Empire Physician Medical Group
                                                3990 Concours Street, Suite 500
                                                                                    Ontario, CA 91765-7970
14963750
           Empire Physicians Medical Group
                                                 3990 Concours Suite 500
                                                                              Ontario, CA 91764
            Encoursa
14963753
                          38 Fulton St W #30
                                                  Grand Rapids, MI 49503
14963756
            Enedina Diaz
                              Address Redacted
14963757
            Engineering Design Group
                                          2121 Montiel Rd
                                                                San Marcos, CA 92069
            Enjoli Body
                             Address Redacted
14963758
14963759
            Enrique Perez
                               Address Redacted
14963760
            Enterix Inc.
                            236 Fernwood Avenue
                                                       Edison, NJ 08837
           Enterprise Fleet Management
                                             6330 Marindustry Dr
14963761
                                                                      Ste 210
                                                                                   San Diego, CA 92121
14963763
           Environmental Protection Agency
                                                 75 Hawthorne Street
                                                                          San Francisco, CA 94105
                                 1360 Truxton Avenue, Suite 300
14963767
            Equiscript, LLC
                                                                    North Charleston, SC 29405
14963768
            Eric Brown
                            Address Redacted
14963769
            Eric Eschweiler
                                Address Redacted
14963770
                            Address Redacted
           Eric Joseph
14963771
            Eric Mclaughlin
                                 Address Redacted
           Eric Pekarek
14963772
                              Address Redacted
```

```
14963773
            Erica Aceves
                              Address Redacted
14963774
           Erica Carrillo
                              Address Redacted
14963775
            Erica Herrera
                              Address Redacted
14963776
            Erica Muralles
                               Address Redacted
14963777
            Erica Quezada
                               Address Redacted
14963778
           Erica Ruano
                             Address Redacted
14963779
            Erica Torres
                             Address Redacted
14963780
           Erick Lopez Romero
                                    Address Redacted
14963781
            Erik Luansing
                               Address Redacted
14963782
           Erik Polanco
                             Address Redacted
14963783
           Erika Amarillas
                                Address Redacted
14963784
            Erika Benitez
                              Address Redacted
14963785
            Erika Ceja
                           Address Redacted
14963786
           Erika Chavarin
                               Address Redacted
14963787
            Erika Heredia
                              Address Redacted
14963788
            Erika Lopez
                             Address Redacted
14963789
                            Address Redacted
           Erika Perez
14963790
           Erika Robles
                             Address Redacted
14963791
           Erika Sanchez
                               Address Redacted
14963792
            Erin Daughhetee
                                 Address Redacted
14963793
                             Address Redacted
           Erin Metcalf
14963794
            Ernesto Loza
                              Address Redacted
14963795
                                 Address Redacted
           Esau Maldonado
14963801
            Esgardo Duarte
                                Address Redacted
14963802
           Esla Gastellum
                               Address Redacted
14963803
           Esmeralda Flores
                                 Address Redacted
14963804
            Esmeralda Marquez
                                    Address Redacted
14963805
            Esmeralda Matute
                                  Address Redacted
14963806
           Esmeralda Padilla
                                  Address Redacted
14963807
            Esmeralda Quiroz
                                  Address Redacted
           Esperanza Gamez
Esperanza Velasquez
14963808
                                  Address Redacted
14963809
                                     Address Redacted
           Essam Beshay
14963810
                               Address Redacted
14963812
            Estefani Altamirano
                                    Address Redacted
14963813
            Esther Arias
                             Address Redacted
14963814
           Esther Centeno
                               Address Redacted
14963815
            Esther Viramontes
                                   Address Redacted
14963816
           Estrada Jesenia
                               Address Redacted
14963818
            Etienne McEnroe
                                 Address Redacted
14963821
           Eugene Washington PCORI/Regents of UCR
                                                          1111 Franklin St.,12th floor
                                                                                          Offc Sec & Chief of Staff for
            Regents
                         Oakland, CA 94607
                             Address Redacted
14963822
            Eva Sanchez
14963823
            Evan Saoor
                            Address Redacted
14963824
           Evan Schwenk
                               Address Redacted
14963825
            Evangelina Gonzalez
                                     Address Redacted
14963826
           Evanie Nugent
                               Address Redacted
14963827
            Evanston Insurance Company
                                             10275 West Higgins Road, Suite 750
                                                                                    Rosemont, IL 60018
           Evanston Insurance Company
14963828
                                             201 California Street, Suite 1450
                                                                                San Francisco, CA 94111
                            Address Redacted
14963829
           Evel Hanna
                               Address Redacted
14963830
           Evelyn Chavez
14963831
                               Address Redacted
            Evelyn Estrada
14963832
            Evelyn Garcia
                               Address Redacted
14963833
            Evelyn Garcia
                               Address Redacted
14963834
            Evelyn Ortega Cruz
                                   Address Redacted
14963835
           Evelyn Robles Hernandez
                                         Address Redacted
14963837
           Everardo Gonzalez
                                   Address Redacted
14963838
            Everett Simerson
                                 Address Redacted
14963839
            Exclusive Surgeries Solutions
                                             8671 W Union Hills Dr Ste 503
                                                                               Peoria, AZ 85382
14963841
                                  30214 Sussex Hwy., Unit 7
           Express Pharmacy
                                                                 Laurel, DE 19956
14963844
            Ezabelle Hayek
                                Address Redacted
14963845
                               Address Redacted
            Ezekiel Deleon
14963849
            FACEBOOK
                              1 Hacker Way
                                                 Meta
                                                            Menlo Park, CA 94025
14963852
           FAIRFIELD INN
                                  10400 Fernwood Road
                                                            Bethesda, MD 20817
14963860
           FASTSIGNS
                              1240 E. Plaza Blvd, Suite 602
                                                               National City, CA 91950
14963866
           FAVOR CREATIONS
                                       1210 Driftwood Dr
                                                              Pittsburgh, PA 15243-1737
14963868
            FEBC Worker Benefit Fund of Southern Ca.
                                                          9121 W Rusell Rd Ste 219
                                                                                        Las Vegas, NV
            81948-1239
14963869
                         P.O. Box 7221
                                            Pasadena, CA 91109-7321
            FEDEX
14963870
           FELDESMAN TUCKER LEIFER FIDELL LLP
                                                              1129 20th Street, NW, 4th Floor
                                                                                                 Washington, DC
            20036
14963873
           FERGUSON H&C
                                   81-925 B Industrial
                                                           Indio, CA 92201-2020
14963874
           FERGUSON H&C - Cathedral City #2769
                                                        36283 Plaza Drive
                                                                               Cathedral City, CA 92234–7323
14963876
           FERNANDO BELTRAN JR
                                            Address Redacted
14963881
            FERRELLGAS
                                P.O. Box 173940
                                                      Denver, CO 80217-3940
14963883
                                78990 Varner Rd
                                                      Indio, CA 92203
            FIESTA FORD
                                                        Santa Monica, CA 90404
14963884
           FIGS
                       2834 Colorado Ave, Suite 100
14963885
            FILINGSUPPLIES.COM
                                         11956 Bernardo Plaza Dr, Suite 524
                                                                               San Diego, CA 92128-2538
14963886
           FILLINGSUPPLIES.COM
                                          11956 Bernardo Plaza Dr, Suite 524
                                                                                 San Diego, CA 92128-2538
```

	01.70	
14963887 14963888		
14963894	140 Austin, TX 78746	
14963897		
14963898		2595
14963899		
14963900		
14963901 14963903		
14963903		
14963910	FLOWERS DIRECT ACCESS INC 4301 Papermill Dr Flowers Direct, Inc. Knoxville	, TN
14963911	37909–2069 I FLOYD SKEREN & MANUKIAN 7525 Metropolitan Drive, Suite 304 San Diego, CA 9210	10
14963912		.0
14963913		
14963914		
14963916		
14963919		
14963922		7.4
14963923	94267-0021	A
14963924	FRANCHISE TAX BOARD P.O. Box 942867 CATHY AGUILAR SACRAMENTO, 94267–0021	CA
14963925	5 FRANCHISE TAX BOARD P.O. Box 942867 EDLIN ZAVALA SACRAMENTO, C 94267–0011	A
14963926		TO, CA
14963927		O, CA
14963928	B FRANCHISE TAX BOARD P.O. Box 942867 FRALEY, EVELYN SACRAMENTO,	CA
14963929		, CA
14963930		A
14963931	,	CA
14963932	, ,	TO, CA
14963933		A
14963934		TO, CA
14963935		CA
14963936	, , , , , , , , , , , , , , , , , , ,	O, CA
14963937		CA
14963938		O, CA
14963939	the state of the s	O, CA
14963940	, , , , , , , , , , , , , , , , , , , ,	CA
14963941	, ,	O, CA
14963942		RED DEBT
14963943		EBT
14963944		
14963945	BOARD SACRAMENTO, CA 94240–0000 5 FRANCHISE TAX BOARD – KAREN HEBETS P.O. Box 942840 SACRAMENTO, CA 94	1240
14963945	5 FRANCHISE TAX BOARD – LUIS TRIGUEROS P.O. Box 1328 COURT ORDERED DE	
14963947		RED DEBT
14963951	COLLECTIONS RANCHO CORDOVA, CA 95741–1328 I FRANK FEDER MEDIATORS 2173 Salk Avenue, Suite 250 Carlsbad, CA 92008	
14963953		
14963958	B FRIENDS UNITING NEIGHBORS INC 31805 Temecula Pkwy, #218 Temecula, CA 92592	-9804
14963959	FRIO ZONE REFRIGERATION 1563 Jayken Way, Suite B Chula Vista, CA 91911	
14963960		
14963961 14963962	•	San
1.703702	Jacinto, CA 92582	Juli

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 25 of 76

```
14963964
           FRONTIER COMMUNICATIONS
                                                 401 Merritt 7 Ste 7
                                                                        760-340-9177-Cook St
                                                                                                   Norwalk, CT
           06851-1069
14963966
           FRONTIER COMMUNICATIONS
                                                 P.O. Box 920041
                                                                      Dallas, TX 75392-0041
                                                                P.O. Box 920041
           FRONTIER COMMUNICATIONS-DHS#2ALARM
14963968
                                                                                     Dallas, TX 75392-0041
14963967
           FRONTIER COMMUNICATIONS/VERIZON
                                                           P.O. Box 920041
                                                                                60-399-4526 CV
                                                                                                      Dallas, TX
           75392-0041
14963969
           FS *ZEBRA TECH
                                   3 Overlook Point
                                                        Lincolnshire, IL 60069
14963970
                      175 N Street, NE
           FTCA
                                           Civil Division US Department of Justice
                                                                                      Washington, DC 20002
                                          9770 Silicon Prairie Pkwy Fitchburg, WI 53593
53603 Polk St Coachella, CA 92236
14963971
           FULL COMPASS SYSTEM
                                         53603 Polk St
14963972
           FULTON DISTRIBUTING
14963973
           FUN SERVICES
                                23720 Via Del Rio
                                                       Yorba Linda, CA 92887
           FW COOK
                           685 Third Avenue, 28th Floor
                                                           New York, NY 10017
14963974
14963846
           Fabio Leon
                           Address Redacted
14963847
           Fabiola Mota Carreon
                                    Address Redacted
14963848
                               Address Redacted
           Fabiola Salazar
14963850
           Fadi Esho
                          Address Redacted
14963851
                           Address Redacted
           Fadi Sabah
14963853
           Fairouze Tupo
                              Address Redacted
14963854
           Faiz Matti
                          Address Redacted
14963855
           FalconStor
                           501 Congress Avenue, Suite 150
                                                              Austin, TX 78701
14963856
                                    22435 Alessandro Blvd, Suite 106
           Family Dentistry Inc.
                                                                        Moreno Valley, CA 92553
                            Address Redacted
14963857
           Farah Tobya
14963858
           Fareshta Quedees
                                 Address Redacted
14963859
           Farzin Morris
                             Address Redacted
14963861
           FastSigns/2009
                               17198 Yuma St
                                                   Victorville, CA 92395
14963862
           Faten Khalaf
                             Address Redacted
14963863
           Fatima Castillo
                               Address Redacted
14963864
           Fatimah Abdulridha
                                   Address Redacted
14963865
           Faviola Moscoso
                                Address Redacted
14963867
           Fazela Wasimi
                              Address Redacted
14963871
           Felicia Melendrez
                                 Address Redacted
14963872
           Felipe Cendejas
                               Address Redacted
           Fernando Beltran
14963875
                                Address Redacted
14963877
           Fernando Delgado
                                 Address Redacted
           Fernando Leal
                              Address Redacted
14963878
14963879
           Fernando Perez
                               Address Redacted
14963880
           Ferrellgas
                          7500 College Boulevard, Suite 1000
                                                                Overland Park, KS 66210
14963882
           Fidel Gamez Ochoa
                                   Address Redacted
14963889
           Find Food Bank
                                83775 Citrus Ave
                                                     Indio, CA 92201
14963890
                               12735 Gran Bay Parkway, Suite 150
           Firehouse Subs
                                                                     Jacksonville, FL 32258
                                                         Riverside CTY Children &Families Comm
14963891
           First 5 Riverside
                                585 Technology Court
                                                                                                      Riverside, CA
           92507
14963892
           First American Equipment Finance
                                                211 High Point Drive
                                                                         Victor, NY 14564
14963893
           First Desert Obstetrics & Gyno Med Assoc
                                                       81767 Doctor Carreon Blvd #101
                                                                                          Indio, CA 92201
14963895
           First Health/CCN/Coventry Health Care
                                                     5210 E. Williams Circle, Suite 220
                                                                                         Tucson, AZ 85711
14963896
           First Insurance Funding
                                      450 Skokie Blvd, Suite 1000
                                                                     Northbrook, IL 60062
14963902
           Fiyinfoluwa Fawibe
                                   Address Redacted
14963905
           Flordeliza Burns
                                Address Redacted
14963906
                               750 East Main Street
           Florence Ugboh
                                                        Barstow, CA 92311
           Florence Ugboh
                               Address Redacted
14963907
14963908
                                 2452 Fenton Street, Ste 102
           Florencio Burquez
                                                                Chula Vista, CA 91914
14963909
           Florida Department of Revenue P.O. Box 6668
                                                                Tallahassee, FL 32314
14963915
           Foothills Psychological Services, Inc.
                                                  9089 Baseline Road, Suite 200
                                                                                   Rancho Cucamonga, CA
           91730
           Ford Motor Company
14963917
                                    P.O. Box 6248
                                                       Customer Relationship Center
                                                                                        Dearborn, MI 48126
                          P.O. Box 1190
                                                                                  Springfield, MO 65801-1190
14963918
           Forvis, LLP
                                             910 E. St. Louis Street, Suite 200
14963920
           Foxit Software Incorporated
                                         41841 Albrae Street
                                                                 Fremont, CA 94538
14963921
           Frances Arellano
                                Address Redacted
14963948
                                Address Redacted
           Francisco Flores
14963949
           Francisco Lopez
                                Address Redacted
14963950
           Francisco Ochoa
                                Address Redacted
14963952
           Frank Figueroa
                               Address Redacted
14963954
           Frank Tabarez
                              Address Redacted
14963955
           Frankie Ruiz
                             Address Redacted
14963956
           Freddy Ortiz
                            Address Redacted
14963957
                                    Address Redacted
           Fredi Ponce Gutierrez
14963963
           Frontier Communications
                                        401 Merritt 7
                                                         Norwalk, CT 06851
           Frontier Communications
14963965
                                        P.O. Box 740407
                                                             Cincinnati, OH 45274–0407
14963975
           GABRIEL MALDONADO
                                         Address Redacted
14963987
           GAINES PEST CONTROL
                                          1627 Doncarol Ave
                                                                 El Cajon, CA 92019
                                                 Madison Heights, MI 48071
14963988
                         26010 PINEHURST
           GALCO
14963992
           GALLS
                        1340 Russell Cave Rd
                                                  Lexington, KY 40505
           GANDER RV & OUTDOORS
14963993
                                             250 Parkway Drive Ste. 270
                                                                           Lincolnshire, IL 60069
           GARDAWORLD SECURITY SERVICES
14963994
                                                      P.O. Box 843886
                                                                             Kansas City, MO 64184-3886
                                                             Nashville, TN 37214
14963999
           GAYLORD HOTELS
                                     2800 Opryland Drive
                                                       10055 Slater Avenue, Suite 214
14964000
           GEMINI DIVERSIFIED SERVICES, INC
                                                                                         Fountain Valley, CA
           92708
14964012
           GENTRY PLAZA, LLC
                                       31225 La Baya Drive, Suite 206
                                                                         Westlake Village, CA 91362
```

```
14964027
                            358 Goucher St
                                               Johnstown, PA 15905-3423
14964031
           GILEAD SCIENCES, INC.
                                       P.O. Box 44000
                                                          Dept. 44641
                                                                            San Francisco, CA 94144
                                                                       Carlsbad, CA 92009
14964038
           GLASS BOX TECHNOLOGY, INC.
                                                2855 Camino Serbal
                                     2550 Fifth Avenue, Suite 605
                                                                   San Diego, CA 92103
14964039
           GLASS MEDIATIONS
                                                          P.O. Box 740415
14964040
           GLAXOSMITHKLINE PHARMACEUTICALS
                                                                              Atlanta, GA 30374-0415
14964042
           GLOBAL EQUIPMENT COMPANY
                                                 29833 NETWORK PLACE
                                                                              Chicago, IL 60673
14964043
           GLOBAL INDUSTRIAL
                                      11 Harbor Park Drive
                                                             Port Washington, NY 11050
14964044
           GLOBAL MEMORY PROCUR
                                           216 Avenida Fabricante Suite 103
                                                                              San Clemente, CA 92672
                                      17130 Van Buren Blvd 25
14964045
           GLOBAL PROTECTION
                                                                  Riverside, CA 92504-5905
          GLORIA ARANDA
14964046
                                  Address Redacted
                           2155 E GoDaddy Way
14964053
           GODADDY
                                                    Tempe, AZ 85284
14964054
           GOFF, YVONNE
                               Address Redacted
14964055
           GOKEYLESS
                             955 Mound Rd.
                                                Mound Business Park
                                                                        Miamisburg, OH 45342
14964056
           GOLDEN BRUSH SIGNS
                                       108 E. FLORIDA AVE.
                                                                 HEMET, CA 92543
           GOLDEN OFFICE TRAILERS, INC.
                                                                  Wildomar, CA 92595
14964057
                                                 P.O. Box 669
14964059
           GOLDEN STATE WATER COMPANY – 57660282286
                                                                 P.O. Box 9016
                                                                                   San Dimas, CA
           91773-9016
14964060
           GOLDEN STATE WATER COMPANY - 63889418586
                                                                 P.O. Box 9016
                                                                                   San Dimas, CA
           91773-9016
14964061
           GOLDSTEIN, PHILIPPE
                                      P.O. Box 434441
                                                           San Diego, CA 92143
14964063
           GOOD LITE COMPANY
                                      1155 Jansen Farm Dr
                                                               Elgin, IL 60123
           GOODWIN PROCTER, LLP
14964065
                                          100 Northern Avenue
                                                                  Boston, MA 02210
14964066
           GOOGLE PLAY
                              1600 Amphitheatre Parkway
                                                             Mountain View, CA 94043
          GOSCH COLLISION – HEMET – FORD
GOSQ.COM 1455 Market Street
                                                    150 CARRIAGE CIRCLE
14964068
                                                                                 Hemet, CA 92545
14964069
                                                  San Francisco, CA 94103
                                      P.O. Box 706
14964072
           GRADEXCHANGE, INC
                                                       New Haven, CT 06503
                           DEPT. 869879205
14964074
           GRAINGER
                                                 Palatine, IL 60038-0001
                                           P.O. Box 2524
14964077
           GRAPHICS YOU CAN TRUST
                                                              Borrego Springs, CA 92004
           GREATER PALM SPRINGS PRIDE
                                                329 W. Mariscal Rd.
14964079
                                                                       Palm Springs, CA 92262
                                                          4804 Laurel Canyon Blvd., #554
14964082
           GREENFIELD MANAGEMENT STRATEGIES
                                                                                            Valley Village, CA
           91607
                                     4301 W Boy Scout Blvd #800
14964084
          GREENWAY HEALTH
                                                                     Tampa, FL 33607
14964086
           GREENWAY, HUBERT
                                      Address Redacted
14964088
           GREGORY M GARRISON, A PROFESSIONAL Corp
                                                                Address Redacted
14964090
           GREGS RV SERVICE
                                    225 LINCOLN HIGHWÂY SUITE #G
                                                                           Fairless Hills, PA 19030
14964095
           GROSSMONT HOSPITAL MEDICAL STAFF
                                                         P.O. Box 158
                                                                          La Mesa, CA 91944
                                            1260 S STATE ST
14964104
           GUARD DOG SELF STORAGE
                                                                  SAN JACINTO, CA 92583
14964106
                                 8825 MURRAY DRIVE
          GUITAR CENTER
                                                            LA MESA, CA 91942
14964108
           GYNEX
                       P.O. Box 3189
                                          Redmond, WA 98073
14963976
                                 Address Redacted
          Gabriel Maldonado
14963977
           Gabriela Arechiga
                                Address Redacted
14963978
           Gabriela Barragan
                                Address Redacted
14963979
           Gabriela Garcia
                              Address Redacted
14963980
           Gabriela Leon
                            Address Redacted
14963981
           Gabriela Llamas
                              Address Redacted
14963982
           Gabriela Manzo
                              Address Redacted
14963983
           Gabriela Marentes
                                Address Redacted
14963984
           Gabriela Sanchez
                               Address Redacted
                              Address Redacted
14963985
           Gabrielle Garcia
           Gaetano Vaccaro
                               Address Redacted
14963986
14963989
           Gallagher Benefit Services, Inc
                                           500 N. Brand Boulevard Suite 100
                                                                              Michelle Gonzalez
                                                                                                    Glendale, CA
14963990
           Gallagher Benefit Services, Inc
                                           P.O. Box 95148
                                                              Chicago, IL 60694-5148
           Gallagher Insurance 2345 Grand Blvd, Suite 400
14963991
                                                               Kansas City, MO 64108
           Garden Grove Community Pharmacy
14963995
                                                                            Garden Grove Hospital Medical
                                                12601 Garden Grove Blvd
           Center
                      Garden Grove, CA 92843
14963996
                          Address Redacted
           Gary Rotto
           Gaston Del Bosque
14963997
                                 Address Redacted
14963998
           Gayla Turner
                           Address Redacted
14964001
           Gemma Robles
                             Address Redacted
14964002
           Genaro Figueroa
                              Address Redacted
           General Security Indemnity Co of Arizona
14964003
                                                     2338 W. Royal Palm Road, Suite J
                                                                                      Phoenix, AZ 85021
           General Security Indemnity Co of Arizona
14964004
                                                    28 Liberty Plaza, Suite 5400 New York, NY 10005
14964005
           General Security Indemnity Co of Arizona
                                                     One Seaport Plaza
                                                                         199 Water Street, 21st floor
                                                                                                       New
           York, NY 10038
14964006
          Genesis Gonzalez
                                Address Redacted
           Genevieve Ackerman
14964007
                                   Address Redacted
           Genevieve Romero
14964008
                                 Address Redacted
14964009
           Gennady Nosovitsky
                                  Address Redacted
14964010
          Gennifer Martinez
                                Address Redacted
14964011
           Genoveva Godinez Hernandez
                                          Address Redacted
14964013
           George Brown
                             Address Redacted
          George C. Jared, D.D.S., Inc.
George Giles Address R
14964014
                                        Address Redacted
          George Jared, D.D.S Address Redacted
George Jared, D.D.S Address Po
14964015
14964016
                                  Address Redacted
                              Address Redacted
14964017
           George Napoles
14964018
           George Zambrana
                                Address Redacted
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 27 of 76

```
14992268
           Gerald N. Sims, Esq.
                                    Susan C. Stevenson, Esq.
                                                                Pyle Sims Duncan & Stevenson, APC
                                                                                                        1620 Fifth
            Avenue, Suite 400
                                  San Diego, CA 92101
14964019
           Geraldine Hurtado
                                  Address Redacted
14964020
                                   Address Redacted
           Geraldine Morataya
14964021
           Geraldo Bravo Perez
                                    Address Redacted
14964022
           Gerarda Ruiz
                             Address Redacted
14964023
           Gerardo Ruiz
                             Address Redacted
14964024
           Gerardo Vargas
                               Address Redacted
14964025
           German Calderon
                                 Address Redacted
14964026
           German Garate
                               Address Redacted
14964028
           Ghaith Kamil
                              Address Redacted
14964029
           Gilbert Espinoza
                                Address Redacted
14964030
           Gildas Club Desert Cities
                                        73555 Alessandro Dr
                                                                 Palm Desert, CA 92260
14964032
           Giovanna Hernandez
                                    Address Redacted
14964033
           Gisela Gomez
                              Address Redacted
14964034
           Gisela Leal
                           Address Redacted
14964035
           Gisselle Padilla
                               Address Redacted
14964036
           Gizzele Victoria Valencia
                                        Address Redacted
14964037
           Gladis Perez
                             Address Redacted
14964041
           Glen Oaks Escrow
                                  1662 Hillhurst Avenue, Suite B
                                                                    Los Angeles, CA 90027
14964047
           Gloria Barba
                             Address Redacted
                                  Address Redacted
14964048
           Gloria De La Cruz
14964049
                                Address Redacted
           Gloria Manrique
14964050
           Gloria Mazuera
                               Address Redacted
14964051
           Gloria Sierra
                             Address Redacted
14964052
           Gloria Sunu
                            Address Redacted
14964058
           Golden State Water Company
                                            630 East Foothill Boulevard
                                                                           San Dimas, CA 91773-1212
14964062
           Gonzalo Cardenas
                                  Address Redacted
14964064
           Goodwin Procter LLP
                                     601 South Figueroa Street, Suite 4100
                                                                            Los Angeles, CA 90017
14964067
           Gordon & Rees LLP
                                    1111 Broadway, Suite 1700
                                                                  Oakland, CA 94607
14964070
           Graciela Juarez
                               Address Redacted
14964071
           Graciela Lopez
                               Address Redacted
           Graicee Cifuentes
14964073
                                 Address Redacted
14964075
           Grant Farmiloe
                               Address Redacted
                                                       San Jose, CA 95110
14964076
           Graphic Products
                                961 Stockton Ave
14964078
                                        15450 West Sand Street
                                                                   Victorville, CA 92392
           Greater Hope Foundation
14964080
           Greenberg Glusker Fields Claman
                                               2049 Century Park East, Suite 2600
                                                                                     Los Angeles, CA 90067
14964081
           Greenfield Management Strategies
                                                4804 Laurel Canyon Blvd
                                                                             North Hollywood, CA 91607
14964083
                                   4301 W. Boy Scout Blvd., Suite 800
           Greenway - Intergy
                                                                         Tampa, FL 33607
14964085
           Greenway Health, LLC
                                      4301 West Boy Scout Blvd, Ste 800
                                                                            Karen Mulroe
                                                                                              Tampa, FL 33607
           Greenway Health, LLC
14968312
                                      c/o Law Office of Christine E. Baur
                                                                            4653 Carmel Mountain Road
                                                                                                           Suite 308
                      San Diego, CA 92130
           #332
14964087
           Greg Rodriquez
                               Address Redacted
14964089
                               Address Redacted
           Gregory Smith
14964091
           Gricelda Ceja
                             Address Redacted
           Griselda Garcia-Castillo
                                       Address Redacted
14964092
14964093
           Griselda Romero
                                Address Redacted
14964094
           Grissel Crisostomo
                                  Address Redacted
14964096
           Grossmont Imaging
                                   8860 Center Drive, Suite 350A
                                                                    La Mesa, CA 91942
14964097
           Grow Well Crecer Bien Project – HSR
                                                    900 University Ave
                                                                           UC Riverside
                                                                                             Riverside, CA
            92521
14964098
           Guadalupe Arreola
                                  Address Redacted
           Guadalupe Castaneda Castillo
14964099
                                            Address Redacted
           Guadalupe Garcia Silva
14964100
                                      Address Redacted
14964101
                                   Address Redacted
           Guadalupe Hurtado
14964102
           Guadalupe Ibanez
                                 Address Redacted
14964103
                              Address Redacted
           Guanzhong Lo
14964105
           Guillermo Saenz Zavala
                                       Address Redacted
14964107
                                Address Redacted
           Gustavo Magana
14964110
           HAGGAR GROUP, (Now under Gentry Plaza,)
                                                            31225 La Baya Drive, Suite 206
                                                                                              Westlake Village, CA
           91362
14964113
           HALO BRANDED SOLUTIONS
                                               3182 MOMENTUM PLACE
                                                                               CHICAGO, IL 60689-5331
14964115
           HAMBURGER PATTYS
                                        9850 Owensmouth Ave # 5
                                                                      Chatsworth, CA 91311
14964117
           HAMPTON INN & SUITE
                                          7930 Jones Branch Drive
                                                                      McLean, VA 22102
                                                                             Tucson, AZ 85741
14964118
           HANATEK SOLUTIONS, LLC
                                              5234 W. Spring Willow CT
                                                              425 Market Street
14964121
           HANSON BRIDGETT, LLP, IOLTA ACCOUNT
                                                                                    26th Floor
                                                                                                   San Francisco, CA
           94105
14964122
           HAPPY CHEF
                               1520 Trousdale Dr
                                                     Burlingame, CA 94010
           HARBOR FREIGHT TOOLS
14964123
                                            1750 N IMPERIAL AVE
                                                                        EL CENTRO, CA 92243
14964124
           HARLAND CLARKE
                                      15955 LA CANTERA PARKWAY
                                                                           San Antonio, TX 78256
14964128
           HARVARD SPH
                                 677 Huntington Ave
                                                         Boston, MA 02115
14964130
           HATTON, PETRIE & STACKLER APC
                                                      12 Journey, Suite 225
                                                                               Aliso Viejo, CA 92656
                                                                Indio, CA 92201
14964134
           HAWK DESIGNS
                                  80-975 Indio Blvd. #B-11
                                                                Los Angeles, CA 90084-3951
14964136
           HB Rady Financial Services
                                          P.O. Box 843951
14964137
           HCPRO MARKETPLACE
                                         75 Sylvan St
                                                           Ste A-101
                                                                          Danvers, MA 01923
                                                           Atlanta, GA 30339
14964138
           HD SUPPLY
                             3400 Cumberland Blvd SE
14964139
           HEADSETS.COM
                                  1854 Air Lane Dr #6
                                                           Nashville, TN 37210
           HEALTH CARE LOGISTICS
14964140
                                            P.O. Box 400
                                                              Circleville, OH 43113–0400
```

```
HEALTH CENTER PARTNERS OF SOUTHERN CA
                                                               3710 Ruffin Road
                                                                                    San Diego, CA 92123
14964144
          HEALTH FINANCIAL SYSTEMS
                                             8109 Laguna Boulevard Elk Grove, CA 95758
                                                Van Nuys, CA 91410-0406
14964145
           HEALTH NET
                             P.O. Box 10406
                                                                3710 Ruffin Road
14964149
           HEALTH QUALITY PARTNERS OF SOUTHERN CA
                                                                                     San Diego, CA 92123
14964142
           HEALTH eCAREERS
                                    33292 Collection Center Dr.
                                                                 Chicago, IL 60693-0332
                                                                   NORCROSS, GA 30071
14964152
          HEALTHASYST LLC
                                    746 HOLCOMB BRIDGE RD
14964153
           HEALTHCARE LEADERSHIP SOLUTIONS
                                                        8901 West 125th Terrace
                                                                                   Overland Park, KS 66213
                                                  Cincinnati, OH 45263-9428
14964154
          HEALTHICITY
                              P.O. Box 639428
                                             3385 Seabreeze Lane
14964155
           HEALTHPLUS STAFFING, LLC
                                                                    Margate, FL 33063
                                            39615 Washington Ave.
14964158
          HEARING AID HEALTHCARE
                                                                     Palm Desert, CA 92211
           HEART PLUBING
14964159
                                 P.O. BOX 1090
                                                    Barstow, CA 92312
14964160
                                 Address Redacted
           HEATHER PATEL
14964162
          HECTOR LOMELI
                                 Address Redacted
14964170
          HELIX WATER DISTRICT-216023
                                                P.O. Box 513597
                                                                     Los Angeles, CA 90051-3597
14964171
           HELIX WATER DISTRICT-216024
                                                P.O. Box 513597
                                                                    Los Angeles, CA 90051–3597
14964172
           HELY AND WEBER
                                  1185 E Main St
                                                      Santa Paula, CA 93060
          HEMET SAN JACINTO CHAMBER OF COMMERCE
14964173
                                                               615 N San Jacinto
                                                                                     Hemet, CA 92544
                                                    Melville, NY 11747
14964176
           HENRY SCHEIN
                              135 Duryea Road
14964177
           HENRY SCHEIN, INC
                                    P.O. Box 7156
                                                      Pasadena, CA 91107-7156
          HERC RENTALS, INC P
HESHAM E MOKADEM, MD
14964179
                                    P.O. Box 2369
                                                       Atlanta, GA 31193
14964187
                                           3975 Jackson St.
                                                               Ste. 110
                                                                            Riverside, CA 92503
           HI DESERT FIRE PROTECTION SERVICES, INC.
                                                             P.O. Box 400182
                                                                                 Hesperia, CA 92340
14964188
14964191
           HIGH DESERT PONY
                                    19001 Centennial St
                                                           Hesperia, CA 92345
14964192
           HIGH TECH MAILING SERVICES
                                               P.O. Box 249
                                                               Palm Desert, CA 92261
14964195
           HILTON HOTELS & RESORT
                                           7930 Jones Branch Drive McLean, VA 22102
                                     3482 Keith Bridge Rd
           HIMFORMATICS, LLC
14964196
                                                           Ste 127
                                                                          Cummings, GA 30041
                      350 N. Orleans St., Suite S10000 Chicago, IL 60654
, LLC P.O. Box 847891 Dallas, TX 75284–7891
14964197
           HIMSS
          HIRERIGHT, LLC
14964198
          HOLDER, DAVID WILLIAM
HOLIDAY INN

2 P.O. Box 210
14964200
                                          P.O. Box 210123
14964201
                                                               Chula Vista, CA 91921
14964202
                             3 Ravinia Drive, Suite 2900
                                                           Atlanta, GA 30346
14964205
                            111 East 5th Avenue
                                                  Vancouver, BC V5T 4L1 Canada
           HOOTSUITE
14964210
                        P.O. Box 742881
                                            Los Angeles, CA 90074-2881
          HP, INC.
           HR VANTAGE POINT, LLC
14964211
                                          16206 Country Road 1735
                                                                      Lubbock, TX 79424
14964218
                      P.O. Box 1959
                                        Borrego Springs, CA 92004
           HVAC
          HVMC MEDICAL STAFF
                                       1117 East Devonshire Avenue
14964219
                                                                       Hemet, CA 92543
14964220
          HYATT
                       150 North Riverside Plaza
                                                   Chicago, IL 60606
          HYATT REGENCY HUNTINGTON BEACH RESORT
                                                                21500 PACIFIC COAST
14964221
                          HUNTINGTON BEACH, CA 92648
           HIGHWAY
                                      4111 S. Ocean Dr
14964222
           HYDE BEACH RESORT
                                                           Hollywood, FL 33019
14964109
           Hadeel Nona Juwaideh
                                   Address Redacted
15009248
                          United States Trustee's Office
           Haeji Hong
                                                         880 Front Street, Suite 3230
                                                                                      San Diego, CA 92101
14964111
           Hailey Ly
                         Address Redacted
           Haissel Campos Campos
14964112
                                     Address Redacted
14964114
           Halton Scott-Kelly
                                Address Redacted
14964116
           Hamed Firoozi
                             Address Redacted
14964119
           Hanger Prosthetics & Orthotics
                                          10910 Domain Drive, Suite 300
                                                                          Austin, TX 78758
14964120
           Hannah Noblado Address Redacted
                                 Address Redacted
14964125
           Harrsha Thirunagari
14964126
                          Address Redacted
           Harry Ilsley
           Harry Seifert
14964127
                           Address Redacted
14964129
           Hassan Pano
                           Address Redacted
                                    391 W Central Ave
14964131
           Hawatmeh Dental Group
                                                                                  Brea, CA 92821
                                                           Dr Ayed Hawatmeh
          Hawatmeh Dental Group, P.C.
Hawatmeh Dental Group, P.C.
14964132
                                         3495 E CONCOURS Unite A Ontario, CA 91764
14964133
                                                                    The Maul Firm PC-Anthony F.
                                          101 Broadway, Suite 3A
           Maul
                     Oakland, CA 94607
           Haydee Padilla
                             Address Redacted
14964135
14964143
           Health Equity, Inc
                                15 W Scenic Pointe Dr, Suite 100
                                                                 Draper, UT 84020
                         P.O. Box 9103
14964146
           Health Net
                                            Van Nuys, CA 91409-9103
           Health Net of California Refunds
14964147
                                            File #056527
                                                             Los Angeles, CA 90074-6527
14964148
           Health Net.
                          P.O. Box 10196
                                             Van Nuys, CA 91410-0196
14964150
                                                    5600 Fishers Lane
           Health Resources & SVCS Administration
                                                                         Rockville, MD 20857
14964151
           Health to Hope Clinics
                                   2880 Hulen Pl
                                                     Riverside, CA 92507
                                500 11th Avenue, North, Suite 1000
14964156
           HealthStream, Inc
                                                                   Nashville, TN 37203
                                                      Building C, Suite 207
                            211 East Six Forks Road
14964157
           HealthySteps
                                                                             Raleigh, NC 27609
14964161
           Hector Barajas
                             Address Redacted
14964163
           Hector Ramirez
                              Address Redacted
14964164
           Hector Trejo
                           Address Redacted
14964165
           Heidi Galicia
                            Address Redacted
           Heidi Gomez
14964166
                            Address Redacted
                          Address Redacted
14964167
           Heidy Cruz
14964168
           Helen Simbana
                             Address Redacted
                                    P.O. Box 249
14964169
           Helendale School District
                                                       15350 Riverview Road
                                                                                 Helendale, CA 92342
          Hemet Unified School District
                                          1791 W Acacia Ave
14964174
                                                               District Office
                                                                                   Hemet, CA 92545
                                                                   Hemet Global Medical Center
14964175
           Hemet Valley Medical Center
                                          1117 E Devonshire Ave
                                                                                                  Hemet, CA
           92543
14964178
           Herbert Harris
                            Address Redacted
14964180
          Heriberto Galvan Jr. Address Redacted
```

```
14964181
          Heritage Provider Network
                                       8510 Balboa Boulevard Suite 285
                                                                        Northridge, CA 91325
          Heritage Victor Valley Medical
                                                                       Victorville, CA 92395
14964182
                                          12370 Hesperia Rd Suite 6
14964183
          Herlinda Arambula
                                Address Redacted
          Hermelinda Pizano
                                Address Redacted
14964184
14964185
          Hernan Sanchez
                              Address Redacted
                                275 W Wisconsin Ave, Ste. 210
14964186
                                                                Administrative Office
          Herzing University
                                                                                        Milwaukee, WI
           53203
14964189
          Hiba Al-Dulaimi
                              Address Redacted
14964190
          Higgs, Fletcher & Mack LLP
                                    401 West A Street, Suite 2600
                                                                       San Diego, CA 92101
14964193
          Hilda Garcia
                          Address Redacted
14964194
          Hilda Leon
                          Address Redacted
14964199
          Hisako Sawai
                            Address Redacted
          Holman Family Counseling Inc
                                          P.O. Box 8011
14964203
                                                            Canoga Park, CA 91309
14964204
          Hooper, Lundy & Bookman, P.C.
                                            1875 Century Park East, Suite 1600
                                                                               Los Angeles, CA
           90067-2517
14964206
          Hope Through Housing Foundation
                                              9421 Haven Ave
                                                                 Rancho Cucamonga, CA 91730
          Horizon Valley Medical Group
                                          19111 Town Center Drive
14964207
                                                                     Apple Valley, CA 92308
14964208
          Hortensia Delgado
                                Address Redacted
14964209
          Hortensia Delgado
                                Address Redacted
14964212
          Hugo David Martinez Herrera
                                         Address Redacted
14964213
          Humaira Quedess
                               Address Redacted
          Humana Health Care Plans
                                     P.O. Box 931655
                                                           Atlanta, GA 31193
14964214
14964215
          Humberto Barboza Address Redacted
14964216
          Husam E. Aldairi, DDS
                                    101 Broadway, Suite 3A
                                                              The Maul Firm PC–Anthony F. Maul
                                                                                                   Oakland,
          CA 94607
14964217
          Hussain Al Aloosi
                                Address Redacted
          I2I POPULATION HEALTH
14964223
                                         377 Riverside Dr, Ste 300
                                                                    Franklin, TN 37064
                                   P.O. Box 676673
14964226
          IBM CORPORATION
                                                       Dallas, TX 75267-6673
14964227
          IBM CR LLC
                            6303 Barfield Rd
                                              Atlanta, GA 30328
14964229
          ICOMTECH.INC
                               16782 Von Karman Ave., Unit 25
                                                                 Irvine, CA 92606
          ID CARD GROUP
14964231
                                P.O. Box 23308
                                                  Tigard, OR 97281
14964232
                                     37 W 26th St 10th floor
          ID SECURITY ONLINE
                                                              New York, NY 10010
14964235
                    P.O. Box 4349
                                      Rancho Cucamonga, CA 91729
          IEHP
14964236
          IEHP - MOU PNE
                                P.O. Box 1800
                                                  Rancho Cucamonga, CA 91729
                                            10801 Sixth Street
          IEHP - SPECIALTY SERVICES
14964237
                                                                 Legal Department
                                                                                      Rancho Cucamonga, CA
           91730
14964238
          IEHP - URGENT CARE
                                     10801 Sixth Street
                                                          Legal Department
                                                                               Rancho Cucamonga, CA
           91730
14964239
          IEHP BEHAVIORAL HEALTH
                                            10801 Sixth Street
                                                                 Legal Department
                                                                                     Rancho Cucamonga, CA
          91730
14964240
          IEHP DIRECT PCP
                                 10801 Sixth Street
                                                      Legal Department
                                                                           Rancho Cucamonga, CA 91730
14964241
          IEHP Medi-Cal
                            10801 Sixth Street
                                                  Legal Department
                                                                       Rancho Cucamonga, ČA 91730
14964242
          IEHP VISION AGREEMENT
                                          10801 Sixth Street
                                                               Legal Department
                                                                                   Rancho Cucamonga, CA
          91730
14964243
          IHEANACHO EMERUWA
                                        Address Redacted
14964250
          IMPERIAL CAL PRODUCTS INC
                                              425 Apollo St
                                                               Brea, CA 92821
                                                                 Imperial, CA 92251-0937
          IMPERIAL IRRIGATION DISTRICT
14964253
                                                P.O. Box 937
                       6433 Champion Grandview Way, Building 1
14964255
                                                                  Austin, TX 78750
          INDEED
          INDEPENDENT ELECTRIC
                                         P.O. Box 1415
14964257
                                                           Thermal, CA 92274–1415
14964261
          INDIO CAR WASH, INC
                                      83360 HWY 111
                                                          Indio, CA 92201
                                                 320 Whittington PKWY
14964262
          INFO BEYOND TECHNOLOGY LLC
                                                                           STE 117
                                                                                        Louisville, KY
           40222
          INLAND EMPIRE GLASS
14964265
                                       25280 Third Street
                                                             San Bernardino, CA 92410
                                                     3579 Arlington Avenue 300
14964267
           INLAND EMPIRE OCCUPATIONAL MED
                                                                                  Riverside, CA 92506
14964269
                                                     124 W Main St, Suite 240
          INLAND VALLEY INVESTMENTS, LLC
                                                                                El Cajon, CA 92020
14964275
          INSTITUTE FOR HEALTHCARE COMMUNICATION
                                                                1010 Sync Street, Suite
                   RELIAS
                                Morrisville, NC 27560
           100
14964276
          INSTITUTE FOR HEALTHCARE IMPROVEMENTS
                                                              53 State Street, 18th Floor
                                                                                          Boston, MA
          02109
14964279
          INTEGRAL HEALTHCARE SOLUTIONS, INC
                                                         P.O. Box 1661
                                                                           Tucson, AZ 85702
14964280
          INTEGRATED HEALTH PARTNERS
                                                P.O. Box 572169
                                                                     Tarzana, CA 91357
14964283
          INTERCONTINENTAL NEW ORLEANS
                                                   444 St Charles Ave
                                                                          New Orleans, LA 70130
          INTERNATIONAL E-Z UP
                                       1900 SECOND STREET
14964286
                                                                 Norco, CA 92860
          INTERSTATE PAVING GRADING
14964287
                                              128 S Maple Ave, South
                                                                         San Francisco, CA 94080
                       2700 Coast Ave
                                          Mountain View, CA 94043
14964288
          INTUIT
          IRIS HINZO
                           3827 W. 182nd Street
14964298
                                                   Torrence, CA 90504
14964302
          IRON MOUNTAIN
                                 P.O. Box 601002
                                                     Pasadena, CA 91189-1002
          IRON MOUNTAIN – AL024
                                         P.O. Box 601002
                                                             Pasadena, CA 91189-1002
14964303
14964304
          IRON MOUNTAIN -SHRED
                                         P.O. Box 601002
                                                             Pasadena, CA 91189-1002
14964305
                   1111 Constitution Ave, NW
                                                Internal Revenue Service Federal Bldg
                                                                                      Washington, DC
          IRS
          20224
                                                                      Calgary, AB T2G 1M8 Canada
14964319
          ISTOCK
                        1240 20th Ave SE, Suite 313
                                                      Getty Images
14964321
          ISTORAGE-DESERT HOT SPRINGS
                                                 15305 Little Morongo Rd
                                                                            Desert Hot Springs, CA 92240
          IVERSON DENTAL LABS
                                        14437 Meridian Parkway
14964323
                                                                  Riverside, CA 92518
14964225
           Ian Castellanos
                             Address Redacted
14964228
          Ibtesam Mikha
                             Address Redacted
14964233
           Idalia Craycraft
                             Address Redacted
```

14964234

Iden Hankala

Address Redacted

```
14964244
           Ihuoma Chukwueke
                                   Address Redacted
           Iliana De La Cruz
14964245
                                  Address Redacted
14964246
           Iman Fobia
                            Address Redacted
14964247
           Imelda Chan
                             Address Redacted
14964248
           Imelda Chiu
                             Address Redacted
14964249
           Imelda Lorena Sosa
                                   115 Desert Shore Dr sp #5
                                                                 Thermal, CA 92274
14964251
           Imperial Health Holdings Medical Group
                                                      P.O. Box 60075
                                                                           Pasadena, CA 91106
           Imperial Irrigation District
                                         333 E. Barioni Blvd.
                                                                  Imperial, CA 92251
14964252
14964256
           InDemand Interpreting
                                      8840 Cypress Waters Blvd., Suite 300
                                                                              Dallas, TX 75019
14964254
           Inas Mousa
                            Address Redacted
14964258
           India Holmberg
                                Address Redacted
14964259
           India Holmberg
                                Address Redacted
14964260
           Indiana Pharmacy
                                  P.O. Box 396
                                                    Greensburg, IN 47240
           InfoBeyond Technologies
                                        320 Whittington Pkwy #303
14964263
                                                                        Louisville, KY 40222
14964264
           Ingrid Galvez
                              Address Redacted
14964266
           Inland Empire Health Plan
                                         10801 6th St #120
                                                                Rancho Cucamonga, CA 91730
14964268
           Inland Faculty Medical Group
                                            3990 Concours Suite 500
                                                                         Ontario, CA 91764
           Inland Valley Investments, LLC
14964270
                                              1680 W Barbara Worth Dr.
                                                                             El Centro, CA 92243
14964271
           Inland Wellness Information Network
                                                   1859 W Redlands Blvd
                                                                              San Bernardino County Medical
           Society
                        Redlands, CA 92373
           Inland Women Care
                                    9220 Haven Ave, Suite 210
14964272
                                                                  Rancho Cucamonga, CA 91730
14964273
           Inmar Rx Solutions
                                   1 W. 4th St., Suite 500
                                                             Winston-Salem, NC 27101
14964274
                         P.O. Box 58790
                                              Philadelphia, PA 19102
           Instamed
14964277
           Intact Insurance Management Liability
                                                    199 Scott Swamp Road
                                                                               Farmington, CT 06032
                                          9040 S Rita Rd, Suite 1270
14964278
           Integral Healthcare Services
                                                                         Tucson, AZ 85747
14964281
           Integrated Sings
                                1160 Pioneer Way, Suite M
                                                               El Cajon, CA 92020-1944
14964282
           Intercare Holdings Insurance Services
                                                   P.O. Box 579
                                                                     Roseville, CA 95661
                                        1111 Constitution Ave., NW
14964284
           Internal Revenue Service
                                                                        Washington, DC 20224
14964285
           Internal Revenue Service
                                        P.O. Box 7346
                                                           Insolvency Operations Unit-G.S. Moxley
                                                                                                       Philadelphia, PA
            19101-7346
14964289
           Irais Cristales
                              Address Redacted
           Irene Cervantes
                                Address Redacted
14964291
14964292
           Irene Meza
                            Address Redacted
14964293
           Irene Romero
                              Address Redacted
14964294
           Irene Velasquez
                                Address Redacted
14964295
           Irene Villarreal Aguilar
                                      Address Redacted
14964296
           Iridian Martinez
                                Address Redacted
14964297
           Iris Estrada-Jimenez
                                    Address Redacted
14964299
           Iriz Cuellar
                            Address Redacted
14964300
           Irma Alvarado
                               Address Redacted
14964301
           Irma Garcia
                            Address Redacted
           Isa Yanez
14964306
                           Address Redacted
14964307
           Isaac Algazi
                             Address Redacted
14964308
           Isaac Lomeli
                             Address Redacted
14964309
           Isabel Alvarez
                               Address Redacted
14964310
           Isabel De La Riva
                                  Address Redacted
14964311
           Isabel Duran
                             Address Redacted
14964312
           Isamara Aispuro
                                Address Redacted
14964313
           Isaura Stoffel
                              Address Redacted
14964314
           Isela Ochoa
                            Address Redacted
14964315
           Isela Sandoval
                               Address Redacted
14964316
           Isis Renteria
                             Address Redacted
14964317
           Ismael Delval
                              Address Redacted
14964318
           Issa Hanna
                           Address Redacted
14964322
           Itaty Flores
                            Address Redacted
14964324
           Ivette De La Torre
                                  Address Redacted
14964325
                             Address Redacted
           Ivette Lujano
14964326
           Ivette Torres
                             Address Redacted
14964327
           Ivette Zamora
                              Address Redacted
14964328
           Ivonne Sierra Garduno
                                      Address Redacted
14964329
           Ivy Cao
                         Address Redacted
14964330
           Ixzel Bonilla-Tule
                                  Address Redacted
14964331
           J&T Tire and Auto
                                  896 Palm Canyon Dr.
                                                            Borrego Springs, CA 92004
14964332
           J.W. DESERT SPRINGS MARRIOTT
                                                    74–855 Country Club Drive
                                                                                    Palm Desert, CA 92260
           JAIME A RODRIGUEZ
14964345
                                        Address Redacted
           JAKES PALM SPRING
                                                                    PALM SPRINGS, CA 92262
14964349
                                       664 N PALM CANYON
           JAKES TIRE AND AUTO
14964350
                                          P.O. Box 593
                                                            896 Palm Canyon Dr.
                                                                                     Borrego Springs, CA 92004
14964351
                                          1355 Shoreway Rd
                                                                 Belmont, CA 94002
           JAMECO ELECTRONICS
           JAMES L. WERMERS
JAN G SULLIVAN
14964356
                                       Address Redacted
14964359
                                    Address Redacted
14964386
           JASON KIRK
                               Address Redacted
14964391
           JAVIER LEDEZMA RODRIGUEZ D.B.A. JLR JANI
                                                                  1711 Peerless Drive
                                                                                          El Cajon, CA 92021
14964394
           JAY KYUNG LEE MD
                                       Address Redacted
           JB CABINETS & CONSTRACTION, INC
14964399
                                                        204 GREENFIELD DR. STE B
                                                                                           EL CAJON, CA 92020
14964407
           JEFFREY ZONIS
                                  Address Redacted
14964408
           JEFFRIES & CORIGLIANO LLP
                                                1560 Broadway ste 914
                                                                           New York, NY 10036
14964410
           JENNA LECOMTE-HINELY
                                             Address Redacted
                                      Address Redacted
14964426
           JENNIFER WALKER
```

```
14964433
           JERONIMO SILVA SANDOVAL
                                               Address Redacted
14964469
           JETBLUE AIRWAYS
                                    27-01 Queens Plaza N
                                                             Long Island City, NY 11101
14964470
                                     47111 Monroe St
                                                          Desert Care Network
                                                                                  Indio, CA 92201
           JFK Memorial Hospital
14964471
           JIFFY LUBE
                            P.O. Box 620130
                                                 Middleton, WI 53562
14964475
           JIM COLEMAN LTD
                                    Address Redacted
14964477
           JIMS BACKFLOW SERVICE
                                           27790 Lake Street
                                                                 Hemet, CA 92544
14964479
           JK DATA CORP
                                P.O. Box 9278
                                                   Santa Rosa, CA 95405
14964480
                                    Address Redacted
           JOAN FEDORCHUK
14964511
           JOHN WALTER
                                Address Redacted
14964514
           JOHNSON CONTROLS SECURITY SOLUTIONS
                                                              P.O. Box 371967
                                                                                   Pittsburgh, PA 15250-7967
14964515
           JOHNSTONE SUPPLY
                                      11632 NE Ainsworth Circle
                                                                    Portland, OR 97220
14964519
           JOLON SABO
                              Address Redacted
14964524
           JORDAN EQUIPMENT SERVICES
                                                 42242 Gateshead Court
                                                                           Temecula, CA 92592
14964571
           JOSEPH P TRAINER
                                    Address Redacted
14964583
                                    Address Redacted
           JOVANI MORALES
14964586
           JRebekah Hernandez
                                   Address Redacted
14964587
           JUAN A VALDEZ CERVANTES
                                               Address Redacted
14964599
           JUAN POLLO
                              Address Redacted
           JUANS ROTISSERIE CHICKEN
                                              2241 W Esplandade Ave, Ste 101C
14964607
                                                                                   San Jacinto, CA 92582
14964609
           JUDICATE WEST
                                 1851 East First Street, Suite 1600
                                                                    Santa Ana, CA 92705
14964618
           JULIAN AUTO BODY & PAINT
                                               3582 Highway 78
                                                                    Julian, CA 92036
           JULIAN CHAMBER OF COMMERCE
14964619
                                                   P.O. Box 1866
                                                                      Julian, CA 92036
           JULIAN COMMUNITY SERVICES DISTRICT
                                                           P.O. Box 681
                                                                             2656 Farmer Road
14964620
                                                                                                   Julian, CA
           92036-0681
14964621
           JULIAN COMMUNITY SVCS DISTRICT-2717 A ST
                                                                 P.O. Box 681
                                                                                   2656 Farmer Road
                                                                                                         Julian,
           CA 92036-0681
           JULIAN GOLD RUSH HOTEL
                                                              Julian, CA 92036
14964622
                                             2032 Main St
14964623
           JULIAN IRON WORKS, INC
                                           P.O. Box 1146
                                                              1840 Porter Lane, B
                                                                                     Julian, CA 92036-1146
                                            P.O. Box 2073
14964627
           JULIAN YOUTH BASEBALL
                                                              Julian, CA 92036
14964629
                                Address Redacted
           JULIE A. SMITH
                                                     4850 Pedley Rd
14964635
           JURUPA UNIFIED SCHOOL DISTRICT
                                                                         Jurupa Valley, CA 92509
14964637
           JURUPA VALLEY CHAMBER OF COMMERCE
                                                             P.O. Box 3798
                                                                                Jurupa Valley, CA 92519
                                      10489 Liberty Lane
           JUST BRAILLE SIGNS
14964638
                                                             Chisago City, MN 55013
14964333
           Jackeline Maturino
                                 Address Redacted
14964334
           Jackeline Vargas
                                Address Redacted
14964335
           Jackson & Coker
                                2655 Northwinds Pkwy
                                                          Alpharetta, GA 30009
14964336
           Jacob Kolb
                           Address Redacted
                           Address Redacted
14964337
           Jacob Perry
14964338
           Jacobo Audelia
                              Address Redacted
14964339
           Jacqueline Mendoza
                                  Address Redacted
14964340
           Jacqueline Parada
                                 Address Redacted
14964341
           Jacqueline Ramos
                                Address Redacted
14964342
           Jacqueline Thompson
                                    Address Redacted
14964343
           Jacquelyn Flores
                               Address Redacted
14964344
           Jahanzeb Ashraf
                               Address Redacted
14964346
           Jaime Gonzales
                               Address Redacted
14964347
           Jaime Lopez
                            Address Redacted
14964348
                           Address Redacted
           Jairo Rojas
                              Address Redacted
14964352
           James Bentley
                             Address Redacted
14964353
           James Everett
14964354
                             Address Redacted
           James Hebets
14964355
           James Huot
                           Address Redacted
14964357
           James Radford
                              Address Redacted
14964358
           Jamie Blakeley
                              Address Redacted
           Jana Webb
                          Address Redacted
14964360
14964361
           Janae Farmiloe
                              Address Redacted
14964362
           Janele Diaz
                           Address Redacted
14964363
           Janessa Orosco-Lopez
                                     Address Redacted
                           Âddress Redacted
14964364
           Janet Duran
14964365
           Janet Medina
                            Address Redacted
14964366
           Janet Ratniewski
                                Address Redacted
14964367
           Janette Guadarrama
                                  Address Redacted
14964368
           Janette Zetina
                             Address Redacted
14964369
           Janice Jones
                            Address Redacted
14964370
           Janice Lopez
                            Address Redacted
14964371
           Janine Krachman
                                Address Redacted
14964372
           Janna Brice
                           Address Redacted
14964373
           Janneth Contreras
                                Address Redacted
14964374
           Jaquelina Tello Botello
                                     Address Redacted
14964375
           Jaqueline Fontes Gonzalez
                                        Address Redacted
           Jarrard Phillips Cate & Hancock
14964376
                                            P.O. Box 736009
                                                                 Dallas, TX 75373-6009
                               Address Redacted
14964377
           Jarrett Saunders
14964378
           Jasmin Brown
                             Address Redacted
                                 Address Redacted
14964379
           Jasmin Maldonado
14964380
           Jasmin Ramirez Alvarez
                                      Address Redacted
14964381
           Jasmine Alvarado
                                Address Redacted
14964382
           Jasmine Barbosa
                                Address Redacted
14964383
           Jasmine Carrillo
                               Address Redacted
```

```
14964384
            Jasmine Gallardo
                                  Address Redacted
14964385
            Jason Francis
                               Address Redacted
14964387
            Jason Shelly
                              Address Redacted
                                  Address Redacted
14964388
            Jason Threadgill
14964389
            Jatzibe Valle
                              Address Redacted
14964390
            Javier Hernandez
                                  Address Redacted
14964392
            Javier Pastrana
                                Address Redacted
                                Address Redacted
14964393
            Javon M Davis
14964395
            Jayleen Picazo
                                Address Redacted
14964396
            Jayne Antolin
                               Address Redacted
14964397
            Jazmin Ramirez
                                 Address Redacted
14964398
            Jazmine Olmos
                                 Address Redacted
14964400
            Jeanette Barajas
                                 Address Redacted
14964401
            Jeanette Maldonado
                                     Address Redacted
14964402
            Jeanette Toussaint
                                   Address Redacted
14964403
            Jeannette Gonzalez
                                    Address Redacted
14964404
            Jeffrey Alvarez
                                 Address Redacted
14971707
            Jeffrey K. Garfinkle, Esq., Bar No. 153496
                                                          BUCHALTER
                                                                              18400 Von Karman Ave., Suite
                     Irvine, CA 92612
                                            Telephone: (949) 224-6254
                                                                            Email: jgarfinkle@buchalter.com
            800
14964405
            Jeffrey Mendoza
                                  Address Redacted
14971736
                                      PACHULSKI STANG ZIEHL & JONES LLP
                                                                                       10100 Santa Monica Blvd., 13th
            Jeffrey N. Pomerantz
                       Los Angeles, CA 90067-4100
            Floor
            Jeffrey Turchi
                               Address Redacted
14964406
14964409
            Jenenne Velasquez
                                    Address Redacted
14964411
            Jenna Risa LeComte-Hinely
                                             Address Redacted
14964412
            Jennie Garcia
                               Address Redacted
14964413
            Jennifer Armenta
                                  Address Redacted
14964414
            Jennifer Bonilla
                                 Address Redacted
14964415
            Jennifer Boyzo
                                Address Redacted
14964416
            Jennifer Dickinson
                                    Address Redacted
14964417
            Jennifer Espinoza
                                   Address Redacted
14964418
            Jennifer Garcia
                                Address Redacted
14964419
            Jennifer Jimenez
                                  Address Redacted
14964420
            Jennifer Morataya
                                   Address Redacted
14964421
            Jennifer Nowotney
                                    Address Redacted
14964422
                                  Address Redacted
            Jennifer Ramirez
14964423
            Jennifer Ramirez
                                  Address Redacted
14964424
            Jennifer Rasmussen
                                     Address Redacted
14964425
            Jennifer Rodriguez-Juarez
                                           Address Redacted
14964427
            Jennifer Wermers
                                   Address Redacted
14964428
            Jennifer Winesburg
                                    Address Redacted
14964429
                               Address Redacted
            Jenniffer Teal
14964430
            Jenny Duran
                              Address Redacted
14964431
            Jeremy Alcarion
                                  Address Redacted
14964432
            Jeremy Noble
                               Address Redacted
14964434
            Jerry Vega
                            Address Redacted
14964435
            Jesse Hodge
                              Address Redacted
14964436
            Jessica Alanis
                               Address Redacted
14964437
            Jessica Besgrove
                                  Address Redacted
14964438
            Jessica Cardenas
                                  Address Redacted
            Jessica Cervantes
14964439
                                  Address Redacted
14964440
            Jessica Chavez
                                Address Redacted
14964441
                                        Address Redacted
            Jessica Flores Mercado
14964442
            Jessica Garcia
                               Address Redacted
14964443
            Jessica Hernandez
                                   Address Redacted
14964444
            Jessica Jara
                             Address Redacted
14964445
                               Address Redacted
            Jessica Leyva
14964446
            Jessica Lopez
                               Address Redacted
14964447
            Jessica Martinez
                                 Address Redacted
14964448
            Jessica Martinez Rodriguez
                                            Address Redacted
14964449
            Jessica Moreno
                                Address Redacted
14964450
            Jessica Olvera
                                Address Redacted
14964451
            Jessica Sanchez
                                 Address Redacted
14964452
            Jessica Smith
                               Address Redacted
14964453
            Jessica Tarin
                              Address Redacted
14964454
            Jessica Vallejo
                                Address Redacted
14964455
            Jessie Flores
                              Address Redacted
14964456
            Jessie Guillen
                               Address Redacted
14964457
            Jessie Longoria
                                 Address Redacted
14964458
                               Address Redacted
            Jestina Garcia
14964459
            Jesus Aispuro
                               Address Redacted
14964460
                               Address Redacted
            Jesus Barajas
14964461
            Jesus Barragan
                                Address Redacted
14964462
            Jesus De Santiago
                                   Address Redacted
14964463
            Jesus Estrada
                               Address Redacted
14964464
            Jesus Manjarrez-Herrera
                                         Address Redacted
14964465
            Jesus Rangel
                              Address Redacted
14964466
            Jesus Roa-Maya
                                  Address Redacted
```

```
14964467
            Jesus Rosales
                              Address Redacted
14964468
            Jesus Sanchez
                               Address Redacted
14964472
            Jilbert Bakramian, D.D.S.
                                          Address Redacted
                                        2801 West Empire Avenue
14964473
            Jilbert Bakramian, DDS
                                                                       Bleau Fox, a PLC - Megan Ann
            Childress
                          Burbank, CA 91504
14964474
            Jilliana Dickerson
                                  Address Redacted
14964476
            Jimmy Sears
                              Address Redacted
14964478
            Jisell Torres
                             Address Redacted
14964481
            Joann Enderica
                                Address Redacted
14964482
            Joanna Gonzalez
                                 Address Redacted
14964483
            Joanna Hill
                             Address Redacted
14964484
            Joanna Ibarra
                              Address Redacted
14964485
            Joanna Ibarra
                              Address Redacted
14964486
            Joanna Martinez Sanchez
                                         Address Redacted
                                Address Redacted
14964487
            Joanna Montes
14964488
            Joanna Ocampo
                                Address Redacted
14964489
            Joanna Roldan Ortega
                                      Address Redacted
14964490
            Joanna Sung
                              Address Redacted
14964491
            Joany Saldana
                               Address Redacted
14964492
            Joaquin Loustaunau
                                    Address Redacted
14964493
            Jocelyn Mendoza Sandoval
                                           Address Redacted
14964494
            Jocelyne Gutierrez
                                   Address Redacted
14964495
                                 Address Redacted
            Jodiann Ledford
14964496
            Joe Tabarez
                             Address Redacted
14964497
            Joel Camargo
                              Address Redacted
14964498
            Joel Cazares
                              Address Redacted
            Johanna Molina
14964499
                                Address Redacted
14964500
            Johanna Valdez
                                Address Redacted
14964501
            John Aguilar
                              Address Redacted
14964502
            John Chiu
                           Address Redacted
14964503
            John Davidson
                                Address Redacted
14964504
            John Elliot
                            Address Redacted
14964505
            John Hancock
                               Address Redacted
14964506
            John Petitt
                            Address Redacted
14964507
            John Sarrouf
                              Address Redacted
14964508
            John Stoneman
                                Address Redacted
14964509
            John Stoneman
                                Address Redacted
14964510
            John Subka
                             Address Redacted
14964512
            Johnathon Jensen
                                  Address Redacted
14964513
            Johnny Garcia
                               Address Redacted
14964516
            JoiLyn Anderson
                                  Address Redacted
14964517
                             Address Redacted
            Jolien Mora
14964518
            Jolon Sabo
                            Address Redacted
14964520
            Jonathan Ballesteros
                                     Address Redacted
14964521
            Jonathan Galindo
                                  Address Redacted
14964522
            Jonathan Gonzalez
                                   Address Redacted
14964523
            Jonathan Torres Barriga
                                        Address Redacted
14964525
            Jordan Hoberman
                                  Address Redacted
14964526
            Jordan Thompson
                                  Address Redacted
14964527
            Jorge Blanco
                              Address Redacted
14964528
            Jorge Capo Rodriguez
                                      Address Redacted
14964529
            Jorge Morataya
                                Address Redacted
14964530
            Jorge Naranjo
                               Address Redacted
14964531
            Jorge Otanez Cervantes
                                        Address Redacted
14964532
            Jorge Parada
                              Address Redacted
14964533
            Jorge Reves
                             Address Redacted
14964534
            Jorge Rubio
                             Address Redacted
14964535
            Jorge Torres Chavez
                                     Address Redacted
14964536
            Josalyn Inda
                             Address Redacted
14964537
            Jose Arias Martinez
                                    Address Redacted
14964538
            Jose Banuelos
                               Address Redacted
14964539
            Jose Campos
                              Address Redacted
                             Address Redacted
14964540
            Jose Castro
14964541
            Jose Dominguez
                                 Address Redacted
            Jose Dominguez
14964542
                                 Address Redacted
14964543
            Jose Fernandez
                                Address Redacted
14964544
            Jose Fuentes
                              Address Redacted
14964545
            Jose Gomez
                             Address Redacted
14964546
            Jose Jerez
                           Address Redacted
14964547
                             Address Redacted
            Jose Lujano
14964548
            Jose Macias
                             Address Redacted
14964549
            Jose Magana
                              Address Redacted
14964550
            Jose Martinez
                               Address Redacted
14964551
            Jose Nuno
                           Address Redacted
14964552
            Jose Ortiz
                           Address Redacted
14964553
            Jose Pastrana
                              Address Redacted
14964554
            Jose Preciado
                              Address Redacted
14964555
            Jose Ramirez
                              Address Redacted
```

```
14964556
           Jose Ramirez
                              Address Redacted
14964557
           Jose Rangel
                            Address Redacted
14964558
           Jose Ricon
                            Address Redacted
14964559
           Jose Troncoso Villafana
                                       Address Redacted
14964560
           Jose Vega
                           Address Redacted
14964561
                                              Address Redacted
           Jose-Daniel Hernandez-Rivera
14964562
           Josefina Araujo
                                Address Redacted
14964563
           Josefina Armenta
                                 Address Redacted
14964564
           Joselin Pichardo
                                 Address Redacted
           Joselito Dominguez
14964565
                                   Address Redacted
14964566
           Joselyn Barba
                               Address Redacted
14964567
           Joseph Carroll
                               Address Redacted
14964568
           Joseph Dahman
                                Address Redacted
           Joseph Francis
14964569
                               Address Redacted
14964570
           Joseph Gustafson
                                 Address Redacted
14964572
            Joseph Schwartz
                                 Address Redacted
14964573
           Josephina Cota
                               Address Redacted
14964574
           Josephine Morales
                                   Address Redacted
14964575
           Joshua Ferreira
                               Address Redacted
14964576
           Joshua Mandap
                                Address Redacted
14964577
           Joshua Thompson
                                  Address Redacted
14964578
           Josselyn Vasquez
                                  Address Redacted
14964579
           Josue Castaneda
                                Address Redacted
14964580
           Josue Hernandez
                                 Address Redacted
14964581
           Josue Leon
                            Address Redacted
14964582
           Jovana Fierro
                              Address Redacted
14964584
           Jovita Ramirez
                               Address Redacted
14964585
           Joy Payne
                           Address Redacted
14964588
           Juan Almaraz
                              Address Redacted
14964589
           Juan Cabello Zuniga
                                    Address Redacted
14964590
           Juan Chavez
                             Address Redacted
                            Address Redacted
14964591
           Juan Garcia
14964592
           Juan Garcia
                            Address Redacted
14964593
           Juan Garcia Rosas
                                  Address Redacted
14964594
           Juan Lemus
                            Address Redacted
14964595
           Juan Loaiza
                            Address Redacted
14964596
           Juan Lopez
                            Address Redacted
           Juan Lucero Maduena
14964597
                                     Address Redacted
14964598
            Juan Mendoza
                               Address Redacted
14964600
           Juan Ruiz
                           Address Redacted
14964601
           Juan Sanchez
                              Address Redacted
14964602
                              Address Redacted
           Juana Araujo
14964603
           Juana Galindo
                               Address Redacted
14964604
           Juana Sanchez
                               Address Redacted
14964605
           Juana Silva
                            Address Redacted
14964606
           Juanita Williams
                                 Address Redacted
14964608
            Juany Gomez-Coria
                                    Address Redacted
           Judith Caceres
14964610
                               Address Redacted
14964611
           Judith Caro
                            Address Redacted
           Judith Hermosillo
14964612
                                  Address Redacted
14964613
                             Address Redacted
           Judy A Britt
14964614
           Julia Anguiano
                               Address Redacted
                           Address Redacted
14964615
           Julia Cole
14964616
           Julia Miranda
                              Address Redacted
14964617
           Julia Segura
                             Address Redacted
14964624
           Julian Medical Clinic
                                     2721 Washington Street
                                                                 Julian, CA 92036
                                                              Julian, CA 92036-2512
14964625
           Julian Medical Foundation
                                          P.O. BOX 2512
14964626
            Julian Thompson
                                 Address Redacted
14964628
                               Address Redacted
           Julianna Duran
14964630
           Julienne Haddad
                                 Address Redacted
14964631
           Julieta Perez
                             Address Redacted
14964632
           Julio Lopez
                            Address Redacted
           Julliana Morales
14964633
                                Address Redacted
14964634
           Juniper EX3300
                                1133 Innovation Way
                                                          c/o Juniper Networks, Inc.
                                                                                        Sunnyvale, CA 94089
                                                5565 Troth St
14964636
           Jurupa USD-Troth St Elementary
                                                                   Jurupa Valley, CA 91752
14964639
           Justyne Navarro
                                Address Redacted
14964640
           Juventino Coronado
                                    Address Redacted
            K2 AUTO DETAILING
                                                                 Ramona, CA 92065
14964641
                                        24001 Gymkhana Rd
                                 3029 Horseshoe Ln
                                                        Ste D
14964642
           K2 SCIENTIFIC
                                                                    Charlotte, NC 28208
14964643
           KAINOS PLANNING LLC
                                           355 S. Teller Street, Suite 200
                                                                            Lakewood, CO 80226
                                                             Kaplan Higher Education
                           1515 West Cypress Creek Road
14964650
           KAPLAN
                                                                                          Fort Lauderdale, FL
            33309
14964683
            KAS DENTAL LAB, INC.
                                          571 North Poplar Street #H
                                                                         Orange, CA 92868
                                                    Dallas, TX 75267-6110
14964702
                              P.O. Box 676110
            KAZOO, INC
14964703
            KBH Healthcare Consulting
                                           3283 BROKEN ARROW RD
                                                                            BORREGO SPRINGS, CA 92004
            KEITH H. RUTMAN (CSB # 144175)
15000583
                                                     501 West Broadway Ste 1650
                                                                                     San Diego, California 92101
14964707
           KEITH O. MURPHY
                                     Address Redacted
```

```
14964710
           KELLI KOMADA
                                  Address Redacted
           KELLIS ANSWERING SERVICE INC.
14964711
                                                     501 Heritage Drive
                                                                            Oxford, MS 38655
14964712
           KELLY HOHENBRINK
                                        Address Redacted
14964713
           KELLY MALDONADO
                                        Address Redacted
14964723
                         1974 RAYMOND DR
                                                   Northbrook, IL 60062
           KENSA
14964729
           KEY SURGICAL
                                 8101 Wallace Road
                                                         Eden Prairie, MN 55344
14964745
           KNIGHT SECURITY & FIRE SYSTEMS
                                                       2418 Auto Park Way
                                                                                Escondido, CA 92029
14964746
           KOMEN SAN DIEGO RACE FOR THE CURE
                                                            6116 Innovation Way
                                                                                     Carlsbad, CA 92009
14964748
           KPA Strategies, Inc
                                   100 Howe Avenue 186N
                                                               Sacramento, CA 95825
                                                    890 W. Stetson Ave, Ste B
14964749
           KPC GLOBAL MANAGEMENT, LLC
                                                                                  Hemet, CA 92543
14964750
           KPC Healthcare
                               1301 N Tustun Avenue
                                                          Santa Ana, CA 92705
                                       31276 Dunham Way
14964751
           KPSP – Local Channel 2
                                                                Thousand Palms, CA 92276
                                               31276 Dunham Way
14964752
           KPSP- Gulf California Broadcast
                                                                       Thousand Palms, CA 92276
           KUNA - FM - Gulf California Broadcast
                                                      P.O. BOX 873808
                                                                            KANSAS CITY, MS 64187
14964760
14964644
           Kaiser Foundation Health Plan, Inc.
                                                 1 Kaiser Plz 19th Floor
                                                                           Oakland, CA 94612
14964645
           Kaiser Foundation Hospitals
                                          4460 HACIENDA DR BUILIDING A 3RD FLOOR
                                                                                               Oakland, CA
           94660
14964647
           Kaiser Permanente
                                  Attn Regional Claims Recovery
                                                                    P.O. Box 741639
                                                                                         Los Angeles, CA
           90074-1639
14964646
           Kaiser Permanente
                                  P.O. Box 7004
                                                     Downey, CA 90242
14964648
           Kajsa Wakamiya
                                Address Redacted
           Kala Dharma
14964649
                             Address Redacted
14964651
                               6301 Kaplan University Avenue
           Kaplan College
                                                                 Fort Lauderdale, FL 33309
14964652
           Karen Alvarenga
                                Address Redacted
14964653
           Karen Anderson
                                Address Redacted
14964654
           Karen Davidson
                                Address Redacted
14964655
           Karen Hebets
                             Address Redacted
14964656
           Karen Hebets
                             Address Redacted
                                        Address Redacted
14964657
           Karen Hernandez Rosales
                               Address Redacted
14964658
           Karen L Lopez
14964659
           Karen Lopez
                             Address Redacted
14964660
                              Address Redacted
           Karen Morrow
           Karen Ochoa
14964661
                             Address Redacted
14964662
           Karen Ramirez Lopez
                                     Address Redacted
14964663
           Karin Moreno
                              Address Redacted
14964664
           Karin Vogler
                             Address Redacted
14964665
           Karina Arias Hernandez
                                       Address Redacted
14964666
           Karina Ayala
                             Address Redacted
14964667
           Karina Diaz
                            Address Redacted
14964668
                                       Address Redacted
           Karina Hernandez Arias
14964669
           Karina Lazaro
                              Address Redacted
14964670
           Karina Oropeza
                               Address Redacted
14964671
           Karina Pedraza
                               Address Redacted
                                  Address Redacted
14964672
           Karina Ouintanilla
14964673
           Karla Figueroa
                               Address Redacted
14964674
           Karla Gonzalez
                               Address Redacted
14964675
           Karla Jimenez Meraz
                                    Address Redacted
14964676
           Karla Magana
                              Address Redacted
14964677
           Karla Muro
                            Address Redacted
14964678
           Karla Perezn
                             Address Redacted
14964679
           Karla Soto-Rosselle
                                   Address Redacted
14964680
           Karla Vazquez
                              Address Redacted
14964681
                           Address Redacted
           Karrar Ali
14964682
           Karynne Morales
                                 Address Redacted
14964684
           Kassandra Arreola
                                  Address Redacted
14964685
           Kassandra Cervantes
                                    Address Redacted
           Katelynn Luansing
                                  Address Redacted
14964686
14964687
           Katelynn Tinoco
                                Address Redacted
14964688
           Katherine Cervantes
                                   Address Redacted
14964689
           Katherine Garay
                                Address Redacted
14964690
           Katherine Hernandez
                                    Address Redacted
14964691
           Katherine Lewis
                                Address Redacted
14964692
           Katherine Vargas
                                Address Redacted
14964693
           Katherine Wood
                                Address Redacted
14964694
           Kathleen Kortlang
                                  Address Redacted
           Kathleen Rincon
                                Address Redacted
14964695
14964696
           Kathryn Stewart
                                Address Redacted
           Kathy Brahier
                              Address Redacted
14964697
14964698
           Katia Banuelos
                               Address Redacted
14964699
           Kay Bradshaw - Smith
                                      Address Redacted
           Kayla Buenrostro
14964700
                                 Address Redacted
14964701
           Kaylee Callahan
                                Address Redacted
14964704
           Keenan Freeman
                                Address Redacted
           Keilyn Gallo Figueroa
                                     Address Redacted
14964705
14964706
           Keinya Patrick
                              Address Redacted
14964708
           Kelli Boyd
                           Address Redacted
14964709
           Kelli Hewlett
                             Address Redacted
14964714
                             Address Redacted
           Kelvin Gully
```

```
14964715
           Kelym Magallanes
                                 Address Redacted
14964716
           Kendra Coburn
                              Address Redacted
14964717
           Kenneth Acha
                             Address Redacted
           Kenneth Charp
14964718
                              Address Redacted
14964719
           Kenneth Johnson
                                Address Redacted
14964720
           Kenneth Nordlund
                                 Address Redacted
14964721
           Kenneth R Johnson
                                  Address Redacted
14964722
           Kenneth Soda
                             Address Redacted
14964724
           Kenya Roberts
                              Address Redacted
14964725
           Kenya Vargas
                             Address Redacted
14964726
           Kerry Hubbert
                             Address Redacted
14964727
           Kevin Cortez
                            Address Redacted
14964728
           Kevin Hill
                          Address Redacted
14964730
           Keyvan Malekshamran
                                     Address Redacted
14964731
           Khadija Hamisi
                               Address Redacted
14964732
           Kiara Gutierrez Paez
                                   Address Redacted
14964733
           Kiara Polanco
                             Address Redacted
14964734
           Kiara Ventura Castro
                                   Address Redacted
14964735
           Kim Tiamzon
                             Address Redacted
14964736
           Kimberly Arredondo
                                   Address Redacted
14964737
           Kimberly Briscoe
                                Address Redacted
14964738
           Kimberly Giordano
                                  Address Redacted
14964739
           Kimberly Knight
                                Address Redacted
           Kimberly Mejia
Kimberly Moreno
14964740
                               Address Redacted
14964741
                                Address Redacted
14964742
           Kimberly Skeete
                               Address Redacted
14964743
           Kingston Service Center
                                      P.O. Box 30985
                                                          Salt Lake City, UT 84130
14964744
           Kmart Pharmacy-Sears Holding MGMT Corp.
                                                          3333 Beverly Road
                                                                                Hoffman Estates, IL 60179
14964747
           Komonique Kenkay
                                  Address Redacted
14964753
           Kristal DeSantiago
                                 Address Redacted
14964754
           Kristen Lord
                            Address Redacted
14964755
           Kristin Ferry
                            Address Redacted
14964756
           Kristine Rausch
                               Address Redacted
14964757
           Kristy Turrubiate
                                Address Redacted
14964758
           Krystal Rodriguez
                                 Address Redacted
14964759
           Krystal Smith
                             Address Redacted
14964761
           Kurt Hochenauer
                                Address Redacted
14964762
           Kyle Ransom
                             Address Redacted
14970485
           Kyle S. Hirsch
                              BRYAN CAVE LEIGHTON PAISNER LLP
                                                                          2 North Central Avenue, Suite
           2100
                     Phoenix, Arizona 85004
14964764
           LA CASA DEL ZORRO
                                      3845 Yaqui Pass Rd
                                                              Borrego Springs, CA 92004
                                                            Anza, CA 92539
14964765
           LA COCINA MEXICAN
                                       56400 HWY 371
14964766
           LA Department of Public Health
                                             313 N. Figueroa Street, Room 806
                                                                                 Communications & Public
                       Los Angeles, CA 90012
14964772
           LA SALLE MEDICAL ASSOCIATES
                                                   1680 S Garfield Ave Suite 205
                                                                                   Alhambra, CA 91801
14964773
           LA SALLE MEDICAL IPA
                                         1855 Redlands Blvd 2nd Floor
                                                                         Redlands, CA 92373
                                              Burlington, NC 27216-2140
14964774
           LABCORP
                           P.O. Box 12140
14964775
           LABEL IT INC O/A POLYLABEL.COM
                                                     6691 Ellis Road
                                                                         Cambridge, ON N3C 2V4 Canada
                                                       San Francisco, CA 94139–9148
14964776
           LABOR FINDERS
                                 P.O. Box 399148
14964777
           LABREPCO
                            101 Witmer Road, Suite 700
                                                          Horsham, PA 19044
           LAC T. VU
14964778
                           5555 Reservoir Dr. Ste 205
                                                         San Diego, CA 92120
14964780
           LAIDLAW PLUMBING & AIR (DONALD LAIDLAW)
                                                                   P.O. Box 2705
                                                                                     Borrego Springs, CA
           92004
14964783
           LAKESHORE
                              2695 E. Dominguez St
                                                       Carson, CA 90895
                                     P.O. Box 96030
14964786
           LAMAR COMPANIES
                                                         Baton Rouge, LA 70896
14964792
           LASCAR ELECTRONICS
                                        4258 W 12th St
                                                            Erie, PA 16505
                                  Address Redacted
14964796
           LAURA E GOMEZ
14964803
           LAW OFFICE OF DALE L. HENDERSON
                                                       1200 Harbor Dr N Unit 6D
                                                                                    Dale Lee
                          Oceanside, CA 92054-1063
           Henderson
14964804
           LAW OFFICE OF KAREN J. SLOAT, APC
                                                        42600 Caroline Ct., Suite 101
                                                                                       Palm Desert, CA 92211
14964808
           LAYTON J PHILLIPS
                                     Address Redacted
14964809
           LAZ PARKING
                               P.O. Box 928543
                                                    San Diego, CA 92192
                                         Hartford, CT 06102-5066
14964812
           LEAF
                      P.O. Box 5066
14964813
           LEANNA G STRECKER
                                        Address Redacted
14964819
           LEMON GROVE GLASS & SUPPLY, INC.
                                                        7457 Broadway
                                                                           Lemon Grove, CA 91945
           LENNY PEPPER
14964820
                                Address Redacted
                                                      Hong Kong, China
14964821
           LENOVO
                         New Town Plaza Phase 1
           LEVEL TEN SYSTEMS
                                                       Hamburg, NJ 07419
14964838
                                      3670 NJ-94
                                                              10250 Constellation Blvd, Ste 1700
14964839
           LEVENE, NEALE, BENDER, YOO & BRILL LLP
                                                                                                  Los Angeles, CA
           90067-6200
           LG ELECTRONICS
                                                       Englewood Cliffs, NJ 07632
14964840
                                   111 Svlvan Ave
14964844
                              1336 MONTE VISTA AVE #2
                                                               UPLAND, CA 91786
           LIFE SAVING
14964845
           LIFESIGNS INC
                                2222 Laverna Ave.
                                                       Los Angeles, CA 90041
                                           1000 W. Maude Avenue
14964858
           LINKEDIN CORPORATION
                                                                      Sunnyvale, CA 94085
                                     935 N Main St
14964863
           LITTLE GRADUATES
                                                        Stafford Township, NJ 08050
                                                   78206 Varner Road Ste. B
           LOCKS AROUND THE CLOCK, INC
14964870
                                                                               Palm Desert, CA 92211
14964871
           LOEB & LOEB, LLP
                                    10100 Santa Monica Blvd., Suite 2200
                                                                           Los Angeles, CA 90067
           LOEWS CORONDO BAY
                                        4000 Loews Corondo Bay Road
14964872
                                                                          San Diego, CA 92118
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 37 of 76

```
14964873
           LOGMEIN USA, INC
                                      P.O. Box 50264
                                                          Los Angeles, CA 90074-0264
                             41695 Elm Street, #101
14964874
                                                         Murrieta, CA 92562
           LOGO JOES
14964875
           LOMA LINDA UNIVERSITY
                                              11139 Anderson St.
                                                                     Loma Linda, CA 92350
                                          7839 University Avenue # 107
                                                                           La Mesa, CA 91942
14964880
           LORENZ ADVERTISING
14964882
           LORMAN EDUCATION SERVICES
                                                    2510 Alpine Road
                                                                          Eau Claire, WI 54703
14964891
           LOW VOLT AUTOMATION TECHNOLOGIES, INC.
                                                                    3460 Marron Rd., Suite 103-372
                                                                                                       Oceanside, CA
           92056
           LOWES
                         1000 Lowes Boulevard
14964892
                                                    Mooresville, NC 28117
14964893
           LSOFT TECHNOLOGIES
                                          7177 Danton Promenade
                                                                      Mississauga, ON L5N 5P3 Canada
           LUPE PACHECO
14964915
                                  Address Redacted
14964918
           LWDA
                        455 Golden Gate Avenue
                                                     10th Floor
                                                                     San Francisco, CA 94102
14964763
           La Buena Vida Pharmacy
                                        51-335 Cesar Chavez Suite 116
                                                                           Coachella, CA 92236
14964767
           La Mesa Family Medical Group
                                              8881 Fletcher Parkway, Suite 235
                                                                                   La Mesa, CA 91942
           La Mesa OB-GYN Medical Group
14964768
                                                 8851 Center Dr Ste 210
                                                                            La Mesa, CA 91942
                                                8881 Fletcher Parkway Suite 200
14964769
           La Mesa Pediatric Medical Group
                                                                                   La Mesa, CA 91942
14964770
           La Mesa Pediatrics Medical Group
                                                 8881 Fletcher Parkway, Ste 200
                                                                                   La Mesa, CA 91942
                                                                      La Quinta, CA 92247
14964771
           La Quinta Heating & Air Conditioning
                                                    P.O. Box 557
           LaSalle Medical Associates
14964791
                                          1855 Redlands Blvd 2nd Floor
                                                                            Redlands, CA 92373
14964779
           Lady Fuentes
                              Address Redacted
14964781
           Lake Elsinore Clinica Medica Familiar
                                                    31739 Riverside Dr Suite A-1
                                                                                     Lake Elsinore, CA 92530
14964782
           Laken Moreno
                               Address Redacted
                                                              Carson, CA 90895
14964784
           Lakeshore Learning
                                   2695 E. Dominguez St
14964785
           Lali Reddy
                           Address Redacted
14964787
           Lance Harrington
                                 Address Redacted
14964788
           Lance Rosenberger
                                   Address Redacted
14964789
           Larissa Lacey
                              Address Redacted
           Lars Stensman
14964790
                               Address Redacted
14964793
           Latrecia Pledger
                                Address Redacted
14964794
           Laura Alcaraz
                              Address Redacted
14964795
           Laura Anderson
                                Address Redacted
14964797
           Laura Garcia-Martinez
                                      Address Redacted
14964798
           Laura Goetz
                             Address Redacted
14964799
           Laura Gomez
                              Address Redacted
                                Address Redacted
14964800
           Laura Manzano
                               Address Redacted
14964801
           Laura Preciado
14964802
           Lauren Carreon
                                Address Redacted
14964805
           Law Office of Rene H Pimentel
                                              4094 Brockton Ave.
                                                                      Attorney Client Trust Acct
                                                                                                    Riverside, CA
           92501
14964806
           Lawrence Foster
                                Address Redacted
14964807
           Lawyers for Employee & Consumer Rights
                                                        4100 West Alameda Avenue, Third Floor
                                                                                                   Burbank, CA
           91505
14964810
           Lazaro Renteria
                                Address Redacted
14964811
           Le Le Wang
                             Address Redacted
14964814
                             Address Redacted
           Leanne Pang
14964815
           Lee Bella Mandap
                                  Address Redacted
14964816
           Lee Davis
                           Address Redacted
14964817
           Lee Heid
                          Address Redacted
                                 Address Redacted
14964818
           Leland Barrett III
14964822
           Leobardo Vivas
                                Address Redacted
14964823
           Leonel Magarro
                                Address Redacted
           Leopoldo Tijerina
14964824
                                  Address Redacted
14964825
           Lesley Arambula
                                 Address Redacted
           Lesley Perez
                             Address Redacted
14964826
14964827
           Leslic Martinez Martinez
                                        Address Redacted
14964828
           Leslie Cruz
                            Address Redacted
14964829
           Leslie Mercado Zuniga
                                      Address Redacted
                              Address Redacted
14964830
           Leslie Myrick
14964831
           Leslie Padilla
                              Address Redacted
14964832
           Leslie Rivera
                             Address Redacted
14964833
           Leslie Velasquez
                                 Address Redacted
14964834
           Leticia Green
                              Address Redacted
14964835
           Leticia Magana-Cisneros
                                        Address Redacted
           Leticia Martinez
14964836
                                Address Redacted
14964837
           Leticia Rios
                             Address Redacted
           Lianna Rosario
14964841
                               Address Redacted
14964842
           Liesl Wagner
                              Address Redacted
           Life Point Medical Group and Urgent Care
14964843
                                                        9057 Central Avenue
                                                                                 Montclair, CA 91763
14964846
           Lilia Salcedo
                             Address Redacted
14964847
           Lilian Arias
                            Address Redacted
14964848
                               Address Redacted
           Liliana Aguilar
           Lilibeth Martinez
                                 Address Redacted
14964849
14964850
                                Address Redacted
           Lilliana Sanson
14964851
           Linda Gomez
                              Address Redacted
14964852
           Linda Jones
                            Address Redacted
14964853
           Linda Njoku
                             Address Redacted
14964854
           Linda Yanaga
                              Address Redacted
14964855
           Lindsay Lugo
                              Address Redacted
14964856
                             Address Redacted
           LingLing Shi
```

```
14964857
           Linh Ngo
                         Address Redacted
14964859
           Lisa Caspersen
                              Address Redacted
14964860
           Lisa Cravea
                            Address Redacted
           Lisa Odom
                           Address Redacted
14964861
14964862
           Lissa Romero
                             Address Redacted
14964864
           Lizbeth Colon
                             Address Redacted
14964865
           Lizbeth Hernandez De Torres
                                           Address Redacted
14964866
                                     Address Redacted
           Lizbeth Moreno Zepeda
14964867
           Lizeth Martinez
                               Address Redacted
14964868
           Lizette Gonzalez
                                Address Redacted
14964869
           Lobna Yaqo
                            Address Redacted
14964876
           Lopez and Noyes Dental
                                      230 Landis Ave
                                                          Chula Vista, CA 91910
14964877
           Lorena Diaz
                            Address Redacted
14964878
           Lorena Lara
                            Address Redacted
14964879
           Lorena Newsom
                               Address Redacted
14964881
           Loriann Villalobos
                                  Address Redacted
14964883
                              Address Redacted
           Lorraine Salas
14964884
           Louis B Alexander & Laura R Alexander
                                                     Address Redacted
14964885
           Lourdes Enriquez
                                Address Redacted
14964886
           Lourdes Hernandez
                                  Address Redacted
14964887
           Lourdes Lopez Gomez
                                     Address Redacted
14964888
           Lourdes Quiroz Gomez
                                     Address Redacted
14964889
           Lourdes Valdez
                               Address Redacted
14964890
           Loussa Anjejo
                              Address Redacted
14964894
           Lubna Hanna
                             Address Redacted
14964895
           Lucero Cruz
                            Address Redacted
           Lucia Hernandez
14964896
                                Address Redacted
14964897
           Lucia Morataya
                               Address Redacted
           Luciano Madrigal-Lopez
14964898
                                       Address Redacted
14964899
                             Address Redacted
           Lucila Beltran
14964900
           Lucila Beltran-Beltran
                                     Address Redacted
14964901
           Lucille Aceves
                              Address Redacted
14964902
           Lucille Loa
                           Address Redacted
14964903
           Luis A Margarito Julian
                                      Address Redacted
14964904
           Luis Carias
                           Address Redacted
14964905
           Luis Cervantes Gomez
                                     Address Redacted
14964906
           Luis F Rojas Uribe
                                  Address Redacted
14964907
           Luis Fernandez
                               Address Redacted
14964908
           Luis Jimenez Garcia
                                   Address Redacted
14964909
           Luis Troncoso
                              Address Redacted
14964910
           Luisa Lopez
                            Address Redacted
14964911
           Luisa Rodriguez
                                Address Redacted
14964912
           Luisa Samper
                             Address Redacted
14964913
                       100 CenturyLink Drive
           Lumen
                                                 Monroe, LA 71203
14964914
           Lupe Pachecho
                              Address Redacted
14964916
           Lupita Macias
                             Address Redacted
14964917
           Lurriena Lucero
                               Address Redacted
14964919
           Lyle Cook
                          Address Redacted
14964920
           Lynetta Beltran
                               Address Redacted
14964921
           Lynette Carbajal
                               Address Redacted
14964922
           Lynn Alexander
                               Address Redacted
14964923
           Lyssete Sanchez
                               Address Redacted
14964924
           M Carmen Santibanez
                                     Address Redacted
14964925
           M&M DISTRIBUTION
                                      P.O. BOX 122
                                                         KECHI, KS 67067
14964926
           M&M DISTRIBUTION, LLC
                                           2403 E. Loper St.
                                                                 Wichita, KS 67219
14964927
           M. Del Rosario Escobar
                                      Address Redacted
14964934
           MACDONALD SERVICE GROUP/FLOOD RESPONSE
                                                                                              Thousand Palms, CA
                                                                    27-104 Corporate Way
14964937
           MACGILL
                           1000 N. Lombard Rd.
                                                    Lombard, IL 60148
           MACRO HELIX LLC
14964939
                                     P.O. Box 742256
                                                          Atlanta, GA 30374-2256
                                954 W. Highland Avenue
14964943
           MADERO, INC.
                                                            Redlands, CA 92373
14964956
           MAGELLA MEDICAL GROUP, INC
                                                   P.O. Box 281034
                                                                       Atlanta, GA 30384-1034
14964957
           MAGELLAN BEHAVIORAL HEALTH
                                                     300 Continental Boulevard Suite 240
                                                                                           El Segundo, CA
           90245
           MAGTEK
14964960
                          1710 Apollo Court
                                                 Seal Beach, CA 90740
                                                         Santa Ana, CA 92704
           MANAMED, INC.
14964967
                                  1511 W Alton Ave.
                                                         3950 S LAS VEGAS BLVD
14964969
           MANDALAY BAY RESORT AND CASINO
                                                                                        LAS VEGAS, NV
           89119
14964971
           MANIFEST MEDEX
                                    6001 Shellmound Street, Suite 500
                                                                        Emeryville, CA 94518
14965026
           MARIA ALVAREZ
                                   Address Redacted
14965032
           MARIA BARRERA
                                   Address Redacted
14965072
                                 Address Redacted
           MARIA FLORES
           MARIA G MORENO GARCIA
14965075
                                             Address Redacted
14965091
           MARIA HERNANDEZ
                                      Address Redacted
14965111
           MARIA MARTINEZ
                                    Address Redacted
14965129
                                 Address Redacted
           MARIA OCHOA
14965132
           MARIA P SALDIVAR
                                     Address Redacted
           MARIBEL RAMIREZ
                                     Address Redacted
14965181
```

```
14965232
           MARK DAVIDSON PHOTOGRAPHY
                                                 555 S. Sunrise Way, Suite 210
                                                                                 Palm Springs, CA 92264
          MARK DUEBNER DESIGN 2730 Sunrise Sonata Lane Palm Springs, CA 92262 MARRIOTT HOTELS 7750 Wisconsin Avenue Bethesda, MD 20814
14965233
14965248
           MARSH & MCLENNAN AGENCY
14965249
                                               62886 Collection Center Drive
                                                                               Chicago, IL 60693-0628
14965254
           MARTEN L PHILIP
                                 Address Redacted
14965264
           MARTHA MUNIZ
                                 Address Redacted
14965274
           MARTHAS VILLAGE AND KITCHEN
                                                  83791 Date Ave
                                                                      Indio, CA 92201
           MARTIN MERCADO
14965282
                                   Address Redacted
14965290
           MARY E. FOLEY
                                Address Redacted
           MARY OLIVERE
14965297
                                Address Redacted
           MASTER CARE COMMERCIAL JANITORIAL
                                                                                Palm Springs, CA 92262
32 Hesperia, CA 92345
14965309
                                                          3350 Savanna Trail
                                                      16666 Smoketree St. Ste. B2
14965311
           MASTERS REFRESHMENT SERVICES, LLC
                                              81410 US HIGHWAY 111
14965313
           MATHISBROTHERS FURNITURE
                                                                           Indio, CA 92201
14965323
           MAUREEN FLEMMING-MULLINS
                                                Address Redacted
14965349
                                  60 Crossways Park Drive West, Suite 301
           MAZARS USA LLP
                                                                          Woodbury, NY 11797-9816
14965427
           MCKESSON RITE AID 5670
                                          5670 East 2nd Street
                                                             Long Beach, CA 90803
14965478
                                          68 South Service Rd, Suite 100
           MDamerica Supply Group Inc
                                                                      Melville, NY 11747
14965480
           MEARS TRANSPORTATION
                                          324 W Gore St
                                                           Orlando, FL 32806
14965483
           MED TECH SOLUTIONS
                                       P.O. Box 748882
                                                           Los Angeles, CA 90074-8882
           MEDCENTERDISPLAY, LLC
                                          11408 Otter Creek South
14965484
                                                                     Mabelvale, AR 72103
14965485
           MEDIA FIRE
                            21175 State Highway 249
                                                       Houston, TX 77070
           MEDICAID SCHIP MSDA
                                       4411 Connecticut Avenue, N.W., #401
14965487
                                                                              Washington, DC 20008
14965489
           MEDICAL GROUP MANAGEMENT ASSOCIATION
                                                               104 Iverness Terrace E
                                                                                        Englewood, CO
14965490
           MEDICAL STAFF FUND
                                       3865 Jackson Street
                                                             Riverside, CA 92503
14965492
           MEDPRO GROUP
                                5814 Reed Rd
                                                  Fort Wayne, IN 46835
14965507
           MELISSA BUSIC
                                Address Redacted
14965525
           MELTWATER
                             P.O. BOX 23721
                                                 14005 LIVE OAK AVE
                                                                           IRWINDALE, CA 91706
14965526
           MELTWATER NEWS US INC.
                                          Dept LA 23721 Pasadena, CA 91185–3721
14965529
           MENTAL HEALTH NETWORK (MHN)
                                                   2370 Kerner Blvd
                                                                       San Rafael, CA 94901
          MERCER (US) Inc.

MERCHAIN CATERING

CO INC

MERCHAIN CATERING

3685 N. Jumpero ...

2000 Galloping Hill Road

Amherst, NY
14965533
                                                   Newark, NJ 07188-0793
14965536
                                     3685 N. Junipero Ave, #2 Palm Springs, CA 92262
14965537
          2000 C
P.O. Box 1652
MERITAIN, INC
META LINE CO
                                                           Mailstop K1–1004A Kenilworth, NJ 07033
14965544
                                           Amherst, NY 14226
14965546
                               P.O. Box 1652
                                                 Amherst, NY 14226
14965548
           META LIND SOLUTIONS, INC
                                            433 Sandra Place
                                                                Eagle, ID 83616
                                            2733 Via Orange Way Ste 103
14965553
           METRO FIRE & SAFETY INC.
                                                                           Spring Valley, CA 91978
14965560
           MFI MEDICAL EQUIPMENT, INC.
                                               10695 Treena St., Ste. 105
                                                                           San Diego, CA 92131
                                               3799 LAS VEGSA BLVD
14965561
           MGM GRAND HOTEL & CASINO
                                                                           Las Vegas, NV 89109
14965562
                                                  Englewood, CO 80112-5306
           MGMA
                      104 Inverness Terrace East
           MICHAEL A CARUSO
14965563
                                    Address Redacted
14965578
           MICHAEL RAUM
                                Address Redacted
14965602
           MICRODAQ.COM
                                 29 Pembroke Rd
                                                     Concord, NH 03301
14965603
           MICROSOFT
                           One Microsoft Way
                                                  Redmond, WA 98052
           MILLARS PLUMBING AND BACKFLOW TESTING

33485 Wild Lilac Rd.

Piverside CA 92504
14965626
                                                                                       Menifee, CA 92584
                                                                Riverside, CA 92504
           MINT CONDITION-ING
                                     6859 Arlington Avenue
14965629
14965630
           MIRACLE SPRINGS RESORT & SPA
                                                 10625 PALM DRIVE
                                                                         DESERT HOT SPRINGS, CA
           92240
14965631
           MIRACLES AND DREAMS FOUNDATION
                                                       3230 Vine Street, Suite 200
                                                                                    Riverside, CA 92507
           MIRAGE GAS STATION 755 PALM CANYON DRIVE
                                                                     BORREGO SPRINGS, CA 92004
14965632
                                                         Las Vegas, NV 89109
                               3400 S Las Vegas Blvd
14965633
           MIRAGE HOTEL
14965643
           MIRION TECHNOLOGIES
                                       P.O. Box 101302
                                                            Pasadena, CA 91189-0005
           MOBILE HEALTH CLINICS ASSOCIATION
                                                        2275 Schuetz Rd
14965649
                                                                           Saint Louis, MO 63146
                                                         La Quinta, CA 92253
14965650
          MOBILE LUBE EAT INCOME.

MOBILE MINI, LLC- CA P.O. Box 7144 Pasaucing
MODERN OPTICAL INTERNATIONAL 56 Chancellor

12742 Limonite Ave

Rars
           MOBILE LUBE EXPRESS
                                       P.O. Box 1146
14965651
                                                         Pasadena, CA 91109-7144
                                                   56 Chancellor Dr
14965652
                                                                       Roselle, IL 60172
                                                                    Corona, CA 92880
14965656
                                      1010 WEST MAIN
14965663
           MOJAVE AUTO GROUP
                                                            Barstow, CA 92311
14965665
           MOLINA HEALTH CARE-SAN DIEGO
                                                    200 Oceangate, Suite 100
                                                                              URGENT CARE
           AGREEMENT Long Beach, CA 90802
MOLINA HEALTHCARE OF CA 200
14965667
                                             200 Oceangate, Suite 100
                                                                        Long Beach, CA 90802
                                                       P.O. Box 30567
           MOLINA HEALTHCARE OF CALIFORNIA
14965669
                                                                         Los Angeles, CA 90030–0567
           MOLINA HEALTHCARE, CA BEHAVIORAL HEALTH
14965670
                                                                  200 Oceangate, Suite 100
                                                                                           Long Beach, CA
           90802
           MONSTER.COM
14965689
                               133 Boston Post Road
                                                        Weston, MA 02493
14965690
           MOORE MEDICAL
                                  P.O. Box 99718
                                                     Chicago, IL 60696
           MOUTHWATCH
                               460 MAIN STREET STE B
14965694
                                                          Metuchen, NJ 08840
           MPOWER TELEPACIFIC-103679
                                            P.O. Box 509013 San Diego, CA 92150–9013
14965697
14965695
                                 303 Colorado St. Suite 2075
                                                           Steve Wexler Austin, TX 78701
           MPower TelePacific
14965699
           MR. CRANE
                           647 N. Hariton St
                                                Orange, CA 92868
14965700
           MR. KABOB
                            Address Redacted
14965701
           MR. ROOTER PLUMBING
                                        P.O. Box 3385
                                                          Palm Desert, CA 92261
                                    2281 W. Eslanade Ave, 103B
66575 Second St. Desert Hot S
14965702
           MR. YOU CHINESE FOOD
                                                                      San Jacinto, CA 92582
                                                        Desert Hot Springs, CA 92240–3711
Desert Hot Springs, CA 92240–3711
           MSWD 26-071001-14
14965703
           MSWD 26-192001-12
                                    66575 Second St.
14965704
                                   66575 Second St.
14965705
           MSWD 26-192002-1
                                                       Desert Hot Springs, CA 92240-3711
           MULLEN COUGHLIN, LLC
14965707
                                     426 W. Lancaster Avenue, Suite 200 Devon, PA 19333
          MUSHROOM NETWORKS, INC
                                           5703 Oberlin Dr. Suite 208
                                                                        San Diego, CA 92121
14965710
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 40 of 76

```
14965714
           MY WHITE BOARDS
                                       220 Cherry Street
                                                             Shrewsbury, MA 01545
14965715
            MYHRSCREENS
                                  401 Legacy Park, Suite B
                                                               Ridgeland, MS 39157
14965722
            MYRON ZAMORA
                                    Address Redacted
                                        Address Redacted
14964928
            Ma De Lourdes B Garcia
14964929
            Ma G Botello
                              Address Redacted
14964930
           Ma Guadalupe Diaz
                                    Address Redacted
14964931
            Ma Guadalupe Gonzalez Molinar
                                                Address Redacted
                                 Address Redacted
14964932
            Ma Soto-Munoz
14964933
            Macario Avila
                               Address Redacted
            Mace Al-Waidh
14964935
                                 Address Redacted
14964936
            Macedonio Orozco
                                   Address Redacted
14964938
            Maclovio Garcia
                                 Address Redacted
14964940
            Madeline Tucker
                                 Address Redacted
14964941
                                 Address Redacted
            Madelrosa Toral
14964942
                                Address Redacted
            Madelyn Milon
14964944
            Madison Curtiss
                                Address Redacted
14964945
            Mae Quinagon
                               Address Redacted
14964946
            Maeki Ajei
                            Address Redacted
14964947
            Maelil Ajei
                            Address Redacted
14964948
            Magali Albarran
                                 Address Redacted
14964949
            Magali Cardenas
                                 Address Redacted
14964950
            Magaly Velasquez, D.D.S
                                         Address Redacted
14964951
            Magaly Velasquez, DDS
                                        2801 West Empire Avenue
                                                                      Bleau Fox, a PLC - Megan Ann
            Childress
                          Burbank, CA 91504
14964952
            Magdalena Armendariz
                                       Address Redacted
14964953
            Magdalena Morales
                                    Address Redacted
14964954
            Magdaleno Salomon
                                    Address Redacted
14964955
            Maged Ibrahim
                               Address Redacted
14964958
            Magellan Health Services of California
                                                     300 Continental Boulevard Suite 240
                                                                                            El Segundo, CA
            90245
14964959
            Magnolia Clinica Medica Familiar
                                                 9939 Magnolia Ave
                                                                         Riverside, CA 92503
14964961
            Mai Nguyen
                             Address Redacted
14964962
                               Address Redacted
            Majida Tawfiq
14964963
            Makadia & Shah
                                 115 N McKinley St Ste 105
                                                                Corona, CA 92879
14964964
            Malcom Lambert
                                 Address Redacted
14964965
            ManageEngine
                               4141 Hacienda Drive
                                                        Zoho Corporation
                                                                               Pleasanton, CA 94588
14964966
            Manal Alalaq
                              Address Redacted
                                                                     Pomona, CA 91768
14964968
            Manar Jamal-Fontana & Hemet
                                              2005 W Holt Ave
14964970
            Mangaljit Singh
                                Address Redacted
14964972
            Manuel Carbajal
                                 Address Redacted
14964973
            Manuel Chico
                               Address Redacted
14964974
            Manuel Hernandez
                                   Address Redacted
14964975
            Manuel Salinas-Aispuro
                                        Address Redacted
14964976
           Manuel Sandoval
                                 Address Redacted
14964977
            Manuel Soto Valanzuela
                                        Address Redacted
            Manuel Soto Valenzuela
14964978
                                        Address Redacted
14964979
            Manuela Aguilar
                                 Address Redacted
14964980
            Manuela Rubio Aguilar
                                       Address Redacted
14964981
            Marbella Luque Verdugo
                                        Address Redacted
14964982
            Marc E Stevenson
                                  Address Redacted
14964983
            Marcela Andersen
                                  Address Redacted
14964984
            Marcela Martinez
                                  Address Redacted
14964985
            Marcela Miranda Rodriguez
                                           Address Redacted
14964986
            Marcela Tapia
                               Address Redacted
14964987
            Marcelina Ortega
                                 Address Redacted
14964988
            Marcelino Lopez
                                 Address Redacted
14964989
            Marcellus Omar-Joya
                                      Address Redacted
14964990
            Marcelo Garcia
                                Address Redacted
14964991
            Marcelo Toledo D.D.S., Inc.
                                           401 West A Street, Suite 1900
                                                                             Wilson Elser-Robert W. Harrison
                                                                                                                 San
            Diego, CA 92101
14964992
            Marcelo Toledo D.D.S., Inc.
                                           Address Redacted
14964993
            Marcelo Toledo D.D.S., Inc.
                                           Address Redacted
           Marcelo Toledo, DDS
14964994
                                      401 West A Street, Suite 1900
                                                                       Wilson Elser-Robert W. Harrison
                                                                                                            San Diego,
            CA 92101
14964995
            Marcial Ceballos
                                 Address Redacted
14964996
            Marciano Herrera
                                  Address Redacted
14964997
            Marco Esquivel
                                Address Redacted
            Marco Gaona De La Torre
14964998
                                         Address Redacted
14964999
            Marco V Narcisco
                                  Address Redacted
14965000
            Marcos Cesena
                                Address Redacted
14965001
            Marcus Epstein
                                Address Redacted
14965002
            Margaret Keenan
                                 Address Redacted
14965003
            Margaret Smith
                                Address Redacted
                                          Address Redacted
14965004
            Margarita Arroyo Enriquez
14965005
            Margarita Carpenter
                                    Address Redacted
14965006
            Margarita Castro
                                 Address Redacted
14965007
            Margarita Estrada
                                  Address Redacted
14965008
           Margarita Hernandez Del Julian
                                              Address Redacted
```

```
Address Redacted
14965009
            Margarita Muralles
14965010
            Margarita Robles Martinez
                                          Address Redacted
14965011
            Margarita Rodriguez
                                     Address Redacted
14965012
            Margarita Salazar
                                  Address Redacted
14965013
            Margarita Saucedo
                                   Address Redacted
14965014
            Margarita Tinoco
                                  Address Redacted
14965015
            Margarita Tobar
                                 Address Redacted
            Margarito Flores
14965016
                                 Address Redacted
14965017
            Margot Cardona
                                 Address Redacted
14965018
            Mari Barosav
                              Address Redacted
14965019
            Mari Zahroon
                               Address Redacted
14965020
            Maria A Rodriguez
                                   Address Redacted
14965021
            Maria Abrica
                              Address Redacted
            Maria Aguilar
14965022
                               Address Redacted
14965023
            Maria Aguillar
                               Address Redacted
14965024
            Maria Alonzo
                               Address Redacted
14965025
            Maria Alvarez
                               Address Redacted
14965027
            Maria Alvarez
                               Address Redacted
14965028
            Maria Amaya
                              Address Redacted
14965029
            Maria Angelica Rodriguez
                                          Address Redacted
            Maria Ayala
                             Address Redacted
14965030
                                 Address Redacted
14965031
            Maria B Navarro
14965033
            Maria Bernal
                              Address Redacted
14965034
            Maria Bernal Hernandez
                                        Address Redacted
14965035
            Maria Bernardo
                                Address Redacted
14965036
            Maria Bibian
                              Address Redacted
14965037
            Maria Bristol
                              Address Redacted
14965038
            Maria C Gallardo
                                  Address Redacted
14965039
            Maria Carrillo
                               Address Redacted
14965040
            Maria Castro
                              Address Redacted
14965041
            Maria Castro Leon
                                   Address Redacted
14965042
            Maria Ceja
                            Address Redacted
            Maria Cienfuegos
14965043
                                  Address Redacted
14965044
                                Address Redacted
            Maria Cisneros
14965045
            Maria Colunga
                                Address Redacted
14965046
            Maria Concecion Vallejo
                                         Address Redacted
14965047
            Maria Corrales
                                Address Redacted
14965048
            Maria Cruz
                            Address Redacted
14965049
            Maria Cueva
                              Address Redacted
14965050
            Maria De Del Rio
                                  Address Redacted
14965053
            Maria De Jesus Ramos
                                       Address Redacted
14965054
            Maria De Jesus Torres
                                      Address Redacted
14965055
            Maria De La Luz Medina Amor
                                               Address Redacted
            Maria DePilar Vargas Loza
14965058
                                           Address Redacted
14965056
            Maria Del Pilar Meza
                                      Address Redacted
14965057
            Maria Delgado Galindo
                                       Address Redacted
14965059
            Maria Donoza
                               Address Redacted
            Maria E Baltazar
14965060
                                 Address Redacted
14965061
            Maria E Lara
                              Address Redacted
14965062
            Maria E Salazar
                                Address Redacted
            Maria Elena Gutierrez
14965063
                                      Address Redacted
14965064
            Maria Esparza
                               Address Redacted
14965065
            Maria Esquivel
                                Address Redacted
14965066
            Maria Esquivel-Gonzalez
                                          Address Redacted
14965067
            Maria F Escobar Martinez
                                          Address Redacted
14965068
            Maria Felix Gomez
                                    Address Redacted
14965069
            Maria Fernandez
                                 Address Redacted
14965070
            Maria Figueroa
                                Address Redacted
14965071
            Maria Flores
                              Address Redacted
14965073
            Maria Fonseca
                               Address Redacted
14965074
            Maria G Moreno Garcia
                                        Address Redacted
14965076
            Maria G Ortiz
                               Address Redacted
14965077
            Maria Gallardo
                                Address Redacted
14965078
            Maria Garcia
                              Address Redacted
14965079
            Maria Garcia
                              Address Redacted
            Maria Garcia Ramirez
14965080
                                      Address Redacted
14965081
            Maria Garcia-Perez
                                    Address Redacted
14965082
            Maria Gonzalez
                                Address Redacted
14965083
            Maria Gradilla
                               Address Redacted
14965084
            Maria Guadalupe Camargo
                                           Address Redacted
14965085
            Maria Guadalupe Rosas
                                        Address Redacted
14965086
            Maria Guerra-Ayala
                                     Address Redacted
14965087
            Maria Guillen
                               Address Redacted
14965088
            Maria Gutierrez
                                Address Redacted
14965089
            Maria Hernandez
                                  Address Redacted
14965090
            Maria Hernandez
                                  Address Redacted
14965092
            Maria Herrera
                               Address Redacted
14965093
            Maria Josefina Calderas Partida
                                               Address Redacted
```

```
14965094
            Maria L Santos
                                Address Redacted
14965095
            Maria Lomeli
                               Address Redacted
14965096
                              Address Redacted
            Maria Lopez
14965097
            Maria Lopez Estrada
                                     Address Redacted
14965098
            Maria Lopez Hernandez
                                        Address Redacted
            Maria Loyola
14965099
                              Address Redacted
14965100
            Maria Lucas-Gonzalez
                                       Address Redacted
            Maria Luisa Godinez Carrillo
14965101
                                             Address Redacted
14965102
            Maria M Rodriguez
                                    Address Redacted
14965103
            Maria M Silva
                               Address Redacted
14965104
            Maria Madrigal
                                Address Redacted
14965105
            Maria Madrigal
                                Address Redacted
14965106
            Maria Magana
                                Address Redacted
            Maria Magee
14965107
                              Address Redacted
14965108
            Maria Mancilla
                                Address Redacted
14965109
            Maria Manzo
                              Address Redacted
14965110
            Maria Martin
                              Address Redacted
14965112
            Maria Martinez
                                Address Redacted
14965113
            Maria Mata
                             Address Redacted
14965114
            Maria Mata Sanchez
                                     Address Redacted
14965115
                               Address Redacted
            Maria Medina
14965116
            Maria Mejia
                             Address Redacted
            Maria Melendrez
14965117
                                  Address Redacted
14965118
            Maria Mendoza-Ibarra
                                       Address Redacted
14965119
            Maria Meraz
                              Address Redacted
14965120
            Maria Morales
                                Address Redacted
                             Address Redacted
14965121
            Maria Motta
14965122
            Maria Moya
                             Address Redacted
14965123
            Maria Mujica
                              Address Redacted
14965124
            Maria Natividad
                                 Address Redacted
14965125
            Maria Navarro
                                Address Redacted
14965126
            Maria Ocampo De La Torre
                                           Address Redacted
14965127
            Maria Ochoa
                              Address Redacted
14965128
            Maria Ochoa
                              Address Redacted
                                Address Redacted
14965130
            Maria Oliveros
14965131
            Maria Ottinger
                                Address Redacted
14965133
            Maria Placencia
                                 Address Redacted
14965134
            Maria Posadas
                                Address Redacted
14965135
            Maria Prado
                             Address Redacted
            Maria Quintana
14965136
                                Address Redacted
14965137
            Maria Quiroga
                                Address Redacted
            Maria Rafael
14965138
                              Address Redacted
14965139
            Maria Rangel
                              Address Redacted
14965140
            Maria Reves
                             Address Redacted
14965141
            Maria Reyes
                              Address Redacted
14965142
            Maria Robles Montellano
                                         Address Redacted
14965143
            Maria Rodriguez
                                 Address Redacted
            Maria Rodriguez
                                 Address Redacted
14965144
14965145
            Maria Rodriguez
                                  Address Redacted
14965146
            Maria Rojas
                             Address Redacted
14965147
            Maria Romero
                                Address Redacted
14965148
            Maria Rueda
                              Address Redacted
14965149
            Maria Ruiz
                            Address Redacted
14965150
            Maria S Millan Juarez
                                      Address Redacted
            Maria Sanchez
14965151
                               Address Redacted
14965152
            Maria Sanchez
                                Address Redacted
14965153
            Maria Sanchez
                                Address Redacted
14965154
            Maria Sandobal Cortez
                                       Address Redacted
14965155
            Maria Santa Maria
                                   Address Redacted
14965156
            Maria Sara Graciano
                                     Address Redacted
14965157
            Maria Segura
                              Address Redacted
14965158
            Maria Servin
                              Address Redacted
14965159
            Maria Teran
                             Address Redacted
14965160
            Maria Torres
                              Address Redacted
            Maria Torres Ibarra
14965161
                                    Address Redacted
14965162
            Maria Torres Rojas
                                    Address Redacted
14965163
            Maria Uribe
                             Address Redacted
14965164
            Maria Valdez
                              Address Redacted
14965165
            Maria Villareal
                                Address Redacted
14965166
            Maria Villasenor
                                 Address Redacted
14965167
            Maria Zea
                           Address Redacted
            Maria de Jesus Garcia Fregozo
                                              Address Redacted
14965051
14965052
            Maria de Jesus Hernandez Molina
                                                 Address Redacted
14965168
            Mariam Abdallah
                                  Address Redacted
14965169
            Mariam Najjar
                                Address Redacted
            Mariam Polus
                               Address Redacted
14965170
14965171
            Mariana Cervantes
                                   Address Redacted
14965172
            Mariana Fernandez
                                   Address Redacted
```

```
14965173
            Mariana Hernandez
                                    Address Redacted
14965174
            Mariana Mansour
                                  Address Redacted
14965175
            Marianne Padilla
                                 Address Redacted
14965176
            Marianne Rhodes
                                  Address Redacted
14965177
            Maribel Fleming
                                 Address Redacted
           Maribel Gamez
14965178
                                Address Redacted
14965179
            Maribel Lopez-Cruz
                                     Address Redacted
14965180
            Maribel Macias Ortiz
                                     Address Redacted
14965182
            Maribel Rios
                              Address Redacted
14965183
           Maribel Rivera
                                Address Redacted
14965184
            Maribel Torres
                                Address Redacted
14965185
            Maribel Trujillo-Rodriguez
                                           Address Redacted
14965186
            Maribel Valdivia
                                 Address Redacted
14965187
            Maricarmen Aguirre
                                     Address Redacted
            Maricel Delacruz
14965188
                                  Address Redacted
14965189
            Maricela Calderon
                                   Address Redacted
14965190
            Maricela Calderon
                                   Address Redacted
14965191
            Maricela Chavarria
                                   Address Redacted
14965192
            Maricela Estock
                                 Address Redacted
14965193
            Maricela Palacio
                                 Address Redacted
14965194
                               Address Redacted
            Maricela Soto
14965195
            Mariel De Leon
                                Address Redacted
14965196
            Mariel Hernandez
                                  Address Redacted
14965197
            Mariela Felix
                              Address Redacted
                               Address Redacted
14965198
           Marina Cortez
14965199
            Marina Cortez Verdugo
                                       Address Redacted
14965200
            Marina Dsouza
                                Address Redacted
14965201
            Marina Ortuno
                               Address Redacted
14965202
           Marina Vazquez
                                 Address Redacted
            Marino D Carreno
14965203
                                   Address Redacted
14965204
            Mario A Benitez
                                 Address Redacted
14965205
            Mario Feria
                             Address Redacted
14965206
           Mario Garcia
                              Address Redacted
14965207
            Mario Gonzalez
                                 Address Redacted
14965208
            Mario Kio
                           Address Redacted
14965209
            Mario Luna
                             Address Redacted
14965210
            Mario Mentado
                                Address Redacted
14965211
            Mario Murillo
                               Address Redacted
14965212
            Mario Yarza
                              Address Redacted
14965213
                                 Address Redacted
            Marisela Escoto
14965214
            Marisela Grijalva
                                  Address Redacted
           Marisela Hanel
14965215
                                Address Redacted
14965216
            Marisela Meraz
                                Address Redacted
14965217
            Marisela S Marquez
                                    Address Redacted
14965218
            Marisol Castro
                               Address Redacted
14965219
            Marisol Hernandez
                                   Address Redacted
14965220
            Marisol Ruiz
                              Address Redacted
14965221
            Marisol Solares
                                Address Redacted
14965222
            Marisol Solis
                              Address Redacted
14965223
            Marissa Rodriguez
                                   Address Redacted
14965224
            Marissa Soto
                              Address Redacted
14965225
            Maritza Bojorquez
                                   Address Redacted
14965226
            Maritza Mendoza
                                  Address Redacted
14965227
            Maritza Zaragoza
                                  Address Redacted
           Marivel Valdivia
14965228
                                 Address Redacted
14965229
            Marivette Juarez
                                 Address Redacted
14965230
                            Address Redacted
           Mark Brom
14965231
            Mark Connelly
                                Address Redacted
14965234
           Mark E Caleb
                               Address Redacted
14965235
            Mark Naah
                            Address Redacted
14965236
            Marla Marie Giglio
                                    Address Redacted
14965237
            Marleen Al Sapty
                                  Address Redacted
           Marlen Otero
14965238
                              Address Redacted
14965239
            Marlene M. Thompson, D.D.S., Inc.
                                                   Address Redacted
14965240
            Marlene M. Thompson, D.D.S., Inc.
                                                   Address Redacted
                                                401 West A Street, Suite 1900
                                                                                 Wilson Elser-Robert W.
14965241
            Marlene Thompson, D.D.S., Inc.
            Harrison
                          San Diego, CA 92101
14965242
            Marleni Baranda
                                 Address Redacted
14965243
            Marly Gonzalez Pagan
                                      Address Redacted
14965244
            Marlyne Abon
                               Address Redacted
14965245
            Marnie Hesson
                                Address Redacted
14965246
            Marquelle Bradford
                                    Address Redacted
14965247
            Marquis Smith
                               Address Redacted
14965250
            Marsh & McLennan Insurance Agency
                                                     Towne Centre Drive, Suite 100
                                                                                        San Diego, CA 92122
14965251
            Marsh McLennan Agency (MMA)
                                                  62886 Collection Center Drive
                                                                                    Chicago, IL 60693-0628
14965252
            Marsh McLennan Agency (MMA)
                                                 9171 Towne Centre Drive, Suite 100
                                                                                         Jennifer Weld
                                                                                                            San Diego,
            CA 92122
14965253
           Marta Garcia
                              Address Redacted
```

```
14965255
            Martha Centeno
                                Address Redacted
            Martha Contreras-Orozco
14965256
                                         Address Redacted
14965257
            Martha Coreas
                               Address Redacted
                                 Address Redacted
14965258
            Martha Deichler
14965259
            Martha Guiterrez
                                 Address Redacted
14965260
           Martha Hernandez
                                   Address Redacted
14965261
            Martha Lopez
                               Address Redacted
            Martha M Guzman
14965262
                                   Address Redacted
14965263
            Martha Morales
                                Address Redacted
14965265
            Martha Ortiz
                              Address Redacted
14965266
            Martha Perez
                              Address Redacted
14965267
            Martha Putman
                                Address Redacted
14965268
            Martha Reymundo de Orellana
                                              Address Redacted
            Martha Romanillo
14965269
                                  Address Redacted
14965270
            Martha Romero
                                Address Redacted
14965271
            Martha Trejo
                              Address Redacted
14965272
            Martha Vargas
                               Address Redacted
14965273
            Martha Villalobos
                                  Address Redacted
            Marthas Village and Kitchen
14965275
                                            83971 Date Avenue
                                                                    Indio, CA 92201
14965276
            Martin Borbonio
                                 Address Redacted
14965277
            Martin Felix
                             Address Redacted
                                 Address Redacted
14965278
            Martin J Stavich
14965279
            Martin Juarez
                              Address Redacted
14965280
            Martin Lonnekah
                                  Address Redacted
14965281
            Martin Lopez
                              Address Redacted
14965283
            Martin Nunez
                              Address Redacted
14965284
            Martin Soto Ixta
                                 Address Redacted
            Martina Zacarias Zamora
14965285
                                         Address Redacted
14965286
            Marwah Aldulaimi
                                   Address Redacted
14965287
            Marwah Aldulaimi
                                   Address Redacted
14965288
            Mary Beyers
                              Address Redacted
14965289
            Mary Chico
                             Address Redacted
14965291
            Mary Grammont
                                 Address Redacted
14965292
                              Address Redacted
            Mary Hetsko
           Mary Ingram
Mary Litch
14965293
                              Address Redacted
14965294
                            Address Redacted
14965295
            Mary Metzler
                              Address Redacted
14965296
            Mary Mitten
                             Address Redacted
14965298
            Mary P Silsbe
                               Address Redacted
            Mary Romero
14965299
                               Address Redacted
14965300
            Mary Sullivan
                               Address Redacted
14965301
            Maryam Al-Shaikhli
                                     Address Redacted
14965302
            Maryam Iskander
                                  Address Redacted
14965303
            Maryam Poursartip
                                   Address Redacted
14965304
            Maryam Yousif
                                Address Redacted
                                   Address Redacted
14965305
            Maryan Hernandez
14965306
            Maryrose Channell
                                   Address Redacted
            Maryville University
14965307
                                     650 Maryville University Drive
                                                                        St. Louis, MO 63141
14965308
            Masroor Samandari
                                    Address Redacted
            Master Shield
14965310
                              Address Redacted
            Mateo Williamson
14965312
                                   Address Redacted
14965314
            Matilda Fernandez
                                   Address Redacted
14965315
            Matt Phillips
                              Address Redacted
14965316
            Matthew Fagan
                                Address Redacted
14965317
            Matthew Hamana
                                  Address Redacted
14965318
            Matthew J Juric
                                Address Redacted
            Matthew Keane
14965319
                                Address Redacted
14965320
            Matthew Macias
                                 Address Redacted
14965321
            Maura Tuso, D.D.S
                                    Address Redacted
14965322
            Maureen Fleming-Mullins
                                          Address Redacted
14965324
            Mauricio Polanco
                                  Address Redacted
14965325
            Maurilia Landeros
                                  Address Redacted
14965326
            May Basheer
                              Address Redacted
14965327
            May Hasan
                            Address Redacted
14965328
            May Hlaihel
                             Address Redacted
14965329
            Maya Salmeron
                                Address Redacted
14965330
            Mayara Dantas
                                Address Redacted
                                  35-325 Date Palm Drive, Suite 101
                                                                        Cathedral City, CA 92234
14965331
            Mayfield College
14965332
            Maykoll A Perdomo
                                    Address Redacted
14965333
            Mayra Barraza
                               Address Redacted
            Mayra Fernandez Avon
14965334
                                       Address Redacted
14965335
            Mayra Figueroa Vazquez
                                         Address Redacted
14965336
            Mayra Garcia
                              Address Redacted
14965337
            Mayra Garza
                              Address Redacted
14965338
            Mayra Hernandez
                                  Address Redacted
14965339
            Mayra Mendivil
                                 Address Redacted
14965340
            Mayra Navarrette
                                  Address Redacted
14965341
            Mayra Nogales
                                Address Redacted
```

```
14965342
           Mayra Peterson
                              Address Redacted
14965343
           Mayra Salce
                           Address Redacted
14965344
           Mayra Santana
                             Address Redacted
14965345
           Mayra Terry
                           Address Redacted
14965346
           Mayra Valdivia
                              Address Redacted
14965347
           Mayra Zamora Cruz
                                 Address Redacted
14965348
           Mays Alfalahi
                             Address Redacted
14965350
           McAlister Institute
                                1400 N Johnson Avenue, Suite 101
                                                                   El Cajon, CA 92020
14965351
           McKesson
                         6535 N. State Highway 161
                                                      Irving, TX 75039
                                                Irving, TX 75039
14965352
           McKesson
                         6555 State Hwv 161
14965353
           McKesson
                         9501 Norwalk Blvd
                                                Santa Fe Springs, CA 90670
           McKesson - 247575 - Retail Account
                                                 590 PALM CANYON DR STE 2
14965354
                                                                                   BORREGO SPRINGS, CA
           92004
14965355
           McKesson - 249668 -OTC
                                        590 PALM CANYON DR. STE 2
                                                                          Borrego Springs, CA 92004
14965357
           McKesson-444106-340B
                                         590 PALM CANYON DR. STE20
                                                                            Borrego Springs, CA 92004
14965377
           McKesson - 598891 - Rite Aid 05670
                                                 82451-C HIGHWAY 111
                                                                             INDĬO, ĈA 92201
           McKesson - 712892 - Cat City
14965393
                                           68555 RAMON RD BLDG D105
                                                                             CATHEDRAL CITY, CA 92234
14965399
           McKesson - 885450 - WALMART 1555
                                                    2150 N. WATERMAN AVE.
                                                                                  EL CENTRO, CA 92243
           McKesson - 885482 - Walmart - 3464
                                                  3943 Grand Avenue
14965410
                                                                         Chino, CA 91710
                                                                          Redlands, CA 92373
14965411
           McKesson - 902248 - SAV ON 1319
                                                 450 E CYPRESS AVE
           McKesson - 902249 - SAV ON 1569
                                                 1751 N SUNRISE WAY #1
                                                                              PALM SPRINGS, CA 92262
14965412
14965413
                                                  40101 MONTEREY AVE
           McKesson - 902250 - SAV ON - 2586
                                                                              RANCHO MIRAGE, CA
           92270
                                                                        Spring Valley, CA 91977
SPRING VALLEY, CA 91977
14965414
           McKesson - 902251 - SAV ON 0758
                                                 543 SWEETWATER RD
14965415
           McKesson - 902252 - SAV ON - 0723
                                                  9831 CAMPO RD
           McKesson - 902253 - SAV ON - 2783
14965416
                                                  2899 JAMACHA RD
                                                                          EL CAJON, CA 92019
14965417
           McKesson - 902255 - SAV ON 0704
                                                 2955 ALPINE BLVD
                                                                         ALPINE, CA 91901
           McKesson – 902256 – SAV ON 0725
14965418
                                                 1459 MAIN ST
                                                                    RAMONA, CA 92065
           McKesson - 902258 - SAV ON 4713
14965419
                                                 1509 E VALLEY PKWY
                                                                            ESCONDIDO, CA 92027
           McKesson Corporation
14965424
                                                                    Lori Schechter
                                    6555 North State Highway 161
                                                                                      Irving, TX 75039
14965425
           McKesson Pharmaceutical - 345433
                                              455 NORTH MAGNOLIA AVE
                                                                               EL CAJON, CA 92020
14965426
                                                                        Atlanta, GA 30384-0884
           McKesson Pharmacy Systems-00821366
                                                   P.O. Box 100884
           Mckesson – 270545 – La Mesa
                                                                      LA MESA, CA 91942
14965356
                                           8851 CENTER DR. #210
14965358
           Mckesson - 536421 - Walmart - 4169
                                                  300 N 2ND ST EL C
                                                                         EL CAJON, CA 92021
           Mckesson - 536422 - Walmart 3084
                                                1266 E VALLEY PARKWAY
14965359
                                                                               ESCONDIDO, CA 92027
           Mckesson – 536473 – Walmart 5996
14965360
                                                1330 E GRAND AVE
                                                                        Escondido, CA 92027
           Mckesson - 536475 - Walmart 3494
14965361
                                                732 CENTER DRIVE
                                                                        SAN MARCOS, CA 92069
           Mckesson - 536528 Walmart - 1756
14965362
                                                17251 Foothill Blvd
                                                                       Fontana, CA 92335
14965363
           Mckesson - 536580 - Walmart 1805
                                                79295 HWY 111
                                                                    La Quinta, CA 92253
           Mckesson - 536583 - Walmart 1832
                                                WALMART 1832/BORREGO PHS
14965364
                                                                                    PALM SPRINGS, CA
           92262
          Mckesson – 536634 – Walmart 1899
Mckesson – 536635 – Walmart 2028
14965365
                                                6250 VALLEY SPRINGS PKWY
                                                                                  Riverside, CA 92507
14965366
                                                5200 VAN BUREN BLVD
                                                                             RIVERSIDE, CA 92503
           Mckesson - 536636 - Walmart 3524
14965367
                                                605 FLETCHER PARKWAY
                                                                               EL CAJON, CA 92020
14965368
           Mckesson - 536688 - Walmart 2253
                                                13487 CAMINO CANADA
                                                                             El Cajon, CA 92021
           Mckesson - 536689 - Walmart 5096
14965369
                                                34500 MONTEREY AVE
                                                                            Palm Desert, CA 92260
14965370
           Mckesson - 597979 Rite Aid 05651
                                               6405 EL CAJON BLVD
                                                                         SAN DIEGO, CA 92115
           Mckesson - 598032 Rite Aid 05716
                                               3849 CHICAGO AVENUE
                                                                            RIVERSIDE, CA 92507
14965371
                                                                      LEMON GROVE, CA 91945
           Mckesson - 598140 Rite Aid 05618
14965372
                                               7224 BROADWAY
           Mckesson - 598301 Rite Aid 05679
14965373
                                               12900 PALM DR
                                                                    DESERT HOT SPG, CA 92240
           Mckesson - 598355 Rite Aid 05632
14965374
                                               9532 WINTER GARDENS BLVD
                                                                                  Lakeside, CA 92040
14965375
           Mckesson - 598569 Rite Aid 06486
                                               1180 N STATE STREET
                                                                          BORREG PHS
                                                                                             SAN JACINTO, CA
           92583
14965376
           Mckesson - 598570 Rite Aid 05712
                                               6150 VAN BUREN BLVD
                                                                           RIVERSIDE, CA 92503
           Mckesson – 598892 Rite Aid 05683
14965378
                                               366 SOUTH PALM CANYON DR
                                                                                  PALM SPRINGS, CA
           92262
           Mckesson - 598945 Rite Aid 05656
14965379
                                               8694 LAKE MURRAY BLVD
                                                                               San Diego, CA 92119
14965380
           Mckesson - 598998 Rite Aid- 05718
                                                8015 LIMONITE AVENUE
                                                                              RIVERSIDE, CA 92509
           Mckesson – 599052 Rite Aid 05676
14965381
                                                                         CATHEDRAL CITY, CA 92234
                                               69155 RAMON ROAD
14965382
           Mckesson - 599106 Rite Aid 05694
                                               1133 PHS 1133 N Mount Vernon Ave
                                                                                     Colton, CA 92324
14965383
           Mckesson - 599160 Rite Aid 06246
                                               41800 WASHINGTON ST 113B
                                                                                 Bermuda Dunes, CA 92203
14965384
           Mckesson - 599267 Rite Aid 05681
                                               72-875 HWY 111
                                                                    PALM DESERT, CA 92260
           Mckesson - 599321 Rite Aid 05620
                                               661 SWEETWATER ROAD
14965385
                                                                             SPRING VALLEY, CA 91977
           Mckesson - 599374 Rite Aid 05623
14965386
                                               2516 JAMACHA ROAD
                                                                          El Cajon, CA 92019
           Mckesson - 599428 Rite Aid 05678
14965387
                                               51101 HARRISON STREET
                                                                             COACHELLA, CA 92236
           Mckesson - 599481 Rite Aid 05625
14965388
                                               1135 AVOCADO AVENUE
                                                                             EL CAJON, CA 92020
           Mckesson - 599482 Rite Aid 05684
                                                                              PALM SPRINGS, CA 92262
14965389
                                               111 SOUTH SUNRISE WAY
           Mckesson - 599535 Rite Aid 06532
14965390
                                               47985 MONROE ST BLDG A
                                                                               Indio, CA 92201
           Mckesson - 599857 Rite Aid 05628
14965391
                                               315 W Washington Avenue
                                                                            Escondido, CA 92025
14965392
           Mckesson - 632750 Barstow Pharmacy
                                                  750 E MAIN ST
                                                                      Barstow, CA 92311
14965394
           Mckesson - 885443 - Walmart 2708
                                                32225 Highway 79 South
                                                                           Temecula, CA 92592
           Mckesson - 885445 - Walmart - 5663
14965395
                                                  8844 LĬMOŇITE AVE
                                                                           JURUPA VALLEY, CA 92509
           Mckesson – 885446 – Walmart – 2826
Mckesson – 885447 – Walmart – 3026
14965396
                                                  83053 AVENUE 48
                                                                        COACHELLA, CA 92236
14965397
                                                  72314 HIGHWAY 111
                                                                           PALM DESERT, CA 92260
           Mckesson - 885449 - Walmart 2181
14965398
                                                82-491 AVENUE 42
                                                                        INDIO, CA 92203
           Mckesson – 885451 – Walmart – 1600
Mckesson – 885453 – Walmart 1692
14965400
                                                                                 RIDGECREST, CA 93555
                                                 201 EAST BOWMAN ROAD
14965401
                                                1120 SOUTH MT VERNON AVE
                                                                                   COLTON, CA 92324
          Mckesson - 885454 - Walmart - 1879
14965402
                                                 301 MONTARA ROAD
                                                                            BARSTOW, CA 92311
```

	0170
14965403	Mckesson – 885455 Walmart – 1859 2540 Rockwood Ave Calexico, CA 92231
14965404	Mckesson – 885456 – Walmart – 1862 1366 SOUTH RIVERSIDE AVE RIALTO, CA 92376
14965405	Mckesson – 885462 – Walmart – 1917 170 TOWN CENTER PKWY SANTEE, CA 92071
14965406	Mckesson – 885466 – Walmart 5193 12721 MORENO BEACH DR MORENO VALLEY, CA
	92555
14965407	Mckesson – 885468 – Walmart – 5425 1861 S SAN JACINTO AVE SAN JACINTO, CA 92583
14965408	Mckesson – 885470 – Walmart 2479 2412 COLLEGE AVE SAN DIEGO, CA 92115
14965409	Mckesson – 885473 – Walmart 2150 710 DENNERY RD SAN DIEGO, CA 92154
14965420	Mckesson – Anza Valley Pharmacy 536741 56555 CA–371 Anza, CA 92539
14965421	Mckesson – Desert Hospital –536743 1180 N INDIAN CANYON E140 PALM SPRINGS, CA
	92262
14965422	Mckesson – Hillcrest 885479 120 UNIVERSITY AVE SAN DIEGO, CA 92103
14965423	Mckesson – Navarro 10727/LTSALUD PHS 7205 CORP CTR DR STE 104 MIAMI, FL 33126
14965428	Mckesson-028842-BETTER LIFE PCY 528 E MAIN ST Borrego PHS EL CAJON, CA
	92020
14965429	Mckesson-029347-CVS PHCY 16094 3333 ARLINGTON AVE Borrego PHS RIVERSIDE, CA
1.4065.420	92506
14965430	Mckesson-029362-CAREMARK NJ SP 180 PASSAIC AVE Borrego PHS FAIRFIELD, NJ
14065421	07004 Mckesson-029375-CAREMARK TN SP 8370 WOLF LAKE DR STE 107 Borrego
14965431	
14965432	PHS BARLETT, TN 38133 Mckesson-029376-CAREMARK TX SP 1001 SPINKS ROAD STE 280 Borrego PHS FLOWER
14703432	MOUND, TX 75028
14965433	Mckesson-029382-CAREMARK NV SP 7251 SOUTH EASTERN AVE Borrego PHS LAS
14703433	VEGAS, NV 89119
14965434	Mckesson-029385-CVS PHCY 9746/BorregoPHS 6840 LINCOLN AVENUE RIVERSIDE, CA
14705454	92506
14965435	Mckesson-029387-CVS PHCY 9799/BorregoPHS 6215 RIVERSIDE AVE RIVERSIDE, CA
	92506
14965436	Mckesson-029388 CVS PHCY 9849/BorregoPHS 8280 MAGNOLIA AVE RIVERSIDE, CA
	92504
14965437	Mckesson-802906-CVS PHCY 5497/BorregoPHS 84010 Avenue 50 Coachella, CA 92236
14965438	Mckesson-802907-CVS PHCY 9114/BorregoPHS 1299 Broadway El Canjon, CA 92021
14965439	Mckesson-802909-CVS PHCY 9616/BorregoPHS 31575 Date Palm Dr Cathedral City, CA 92234
14965440	Mckesson–811689–Premier PS CA/BorregoPHS 410 Cloverleaf Dr Baldwin Park, CA 91706
14965441	Mckesson-837123-CVS PHCY 3055/BorregoPHS 38012 47TH STREET E PALMDALE, CA
1.40.65.440	93552
14965442	Mckesson-837124-CVS PHCY 16937 2295 N IMPERIAL AVENUE Borrego PHS EL CENTRO,
14965443	CA 92243 Mckesson–837127–CVS PHCY 10931 1745 S. Imperial Ave Borrego PHS El Centro, CA
14903443	Mckesson–837127–CVS PHCY 10931 1745 S. Imperial Ave Borrego PHS El Centro, CA 92243
14965444	Mckesson-837132-CVS PHCY 7127/BorregoPHS 1302 W. Mission Rd. San Marcos, CA 92069
14965445	Mckesson-837135-CVS PHCY 4768/BorregoPHS 510 C Street San Diego, CA 92101
14965446	Mckesson-837141-CVS PHCY 8842/BorregoPHS 426 E. Chase Avenue El Cajon, CA 92020
14965447	Mckesson-837144-CVS PHCY 8857/BorregoPHS 1441 N Santa Fe Vista, CA 92084
14965448	Mckesson-837152-CVS PHCY 9103/BorregoPHS 318 W. El Norte Parkway Escondido, CA
	92026
14965449	Mckesson-837154-CVS PHCY 9104/BorregoPHS 2427 E. Valley Pkwy Escondido, CA 92027
14965450	Mckesson-837156-CVS PHCY 9108/BorregoPHS 1655 S. Center City Escondido, CA 92025
14965451	Mckesson-837157-CVS PHCY 9109/BorregoPHS 6265 El Cajon Blvd San Diego, CA 92115
14965452	Mckesson-837159-CVS PHCY 9115/BorregoPHS 645 Saturn Blvd San Diego, CA 92154
14965453	Mckesson-837160-CVS PHCY 9120/BorregoPHS 800 Palm Ave Imperial Beach, CA 91932
14965454	Mckesson—837162—CVS PHCY 9137/BorregoPHS 7393 Jackson San Diego, CA 92119
14965455	Mckesson-837164-CVS PHCY 9140/BorregoPHS 555 Broadway Ste 1054 Chula Vista, CA
14965456	91910 Mckesson–837173–CVS PHCY 9145 9730 Mission Gorge Borrego PHS Santee, CA 92071
14965457	Mckesson-837174-CVS PHCY 9166/BorregoPHS 734 S. RANCHO SANTA FE SAN MARCOS, CA
14705457	92078
14965458	Mckesson-837175-CVS PHCY 9173/BorregoPHS 2760 Fletcher Pky El Cajon, CA 92020
14965459	Mckesson-837178-CVS PHCY 9175/BorregoPHS 4829 CLAIREMONT DR SAN DIEGO, CA
- 1, 00 10,	92117
14965460	Mckesson-837182-CVS PHCY 9193/BorregoPHS 1380 43RD ST SAN DIEGO, CA 92113
14965461	Mckesson-837184-CVS PHCY 9194/BorregoPHS 1810 Main Street Ramona, CA 92065
14965462	Mckesson-837193-CVS PHCY 9524/BorregoPHS 3350 Palm Ave San Diego, CA 92154
14965463	Mckesson-837194-CVS PHCY 9549/BorregoPHS 1980 College Blvd Oceanside, CA 92056
14965464	Mckesson-837197-CVS PHCY 9574/BorregoPHS 883 S. Santa Fe Avenue Vista, CA 92083
14965465	Mckesson-837201-CVS PHCY 9634/BorregoPHS 2521 East Avenue S Palmdale, CA 93550
14965466	Mckesson-837210-CVS PHCY 9657/BorregoPHS 44430 Challenger Way Lancaster, CA 93535
14965467	Mckesson-837218-CVS PHCY 9740/BorregoPHS 15210 EAST ROSECRANS AVEN LA MIRADA,
14065460	CA 90638 Melegger 995460, WALMART 1022/DORDECO DUS 12540 EQOTUBLE DEVID DANCHO
14965468	Mckesson–885460–WALMART 1922/BORREGO PHS 12549 FOOTHILL BLVD RANCHO
1/10/5/160	CUCAMONGA, CA 91739 Mckesson-885474-CVS PHCY 9112/BorregoPHS 7100 BROADWAY LEMON GROVE, CA
14965469	Mckesson-885474-CVS PHCY 9112/BorregoPHS 7100 BROADWAY LEMON GROVE, CA 91945
14965470	Mckesson-885478-CVS PHCY 9721/BorregoPHS 1401 S Baldwin Arcaida, CA 91007
14965471	Mckesson–925427–CVS PHCY 1520/BorregoPHS 68010 VISTA CHINO CATHEDRAL CITY, CA
	92234

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 47 of 76

```
14965472
           Mckesson-925429-CVS PHCY 16083
                                                     1280 AUTO PARK WAY SOUTH
                                                                                          Borrego
                     ESCONDIDO, CA 92029
           PHS
14965473
           Mckesson-925431-CAREMARK KS SP
                                                       11162 RENNER BLVD
                                                                                   Borrego PHS
                                                                                                     LENEXA, KS
           66219
14965474
           Mckesson-925435-CVS Caremark 2921
                                                       105 MALL BOULEVARD
                                                                                     Borrego
                      MONROEVILLE, PA 15146
           PHS
14965475
           Mckesson-925436-CVS PHCY 9648/BorregoPHS
                                                                82270 HIGHWAY 111
                                                                                           INDIO, CA 92201
           Mckesson-925437-CVS PHCY 9728/BorregoPHS
                                                                                     CORONA, CA 92882
14965476
                                                                1322 W. SIXTH
14965477
           Mckesson-925439-CVS PHCY 9804/BorregoPHS
                                                                425 SUNRISE WAY SOUTH
                                                                                                 PALM SPRINGS, CA
           92262
14965479
           Meagan Thomas
                                Address Redacted
           Mecca Community Pharmacy
                                            91275 Ave 66 Ste 200
14965481
                                                                      Mecca, CA 92254
                                    90480 66th Ave
14965482
           Mecca Travel Center
                                                         Mecca, CA 92254
14965488
           Medi-cal
                          P.O. Box 997413, MS 0000
                                                         CA Department of Health Care Services
                                                                                                   Sacramento, CA
           95899-7413
14965486
           Medica
                        P.O. Box 30990
                                             Salt Lake City, UT 84130
14965491
                                   9323 Chesapeake Drive, Suite #C1
           Medpilot Pharmacy
                                                                        San Diego, CA 92123
14965493
           Megan Brinkman
                                 Address Redacted
14965494
           Megan Maupin
                               Address Redacted
14965495
           Megan Maupin
                               Address Redacted
           Megan Nielson
14965496
                               Address Redacted
                             Address Redacted
14965497
            Megan Ward
14965498
           Mehrnaz Irani
                              950 Civic Center Drive Suite #B
                                                                  Vista Village Family Dentistry–M Irani
                                                                                                           Vista, CA
            92083
14965499
           Mehyo Tabikh
                               Address Redacted
14965500
           Mel Galvez
                            Address Redacted
14965501
           Melanie Genovese
                                  Address Redacted
14965502
           Melanie Painter
                                Address Redacted
14965503
           Melany Gonzalez
                                 Address Redacted
14965504
           Melchor N Magdaleno
                                      Address Redacted
14965505
                              Address Redacted
           Melisa Rivera
           Melissa Ayala
14965506
                              Address Redacted
14965508
           Melissa Cruz
                             Address Redacted
14965509
           Melissa Daniels
                                Address Redacted
14965510
                                 Address Redacted
           Melissa Esquivel
14965511
           Melissa Fonder
                               Address Redacted
14965512
           Melissa Garcia
                               Address Redacted
14965513
           Melissa Global Intelligence
                                          22382 Avenida Empresa
                                                                      Rancho Santa Margarita, CA 92688
14965514
           Melissa Guerrero
                                 Address Redacted
14965515
           Melissa Jimenez
                                Address Redacted
14965516
           Melissa Macias
                               Address Redacted
                                Address Redacted
14965517
           Melissa Morales
14965518
           Melissa Murguia
                                 Address Redacted
14965519
           Melissa Perez Rios
                                   Address Redacted
14965520
           Melissa Seinturier
                                  Address Redacted
14965521
           Melissa Viniegra
                                 Address Redacted
14965522
           Melody Angeles-Riparip
                                        Address Redacted
14965523
           Melody Armstead
                                  Address Redacted
14965524
           Melody Montano
                                 Address Redacted
14965527
           Meltwater Services
                                   13470 Washington Blvd
                                                               Floor 3
                                                                           Marina Del Rey, CA 90292
14965528
           Mental Health Network
                                                             San Rafael, CA 94901
                                       2370 Kerner Blvd
14965530
           Merari Recio
                             Address Redacted
14965531
           Mercay Romero
                                Address Redacted
14965532
           Mercedes Bolanos
                                  Address Redacted
14965534
           Mercer Insurance Company
                                          1166 Avenue of the Americas
                                                                           New York, NY 10036
           Mercer Insurance Company
                                          P.O. Box # 73909
14965535
                                                                118 2nd Avenue SE
                                                                                        Cedar Rapids, IA 52407
                                              Holland, MI 49422-2571
14965538
           Mercy Air
                           P.O. Box 2571
                                      4500 E. Cotton Center Blvd.
14965539
           Mercy Care Pharmacy
                                                                     Mercy Care /Mercy Care Advantage
                                                                                                            Phoenix,
           AZ 85040
14965540
           Mercy Physicians Medical Group
                                               3900 05th Ave Ste 370
                                                                          San Diego, CA 92103
14965541
           Mereyda Rodriguez
                                   Address Redacted
14965542
           Meriem Bojorquez-Valenzuela
                                              Address Redacted
14965543
           Meritain
                         4500 E Cotton Center Blvd
                                                        Phoenix, AZ 85040
14965545
           Meritain Health
                                1405 Xenium Lane North Suite 140
                                                                      Minneapolis, MN 55441
14965547
           Merna Nissan
                              Address Redacted
14965549
                                                   Kansas City, MO 64116
           MetLife
                         4150 N. Mulberry Dr.
                               P.O. Box 783895
                                                     Philadelphia, PA 19178-3895
14965550
           MetLife - LTD
14965551
           MetLife - Patient Refunds
                                         P.O. Box 981282
                                                               El Paso, TX 79998
14965552
           MetLife - SAP
                               P.O. Box 804466
                                                     Kansas City, MO 64180-4466
           Metropolitan Family Med Clinic-Fontana
                                                       742 W. Highland Ave
742 W. Highland Ave
14965554
                                                                                 San Bernardino, CA 92405
14965555
           Metropolitan Family Med Clinic-Foothills
                                                                                  San Bernardino, CA 92405
14965556
           Metropolitan Family Medical Clinic
                                                  742 W. Highland Ave
                                                                            San Bernardino, CA 92405
           Metropolitan Family Medical Clinic
Metropolitan Family Medical Clinic – RIV
                                                  742 W. Highland Ave
742 W. Highland Ave
14965557
                                                                            San Bernardino, CA 92405
14965558
                                                                                  San Bernardino, CA 92405
                                                       742 W. Highland Ave
14965559
            Metropolitan Family Medical Clinic - SB
                                                                                 San Bernardino, CA 92405
           Michael Briones
                                Address Redacted
14965564
14965565
           Michael Caruso
                                Address Redacted
           Michael Ciolino
14965566
                                Address Redacted
```

```
14965567
            Michael Dominguez
                                    Address Redacted
           Michael Hetsko
14965568
                                Address Redacted
14965569
            Michael Hickok
                                Address Redacted
            Michael Hoang, D.M.D
                                       Address Redacted
14965570
14965571
            Michael Hoang, D.M.D, Inc.
                                            Address Redacted
           Michael Hoang, DDS
                                     401 West A Street, Suite 1900
                                                                       Wilson Elser-Robert W. Harrison
14965572
                                                                                                            San Diego,
            CA 92101
           Michael I. Gottfried, Esq.
                                         Roye Zur, Esq.
                                                             Elkins Kalt Weintraub Reuben Gartside LL
                                                                                                           10345 W.
14969738
            Olympic Blvd.
                               Los Angeles, CA 90064
           Michael I. Gottfried, Esq.
                                         Roye Zur, Esq.
14969764
                                                             Elkins Kalt Weintraub Reuben Gartside LL
                                                                                                           10345 W.
            Olympic Blvd.
                               Los Angeles, CA 90064
            Michael Kaser
14965573
                               Address Redacted
            Michael McLean
14965574
                                 Address Redacted
14965575
            Michael Phelps
                                Address Redacted
14965576
            Michael R. Solomon M.D.
                                          Address Redacted
14965577
            Michael Raum
                               Address Redacted
            Michael Rodriguez
14965579
                                   Address Redacted
14965580
            Michael Standberg
                                   Address Redacted
14965581
            Michael Woodward
                                    Address Redacted
14965582
            Michele Archambeault
                                       Address Redacted
14965583
            Michelea Kidwell
                                  Address Redacted
14965584
            Michelle Aman
                                Address Redacted
14965585
           Michelle Barrios
                                 Address Redacted
14965586
            Michelle Berard
                                 Address Redacted
14965587
            Michelle Blankenship
                                      Address Redacted
14965588
            Michelle Cregg
                                Address Redacted
14965589
            Michelle De Hoyos
                                    Address Redacted
14965590
            Michelle Harman
                                  Address Redacted
                                 Address Redacted
14965591
           Michelle Harper
14965592
                                Address Redacted
            Michelle Hayes
14965593
            Michelle Herrera
                                 Address Redacted
14965594
                                  Address Redacted
            Michelle Johnson
14965595
           Michelle Kimmel
                                  Address Redacted
14965596
            Michelle LaCoff
                                 Address Redacted
14965597
                                Address Redacted
            Michelle Lewis
14965598
            Michelle Perchez
                                  Address Redacted
14965599
            Michelle Ramos
                                 Address Redacted
                                 Address Redacted
14965600
           Michelle Rogers
14965601
            Michelle Sarabia
                                 Address Redacted
14965604
           Microsoft NDA
                                One Microsoft Way
                                                        Redmond, WA 98052
14965605
            Midland National
                                  2575 E Camelback Rd, Suite 700
                                                                      Phoenix, AZ 85016
                                Address Redacted
14965606
            Miguel Alvarez
14965607
            Miguel Baay
                              Address Redacted
           Miguel Corona
                                Address Redacted
14965608
14965609
            Miguel De Leon
                                 Address Redacted
            Miguel Gutierrez Cedillo
14965610
                                        Address Redacted
14965611
            Miguel Herrera
                                Address Redacted
            Miguel Ibarra
                              Address Redacted
14965612
            Miguel Jimenez
14965613
                                Address Redacted
           Miguel Q Torres
Miguel Ramos
                                 Address Redacted
14965614
14965615
                               Address Redacted
14965616
            Miguel Torres
                               Address Redacted
            Miguel Valdez
14965617
                               Address Redacted
14965618
            Miguel Vargas
                               Address Redacted
14965619
            Mike Butrus
                             Address Redacted
14965620
            Mike C Littrell
                                Address Redacted
14965621
            Mike Hickok
                              450 B Street, Suite 1410
                                                          Kyle Harris LLP-Laura Kimberly Gantney
                                                                                                        San Diego, CA
            92101
14965622
           Mike Issac
                            Address Redacted
14965623
            Mikia Wallis
                              Address Redacted
14965624
            Mikia Wallis
                              Address Redacted
14965625
            Mildred Vazquez Gandara
                                          Address Redacted
14965627
           Mimi Segrist
                              Address Redacted
14965628
            Minerva Lara Don Diego
                                         Address Redacted
                                Address Redacted
14965634
            Mirella Morgan
            Mirella Padilla
14965635
                               Address Redacted
            Mireya Lopez-Razo
14965636
                                     Address Redacted
            Mireya Luevano
                                 Address Redacted
14965637
14965638
            Miriam A Garcia
                                 Address Redacted
14965639
            Miriam Cruz
                              Address Redacted
            Miriam Mendez
14965640
                                Address Redacted
14965641
            Miriam Ramirez Sanchez
                                         Address Redacted
14965642
            Miriam Valadez
                                 Address Redacted
            Mirna A Gonzalez
                                   Address Redacted
14965644
14965645
            Mirna Gonzalez
                                Address Redacted
14965646
            Mirta Calderon
                                Address Redacted
14965647
            Mission Springs Water District
                                              66575 Second St.
                                                                    Desert Hot Springs, CA 92240-3711
14965648
           Mita Beach
                            Address Redacted
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 49 of 76

```
14965653
           Modi & Shah
                             28039 Scott Rd STE I
                                                      DDS
                                                                 Murrieta, CA 92563
14965654
           Modupe Adeniranye
                                   Address Redacted
14965655
           Mohamed Dowaidari
                                    1635 N Mountain Avenue
                                                                 Mohamed Dowaidari
                                                                                         Upland, CA 91785
                                            740 Oak Avenue Parkway, Suite 120
14965658
           Mohammed AlTekreeti, D.D.S
                                                                                   Adams & Corzine-Michael Karl
           Pazdernik
                          Folsom, CA 95630
           Mohammed AlTekreeti, D.D.S
14965659
                                            Address Redacted
14965657
           Mohammed Almawla
                                    Address Redacted
14965660
           Mohanad Abdulridha
                                    Address Redacted
14965661
           Moises Aviles
                              Address Redacted
           Moises Diaz
14965662
                            Address Redacted
14965664
           Molina Health Care of California
                                              200 Oceangate, Suite 100
                                                                           Long Beach, CA 90802
                                                              Long Beach, CA 90802
14965666
           Molina Health Inc.
                                  200 Oceangate, Suite 100
14965668
           Molina Healthcare of California
                                             200 Oceangate, 6th floor
                                                                         Long Beach, CA 90802
14965671
                               Address Redacted
           Monica Alcaraz
14965672
           Monica Diaz
                             Address Redacted
14965673
           Monica Gonzalez
                                 Address Redacted
14965674
           Monica Lora Martinez
                                     Address Redacted
14965675
           Monica M Osuna
                                 Address Redacted
14965676
           Monica Modi
                             Address Redacted
14965677
           Monica Morlan
                               Address Redacted
14965678
           Monica Olguin
                               Address Redacted
14965679
           Monica Patton
                              Address Redacted
14965680
           Monica Ramirez
                                Address Redacted
14965681
           Monica Sanchez
                                Address Redacted
           Monica Torres-Silva
14965682
                                    Address Redacted
14965683
           Monica Urias
                             Address Redacted
14965684
           Monika Soto
                             Address Redacted
14965685
           Monique Carrier
                                Address Redacted
14965686
           Monique Jimenez
                                 Address Redacted
           Monique Perry
14965687
                               Address Redacted
           Mono Degrace Address Redacted
Moreno Valley Unified School District
                              Address Redacted
14965688
14965691
                                                    25634 Alessandro Blvd.
                                                                               Moreno Valley, CA 92553
14965692
           Morgan Hodge
                               Address Redacted
14965693
           Mouayed Marzook
                                  Address Redacted
14965696
           Mpower TelePacific
                                   303 Colorado St., Suite 2075
                                                                   Austin, TX 78701
14965698
           Mr D. TIRE SERVICE
                                                         Barstow, CA 92311
                                      1010 E Main St
14965706
           Muhammad Arif
                                Address Redacted
14965708
                                               New York, NY 10003
           Multiplan
                          115 Fifth Avenue
14965709
           Muna Sleva
                           Address Redacted
14965711
           Mustafa Amanyar
                                 Address Redacted
14965712
           Mustafa Bilal
                             1210 S. Brookhurst Street
                                                          Mustafa Bilal
                                                                            Anaheim, CA 92804
14965713
           Muwafag Geryagose
                                   Address Redacted
14965716
           Mylien Ho
                           Address Redacted
14965717
           Myra Hernandez
                                Address Redacted
14965718
           Myranda Poeppel
                                 Address Redacted
14965719
           Myrna Ramirez
                               Address Redacted
14965720
           Myrna Torresdey
                                 Address Redacted
14965721
           Myro Zamora
                             Address Redacted
14965723
           NACCME
                           104 Windsor Center Dr # 200
                                                          East Windsor, NJ 08520
14965746
           NANCY BLEVINS
                                   Address Redacted
                                   3810 Palm Dr
           NANCY PEALING
14965756
                                                     Bonita, CA 91902
14965761
           NANETTE HOLZER
                                    Address Redacted
14965764
                                    2476 Stirrup RD
           NAPA – BORREGO
                                                         Borrego Springs, CA 92004
           NAPONIC, MEARL 8851 Center Dr. # 100 L
NATIONAL COUNCIL FOR BEHAVIORAL HEALTH
14965766
                                                            La Mesa, CA 91942
                                                                   P.O. Box 745709
14965784
                                                                                         Atlanta, GA
           30374-5709
           NATIONAL LGBTQ TASK FORCE
                                                  1050 Connecticut Ave NW, Suite 65500
                                                                                            Washington, DC
14965785
           NATIONAL MINORITY AIDS CO
14965786
                                                 1000 Vermont Avenue, NW, Suite 200
                                                                                         Washington, DC
           20005-4903
14965787
           NATIONAL NETWORK ORAL HEALTH
                                                        181 E 56th Avenue, Suite 410
                                                                                        Denver, CO 80216
14965788
           NATIONAL NOTARY ASSOCIATION
                                                     9350 DeSoto Avenue
                                                                              Chatsworth, CA 91311
14965789
           NATIONAL PHARMACEUTICAL SERVICES
                                                            P.O. Box 407
                                                                              Boys Town, NE 68010
14965790
           NATIONAL SERVICE & CONTROLS, INC
                                                          34880 Avenue G
                                                                              Yucaipa, CA 92399
14965791
           NATIONAL STUDENT CLEARINGHOUSE
                                                          2300 Dulles Station Blvd., Suite 220
                                                                                                Herndon, VA
           20171
14965792
           NATIONAL TECHNICAL INFORMATION SERVICE
                                                                   5301 Shawnee Rd.
                                                                                         Alexandria, VA 22312
                                       5500 Meadows Road, Suite 500
                                                                         Lake Oswego, OR 97035
14965794
           NAVEX GLOBAL, INC
14965796
           NAVINET
                           179 Lincoln Street
                                                 Boston, MA 02111
14965800
                          4917 Waters Edge Drive, Suite 165
                                                               Raleigh, NC 27606
           NCCHCA
14965801
                       1100 13th St. NW. Third Floor
                                                         Washington, DC 20005
           NCOA
14965804
           NEIGHBORHOOD FIRE PROTECTION
                                                      4580 Orange Hill
                                                                            Fallbrook, CA 92028
                                   P.O. Box 549271
14965807
           NEJM Knowledge+
                                                        Waltham, MA 02454-9271
14965819
           NETFORTRIS/FONALITY
                                          5340 Legacy Drive, Suite No. 155
                                                                              Plano, TX 75024
                                            7800 South Elati St, Suite 300
14965820
           NETGAIN SOLUTIONS LLC
                                                                             Littleton, CO 80120
           NETWORK SERVICES SOLUTIONS LLC- MISC 3
NETWORK SERVICES SOLUTIONS LLC-QWEST-400
                                                               3700 Barron Way
14965822
                                                                                     Reno, NV 89511-2388
14965823
                                                                      3700 Barron Way
                                                                                           Reno, NV
           89511-2388
```

```
14965824
           NETWORK SERVICES SOLUTIONS LLC-QWEST-401
                                                                      3700 Barron Way
                                                                                            Reno, NV
            89511-2388
14965825
           NETWORKS BY DESIGN
                                          231 W TENTH ST
                                                                 TRACY, CA 95376
14965826
           NEVA CHAUPPETTE, PSY. D.
                                               Address Redacted
14965830
           NEW DIRECTIONS
                                    P.O. Box 6729
                                                       Leawood, KS 66206-0729
14965832
           NEW INNOVATIONS, INC.
                                                                      Uniontown, OH 44685
                                           3540 Forest Lake Drive
14965837
           NEW YORK GIANT PIZZA
                                           356 North Magnolia
                                                                   El Cajon, CA 92020
                                                 City of Industry, CA 91745
14965838
                           17560 Rowland St
           NEWEGG
14965839
           NEXT WAREHOUSE
                                      14712 FRANKLIN AVE. SUITE #F
                                                                            Tustin, CA 92780
                                                                              Laguna Niguel, CA 92677
14965840
           NEXUS CONSULTING
                                       30251 Golden Lantern, Suite E #151
14965842
           NFP
                     3470 Mt. Diablo Blvd. Suite A100
                                                          Lafayette, CA 94549
                                                   1250 Capital of Texas HWY. South
14965843
           NFP INSURANCE SERVICES, INC.
                                                                                         Building Two, Suite
           600
                     Austin, TX 78746
14965844
           NFP PROPERTY & CASUALTY SERVICES, INC.
                                                                8201 North Hayden Road
                                                                                             Scottsdale, AZ
            85258
14965847
           NICHOLSON PROPERTIES, INC
                                                7770 Regents Road, Suite 113
                                                                                 San Diego, CA 92122
14965851
           NICOLE GRIEGO
                                  Address Redacted
14965870
           NON PROFIT WEB ADVISOR
                                              Barton Oaks Plaza One, 901 S MoPac
                                                                                      Expy #140
                                                                                                     Austin, TX
           78746
14965890
           NORTHERN TOOL + EQUIPMENT
                                                   2800 Southcross Drive West
                                                                                  Burnsville, MN 55306
14965893
                                           Chantilly, VA 20153-0832
           NPDB
                       P.O. Box 10832
                                 1867 E FLORIDA
                                                       Springfield, MO 65803
14965894
           NSC GROUP LL
14965902
           NU-VUE WINDOW FILMS, INC
                                                2235 Enterprise Way, Ste 170
                                                                                 Escondido, CA 92029
14965724
           Nada Alsawah
                              Address Redacted
14965725
           Nada Juma
                           Address Redacted
14965726
           Nadia Gonzalez
                               Address Redacted
14965727
           Nadia Hermiz
                              Address Redacted
14965728
           Nadia Kurdi
                            Address Redacted
14965729
           Nadine Guerrero
                                Address Redacted
14965730
           Nagham Algamal
                                 Address Redacted
14965731
                               Address Redacted
           Nagham Butris
14965732
           Nahla Bahnan
                              Address Redacted
14965733
           Nahran Al Ogaidi
                                 Address Redacted
14965734
           Nahum Bautista
                                Address Redacted
14965735
           Nahun Castillo
                               Address Redacted
14965736
           Nailah Jackson
                               Address Redacted
14965737
           Naim Uka
                           Address Redacted
14965738
           Najah Tohey
                             Address Redacted
14965739
           Najla Al Habeeb
                                Address Redacted
14965740
           Nakita M. Hooper
                                 Address Redacted
14965741
           Nakyeyune Harriet
                                  Address Redacted
14965742
           Nalleli Parra
                             Address Redacted
14965743
           Nallely Sanchez
                                Address Redacted
14965744
           Nan Nanessence
                                Address Redacted
14965745
            Nancy Benton
                              Address Redacted
14965747
           Nancy Contreras
                                Address Redacted
14965748
           Nancy Flores-Cordero
                                      Address Redacted
14965749
           Nancy Guzman
                               Address Redacted
14965750
           Nancy Leal
                            Address Redacted
           Nancy Lopez Santoyo
14965751
                                     Address Redacted
14965752
           Nancy Mendez Cisneros
                                       Address Redacted
14965753
           Nancy Navarro
                               Address Redacted
14965754
           Nancy Orozco
                              Address Redacted
           Nancy Palacios
Nancy Romero
14965755
                               Address Redacted
14965757
                               Address Redacted
14965758
           Nancy Sandoval
                                Address Redacted
           Nancy Ybarra
Nancy Zuniga Rocha
Naomi Matthew
14965759
                              Address Redacted
14965760
                                    Address Redacted
14965762
                                Address Redacted
14965763
           Naomie Novelo
                               Address Redacted
14965765
           Napat Nguyen
                              Address Redacted
            Narda Velazquez
14965767
                                 Address Redacted
                                  Address Redacted
14965768
           Narder M Tammar
14965769
           Natalia Arias Hernandez
                                       Address Redacted
14965770
           Natalia Cardenas
                                 Address Redacted
14965771
           Natalia Mercado
                                Address Redacted
14965772
           Natalie Gonzalez
                                 Address Redacted
14965773
           Natalie Padilla
                               Address Redacted
14965774
           Natalie Urzua
                              Address Redacted
14965775
           Natalie Whang
                               Address Redacted
           Nataliya Hammer
14965776
                                 Address Redacted
14965777
           Nataly Rosales
                               Address Redacted
14965778
           Natanael Jarquin
                                Address Redacted
           Nathalia Cedano
14965779
                                Address Redacted
14965780
           Nathalie Hernandez
                                   Address Redacted
14965781
           Nathaly Martinez
                                 Address Redacted
14965782
           Nathan Kamer
                              Address Redacted
14965783
           National Assoc of Community Health Ctr
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 51 of 76

```
14965793
           Navdeep Atwal
                                Address Redacted
14965795
           Navex Global-PolicyTech-EthicsPoint
                                                     5500 Meadows Road, Suite 500
                                                                                        HRIS Integration
                                                                                                              Lake
            Oswego, OR 97035
            Nayeli Bustamante
14965797
                                   Address Redacted
14965798
            Nazdar Gorgees
                                Address Redacted
           Nazhly S Rodriguez
14965799
                                    Address Redacted
14965802
            Neftali Galarza
                                Address Redacted
14965803
           Neida Flores
                             Address Redacted
14965805
            Neighborhood Healthcare
                                         425 N Date St
                                                            Escondido, CA 92025
14965806
           Nejat Refou
                             Address Redacted
14965808
            Nekkeya Kelly
                               Address Redacted
14965809
            Nelissa Nestor-John
                                     Address Redacted
14965810
           Nely Curiel
                            Address Redacted
14965811
            Nenita Bostick
                               Address Redacted
14965812
                                 Address Redacted
            Nereida Terrazas
14965813
            Nereida Yanez
                               Address Redacted
14965814
            Nereyda L Perez
                                 Address Redacted
14965815
            Nery Valdez
                             Address Redacted
14965816
                                        2405 Transportation Avenue
           Ness Dental Corporation
                                                                        National City, CA 91950
            Ness Dental Corporation
14965817
                                        600 B Street, Suite 2491
                                                                    Sharif Faust Lawyers Ltd-Matthew J Faust
                                                                                                                  San
            Diego, CA 92101
                         3060 Olsen Drive
14965818
            NetApp
                                               San Jose, CA 95128
14965821
                         2300 Oracle Way
                                                Austin, TX 78741
            NetSuite
            Neven Al Khayyat
14965827
                                  Address Redacted
14965828
           Neven Alkas
                             Address Redacted
            Nevena Antonov
14965829
                                 Address Redacted
14965831
            New Directions Behavioral Health
                                                 P.O. Box 6729
                                                                    Leawood, KS 66206-0729
14965833
            New Millennium Dental Group-A Arakelyan
                                                           10917 Paramount Blvd
                                                                                      Downey, CA 90241
            New Millennium Dental Group-A Arakelyan
14965834
                                                           19523 E. Cypress Street
                                                                                       Covina, CA 91724
            New Millennium Dental Group-A Arakelyan
                                                                                          Bleau Fox, a PLC - Megan Ann
14965835
                                                           2801 West Empire Avenue
            Childress
                          Burbank, CA 91504
14965836
           New Millennium Dental Group-A Arakelyan
                                                           599 Inland Center Drive, Suite 110
                                                                                                 San Bernardino, CA
            92408
14965841
            Neyda Rodriguez-Loo
                                       Address Redacted
            Nicanor Villa
14965845
                              Address Redacted
14965846
            Nicholas (Nick) Priest, Esq.
                                           Address Redacted
14965848
            Nicolas Pantoja
                                Address Redacted
14965849
            Nicolas Vasquez
                                 Address Redacted
14965850
            Nicole Flores
                              Address Redacted
14965852
           Nicole Hackett
                               Address Redacted
14965853
            Nicole Robbins
                                Address Redacted
14965854
           Nicole Rodriguez
                                  Address Redacted
14965855
            Nicole Schmidt
                                Address Redacted
14965856
           Nicole Suetos
                              Address Redacted
14965857
            Nidia Herrera
                              Address Redacted
14965858
           Nihad Gabr
                            Address Redacted
14965859
            Nikki Washington
                                  Address Redacted
14965860
           Nikol Krachtus
                                Address Redacted
14965861
            Nikolai Pregler
                                Address Redacted
14965862
           Nithya Venugopal
                                  Address Redacted
14965863
            Noe Alvarenga
                                Address Redacted
14965864
            Noehmy Ochoa
                                Address Redacted
            Noelhy Delgadillo
14965865
                                  Address Redacted
14965866
            Noemi Iturriaga Gonzalez
                                         Address Redacted
14965867
                                Address Redacted
           Noemi Sanchez
14965868
            Noemy Hernandez-Angel
                                         Address Redacted
                               Address Redacted
14965869
           Nohemi Lopez
14965871
            Noor Al Aloosi
                                Address Redacted
14965872
           Noor Al Mubark
                                 Address Redacted
14965873
            Noor Al Yousuf
                                 Address Redacted
14965875
            Noor Al-Timimi
                                 Address Redacted
14965874
            Noor Alsendi
                              Address Redacted
14965876
           Noor Mansour
                               Address Redacted
14965877
            Nora Chacon Enriquez
                                      Address Redacted
14965878
           Nora Gutierrez Castaneda
                                         Address Redacted
            Norma Alicia Quinones Celis
14965879
                                            Address Redacted
            Norma Campos
14965880
                                Address Redacted
14965881
            Norma Cervantes
                                  Address Redacted
14965882
            Norma Diaz
                             Address Redacted
14965883
            Norma Gonzalez
                                 Address Redacted
14965884
           Norma Leon
                             Address Redacted
14965885
           Norma Mendez
                                Address Redacted
14965886
            Norma Palomino
                                 Address Redacted
14965887
            North County Womens Health Services
                                                     488 E. Valley Parkway, Suite 400
                                                                                          Escondido, CA 92025
            North County Womens Specialists
14965888
                                                 488 E. Valley Parkway, Suite 400
                                                                                      Escondido, CA 92025
            North Rialto Drug
                                  531 E Foothill Blvd
14965889
                                                          Rialto, CA 92376
14965891
            Noteware & Rosa Government Relations
                                                       1201 K St Ste 1030
                                                                               Sacramento, CA 95814
14965892
           Noteware Government Relations
                                               1201 K Street, Suite 1030
                                                                             Sacramento, CA 95814
```

```
14965895
           Nubia Lopez
                            Address Redacted
           Nudos Pharmacy
14965896
                              455 N Magnolia Ave
                                                       El Cajon, CA 92020-3606
           Nuevo Water Company
14965897
                                     30427 11th St.
                                                       Nuevo, CA 92567
14965898
           Nury Lopez
                           Address Redacted
14965899
           Nury Lopez
                           Address Redacted
           Nuvie Nunez
14965900
                            Address Redacted
14965901
           Nuview Union School District
                                           29780 Lakeview Ave
                                                                   Nuevo, CA 92567
14965903
           Nygelle Pace
                            Address Redacted
14965906
           OB Hospitalist Group
                                   777 Lowndes Hill Road Building 1
                                                                       Greenville, SC 29607
                                                  Greenville, SC 29606-7717
14965908
                                P.O. Box 16717
           OBHG California
                                      1250-I Newell Ave. #209
14965909
           OBJECT HEALTH, LLC
                                                                  Walnut Creek, CA 94596
14965910
           OCCUPATIONAL HEALTH CENTERS OF CA
                                                           P.O. BOX 3700
                                                                              RANCHO CUCAMONGA, CA
           91729-3700
14965912
           OCCUPATIONAL SERVICES, INC.
                                                 6397 Nancy Ridge Dr
                                                                         San Diego, CA 92121
                             5225 Canyon Crest Dr Ste 71
14965913
           OCTOCLEAN
                                                            Riverside, CA 92507
14965916
           OEM OPTIC
                            22431 Antonio Pkwy B160
                                                         Rancho Santa Margarita, CA 92688
14965920
           OFF DUTY OFFICERS, INC.
                                                                 Vista, CA 92081
                                          2365 La Mirada Dr
                                    P.O. BOX 29248
14965921
           OFFICE DEPOT BSD
                                                        Phoenix, AZ 85038-9248
14965922
           OFFICE DEPOT-AMEX
                                       6600 North Military Trail
                                                                  Boca Raton, FL 33496
14965930
           OLIVE GARDEN
                                8984 International Dr
                                                        Orlando, FL 32819
           OMNI SHOREHAM HOTEL
14965936
                                          2500 Calvert St NW
                                                                 Washington, DC 20008
           OMNICARD
                                                       Pleasanton, CA 94588
14965937
                            6220 Stoneridge Mall Rd
                                              Cincinnati, OH 45263-7472
14965938
           OMNISYS
                          P.O. Box 637472
14965939
           ON HOLD MARKETING WORKS
                                               P.O. Box 79948
                                                                   Houston, TX 77279
           ON THE FLY TERMITE & PEST CONTROL
14965940
                                                         78-206 Varner Rd.
                                                                               D235
                                                                                          Palm Desert, CA
           92211
           ON THE FLY TERMITE & PEST CONTROL – Anza
14965941
                                                               78206 Varner Rd. STE D #235
                                                                                                Palm Desert, CA
           92211
14965945
           OPEN TEXT INC.
                                 24685 Network Place
                                                         Chicago, IL 60673-1246
14965947
           OPHTHALMIC INSTRUMENTS INC.
                                                  1 Musick
                                                                Irvine, CA 92618
14965949
           OPTUM 360
                           2175 Park Pl
                                            El Segundo, CA 90245-4705
14965950
           OPTUM HEALTH - Private Sector
                                               13625 Technology Dr
                                                                        Eden Prairie, MN 55344
14965954
           ORACLE AMERICA, INC.
                                        15612 Collections Center Drive
                                                                         Bank of America Lockbox
                        Chicago, IL 60693
           Services
14965955
           ORANGE BOOK
                                625 N 4th St
                                                Mt. Vernon, WA 98273
14965956
           ORANGE GLEN HS FOOTBALL BOOSTERS CLUB
                                                                1254 N Ivy
                                                                                Escondido, CA 92026
14965957
           ORANGECREST LANDSCAPING & MAINTENANCE
                                                                  19192 Wyler Rd
                                                                                      Perris, CA 92570
                                                P.O. Box 780518
14965959
           ORASURE TECHNOLOGIES, INC
                                                                    Philadelphia, PA 19178-0518
           OREILLY AUTO PARTS
                                                          Springfield, MO 65801–9464
14965960
                                       P.O. Box 9464
           ORIENTAL TRADING
14965962
                                     P.O. Box 2308
                                                        Oriental Trading Company
                                                                                    Omaha, NE 68103-2308
14965965
                                       13468 Mesquite Rd
                                                             Whitewaters, CA 92282
           ORTEGA, MARCELINA
14965966
           ORTHOPEDIC OUTFITTERS INC
                                                3320 Railroad Ave Ste B
                                                                           Redding, CA 96001-3510
14965967
           ORTHOPLUS
                             3606 OCEAN RANCH BLVD
                                                             Oceanside, CA 92056
14965973
                                6024 CULLIGAN WAY
           OSI BATTERIES
                                                           Minnetonka, MN 55345
14965978
           OUR COMMUNITY USA LLC
                                            2000 Palm Beach Lakes Blvd
                                                                           West Palm Beach, FL 33409
                             799 W. Coliseum Way Mic
14525 Highway 7 – Suite 315
14965979
           OVERSTOCK
                                                      Midvale, UT 84047
14965980
           OZ GROUP INC
                                                               Minnetonka, MN 55345
14965904
                         Address Redacted
           Oanh Lam
14965905
           Oasis Mobile Home Park
                                      5145 CA-78
                                                       Borrego Springs, CA 92004
14965907
           Obdulia Ayala-Aguirre
                                     Address Redacted
14965911
           Occupational Safety & Health ADMIN
                                                  1515 Clay Street, Suite 1901
                                                                                Oakland, CA 94612
14965914
           Odessey Mora
                             Address Redacted
14965915
           Odeth Čoloma
                             Address Redacted
14965917
           Ofelia Barajas
                             Address Redacted
14965918
           Ofelia Gomez
                            Address Redacted
14965919
           Off Duty Officers Incorporated
                                           2365 La Mirada Dr
                                                                  Vista, CA 92081
14965923
           Office of State Health Planning & Dev
                                                  2020 West El Camino Avenue, Suite 1016
                                                                                           Song-Brown Grant
           AGMT
                       Sacramento, CA 95833
14965924
           Office of the United States AG
                                           950 Pennsylvania Avenue, NW
                                                                            Washington, DC 20530
14965925
           Ohio Department of Taxation
                                          4485 Northland Ridge Blvd.
                                                                        Columbus, OH 43229
                                Address Redacted
14965926
           Olga Meza Muniz
14965927
           Olga Ornelas
                            Address Redacted
14965928
           Olga Santiago Martinez
                                     Address Redacted
14965929
           Olimpia Reyes
                             Address Redacted
14970484
           Olivia J. Scott
                             BRYAN CAVE LEIGHTON PAISNER LLP
                                                                         1920 Main Street, Suite 1000
                                                                                                        Irvine,
           California 92614–7276
           Olympic Public Storage
14965931
                                     118 12th Street
                                                        Ramona, CA 92065
14965932
           Omar Aguirre
                             Address Redacted
14965933
           Omar Castro Ramos
                                  Address Redacted
14965934
           Omar Santos-Lopez
                                  Address Redacted
14965935
           Omid Zebariadi
                              Address Redacted
14965942
           One Ring Networks
                                  2030 Powers Ferry Road SE, Suite 200
                                                                         Atlanta, GA 30339
           One Ring Networks
                                            60 Athens, TX 75751–1360
Lindon, UT 84042–1227
14965943
                                  P.O. Box 1360
14965944
                          2578 W 600 N
           Open Edge
14965946
                               DEPT LA 24661
                                                   Pasadena, CA 91185-4661
           OpenSesame Inc
           Optum – United Behavioral Health
14965948
                                              13625 Technology Dr
                                                                       Eden Prairie, MN 55344
14965951
           Optum Pharmacy–Avella of Deer Valley
                                                   13625 Technology Dr
                                                                            BriovaRx of Maine
                                                                                                  Eden Prairie.
           MN 55344
```

```
14965952
           Optum360
                         13625 Technology Dr
                                                  Eden Prairie, MN 55344
           OptumHealth Financial Services, Inc
14965953
                                                4316 Rice Lake Road
                                                                       Duluth, MN 55811
14978985
           Oracle America, Inc. SII to NetSuite, Inc.
                                                   c/o Shawn M. Christianson, Esq.
                                                                                     Buchalter PC
                                                                                                      425
                                   San Francisco, CA 94105
           Market St., Suite 2900
14965958
           Oras Yousif
                          Address Redacted
14965961
           Org for Safety Asepsis &Prevention
                                               P.O. Box 6297
                                                                 20 avenue de Florimont
                                                                                           Annapolis, MD
           21401
           Orlondo Johnson
14965963
                               Address Redacted
14965964
           Oro Grande School District
                                      P.O. Box 386
                                                         Oro Grande, CA 92368
14965968
           Oscar Dominguez
                               Address Redacted
14965969
           Oscar E Sanchez
                               Address Redacted
14965970
           Oscar Fuentes
                            Address Redacted
14965971
           Oscar Rivero Cruz
                                Address Redacted
14965972
           Oshin Safarian
                             Address Redacted
14965974
           Osvaldo A Loya
                              Address Redacted
14965975
           Ottoniel Molina–Espinoza
                                      Address Redacted
14965976
           Oualid Tellissi
                             Address Redacted
14965977
           Oudinarath Donelvilay
                                   59155 Gilman Road
                                                          Anza, CA 92539
14965981
                            14525 Highway 7 Suite 315
           Oz Group Inc
                                                         Customer Contact Services-Chris Rosales
                                                                                                  Minnetonka,
           MN 55345
14965982
           P&M CUSTOM UPHOLSTERY
                                            83095 Indio Blvd
                                                                 Indio, CA 92201
           PACIFIC ALARM SERVICE, ARLANZA
14965984
                                                     521 Wellwood Ave
                                                                           Under SSD Alarm
                                                                                                Beaumont, CA
           92223
14965985
           PACIFIC ALARM SERVICE, EHC
                                               14407 Meridian Parkway
                                                                          Under SSD Alarm
                                                                                               Riverside, CA
           92518
14965986
           PACIFIC ALARM SERVICE, EHC BLDG B
                                                       14407 Meridian Parkway
                                                                                  Under SSD
           ALARM
                        RIverside, CA 92518
14965987
           PACIFIC ALARM SERVICE, INC, DHS
                                                    14407 Meridian Parkway
                                                                               Riverside, CA 92518
           PACIFICARE – PPO
                                                       Cypress, CA 90630
14965991
                                  5995 Plaza Drive
14965994
           PAESSLER AG
                              Thurn-und-Taxis-Str. 14
                                                          Nuremberg, 90411 Germany
14965997
                                                          Grand Prairie, TX 75050
           PAINT SUPPLY
                               905 Avenue T, Suite 312
14965999
           PALM DESERT RESUSCITATION
                                               73700 Dinah Shore Dr #107
                                                                             Palm Desert, CA 92211
           PALM MIRROR AND GLASS, INC.
14966000
                                                68325 East Ramon
                                                                      Cathedral City, CA 92234
14966001
           PALM PLAZA PROPERTY
                                         P.O. Box 56014
                                                            Sherman Oaks, CA 91413
14966002
           PALM SPRINGS POLICE DEPT-ALARM UNIT
                                                           P.O. Box 3294
                                                                             Palm Springs, CA 92263
           PALM SPRINGS PRIDE
                                      329 W. Mariscal Rd
                                                             Palm Springs, CA 92262
14966003
           PALMS TO PINES PRINTING & PROMO PRODUCTS
14966006
                                                                 P.O. Box 2827
                                                                                   Palm Desert, CA 92261
           PANDA EXPRESS
                                 1717 Walnut Grove Ave #100
14966015
                                                                Rosemead, CA 91770
14966016
           PANERA
                         3630 S Geyer Rd, Suite 100
                                                      St Louis, MO 63127
14966020
                                                                       Camarillo, CA 93010
          PAPAEFTHIMIOU APC
                                      1601 Carmen Drive, Suite 212D
                                                   18 Washington Street, Suite 308
14966021
           PAR8O, INC (Now under CLOUDMED)
                                                                                    Canton, MA 02021
14966022
           PARADISE CHEVROLET TEMECULA
                                                    27360 Ynez Road
                                                                        Paradise Chevrolet Cadillac
                                                                                                      Temecula.
           CA 92591
14966023
           PARADISE SHORES AQUARIUM PROFESSIONALS
                                                                2738 Tonto Way
                                                                                    San Diego, CA 92117
                                                                  Chicago, IL 60693-0126
14966024
           PARAGARD DIRECT
                                    12601 Collection Center Dr.
14966025
           PARALLEL CAPITAL PARTNERS
                                               P.O. Box 741334
                                                                    Los Angeles, CA 90074-1334
14966026
           PARATA SYSTEMS, LLC
                                        P.O. Box 638203
                                                            Cincinnati, OH 45263-8203
           PARCHMENT TRANSCRIPT
14966027
                                           7001 N Scottsdale Rd, Suite 1050
                                                                             Scottsdale, AZ 85253
                                                            Las Vegas, NV 89109
14966029
           PARIS LAS VEGAS
                                  3655 S Las Vegas Blvd
14966031
           PARKING NETWORK, INC.
                                          350 S Figueroa St #420
                                                                   Los Angeles, CA 90071
                                                                 Chicago, IL 60696-0272
14966033
           PARMED PHARMACEUTICALS
                                              P.O. Box 90272
14966035
           PARTS SOURCE
                                777 LENA DRIVE
                                                     Aurora, OH 44202
14966036
           PARTY CITY
                             25 Green Pond Road Suite 1
                                                          Rockaway, NJ 07866
14966038
           PATIENT POINT HOSPITALS SOLUTIONS
                                                        11408 Otter Creek South Road
                                                                                       Mabelvale, AR
           72103
14966049
           PATRICIA HERRERA
                                     Address Redacted
                                    Address Redacted
14966057
           PATRICIA RAMIREZ
14966062
           PATRICIA Y. PASTRANA
                                        Address Redacted
14966068
           PATTERSON DENTAL
                                     P.O. Box 732865
                                                          Dallas, TX 75373-2865
14966072
           PAULA NGUYEN SAUNDERS
                                            Address Redacted
14966079
           PAYLESS 4 LIGHTING
                                      23699 Shelburne Rd
                                                            Shaker Heights, OH 44122-2171
14966081
           PAYPAL
                        2211 N 1st St
                                         San Jose, CA 95131
                                                      Jacksonville, FL 32204
14966083
           PC LIQUIDATIONS
                                  140 Stockton St.
14966089
           PEDSTEST ONLINE, LLC
                                        1013 Austin Ct.
                                                           Nolenville, TN 37135
                                             P.O. Box 208
           PEGLEG MINI STORAGE UNIT
14966091
                                                               Borrego Springs, CA 92004
14966092
                                                   Reading, PA 19607
           PENSKE
                        2675 Morgantown Road
14966093
           PENSKE FORD
                              8970 La Mesa Blvd
                                                     La Mesa, CA 91942
                                            11883 MAGNOLIA ST 40
14966094
           PERFORMANCE AUTO CARE
                                                                         Riverside, CA 92503
           PETER M. DOWNEY, APC
14966100
                                         Address Redacted
           PETES ROAD SERVICE
                                      2230 E. Orangethorpe Ave.
14966102
                                                                   Fullerton, CA 92831
14966110
           PHOENIX FINANCIAL SERVICES LLC
                                                    8902 Otis Avenue, Suite 103A
                                                                                    Indianapolis, IN
           46216-1077
          PHSI Pure Water Finance PHYLLIS TARBELL
14966111
                                      475 Half Day Rd
                                                          Lincolnshire, IL 60069
14966114
                                   Address Redacted
           PIETZCH, BONNETT & WOMACK, P.A.
14966118
                                                     3101 N. Central Ave, Suite 940
                                                                                     Phoenix, AZ 85012
           PILOTSHOP.COM
                                 225 Airport Circle
                                                      Corona, CA 92878
14966120
                                   71680 HWY 111 STE D
14966122
           PINOTS PALATTTE
                                                             Rancho Mirage, CA 92270
14966123
          PITNEY BOWES
                                P.O. BOX 371887
                                                     PITTSBURGH, PA 15250-7887
```

```
14966124
          PITNEY BOWES -8000-9090-0980-8705
                                                     5 Churchill Place, 10th Floor
                                                                                  London, E14 5HU United
           Kingdom
14966125
          PITNEY BOWES -8000-9090-1041-5821
                                                     3001 Summer St
                                                                        Stamford, CT 06905
          PITNEY BOWES GLOBAL FINANCIAL-0017392137
                                                               P.O. Box 981022
14966126
                                                                                   Boston, MA
          02298-1022
          PITNEY BOWES INC
14966127
                                                       Pittsburgh, PA 15250-7896
                                   P.O. Box 371896
14966128
          PJ PARTS
                         1807 FM 2532
                                           Sumner, TX 75486
                                                        EL CAJON, CA 92020-3606
14966129
          PJN Pharmacies
                             455 N MAGNOLIA AVE
          PLANET HOLLYWOOD RESORT
14966130
                                              3667 S Las Vegas Blvd
                                                                       Las Vegas, NV 89109
          PLATINUM WINDOW TINTING
                                             2830 E Vista Chino Rd
14966131
                                                                      Palm Springs, CA 92262
                                   83827 Tamarisk St.
                                                         Indio, CA 92201
14966132
          PLAZA TOWING INC
                       2125 Park Blvd, MS32
                                               San Diego, CA 92101
14966133
          PLCRC
14966134
          PLURALSIGHT
                              42 Future Way
                                                Draper, UT 84020
14966135
                          81-730 HWY 111 #7
                                                 Indio, CA 92201
          PMCS LLC
          PMI-SD CHAPTER
                                 8895 TOWNE CENTRE DRIVE
14966136
                                                                   SAN DIEGO, CA 92122
14966137
           PNC EQUIPMENT FINANCE, LLC – 161391
                                                       P.O. Box 51657
                                                                         Los Angeles, CA 90051–5957
                                       940 SOUTH ROCHESTER AVE. # D
14966138
          PNP OFFICE FURNITURE
                                                                          ONTARIO, CA 91761
14984497
          POLSINELLI LLP
                                Randye B. Soref
                                                   2049 Century Park East, Suite 2900
                                                                                      Los Angeles, CA
          90067
14966140
          POLYCASE
                           1305 Chester Industrial Parkway
                                                            Avon, OH 44011
14966141
          PORKYLAND
                             4645 Carmel Mountain Rd., Suite 201
                                                                  San Diego, CA 92130
14966144
                                                                 Broomfield, CO 80021
          PORTABLE HANDWASHING
                                           8810 WEST 116TH
          POST MASTER – LIS DOOT 15 GILPIN AVE
14966145
                                                           Hauppauge, NY 11788
14966146
                                          15 Gilpin Ave.
                                                             Hauppauge, NY 11788
14966147
                                                     11251 Rancho Carmel Dr.
                                                                                 San Diego, CA
          92199-9621
          PRECISION BACKFLOW 83–453 Ruby Ave PRECISION GLASS & MIRROR, INC. 783 P
14966153
                                                            Indio, CA 92201
                                                  783 Palmyrita Avenue, Suite B
14966154
                                                                                 Riverside, CA 92507
          PREMIER LEARNING SOLUTIONS
14966161
                                                660 American Ave, Suite 203
                                                                              King of Prussia, PA 19406
14966163
          PRICE SELF STORAGE
                                     4635 Morena Blvd
                                                           San Diego, CA 92117
          PRICES NURSERY
                                                   LA QUINTA, CA 92248
                                 P.O. BOX 6329
14966164
          PRIME CLINICAL SYSTEMS
                                          3675 East Huntington Drive, Suite A
14966167
                                                                               Pasadena, CA 91107
          PRIMETIME GARAGE DOORS AND GATES
                                                        3525 Del Mar Heights Rd #858
14966168
                                                                                        San Diego, CA
          92130
14966173
          PRINTING FOR LESS
                                   100 PFL Way
                                                    Livingston, MT 59047
                           8504 Commerce Ave.
14966174
          PRINTSTAR
                                                   San Diego, CA 92121
14966175
          PRIORITY HEALTHCARE DIST
                                             P.O. BOX 978510
                                                                  DBA CURASCRIPT SD
           75397-8510
14966180
          PROCARE CLOUD SERVICE
                                          1 West Main St., Ste 201
                                                                     Medford, OR 97501
          PROFESSIONAL GLASS INSTALLERS
                                                                     141 N 5TH ST
14966181
                                                   P.O. Box 1051
                                                                                       Brawley, CA
14966183
          PROFLAME / AMERIGAS
                                       P.O. Box 7155
                                                          Pasadena, CA 91109-7155
14966184
          PROGRESSIVE MANAGEMENT SYSTEMS
                                                       1521 West Cameron Ave
                                                                                  West Covina, CA 91790
          PROPER SOLUTIONS STAFFING
                                              74-900 Highway 111
                                                                                  Indian Wells, CA 92210
14966187
                                                                      Ste. 114
14966188
          PROPIO LANGUAGE SERVICES
                                              P.O. Box 12204
                                                                Overland Park, KS 66282-2204
          PROSPECT MEDICAL SYSTEMS
                                              600 City Parkway West, Suite 800
14966189
                                                                                Formerly Vantage Medical
                     Orange, CA 92868
          PROTECTION ONE SECURITY
14966190
                                            P.O. Box 219044
                                                                Kansas City, MO 64121-9044
                                                                 Nuremberg, 90411 Germany
14966191
                     Thurn-und-Taxis-Str. 14
                                                Paessler AG
14966192
          PSYCH CENTERS OF SAN DIEGO
                                               P.O. Box 609001
                                                                   San Diego, CA 92160
          PUEBLO UNIDO CDC
14966194
                                    78150 Calle Tampico
                                                           La Quinta, CA 92253
14966195
          PULSEWAY
                           6 – 9 Trinity Street
                                                MMSOFT Design Ltd.
                                                                         Dublin 2, D02 EY47 Ireland
          PURDY CLEAN CAR WASH & STORAGE
                                                     2185 West Main St
14966196
                                                                           Barstow, CA 92311
14966197
          PURE WATER TECHNOLOGY OF SAN DIEGO
                                                           4683 Mission Gorge Place,
                                                                                      San Diego, CA
          92120
14966198
          PYANGO, LLC
                              269 Montgomery Dr
                                                    Mokelumne Hill, CA 95245
14965983
          Pablo DeLeon
                            Address Redacted
14965993
          PacifiCare Health Systems
                                      9900 Bren Road East
                                                             Minnetonka, MN 55343
14965988
          Pacific Alarm Services
                                   521 WELLWOOD AVE
                                                             BEAUMONT, CA 92223
                                           2575 E. Camelback Rd., Ste. 700
14965989
          Pacific Life Insurance Company
                                                                            Phoenix, AZ 85016
14965990
          Pacificare
                       P.O. Box 1459
                                          UnitedHealth Group
                                                                Minneapolis, MN 55440-1459
14965992
                                        P.O. Box 31053
          Pacificare Behavioral Health
                                                           Laguna Hills, CA 92654–1053
14965995
          Paige Thompson
                              Address Redacted
                                                         Newport Beach, CA 92660-2616
14965996
          Pain Management
                               3857 Birch St. Suite 605
          Palm Care Pharmacy
                                 505 N Mollison Ave Suite 102
14965998
                                                                El Cajon, CA 92021
14966004
          Palm Springs Unified School District
                                               150 District Center Drive
                                                                         Palm Springs, CA 92264
                                Address Redacted
14966005
          Palmer Fontanarosa
                             Address Redacted
14966007
          Paloma Rochin
          Palomar Excess & Surplus Insurance Co
14966008
                                                  4400 W 78th Street Suite 120
                                                                                Bloomington, MN 55435
14966009
          Palomar Excess & Surplus Insurance Co
                                                 7979 Ivanhoe Avenue, Suite 500
                                                                                  La Jolla, CA 92037
14966010
          Palomar Medical Center
                                    2125 Citracado Parkway, Suite 130
                                                                   Escondido, CA 92029
                                                              Poway, CA 92064
14966011
          Palomar Pomerado Health
                                      15615 Pomerado Road
14966012
          Pam Sime
                        Address Redacted
14966013
          Pamela Coleman
                              Address Redacted
14966014
           Pamela McEvoy
                              Address Redacted
14966017
                           Address Redacted
          Paola Avalos
14966018
          Paola Garcia
                           Address Redacted
```

14966019

Paola Gonzalez

Address Redacted

```
14966028
                 Pardip Singh
                                            Address Redacted
14966030
                 Parke Vista Pharmacy/Wellpartner
                                                                                                                    Riverside, CA 92503-4001
                                                                          3838 Sherman Dr, Ste 1
14966032
                 Parkview Medical Plaza/ WellPartner
                                                                              20800 SW 115th Avenue Suite 100
                                                                                                                                       Tualatin, OR 97062
                                              Address Redacted
14966034
                 Particia Lopez
14966037
                 PathSolutions, Inc.
                                                     1030 East El Camino Real#173
                                                                                                        Sunnyvale, CA 94087
14966039
                                             Address Redacted
                 Patric Schine
14966040
                 Patricia Arredondo
                                                     Address Redacted
14966041
                 Patricia Arroyo-Perez
                                                         Address Redacted
14966042
                  Patricia Bahena
                                                Address Redacted
14966043
                 Patricia Bulters
                                                Address Redacted
14966044
                 Patricia Depriest
                                                  Address Redacted
14966045
                  Patricia English
                                                Address Redacted
                 Patricia Garcia
14966046
                                               Address Redacted
14966047
                  Patricia Garcia-Zaldivar
                                                            Address Redacted
                                                   Address Redacted
14966048
                 Patricia Gutierrez
14966050
                  Patricia Ibarra
                                              Address Redacted
14966051
                 Patricia Jacobo
                                               Address Redacted
14966052
                 Patricia Lovenich
                                                   Address Redacted
                 Patricia Luna
14966053
                                             Address Redacted
14966054
                  Patricia Martinez
                                                  Address Redacted
14966055
                 Patricia Mendoza
                                                   Address Redacted
14966056
                 Patricia Pastrana
                                                  Address Redacted
14966058
                 Patricia Rodriguez
                                                    Address Redacted
14966059
                  Patricia Romero
                                                 Address Redacted
                 Patricia Ruiz
14966060
                                             Address Redacted
14966061
                 Patricia Torres
                                               Address Redacted
14966063
                  Patrick Craychee
                                                  Address Redacted
14966064
                 Patrick Halili
                                             Address Redacted
14966065
                  Patrick Rhodes
                                               Address Redacted
14966066
                 Patrick Thomas
                                                Address Redacted
14966067
                  Patrise Kae Amiscaray
                                                          Address Redacted
14966069
                 Patty Mendoza
                                               Address Redacted
14966070
                 Paul Gorgis
                                           Address Redacted
14966071
                 Paul Rancour
                                             Address Redacted
14966073
                  Paula Urzua
                                           Address Redacted
                 Paulina Abrica
14966074
                                               Address Redacted
14966075
                  Paulina Munoz
                                               Address Redacted
14966076
                 Paulina Rodriguez Esparza
                                                               Address Redacted
14966077
                  Pauline Erwin-Silva
                                                       Address Redacted
                 Pauline Sechang Address February State Sta
14966078
                                                 Address Redacted
14966080
                                                                              San Diego, CA 92111
14966082
                 Paz Gutierrez
                                             Address Redacted
14966084
                  Pedro Alvarez
                                              Address Redacted
14966085
                 Pedro Caudillo
                                               Address Redacted
14966086
                 Pedro Dairo
                                           Address Redacted
14966087
                  Pedro Gomez
                                             Address Redacted
14966088
                  Pedro Ramirez
                                               Address Redacted
14966090
                 Peg Leg Mini Storage
                                                        1527 Palm Canyon Dr
                                                                                                Borrego Springs, CA 92004
14966095
                 Perla Cervantes Loa
                                                      Address Redacted
14966096
                 Perla Luna
                                          Address Redacted
14966097
                  Perpemelo Balasta
                                                    Address Redacted
14966098
                  Perris Elementary School District
                                                                        143 East First Street
                                                                                                             Perris, CA 92570
14966099
                                         Address Redacted
                 Peter Hara
14966101
                  Peter Philp
                                          Address Redacted
14966103
                 Petra Hinojosa
                                               Address Redacted
14966104
                 Pfizer Inc.
                                         235 E 42nd St
                                                                     New York, NY 10017-5703
                 Pharmacy First
                                               11880 College Blvd., Ste. 420
                                                                                                 Overland Park, KS 66210
14966105
14966106
                  Pharmacy First – Borrego Pharmacies
                                                                               11880 College Blvd., Ste. 420
                                                                                                                                Overland Park, KS 66210
14966107
                 Philadelphia Insurance Companies
                                                                                                                  Bala Cynwyd, PA 19004
                                                                          One Bala Plaza, St 100
14966108
                 Philadelphia Insurance Companies
                                                                          P.O. Box 70251
                                                                                                         Philadelphia, PA 19176-0251
14966109
                 Philip Watkinson
                                                  Address Redacted
                  Phuc Pham
14966112
                                          Address Redacted
14966113
                 Phyllis Jastrok
                                              Address Redacted
                  Physician Health Network Medical Corp.
14966115
                                                                                   732 CARNEGIE DR 125
                                                                                                                               San Bernardino, CA 92408
                                                                       1525 FLORIDA AVENUE, #A
14966116
                  Physicians for Healthy Hospitals
                                                                                                                           HEMET, CA 92543
                                                          631 Park Avenue
14966117
                                                                                          King of Prussia, PA 19406
                  Pierce Leahy Archives
14966119
                  PillPack
                                      250 Commercial St, Suite 2012
                                                                                          Manchester, NH 03101
                  Pima Medical Institute
                                                                                                       Tucson, AZ 85711-3019
14966121
                                                         40 N Swan Road Suite 100
                  Pioneer Funding Group, LLC
15002692
                                                                  232 W. 116th St.
                                                                                                   P.O. Box 1735
                                                                                                                                New York, NY 10026
14966139
                  PointCare
                                        1212 Broadway Plaza, Suite 2100
                                                                                               Walnut Creek, CA 94596
14966142
                                                Address Redacted
                 Porsche Lozada
14966143
                 Porsha Wilson
                                               Address Redacted
14966148
                  Pourshirazi & Homayoun
                                                             27192 NEWPORT ROAD, SUITE 2
                                                                                                                        MENIFEE, CA 92584
14966149
                  Pourshirazi & Youssefi
                                                           8152 E. BAILEY WAY
                                                                                                    ANAHEIM, CA 92808
                                                                    c/o Brinkman Law Group, PC
14971767
                  Pourshirazi & Youssefi Dental
                                                                                                                      543 Country Club Drive, Suite
                                                                        Email: firm@brinkmanlaw.com
                             Wood Ranch, CA 93065
14966150
                 Pourshirazi & Youssefi Dental Corp
                                                                            31569 Canyon Estates Dr. Suite 120
                                                                                                                                      Homayoun
                                          Lake Elsinore, CA 92532
                  Pourshirazi
```

```
14966151
           PracticeLink, LTD.
                                 P.O. Box 100
                                                  Hinton, WV 25951
14966152
           Precious Mwelwa
                               Address Redacted
14966156
           Premier Health Care Management
                                             124 W Main St Ste 120
                                                                      El Cajon, CA 92020
                                            124 West Main Street Suite 120
14966157
           Premier Healthcare Management
                                                                             Daryl R. Priest
                                                                                               El Cajon, CA
           92020
14966158
           Premier Healthcare Management
                                            3465 CAMINO Del Rio S, Ste. 250
                                                                                c/o Stephen H Fitch
                                                                                                      San
           Diego, CA 92108
           Premier Healthcare Management, Inc
14966159
                                                450 B Street, Suite 1410
                                                                          Kyle Harris LLP-Laura Kimberly
                       San Diego, CA 92101
           Premier Healthcare Management, Inc.
14966160
                                                124 W Main Street
                                                                      Ste 240
                                                                                  El Cajon, CA 92020
                                        410 Cloverleaf Dr.
14966162
           Premier Specialty Pharmacy
                                                             Baldwin Park, CA 91706
14966155
           Premier and Summit Healthcare Management
                                                      124 W Main St Ste 120
          Primus Cable 2011 224 W 3990 Concours Suite 500 3710 RIJECTAL BO
                                                                                El Cajon, CA 92020
                                                                Ontario, CA 91764
14966165
14966166
                                         3710 RUFFIN ROAD
                                                                  SAN DIEGO, CA 92123
                            29415 Hunco Way
           Primus Cable
                                                 Lake Elsinore, CA 92530
14966169
14966170
           Principal
                       P.O. Box 9394
                                          Des Moines, IA 50392
14966171
           Principal Life Insurance Company
                                             711 High St
                                                             Des Moines, IA 50392
14966172
           Principal Life Insurance Company
                                             711 High St
                                                             Des Mones, IA 50392
14966176
           Priscilla Angel
                             Address Redacted
14966177
           Priscilla Fortanel
                               Address Redacted
14966178
           Priscilla Ramos
                             Address Redacted
                              Address Redacted
14966179
           Privada Gervais
14966182
           Professional Medical Warehouse, Inc.
                                                3500 E. Tachevah Drive, Suite F
                                                                                 Palm Springs, CA 92262
14966185
           Project ECHO
                         1 University of New Mexico
                                                          MSC07 4245
                                                                           Albuquerque, NM 87131
                                                                       El Cajon, ĈA 92020
14966186
           Promenade Square, LLC
                                   124 W Main Street
                                                           Ste 240
                                            P.O. Box 609001
14966193
           Psychiatric Centers of San Diego
                                                                San Diego, CA 92160
14966200
           QPCS, LLC
                           2400 ROCKEFALLER DRIVE
                                                           Ceres, CA 95307
          QUALITY CHEVROLL
QUALITY CHOICE
QUALITY COLLISION
68–404 Com
188 Lee Avenue
724 N
           QUALITY CHEVROLET OF ESCONDIDO
                                                      1550 AUTO PARK WAY
14966201
                                                                                  ESCONDIDO, CA 92029
                                43157 W 9 Mile Rd.
14966202
                                                         Novi, MI 48375
                                     68-404 Commercial Road
14966203
                                                                 Cathedral City, CA 92234
          QUALITY FORMS 188 L
QUALITY LOGO PRODUCTS
                                           renue Brooklyn, NY 11211
724 NORTH HIGHLAND AVE
14966204
14966205
                                                                             Aurora, IL 60506
14966206
           QUANTIX CONSULTING, INC
                                            W138S7143 Sherwood Cir
                                                                         Muskego, WI 53150
14966207
           QUANTUM CORPORATION
                                           224 Airport Parkway, Suite 550
                                                                           San Jose, CA 95110
                                              Dallas, TX 75373-5777
14966209
           OUENCH
                         P.O. BOX 735777
                                                          Pasadena, CA 91110-2411
14966211
           QUEST DIAGNOSTICS
                                     P.O. Box 912411
14966212
           QUEST DIAGNOSTICS – 42027189
                                                P.O. Box 740709
                                                                     Atlanta, GA 30374-0709
           QUEST DIAGNOSTICS - 92004434
                                                P.O.BOX 50368
14966213
                                                                    LOS ANGELES, CA 90074-0368
14966214
           QUEST DIAGNOSTICS – 92234003
                                                P.O.BOX 50368
                                                                    LOS ANGELES, CA 90074-0368
14966215
           QUEST DIAGNOSTICS - TROTH - 91752473
                                                         P.O.BOX 50368
                                                                             LOS ANGELES, CA
           90074-0368
           QUICK FIX AUTO GLASS
                                                            Palm Desert, CA 92255
14966216
                                        P.O. Box 11784
14966217
           QUIEL SIGNS
                             272 South I Street
                                                  San Bernardino,, CA 92410
14966218
           OUILL.COM
                            P.O. Box 37600
                                               Philadelphia, PA 19101–0600
14966219
           QUINMAR JANITORIAL, INC.
                                            P.O. Box 1026
                                                             Highland, CA 92346
                                                      Los Angeles, CA 90084-9665
14966220
           QUINN COMPANY
                                  P.O. Box 849665
14966199
           Qais Khadher
                           Address Redacted
14966208
                                224 Airport Parkway, Suite 550
           Quantum Tape Library
                                                                    San Jose, CA 95110
                                       630 Allendale Road, Suite 200
14966210
           Quench Rental Agreement
                                                                      King of Prussia, PA 19406
           RAFAEL GONZĂLEZ
                                    Address Redacted
14966232
14966239
           RAHEEM, TONEE
                                 Address Redacted
14966256
           RAMONA CROSSINGS LLC.
                                          14916 Gavan Vista Rd
                                                                    Poway, CA 92064
                                                 4419 Brighton Court
14966257
           RAMONA DIAGNOSTIC SERVICES
                                                                        Hemet, CA 92544
14966260
           RAMONA SELF STORAGE
                                       118 12th Street
                                                           Ramona, CA 92065
14966261
                                                   11626 Sterling Ave Suite D
           RAMSEY BACKFLOW & PLUMBING
                                                                                Riverside, CA 92503
                                                                          SAN DIEGO, CA 92128
14966264
           RANCHO BERNARDO INN
                                         17550 BERNARDO OAKS DR
           RANDY BUSCHBAUM-DBA THE SIGH WAREHOUSE
                                                                   79090 Ave 42, B-101
14966268
                                                                                           Bermuda Dunes, CA
14966272
           RANSOM BROTHER LUMBER SUPPLY
                                                     532 B ST
                                                                   Ramona, CA 92065
14966273
           RAP FOUNDATION
                                  41550 Eclectic St.
                                                       Palm Desert, CA 92260
14966275
           RAPID RECALLING.COM
                                       5724 SW 40th Place
                                                               Ocala, FL 34474
                                   27927 Kalmia Avenue
14966292
           RAYMOND DELONE
                                                             Moreno Valley, CA 92555
                                                             Upland, ČA 91786
14966294
           REACH OUT
                            1126 W.Foothill Blvd Ste 250
14966296
           READY REFRESH (CV ADMIN – 0035537414)
                                                          900 Long Ridge Road, Building 2
                                                                                             Stamford, CT
           06902-1138
           READY REFRESH – MVK
14966295
                                        P.O. Box 856158
                                                             Louisville, KY 40285-6158
           READY REFRESH BY NESTLE (ARROWHEAD)
14966298
                                                             P.O. Box 856158
                                                                                 0026921411 SF
                                                                                                    Louisville.
           KY 40285-6158
           READY REFRESH BY NESTLE (ARROWHEAD)
14966297
                                                             P.O. Box 856158
                                                                                 Louisville, KY 40285-6158
           REGISTRY OF CHARITABLE TRUSTS
14966312
                                                 P.O. Box 903447
                                                                    Sacramento, CA 94203–4470
14966313
           REGULATORY, RISK, COMPLIANCE SPECIALISTS
                                                                5700 Stoneridge Mall Road, Suite
           #350
                     Pleasanton, CA 94588
           REID T SHIGENO
14966314
                                 Address Redacted
           RELAY FOR LIFE OF SAN JACINTO
RELIANCE PUBLIC RELATIONS, INC
14966315
                                                  181 N Ramona Blvd
                                                                         San Jacinto, CA 92583
14966316
                                                   P.O. Box 1944
                                                                      El Centro, CA 92244
                                                                       Asheville, NC 28801
           RENAISSANCE ASHEVILLE HOTEL
                                                  31 Woodfin Street
14966318
14966324
           REPUBLIC SERVICES-3-0467-0027041AL-2707
                                                             P.O. Box 78829
                                                                                 Republic Services
                    Phoenix, AZ 85062-8829
```

#467

```
14966325
           REPUBLIC SERVICES-3-0467-0701707
                                                     P.O. Box 78829
                                                                         Republic Services #467
                                                                                                   Phoenix, AZ
           85062-8829
14966326
           REPUBLIC SVCS 3-0467-0701099-CHILDRENS
                                                             P.O. Box 78829
                                                                                 Republic Services
                     Phoenix, AZ 85062-8829
           #467
14966328
           RESERVATION COUNTER LLC
                                              4700 W Daybreak Parkway, Suite 100
                                                                                     South Jordan, UT 84009
                                                          Bethesda, MD 20817
14966330
                                10400 Fernwood Road
           RESIDENCE INN
14966331
           RETAIL MANAGEMENT SOLUTIONS (RMS)
                                                           4535 Lacey Blvd SE
                                                                                  Lacey, WA 98503
14966338
                     P.O. Box 77631
           RGIS
                                        Detroit, MI 48277
14966342
           RHINO LININGS
                                Address Redacted
14966344
           RHYME UNIVERSITY
                                     P.O. BOX 64784
                                                          Saint Paul, MN 55164
14966355
           RICOH # 13673708
                                  P.O. Box 31001-0850
                                                           Pasadena, CA 91110-0850
14966354
           RICOH - 13699059
                                  P.O. Box 31001-0850
                                                           Pasadena, CA 91110-0850
           RICOH USA, INC-CUST 13727808
RIGHT SPACE STORAGE 53
                                                P.O. Box 31001-0850
14966357
                                                                         Pasadena, CA 91110-0850
14966358
                                        53-301 HWY 111
                                                              Coachella, CA 92236
                                         8200 Auto Drive
14966366
           RIVERSIDE CHEVROLET
                                                             Riverside, CA 92504
14966367
           RIVERSIDE COMMUNITY HEALTH FOUNDATION
                                                                 4275 Lemon Street
                                                                                       Riverside, CA 92501
14966369
           RIVERSIDE COMMUNITY HOSPITAL
                                                    4445 Magnolia Avenue – M4
                                                                                   Riverside, CA
           92501-4199
14966371
           RIVERSIDE COUNTY CLERK
                                            38-686 El Cerrito Road
                                                                       Palm Desert, CA 92211
14966372
           RIVERSIDE COUNTY FIRE DEPARTMENT
                                                         2300 Market St. # 150
                                                                                 Riverside, CA 92501
           RIVERSIDE COUNTY SHERIFF – ELENA WALTER
14966374
                                                                46200 OASIS ST. RM B15
                                                                                             INDIO, CA
14966375
           RIVERSIDE COUNTY SHERIFF - RUIZ, FRANK
                                                                                               RIVERSIDE, CA
                                                             4095 LEMON ST., 4TH FLOOR
14966376
           RIVERSIDE COUNTY SHERIFF - SANDRA MORA
                                                                46200 OASIS STREET, ROOM B15
                                                                                                     INDIO, CA
           92201
14966377
           RIVERSIDE COUNTY TREASURER-TAX COLLECTOR
                                                                    38-686 El Cerrito Rd
                                                                                            Palm Desert, CA
           92211
14966378
           RIVERSIDE COUNTY TREASURER-TAX COLLECTOR
                                                                    P.O. Box 12005
                                                                                        Riverside, CA
           92502-2205
           RIVERSIDE CTY CHILDREN &FAMILIES COM RIVERSIDE CTY SHERIFFS OFFICE 46200
14966379
                                                             585 Technology Court
                                                                                      Riverside, CA 92507
                                                  46200 OASIS ST., ROOM B15
14966380
                                                                                   FLORES.
                        INDIO, CA 92201
           CARINA
14966381
           RIVERSIDE DEPT PUBLIC HEALTH IMMUNE PRGM
                                                                  P.O. Box 7600
                                                                                     3900 Sherman Drive, Suite
                 Riverside, CA 92513
           RIVERSIDE PUBLIC UTILITIES
14966386
                                              3900 Main St
                                                               Riverside, CA 92522-0144
14966388
           RIVERSIDE UNIFIED SCHOOL DISTRICT
                                                       P.O. Box 2800
                                                                          Riverside, CA 92516-2800
14966390
           RIVIERA DENTAL COMPANY
                                             P.O. Box 1894
                                                                Borrego Springs, CA 92004
14966391
           RIVIERA PALM SPRINGS
                                         1600 N INDIAN CANYON
                                                                      Palm Springs, CA 92262
14966392
           RJW, Notary Public
                                  81944 HWY 111, Ste E
                                                           Indio, CA 92201
14966395
           ROBERT H. RICCIARDI
                                       Address Redacted
14966405
           ROBERTS, PATRICIA.
                                     Address Redacted
14966420
           RODOLFO, REY
ROHR, INC
                                Address Redacted
14966425
                           8200 Arlington Ave
                                                  Riverside, CA 92503
14966428
           RON ANDERSEN
                                 Address Redacted
14966458
           ROSALES, NATALY
                                    Address Redacted
14966467
           ROSE MACISAAC
                                  Address Redacted
14966472
           ROSEANN ELIZALDE
                                     Address Redacted
           ROTARY CLUB OF CATHEDRAL CITY
14966483
                                                     P.O. Box 654
                                                                       Cathedral City, CA 92234
14966485
           ROTH STAFFING COMPANIES, LP
                                                 450 N. State College Blvd
                                                                             Orange, CA 92868
14966486
           ROTO-ROOTER
                                2141 Industrial Ct., Ste. D
                                                            Vista, CA 92081
14966487
           ROTTO, GARY
                               Address Redacted
14966494
           ROYAL CRESCENT HOTEL
                                          16 Royal Crescent
                                                                Bath, BA1 2LS United Kingdom
           RR DONNELLEY 14100 Lear Blvd, Suite 130 RS REPAIR & CUSTOM TRAILERS INC. 626
14966496
                                                              Reno, NV 89506
14966497
                                                      6260 Rutland Ave., Suite 21
                                                                                    Riverside, CA 92503
14966498
           RSA CONFERENCE
                                   176 Middlesex Turnpike
                                                              Bedford, MA 01730
14966499
                     72-048 Wobum Ct.
                                           Thousand Palms,, CA 92276
           RSD
14966502
           RUBBER STAMPS
                                  P.O. BOX 445
                                                    Butler, WI 53007
14966507
           RUBIOS
                        1902 Wright Place, Suite 300
                                                       Carlsbad, CA 92008
14966510
           RUBY SERNA
                              Address Redacted
14966512
           RUDYS TERMITE & PEST CONTROL INC.
                                                        43-639 N. Jackson Street
                                                                                    Indio, CA 92201
14966513
           RUHS Foundation
                                P.O. Box 9850
                                                   Moreno Valley, CA 92552
14966515
                                Address Redacted
           RUSSELL WEBB
14966521
           RX SECURITY
                              1200 N. Federal Hwy Suite 200, #50-126
                                                                        Boca Raton, FL 33432
14966522
           RX SYSTEMS, INC
                                  121 Point West Blvd.
                                                          St. Charles, MO 63301
14966221
           Rachel Abel
                           Address Redacted
14966222
           Rachel Chen
                           Address Redacted
14966223
           Rachel Hayes
                            Address Redacted
14966224
           Rachel Huerta
                             Address Redacted
14966225
           Radhwan Shamoon
                                 Address Redacted
14966226
           Rady Childrens Hospital
                                     3020 Childrens Way
                                                             San Diego, CA 92123
14966227
           Rafael A Sanchez
                                Address Redacted
14966228
           Rafael Arcone
                             Address Redacted
14966229
           Rafael Escoto
                             Address Redacted
14966230
           Rafael Garcia
                            Address Redacted
14966231
                             Address Redacted
           Rafael Gomez
14966233
           Rafael Juan
                           Address Redacted
14966234
                             Address Redacted
           Rafael Magana
```

```
14966235
            Rafael Roman Serrano
                                      Address Redacted
14966236
           Rafael Zarate
                              Address Redacted
14966237
            Rafah Pulus
                             Address Redacted
14966238
            Raghda Jouni
                              Address Redacted
14966240
            Rajanae Eason
                               Address Redacted
            Rajesh Shah
14966241
                             115 N McKinley Street Suite 105
                                                                 Rajesh Shah
                                                                                  Corona, CA 92879
14966242
            Rajit Gohil
                            Address Redacted
14966243
            Rakhima Chakhalova
                                     Address Redacted
14966244
            Ralph Leggett
                              Address Redacted
14966245
            Rami Issim Maroki
                                   Address Redacted
14966246
            Ramin Amani
                              Address Redacted
14966247
            Ramiro Avila
                              Address Redacted
14966248
            Ramiro Lopez
                               Address Redacted
14966249
            Ramon Angel
                              Address Redacted
14966250
            Ramon Castillo
                                Address Redacted
14966251
            Ramon Diaz Romero
                                     Address Redacted
14966252
            Ramon Fuentes
                                Address Redacted
14966253
            Ramon H Rodriguez
                                    Address Redacted
14966254
           Ramon Ortiz Rivera
                                    Address Redacted
14966255
            Ramon Soria
                              Address Redacted
14966258
            Ramona Medical Diagnostic Services
                                                    1516 Main St, Suite 103
                                                                               Ramona, CA 92065
14966259
            Ramona Rodriguez
                                   Address Redacted
14966262
            Ran Regev
                            Address Redacted
14966263
            Rana Toma
                            Address Redacted
14966265
            Rancho Medical Clinic
                                      13050 Heacock St
                                                             Moreno Valley, CA 92553
14966266
            Rancy Hermiz
                               Address Redacted
14966267
            Randall Woodend
                                  Address Redacted
            Randy Fedorchuk
14966269
                                  Address Redacted
14966270
            Randy Hernandez
                                  Address Redacted
14966271
            Randy Rath
                            Address Redacted
14966274
            Raphael Alvarez
                                 Address Redacted
14966276
            Raquel Benavides
                                  Address Redacted
14966277
            Raquel Canfield
                                Address Redacted
14966278
            Raquel Ledesma
                                 Address Redacted
14966279
            Raquel Lopez
                              Address Redacted
            Raquel Ortiz
14966280
                             Address Redacted
14966281
            Raquel Palacios
                                Address Redacted
14966282
            Rariba Ghorab
                               Address Redacted
14966283
            Rasaura Navarro
                                 Address Redacted
14966284
            Rasha Hanna
                              Address Redacted
14966285
            Raul Castro
                            Address Redacted
14966286
            Raul Gutierrez Gonzalez
                                        Address Redacted
14966287
            Raul Jimenez
                              Address Redacted
                           Address Redacted
14966288
            Raul Pena
14966289
            Raul Ruacho
                             Address Redacted
14966290
           Raul Sotelo
                            Address Redacted
14966291
            Raul Villasenor
                                Address Redacted
14966293
            Raymond Shaheen
                                   Address Redacted
                                                           P.O. Box 856158
14966299
            Ready Refresh By Nestle-PSFH 6701447331
                                                                                Louisville, KY 40285-6158
14966300
            Rebeca Garcia Centeno
                                       Address Redacted
14966301
            Rebeca Mariscal
                                 Address Redacted
14966302
            Rebeca Siguenza
                                 Address Redacted
14966303
            Rebecca Martin
                                Address Redacted
14966304
            Rebecca Ross
                              Address Redacted
14966305
           Rebecca Sayre
                               Address Redacted
14966306
            Rebeka Oviedo
                                Address Redacted
            Regal Medical Group
14966307
                                     8510 Balboa Blvd STE 150
                                                                    Northridge, CA 91325
14966308
            Regence
                         1800 Ninth Avenue
                                                 Seattle, WA 98101
14966309
            Regents - University of CA, Riverside
                                                                                Riverside, CA 92521
                                                     900 University Avenue
14966310
            Regents of the University of Colorado
                                                    3100 Marine St Ste 481572
                                                                                   Boulder, CO 80309-0001
14966311
            Regina Castillo
                                Address Redacted
14966317
            Remigio Gonzalo
                                  Address Redacted
           Rene Bulle
14966319
                            Address Redacted
14966320
            Rene Hickey
                             Address Redacted
14966321
            Renee Jimenez
                               Address Redacted
            Renova Energy Corp.
                                                                     Palm Desert, CA 92260
14966322
                                     75–181 Mediterranean Ave.
14966323
            Republic Services
                                  18500 North Allied Way
                                                              Phoenix, AZ 85054
                                                     32600 Guevara Dr.
14966327
            Res. Commercial Electrical Contracting
                                                                             Temecula, CA 92592
14966329
            Reshae Crawford
                                 Address Redacted
14966332
            RevenueHealth Systems – PMG Technologies
                                                            100 Rue du Landais Street
                                                                                         Test and Research
                       Blainville, QC J7C 5C9 Canada
            Centre
14966333
            Reyna Reynaga-Vargas
                                        Address Redacted
14966334
            Reynaldo Castillo
                                  Address Redacted
            Reynaldo Cruz-Perez
14966335
                                      Address Redacted
14966336
            Reynaldo Hacinas
                                  Address Redacted
            Reynaldo Saucedo
14966337
                                  Address Redacted
14966339
            Rhett Armstrong
                                 Address Redacted
14966340
                           Address Redacted
           Rhett Papa
```

```
14966341
            Rhianna Atwood
                                  Address Redacted
14966343
            Rhule B Lockett
                                 Address Redacted
14966345
            Rhythm Tech Productions, LLC
                                               264 N Pennsylvania Ave
                                                                            Colton, CA 92324
            Ricardo Navarro
14966346
                                 Address Redacted
14966347
            Ricardo Vela
                              Address Redacted
14966348
            Ricciardi, Robert H
                                    Address Redacted
14966349
            Richard Arcelao
                                 Address Redacted
            Richard Mizer
14966350
                               Address Redacted
14966351
            Rick Reed
                            Address Redacted
14966352
            Rick Seman
                             Address Redacted
14966353
            Ricky Lewis
                              Address Redacted
                             P.O. Box 31001-0850
                                                       Pasadena, CA 91110
14966356
            Ricoh USA
                                                                    El Cajon, CA 92019
14966359
            Rigoberto Antonio-Hernandez
                                              2152 Euclid Ave
            Rigoberto Rodriguez
14966360
                                     Address Redacted
14966361
                               Address Redacted
            Rita Andersen
14966362
            Rita Martinez
                               Address Redacted
14966363
            Rita Saeed
                            Address Redacted
14966364
            Rita Salas
                           Address Redacted
14966365
            Rite Aid - Thrifty Payless Inc.
                                              1200 Intrepid Avenue, 2nd Floor
                                                                                  Philadelphia, PA 19112
14966368
            Riverside Community Hospital
                                              4445 Magnolia Ave
                                                                       Graduate Medical Education Faculty
                                                                                                               Riverside,
            CA 92501
            Riverside County
14966370
                                  Attn Prof Firefighters Benevolent Fund
                                                                            P.O. Box 1131
                                                                                                Riverside, CA
            92502
14966373
            Riverside County Office of Education
                                                     3939 Thirteenth Street
                                                                                Riverside, CA 92501
14966382
            Riverside District office
                                        3737 Main Street, Suite 300
                                                                        Riverside, CA 92501-3337
14966383
            Riverside Latino Commission
                                             1612 First Street
                                                                   Alcohol & Drug Abuse
                                                                                               Coachella, CA 92236
                                            4102 Orange Street
14966384
            Riverside Police Department
                                                                    Riverside, CA 92501
14966385
            Riverside Public Utilities
                                         3900 Main St
                                                            Riverside, CA 92522-0144
14966387
            Riverside Recovery Resources
                                              3757 Elizabeth Street
                                                                        Riverside, CA 92506
                                                                                     Moreno Valley, CA 92555
            Riverside University Health System
14966389
                                                   26600 Cactus Ave. Suite #300
14966393
            Robert Davis
                              Address Redacted
14966394
            Robert Fontanilla
                                  Address Redacted
            Robert Haynes
14966396
                                Address Redacted
14966397
            Robert Jimenez
                                Address Redacted
                              Address Redacted
14966398
            Robert Reves
14966399
            Robert Segnit
                               Address Redacted
14966400
            Robert Siskin
                               Address Redacted
14966401
            Robert Torres
                               Address Redacted
14966402
            Roberto Contreras
                                   Address Redacted
14966403
                                      Address Redacted
            Roberto Romero Cota
14966404
            Roberto Vazquez
                                  Address Redacted
14966406
            Robin Parish
                              Address Redacted
14966407
            Robin Ramirez
                                Address Redacted
14966408
            Robin Stettoe
                               Address Redacted
14966409
            Robyn Long
                              Address Redacted
14966410
            Rocio Jimenez
                                Address Redacted
14966411
            Rocio Murguia
                                Address Redacted
14966412
            Rocio Orozco
                               Address Redacted
14966413
            Rocio Pablo
                             Address Redacted
14966414
            Rocio Romo
                              Address Redacted
14966415
            Rocio Torres
                              Address Redacted
14966416
            Rocky Steele
                              Address Redacted
14966417
            Rod Hagan
                            Address Redacted
            Rodney Koenig
Rodolfo Hernandez
14966418
                                Address Redacted
14966419
                                    Address Redacted
14966421
            Rodrigo Refugio
                                  Address Redacted
14966422
            Rogelio Robles
                                Address Redacted
14966423
            Rogelio Solis Rodriguez
                                        Address Redacted
14966424
            Roger Peck
                             Address Redacted
14966426
            Rolando Viray
                                Address Redacted
14966427
            Romoland School District
                                          25900 Leon Road
                                                                Homeland, CA 92548
14966429
            Ron Palustre
                              Address Redacted
14966430
            Ron Starzinski
                                Address Redacted
14966431
            Ronagene Tecson
                                  Address Redacted
14966432
            Ronald Amos
                               Address Redacted
14966433
            Ronald Andersen
                                  Address Redacted
            Ronald Jay Palustre
14966434
                                    Address Redacted
14966435
            Ronald Reiter
                               Address Redacted
14966436
            Ronald Rodgers
                                 Address Redacted
14966437
            Ronald Scott
                              Address Redacted
14966438
            Ronald Tucker
                                Address Redacted
14966439
                               Address Redacted
            Ronen Efraim
14966440
            Ronnie Pena
                              Address Redacted
14966441
            Rosa Aguilera
                               Address Redacted
14966442
            Rosa Alcaraz
                              Address Redacted
14966443
            Rosa Amelia Cuevas Gastellum
                                               Address Redacted
14966444
            Rosa Castillo
                              Address Redacted
14966445
            Rosa Elena Martinez
                                     Address Redacted
```

```
14966446
           Rosa Espinoza
                              Address Redacted
14966447
           Rosa Galvan
                             Address Redacted
14966448
           Rosa Garcia
                            Address Redacted
14966449
           Rosa Gonzalez
                               Address Redacted
14966450
           Rosa Lopez Lopez
                                  Address Redacted
14966451
           Rosa Ortiz de Iniguez
                                    Address Redacted
14966452
           Rosa Perez
                           Address Redacted
14966453
           Rosa Silva
                           Address Redacted
14966454
           Rosa Sosa Gonzalez
                                   Address Redacted
14966455
           Rosa Valenzuela Walker
                                       Address Redacted
14966456
           Rosalba Flores
                              Address Redacted
14966457
           Rosalba Junco
                              Address Redacted
14966459
           Rosalia Cisneros
                                Address Redacted
14966460
           Rosalina Hickingbottom
                                       Address Redacted
14966461
                                  Address Redacted
           Rosalinda Simpson
14966462
           Rosalva Sosa
                             Address Redacted
14966463
           Rosario Alanis
                              Address Redacted
14966464
           Rosario Lopez
                              Address Redacted
14966465
           Rosario Rodriguez
                                  Address Redacted
14966466
           Rosaura Moreno
                                Address Redacted
14966468
           Rose Patricia Valadez Hernandez
                                               Address Redacted
14966469
           Rose Russo
                            Address Redacted
14966470
           Rose Silva
                           Address Redacted
14966471
           Rose Stephens
                              Address Redacted
14966473
           Roselia Ruiz
                             Address Redacted
14966474
           Roselle Jumamil
                                Address Redacted
14966475
           Rosemarie MacIsaac
                                    Address Redacted
14966476
           Rosemarie Williams
                                   Address Redacted
14966477
           Rosemary Hogenson
                                    Address Redacted
14966478
           Rosen Shingle Creek
                                    Address Redacted
14966479
           Rosendo Perez
                              Address Redacted
14966480
                              Address Redacted
           Rosie Arteaga
14966481
           Rosie Munoz
                             Address Redacted
14966482
           Rossy Perez
                            Address Redacted
14966484
           Roth Staffing
                             5962 La Place Court, Suite 265
                                                              Carlsbad, CA 92008
14966488
           Roxana Dominguez
                                   Address Redacted
14966489
           Roxana Robles
                               Address Redacted
           Roxanna Saavedra
14966490
                                  Address Redacted
14966491
           Roxanne Maligaya
                                  Address Redacted
14966492
           Roxanne Parkins
                                Address Redacted
14966493
           Roy Legarreta
                              Address Redacted
           Roys Desert Resource Center
14966495
                                           19531 Mc Lane St
                                                                 North Palm Springs, CA 92258
14966500
           Ru Mee Han
                            Address Redacted
14966501
           Ruaa Alsammarraie
                                   Address Redacted
14966503
           Ruben Morales
                               Address Redacted
14966504
           Ruben Ortiz
                            Address Redacted
14966505
           Ruben Rodriguez
                                Address Redacted
14966506
                                       Address Redacted
           Rubi Hernandez Aguilar
14966508
           Ruby Ayala-Garcia
                                   Address Redacted
           Ruby Fuentes
14966509
                             Address Redacted
14966511
           Rudolph Castillo
                                Address Redacted
14966514
           Russell Simon
                              Address Redacted
14966516
           Rustan Hicks
                             Address Redacted
14966517
           Rusul Alnaeb
                             Address Redacted
14966518
           Ruth Royal
                           Address Redacted
14966519
           Ruth Solares Aguilar
                                    Address Redacted
                           Address Redacted
14966520
           Ruth Tovar
14966523
           Ryan Dawood
                              Address Redacted
14966524
           S AND G PUMPING SERVICE
                                              73-680 Hwy 111, Suite 123
                                                                             Palm Desert, CA 92260
14966525
           S&R Towing Inc.
                                 P.O. Box 4366
                                                    Carlsbad, CA 92018-4366
14966528
           SABAH HADDAD
                                   166 N. 1ST ST #16
                                                          EL CAJON, CA 92021
14966530
           SABAH MANSOUR
                                    Address Redacted
14966532
           SACRAMENTO TRANSPORTATION SERVICE
                                                              P.O. Box 521
                                                                                West Sacramento, CA 95605
14966533
           SAFE KIDS WORLDWIDE
                                          1 Inventa Place 6th Fl. West
                                                                         Silver Spring, MD 20910
                                                              Cincinnati, OH 45263-3197
14966534
           SAFELITE AUTOGLASS
                                         P.O. Box 633197
14966535
           SAFIROS SANITATIONS
                                         P.O. Box 3071
                                                            National City, CA 91951
                             Address Redacted
14966540
           SAKI ONDA
14966553
           SAM ROBINSON
                                  Address Redacted
                                      11301 Old Georgetown Road
14966560
           SAMBA INSURANCE
                                                                      Rockville, MD 20852
14966563
           SAMESKY HEALTH, INC
                                          P.O. Box 670323
                                                                Dallas, TX 75267-0323
                             2101 SE Simple Savings Dr
14966567
           SAMS CLUB
                                                            Bentonville, AR 72716
14966568
                            1170 Peachtree Street, 9th Floor
           SAMSARA
                                                              Atlanta, GA 30309
14966573
           SAMUEL VENTURA
                                     Address Redacted
           SAN BERNARDINO &RIVERSIDE CTY FIRE EQUIP
14966574
                                                                   932 North D St.
                                                                                       San Bernardino, CA
14966575
           SAN BERNARDINO AREA CHAMBER OF COMMERCE
                                                                       P.O. Box 658
                                                                                         546 W. 6th St
                                                                                                           San
           Bernardino, CA 92402
```

	of 76
14966577	SAN BERNARDINO COUNTY CLERK 222 W. Hospitality Lane 1st Floor San Bernardino, CA
14966581	92415 SAN BERNARDINO VALLEY CONCERT 536 W. 11th St. THE GARCIA CENTER OF THE
14966582	ARTS San Bernardino, CA 92410 SAN DIEGO COUNTY AUDITOR AND CONTROLLER P.O. Box 129037 OFFICE OF REVENUE
14966583	AND RECOVERY San Diego, CA 92112–9037 SAN DIEGO COUNTY SHERIFF – OCAMPO, MARIA P.O. Box 85306 SAN DIEGO, CA
14966584	92186–5306 SAN DIEGO COUNTY SHERIFF–ANA R. HERRERA 325 S MELROSE, SUITE 2400 VISTA, CA 92081
14966585	SAN DIEGO COUNTY SHERIFF-CLAUDIA BARRON 325 S MELROSE, SUITE 2400 VISTA, CA 92081
14966586	SAN DIEGO COUNTY SHERIFF-DURAN, JULIANNA P.O. Box 85306 SAN DIEGO, CA 92186
14966587 14966588	SAN DIEGO COUNTY TREASURER-TAX COLL 1600 Pacific Hwy San Diego, CA 92101 SAN DIEGO COUNTY TREASURER-TAX COLL P.O. Box 129009 San Diego, CA 92112
14966589	SAN DIEGO CTY SHERIFF 325 S. MELROSE DRIVE, SUITE 2400 FAVIOLA GOMEZ MURILLO VISTA, CA 92081
14966590 14966591	SAN DIEGO CTY SHERIFF-DELLISA A GARCIA 325 S. MELROSE, SUITE 2400 VISTA, CA 92081 SAN DIEGO CTY SHERIFF-EFIGENIA M LOPEZ 325 S MELROSE, SUITE 2400 VISTA, CA
14966592	92081 SAN DIEGO CTY SHERIFF-GARIBALDI, CARLO P.O. Box 85306 SAN DIEGO, CA
14966593	92186–5306 SAN DIEGO CTY SHERIFF–JESSICA MARTINEZ 325 S MELROSE DRIVE, SUITE 2400 VISTA,
14966594	CA 92081 SAN DIEGO CTY SHERIFFS OFFICE 250 E MAIN STREET VALASQUEZ, LESLIE EL
14966595	CAJON, CA 92020–3949 SAN DIEGO CTY SHERIFFS OFFICE 325 S MELROSE, SUITE 2400 MURO,
14966596	SANDRA VISTA, CA 92081 SAN DIEGO CTY SHERIFFS OFFICE 325 SOUTH MELROSE DR STE 2400 ORTEGA, MARCIELINA VISTA, CA 92081–6692
14966597	SAN DIEGO EAST CTY CHAMBER OF COMMERCE 201 S. Magnolia Ave. El Cajon, CA 92020
14966601 14966602	SAN DIEGO LIVE SCAN 12937 Pomerado Rd Poway, CA 92064 SAN DIEGO POLICE DEPARTMENT P.O. Box 121431 MS 735 San Diego, CA 92112
14966604 14966605	SAN DIEGO SAFARI PARK P.O. Box 120551 San Diego, CA 92112 SAN DIEGO SHERIFFS DEPT P.O. Box 939062 Licensing Division San Diego, CA
14966606	92193–9062 SAN DIEGO SOCIETY FOR HUMAN Chapter 130 San Diego Soc of HR Mgmt Chapter 130 San Diego Soc of HR Mgmt
14966607	Chapter 130 San Diego, CA 92103 SAN DIEGO STATE UNIVERSITY 5250 Campanile Dr San Diego, CA 92182–1919
14966608	SAN DIEGO ZOO GLOBAL P.O. Box 120551 Warner Building – Event Sales San Diego, CA 92112
14966609	SAN JACINTO CHAMBER OF COMMERCE 2323 S. San Jacinto Ave. San Jacinto, CA 92583 SAN PEDRO OBGYN MEDICAL GROUP 1300 W. 65TH St. Suite 1A San Pedro, CA 90732
14966613 14966615	SAN PEDRO OBGYN MEDICAL GROUP 1300 W. 65TH St. Suite 1A San Pedro, CA 90732 SANDAG 1129 La Media Road San Diego, CA 92154
14966617	SANDMAN.COM 645 Lunt Ave Elk Grove Village, IL 60007
14966649	SANOFI PASTEUR, INC. 12458 Collections Center Dr Chicago, IL 60693
14966655	SANTOS YACUTE Address Redacted
14966656	SARA CASTELLON Address Redacted
14966675	SAV-ON CARPETS 83–151 Indio Blvd Indio, CA 92201
14966678	SBC TAX COLLECTOR 268 W. Hospitality Lane, First Floor San Bernardino, CA 92415–0360
14966679	SBCUSD 777 North F Street San Bernardino, CA 92410 SCHOOL NURSE SUPPLY 1745 Wallace Ave St. Charles, IL 60174
14966681 14966682	SCHOOL NORSE SUPPLY 1743 Wallace Ave St. Charles, IL 00174 SCHOOL-BASED HEALTH ALLIANCE 1010 Vermont Avenue, NW, Suite 600 Washington, DC 20005
14966688	SCRIP HESSCO P.O. BOX 6726 CAROL STREAM, IL 60197–6726
14966691	SDG&E 8326 Century Park Court San Diego, CA 92123–1530
14966692 14966693	SDG&E P.O. Box 25111 Santa Ana, CA 92799–5111 SDSU RESEARCH FOUNDATION 5250 Campanile Drive San Diego, CA 92182–1914
14966695	SEAN BARLOW Address Redacted
14966696 14966697	SEARS 3333 Beverly Rd Hoffman Estates, IL 60179 SECOND OPINION TELEMEDICINE SOLUTIONS 20695 S Western Ave Ste 200 Torrance, CA
14966700 14966702	90501 SEGAL MEDICARE EXPERTS, LLC 1982 Crescent Drive Signal Hill, CA 90755–5600 SELDA SHAMON Address Redacted
14966706	SENTINEL SECURITY SYSTEMS P.O. Box 300577 Escondido, CA 92030
14966708 14966716	SERENDIB LAW FIRM, APC SERRATORE LAW, APC 350 South Grand Avenue, Suite 355 Orange, CA 92868 Los Angeles, CA 90071
14966718 14966719	SERVIN, TERESA Address Redacted SERVPRO OF PALM DESERT 73–605 Dinah Shore Drive Bldg 1000J Palm Desert, CA
14966723	92211 SEVILLE CLASSICS 19401 Harborgate Way Torrance, CA 90501
14966724	SEWALL, SHARON A. Address Redacted
14966726 14966730	SEYFARTH SHAW LLP 3807 Collections Center Drive Chicago, IL 60693 SHAMROCK TECHNOLOGIES, LLC 222 North Pacific Coast Hwy, Ste 1620 El Segundo, CA 90245
	79 - 10

```
14966734
            SHANNON LYNN HART
                                             Address Redacted
            SHANNON SYSTEMS LLC
14966736
                                              P.O. Box 781147
                                                                       Philadelphia, PA 19178–1147
            SHANNON SYSTEMS LLC (B2BGateway.Net)
                                                                   173 Spark Street Brockton, MA 02302
14966737
                                                                     8344 Clairemont Mesa Blvd, Suite 201
            SHARECARE HEALTH DATA SERVICES, LLC
14966741
                                                                                                                 San Diego, CA
                                 1751 Richardson Street, Suite 5400
14966742
            SHAREGATE
                                                                          Montreal, QC H3K 1G6 CANADA
14966746
            SHARP HEALTH PLAN
                                           8520 Tech Way, Suite 200
                                                                             San Diego, CA 92123
14966747
            SHARP HEALTHCARE
                                           8695 Spectrum Center Blvd
                                                                             San Diego, CA 92123
14966758
            SHEPPARD MULLIN RIGHTER & HAMPTON LLP
                                                                        333 South Street, 43rd Floor
                                                                                                          Los Angeles, CA
            90071-1422
                                                       Bethesda, MD 20817
14966759
            SHERATON
                                10400 Fernwood Pl
            SHERWIN–WILLIAMS 101 W. Prospect Ave.
                                                                       Cleveland, OH 44115
14966762
                               Address Redacted
14966765
            SHEYLA CHEA
14966766
            SHI INTERNATIONAL CORP
                                                 P.O. BOX 952121
                                                                          DALLAS, TX 75395-2121
            SHOP KOMEN 13770 Noel Road Suite 801889 Dallas, TX 75380 SHOPLET SELECT 39 Broadway 20th Floor SHRED AND GO, INC. 28061 Jefferson Ave suite 8 Temecula, CA 92590
14966769
14966771
14966772
                                                                            Temecula, CA 92590
            SHRED AND GO, INC. 28061 Jefferson Ave suite 8 Temecula, CA SHRED—IT 28883 Network Place Chicago, IL 60673—1288 SHRM 1800 Duke Street Alexandria, VA 22314 SHURE STEP.COM P.O. Box 920706 Houston, TX 77292 SICKELS FABRIC & UPHOLSTERY 311 E. Valley Parkway Esc. SIGIS 111 Deerwood Rd, Suite 200 San Ramon, CA 94583 SIGNARAMA 41–945 Boardwalk, Ste. L Palm Desert, CA 92211 SIGNATURE TRUCK TORS 82573 India Rlyd. VEHICLE ACCES
14966773
14966774
14966775
14966778
                                                                                     Escondido, CA 92025
14966782
14966783
            SIGNATURE TRUCK TOPS
14966784
                                             82573 Indio Blvd
                                                                      VEHICLE ACCESSORIES & AUTO
            REPAIR
                         Indio, CA 92201
            SIMPLY STAMPS
                                     2021 St. Augustine Rd E
14966795
                                                                  Jacksonville, FL 32207
            SINA AGHAIE Address Redacted
SJ Medical Group 1695 S San Jacinto Ave.
SKILL EDUCATORS 340 S Lemon Ave 3410
SKILL PATH SEMINARS P.O. Box 804441
Kansas City, MO 64180
14966796
14966799
14966800
14966801
            SKYRIVER COMMUNICATION 7310 Miramar Rd, #600
14966803
                                                                                 Now One Ring Networks
                                                                                                                 San Diego,
            CA 92126
                                 300 Cadman Plaza West, Suite 1303
            SMARTSIGN
14966804
                                                                        Brooklyn, NY 11201
            SMILEMAKERS, INC. 425 Sha Lane Spartanburg, SC 29307
14966806
            SMITH, CHARLESETTA
14966807
                                            Address Redacted
            SO CAL MOBILE FLEET
                                            10051 Beech Ave.
14966813
                                                                     Fontana, CA 92335
            SO CAL TRUCK 10460 Mission Gorge Rd. Santee, CA 92 SOCAL AUTO BODY 28745 US Hwy 58 Barstow, CA 92 SOCALGAS P.O. BOX C Montery Park, CA 91756–5111
14966814
                                                                    Santee, CA 92071
14966815
                                                                 Barstow, CA 92311
14966817
            SOCIETY OF TEACHERS FAMILY MEDICINE 11400 Tomahawk Creek Parkway
14966818
                                                                                                          Leawood, KS
14966819
            SOFIA ENRIQUEZ
                                      Address Redacted
14966821
            SOFTCHOICE CORPORATION 16609 Collections Center Drive
                                                                                         Chicago, IL 60693
                                     MAGDELEN CENTRE, OXFORD SCIENCE
            SOLUTION 7 LTD
14966822
                         OXFORDSHIRE OXFORD, OX4 4GA United Kingdom
            SONESTA HOTEL PHILADELPHIA
14966824
                                                        1800 Market St Sonesta Philadelphia Rittenhouse
            Square Philadelphia, PA 19103
            SONIA B HUETE Address Redacted
SONIA M VILLAREAL Address Red
14966831
                                          Address Redacted
14966834
14966840
                           11 American Legion Pl Greenfield, IN 46140
            SONICU
            SOROPTIMIST OF BORREGO SPRINGS
                                                  NGS P.O. Box 504 Borrego Springs, CA 92004
4085 Tudor Centre Drive Anchorage, AK 99507
14966848
14966854
            SOUTHCENTRAL FOUNDATION
            SOUTHCOAST HEATING & AIR CONDITIONING P.O. Box 675055
14966855
                                                                                              Dallas, TX 75267-5055
                                           3306D Pepperell Pkwy Opelika, AL 36801
2702 Love Field Dr. Dallas, TX 75235
14966856
            SOUTHEASTERN FIXTURES
14966859
            SOUTHWEST AIRLINES
                                                        P.O. Box 98890 Las Vegas, NV 89193–8890
14966861
            SOUTHWEST GAS CORPORATION
                                                    P.O. Box 2006 Thousand Palms, CA 92276
14966862
            SOUTHWEST PLUMBING, INC.
14966863
            SOUTHWEST TRAILER SALES
                                                    2430 Main Street
                                                                          Ramona, CA 92065
14966866
            SPACE MAINTAINERS LABORATORIES
                                                            9129 Lurline Avenue Chastsworth, CA 91311
                                                                           Barstow, CA 92311
                                                1421 W. Main Street
14966867
            SPARKLE AUTO LAUNDRY
                                                                     400 Atlantic St Stamford, CT 06901
14966872
            SPECTRUM (TIME WARNER CABLE)PALM DR
                                                                    Pittsburgh, PA 15251
            SPECTRUM - 078920201 P.O. Box 223085
14966870
                                                                 Pittsburgh, PA 15251–2085
Pittsburgh, PA 15251–2085
            SPECTRUM - 096925101
                                             P.O. BOX 223085
14966871
14966873
            SPECTRUM 8245 10 061 4291025
                                                     400 Atlantic St
                                                                          RE 1971 UNIVERSITY AVE
                                                                                                              Stamford, CT
            06901
            SPECTRUM 8448 41 089 0476041
                                                                          DHS MAIN-PIERSON BLVD
14966875
                                                     400 Atlantic St
                                                                                                               Stamford, CT
14966877
            SPECTRUM-069495301 P.O. Box 223085 Pittsburgh, PA 15251
                                                                            1600 IOWA 210
            SPECTRUM-8245 10 061 4406979
                                                    P.O. Box 60229
14966878
                                                                                                  Los Angeles, CA
            90060-0229
                                                             P.O. Box 60229
            SPECTRUM-8245 10 062 1181144 (Troth)
14966879
                                                                                    Los Angeles, CA 90060-0229
            SPICE OF LIFE, INC. 1219 Linda Vista Drive San Marcos, CA 92078
SPORTS BOOSTERS, INC. 5095 Murphy Canyon Rd, Suite 120 San E
SPOTLIGHT 29 CASINO 46–200 Harrison Place Coachella CA 92234
14966880
            SPORTS BOOSTERS, INC. 5095 Murphy Canyon Rd, Suite 120 San Diego, CA 92123 SPOTLIGHT 29 CASINO 46–200 Harrison Place Coachella, CA 92236 SPRINGHILL SUITES BY MARRIST 2555 VENTURE OAKS WAY Sacramento, C. SPRINGT DO BOY 4181
14966881
14966882
14966883
                                                                                             Sacramento, CA 95833
            SPRINT P.O. Box 4181 Carol Stream, IL 60197–4181
SPY TECH INTERNATIONAL INC 4111 W Waters Ave
14966884
14966885
                                                                                  Tampa, FL 33614-8116
            SSD ALARM – 400932 1740 N Lemon Street Anaheim, CA 92801–1007
14966887
```

```
14966888
           SSD ALARM - 401455
                                     1740 N Lemon Street
                                                             Anaheim, CA 92801-1007
           SSD ALARM - 401645
                                     1740 N Lemon Street
                                                             Anaheim, CA 92801-1007
14966889
14966886
                          1740 N Lemon Street
                                                  Anaheim, CA 92801-1007
           SSD Alarm
           ST ELIZABETH OF HUNGARY CATHOLIC CHURCH
14966890
                                                                  66-700 Pierson Blvd
                                                                                          Desert Hot Springs, CA
           ST MARY FOUNDATION
                                                                 APPLE VALLEY, CA 92307
14966891
                                        18300 HIGHWAY 18
14966893
           ST. ELIZABETH OF HUNGARY CATHOLIC CHURCH
                                                                   P.O. Box 366
                                                                                    Julian, CA 92036-0366
14966899
           STAPLES
                                             Chicago, IL 60694-1217
                         P.O. Box 71217
14966902
           STARTECHTEL.COM
                                     206 N. Towne Avenue
                                                             Pomona, CA 91767
                                                             Dublin, OH 43016
14966905
                                     6095 Emerald Pkwy
           STAYBRIDGE SUITES
                                            P.O. BOX 604069
14966906
           STEALTH PARTNER GROUP
                                                                 Charlotte, NC 28260-4069
14966907
           STEELE PLUMBING, INC.
                                                           1304 Pepper Dr
                                         P.O. Box 2264
                                                                              El Cajon, CA 92021
14966934
           STEPHEN R OKANE
                                    Address Redacted
14966937
           STERICYCLE
                             P.O. Box 6578
                                                Carol Stream, IL 60197
           STERILIZATION ASSURANCE SERVICE
14966938
                                                       24876 Taylor St.
                                                                           Loma Linda, CA 92350
14966942
           STEVEN BENSON
                                 Address Redacted
14966951
                      11400 Tomahawk Creek Parway, Suite 240
           STFM
                                                                 Lakewood, KS 66211
14966953
           STOP LOSS INSURANCE SERVICES, INC.
                                                        940 Adams Street, Suite G
                                                                                    Benicia, CA 94510-2950
                                12955 Enterprise Way
                                                        Bridgeton, MO 63044
14966954
           STORE SUPPLY
14966955
           STORQUEST - CATHEDRAL CITY / DATE PALM
                                                               32500 Date Palm Dr
                                                                                      Cathedral City, CA
                                     83614 Dr. Carreon
                                                           Indio, CA 92201
14966956
           STORQUEST - INDIO
14966957
           STRATUS INDEMAND, INC.
                                           P.O. BOX 675248
                                                                 Detriot, MI 48267-5248
                         325 Sub Way
14966958
           SUBWAY
                                           Milford, CT 06461
14966968
           SUN BIRD RV
                             83–237 REQUA AVE.
                                                       INDIO, CA 92201
           SUN ROAD MARINA
14966969
                                    955 Harbor Island Drive
                                                               San Diego, CA 92101
14966970
           SUN-EEZ WINDOW TINTING
                                             77852 Wildcat Dr., Suite 1
                                                                         Palm Desert, CA 92211
14966971
           SUNLINE TRANSIT AGENCY
                                            32505 Harry Oliver Trail
                                                                        Thousand Palms, CA 92276
                                           53301 HIGHWAY 111
14966972
           SUNRIDGE SELF STORAGE
                                                                    Coachella, CA 92236
                                     4400 Earth City Expy
14966974
           SUPER BRIGHT LEDS
                                                             Earth City, MO 63045
14966975
           SUPER SHUTTLE
                                 4610 South 35th Street
                                                          Phoenix,, AZ 85040
14966976
           SUPER STORAGE
                                 148 Robelini Drive
                                                        Vista, CA 92083
14966977
           SUPPORT WAREHOUSE
                                                            Dallas, TX 75350-4598
                                       P.O. Box 204598
                                                     San Mateo, CA 94403
14966980
           SURVEY MONKEY
                                   1 Curiosity Wy
           SUSAN G. KOMEN – INLAND EMPIRE
14966986
                                                     Address Redacted
14966987
           SUSAN G. KOMEN SAN DIEGO
                                              Address Redacted
14966994
           SUSANA RAMIREZ
                                   Address Redacted
                       3970 ROGER B
14966998
           SUSPA
                                           Grand Rapids, MI 49548
14966999
           SUSTAINABLE SUPPLY
                                       8810 W 116th Cir Ste B
                                                                  Broomfield, CO 80021
                               5501 US-30 West
14967007
           SWEETWATER
                                                    Fort Wayne, IN 46818
                                                P.O. Box 1769
                                                                  Sparks, NV 89432
14967011
           SYNERGY QUEST CONSULTING
14967012
           SYSCO SAN DIEGO, INC
                                        12180 KIRKHAM ROAD
                                                                    POWAY, CA 92064
14966526
           Saad Fattouhi
                            Address Redacted
14966527
           Saara Kamel
                           Address Redacted
14966529
           Sabah Mansour
                              Address Redacted
14966531
           Sabrina Lucas
                             Address Redacted
14966536
           Safoura Massoumi Prof Dental Corp
                                               555 S Rancho Santa Fe Rd Ste 101
                                                                                   Childrens Primary & Adult
                           San Marcos, CA 92078
           Dental Care
           Sage Eckenroth
14966537
                              Address Redacted
14966538
           Sahira Zori
                          Address Redacted
14966539
           Saint Georges University
                                      University Centre
                                                           Grenada, West Indies
14966541
           Sal Hanna
                         Address Redacted
14966542
           Saleena Ek
                          Address Redacted
14966543
           Salina Huerta
                            Address Redacted
14966544
           Sally Batac
                          Address Redacted
14966545
           Sally Rosario
                            Address Redacted
14966546
           Sally Villanueva
                               Address Redacted
14966547
           Salomon Arredondo
                                  Address Redacted
14966548
           Salustio Yanez
                             Address Redacted
14966549
           Salvador Ceja
                             Address Redacted
14966550
           Salvador Garcia
                              Address Redacted
14966551
           Salvador Rangel-Garcia
                                     Address Redacted
14966552
           Salvador Rojas
                             Address Redacted
14966554
           Sam Webb
                          Address Redacted
14966555
           Samantha Prior
                              Address Redacted
14966556
           Samantha Rojas
                              Address Redacted
14966557
           Samantha Romero
                                Address Redacted
14966558
           Samantha Santillan
                                 Address Redacted
14966559
           Samara Yakeera
                              Address Redacted
           Samer Basaka
14966561
                             Address Redacted
14966562
           Samer Marroki
                             Address Redacted
14966564
           Sami Sleman
                            Address Redacted
14966565
           Samir Raban
                            Address Redacted
14966566
           Samira Khammi
                              Address Redacted
14966569
           Samuel Robinson
                                Address Redacted
                                     Address Redacted
14966570
           Samuel Sanchez-Castro
14966571
           Samuel Santana
                              Address Redacted
14966572
           Samuel Silva
                            Address Redacted
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 64 of 76

```
14966576
            San Bernardino City USD
                                           777 North F Street
                                                                  San Bernardino,, CA 92410
                                                          662 S. Tippecanoe Ave.
            San Bernardino County Preschool Services
14966578
                                                                                     San Bernardino, CA 92415
                                                                                   San Bernardino, CA 92401-1411
14966579
            San Bernardino District Office
                                               464 W. Fourth Street, Suite 239
                                                                            County of San Bernardino
14966580
            San Bernardino Project Connect
                                                385 N. Arrowhead Ave
                                                                                                           San Bernardino,
            CA 92415
            San Diego Health Connect
                                                                        San Diego, CA 92123
14966598
                                           5575 Ruffin RD Ste 240
14966599
            San Diego Health Connect
                                           5575 Ruffin Rd, Ste 240
                                                                        Health Information Exchange
                                                                                                          San Diego, CA
            92123
14966600
            San Diego Hunger Coalition
                                             845 15th Street, Suite 103
                                                                           San Diego, CA 92101
            San Diego Regional Healthcare Info Exchg 5575 Ruttin Ka S
San Diego Regional Healthcare Info Exchg 2045 S. San Jacinto Alexander
14966603
                                                          5575 Ruffin Rd Ste 225
                                                                                      San Diego, CA 92123–1381
                                                                                 San Jacinto, CA 92583
14966610
            San Marcos Medical Group
                                                                     Metropolitan Family Medical Clinic
14966611
                                            4990 Arlington Ave
                                                                                                             Riverside, CA
            92504-2757
            San Marcos Medical Group, Inc
                                                742 W Highland Ave
                                                                          San Bernardino, CA 92405
14966612
14966614
            San Ysidro Health Center
                                          1601 Precision Park Lane
                                                                        San Diego, CA 92173
14966616
            Sande Rubio
                              Address Redacted
14966618
            Sandra Alkad
                               Address Redacted
14966619
            Sandra Arellano
                                 Address Redacted
                                 Address Redacted
14966620
            Sandra Artinian
14966621
            Sandra Charette
                                 Address Redacted
            Sandra Coronel
14966622
                                 Address Redacted
14966623
            Sandra Cruz
                              Address Redacted
14966624
            Sandra Delgadillo
                                   Address Redacted
14966625
            Sandra Dominguez
                                    Address Redacted
14966626
            Sandra Franco
                                Address Redacted
14966627
            Sandra Gutierrez Valle
                                       Address Redacted
14966628
            Sandra Hansberger
                                    Address Redacted
14966629
            Sandra Hernandez
                                   Address Redacted
14966630
            Sandra Huerta
                                Address Redacted
            Sandra Hurtado
                                 Address Redacted
14966631
14966632
            Sandra Jerez
                              Address Redacted
14966633
            Sandra Junco
                               Address Redacted
14966634
            Sandra Mancilla
                                 Address Redacted
14966635
            Sandra Mora
                               Address Redacted
14966636
            Sandra Muro
                              Address Redacted
14966637
            Sandra Nava Garrido
                                      Address Redacted
14966638
            Sandra Pealing
                                Address Redacted
14966639
            Sandra Ramirez
                                 Address Redacted
14966640
            Sandra Rodriguez
                                   Address Redacted
14966641
            Sandra Ruiz
                             Address Redacted
14966642
            Sandra Sepulveda
                                   Address Redacted
            Sandra Singh
                               Address Redacted
14966643
14966644
            Sandra Vasquez
                                 Address Redacted
            Sandra Veliz
                              Address Redacted
14966645
14966646
            Sandra Yakeera
                                 Address Redacted
            Sandy Adwer
Sandy Matti
14966647
                               Address Redacted
14966648
                              Address Redacted
            Santa Barbara Business College
                                                                     Santa Barbara, CA 93101
14966650
                                                506 Chapala St
14966651
            Santana Cruz
                               Address Redacted
                                          401 West A Street, Suite 1900
                                                                            Wilson Elser-Robert W. Harrison
14966652
            Santiago A. Rojo, D.D.S.
                                                                                                                  San
            Diego, CA 92101
14966653
            Santiago A. Rojo, D.D.S., Inc.
                                              Address Redacted
            Santiago Rojo, DDS
                                     401 West A Street, Suite 1900
14966654
                                                                        Wilson Elser-Robert W. Harrison
                                                                                                              San Diego,
            CA 92101
14966657
            Sara Gerardo
                               Address Redacted
14966658
            Sara Nazco
                             Address Redacted
                            Address Redacted
14966659
            Sara Saco
14966660
            Sara Valdez
                             Address Redacted
14966661
                                 Address Redacted
            Sarah Damman
14966662
            Sarah Dushane
                                Address Redacted
14966663
            Sarah Rogers
                               Address Redacted
14966664
            Sarahi Olvera
                               Address Redacted
14966665
            Saraya Flood
                               Address Redacted
14966666
            Sarmad Kido
                               Address Redacted
14966667
            Sarmed Aziz
                              Address Redacted
                               Address Redacted
14966668
            Sarra Johnson
14966669
            Saul Contreras
                                Address Redacted
            Saul Escamilla
                                Address Redacted
14966670
                               Address Redacted
14966671
            Saul Hinojosa
14966672
            Saul Munguia
                               Address Redacted
14966673
            Savannah Barrientos
                                     Address Redacted
14966674
            Savina Pifirro
                               Address Redacted
14966676
            Sayra Gonzalez Martinez
                                          Address Redacted
                                 Address Redacted
14966677
            Sayuri Barragan
14966680
            School Aged-Gap Elimination Initiative
                                                        1502 SOUTH LA BRUCHERIE RD
                                                                                                EL CENTRO, CA
            92243
14966683
            Schwartz, Stanley H. M.D.
                                           Address Redacted
14966684
                             Address Redacted
            Scott Allen
```

```
14966685
            Scott Blaisdell
                               Address Redacted
                              Address Redacted
14966686
            Scott Larned
14966687
            Scott Turchi
                             Address Redacted
                             50 West Broadway, Suite 1100
14966689
            Scrip World
                                                                Salt Lake City, UT 84101
14966690
            Scripps
                        312 Walnut St, Suite 2800
                                                       Cincinati, OH 45202
14966694
            Sean Barlow
                              Address Redacted
14966698
            Secretary of the US DHHS
                                          200 Independence Ave SW
                                                                         Washington, DC 20201
14966699
            SecurCare Self Storage – San Bernardino
                                                        981 West Mill Street
                                                                                 San Bernardino, CA 92410
14966701
            Seiry Moreno
                              Address Redacted
            Selene Mora
14966703
                             Address Redacted
14966704
            Selene Simon
                              Address Redacted
                              Address Redacted
14966705
            Selma Pulido
14966707
            Sequoia Cipes
                               Address Redacted
14966709
            Sergio Barajas
                               Address Redacted
14966710
            Sergio Bautista Desales
                                       Address Redacted
14966711
            Sergio Espinoza
                                 Address Redacted
14966712
            Sergio Fuentes
                               Address Redacted
14966713
            Sergio Navarro
                                Address Redacted
14966714
            Sergio Ruiz
                             Address Redacted
14966715
            Sergio Soqui
                              Address Redacted
14966717
            Servando Servin
                                 Address Redacted
14966720
            Setareh Jones
                               133 West Main Street, Suite 100
                                                                  El Cajon, CA 92020
14966721
            Setareh Jones
                              Address Redacted
14966722
            Seth Gustafson
                                Address Redacted
14966725
            Seyedsorouh Karimi
                                     Address Redacted
14966727
            Shadha Rammo
                                Address Redacted
14966728
            Shakila Mahmodi
                                  Address Redacted
14966729
            Shamrock Consulting
                                      222 North Pacific Coast Hwy, Ste 1620
                                                                                El Segundo, CA 90245
14966731
            Shams Sherzai
                               Address Redacted
14966732
            Shana Rice
                            Address Redacted
                                   Address Redacted
14966733
            Shannon Dunaway
14966735
            Shannon Murillo
                                  Address Redacted
14966738
            Shannon T Taylor
                                  Address Redacted
            Shannon Younglove
14966739
                                    Address Redacted
            Sharareh Gandy
                                 Address Redacted
14966740
14966743
            Sharlyn Ziprick
                                Address Redacted
14966744
            Sharon Smiar
                              Address Redacted
            Sharp Grossmont Hospital
                                                                           La Mesa, CA 91942
14966745
                                          5555 Grossmont Center Drive
14966748
            Sharp Rees-Stealy Medical Group
                                                 P.O. Box 939089
                                                                        San Diego, CA 92193-9089
14966749
            Shaun DeWitt
                               Address Redacted
14966750
            Shauna Jacobs
                               Address Redacted
            Shawn Carter
                              Address Redacted
14966751
14979359
            Shawn M. Christianson, Esq.
                                            Buchalter, A Professional Corporation
                                                                                     425 Market Street, Suite
                      San Francisco, Ĉalifornia 94105-3493
            2900
14966752
            Shawn Viramontes
                                   Address Redacted
            Shawte Clary
14966753
                              Address Redacted
14966754
            Shaymaa Al Omairi
                                    Address Redacted
14966755
            Sheetal Luthra
                               Address Redacted
            Shefner Moore
14966756
                                Address Redacted
            Sheila Dilley
14966757
                              Address Redacted
            Sherazi Momin
14966760
                                Address Redacted
14966761
            Sherri Langhorne
                                  Address Redacted
14966763
            Sheryl Drew
                             Address Redacted
14966764
            Sheyla Chea
                             Address Redacted
14966767
                                Address Redacted
            Shivani Sikand
14966768
            Shonique Williams
                                   Address Redacted
            Shop Pop Displays
14966770
                                   222 Browertown Rd
                                                            Little Falls, NJ 07424
14966776
            Siana L Thieme
                                Address Redacted
14966777
            Siba Kakos
                            Address Redacted
14966779
            Sidney Dunstan
                                Address Redacted
14966780
            Sierra Beltran
                               Address Redacted
            Sierra Medical Imaging
14966781
                                       231 W Fir Ave.
                                                            Clovis, CA 93611
            Signs Done Fast Inc
14966785
                                    3205 Moore St
                                                        San Diego, CA 92110
14966786
            Signup Training
                                 1605 Gilmore Park Ave, Apt202
                                                                    Columbus, OH 43204
14966787
            Silas Gyimah
                              Address Redacted
                             Address Redacted
14966788
            Silsti Tomas
            Silver Valley Unified School District
14966789
                                                    35320 Daggett-Yermo Rd
                                                                                  Yermo, CA 92398
                                 Address Redacted
14966790
            Silvia Arambula
            Silvia Calderon
14966791
                                Address Redacted
14966792
            Silvia Gonzalez
                                Address Redacted
14966793
            Silvia R Luviano
                                 Address Redacted
                                    6900 E Camelback Road Suite 700
14966794
            SimonMed Imaging
                                                                          Scottsdale, AZ 85251
            Sindy Ayon
Sissy Shoemaker
14966797
                             Address Redacted
14966798
                                  Address Redacted
14966802
            Skinny Gene Project
                                     10620 Treena St. Suite 230
                                                                    San Diego, CA 92131
14969355
                                                   501 W. Broadway, Suite 1360
            Small Law PC
                               Kelly Ann Tran
                                                                                     San Diego, CA 92101
14966805
            Smile Factory
                               41865 Boardwalk, Ste 206
                                                             Palm Desert, CA 92211
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 66 of 76

```
14969921
           Snell & Wilmer L.L.P.
                                       Andrew B. Still
                                                           600 Anton Blvd., Suite 1400
                                                                                           Costa Mesa, CA
            92626-7689
14969920
           Snell & Wilmer L.L.P.
                                      Michael B. Reynolds
                                                                600 Anton Blvd., Suite 1400
                                                                                                Costa Mesa, CA
            92626-7689
14966808
            So Cal Auto Body
                                  28745 US Hwy 58
                                                         Barstow, CA 92311
           So Cal Edison
14966809
                                                           2-39-323-5767,700471712862
                                                                                              Rosemead, CA
                               2244 Walnut Grove Ave
            91770
           So Cal Edison
14966810
                               2244 Walnut Grove Ave
                                                           2-40-717-9142,700443804750
                                                                                              Rosemead, CA
            91770
14966811
           So Cal Edison
                               P.O. Box 300
                                                 Rosemead, CA 91772
14966812
            So Cal Edison
                               P.O. Box 600
                                                 2-37-539-5720,700524094983
                                                                                     Rosemead, CA 91771
14966816
           SoCalGas
                           555 West Fifth Street
                                                    Los Angeles, CA 90013
14966820
            Sofia Mendoza
                               Address Redacted
14966823
            Sonal Briones
                              Address Redacted
14966825
            Sonia Aguiniga
                                Address Redacted
14966826
            Sonia Aguirre
                              Address Redacted
14966827
            Sonia Aldena–Lemus
                                     Address Redacted
14966828
            Sonia Anguiniga
                                 Address Redacted
           Sonia Avina
14966829
                             Address Redacted
14966830
            Sonia Avala
                             Address Redacted
14966832
            Sonia Brito
                            Address Redacted
            Sonia L Rangel
                                Address Redacted
14966833
14966835
            Sonia Maldonado
                                  Address Redacted
14966836
            Sonia Martinez
                                Address Redacted
14966837
            Sonia Odish
                             Address Redacted
14966838
            Sonia Rivera
                             Address Redacted
14966839
           Sonia Salinas
                              Address Redacted
                           2001 Logic Drive
14966841
            SonicWall
                                                 San Jose, CA 95124
14966842
           Sophia Castruita
                                 Address Redacted
14966843
            Sophia Jackson
                                Address Redacted
            Sophia Rodriguez
14966844
                                  Address Redacted
14966845
            Sophia Ruiz-Gonzalez
                                      Address Redacted
14966846
           Sophie Lee
                            Address Redacted
14966847
            Soren Kivork
                              Address Redacted
14966849
            Sotero Ruiz
                            Address Redacted
14966850
            Souad Awad
                             Address Redacted
            Soufyan G Alogaidi
14966851
                                    Address Redacted
14966852
            Sound Physicians
                                  1498 Pacific Avenue, Suite 400
                                                                    Tacoma, WA 98402
14966853
            Sourcewell
                            P.O. Box 219
                                              202 12th Street NE
                                                                      Staples, MN 56479
14966857
                                                       2244 Walnut Grove Avenue
            Southern CA Edison
                                    P.O. Box 800
                                                                                      Rosemead, CA 91770
14966858
           Southern California Real Estate Services
                                                      15901 Red Hill Ave, Suite 205
                                                                                         Tustin, CA 92780
                                      P.O. Box 98510
14966860
           Southwest Corporation
                                                           8360 South Durango Drive
                                                                                          Las Vegas, NV
            89193-8510
14966864
           Sovereign Receivables
                                      P.O. Box 665
                                                         Aldan, PA 19018
14966865
            Sowmya Chinta
                                Address Redacted
            Spectrum
14966868
                          400 Atlantic Street
                                                  Stamford, CT 06901
14966869
            Spectrum
                          P.O. Box 60074
                                               City of Industry, CA 91716–0074
14966874
           Spectrum 8245 10 061 4349559
                                               P.O. Box 60229
                                                                   RE 1970 University
                                                                                           Los Angeles, CA
           90060
           Spectrum Enterprise
                                                        Stamford, CT 06901
14966876
                                    400 Atlantic St
14966892
            St. Bernardine Ĥospital
                                       2101 N Waterman Ave
                                                                  San Bernardino, CA 92404
14966894
            St. Mary Medical Center Foundation
                                                   18300 Kasota Road
                                                                           Apple Valley, CA 92307
            Stacey Alcala
14966895
                              Address Redacted
           Stacey Avalos
Stacey Lemus
14966896
                               Address Redacted
14966897
                              Address Redacted
14966898
            Stacy Silva
                            Address Redacted
            Starr Indemnity & Liability Company
                                                    399 PARK AVE. 8TH FLOOR
14966900
                                                                                      New York, NY 10022
14966901
            Starr Indemnity & Liability Company
                                                    399 Park Avenue, Suite 2000
                                                                                    New York, NY 10022
14966903
                                                1615 Murray Canyon Road, Suite 700
                                                                                         Division of MediCal Fraud &
           State of CA Dept of Justice, OAG
            Elder Abuse
                             San Diego, CA 92108
14966904
            State of California
                                  P.O. Box 85501
                                                       San Diego, CA 92186-5326
14966908
            Stephan Family Dental
                                      860 Jamacha Road, Suite 201
                                                                       EL Cajon, CA 92019
14966909
           Stephanie Alfaro
                                 Address Redacted
14966910
            Stephanie Arzamendi
                                     Address Redacted
14966911
            Stephanie Cebreros-Ocampo
                                            Address Redacted
14966912
            Stephanie Chen
                                Address Redacted
14966913
            Stephanie Delgado Vazquez
                                           Address Redacted
14966914
            Stephanie Fernandez
                                     Address Redacted
14966915
            Stephanie Garcia
                                 Address Redacted
14966916
            Stephanie Hernandez
                                     Address Redacted
14966917
            Stephanie Heydt
                                 Address Redacted
14966918
            Stephanie Lainez
                                 Address Redacted
14966919
            Stephanie Lopez
                                 Address Redacted
            Stephanie Manzo
                                 Address Redacted
14966920
14966921
            Stephanie Martinez
                                   Address Redacted
            Stephanie Mix
                               Address Redacted
14966922
14966923
            Stephanie Munoz
                                  Address Redacted
14966924
                                   Address Redacted
           Stephanie Ramirez
```

```
14966925
           Stephanie Rivas
                               Address Redacted
14966926
           Stephanie Sakuma
                                  Address Redacted
14966927
           Stephanie Smith
                                Address Redacted
14966928
           Stephanie Turincio
                                  Address Redacted
14966929
           Stephanie Velez
                                Address Redacted
14966930
           Stephany Delafuente
                                    Address Redacted
14966931
            Stephany Delafuente
                                    Address Redacted
           Stephen Hawkins
14966932
                                 Address Redacted
14966933
            Stephen Hooton
                                Address Redacted
14966935
           Stephen Shewey
                                Address Redacted
14966936
            Stephen Stevens
                                Address Redacted
14966939
           Steve Ha
                         Address Redacted
14966940
           Steve Perez
                            Address Redacted
14966941
           Steven Benson
                              Address Redacted
14966943
                                       Address Redacted
           Steven Castro Gastelum
14966944
           Steven Crose
                             Address Redacted
14966945
           Steven Gunther
                               Address Redacted
14966946
           Steven Martinez
                                Address Redacted
14966947
           Steven Osuna
                             Address Redacted
14966948
           Steven R Bowers
                                 Address Redacted
14966949
           Steven Valenzuela
                                  Address Redacted
14966950
           Stewart Wilkey
                               Address Redacted
14966952
           Stonewall Pharmacy
                                   68555 Ramon Road, Suite D 105
                                                                       Cathedral City, CA 92234
14966959
            Suhaila Arif
                            Address Redacted
14966960
           Suheir Kasabri
                              Address Redacted
14966961
           SuiteCentric Test Vendor
                                        5857 Owens Ave Suite 300
                                                                      Carlsbad, CA 92008
14966962
           SuiteCentric, LLC
                                  5857 Owens Avenue, Suite 300
                                                                    Carlsbad, CA 92008
14966963
           Sukeytra Thornton
                                  Address Redacted
           Sukhhwinder Singh
14966964
                                   Address Redacted
                                              124 W Main Street, Ste 240
                                                                             c/o Daryl R Priest
14966965
           Summit Healthcare Management
                                                                                                   El Cajon, CA
            92020
14966966
           Summit Healthcare Management
                                              35 Braintree Hill Park, Suite 303
                                                                                 Braintree, MA 02184
14966967
           Summit Healthcare Management, Inc.
                                                   101 Broadway, Suite 3A
                                                                               The Maul Firm PC-Anthony F.
                      Oakland, CA 94607
           Maul
                               37774 Montezuma Vallev
14966973
           Sunshine Stone
                                                            Ranchita, CA 92066
14966978
                              Address Redacted
           Sura Alqazzaz
14966979
           Surescripts
                           2550 South Clark Street
                                                      Arlington, VA 22202
14966981
           Susan Austin
                             Address Redacted
14966982
            Susan Barba Ruiz
                                 Address Redacted
14966983
           Susan Ellis
                           Address Redacted
14966984
           Susan Fletcher
                              Address Redacted
14966985
           Susan Fosseen
                              Address Redacted
14966988
           Susan Kulimakin
                                 Address Redacted
                            Address Redacted
14966989
           Susan Owen
14966990
           Susan Wagner
                              Address Redacted
14966991
           Susan Zunich
                             Address Redacted
14966992
           Susana Dominguez
                                   Address Redacted
14966993
           Susana Jimenez
                               Address Redacted
14966995
           Susana Sanchez
                               Address Redacted
14966996
           Susanna Dominguez
                                    Address Redacted
14966997
                            Address Redacted
            Susie Miller
14967000
           Suzan Moshi
                             Address Redacted
14967001
           Suzanne Armour
                                Address Redacted
14967002
           Suzanne Dixon
                               Address Redacted
14967003
           Suzanne Johnson
                                Address Redacted
14967004
           Suzanne Moore
                               Address Redacted
14967005
           Suzette Martinez
                                Address Redacted
14967006
           Swarat Signs
                             8877 Ramona Trail
                                                    Morongo Valley, CA 92256
14967008
           Sydney Willard
                               Address Redacted
14967009
           Sylvia Carrillo
                              Address Redacted
14967010
           Sylvia Searleman
                                 Address Redacted
                          12920 SE 38th Street
14967154
           T-Mobile
                                                   Bellevue, WA 98006-1350
14967155
           T-Mobile
                          P.O. Box 790047
                                                St. Louis, MO 63179-0047
14967014
           TABLEAU SOFTWARE, INC
                                              P.O. Box 204021
                                                                   Dallas, TX 75320-4021
14967017
           TALENT ADVISORY GROUP
                                              24450 W 207th Street
                                                                       Spring Hill, KS 66083
                                       1845 Borman Court
14967018
           TALX CORPORATION
                                                               St Louis, MO 63146
14967024
           TAMERA DEBOE
                                  9372 HENY CIRCLE
                                                            SANTEE, CA 92071
                                            713 BROADWAY STE I
           TAQUIZAS LOS CHUCHYS
                                                                         CHULA VISTA, CA 91910
14967034
14967037
           TARGET
                          1000 Nicollet Mall
                                                 Minneapolis, MN 55403
14967038
           TARIK ALJUMAILI
                                    Address Redacted
           TAX1099.COM(Under Zenwork Inc))
14967045
                                                   1 E Center St Ste 250
                                                                            Fayetteville, AR 72701
14967047
                         990 Highland Drive, Suite 312
                                                          Solana Beach, CA 92075
           TCOYD
           TEAM SPORTS
14967048
                                82-227 Hwy 111, Suite D-8
                                                               Indio, CA 92201
           TECHNOLOGY INTEGRATION GROUP
                                                        10240 Flanders Court
14967050
                                                                                 San Diego, CA 92004
                                                         Csan Francisco, CA 94107
14967051
           TECHSOUP.ORG
                                  455 Brannan Street
                                                                   Chicago, IL 60693-0324
14967055
           TELETRAC, INC
                                 32472 Collections Center Drive
14967056
           TEMECULA TROPHY & DESIGN
                                                 27314 Jefferson Ave, Suite 5
                                                                                 Temecula, CA 92590
14967058
           TEMPERATURE GUARD
                                         P.O. Box 1154
                                                             Farmington, CT 06034
```

```
14967059
          TEMPTIME
                          P.O. Box 28744
                                             New York, NY 10087-8744
          TENET HEALTH SYSTEM DESERT, INC.
                                                     P.O. Box 845610
14967060
                                                                         Dallas, TX 75284-5610
                                                    Traverse City, MI 49684
14967062
          TENTCRAFT, INC.
                                 2662 Cass Road
          TERESA MARTINEZ
                                   Address Redacted
14967067
                                   9 TECH CIRCLE-SUITE 103
14967083
          THD AMERICA, INC
                                                                 NATICK, MA 01760
          THE AMERICAN COLLEGE WASHINGTON DC
14967084
                                                           25 Massachusetts Avenue, NW Suite
                   Washington, DC 20001-7401
          THE CATHEDRAL CITY SENIOR CENTER
14967086
                                                       37–171 West Buddy Rogers Avenue
                                                                                           Cathedral City, CA
           92234
14967088
          THE CITY OF SAN DIEGO
                                        P.O. Box 129030
                                                            San Diego, CA 92112
          THE COMPLIANCE CENTER
14967089
                                           2150 Liberty Dr.
                                                              Niagara Falls, NY 14304
14967091
          THE DESERT SUN
                                 750 N. Gene Autry Trail
                                                           Palm Springs, CA 92262
                                         4691 Northwest Pkwy
14967092
          THE ECOPLUMBERS, INC.
                                                                 Hilliard, OH 43026
14967093
          THE FIRST TEE OF SILICON VALLEY
                                                   1922 The Alameda, Suite 214
                                                                                 San Jose, CA 95126
          THE FISH GUY
14967094
                              370 Mankato St, Apt #A
                                                        Chula Vista, CA 91910
14967095
          THE FISHER CONSULTING GROUP
                                                105 NW Railroad Avenue #1314
                                                                                  Hammond, LA 70404
          THE HEBETS FAMILY TRUST
14967100
                                           P.O. BOX 2104
                                                              Borrego Springs, CA 92004-2104
14967101
          THE HOME DEPOT
                                  2455 Paces Ferry Rd SE
                                                            Atlanta, ĞA 30339
          THE INLINE GROUP, LLC
14967102
                                        1826 Lakeway Drive
                                                               Lewisville, TX 75057
14967103
          THE INSPECTORS COMPANY
                                            8637 Troy St
                                                            Springs Valley, CA 91977
14967104
          THE JULIAN NEWS
                                  P.O. Box 639
                                                   Julian, CA 92036
                                                                               San Diego, CA 92122
14967105
          THE LARABEE LAW FIRM
                                         8910 University Center Lane, Suite 400
14967106
          THE MEDICAL PROTECTIVE COMPANY
                                                      5814 Reed Rd
                                                                       Fort Wayne, IN 46835
14967108
           THE MISSION INN
                                 3649 Mission Inn Avenue
                                                            Riverside, CA 92501-3364
          THE MOLINA FOUNDATION
14967109
                                          P.O. Box 2677
                                                            Los Alamitos, CA 90720
          THE OFFICE EXPRESS
                                     3198-c Airport Loop
14967110
                                                            Costa Mesa, CA 92626
                                        3512 14TH Street
14967111
          THE PRESS-ENTERPRISE
                                                            Riverside, CA 92501
                                            P.O. Box 842349
14967113
          THE REINALT-THOMAS Corp
                                                                Discount Tire/Americas Tire
                                                                                             Los Angeles, CA
           90084-2349
14967114
          THE ROADRUNNER CLUB L.P.
                                             130 Garden St.
                                                               Santa Barbara, CA 93101
14967115
          THE SD FLOWER SHOP
                                      5101 WARING RD
                                                            SAN DIEGO, CA 92120
                                        2650 Virginia Ave NW
14967116
          THE WATERGATE HOTEL
                                                                  Washington, DC 20037
          THE WESTIN BONAVENTURE HOTEL AND SUITES
                                                                7750 Wisconsin Ave.
14967117
                                                                                       Bethesda, MD
           20814
                                           7750 Wisconsin Ave.
14967118
           THE WESTIN KANSAS CITY
                                                                  Bethesda, MD 20814
          THE WILLARD INTERCONTINENTAL
14967119
                                                   1401 Pennsylvania Avenue Northwest
                                                                                        Washington, DC
14967122
          THERACOM
                           P.O.Box 640105
                                               Cincinnati, OH 45264–0105
          THERMAL TRUCK AND RV, INC
14967128
                                               57100 Desert Cactus
                                                                     Thermal, CA 92274
14967129
                                                                           Noosaville, Qld 4566 Australia
          THINLINX
                          28 Eenie Creek Road
                                                 Noosa Civic Commercial
14967130
          THIRD PARTY FILING SERVICES, LLC
                                                    2532 N 4th St #445
                                                                          Flagstaff, AZ 86004
14967132
          THOMAS AND COMPANY
                                         1 Vantage Way # A105
                                                                 Nashville, TN 37228
14967136
          THOMPSON ENGINEERING
                                          2205 Fleetwood Dr.
                                                                Riverside, CA 92509
14967137
          THOMSON REUTERS – WEST PAYMENT CENTER
                                                              P.O. Box 6292
                                                                                Carol Stream, IL
           60197-6292
                                                      Jordan, MN 55352
14967138
          THREAD LOGIC
                               16775 Greystone Ln
          THREAT BLASTER MEDIA
14967139
                                         550M Ritchie Hwy, #300
                                                                   Severna Park, MD 21146
14967144
          TIG
                   10240 Flanders Court
                                           San Diego, CA 92121
14967149
          TIMS MOBILE TRUCK REPAIR, INC.
                                                  2277 La Crosse Ave #302
                                                                             Colton,, CA 92324
                             13849 Armargosa Road #102
3841 W FLORIDA AVE
45334 GOLF CENTER PKWY
14967150
                                                            Victorville, CA, CA 92392
          TINT DOCTOR
14967152
          TIRESLES SCHWAB
                                                             Hemet, CA 92545
14967153
          TKB Bakery & Deli
                                                                INDIO, CA 92201
                                    44503 JACKSON ST
14967168
          TOPS N BARRICADES
                                                            Indio, CA 92201
14967170
          TOWING SOLUTIONS
                                    22180 CAJALCO ROAD
                                                               PERRIS, CA 92570
14967171
                                                               Fort Worth, TX 76109
          TOWNEPLACE SUITES
                                     4200 International Plaza
14967172
          TOWNSEND THUASNE USA
                                          4615 SHEPARD ST
                                                                 BAKERFIELD, CA 93313
14967173
                     3810 Pheasant Ridge Drive NE
                                                     Minneapolis, MN 55449
          TPAC
          TPX Communications – 279814 –EC
TPX Communications – 404485 ESCND
14967174
                                               P.O. Box 60767
                                                                  Los Angeles, CA 90060-0767
14967175
                                                  P.O. Box 60767
                                                                     Los Angeles, CA 90060-0767
14967177
          TPX Communications -395654.ESC
                                              P.O. Box 60767
                                                                 Los Angeles, CA 90060-0767
14967178
          TPX Communications-405947 CC
                                              P.O. Box 60767
                                                                Los Angeles, CA 90060-0767
                                                            San Diego, CA 92150-9013
14967176
          TPx Communications – Iowa
                                        P.O. Box 509013
14967182
          TRAINING LEADER
                                 2277 TRADE CENTER WAY, SUITE 101
                                                                            NAPLES, FL 34109
          TRANSACTION PRO.COM
                                        14 Hampshire Dr
14967183
                                                           Hudson, NH 03051
          TRANSGENDER COMMUNITY COALITION
                                                        P.O. Box 580132
14967185
                                                                            North Palm Springs, CA 92240
                                              P.O. BOX 505525
                                                                   St. Louis, MO 63150-5525
          TRI STATE DISTRIBUTION, INC
14967193
          TRI STATE DISTRIBUTION, INC
                                                                  St. Louis, MO 63150-5525
14967194
                                              P.O. Box 505525
          TRICARE-HEALTH NET FEDERAL SVCS-TRICARE
                                                                P.O. Box 9410
14967197
                                                                                  Virginia Beach, VA
           23450-9410
14967199
          TRIWEST
                         16010 North 28th Avenue
                                                    Phoenix, AZ 85053
                                                    Rancho Cucamonga, CA 91730
14967201
                               9531 Milliken Ave.
          TRL Systems, Inc
14967202
          TRONCOSO, PRUDENCIO
                                        Address Redacted
          TRUEVOLUTION
                                4164 Brockton Avenue, Suite A
14967206
                                                                Riverside, CA 92501
                                                  San Francisco, CA 94111
14967208
          TRUEWORK
                            325 Pacific Avenue
                                           P.O. Box 12310
14967209
          TRULY NOLEN BRANCH 063
                                                             Palm Desert, CA 92255
                                                                     Perris, CA 92571-9374
14967210
          TRULY NOLEN BRANCH 064
                                           1622 Illinois Ave, Ste 11
                                           1622 Illinois Ave Ste 11
14967211
          TRULY NOLEN OF AMERICA
                                                                     Perris, CA 92571
```

```
14967214
           TRULY NOLEN OF AMERICA, INC-064-64017708
                                                                   1622 ILLINOIS AVE STE 11
                                                                                                    PERRIS, CA
           92571-9374
14967215
           TRULY NOLEN OF AMERICA, INC-064-64019198
                                                                   1622 ILLINOIS AVE, STE 11
                                                                                                    PERRIS, CA
           92571
14967212
           TRULY NOLEN OF AMERICA, INC.
                                                                         Palm Desert, CA 92255-2310
14967213
           TRULY NOLEN OF AMERICA, INC. - 64019607
                                                                1622 Illinois Ave Ste 11
                                                                                            Perris, CA 92571
14967216
           TTY, LLC
                           2200 N Central Ave, Suite 400
                                                             Phoenix, AZ 85004
14967219
                                                   Ste 300
           TWILIO, INC
                               375 Beale Street
                                                                San Francisco, CA 94105
14967220
           TYCO INTERGRATED SECURITY LLC
                                                        P.O. Box 371967
                                                                             Pittsburgh, PA 15250-7967
14967221
           TYLER S ANDREWS
                                      Address Redacted
14967013
           Tabitha Oldham
                                Address Redacted
14967015
           Tai Hoag
                          Address Redacted
14967016
           Talen Georges
                              Address Redacted
15002693
           Talent Advisory Group
                                      24450 W 207th Street
                                                                Spring Hill, KS 66083
14967019
           Tamara Escobar Prado
                                      Address Redacted
14967020
           Tamara George
                               Address Redacted
14967021
           Tamara Torres Escobed
                                       Address Redacted
14967022
           Tamara Tucker-Ham
                                     Address Redacted
14967023
           Tamatha Bailey
                                Address Redacted
14967025
           Tami Bereki
                             Address Redacted
14967026
           Tangela Derogene
                                  Address Redacted
           Tania Chang
                             Address Redacted
14967027
14967028
           Tania Cuevas
                              Address Redacted
14967029
            Tania Juarez-Rivera
                                    Address Redacted
14967030
           Tannia Gonzalez
                                 Address Redacted
14984498
           Tanya Behnam
                               POLSINELLI LLP
                                                      2049 Century Park East, Suite 2900
                                                                                            Los Angeles, CA
           90067
14967031
           Tanya Olmos
                              Address Redacted
14967032
           Tanya Patros
                             Address Redacted
14967033
           Tanya Sanchez
                               Address Redacted
14967035
           Tara Al Isso
                            Address Redacted
14967036
                              Address Redacted
           Tara Hoffman
           Tarquin Williams
14967039
                                 Address Redacted
14967040
                             Address Redacted
           Tasha Smith
14967041
           Tatiana Enriquez
                                 Address Redacted
14967042
           Tatiana Garcia
                               Address Redacted
14967043
           Tatiana Gonzalez
                                 Address Redacted
14967044
           Tax Agency CA
                                P.O. Box 942857
                                                      Sacramento, CA 94257-0500
14967046
           Taylor Anderson
                                 Address Redacted
14967049
                               Address Redacted
           Tecarra Turner
14967052
           Tegile
                       5601 Great Oaks Pkwy
                                                  San Jose, CA 95119
           Tejas Modi
14967053
                            28039 Scott Road Suite 1
                                                        Heritage Plaza Dental-Tejas Modi
                                                                                             Murrieta, CA 92563
14967054
           Teletrac Subsciber Agreement
                                             310 Commerce, Suite 100
                                                                          Irvine, CA 92602
14967057
           Temecula Valley Hospital
                                                                      Temecula, CA 92592
                                         31700 Temecula Parkway
14967061
           Tennessee Department of Revenue
                                                500 Deaderick Street
                                                                         Nashville, TN 37242
14967063
           Teresa Arias
                             Address Redacted
14967064
           Teresa Bolanos
                               Address Redacted
14967065
           Teresa Gonzalez
                                Address Redacted
14967066
           Teresa League
                               Address Redacted
                                 Address Redacted
14967068
           Teresa Rodriguez
14967069
           Teresa Rodriguez
                                 Address Redacted
14967070
           Teresa Sanchez
                               Address Redacted
                              Address Redacted
14967071
           Teresa Tobey
14967072
           Teresa Torres
                              Address Redacted
14967073
           Teresa West
                             Address Redacted
14967074
           Teresita Carrera Diaz
                                     Address Redacted
                                 Address Redacted
14967075
           Teresita Morales
14967076
           Teri Riggle
                            Address Redacted
14967077
                                Address Redacted
           Terressa Powell
14967078
           Terri L Ohly
                             Address Redacted
14967079
           Terri Steidl
                            Address Redacted
14967080
           Texas Comptroller of Public Accounts
                                                    111 East 17th Street
                                                                            Austin, TX 78774
14967081
           Thaer Yalda
                            Address Redacted
14967082
           Thaghar Oda
                             Address Redacted
14967085
           The Bradley Group, LLC
                                        2857 Darius Way
                                                              San Leandro, CA 94577
           The City of San Diego
                                                           Business Tax Program
14967087
                                      P.O. Box 122289
                                                                                     San Diego, CA 92112-2289
           The Countertop Shop, Inc.
14967090
                                         1864 Commercial St.
                                                                  Escondido, CA 92029
                                  590 Palm Canyon Drive
           The Frugal Coyote
14967096
                                                              Borrego Springs, CA 92004
           The Greater Coachella Valley
14967097
                                            Attn Chamber of Commerce
                                                                           82-921 INDIO BLVD
                                                                                                      INDIO, CA
           92201
14967098
           The Greek Palace
                                 8878 Clairemont Mesa Blvd.
                                                                 San Diego, CA 92123
14967099
           The Hebets Company
                                     2575 East Camelback Road, Suite 700
                                                                             Phoenix, AZ 85016
           The Miss Julian Scholarship Pageant
14967107
                                                  2129 Main Street
                                                                        Julian, CA 92036
14967112
           The Receivable Management Services LLC
                                                        P.O. Box 19646
                                                                             Minneapolis, MN 55419
14967120
           Theodore Guice
                                Address Redacted
                                  Address Redacted
14967121
           Theodore Marbley
14967123
           Theresa Corona
                                Address Redacted
14967124
           Theresa Forrest
                               Address Redacted
```

```
14967125
           Theresa Lopez-Zepeda
                                     Address Redacted
14967126
           Theresa Moon
                             Address Redacted
14967127
           Theresa Richards
                               Address Redacted
           Third Step, LLC – The Fight
14967131
                                         611 S. Catalina St. #307
                                                                    Los Angeles, CA 90005
14967133
           Thomas É. Freese, PH.D.
                                      Address Redacted
14967134
           Thomas Moore
                              Address Redacted
14967135
           Thomas San Millan
                                  Address Redacted
                       Address Redacted
14967140
           Tien Vo
14967141
           Tiffany Bennett
                              Address Redacted
14967142
           Tiffany Godinez
                               Address Redacted
14967143
           Tiffany Hebets
                              Address Redacted
           Time Warner Cable–Wellness
14967145
                                          P.O. Box 60074
                                                              City of Industry, CA 91716-0074
14967146
           Timothy Buford
                               Address Redacted
14967147
           Timothy Martinez
                                Address Redacted
14967148
           Timothy Martinez, D.D.S
                                      Address Redacted
14967151
           Tiodorita Benitez
                                Address Redacted
14967156
                            Address Redacted
           Todd Brown
14967157
           Todd Woodson
                              Address Redacted
14967158
           Toma & Petros
                              3530 Camino Del Rio N Suite 109
                                                                 San Diego, CA 92108
14967159
           Tomas Carrasco
                               Address Redacted
           Tomas Esteban
14967160
                              Address Redacted
14967161
           Tonatiuh Signoret
                                Address Redacted
14967162
           Tonda McCray
                              Address Redacted
14967163
           Toni Perez
                          Address Redacted
14967164
           Tony Alberts
                            Address Redacted
14967165
           Tony Alberts
                            Address Redacted
14967166
           Tony Amores
                             Address Redacted
14967167
           Top Grill INC
                             2843 PANSY WAY
                                                    SAN JACINTO, CA 92582
                                   1983 W. 190th Street Ste 100
14967169
           Tower Energy Group
                                                                  Torrance, CA 90504
14967179
           Tracey Shepard
                              Address Redacted
14967180
           Tracie Davis
                            Address Redacted
14967181
           Tracy Lee
                          Address Redacted
14967184
           Transfirst
                         92 Groveland Park Blvd
                                                    Sound Beach, NY 11789-2507
14967186
           Transwestern Insurance Administrators
                                                  P.O. Box 45019
                                                                      Fresno, CA 93718
                         P.O. Box 660307
14967187
           Travelers
                                             Dallas, TX 75266-0307
14967188
           Travelers Casualty & Surety Co of America
                                                      One Tower Square
                                                                            Hartford, CT 06183
14967189
           Travis Lyon
                           Address Redacted
14967190
           Trend Micro
                            225 East John Carpenter Fwy, Ste 1500
                                                                   Irving, TX 75062
14967191
           Treniese Gill
                            Address Redacted
14967192
           Trevor J Boreham
                                Address Redacted
14967195
           TriCare Program
                               7700 Arlington Boulevard, Suite 501
                                                                     Falls Church, VA 22042-5101
                                                       Florence, SC 29502-2100
14967196
           Tricare West Region
                                  P.O. Box 202100
14967198
           Trisha Mork
                           Address Redacted
14967200
           Triwest-CA - Tricare - United HealthCare
                                                      P.O. Box 202112
                                                                           Florence, SC 29502-2112
14967203
           Troy Kurz
                          Address Redacted
14967204
           Troy Quibuyen
                              Address Redacted
14967207
           TruEvolution – Ryan White
                                        4175 Brockton Avenue
                                                                  Riverside, CA 92501
14967205
           Truepill
                        3121 Diablo Ave
                                            Hayward, CA 94545
14967217
           Tulin Sinjar
                           Address Redacted
           Turning Point Crisis Ctr-Community Rsrch
                                                      1738 South Tremont Street
14967218
                                                                                   Julian Medical
                     Oceanside, CA 92054
14967222
           U S DEPARTMENT OF EDUCATION
                                                   P.O. Box 105081
                                                                        NATIONAL PAYMENT
           CENTER
                         ATLANTA, GA 30348-5081
14967223
           U S DeptOF EDUCATION-GUADALUPE IBANEZ
                                                               P.O. Box 790356
                                                                                    ST LOUIS, MO
           63179-0356
14967224
           U S DeptOF EDUCATION-JOSE ARIEL VEGA
                                                            P.O. Box 790356
                                                                                ST LOUIS, MO 63179-0356
           U-First Dental Care
                                  9130 Foothill Blvd
                                                        Rancho Cucamonga, CA 91730
14967243
14967225
           U.S DEPARTMENT OF COMMERCE
                                                  1401 Constitution Ave NW
                                                                                Washington, DC 20230
           U.S HEALTHWORKS MEDICAL GROUP PC
                                                          P.O. Box 50042
14967226
                                                                             Escondido South Medical
                      Los Angeles, CA 90074
                                                  P.O. BOX 790448
14967227
           U.S. BANK EQUIPMENT FINANCE
                                                                       ST. LOUIS, MO 63179-0448
14967228
           U.S. Behavioral Health Plan
                                        425 Market St, 14th Fl
                                                                  San Francisco, CA 94105
14967229
           U.S. DEPARTMENT OF LABOR - DFVC
                                                      P.O. BOX 6200-35
                                                                             PORTLAND, OR 97228–6200
14967230
           U.S. DEPT OF TREASURY-MCKIEL, CHRISTINE
                                                               P.O. Box 979101
                                                                                    DEBT MANAGEMENT
                           ST. LOUIS, MO 63197-9000
           SERVICES
           U.S. Postal Service –BS
14967231
                                     2599 Country Club Rd
                                                               Borrego Springs, CA 92004
14967232
                      P.O. BOX 741581
           UBER
                                            Los Ángeles, CA 90074–1581
                                                                     NEW YORK, NY 10017
14967234
           UBIQUITI INC
                              685 THIRD AVENUE, 27TH FLOOR
           UC Berkeley Optometric Eye Center
14967235
                                                200 Minor Addition
                                                                       Berkeley, CA 94720-2020
14967236
                         One Shields Avenue
                                                Davis, CA 95616
           UC Davis
14967237
                             110 WESTWOOD PLAZA, STE. C305
           UC REGENTS
                                                                      UCLA/J&J HCEP AT UCLA ANDERSON
                         Los Angeles, CA 90095-1481
           SCHOOL
                                   9300 Campus Point Drive, M.C. 7433
14967238
           UC San Diego Health
                                                                          La Jolla, CA 92037
           UCLA & the CA Primary Care Assoc (CPCA)
14967239
                                                        1231 I Street, Suite 400
                                                                                  Sacramento, CA 95814
14967241
           UCR Health
                           900 University Avenue
                                                     Riverside, CA 92521
           UCSD MEDICAL CENTER
14967242
                                         Lockbox 556631
                                                              Los Angeles, CA 90074
14967244
           ULINE
                       12575 ULINE DRIVE
                                                Pleasant Prairie, WI 53158
14967246
           ULISIS VALDIOSERA
                                     Address Redacted
```

```
14967248
           UNIDOS POR LA MUSICA
                                           1520 N Mountain Ave STE 138
                                                                           Ontario, CA 91762
14967249
           UNIFIRST CORPORATION
                                                                      Ontario, CA 91761
                                           700 South Etiwanda Ave.
                                                         Chicago, IL 60606
14967251
           UNITED AIRLINES
                                   233 S. Wacker Drive
           UNITED HEALTHCARE GROUP RECOVERY SVCS
14967255
                                                                 9900 Bren Rd E
                                                                                      Minnetonka, MN 55343
14967258
           UNITED HEALTHCARE-COMMUNITY HEALTH PLAN
                                                                      4365 Executive Drive, Suite 500
                                                                                                         San Diego,
           CA 92121
14967260
           UNITED RENTALS
                                   36025 CATHEDRAL CANYON DR
                                                                          CATHEDRAL CITY, CA 92234-7262
           UNITED STATES POSTAL SERVICE
14967262
                                                   11251 Rancho Carmen Dr.
                                                                                 San Diego, CA 92199–9602
           UNITED STATES TREASURY P.O. Box 942878 Sacramento, CA 94278–3340 UNITED STATES TREASURY -CHARLES P FRALEY P.O. Box 145566, STOP 813
14967263
                                                                   P.O. Box 145566, STOP 813G
14967264
                          CINCINNATI, OH 45250-5566
           SUPPORT
           UNIVERSAL PREMIUM FLEET CARD
                                                     2000 Purchase Street
14967267
                                                                              Purchase, NY 10577
           UNIVERSAL SERVICE ADMINISTRATIVE CO
14967268
                                                             700 12th Street, NW, Suite
                    USAC
                                Washington, DC 20005
                              DEPT CH 17699
14967279
           UPDOX, LLC.
                                                   Palatine, IL 60055-7699
14967280
           UPS
                    P.O. Box 894820
                                         Los Angeles, CA 90189-4820
           UPWORK
14967281
                          2625 AUGUSTINE DR SUITE 601
                                                                Santa Clara, CA 95054
                      P.O. Box 79758
                                       Baltimore, MD 21279-0758
14967282
           URAC
           US Attorney for Southern District of CA
                                                     880 Front Street, Room 6293
14967285
                                                                                    Federal Office Building
                                                                                                               San
           Diego, CA 92101
           US BEHAVIORAL HEALTH
14967286
                                            11000 Optum Circle
                                                                    OPTUM HEALTH-Public Sector
                                                                                                       Eden Prairie,
           MN 55344
14967287
           US BEHAVIORAL HEALTH-OPTUM HEALTH
                                                              11000 Optum Circle
                                                                                      PRIVATE SECTOR
                                                                                                              Eden
           Prairie, MN 55344
14967288
           US DEPARTMENT OF EDUCATION-RAFAEL MAGANA
                                                                       P.O. Box 105081
                                                                                             NATIONAL PAYMENT
           CENTER
                         ATLANTA, GA 30348-5081
14967289
           US DeptOF EDUCATION-ALVARO DELGADILLO
                                                                 P.O. Box 105081
                                                                                      NATIONAL PAYMENT
           CENTÈR
                         ATLANTA, GA 30348-5081
           US DeptOF EDUCATION-NEWELL, ELIZABETH
14967290
                                                                P.O. Box 105081
                                                                                     ELIZABETH
           NEWÊLL
                          ATLANTA, GA 30348-5081
                          131 M Street, NE
14967291
           US EEOC
                                               Washington, DC 20507
           US FOODS, INC. – DHSWRC
14967292
                                            P.O. Box 100131
                                                                  Pasadena, CA 91189–0131
           US FOODSERVICES INC, AL
14967294
                                                                  Pasadena, CA 91189-0131
                                            P.O. Box 100131
                                       P.O. Box 100131
14967293
           US Foodservice Inc..Child
                                                             Pasadena, CA 91189-0131
14967295
           US MARSHALS – HUMBERTO A BARBOZA
                                                            350 W 1ST STREET, SUITE 3001
                                                                                                LOS ANGELES, CA
           90012
                                                             Thermal, CA 92274-4998
                                      87200 Airport Blvd
14967296
           US POSTAL SERVICE
14967297
                                                                      Anza, CA 92539-9998
           US POSTAL SERVICE ANZA
                                            P.Ô. Box Fee Payment
14967298
           USA COMMUNICATIONS
                                          P.O. Box 1057
                                                            Kearney, NE 68848
14967299
           USA COMMUNICATIONS - 2401 STIRRUP ROAD
                                                                 920 E 56th St Ste B
                                                                                        Kearney, NE 68847
                                                     920 E 56th Street Ste. B
14967300
           USA Communications-Antillies Wireless
                                                                                Kearney, NE 68847
           USA LIVE BEE REMOVAL 965 N ALLEN ST
USA MOBILITY WIRELESS, INC / SPOK P.O.
USA MULTICULTURAL P.O. Box 461801
14967301
                                                                 Banning, CA 92220
14967302
                                                       P.O. Box 660324
                                                                             Dallas, TX 75266-0324
14967303
                                                            Escondido, CA 92046
                                   P.O. BOX 1817
14967304
           USA PROTECTION
                                                       Hemet, CA 92546
                          9111 Cypress Waters Blvd
1515 3rd Street Lega
                                                       Ste 300
14967305
           UWORLD
                                                                    Dallas, TX 75019
14967233
           Uber Health
                                              Legal Department
                                                                     San Francisco, CA 94158
14967245
           Ulises Rodriguez
                                Address Redacted
                                          700 E. Inland Empire Blvd, Suite 250
14967247
           Ultrasound Institute Medical
                                                                                 Ontario, CA 91764
14967252
           United Care Medical Group
                                         600 City Parkway West
                                                                   Ste 400
                                                                                 Orange, CA 92868
                                                      Atlanta, GA 30374
14967253
           United Healthcare
                                 P.O. Box 740804
14967254
           United Healthcare Attn TRACR Unit
                                                 P.O. Box 740804
                                                                       Atlanta, GA 30374
                                                  P.O. Box 101760
14967257
           United Healthcare Recovery Services
                                                                       Atlanta, GA 30392-1760
           United Healthcare of California
United Methodist Church of Palm Springs P.O. Box 2007

200 Constitution Ave., NW
           United Healthcare of California 5701 Katella Ave, MS CA120.–0500 Cypress, CA 90630–5006 United Methodist Church of Palm Springs P.O. Box 2007 Palm Springs, CA 92263
14967256
14967259
           United States Department of Labor
Building Washington, DC 20210
14967261
                                                                             Frances Perkins
15009338
           United States Trustees Office
                                          880 Front Street, Third Floor
                                                                          Suite 3230,
                                                                                          San Diego, CA 92101
14967250
           United ag
                         54 Corporate Park
                                               Irvine, CA 92606
                                9900 Bren Road East
14967265
           UnitedHealthcare
                                                        Minnetonka, MN 55343
14967266
           Universal Care
                              5455 Garden Grove Blvd, Suite 500 Westminster, CA 92683
14967269
           University of California Regents
                                             1111 Franklin St.,12th floor
                                                                            Offc Sec & Chief of Staff for
                       Oakland, CA 94607
           Regents
14967270
           University of California Riverside
                                               900 University Avenue
                                                                         Riverside, CA 92521
           University of California San Francisco
                                                   P.O. Box 0986, MU 249E
14967271
                                                                               Legal Department
                                                                                                     San Francisco.
           CA 94143
           University of California, San Diego
                                                9500 Gilman Drive
14967272
                                                                       La Jolla, CA 92093-0021
           University of Phoenix 4035 S. Riverpoint Parkway
                                                                   Phoenix, AZ 85040
14967273
           University of San Diego
14967274
                                    5998 Alcala Park San Diego, CA 92110
14967275
           University of Southern California
                                              3551 Trousdale Parkway, ADM 352
                                                                                    Office of the General
                       Los Angeles, CA 90089–5013
           Counsel
           University of St. Augustine
                                         700 Windy Point Drive
14967276
                                                                   San Marcos, CA 92069
14967277
           Unneetha Pruitt
                              Address Redacted
14967278
           Uoc Nguyen
                            Address Redacted
14967283
           Uri Guefen
                           Address Redacted
14967284
           Uriarte, Maricela DDS – IRS Levy
                                               Address Redacted
14967310
           VALENTIN SIGHIREANU
                                          Address Redacted
           VALENZ HEALTH CARE
                                         23048 N. 15th Ave
14967311
                                                                Phoenix, AZ 85027
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 72 of 76

```
14967324
            VALLEY GLASS & MIRROR
                                               170 N Juanita Street
                                                                       Hemet, CA 92543
            VALLEY LOCK & SAFE
                                                                          Cathedral City, CA 92234
14967325
                                          68-100 Ramon Road #C-11
14967326
            VALLEY RADIOLOGY CONSULTANTS MED GROUP
                                                                        613 W Vally Parkway, Ste 330
                                                                                                          Escondido, CA
            92123
14967347
            VANIMAN MANUFACTURING CO.
                                                      25799 Jefferson Ave
                                                                               Murrieta, CA 92562
                                                                 Mesquite, TX 75150
14967350
                                          2330 Insterstate 30
            VAR Technology Finance
14967351
            VASQUEZ & COMPANY LLP
                                               655 N. Central Avenue, Suite 1550
HIGH POINT, NC 27260
                                                                                     Glendale, CA 91203
14967352
                         1515 W. GREEN DR
            VAULT
14967353
            VAULT VERIFY
                                  1597 INDUSTRIAL DRIVE
                                                                  New Smyrna Beach, FL 32168
            VERICOR, LLC
                                                           Holmen, WI 54636-9787
                                 703 Western Avenue
14967357
14967358
            VERIFENT
                             1312 17th Street, Suite 536
                                                            Denver, CO 80202
14967359
            VERIFYFAST
                                498 Seventh Avenue 12th Floor
                                                                   New York, NY 10018
14967360
            VERIFYTODAY.COM
                                        3 Sugar Creek Blvd
                                                                Sugar Land, TX 77479
14967361
                                   P.O. Box 1171130
                                                          Atlanta, GA 30368-7113
            VERITYSTREAM
            VERIZON WIRELESS - 00006
14967363
                                               P.O. Box 660108
                                                                     Dallas, TX 75266-0108
14967364
            VERIZONWIRELESS.COM - 969902732-00007
                                                                1095 Avenue of the Americas
                                                                                                  New York, NY
            10036
            VERTICAL VISIONS, INC
VES FIRE ** OFFICE OF THE OFFI
14967367
14967387
                                           74000 Dinah Shore Dr. Ste 414
                                                                               Palm Desert, CA 92211
14967388
            VFS FIRE & SECURITY SERVICES
                                                     501 West Southern Ave
                                                                                 Orange, CA 92865
14967392
            VICTOR CARBORIN
                                       Address Redacted
14967403
            VICTORIA YOUNG
                                     Address Redacted
14967404
            VIDEO CONFERENCE GEAR
                                               109 Inverness Drive East, Suite C
                                                                                    Englewood, CO 80112
14967405
            VILLAGE CARPETS, INC
                                           8745 N Magnolia Ave Ste A
                                                                           Santee, CA 92071
14967408
            VIP RV SERVICE, LLC
                                         27451 Ethanac Road
                                                                  Menifee, CA 92585
14967416
            VISION Y COMPROMISO
                                           P.O. BOX 708
                                                               SAN LORENZO, CA 94580
14967418
            VISTA PAINT
                                2020 E. Orangethorpe Ave. Suite 210
                                                                        Fullerton, CA 92831
14967421
            VITAMIN D PUBLIC RELATIONS, LLC
                                                         631 Dell Street
                                                                             Solana Beach, CA 92075
14967422
            VITOR CORAL
                                 Address Redacted
14967429
            VORTEX INDUSTRIES, INC
                                              1801 Olympic Blvd.
                                                                       Pasadena, CA 91199-1095
14967306
            Vacek Lubas
                             Address Redacted
14967307
                                    Address Redacted
            Vache Chakmakian
14967308
            Vaibhavi Bharadwai
                                     Address Redacted
14967309
            Val Verde Unified School District
                                                 975 West Morgan St
                                                                          Perris, CA 92571
14967312
            Valeria Alvarado Perez
                                       Address Redacted
14967313
            Valeria Mendez
                                Address Redacted
14967314
            Valeria Quiroz
                                Address Redacted
14967315
            Valeria Sanchez
                                 Address Redacted
14967316
            Valerie DAlessio
                                  Address Redacted
14967317
            Valerie Malkiewicz
                                    Address Redacted
14967318
            Valerie Noblado
                                 Address Redacted
14967319
                             Address Redacted
            Valerie Pina
14967320
            Valerie Roybal
                                Address Redacted
14967321
            Valerie Tapia
                              Address Redacted
14967322
            Valerio Iovino
                               Address Redacted
14967323
            Valley Auto -Anza
                                    58581 CA-371
                                                        Anza, CA 92539
14967327
            Vanesa Maldonado
                                    Address Redacted
14967328
            Vanessa Alvarado
                                  Address Redacted
            Vanessa Alvares
14967329
                                 Address Redacted
            Vanessa Bermudez
14967330
                                   Address Redacted
14967331
            Vanessa Briones
                                 Address Redacted
14967332
            Vanessa Calderon
                                  Address Redacted
14967333
            Vanessa Casillas
                                 Address Redacted
14967334
            Vanessa Coronel
                                 Address Redacted
14967335
            Vanessa Espinoza Garcia
                                         Address Redacted
14967336
            Vanessa Esquer
                                Address Redacted
            Vanessa Gonzalez
14967337
                                   Address Redacted
14967338
            Vanessa Medrano
                                  Address Redacted
14967339
            Vanessa Munoz
                                Address Redacted
14967340
            Vanessa Ordas
                                Address Redacted
14967341
            Vanessa Perilli
                                Address Redacted
14967342
            Vanessa Ramirez
                                  Address Redacted
14967343
            Vanessa Ruiz
                              Address Redacted
14967344
            Vanessa Steele
                                Address Redacted
14967345
            Vanessa Vargas
                                Address Redacted
14967346
            Vania Carreon
                                Address Redacted
14967348
            Vannessa Hernandez
                                     Address Redacted
14967349
            Vantage Medical Group
                                        215 Compton Ave
                                                               Corona, CA 92881
14967354
            Velasquez DDS Professional Dental Corp.
                                                        9130 Foothill Blvd
                                                                                Rancho Cucamonga, CA 91730
14967355
                               Address Redacted
            Venetia Lopez
            Ventegra, Înc.
14967356
                               450 North Brand Blvd, Suite 600
                                                                   Glendale, CA 91203
14967362
            Verizon Wireless
                                  1095 Avenue of the Americas
                                                                   New York, NY 10036
14967365
            Veronica Ariana Solis
                                      Address Redacted
            Veronica Ascencio
14967366
                                   Address Redacted
14967368
            Veronica Carrillo
                                  Address Redacted
                                  Address Redacted
14967369
            Veronica Cedillo
14967370
            Veronica De La Rosa
                                     Address Redacted
14967371
                                Address Redacted
            Veronica Flores
```

```
14967372
           Veronica Garcia
                               Address Redacted
                                       Address Redacted
14967373
           Veronica Garcia-Campos
14967374
           Veronica Gomez
                                Address Redacted
14967375
                                 Address Redacted
           Veronica Kennedy
14967376
           Veronica Lavados
                                 Address Redacted
14967377
           Veronica Lopez
                               Address Redacted
14967378
           Veronica Meza Figueroa
                                      Address Redacted
           Veronica Navarro
14967379
                                 Address Redacted
14967380
           Veronica Quiroz
                                Address Redacted
           Veronica Robles
14967381
                               Address Redacted
14967382
           Veronica Rodriguez
                                   Address Redacted
14967383
           Veronica Rodriguez Torres
                                         Address Redacted
14967384
           Veronica Sanchez
                                 Address Redacted
14967385
           Veronica Vaca
                              Address Redacted
14967386
           Veronica Villafana
                                  Address Redacted
14967389
           Vicente Serrano
                               Address Redacted
14967390
                              Address Redacted
           Victor Barrera
14967391
           Victor Bautista Manzo
                                     Address Redacted
                             Address Redacted
14967393
           Victor Duran
14967394
           Victor Fontaine
                               Address Redacted
14967395
           Victor Leon
                            Address Redacted
14967396
           Victor Madero
                              Address Redacted
14967397
           Victor Reyes-Mendoza
                                      Address Redacted
14967398
           Victor Sebring
                              Address Redacted
14967399
           Victoria Gibbs
                              Address Redacted
14967400
           Victoria Malkhasyan
                                   Address Redacted
14967401
           Victoria Rizzotto
                                Address Redacted
14967402
           Victoria Sanchez
                                Address Redacted
14967406
           Vincent OBrien
                               Address Redacted
14967407
           Violet Valenzuela
                                 Address Redacted
14967409
                                                907 Floyd Ave
           Virginia Commonwealth University
                                                                   Richmond, VA 23284
14967410
           Virginia De La Cruz
                                   Address Redacted
14967411
           Virginia Fernandez
                                  Address Redacted
14967412
           Virginia Houston
                                Address Redacted
           Virginia Mulato-Bacilio
                                      Address Redacted
14967413
           Virginia Saldivar
14967414
                                Address Redacted
14967415
           Viridiana Salazar
                                Address Redacted
14967417
           Vista Hill
                         1012 Main Street, Suite 101
                                                        dba SmartCare Integrated Behave Health
                                                                                                 Ramona, CA
           92065
                                                                 Vista, CA 92083
           Vista Village Pediatrics
                                     950 Civic Center Dr. #A
14967419
14967420
           Vitamin D Public Relations, LLC
                                              5900 Balcones Dr. Suite 100
                                                                             Denise Gitsham
                                                                                                 Austin, TX
           78731-4298
14967423
           Vivanda Naoom
                               Address Redacted
14967424
           Vivian Garcia
                             Address Redacted
14967425
           Viviana Martinez
                                Address Redacted
14967426
           Viviana Portillo
                               Address Redacted
14967427
           Viviana Sanson
                               Address Redacted
14967428
                        3401 Hillview Ave
           Vmware
                                               Palo Alto, CA 94304
                      20 Washington Avenue South, 4th Floor
                                                              Minneapolis, MN 55401
14967430
                                                401 West A Street, Suite 1900
14967431
           W.A. Stephan, a Dental Corporation
                                                                                 Wilson Elser–Robert W.
           Harrison
                        San Diego, CA 92101
14967432
           W.A. Stephan, a Dental Corporation
                                                4025 Avati Drive
                                                                      San Diego, CA 92117
                                         51607 29 PALMS HWY
           WAGNER RV SERVICES
14967434
                                                                    MORONGO VALLEY, CA 92256
14967435
           WAJDIN DABOO
                                  Address Redacted
                            702 S.W. 8th St
14967439
           WALMART
                                                Bentonville, AR 72716
           WALT DISNEY RESORT
14967442
                                         2901 OSCEOLA PARKWAY
                                                                         Orlando, FL 32821
                                   702 S.W. 8th St
14967447
           WALTMART.COM
                                                       Bentonville, AR 72716
                                                              Washington, DC 20005
14967450
           WASHINGTON PLAZA
                                       10 Thomas Cir NW
14967451
           WASHINGTON SQUARE SHOPPING CENTER LLC
                                                                 P.O. Box 445
                                                                                   Rancho Santa Fe, CA 92067
14967453
           WASTE MANAGEMENT - 17-25770-03009
                                                          P.O. Box 43530
                                                                              Phoenix, AZ 85080
                                                                     Los Angeles, CA 90054-1008
14967454
           WASTE MANAGEMENT - Nuevo
                                                P.O. Box 541065
14967455
           WASTE MANAGEMENT 330-0020141-1584-0
                                                             P.O. Box 541008
                                                                                  Los Angeles, CA 90054-1008
           WASTE MANAGEMENT 8-60990-55008
14967456
                                                       P.O. Box 541008
                                                                            Los Angeles, CA 90054-1008
14967457
           WASTE MANAGEMENT 8-74946-05005
                                                       P.O. Box 541008
                                                                            Los Angeles, CA 90054-1008
14967458
           WAYFAIR
                           4 Copley Place, 7th Floor
                                                       Boston, MA 02116
                       2275 E Cooley Dr
                                                                                    Cooley, CA 92324
14967459
           WCHS
                                             dba Colton Comprehensive Treatment
14967460
           WE KLEAN INC.
                                 427 S Citrus Ave
                                                      Escondido, CA 92027
           WEBB, SHANNON
                                   3323 Niblick Dr.
                                                        La Mesa, CA 91941
14967462
           WEBER & ASSOCIATES PLANNING & DESIGN LLC
14967463
                                                                   2636 Sycamore Dr
                                                                                          Oceanside, CA 92056
14967464
           WEBIPLEX, INC
                                 9 Corporate Park, Suite 240
                                                               Irvine, CA 92606
14967466
           WEITZMAN INSTITUTE LEARNING ACADEMY
                                                               4600 Campus Drive Suite 110
                                                                                                Newport Beach, CA
           92660
14967467
           WELCH, CHERYL
                                  Address Redacted
14967478
           WEPAY
                         50 Convention Way #200
                                                     Redwood City, CA 94063
14967480
           WEST COAST FIRE & INTEGRATION
                                                     P.O. Box 4010
                                                                        Castaic, CA 91310
14967481
                                      14465 Griffith St
           WEST COAST PARTS
                                                           San Leandro, CA 94577
14967482
           WEST COAST RESOLUTION GROUP
                                                     530 B Street, Suite 1700
                                                                               San Diego, CA 92101
```

```
14967483
           WEST INTERACTIVE SERVICES CORPORATION
                                                                  Dept # 1343
                                                                                   Denver, CO 80256-0001
14967487
           WESTGATE RESORT
                                      5601 Windhover Dr.
                                                               Orlando, FL 32819
14967488
           WEX BANK
                             P.O. Box 4337
                                                 Carol Stream, IL 60197-4337
14967489
           WHAT TO EXPECT ORG
                                          P.O. Box 58277
                                                              2512 Virginia Ave NW
                                                                                         Washington, DC 20037
14967492
           WHOVA, INC.
                                7310 Miramar Road, Suite 200
                                                                 San Diego, CA 92126
           WIFIBY CV
14967494
                             231 Oasis Road
                                                 Palm Springs, CA 92262
14967507
           WILLIAMS SCOTSMAN, INC
                                              P.O. Box 91975
                                                                  Chicago, IL 60693-1975
           WIPA Medical Group
                                     P.O. Box 7014
14967509
                                                        Lancaster, CA 93539
14967511
                      7095 Hollywood Blvd, Ste 788
                                                        Los Angeles, CA 90028-8912
14967512
           WOLTERS KLUWER HEALTH / UPTODATE, INC.
                                                                  P.O. Box 412094
                                                                                        Boston, MA 02241-2094
14967514
           WORKDAY, INC.
                                  6110 Stoneridge Mall Road
                                                                 Pleasanton, CA 94588
14967515
            WORLD CONFERENCE FORUM
                                                 1 Orient Way Suite F170
                                                                              Rutherford, NJ 07070
14967517
           WPATH
                         1061 E. Main St. Ste 300
                                                      East Dundee, IL 60118
14967433
                              Address Redacted
            Wafaa Shateh
                                       Address Redacted
14967436
           Waleed Stephan, D.D.S
15000569
           Waleed Stephan, D.D.S.
                                       W.A. Stephan, A Dental Corporation
                                                                              Marcelo G. Toledo, D.D.S.
                                                                                                             Marcelo
           Toledo D.D.S., Inc.
14967437
           Waleed Stephan, DDS
                                     401 West A Street, Suite 1900
                                                                       Wilson Elser-Robert W. Harrison
                                                                                                           San Diego,
           CA 92101
14967438
            Walgreens Pharmacy
                                    200 Wilmot Road
                                                          MS #2002
                                                                          Deerfield, IL 60015
14967440
           Walmart - Contract Pharmacy
                                             702\;SW\;8th\;St
                                                                Bentonville, AR 72712
            Walmart Pharmacy
14967441
                                   702 S.W. 8th St
                                                       Bentonville, AR 72716
14967443
           Walter Druckenmiller
                                     Address Redacted
14967444
            Walter Gonzalez
                                Address Redacted
           Walter Reyes
14967445
                              Address Redacted
14967446
           Walter Snow
                             Address Redacted
14967448
           Waseem Esho
                              Address Redacted
14967449
           Washburn University
                                     1700 SW College Ave.
                                                               Topeka, KS 66621
14967452
           Waste Management
                                   800 Capitol Street, Suite 3000
                                                                    Houston, TX 77002
                              427 S. Citrus Avenue
14967461
           We Klean Inc.
                                                        Aracelis Gutierrez
                                                                              Escondido, CA 92027
14967465
           Wegdan Hanna Haddad
                                       Address Redacted
14967468
                            20800 SW 115th Avenue, Suite 100
                                                                  Tualatin, OR 97062
           Wellpartner
14967469
           Wellpoint Military Care Corporation
                                                  220 Virginia Ave
                                                                        Indianapolis, IN 46204
15034835
           Wells Fargo Auto
                                 PO Box 130000
                                                      Raleigh, NC 27605
                                                      Raleigh, NC 27605
15034836
           Wells Fargo Auto
                                 PO Box 130000
                                                            PO Box 130000
14968677
           Wells Fargo Bank N.A., d/b/a Wells Fargo Auto
                                                                                 Raleigh NC
                                                                                                  27605
14968672
           Wells Fargo Bank N.A., d/b/a Wells Fargo Auto
                                                            PO Box 130000
                                                                                 Raleigh, NC
                                                                                                  27605
14967470
           Wells Fargo Dealer Services
                                           P.O. Box 51963
                                                               Los Angeles, CA 90051–6263
14967471
            Wendi Velasco
                               Address Redacted
14967472
           Wendolyn Campos Perez
                                        Address Redacted
14967473
           Wendy Beltran
                               Address Redacted
           Wendy Campos
Wendy Campos
14967474
                                Address Redacted
14967475
                                Address Redacted
14967476
           Wendy Magana
                                Address Redacted
           Wendy Zamora Villalbazo
14967477
                                         Address Redacted
14967479
           Wesley Detwiler
                                Address Redacted
14967484
           Western Pharmacy Group
                                        286 Euclid Ave Ste 206
                                                                    San Diego, CA 92114–3612
14967485
           Western University - MOU
                                          309 E. Second St
                                                                Pomona, CA 91766-1854
14967486
           Western University of Health Sciences
                                                    309 E. Second Street
                                                                             Pomona, CA 91766
14967490
           White Cross Pharmacy
                                      1717 E. Vista Chino, Suite B2
                                                                       Paml Springs, CA 92262
14967491
            White Smile Dental
                                   3495 East Concours Street, Suite A
                                                                        Corona, CA 91764
14967493
           Wiam Badr
                            Address Redacted
14967495
           Wilber Guzman
                                Address Redacted
           Will Lightbourne-CA Dept of Healthcare 95899-7413
14967496
                                                       P.O. Box 997413
                                                                            MS 0000
                                                                                          Sacramento, CA
14967497
           Willemine Hennessy
                                    Address Redacted
14967498
                                  Address Redacted
           William Cummins
14967499
            William D Clark
                                Address Redacted
14967500
           William Daniel
                               Address Redacted
14967501
            William Mackin
                                Address Redacted
14967502
           William McKenzie
                                   Address Redacted
           William P. Fennell, Esq. L
930 San Diego, CA 92101
15026218
                                       Law Office of William P. Fennell, APLC
                                                                                  600 West Broadway, Suite
14967503
           William Painter
                                Address Redacted
           William Ramos
William Ward
14967504
                                Address Redacted
                              Address Redacted
14967505
14967506
           William Welte
                               Address Redacted
14967508
           Windy Nelson
                              Address Redacted
14967510
           Wipfli LLP
                            P.O. Box 3160
                                               Milwaukee, WI 53201-3160
14967513
           Womens Health & Wellness Center-La Mesa
                                                          8851 Center Dr Suite 210
                                                                                       La Mesa, CA 91942
14967516
           World Prof Assoc for Transgender Health
                                                       1061 E Main St Ste 300
                                                                                  East Dundee, IL 60118
14967518
                             Address Redacted
           Ximena Oros
14967519
           Xochiquetlzal Castaneda
                                        Address Redacted
14967520
           Xochitl Aispuro
                                Address Redacted
            YALE CHÂSE EQUIPMENT AND SERVICES, INC.
14967525
                                                                  P.O. Box 848905
                                                                                        Los Angeles, CA
           90084-8905
14967536
            YASHWANT CHAUDRHRI, M.D. A PROF. CORP
                                                                 Address Redacted
           YASHWANT CHAUDRHRI, M.D.A Prof Corp
15021350
                                                             4537 College Ave
                                                                                   San Dioego, CA 92115
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 75 of 76

```
14967542
           YC Power Systems
                                  8089 Cherry Avenue
                                                          Fontana, CA 92336
                                                 Dallas, TX 75320-4393
                            P.O. Box 204393
14967543
           YELP, INC.
14967556
           YEXT, INC
                            P.O. Box 9509
                                               New York, NY 10087-9509
14967557
           YI PIN CHENG
                                Address Redacted
14967564
           YOLANDA SALVADOR
                                        Address Redacted
14967574
           YRC FREIGHT
                                P.O. Box 100129
                                                    Pasadena, CA 91189-0003
14967521
           Yadira Leal Galvez
                                  Address Redacted
14967522
                              Address Redacted
           Yadira Ortega
14967523
           Yadira Rodriguez
                                 Address Redacted
14967524
           Yadira Sierra
                             Address Redacted
14967526
           Yamell Espinoza
                                Address Redacted
                                 Address Redacted
14967527
           Yaneli Marroquin
14967528
           Yanelly Garnica
                                Address Redacted
14967529
           Yanet Velarde
                              Address Redacted
14967530
           Yanin Lopez Silva
                                  Address Redacted
14967531
           Yanis Hernandez
                                Address Redacted
14967532
           Yao Zhang
                           Address Redacted
14967533
           Yarabhi Arce
                             Address Redacted
14967534
           Yarem Callahan
                                Address Redacted
14967535
           Yaser Taleb
                            Address Redacted
14967537
           Yasmin Madrigal
                                 Address Redacted
14967538
           Yasmin Ruelas Sanchez
                                      Address Redacted
14967539
                               Address Redacted
           Yasmine Garcia
14967540
           Yasser Al Ibraheem
                                  Address Redacted
14967541
           Yassine Dhiri
                             Address Redacted
14967544
           Yeni Lopez
                           Address Redacted
14967545
           Yesenia Aguilar
                                Address Redacted
14967546
           Yesenia Gonzalez
                                 Address Redacted
14967547
           Yesenia Oropeza
                                Address Redacted
14967548
           Yesenia Perez
                              Address Redacted
14967549
           Yesenia Ruedas
                               Address Redacted
14967550
           Yesenia Uroza
                              Address Redacted
14967551
           Yessenia Cintron
                                Address Redacted
14967552
                                   Address Redacted
           Yessenia Fernandez
14967553
           Yessenia Ouitiquit
                                  Address Redacted
14967554
           Yessica Ortega Ortega
                                     Address Redacted
14967555
           Yessica Ramirez Vidal
                                     Address Redacted
14967558
           Yolanda Andrade
                                 Address Redacted
14967559
           Yolanda Espinoza
                                 Address Redacted
14967560
           Yolanda Gonzalez
                                 Address Redacted
14967561
           Yolanda Gutierrez
                                 Address Redacted
                             Address Redacted
14967562
           Yolanda Lira
14967563
           Yolanda Martinez
                                 Address Redacted
14967565
           Yolanda Torres
                               Address Redacted
14967566
           Yolanda Yahuaca
                                 Address Redacted
14967567
           Yolie Pacheco
                              Address Redacted
14967568
           Yomalie Maganda
                                 Address Redacted
14967569
           Yosa Sanchez Rabago
                                     Address Redacted
14967570
           Yosef Saenz
                            Address Redacted
           Yoselin Magdaleno
14967571
                                  Address Redacted
14967572
           Yosr Moussa
                             Address Redacted
14967573
           Yovanka Mendoza
                                  Address Redacted
14967575
           Yukio Cardenas-Garcia
                                      Address Redacted
14967576
           Yuliana Verdeja
                               Address Redacted
14967577
           Yulissa Cruz Čontreras
                                      Address Redacted
14967578
           Yung Leung
                            Address Redacted
           Yuriana Leon Gutierrez
14967579
                                      Address Redacted
14967580
           Yusra Tawfeeq
                              Address Redacted
14967581
           Yvette Manriquez
                                 Address Redacted
14967582
           Yvette Meza
                            Address Redacted
14967583
           Yvette Montes
                              Address Redacted
14967584
           Yvonne Ferguson
                                 Address Redacted
14967585
           Yvonne M Niman Creason
                                         Address Redacted
14967592
           ZAZZLE.COM
                               811 Sandhill Road.
                                                      Reno, NV 89521
           ZEBU COMPLIANCE SOLUTIONS
14967593
                                                  609 2nd Street
                                                                     Portsmouth, OH 45662
                                                  P.O. Box 123866, DEPT 3866
                                                                                   DBA VALENZ
14967594
           ZEBU COMPLIANCE SOLUTIONS
                              Dallas, TX 75312-3866
           ASSURANCE
                                                   STE 250
14967598
           ZENWORK, INC
                                                                Fayetteville, AR 72701
                                 1 E Center St
                                                      Washington, DC 20037
14967600
           ZERO TO THREE
                                  1255 23rd #350
14967601
           ZIEMER PLUMBING, INC
                                          P.O. Box 1016
                                                             221 N. 1st Ave
                                                                                Barstow, CA 92311
                                                             343 Soquel Avenue, Suite 409
           ZINNANTI SURGICAL DESIGN GROUP, INC.
14967603
                                                                                             Santa Cruz, CA
           95062
           ZIPPY DOGS LLC
14967604
                                  6523 CALIFORNIA AVE SW #329
                                                                        Seattle, WA 98136
           ZITO - C-BCHF
                                                    COUDERSPORT, PA 16915
14967605
                                 P.O. BOX 665
14967606
           ZITO BUSINESS
                                 P.O. Box 665
                                                   COMMERCIAL SVCS BILLING-C-BORCOM
                                                                                                    Coudersport, PA
           16915
14967607
           ZITO BUSINESS
                                 P.O. Box 665
                                                   COMMERCIAL SVCS BILLING-C-BORHTH
                                                                                                    Coudersport, PA
           16915
```

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-2 Pg. 76 of 76

14967608	ZITO BUSINESS P.O. Box 665 COMMERCIAL SVCS BILLING-C-WOLCOT Coudersport, PA 16915
14967609	ZITO MEDIA P.O. Box 431 Coudersport, PA 16915–0431
14967611	ZOGICS 309 Pittsfield Road, Suite A Lenox, MA 01240
14967613	ZOHO CORPORATION P.O. Box 894926 Los Angeles, CA 90189–4926
14967615	ZOLL MEDICAL CORPORATION P.O. Box 27028 New York, NY 10087–7028
14967616	ZOOM VIDEO COMMUNICATION, INC P.O. Box 398843 San Francisco, CA 94139–8843
14967617	ZORO TOOLS 909 Asbury Dr Buffalo Grove, IL 60089–4525
14967586	Zahir Yousif Address Redacted
14967587	Zaid Ahmed Address Redacted
14967588	Zaidoom Abdulkadhum Address Redacted
14967589	Zainab Hanna Address Redacted
14967590	Zainab Kadhm Address Redacted
14967591	Zaira Ortiz Address Redacted
14967595	Zeena Juwaideh Address Redacted
14967596	Zelis Payments 570 Carillon Pkwy, Suite 500 St. Petersbeurg, FL 33716
14967597	Zeljka Kostich Address Redacted
14967599	Zenziwe Bryant Address Redacted
14967602	Zinah Alsaffar Address Redacted
14967610	Zitzihani Maganda Address Redacted
14967612	Zohal Abdullatif Address Redacted
14967614	Zoila Alvarado Address Redacted
14967618	Zuleica Ramirez Address Redacted
14967619	Zuleima Hernandez Address Redacted
14967620	Zuleyka Baez Address Redacted
14967621	Zulma Duarte Address Redacted
14967622	Zuri Vasquez Address Redacted
14963628	e3 AUDIOMETRICS 3333 N Kennicott Ave Arlington Heights, IL 60004
14963765	ePLASTICS 5535 Ruffin Road San Diego, CA 92123
14964224	i2i Systems 377 Riverside Drive, Suite 300 Franklin, TN 37064
14964230	iCreative 41625 Eclectic Street, Suite H–1 Palm Desert, CA 92260
14964290	iRedeemHealth 55 Progress P1 Suite 201 Jackson, NJ 08527 iSTORAGE SAN BERNARDINO 1155 S Tippecanoe Ave San Bernardino, CA 92408
14964320 14967240	iSTORAGE SAN BERNARDINO 1155 S Tippecanoe Ave San Bernardino, CA 92408 uConfirm P.O. Box 1971 Woodstock, GA 30188
1490/240	ucommin F.O. Dox 1971 woodstock, GA 30100

TOTAL: 5395

Case 22-02384-LT11 Filed 01/25/24 Entered 01/25/24 11:16:50 Doc 1273-3 Pg. 1 of 1

Notice Recipients

District/Off: 0974-3 User: Admin. Date Created: 1/25/2024

Case: 22-02384-LT11 Form ID: pdfO4 Total: 5

Recipients of Notice of Electronic Filing:

Haeji.Hong@usdoj.gov Haeji Hong

Tania M. Moyron tania.moyron@dentons.com aty

TOTAL: 2

Recipients submitted to the BNC (Bankruptcy Noticing Center):

BORREGO COMMUNITY HEALTH FOUNDATION, Springs, CA 92004 587 Palm Canyon Dr. Suite 208 Borrego

880 Front Street aty David Ortiz DOJ-Ust David A. Ortiz Suite 3230 San Diego, CA 92101

Samuel Ruven Maizel Dentons US LLP 601 South Figueroa Street Suite 2500 Los Angeles, aty

CA 90017

TOTAL: 3